

17335 Golf Parkway Suite 100 Brookfield, WI 53045 USA Tel +1 262 784 2250

milliman.com

Mathieu Doucet, FSA, MAAA Senior Consulting Actuary

mathieu.doucet@milliman.com

December 9, 2022

Henry Lipman, FACHE Medicaid Director NH Department of Health and Human Services 129 Pleasant Street Concord, NH 03301 Sent via email: henry.lipman@dhhs.nh.gov

Re: Amendment #9 - SFY 2023 Medicaid Care Management Program Capitation Rates

Dear Henry:

This letter provides the New Hampshire Department of Health and Human Services (DHHS) with Amendment #9 SFY 2023 managed care organization (MCO) capitation rates for the Medicaid Care Management (MCM) program including the following populations:

- Standard Medicaid, which also includes the Children's Health Insurance Program (CHIP) population
- Granite Advantage Health Care Program (GAHCP), New Hampshire's adult Medicaid expansion program

DHHS retained Milliman to calculate, document, and certify its capitation rate development. We developed the capitation rates using the methodology described in this letter.

Our role is to certify that the SFY 2023 MCO capitation rates produced by the rating methodology are actuarially sound to comply with Centers for Medicare and Medicaid Services (CMS) regulations. We developed actuarially sound capitation rates using published guidance from the American Academy of Actuaries (AAA), the Actuarial Standards Board (ASB), CMS, and federal regulations to ensure compliance with generally accepted actuarial practices and regulatory requirements.

This letter includes an abbreviated summary of the SFY 2023 capitation rate setting methodology and assumptions for discussion with the MCOs. Please see the May 20, 2022 rate report, provided as Attachment A, for additional details of the data, assumptions, and methodology underlying the development of the Amendment #8 SFY 2023 MCM capitation rates. The rate setting guide and checklist from Attachment A also apply to the rates documented in this letter. Attachment B contains the Actuarial Certification for the Amendment #9 SFY 2023 MCM capitation rates. Attachment C contains the data reliance letter.

AMENDMENT #9 PROGRAM CHANGES

The Amendment #9 SFY 2023 capitation rates reflect several program changes from the Amendment #8 SFY 2023 capitation rates presented in our May 20, 2022 certification. The changes are as follows:

- Updated acuity adjustments by population
- Consideration for changes in hospital outpatient cost to charge ratios
- Updated enrollment scenarios reflecting potential timing for the end of the public health emergency (PHE)
- Adjustments to projected state owned hospital expenditures
- Updated applied behavior analysis (ABA) services trends
- Implementation of reimbursement increases for freestanding and hospital based birthing services
- Inclusion of non-emergency medical transportation (NEMT) reimbursement changes
- Implementation of the October 2022 DRG weights
- Implementation of remote patient monitoring services effective January 1, 2023
- Inclusion of advanced lab tests for breast and ovarian cancer and non-alcoholic steatohepatitis (NASH)
- Updated list of carved out gene therapies
- Implementation of reimbursement increases for SUD residential code H0010



SFY 2023 CAPITATION RATE CHANGE

Table 1A below shows the statewide rate change by population from the Amendment #8 SFY 2023 capitation rates to the Amendment #9 SFY 2023 rates in this letter using current SFY 2023 DHHS population estimates and excludes directed payments. Exhibit A shows the rate change by rate cell.

Table 1A New Hampshire Department of Health and Human Services Medicaid Care Management Program SFY 2023 Capitation Rate Change Based on Projected SFY 2023 Enrollment by Rate Cell Excluding Directed Payments April PHE End Date							
Population	Amendment #8	Amendment #9	Rate Change				
Standard Medicaid							
Base Population	\$349.04	\$348.81	-0.1%				
CHIP	211.15	213.50	1.1%				
Behavioral Health Population	1,459.07	1,431.84	-1.9%				
Total Standard Medicaid	\$406.64	\$404.89	-0.4%				
Granite Advantage Health Care	Granite Advantage Health Care Program						
Medically Frail	\$1,097.65	\$1,074.95	-2.1%				
Non-Medically Frail	419.51	414.34	-1.2%				
Behavioral Health Population	2,054.51	2,015.13	-1.9%				
Total GAHCP	\$564.78	\$556.50	-1.5%				
Grand Total	\$467.51	\$463.25	-0.9%				

Table 1B below shows the statewide rate change by population from the Amendment #8 SFY 2023 capitation rates to the Amendment #9 SFY 2023 rates, including directed payments.

Table 1B New Hampshire Department of Health and Human Services Medicaid Care Management Program SFY 2023 Capitation Rate Change Based on Projected SFY 2023 Enrollment by Rate Cell Including Directed Payments April PHE End Date					
Population	Amendment #8	Amendment #9	Rate Change		
Standard Medicaid					
Base Population	\$361.13	\$360.19	-0.3%		
CHIP	216.61	218.32	0.8%		
Behavioral Health Population	1,514.99	1,483.27	-2.1%		
Total Standard Medicaid	\$420.82	\$418.10	-0.6%		
Granite Advantage Health Care	Program				
Medically Frail	\$1,150.99	\$1,113.50	-3.3%		
Non-Medically Frail	432.57	423.65	-2.1%		
Behavioral Health Population	2,097.00	2,050.34	-2.2%		
Total GAHCP	\$583.49	\$570.08	-2.3%		
Grand Total	\$483.43	\$476.60	-1.4%		



The Amendment #9 SFY 2023 capitation rates include directed payments to community mental health centers (CMHCs) and critical access hospitals (both subject to CMS approval). While these amounts are included in the MCO capitation rates, MCOs are not at risk for the amount of these directed payments.

We project an overall MCO medical loss ratio (MLR) for at-risk services of 90.9% for the Standard Medicaid population and 90.2% for the GAHCP population in SFY 2023, which includes:

- An overall 7.6% administrative cost allowance for the Standard Medicaid population, and 8.3% for the GAHCP population
- A 1.5% risk margin applied as a percentage of revenue prior to the directed payments and the 2.0% premium tax allowance

The projected MLR for at-risk services excludes the impact of the directed payments and the 2.0% premium tax allowance in both the numerator and denominator of the MLR calculation, which is consistent with the treatment of directed payments and premium taxes in federal MLR calculations.

The Amendment #9 SFY 2023 MCM capitation rates do not include funding for the COVID-19 vaccines or vaccine counseling services. DHHS will continue to cover the vaccine administration and vaccine counseling costs under a non-risk arrangement for SFY 2023.

Table 2 shows a breakdown of the capitation rate changes by major rate component for the Amendment #9 SFY 2023 capitation rates compared to the Amendment #8 SFY 2023 capitation rates on a percentage and aggregate dollar basis. Please note, the various rate change components include the impact of attributed administrative allowance, risk / profit margin and premium tax beyond the service costs alone.

Table 2 New Hampshire Department of Health and Human Services						
SFY 2023 MCM Capitation Rate Change by Component ¹						
Based on Projected SFY 2023 Enrollment by Rate Cell						
	April	2023 PHE End So	cenario			
	Standa	rd Medicaid		GAHCP		Total
	Rate	12 Month	Rate	12 Month Dollar	Rate	12 Month
Rate Component	Change	Dollar Impact	Change	Impact	Change	Dollar Impact
Covered Services and Fee Schedule Chang	jes					
Remote patient monitoring services ⁴	0.0%	44,502	0.0%	14,051	0.0%	58,553
Birthing services reimbursement ³	0.4%	2,704,903	0.0%	266,587	0.2%	2,971,490
Advanced lab tests ⁴	0.0%	181,054	0.0%	74,476	0.0%	255,530
Ambulance emergency medical	0.1%	1,028,020	0.1%	973,491	0.1%	2,001,511
transportation rates ³						
Friends and family NEMT rates ⁴	0.0%	305,170	0.1%	463,202	0.1%	768,372
Chair car transportation rates ⁴	0.0%	178,968	0.0%	111,769	0.0%	290,737
State-owned hospital professional services ⁴	0.1%	624,648	0.0%	182,776	0.1%	807,424
SUD residential services ⁴	0.0%	72,426	0.1%	546,173	0.0%	618,600
Total program changes	0.7%	5,139,692	0.4%	2,632,525	0.6%	7,772,217
Other changes relative to Amendment #8 S	FY 2023 ratir	a assumptions				
SFY 2023 acuity adjustment compared to	-0.2%	(\$1,270,728)	-1.3%	(\$8,519,200)	-0.7%	(\$9,789,928)
Amendment #8 acuity assumption ³	0.270	(\$1,210,120)	1.070	(\$0,010,200)	0.170	(\$0,100,020)
October 2022 DRG weights ⁴	0.0%	158,231	0.0%	193,003	0.0%	351,234
State-owned hospital staffing constraints ³	-0.4%	(3,272,851)	-0.2%	(1,087,774)	-0.3%	(4,360,625)
Applied behavioral analysis service trends ³	-0.1%	(712,628)	0.0%	3.762	-0.1%	(708,866)
Impact of enrollment increase on the PMPM	-0.5%	(3,973,112)	-1.1%	(7,346,323)	-0.8%	(11,319,435)
value of fixed-dollar directed payments ³		(-,,)		(.,,))		(,,,)
Other ²	-0.1%	(915,576)	-0.1%	(857,612)	-0.1%	(1,773,188)
Total other changes	-1.3%	(9,986,664)	-2.7%	(17,614,143)	-2.0%	(27,600,807)
Total rate change for Amendment #9	-0.6%	(\$4,846,972)	-2.3%	(\$14,981,619)	-1.4%	(\$19,828,591)
relative to Amendment #8 rates						

¹ The various rate change components include the impact of attributed administrative allowance, risk / profit margin, and premium tax beyond the service costs alone.

² The other change component can include items, such as changes in fee-for-service fees from SFY 2022 to SFY 2023, base data updates, refreshed financial items from MCOs, and modifications to program change effective dates among other things.

³This change is retroactive to July 1, 2022. The fiscal impact represents a twelve-month period.

⁴ This change is effective after July 1, 2022. The fiscal impact represents less than a twelve-month period.



The composite capitation rates shown above are based on DHHS projections of MCO enrollment for SFY 2023, which include considerations for the unwind process following the end of the public health emergency (PHE) that DHHS expects to occur no earlier than April 2023. The enrollment projections by month are included in Exhibit F.

It should be emphasized that capitation rates are a projection of future costs for an efficient MCO based on a set of assumptions. Actual MCO costs will be dependent on each MCO's situation and the extent to which future experience conforms to the assumptions made in the capitation rate development calculations.

METHODOLOGY

The methodology used to develop the Amendment #9 SFY 2023 MCM capitation rates is consistent with the Amendment #8 capitation rates detailed in our May 20, 2022 report included in Attachment A.

The sections below provide details on all changes from the Amendment #8 SFY 2022 MCM capitation rates.

Detailed rate development exhibits are provided in Excel format in the attached "Appendices - SFY 2023 MCM Rates (12-09-22).xlsx" file.

Acuity Adjustment

In the original SFY 2023 rate certification, we included an acuity adjustment to account for the lower average acuity levels while enrollment redeterminations are voluntary during the PHE. Our intent was to retroactively calculate the final acuity adjustment based on the actual redetermination and disenrollment activity that would occur during SFY 2023. While the calculation of this adjustment would occur following SFY 2023, it would largely rely on activity (enrollment, redeterminations, and claims) that takes place during SFY 2023. As the PHE appears to be extended to mid-April, it seems likely there will not be sufficient redetermination activity to appropriately measure acuity differences between continuing and disenrolled individuals during SFY 2023.

In light of this, we are making a slight revision to the original acuity adjustment and removing the retrospective component of the adjustment. We now calculate the acuity adjustment on a prospective basis to capture observed acuity and emerging experience differences during recent periods. Specifically, we reviewed the following information to update our original acuity adjustment assumptions for Children, Adults, and Disabled Adults (we continue to set the acuity adjustment to 0.0% for Disabled Children, the Behavioral Health Population, and Dual Eligibles):

- January to June 2021 actual experience relative to estimates using projected CY 2019 data
- SFY 2022 actual experience relative to estimates using projected CY 2019 data
- Growth in the percentage of enrollees with no monthly claims from CY 2019 to SFY 2022

Since the SFY 2023 rates rely on CY 2019 base data, we use the analyses noted above to determine an appropriate adjustment factor that accounts for population acuity and emerging experience. However, we recognize that all observed differences relative to projected CY 2019 data are not driven solely by acuity. As such, we dampened the observed differences by 50%. We did not explicitly factor the growth in the percentage of enrollees with no monthly claims into the adjustment, but rather we relied on this information as additional support for our observations.

Based on our review, we updated the acuity adjustment factors as shown in Table 3 below.

Table 3 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Program Acuity Adjustments					
Eligibility Grouping	Amendment #8 Assumption	Amendment #9 Assumption			
Children ¹	-1.9%	-0.4%			
Disabled Children	0.0%	0.0%			
Adults ²	-1.9%	-3.4%			
Disabled Adults ³	-1.9%	-4.3%			
Behavioral Health Population	0.0%	0.0%			
Dual Eligibles	0.0%	0.0%			

¹ Includes Low Income Children - Age 1 to 18 Years, CHIP, and Foster Care / Adoption rate cells.

² Includes Low Income Adults and Granite Advantage - Non-Medically Frail rate cells.

³ Includes Elderly and Disabled Adults - Age 19 to 64, Elderly and Disabled Adults - Age 65+, and Granite Advantage - Medically Frail rate cells.



Exhibit D provides the details of our analysis.

We retained the 3.9% additional acuity adjustment for the Medically Frail rate cell from our May 20, 2022 rate report that reflected increased acuity during the CY 2018 and CY 2019 base period. This additional acuity adjustment is then combined with the -4.3% Disabled Adults acuity adjustment related to the end of the PHE discussed above to calculate the final -0.5% acuity adjustment for the Medially Frail rate cell.

Hospital Outpatient Cost to Charge Ratios

We reviewed the cost-to-charge ratios (CCRs) that underlie the FFS hospital outpatient payment methodology for SFY 2019 to SFY 2023 to understand how experience during the PHE may impact SFY 2023 reimbursement levels. We used the CY 2019 hospital outpatient billed charges by facility to evaluate the program-wide changes in CCRs from SFY 2019 to SFY 2023.

We observed the CCRs increased approximately 7.4% from SFY 2019 to SFY 2023 for hospitals with a CCR (i.e., excluding out-of-state hospitals or hospitals with other reimbursement structures). Included in this 7.4% increase is the 6.3% increase to hospital outpatient percentage of cost reimbursement due to the 3.1% Medicaid provider rate increases effective January 1, 2020 and January 1, 2021, as mandated by House Bill 4. In the rate development, we apply this 6.3% cost increase to all hospital outpatient costs regardless of the use of a CCR underlying the FFS reimbursement methodology.

The resulting CCR increase from SFY 2019 to SFY 2023 is approximately 0.5% after accounting for the House Bill 4 allowance. Consistent with historical observations, we expect modest annual fluctuations in reported CCRs due to cost changes, billed charge changes, volume changes, and other factors. Due to the limited CCR change observed between SFY 2019 and SFY 2023, we do not make further adjustments beyond the House Bill 4 reimbursement increase and the standard annual hospital operating cost trend.

Projected State Owned Hospital Expenditures

The Amendment #9 SFY 2023 MCM capitation rates include two changes related to projected expenditures at New Hampshire Hospital and Hampstead Hospital.

New Hampshire Hospital Staffing Constraints

Since July 1, 2022, New Hampshire Hospital (NHH) has been operating with 145 beds rather than its usual 193 beds due to staffing constraints, which represents about 75% of historical capacity. NHH and DHHS anticipate this same capacity level to continue through June 2023. As such, we applied an adjustment to reflect the decreased utilization at NHH.

New Hampshire Hospital and Hampstead Hospital Professional Services

Effective January 1, 2023, New Hampshire Hospital and Hampstead Hospital will begin billing MCOs for professional services provided to the MCM population. Reimbursement for professional services is expected to be set at Medicaid fee levels through June 30, 2023, and then increased to Medicare fee levels starting on July 1, 2023.

We used professional services claim information for commercial and Medicare patients provided by NHH to estimate the cost of professional services on a per admission basis. We estimate the per admission professional services cost to be \$1,570.93 under Medicaid fee-for-service reimbursement and \$2,835.30 under Medicare reimbursement.

Applied Behavioral Analysis Trend

For applied behavioral analysis (ABA) services, we reviewed historical data through June 2022 to develop separate utilization and unit cost trends. With MCOs expanding their ABA networks in recent years, we looked at the percentage of members with autism receiving ABA services over time to evaluate the change in utilization, as well as average ABA cost per member to estimate changes in unit cost. We selected an annual utilization trend of 21.7% (varies by rate cell) and a 0.0% annual trend for unit cost. We then trended SFY 2022 experience data at these annual rates to develop our projected SFY 2023 costs PMPM. The resulting annual trend rate from CY 2019 to SFY 2023 is 21.7% compared to the Amendment #8 assumption of 22.5%.



Freestanding and Hospital Based Birthing Services

Effective July 1, 2022, DHHS is increasing reimbursement for freestanding and hospital based birthing services as follows:

- Free-standing birthing centers receive an increased per diem rate to \$921.00, a 44.4% rate increase
- Neonatal DRGs (DRG 789 through 794) are increased by 25%
- Birthing-related DRGs (DRGs 768, 783-788, 795-797, 805-807) are increased by 34% for Critical Access Hospitals and 24% for Non-Critical Access Hospitals

We reviewed CY 2018 and CY 2019 encounter claims data and summarized paid amounts by facility type and DRG. We then applied the mandated increase by rate cell accordingly.

Additionally, we made an adjustment for the closure of Frisbie Memorial Hospital's birthing center. We assumed these deliveries would move to Portsmouth Regional Hospital, Wentworth Douglas Hospital, or other comparable hospitals, resulting in MCO reimbursement levels approximately 3.0% higher than the historical reimbursement level at Frisbie Memorial Hospital. We applied the increase in reimbursement by rate cell based on the distribution of historical CY 2018 and CY 2019 encounter claims at Frisbie's birthing center.

Transportation

The Amendment #9 SFY 2023 MCM capitation rates include allowances for increased reimbursement rates for certain NEMT services as follows:

- <u>Ambulance Emergency Medical Transportation Rates</u>: Effective July 1, 2022, an increase to set Emergency Medical Transportation rates equal to 50% of current Medicare rates for the following codes:
 - A0425
 - A0427
 - A0429
 - A0433
 - A0434

We applied the rate increase to the CY 2018 and CY 2019 MCO encounter data to estimate the impact of this rate increase.

 Friends and Family: Effective January 1, 2023, reimbursement will be increased to \$0.625 per mile for code A0090.

We applied the rate increase to the CY 2018 and CY 2019 MCO encounter data to estimate the impact of this rate increase. We adjusted our projections to reflect a three month ramp up period to allow for MCO and NEMT vendor communications with enrollees.

- <u>Chair Car Transportation Rates</u>: Effective January 1, 2023 the SFY 2023 capitation rates include additional funding for the following codes when they are used for chair car transportation:
 - A0100
 - S0215
 - A0130
 - S0209
 - T2005
 - T2049
 - A0428
 - A0380
 - A0426
 - A0390



This estimated SFY 2023 funding increase, roughly \$0.09 PMPM accounting for the January 1, 2023 effective date, is based on information provided by the MCOs regarding the updated fee schedule relative to their historical contracts with NEMT providers.

October 2022 DRG Weights

We updated the hospital inpatient unit cost trend using the actual 2023 DRG weights effective October 1, 2022. A review of the newly released weights suggests a 0.8% unit cost trend between SFY 2022 and SFY 2023 compared to our Amendment #8 assumption of 0.45% based on historical data.

Remote Patient Monitoring

Effective January 1, 2023, remote patient monitoring and store and forward services will be covered under the MCM contract. We reviewed emerging CY 2021 and CY 2022 claims experience for a Medicare enrolled population for select CPT codes provided by DHHS. We calculated the remote patient monitoring and store and forward services expenditures as a percentage of home health services spending. We determined these services represent about 0.5% of home health expenditures based on a review of emerging Medicare fee-for-service and Medicare Advantage data. Medicare began covering these services in 2019 and utilization has ramped up from 2019 to 2022. We used this same percentage across rate cells to develop a PMPM adjustment to the rates.

Advanced Laboratory Tests and DME Codes

The Amendment #9 SFY 2023 MCM capitation rates include allowances for new laboratory tests and additional DME codes effective January 1, 2023, as follows:

Advanced Testing for Ovarian and Breast Cancer: The MyChoice CDx (0172U) lab test determines homologous recombination deficiency (HRD) status using somatic mutation analysis to detect variants in tumor suppressor genes BRCA1 and BRCA2 for breast and ovarian cancer. The resulting genomic instability score provides an algorithmic measurement of certain biomarkers valuable for selection of cancer treatment using Lynparza or Zejula.

We estimate the annual utilization for this laboratory test ranges from 150 to 230 tests, assuming that 20% to 35% of Medicaid enrollees with either breast or ovarian cancer will be provided the test.

We developed our utilization estimate by reviewing fee for service and encounter data for the MCM program to count the annual number of cases for breast and ovarian cancer. We determined approximately 0.3% of the New Hampshire Medicaid program population, or 780 members, are breast or ovarian cancer patients. We selected our testing percentage range assumption based on information from breastcancer.org citing that 24.1% of breast cancer patients and 30.9% of ovarian cancer patients are tested for a BCRA mutation. We used a +/- 5% range around these values as our testing percentage range.

The reimbursement on the fee schedule for this test is \$2,424.

 <u>Multianalyte Assays with Algorithmic Analysis test (0014M)</u>: This test provides analysis of three biomarkers and carries out an algorithmic analysis using patient data and the lab test results to report a risk score for liver fibrosis and liver–related clinical events that could occur within five years.

NASH is a severe form of nonalcoholic fatty liver disease (NAFLD) characterized by chronic inflammation and liver cell injury This service is important to identify patients with NASH who have a high risk of disease progression to cirrhosis and liver-related clinical events, including hepatic decompensation, Model for End-Stage Liver Disease (MELD) score ≥15, liver transplantation, or death. NASH is a severe form of nonalcoholic fatty liver disease (NAFLD) characterized by chronic inflammation and liver cell injury.

In many cases, this test will replace a liver biopsy procedure. We reviewed the CY 2019 encounter claims data for liver biopsies performed in a professional setting. We determined the estimated impact of offering this test is not material due to the majority of liver biopsies being performed in a facility based setting, and therefore, not subject to a fee schedule based reimbursement. We made no further adjustments to the capitation rates for this new laboratory test.



Henry Lipman, FACHE NH Department of Health and Human Services December 9, 2022 Page 8 of 9

- <u>Other DME Codes</u>: We also increased reimbursement for the following DME codes:
 - E0604 Rental / Based on Utilization
 - A4281 Tubing for Breast Pump, Replacement
 - A4282 Adapter for Breast Pump, Replacement; Portable Vehicle Adapter 9 volt
 - A4285 Polycarbonate Bottle for use with Breast Pump, Replacement

The total funding for these DME codes is not material.

New Gene Therapy Carve-Outs

Effective July 1, 2022 DHHS will add two new gene therapies to the list of carved out drugs for the MCM program.

Zynteglo is a gene therapy that received Food and Drug Administration (FDA) approval on August 17, 2022 for the treatment of adult and pediatric patients with beta-thalassemia who require regular red blood cell transfusions. Zynteglo is anticipated to be a one-time treatment that will cost \$2.8 million with additional administration costs associated with required hospitalizations.

Skysona was approved by the FDA on September 16, 2022 for the treatment of the rare neurological disorder cerebral adrenoleukodystrophy (CALD). Skysona is expected to cost \$3 million per treatment.

Both medications will be covered on a fee-for-service basis for MCM enrollees.

SUD Residential Code H0010

Effective January 1, 2023, DHHS is increasing per diem reimbursement for SUD residential code H0010 to \$577.07. We applied the rate increase to the CY 2018 and CY 2019 MCO encounter data to estimate the impact of this rate increase. We adjusted our projections to reflect a three month ramp up period for implementation.

CAVEATS AND LIMITATIONS ON USE

Milliman has developed certain models to estimate the values included in this letter. The intent of the models was to estimate Amendment #9 SFY 2023 capitation rates. We have reviewed the models, including their inputs, calculations, and outputs for consistency, reasonableness, and appropriateness to the intended purpose and in compliance with generally accepted actuarial practice and relevant actuarial standards of practice (ASOP).

The models rely on data and information as input to the models. We used fee-for-service (FFS) and MCO encounter cost and eligibility data for July 2017 through June 2022, MCO financial data, historical reimbursement information, TPL recoveries, current fee schedules, and other DHHS and MCO information to calculate the New Hampshire MCM program capitation rates shown in this letter. This data was provided by DHHS and participating MCOs. We did not audit this data and other information. If the underlying data or information is inaccurate or incomplete, the results of our analysis may likewise be inaccurate or incomplete. We performed a limited review of the data used directly in our analysis for reasonableness and consistency and did not find material defects in the data. If there are material defects in the data, it is possible they would be uncovered by a detailed, systematic review and comparison of the data to search for data values that are questionable, or for relationships that are materially inconsistent. Such a review was beyond the scope of our assignment. Please see the data reliance letter for a full list of the data relied upon to develop the SFY 2023 capitation rates.

We constructed several projection models to develop the capitation rates shown in this letter. Differences between the capitation rates and actual MCO experience will depend on the extent to which future experience conforms to the assumptions made in the capitation rate development calculations. It is certain that actual experience will not conform exactly to the assumptions used in the SFY 2023 capitation rates, due to differences in health care trend, managed care efficiency, provider reimbursement levels, and many other factors. Actual amounts will differ from projected amounts to the extent that actual experience is higher or lower than expected.



Milliman prepared this letter for the specific purpose of developing SFY 2023 MCM program capitation rates. This letter should not be used for any other purpose. The letter and the models used to develop the values in this letter have been prepared solely for the internal business use of, and is only to be relied upon by, the management of DHHS. We understand this letter may be shared with participating MCOs, CMS, and other interested parties. Milliman does not intend to benefit or create a legal duty to any third party recipient of its work. This letter should only be reviewed in its entirety.

The results of this letter are technical in nature and are dependent upon specific assumptions and methods. No party should rely on these results without a thorough understanding of those assumptions and methods. Such an understanding may require consultation with qualified professionals.

The authors of this letter are consulting actuaries for Milliman, members of the American Academy of Actuaries, and meet the Qualification Standards of the Academy to render the actuarial opinion contained herein. To the best of their knowledge and belief, this letter is complete and accurate and has been prepared in accordance with generally recognized and accepted actuarial principles and practices.

The terms of Milliman's contract with the New Hampshire Department of Health and Human Services effective July 1, 2022, apply to this letter and its use.



Please call John Meerschaert, Greg Herrle, Sarah Wunder or me at 262 796 2250 if you have any questions.

Sincerely,

Mathieu Doucet, FSA, MAAA Senior Consulting Actuary

MD/jf

Attachments (Provided in Excel)



December 9, 2022

New Hampshire Department of Health and Human Services Capitated Contracts Ratesetting Actuarial Certification SFY 2023 Medicaid Care Management Program Capitation Rates

I, John D. Meerschaert, am associated with the firm of Milliman, Inc., am a member of the American Academy of Actuaries, and meet its Qualification Standards for Statements of Actuarial Opinion. I was retained by the New Hampshire Department of Health and Human Services (DHHS) to perform an actuarial certification of the Medicaid Care Management program capitation rates for SFY 2023 for filing with the Centers for Medicare and Medicaid Services (CMS). I reviewed the calculated capitation rates and am familiar with the relevant requirements of 42 CFR 438; the CMS "Attachment A, PAHP, PIHP and MCO Contracts Financial Review Documentation for At-risk Capitated Contracts Ratesetting;" the 2022 to 2023 Medicaid Managed Care Rate Development Guide; and Actuarial Standard of Practice (ASOP) 49.

I examined the actuarial assumptions and actuarial methods used in setting the capitation rates for SFY 2023. To the best of my information, knowledge and belief, the capitation rates offered by DHHS are in compliance with the relevant requirements of 42 § CFR 438.3(c), 438.3(e), 438.4, 438.5, 438.6, and 438.7. The attached actuarial report describes the capitation rate setting methodology.

In my opinion, the capitation rates are actuarially sound, as defined in Actuarial Standard of Practice (ASOP) 49, were developed in accordance with generally accepted actuarial principles and practices, and are appropriate for the populations to be covered and the services to be furnished under the contract.

In making my opinion, I relied upon the accuracy of the underlying claims and eligibility data records prepared by DHHS, as well as encounter data, financial data summaries, and other information prepared by the participating MCOs. A copy of the reliance letter received from DHHS is attached and constitutes part of this opinion. I did not audit the data and calculations, but did review them for reasonableness and consistency and did not find material defects. In other respects, my examination included such review of the underlying assumptions and methods used and such tests of the calculations as I considered necessary.

The capitation rates developed may not be appropriate for any specific MCO. Any MCO will need to review the rates in relation to the benefits provided. Each MCO should compare the rates with its own experience, expenses, capital and surplus, and profit requirements prior to agreeing to contract with DHHS. The MCO may require rates above, equal to, or below the actuarially sound capitation rates.

Actuarial methods, considerations, and analyses used in forming my opinion conform to the appropriate Standards of Practice as promulgated from time-to-time by the Actuarial Standards Board, whose standards form the basis of this Statement of Opinion.

It should be emphasized that capitation rates are a projection of future costs based on a set of assumptions. Actual costs will be dependent on each contracted MCO's situation and experience.

This Opinion assumes the reader is familiar with the New Hampshire Medicaid program, Medicaid eligibility rules, and actuarial rating techniques. The Opinion is intended for the State of New Hampshire and the Centers for Medicare and Medicaid Services and should not be relied on by other parties. The reader should be advised by actuaries or other professionals competent in the area of actuarial rate projections of the type in this Opinion, so as to properly interpret the projection results.

∮ohn D. Meerschaert Member, American Academy of Actuaries December 9, 2022



Lori A. Shibinette

Commissioner

Henry D. Lipman

Director

STATE OF NEW HAMPSHIRE

DEPARTMENT OF HEALTH AND HUMAN SERVICES

DIVISION OF MEDICAID SERVICES

129 PLEASANT STREET, CONCORD, NH 03301 1-844-ASK-DHHS (1-844-275-3447) Fax: 603-271-8431 TDD Access: 1-800-735-2964 www.dhhs.nh.gov

May 19, 2022

Mr. John D. Meerschaert, F.S.A. Milliman, Inc. 15800 Bluemound Road, Suite 100 Brookfield, WI 53005

Re: Actuarial Certification of SFY 2023 Capitation Rates for New Hampshire Medicaid Care Management Program Capitation Rates

Dear Mr. Meerschaert:

I, Henry Lipman, Medicaid Director for the New Hampshire Department of Health and Human Services, hereby affirm that the data prepared and submitted to Milliman, Inc. (Milliman) for the purpose of certifying the SFY 2023 New Hampshire Medicaid Care Management (MCM) program capitation rates were prepared under my direction, and to the best of my knowledge and belief are accurate and complete. This data includes:

- 1. Computer files supporting the SFY 2023 capitation rate calculation, including, but not limited to:
 - Technical Definition for NH MCM Data Book Services Scope V3.doc
 - Reference Files.xls
 - NH Provider Type Codes and Descriptions.xls
 - Eligibility Category Detail.xlsx
 - Medicaid CAHs.xls
 - OP-RHC-FQHC Reimbursement Process as of 2-5-13.doc
 - Provider Payment Algorithms2011.docx
 - QA LOG Care management 111412012 w corrected date at top.xlsx
 - Medicaid Extract and Claims Information.doc
 - NH+Medicaid+rebranded+detailed+FQHC+Provider+Manual+2-1-18.pdf
 - Newborn Reporting Procedures Guidance Statement 20121130.doc
 - NH Care Management Contract Exhibit A 031612.pdf
 - NH MCM Rate Cells Definition 2014-02-20.xls
 - Community Mental Health Agreement 1.22.15.pdf
 - Fiscal Impact Change of Scope & LAL SFY19.xlsx
 - FY20 21 Provider Rate Increases by Senate Bill and Milliman Exh 1.xlsx
 - NHM PDL 20220328.xlsx
 - PG02+PG09 Provider Numbers.xlsx
 - DME MP for Analysis 09_01_2020.xlsx
 - Initial Estimate Funding ALL MCOs (Pmts 0722-0623).xlsx
 - Neuro Psych (96131-96139) Utilization 2021.10.05.xlsx

Mr. John D. Meerschaert, F.S.A. May 19, 2022 Page 2 of 3

- Rate Increase Impacts-Peristeen A4459 & catheters.xlsx
- Metric_Quantity_Request with MMIS Units.xlsx
- 2. Fee schedule files:
 - 1) CY/SFY/FFY 2018 fee schedules:
 - · 2018 DRG Rate Sheet.xls
 - SFY 18 Hosp IP & OP.xls
 - 2018 NH Fee Schedule Covered Procedures 02232018.xlsx
 - 2018 NH Fee Schedule Manually Priced Procedures 02232018.xlsx
 - 2018 Medicaid Rates for Diagnostic and Physician Rates-20190108.xlsx
 - 2018 ASC Fee Schedule.xls
 - 2018 Hospice Rates worksheet-Final.xlsx
 - FQHC Based Rate SFY 2018.xls
 - SFY18 RATE CHANGE LOG.xlsx
 - 2019 FISCAL IMPACT Hospice Rates FINAL.xlsx
 - 2) CY/SFY/FFY 2019 fee schedules:
 - 2019 DRG Rate Sheet.xls
 - Fee Schedule 1-8-19.xlsx
 - Hospice Rates PDF File for Providers Eff 10_1_2018.pdf
 - Report of Covered Procedures 20190110.xlsx
 - Report of Manually Priced Procedures 20190110.xlsx
 - 3) CY/SFY/FFY 2023 fee schedules:
 - SFY23 RATE CHANGE LOG.xlsx
 - FFY22 DRG Rate Sheet .xlsx
 - Covered proedures Fee schedule as of 01012021.xlsx
 - Manual Priced Procedures 01012021.xlsx
 - 2022 Hospice Rates.xlsx
 - ASC Rates_20220125_Fee Sched_1.26.22.xlsx
 - J Code CSR Attachment 22020405.xlsx
- 3. January 2015 October 2021 Medicaid eligibility data and claims from MMIS and MCO encounter files, including:
 - Biweekly claims data (facility, professional and drug).
 - Biweekly enrollment data
 - Provider reference files.
 - Supplemental eligibility/ineligibility files
 - Additional FFS Hospice claims:
 - SFY 2018 Hospice data for Milliman.xlsx
 - 2020_3_4_Hospice_SFY2019 Data for Milliman.xlsx

Mr. John D. Meerschaert, F.S.A. May 19, 2022 Page 3 of 3

- Additional FFS CTS claims:
 - CTS FFS Encounter Data.csv
 - CTS PAP Encounter Data.csv
 - Additional BDAS claims:
 - tblWITS_SFY18_SvcInMedicaidSpan.xlsx
- Plan Selection Period info:
 - Fee for Service Spans for Andrew 1-1-2018 to current.xls
- IMD Claims:
 - NHH_Medicaid_2016-2019.xlsx
- 4. Other supporting documentation, including:
 - MCO contract
 - Financial Reconciliation data from participating health plans
 - MCO administrative cost survey results
 - ATECH services details
 - Behavioral Health Crisis Treatment Center (BHCTC) implementation and termination schedule
 - DRF identification process
 - Hospital directed payment details
 - Community Residential Service expansion details
 - Mobile Crisis Response Team and Emergency Services identification process
 - Home and Community Based Services directed payment details
 - Genetic Testing for Hospitals details
 - Fiscal impact and appropriation amounts for various legislative provisions
 - Other computer files
 - Conversations concerning supplied dat

ed data	01
Jar	20hr
Henry Lipman	

1

Medicaid Director

May 19, 2022

State of New Hampshire Department of Health and Human Services

State Fiscal Year (SFY) 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

John D. Meerschaert, FSA, MAAA Principal and Consulting Actuary

Greg J. Herrle, FSA, MAAA Principal and Consulting Actuary Mathieu Doucet, FSA, MAAA Senior Consulting Actuary

Sarah Wunder, FSA, MAAA Consulting Actuary Rachel Marsiglio, ASA, MAAA Actuarial Manager

Jacob Magnusson, ASA, MAAA Senior Actuarial Manager



15800 W. Bluemound Road Suite 100 Brookfield, WI 53005 USA Tel +1 262 784 2250 Fax +1 262 923 3680



C Milliman

Table of Contents

I.	EXECUTIVE SUMMARY	1
II.	METHODOLOGY OVERVIEW	9
III.	MEDICAL COST PROJECTIONS	.15
IV.	FINAL CAPITATION RATE ADJUSTMENTS	.28
V .	SERVICE CATEGORY ASSIGNMENT	.35
VI.	CMS RATE SETTING CHECKLIST ISSUES	. 38
VII.	RESPONSE TO 2022-2023 MEDICAID MANAGED CARE RATE DEVELOPMENT GUIDE (APRIL 2022)	.43

APPENDICES

Capitation Rate Development Exhibits (Provided in Excel)

Appendices - SFY 2023 MCM Rates.xlsx

Fiscal Impact Exhibits (Provided in Excel)

- Exhibit A: Comparison of Amendment #8 SFY 2023 Capitation Rates to Amendment #7 SFY 2022 Capitation Rates
- Exhibit B: Estimated Fiscal Impact of Medicaid Care Management Program

Other Exhibits (Provided in Excel)

- Exhibit C: Boston Children's Hospital Funding Development
- Exhibit D: Acuity Adjustment Development
- Exhibit E: Community Mental Health Center Cost Component Summary
- Exhibit F: Enrollment Projections
- Exhibit G: Pharmacy Efficiency Savings
- Exhibit H: List of NDCs for Carved-Out Prescription Drugs
- Exhibit I: Rate Cell Logic Letter
- Exhibit J: Summary of Preferred Drug List Analysis

Actuarial Certification

Exhibit K: Actuarial Certification of SFY 2023 New Hampshire Medicaid Care Management Capitation Rates

Reliance Letter

State of New Hampshire Department of Health and Human Services

SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

I. EXECUTIVE SUMMARY

This report documents the development of the state fiscal year (SFY) 2023 managed care organization (MCO) capitation rates for New Hampshire's Medicaid Care Management (MCM) program, including the following populations:

- Standard Medicaid, which also includes the Children's Health Insurance Program (CHIP) population
- Granite Advantage Health Care Program (GAHCP), which is New Hampshire's adult Medicaid expansion program

The New Hampshire Department of Health and Human Services (DHHS) retained Milliman to calculate, document, and certify its capitation rate development. We developed the capitation rates using the methodology described in this report. The SFY 2023 capitation rates are included in Amendment #8 to the MCM program contracts with the three managed care organizations (MCOs).

Our role is to certify that the SFY 2023 capitation rates produced by the rating methodology are actuarially sound to comply with Centers for Medicare and Medicaid Services (CMS) regulations. We developed actuarially sound capitation rates using published guidance from the American Academy of Actuaries (AAA), the Actuarial Standards Board (ASB), CMS, and federal regulations to ensure compliance with generally accepted actuarial practices and regulatory requirements.

SFY 2023 PROGRAM CHANGES

The SFY 2023 capitation rates reflect several program changes from the SFY 2022 capitation rates presented in our December 1, 2021 certification. The changes are as follows:

- Inclusion of hospital admissions at New Hampshire Hospital for age 21 to 64 adults as an at-risk service included in the monthly capitation rates
- Transition of all hospital admission costs for Hampstead Hospital from the inpatient psychiatric admission kick payment to the monthly capitation rates
- Implementation of a minimum fee schedule for admissions at New Hampshire and Hampstead Hospitals
- Inclusion of targeted pharmacy savings through review of several pharmacy efficiency measures
- Implementation of a retroactive acuity adjustment as part of the risk mitigation strategy

COVID-19 CONSIDERATIONS IN SFY 2023 RATE DEVELOPMENT

As of the time of this report, the impact on SFY 2023 capitation rates due to the COVID-19 pandemic and public health emergency (PHE) is difficult to predict. As such, a minimum and maximum medical loss ratio (MLR) will be in effect in SFY 2023 to reflect the uncertainty in the capitation rates due to COVID-19. The minimum and maximum MLR provision is described in more detail in Section III.

In addition, explicit adjustments for COVID-19 are made in the rate development for the following issues, as described in Section IV:

- Medicaid enrollment is projected to decrease through March 31, 2023 and remain flat from April 1, 2023 through the end of the fiscal year. The SFY 2023 capitation rates in Amendment #8 of the MCM contract reflect this updated enrollment distribution, and the preliminary acuity adjustment factor for each rate cell has been updated to reflect the lower enrollment estimates used in Amendment #8 vs. Amendment #7.
- Due to the uncertainty surrounding the unwind process following the end of the PHE, DHHS will implement a non-budget neutral retrospective acuity adjustment. This adjustment will capture the relative acuity of the actual individuals enrolled during SFY 2023 during the continuing PHE and through the redetermination process. The preliminary acuity factor included in this report will be updated retroactively to account for the actual PHE end date, the redetermination process, and relative acuity levels.

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

This report assumes that the reader is familiar with the State of New Hampshire's Medicaid program, its benefits, and rate setting principles. This report was prepared solely to provide assistance to DHHS to set SFY 2023 capitation rates for the Medicaid Care Management program. It may not be appropriate for other purposes. Milliman does not intend to benefit and assumes no duty or liability to other parties who receive this work. This material should only be reviewed in its entirety.

- The capitation rates do not include provisions for expected vaccination administration fees related to COVID-19 in SFY 2023 as these costs remain carved out of the MCM program.
- Since the start of the COVID-19 pandemic and PHE, service utilization levels in the New Hampshire Medicaid program have lagged behind CY 2019 utilization by rate cell and on an aggregate basis. The SFY 2023 capitation rates assume SFY 2023 utilization levels will be consistent with CY 2019. As such, we did not apply utilization trend from CY 2019 to SFY 2023 with the exception of ABA services. This utilization projection is supported by our review of historical quarterly PMPM amounts that indicate emerging costs through June 2021 were below CY 2019 even with the known provider rate increases and acuity differences. We also reviewed DHHS summaries of emerging encounter data through January 2022 show similar cost reductions compared to CY 2019. We will continue to monitor emerging experience and revisit this assumption for the planned January 2023 rate amendment.

The SFY 2023 capitation rates do not include any explicit adjustments for the following issues. The net impact of these issues is implicitly included in the assumption that SFY 2023 service utilization will be consistent with CY 2019 service utilization on an aggregate basis. The MCM program risk mitigation provisions protect DHHS and the MCOs from material overpayments or underpayments.

- COVID-19 testing and treatment cost: The prevalence of COVID-19 infection rates in SFY 2023 is dependent on many variables that make it difficult to predict to include an estimate for the cost of testing for and treating individuals with COVID-19 including, but not limited to:
 - The take-up rate and timing of COVID-19 vaccinations
 - The emergence of COVID-19 variants and the efficacy of vaccines on these variants
 - The implementation of social distancing measures
- To our knowledge, there is not a publicly available model that includes COVID-19 infection rate or hospital admissions through June 2023. In addition, the publicly available models have materially changed short-term and long-term projections of COVID-19 prevalence in reaction to emerging data. While we have reviewed historical MCM data for the cost of testing and treating COVID-19 in CY 2021, given the unpredictable patterns of COVID-19 prevalence to date in New Hampshire and the changing national models there is a range of potential impacts on the SFY 2023 rates.
- Deferred and foregone services: The most significant fiscal impact of COVID-19 to date has been the deferral of non-essential services, either through government-enacted policies, the impact of social distancing on the administration of services, or personal choice. We reviewed MCM emerging claims data incurred through February 2022 by population type (to remove the impact of membership mix changes). As of February 2022, there was still measurable reductions in claim costs for each population compared to the PMPMs prior to the pandemic in CY 2019. However, it is difficult to use this historical data to project the impact of deferred services for SFY 2023 for many reasons.
 - Utilization by service category has varied in CY 2020 and 2021 as the level of COVID-19 diagnoses and hospital admissions has changed in New Hampshire over the course of the pandemic. Therefore, the unknown future prevalence of COVID-19 will be a key variable in future service utilization changes relative to pre-pandemic levels.
 - In the MCM data we reviewed it is difficult to isolate the impact of deferred services from changes in utilization due to other drivers, such as member acuity changes or change in service mix.
- Service Mix Changes: In response to the pandemic, the mix of services used to treat patients has changed, such as the use of telehealth services. It is unknown if these provider and patient changes will persist after the end of the pandemic.

SFY 2023 CAPITATION RATE CHANGE

Table 1A below shows the statewide rate change by population from the Amendment #7 SFY 2022 capitation rates to the Amendment #8 SFY 2023 rates in this report using current DHHS enrollment estimates and **excludes directed**

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

payments. DHHS has been paying the MCOs the rates included in Amendment #7. Exhibit A shows a summary of the rate change by rate cell.

The rate change is shown including and excluding mental health admissions to New Hampshire Hospital (NHH) of less than 60 days for age 21 to 64 population. The SMI/SED 1115 waiver amendment to the Substance Use Disorder Treatment and Recovery Access (SUD-TRA) 1115 demonstration will allow DHHS to receive federal matching funds for these Institution for Mental Disease (IMD) services. As such, these admissions to New Hampshire Hospital will now be included as an at-risk service for MCOs. In the past, DHHS has covered these services using state general funds, and therefore, including them as a Medicaid results in roughly \$15 million of additional services subject to federal matching.

	Medicaid C SFY 2023 Based on Projected	Table 1A rtment of Health and Hu care Management Progr Capitation Rate Chang I SFY 2023 Enrollment b ing Directed Payments	am e oy Rate Cell		
		SFY 2023 Excluding Impact of SMI/SED 1115	5	Percentage Cha Excluding Impact of SMI/SED 1115	nge
Population	SFY 2022	Wavier Amendment	Total	Waiver Amendment	Total
Standard Medicaid					
Base Population	\$328.56	\$344.83	\$346.28	5.0%	5.4%
CHIP	192.68	213.67	214.07	10.9%	11.1%
Behavioral Health Population	1,381.53	1,408.40	1,461.70	1.9%	5.8%
Total Standard Medicaid	\$384.19	\$401.87	\$406.85	4.6%	5.9%
Granite Advantage Health Care Proc	Iram				
Medically Frail	\$1,091.37	\$1,093.39	\$1,097.65	0.2%	0.6%
Non-Medically Frail	418.97	418.07	419.51	-0.2%	0.1%
Behavioral Health Population	1,933.60	1,951.04	2,033.26	0.9%	5.2%
Total GAHCP	\$563.17	\$563.42	\$568.33	0.0%	0.9%
Grand Total	\$441.88	\$453.94	\$458.90	2.7%	3.8%

Table 1B below shows the statewide rate change by population from the Amendment #7 SFY 2022 capitation rates to the Amendment #8 SFY 2023 rates, **including directed payments**.

	Medicaid Ca SFY 2023 Based on Projected	Table 1B ment of Health and Hu are Management Progra Capitation Rate Change SFY 2023 Enrollment b g Directed Payments	im Ə		
		SFY 202 Excluding Impact of SMI/SED 1115	3	Percentage Char Excluding Impact of SMI/SED 1115	nge
Population	SFY 2022	Wavier Amendment	Total	Wavier Amendment	Total
Standard Medicaid					
Base Population	\$352.95	\$356.71	\$358.17	1.1%	1.5%
CHIP	198.39	219.27	219.67	10.5%	10.7%
Behavioral Health Population	1,470.78	1,464.33	1,517.68	-0.4%	3.2%
Total Standard Medicaid	\$410.57	\$416.00	\$420.98	1.3%	2.5%
Granite Advantage Health Care Prog	Iram				
Medically Frail	\$1,144.47	\$1,146.77	\$1,150.99	0.2%	0.6%
Non-Medically Frail	432.79	431.13	432.57	-0.4%	-0.1%
Behavioral Health Population	2,001.29	1,992.81	2,075.12	-0.4%	3.7%
Total GAHCP	\$583.63	\$582.21	\$587.11	-0.2%	0.6%
Grand Total	\$466.35	\$469.57	\$474.53	0.7%	1.8%

State of New Hampshire Department of Health and Human Services

SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

This report assumes that the reader is familiar with the State of New Hampshire's Medicaid program, its benefits, and rate setting principles. This report was prepared solely to provide assistance to DHHS to set SFY 2023 capitation rates for the Medicaid Care Management program. It may not be appropriate for other purposes. Milliman does not intend to benefit and assumes no duty or liability to other parties who receive this work. This material should only be reviewed in its entirety.

Page 3

The SFY 2023 capitation rates include directed payments to community mental health centers (CMHCs) and critical access hospitals (both subject to CMS approval). The HCBS directed payment that was in place during SFY 2022 was eliminated for SFY 2023. While these amounts are included in the MCO capitation rates, MCOs are not at risk for the amount of these directed payments.

We project an overall MCO medical loss ratio (MLR) for at-risk services of 90.8% for the Standard Medicaid population and 90.1% for the GAHCP population in SFY 2023, which includes:

- An overall 7.7% administrative cost allowance for the Standard Medicaid population, and 8.4% for the GAHCP population
- A 1.5% risk margin applied as a percentage of revenue prior to the directed payments and the premium tax allowance

The projected MLR for at-risk services excludes the impact of the directed payments and the 2.0% premium tax allowance in both the numerator and denominator of the MLR calculation, which is consistent with the treatment of directed payments and premium taxes in federal MLR calculations.

The SFY 2023 MCM capitation rates do not include funding for the COVID-19 vaccines. DHHS will continue to cover the vaccine administration costs under a non-risk arrangement for SFY 2023.

Table 2 shows a breakdown of the capitation rate changes by major rate component for the Amendment #8 SFY 2023 capitation rates compared to the Amendment #7 SFY 2022 capitation rates on a percentage and aggregate dollar basis. Please note, the various rate change components include the impact of attributed administrative allowance, risk / profit margin and premium tax beyond the service costs alone.

May 20, 2022

		Table 2 epartment of Hea Capitation Rate C				
		cted SFY 2023 E				
		rd Medicaid		АНСР	Т	otal
Rate Component	Rate Change	12 Month Dollar Impact	Rate Change	12 Month Dollar Impact	Rate Change	12 Month Dollar Impact
•		•				•
Inclusion of IMD admissions of less than 6			ion as a cov	ered Medicaid serv	rice through SM	/II/SED 1115
waiver amendment to the SUD-TRA 1115 de Newly covered New Hampshire Hospital	emonstratio	n				
(NHH) admissions repriced at NHH						
minimum fee schedule	1.1%	\$7,703,888	0.8%	\$3,755,578	1.0%	\$11,459,46
Reversal of IMD exclusion for other		, , - ,		, . , ,	-	, , , -
impacted services	0.1%	\$672,069	0.0%	\$163,043	0.1%	\$835,11
Total impact of SMI/SED 1115 waiver						
amendment	1.2%	\$8,375,957	0.8%	\$3,918,620	1.1%	\$12,294,57
Other program changes						
New Hampshire and Hampstead Hospitals						
– repricing to minimum fee schedule for						
services historically covered by MCM						
program	0.5%	\$3,450,993	0.1%	\$365,521	0.3%	\$3,816,51
Community residential services – additional						
beds	0.1%	745,993	0.0%	121,129	0.1%	867,12
Preliminary SFY 2023 acuity adjustment						
compared to SFY 2022 acuity assumption	0.7%	4,769,045	1.5%	7,211,955	1.0%	11,981,00
Total other program changes	1.3%	\$8,966,032	1.6%	\$7,698,605	1.4%	\$16,664,63
Other changes relative to SFY 2022 rating a						
Non-pharmacy utilization trends	-1.1%	(7,844,706)	-1.1%	(5,248,839)	-1.1%	(13,093,54
Opioid addiction treatment trend adjustment	-0.5%	(3,146,990)	-0.8%	(3,909,040)	-0.6%	(7,056,03
Inpatient and outpatient hospital unit cost						
trends	0.5%	3,184,479	0.6%	2,980,746	0.5%	6,165,22
Applied behavioral analysis service trends	1.1%	7,450,166	0.0%	5,939	0.6%	7,456,10
Pharmacy trends	2.4%	16,751,556	0.3%	1,296,337	1.6%	18,047,89
Preferred drug list (PDL) adjustment	-0.1%	(1,000,190)	-0.9%	(4,036,173)	-0.4%	(5,036,363
Pharmacy savings initiatives	0.0%	121,825	0.1%	406,868	0.0%	528,69
J Code fee schedule changes	0.1%	1,033,584	0.2%	991,855	0.2%	2,025,43
Boston Children's Hospital risk pool funding						
level	0.3%	1,955,549	0.0%	34,726	0.2%	1,990,27
Other	0.3%	2,227,495	0.0%	(21,256)	0.2%	2,206,24
Total Other Changes	3.0%	\$20,732,768	-1.6%	(\$7,498,837)	1.1%	\$13,233,93
Total rate change for SFY 2023 relative						
to SFY 2022 rates (prior to directed						
payment changes)	5.5%	\$38,074,756	0.9%	\$4,118,388	3.6%	\$42,193,14
Directed payment changes						
Removal of HCBS directed payment	-3.3%	(23,012,503)	-0.7%	(3,377,214)	-2.3%	(26,389,717
Impact of enrollment decrease on the	0.070	(20,012,000)	5.175	(0,011,217)	2.070	(20,000,711
PMPM value of fixed-dollar directed						
payments ²	0.4%	2,431,500	0.4%	2,047,234	0.4%	4,478,73
Total directed payment changes	-3.0%	(\$20,581,003)	-0.3%	(\$1,329,981)	-1.9%	(\$21,910,984
Total rate change for SFY 2023 relative						
to SFY 2022 rates (after directed						
payment changes)	2.5%	\$17,493,753	0.6%	\$2,788,408	1.8%	\$20,282,16
¹ The various rate change components include the						

²The SFY 2023 directed payments remain at \$33 million for critical access hospitals and \$5 million for CMHCs. Lower projected enrollment for SFY 2023 compared to SFY 2022 results in higher PMPM amounts to fund the directed payments.

State of New Hampshire Department of Health and Human Services

SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

This report assumes that the reader is familiar with the State of New Hampshire's Medicaid program, its benefits, and rate setting principles. This report was prepared solely to provide assistance to DHHS to set SFY 2023 capitation rates for the Medicaid Care Management program. It may not be appropriate for other purposes. Milliman does not intend to benefit and assumes no duty or liability to other parties who receive this work. This material should only be reviewed in its entirety.

Page 5

The composite capitation rates shown above are based on DHHS' initial projections of MCO enrollment for SFY 2023, which include considerations for the unwind process following the end of the PHE with enrollment returning to pre-PHE levels by March 2023. The enrollment projections by month are included in Exhibit F. Future rate amendments will incorporate updated enrollment estimates that capture additional available information related to the end of the PHE.

We note, the SFY 2023 component changes in Table 2 are expressed relative to the same components in the SFY 2022 rates. As such, the pharmacy savings and preliminary acuity adjustment show increases even though both components result in a reduction to projected SFY 2023 costs. The pharmacy savings analysis identified less opportunity for savings in SFY 2023 than SFY 2022. Similarly, the preliminary acuity adjustment reduces costs by a lower amount in SFY 2023 than SFY 2022 largely due to the current estimate of enrollment timing and trajectory during the PHE unwind.

SFY 2023 CAPITATION RATES

Table 3 shows statewide capitation rates for the base population that DHHS will pay to MCOs for each member meeting these rate cell criteria. Note, the base population rate cells exclude the behavioral health population and the GAHCP populations, which have separate rate cells as shown in Tables 4 and 5. The rate changes by rate cell are shown in the accompanying Excel workbooks.

Table 3 New Hampshire Department of Health and Human Services Medicaid Care Management Program SFY 2023 Base Population Capitation Rates ¹					
Base Population Rate Cell	SFY 2023				
Low Income Children and Adults					
0 to 11 months	\$350.35				
1 to 18	189.00				
19+	494.04				
CHIP	185.02				
Foster Care / Adoption	374.07				
Severely Disabled Children	1,808.17				
Elderly and Disabled Adults – 19 to 64	1,324.83				
Elderly and Disabled Adults – 65+	1,059.06				
Dual Eligibles	322.91				
Newborn Kick Payment	4,721.98				
Neonatal Abstinence Syndrome Kick Payment	9,707.32				
Maternity Kick Payment	2,875.73				

¹ Excludes the behavioral health population, and GAHCP populations.

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

Table 4 shows the statewide capitation rates for the behavioral health population that DHHS will pay to the MCOs for each member meeting these rate cell criteria. The behavioral health population includes all Standard Medicaid and GAHCP population members eligible for enhanced behavioral health services.

Table 4 New Hampshire Department of Health and Human Services Medicaid Care Management Program SFY 2023 Behavioral Health Population Capitation Rates ¹					
Behavioral Health Population Rate Cells	SFY 2023				
Severe / Persistent Mental Illness – Medicaid Only	\$2,553.10				
Severe / Persistent Mental Illness – Dual Eligibles 1,939.34					
Severe Mental Illness – Medicaid Only	1,698.74				
Severe Mental Illness – Dual Eligibles	1,158.41				
Low Utilizer – Medicaid Only	1,567.60				
Low Utilizer – Dual Eligibles	686.42				
Serious Emotionally Disturbed Child	1,154.84				

¹ Excludes the base population, GAHCP Medically Frail, and GAHCP Non-Medically Frail populations.

Table 5 shows the statewide capitation rates for the GAHCP Medically Frail and Non-Medically Frail populations that DHHS will pay to the MCOs for each member meeting these rate cell criteria.

Table 5					
New Hampshire Department of Health and Human Services					
Medicaid Care Management Program					
SFY 2023 GAHCP Population Capitation Rates ¹					
GAHCP Rate Cell	SFY 2023				
Medically Frail	\$1,150.99				
Non-Medically Frail 432.57					
1 Evelvedee the helps viewel he althe menulation					

¹ Excludes the behavioral health population.

Table 6 shows the kick payments DHHS will pay to the MCOs for each hospital inpatient psychiatric admission. While the SFY 2022 psychiatric admission kick payments included services provided at New Hampshire Hospital and Hampstead Hospital, such services are now captured in the monthly capitation rates and are excluded from Table 6 below.

Table 6 New Hampshire Department of Health and Human Services Medicaid Care Management Program SFY 2023 Hospital Inpatient Psychiatric Admission Kick Payments ¹		
Psychiatric Kick Payment	SFY 2023	
Peer Group 01 and 07	\$3,681.70	
Peer Group 02	2,836.28	
Peer Group 06	17,606.39	
Peer Group 09	8,000.60	

¹ Does not apply to dual eligibles or admissions for adults at New Hampshire Hospital.

It should be emphasized that capitation rates are a projection of future costs for an efficient MCO based on a set of assumptions. Actual MCO costs will be dependent on each MCO's situation and the extent to which future experience conforms to the assumptions made in the capitation rate development calculations.

REPORT STRUCTURE

The accompanying Excel workbooks document the development of the SFY 2023 capitation rates for all populations. Exhibit A shows a comparison of the Amendment #8 SFY 2023 capitation rates to the Amendment #7 SFY 2022 capitation rates documented in our December 1, 2021 rate certification. Exhibit B shows the fiscal impact of the

State of New Hampshire Department of Health and Human Services

Page 7

SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

Amendment #8 SFY 2023 capitation rates based on projected SFY 2023 member months. The actuarial certification of the SFY 2023 New Hampshire MCM program capitation rates is included as Exhibit K.

Section II provides an overview of the methodology, including a summary of methodology changes. Section III documents the capitation rate base data and medical cost projections. Section IV summarizes final capitation rate adjustments for all rate cells, including various program adjustments and the administrative / margin allowance. Section V of the report provides information regarding the assignment of service categories. Section VI discusses issues related to the CMS rate setting checklist. Section VII includes comments on items related to the 2022-2023 Medicaid Managed Care Rate Development Guide.

DATA RELIANCE AND IMPORTANT CAVEATS

Milliman has developed certain models to estimate the values included in this report. The intent of the models was to estimate SFY 2023 capitation rates. We have reviewed the models, including their inputs, calculations, and outputs for consistency, reasonableness, and appropriateness to the intended purpose and in compliance with generally accepted actuarial practice and relevant actuarial standards of practice (ASOPs).

The models rely on data and information as input to the models. We used fee-for-service (FFS) and MCO encounter cost and eligibility data for January 2018 through October 2021, MCO financial data, historical reimbursement information, TPL recoveries, current fee schedules, and other DHHS and MCO information to calculate the New Hampshire MCM program capitation rates shown in this report. This data was provided by DHHS and participating MCOs. We did not audit this data and other information. If the underlying data or information is inaccurate or incomplete, the results of our analysis may likewise be inaccurate or incomplete. We performed a limited review of the data used directly in our analysis for reasonableness and consistency and did not find material defects in the data. If there are material defects in the data, it is possible they would be uncovered by a detailed, systematic review and comparison of the data to search for data values that are questionable or for relationships that are materially inconsistent. Such a review was beyond the scope of our assignment. Please see the data reliance letter for a full list of the data relied upon to develop the SFY 2023 capitation rates.

We constructed several projection models to develop the capitation rates shown in this report. Differences between the capitation rates and actual MCO experience will depend on the extent to which future experience conforms to the assumptions made in the capitation rate development calculations. It is certain that actual experience will not conform exactly to the assumptions used in the SFY 2023 capitation rates due to differences in health care trend, managed care efficiency, provider reimbursement levels, and many other factors. Actual amounts will differ from projected amounts to the extent that actual experience is higher or lower than expected.

Milliman prepared this report for the specific purpose of developing SFY 2023 MCM program capitation rates. The report and the models used to develop the values in this report should not be used for any other purpose. This report has been prepared solely for the internal business use of, and is only to be relied upon by, the management of DHHS. We understand this report may be shared with participating MCOs, CMS, and other interested parties. Milliman does not intend to benefit or create a legal duty to any third party recipient of its work. This report should only be reviewed in its entirety.

The results of this report are technical in nature and are dependent upon specific assumptions and methods. No party should rely on these results without a thorough understanding of those assumptions and methods. Such an understanding may require consultation with qualified professionals.

The authors of this report are consulting actuaries for Milliman, members of the American Academy of Actuaries, and meet the Qualification Standards of the Academy to render the actuarial opinion contained herein. To the best of their knowledge and belief, this report is complete and accurate and has been prepared in accordance with generally recognized and accepted actuarial principles and practices.

The terms of Milliman's contract with the New Hampshire Department of Health and Human Services effective July 1, 2017 apply to this report and its use.

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program Page 8

May 20, 2022

II. METHODOLOGY OVERVIEW

This section of the report provides an overview of the SFY 2023 New Hampshire MCM program capitation rate methodology and highlights program changes effective for SFY 2023.

BASE DATA

We developed the SFY 2023 capitation rates using enrollment data, FFS claims, and MCO encounter claims.

We summarized the following base data sources by rate cell and service category:

- CY 2018 and CY 2019 MCO encounter experience for the Standard Medicaid population and the GAHCP Medically Frail population.
- March 2019 to December 2019 MCO encounter experience for the GAHCP Non-Medically Frail population. We selected the March 2019 to December 2019 time period due to claims ramp-up in January 2019 and February 2019.

We used two years of base data for most rate cells to increase the credibility of the rate development calculation, particularly for smaller rate cells. The base experience period is collected from the following sources:

- We obtained MCO encounter data and sub-capitated expenditures directly from the participating MCOs. The SFY 2023 capitation rates include sub-capitated expenditures for services not capitated through an affiliated organization. For related entities, the SFY 2023 rates include actual encounter payments to providers for those services, when available. We removed administrative payments made by the MCOs to related parties from the encounter data. Actual MCO encounter data is used for all CMHC services. The MCOs also provided summarized provider incentive payments and settlements made outside of the claims data and these items were included in the base data.
- We obtained eligibility and MCM enrollment information from DHHS.

We believe the encounter data is of appropriate quality and completeness to use as the primary basis for developing actuarially sound rates for the New Hampshire MCM program. We validated the MCO encounter data using the following process:

- We compared the submitted encounter data to quarterly financial data summaries provided by the MCOs. The quarterly financial data summaries included FFS and sub-capitated payments made by the MCOs to providers by rate cell, broad service category, and quarter. The financial data was not audited but is certified by each MCO as accurate and complete.
- DHHS and Milliman provided an opportunity for MCOs to play a significant role in the base data validation for the SFY 2023 capitation rate development process. As we worked on the development of the SFY 2023 capitation rates, we provided MCOs with a series of detailed data summaries in order to further our understanding of the data, complete the validation process, and offer more transparency on the process leading to the capitation rates.
- Through this detailed review process, Milliman, DHHS, and the MCOs validated the encounter data for use in the capitation rate setting process.

Table 7 below shows a summary of the encounter data reconciliation to the financial statement information provided by the MCOs. Overall, the CY 2018 and CY 2019 base encounter data is about 0.4% lower than financial summaries provided by MCOs.

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

Table 7 New Hampshire Department of Health and Human Services Medicaid Care Management Program CY 2018 and CY 2019 Data Validation Summary			
Quarter	Dollar Difference	Percentage Difference	
1Q2018	(\$112,053)	-0.1%	
2Q2018	39,862	0.0%	
3Q2018	(1,514,174)	-1.2%	
4Q2018	(1,222,374)	-0.9%	
1Q2019	(1,270,878)	-0.7%	
2Q2019	(2,472,969)	-1.3%	
3Q2019	926,640	0.5%	
4Q2019	(225,491)	-0.1%	
CY 2018	(\$2,808,739)	-0.48%	
CY 2019	(3,042,698)	-0.40%	
Grand Total	(\$5,851,438)	-0.44%	

In recent years, the MCM program capitation rate development process has generally relied upon base data from the two most recent complete state fiscal years. However, the impact of the COVID-19 pandemic on costs beginning in early 2020 presents challenges to using SFY 2020 and SFY 2021 base period for setting capitation rates. CY 2018 and CY 2019 reflect the most recent available claims data with sufficient payment runout prior to the impact of the COVID-19 pandemic. As we used the CY 2018 and CY 2019 base period data to project SFY 2023 costs, we also analyzed 2020 and 2021 data through October, to the extent it is useable, to help inform rate development assumptions and understand more recent costs. In an email dated May 11, 2022, CMS approved the state's request for an exception to the base data standards in accordance with 42 CFR § 438.5(c)(3) for SFY 2023 rate development purposes.

Consistent with the SFY 2022 rate development, we continue to rely solely on encounter data due to the decreasing impact of the FFS experience on the MCM capitation rates in recent years. However, we continue to apply an adjustment factor developed from FFS data to reflect the portion of the base experience data historically attributed to FFS. The adjustment factor is discussed in more detail later in this report.

A few years ago, DHHS discovered MCO payment issues related to crossover claims for skilled nursing facility (SNF) services. SNF claims are not a covered service under the MCM program. Using claim number information provided by MCOs, we identified approximately \$330,000 in the CY 2018 and CY 2019 base experience data attributed to improperly paid services, and we removed these claims from the capitation rate development process.

It is our understanding that Community Mental Health Centers (CMHCs) originally did not completely report certifications for GAHCP members eligible for enhanced behavioral health services, due to the lack of a capitation arrangement between the MCOs and CMHCs. As a result, the number of certifications received by DHHS increased rapidly beginning in June 2020.

To correct this issue and accurately assign rate cells for GAHCP members in the base data, we re-processed the enrollment data and assumed any GAHCP individual with a behavioral health certification in December 2020 should have had the certification in prior months. This change results in the reassignment of member months from GAHCP program rate cells to the behavioral health rate cells.

Consistent with the SFY 2022 capitation rates, the claims and enrollment base data for GAHCP members with behavioral health certifications is excluded from rate development due to underreported CMHC expenditures. Therefore, the base experience data for the behavioral rate cells is unchanged.

We did not identify any material concerns with the quality or availability of the data with respect to total claims in aggregate or our ability to allocate encounter data to major service categories. Our data reconciliation efforts are consistent with Actuarial Standard of Practice #23 on data quality.

The base experience data used in the development of the capitation rates does not include services invalidated by courts of law, or by changes in federal statutes, regulations, or approvals.

SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

This report assumes that the reader is familiar with the State of New Hampshire's Medicaid program, its benefits, and rate setting principles. This report was prepared solely to provide assistance to DHHS to set SFY 2023 capitation rates for the Medicaid Care Management program. It may not be appropriate for other purposes. Milliman does not intend to benefit and assumes no duty or liability to other parties who receive this work. This material should only be reviewed in its entirety.

Page 10

State of New Hampshire Department of Health and Human Services

METHODOLOGY OVERVIEW

We used the following methodology to develop the New Hampshire MCM program capitation rates:

- Summarize CY 2018 and CY 2019 MCO encounter experience data for the Standard Medicaid and GAHCP Medically Frail eligible population and the March 2019 to December 2019 MCO encounter data for the GAHCP Non-Medically Frail population.
- 2. Project estimated statewide SFY 2023 MCM program medical costs for all covered services by rate cell.
- 3. Blend the projected SFY 2023 costs based on CY 2018 and CY 2019 encounter data.
- 4. Adjust SFY 2023 projected medical costs for all rate cells for benefits not included in the base experience data, expected administrative expenses, margin, and premium tax.

Sections III and IV of this report document the rate setting methodology in detail.

COMMUNITY MENTAL HEALTH CENTER COST COMPONENT SUMMARY

Exhibit E provides DHHS and the MCOs with a summary of the CMHC cost component included in the SFY 2023 MCM capitation rates. This information can be used by MCOs and CMHCs to set a baseline for contract negotiations.

RETROACTIVE ACUITY ADJUSTMENT

Due to the uncertainty surrounding the unwind process following the end of the PHE, DHHS will implement a non-budget neutral retrospective acuity adjustment. During its call with the National Association of Medicaid Directors on April 8, 2022, CMS indicated states should work to appropriately capture population acuity before, during, and after redetermination, and that a retroactive acuity adjustment would be an appropriate way to handle the unknown acuity related to the PHE unwind. As such, we include a preliminary acuity assumption that will be updated following the contract year to reflect the actual population enrolled throughout SFY 2023.

We specifically intend to assign each enrolled individual a risk score to capture how the average risk of the departed population (those who disenroll following redetermination) compares to the continuing population (those who meet eligibility requirements during redetermination). Since all continuing members will meet largely the same eligibility requirements that existed in the CY 2018 and CY 2019 base period, it is appropriate to use this base period as a starting point to estimate the SFY 2023 costs of the continuing population. The retroactive acuity adjustment factor will then adjust the rates to account for the average SFY 2023 acuity by capturing the number of member months associated with departed members and their relative acuity compared to continuing members at the rate cell level.

In the initial rate development, we use DHHS' membership projections that include the PHE unwind, and we assume Standard Medicaid departed members will have a relative acuity of 0.750, while GAHCP departed members will have a relative acuity of 0.900. We selected these relativities, which are intended to be preliminary placeholders, because they produce an aggregate acuity adjustment impact on the rates that is generally consistent with the enrollment-based adjustment included in the SFY 2022 rates.

The inclusion of a retrospective acuity adjustment helps to mitigate the risk surrounding the end of the PHE by calculating final rates based on the actual enrolled population. This approach allows the MCOs to be at-risk for the actual members they cover.

The current enrollment projections provided by DHHS assume a return to pre-COVID-19 enrollment and acuity levels by March 1, 2023. In the event HHS continues to extend the PHE continues or a majority of redeterminations are not complete in time to accurately capture the relative acuity of the departed and continuing cohorts, we will consider re-evaluating the methodology to retroactively capture the acuity of the enrolled members in SFY 2023.

RISK ADJUSTMENT

The SFY 2023 capitation rates will use an actuarially sound prospective risk adjustment model to adjust the rates for each participating MCO in a manner that is budget-neutral to the state. A brief description of the methodology is provided below. The SFY 2023 risk adjustment process will be consistent with the current methodology in use for SFY

May 20, 2022

This report assumes that the reader is familiar with the State of New Hampshire's Medicaid program, its benefits, and rate setting principles. This report was prepared solely to provide assistance to DHHS to set SFY 2023 capitation rates for the Medicaid Care Management program. It may not be appropriate for other purposes. Milliman does not intend to benefit and assumes no duty or liability to other parties who receive this work. This material should only be reviewed in its entirety.

Page 11

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

2022. We do not anticipate any modifications will be made for SFY 2023 other than updating to version 6.5 of the CDPS+Rx risk weights.

The risk adjustment process will use the CDPS+Rx model to assign scored individuals to a demographic category and disease categories based on their medical claims and drug utilization during the study period. Only a single incidence of an International Classification of Disease Code (ICD) or drug is necessary to establish a CDPS+Rx disease category. We will exclude diagnosis codes related to radiology and laboratory services to avoid including false positive diagnostic indicators for tests run on an individual.

CDPS uses ICD-10 diagnosis codes to assess risk and assigns each member to one or more of 58 possible medical condition categories from 19 major diagnostic categories. Each member is also assigned to one of 11 age / gender categories. All of the 19 major diagnostic categories are "hierarchical" categories in that only the single most severe diagnosis within the major category is counted. This counting rule simplifies the model and strengthens its resistance to additional coding. Single counting within major categories is intended to avoid encouraging a proliferation of different diagnoses reported for a single disease process just to increase payment. However, as with most models, CDPS considers not only a person's single most serious diagnosis, but also diagnoses from other major categories.

The CDPS+Rx model includes the full set of diagnosis categories from the CDPS model, as well as 15 MRX categories from the Medicaid Rx model that are embedded within the CDPS hierarchy. The researchers at UCSD who developed the CDPS+Rx model decided to limit the MRX categories to the 15 that added predictive power to the diagnostic model (i.e., both relatively common and significant predictors of cost) and that were relatively less susceptible to variations in practice patterns.

We will use version 6.5 of the risk weights to score individuals with at least six months of eligibility and claims experience in the base data. Each scored individual receives a demographic relative cost weight and can have multiple disease categories assigned depending on that individual's health status. For each scored member, the weights for all of the disease categories assigned are combined with their demographic information to calculate a total individual risk score. We will continue to calculate customized risk adjustment weights to reflect the emerging cost to the treat the opioid addiction population.

Additionally, each scored member with less than 12 months of experience in the base data period will also be assigned a durational adjustment to compensate for missing diagnoses due to shorter enrollment durations, similar to a missing data adjustment. This additional adjustment will contribute to a more equitable allocation of resources as ACNH's members build up Medicaid experience to be used in the base experience data period for risk adjustment.

Finally, each unscored member will be assigned a demographic-only risk weight that considers the historical costs of unscored members. This provision of the risk adjustment methodology reflects the relative risk of unscored populations compared to scored populations within a rate cell. We are currently evaluating a separate adjustment for members that are auto-assigned to an MCO vs. members who choose an MCO.

For the SFY 2023 contract period, we will continue using the actual enrollment by MCO in each quarter to calculate risk scores in order to capture actual membership changes for each MCO.

For each MCO, the unadjusted plan factors for scored members are derived by performing a weighted average of the cost weights using the count of scored adults and children associated with each CDPS+Rx category. The budget neutral plan factor for scored members is calculated by dividing each individual MCO's unadjusted plan factor by the total population's unadjusted plan factor.

Similarly, the demographic only risk scores for the unscored population is normalized, by rate cell, across MCOs. We will analyze historical cost differences between scored and unscored members and introduce an adjustment to reflect any observed differences (i.e., if unscored members are shown to be less expensive than scored members, we will adjust risk scores accordingly).

The final risk score for each MCO and rate cell will be the weighted average risk scores between the scored and unscored population. These final risk scores will be used to adjust payments made to MCOs.

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program Page 12

May 20, 2022

RISK PROTECTION STRUCTURE

The risk mitigation structure described below is consistent with the structure included in the SFY 2022 capitation rates.

Boston Children's Hospital Risk Pool

Historical spend at Boston Children's Hospital (BCH) has varied significantly between the two MCOs in CY 2018 and CY 2019. With the recent addition of a third MCO, there is additional uncertainty related to the distribution of BCH services among the MCOs.

In order to better allocate funds across MCOs for children's specialty services provided at BCH, DHHS created a budget-neutral risk pool to redistribute funds at the end of each contract period based on actual expenditure levels. Inpatient and outpatient facility services provided at BCH will qualify for the risk pool calculation. Below we outline the approach for developing and administering this risk pool:

- At the beginning of each contract period, Milliman and DHHS will calculate the BCH funding level as a PMPM amount by rate cell and estimate the total risk pool amount based on projected enrollment. This amount will be trued-up at the end of the contract period based on actual enrollment by rate cell and risk adjustment results.
- At the end of the contract period, Milliman and DHHS will collect detailed BCH encounter claims data. Milliman
 and DHHS will calculate final BCH paid claim amounts by MCO.
- The final risk pool amount will then be reallocated based on a distribution of paid claims at BCH across MCOs.
- All funds will be allocated. In the case of a surplus, an MCO may receive funds in excess of their actual expenditures. In the case of a deficit, a MCO may receive less funds than their actual expenditures.
- Since the capitation rates fund these services using actual historical paid claims rather than relative to the Medicaid fee schedule, the settlement of the risk pool will also use actual paid claims rather than re-priced claims.

In order to avoid unintended interactions between the risk pool funding reallocation process and the stop-loss provision of the MCO contract, the high-cost patient stop-loss calculation will exclude BCH claims.

High-Cost Patient Stop-Loss

Effective September 1, 2015, DHHS implemented a stop-loss arrangement for high-cost patients where DHHS will share 50% of the cost above \$500,000 for members that have total expenses valued at Medicaid fee levels above that threshold. In order to avoid unintended interactions between the BCH risk pool funding reallocation process and the stop-loss provision of the MCO contract, the high-cost patient stop-loss excludes BCH claims.

The stop-loss attachment point of \$500,000 is indexed annually at a rate of 3.0% from its inception in SFY 2016 and rounded to the nearest \$1,000. The SFY 2023 attachment point will be \$615,000. The indexing of the attachment point addresses historical inflation and recent provider reimbursement increases and maintains a level of MCO risk consistent with program goals.

Minimum and Maximum MLR Provision

On page 9 of the 2022-2023 Medicaid Managed Care Rate Development Guide, "CMS recommends all states implement a 2-sided risk mitigation strategy for rating periods impacted by the public health emergency." Due to population and cost uncertainty related to the ongoing PHE and the anticipated enrollment redetermination process, we understand DHHS intends to maintain the minimum and maximum medical loss ratio (MLR) provision for SFY 2023. The minimum and maximum MLR provision will continue to use the "regular" MLR definition (not the federal MLR definition) consistent with the rate setting process.

The MCM capitation rates reflect a target MLR, which measures the projected medical service costs as a percentage of the total at-risk MCO capitation rates. The minimum MLR will limit MCO gains if the actual MLR is lower than the minimum MLR. The maximum MLR will limit MCO losses if the actual MLR is higher than the maximum MLR. The

Page 13

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

target MLR for at-risk services is 90.8% for Standard Medicaid and 90.1% for GAHCP based on the SFY 2023 projected enrollment distribution. Please note, each program's target MLR may change in future rate amendments as a result of changes to underlying assumptions, such as enrollment projections, emerging utilization experience, and the retroactive acuity adjustment.

The minimum MLR is set on a program-wide basis for each major population, such that maximum profit achievable is 4%, which is equal to the 1.5% target margin plus the amount between the target MLR and the minimum MLR (2.5%). Based on the target MLRs in this report, the minimum MLR will be 88.3% for the Standard Medicaid population and 87.6% for the GAHCP population. The maximum MLR is also set on a program-wide basis for each major population 3.5% above the target MLR, such that MCOs will have a maximum loss of 2.0%. Based on the target MLRs in this report, the maximum MLR dedicaid population and 93.6% for the GAHCP population. The final minimum MLR for Standard Medicaid and GAHCP will be updated for the final retroactive acuity factor and any other changes implemented in the expected January 1, 2023 contract amendment.

There are several operational requirements related to administering the minimum and maximum MLR settlement:

- The settlement will be done separately for the Standard Medicaid and GAHCP populations.
- The retroactive acuity adjustment will apply to the rates prior to the MLR settlement and prior to other risk
 mitigation provisions.
- Other MCM program risk mitigation provisions will apply prior to the minimum and maximum MLR settlement (i.e., Boston Children's Hospital risk pool, high-cost patient stop-loss arrangement, and prospective risk adjustment).
- The numerator of each MCO's actual MLR will include all payments made to providers, such as fee-for-service payments, sub-capitation payments, incentive payments, and settlement payments. The numerator of each MCO's actual MLR will <u>not</u> include costs related to quality improvement activities or fraud, waste, and abuse prevention.
- Payments and revenue related to directed payments and premium taxes will be excluded from the numerator and denominator of each MCO's actual MLR.
- Payments related to the withhold and incentive program will be excluded from the minimum and maximum MLR settlement.
- The minimum and maximum MLR settlement will occur after the contract year is closed and sufficient paid claims runout is available.

May 20, 2022

III. MEDICAL COST PROJECTIONS

This section of the report describes the projection of the MCO encounter data for all rate cells.

We used the following methodology to project the encounter data used in the calculation of the medical component of the capitation rates:

- 1. Summarize base period MCO encounter experience data for the population eligible for and the services covered by the MCM program.
- 2. Apply adjustments to the base data to project SFY 2023 medical costs.
- 3. Blend the projected base period experience data.

Each of the above steps is described in detail below.

STEP 1: SUMMARIZE BASE EXPERIENCE DATA

In this step, we summarize the encounter base experience by rate cell and service category for the populations eligible to enroll in the MCM program.

Base Data

We summarized detailed CY 2018 and CY 2019 MCO encounter claims data for the Standard Medicaid and GAHCP Medically Frail eligible population and the March 2019 to December 2019 MCO encounter data for the GAHCP Non-Medically Frail population with the following specifications:

- The cost and utilization data reflect the claim header information for claims paid at the header level and line item detail for claims paid at the detail level
- Claims for FQHC and RHC providers reflect their normal prospective per encounter rates
- Prescription drug claims reflect gross ingredient cost and dispensing fees prior to any pharmacy rebates
- We excluded all Part D covered drugs for dual eligibles
- We no longer exclude claims and eligibility data for the portion of any month when an individual had a stay longer than 15 days in an institution for mental disease (IMD). However, we continue to exclude stays over 60 days.
- We excluded hemophilia and certain other high-cost drugs (i.e., Carbaglu, Ravicti, and Zolgensma), as they
 are carved out of the MCM program
- We excluded encounter data for the GAHCP behavioral health population due to completion concerns and discrepancies with the current benefit package

Non-Covered Services Adjustment

MCOs are allowed to provide services not explicitly covered under the MCM program to beneficiaries' in-lieu of a covered service. As part of the capitation rate development process, the encounter data must be adjusted to remove any portion of the cost of in-lieu of services that exceeds the cost of the corresponding covered service.

MCOs currently provide Medical Nutrition and Diabetes Self-Management services defined by Procedure Codes 97802, 97803, and G0108 with an average unit cost of \$19.88 per unit using staff nutritionists for the Standard Medicaid population. For the GAHCP population, the average unit cost is \$18.99 per unit using staff nutritionists. Alternatively, these services would be provided as a covered service by a physician in an office setting at the cost of \$35.71 per unit (based on the Medicaid fee for 99202-office visit). This comparison shows that Medical Nutrition and Diabetes Self-Management services are cost effective. Therefore, we did not make any adjustment to the base period data for non-covered services.

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

Retroactive Eligibility and Enrollment Lag

Recipient enrollment in the FFS program can and does occur retroactively. When an individual applies and qualifies for Medicaid coverage, DHHS reimburses claims that occurred during the retroactive qualification period prior to their application. DHHS backdates the eligibility of the individual to accommodate the retroactive coverage.

The MCOs do not cover these retroactive enrollment periods. Retroactive claims were included in the FFS data provided by DHHS. The enrollment data provided by DHHS excluded retroactive enrollment periods; therefore, a special adjustment was not necessary because we only summarized claims for individuals with non-retroactive enrollment records. As a result, the FFS claims included in this analysis are reflective of the claims and time period that MCOs will cover. Additionally, the encounter data already excludes the retroactive eligibility period and enrollment lag so no adjustments were needed.

Eligibility Category Assignment

We grouped Medicaid enrollees into the SFY 2023 rate cell structure using the following hierarchy based on their eligibility status on the first day of each eligibility span:

- Behavioral health rate cells
- GAHCP rate cells
- Standard Medicaid rate cells

The behavioral health rate cells are structured around the target populations eligible for enhanced behavioral health services at the CMHCs. These individuals were historically identified through the following modifier codes found on CMHC FFS claims:

- Modifier 1 is HW for all BBH eligible patients
- Modifier 2 is U1 for Severe / Persistent Mental Illness (SPMI)
- Modifier 2 is U2 for Severe Mental Illness (SMI)
- Modifier 2 is U5 for Low Utilizer
- Modifier 2 is U6 for Serious Emotionally Disturbed Child (SED)
- Modifier 2 is U7 for Serious Emotionally Disturbed Child, Interagency Involvement (SED-I)

To identify the behavioral health population enrolled in an MCO, we used a separate data file provided by DHHS that includes behavioral health certification information for each individual. We separated each group into rate cells based on Medicare eligibility as follows:

- Severe / Persistent Mental Illness (SPMI) Medicaid Only
- Severe / Persistent Mental Illness (SPMI) Dual Eligibles
- Severe Mental Illness (SMI) Medicaid Only
- Severe Mental Illness (SMI) Dual Eligibles
- Low Utilizers Medicaid Only
- Low Utilizers Dual Eligibles
- Serious Emotionally Disturbed Child (SED) Medicaid Only and Dual Eligibles

The GAHCP population rates cells are defined using the MGIM and MGIA eligibility categories for the Medically Frail and Non-Medically Frail populations, respectively.

We identified and classified all other Standard Medicaid population using the eligibility codes found in the enrollment file. Please refer to our rate cell logic letter dated January 16, 2020 for more details.

The various eligibility groups have been further segmented into rate cells based on analysis of the per capita costs of various population sub-groups. These sub-groups were defined by age within the covered eligibility categories and grouped into rate cells based on similarities in average costs. There are three additional kick payments made for maternity costs and newborn costs.

The population identification is done on a first of eligibility span basis, consistent with capitation rate payment from MMIS under the next day enrollment process.

Page 16

May 20, 2022

State of New Hampshire Department of Health and Human Services

SFY 2023 Capitation Rate Development for Medicaid Care Management Program

We also proactively assigned "Pending Duals" to their respective dual rate cell. Based on DHHS' review of several cases, DHHS ultimately retroactively assigns a dual eligible status to these individuals in more than 90% of cases.

The eligibility category assignment process is consistent with the guidance provided in the Actuarial Standards of Practice #12 for risk classification.

Maternity Kick Payment Definition

The maternity kick payment includes all facility and professional claims associated with deliveries. Maternity kick payment cases are counted as women who have <u>either</u> a maternity delivery DRG <u>or</u> a physician maternity delivery claim (or both). The maternity kick payment only includes women already enrolled in Medicaid at time of delivery.

The maternity kick payment cases are distributed in the following manner:

- Both a maternity delivery DRG and a physician claim = 87%
- A maternity delivery physician claim only = 12%
- A maternity delivery DRG only = 1%

We used the following criteria to identify claims information to calculate the maternity kick payment.

- Hospital inpatient services with DRG codes of:
 - Prior to DRG version 37: 765 768, 774 775
 - DRG version 37: 783 788, 796 798, 805 807
- Hospital outpatient and professional services with a primary diagnosis code of:
 - ICD10: O60 O82, and Z37

This would include delivery related anesthesia services (CPT-4 codes 00850, 00857, 00946, 00955, 01960, 01961, 01967, 01968), as long as a relevant primary diagnosis code is present.

Delivery and post-partum care services (CPT-4 codes 59400, 59409, 59410, 59430, 59510, 59514, 59515, 59610, 59612, 59614, 59618, 59620, 59622)

Prenatal and postpartum services are not included in the maternity kick payment. These services are included in the rate cell corresponding to the person receiving the services. For global reimbursement CPT-4 codes, such as 59400, 59510, 59610, and 59618, we allocated the total allowed and paid amounts to prenatal and delivery services using RBRVS ratios corresponding to the codes without prenatal care: 59410, 59515, 59614, and 59622, respectively. The prenatal care allocation is then assigned to the corresponding monthly rate cell.

Neonatal Abstinence Syndrome Kick Payment Definition

The neonatal abstinence syndrome (NAS) kick payment is provided for babies who have a diagnosis code of P96.1 (ICD10) in their birth month on an inpatient claim. The regular newborn kick payment is provided for all other babies.

Hospital Inpatient Psychiatric Admission Kick Payment Definition

Effective July 1, 2021, DHHS transferred the funding for hospital inpatient psychiatric admissions to a kick payment structure designed to promote the expansion of Designated Receiving Facility (DRF) bed capacity to eliminate emergency department psychiatric boarding. MCOs receive a per admission kick payment from DHHS for every applicable hospital inpatient psychiatric admission.

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

The hospital inpatient psychiatric admission kick payment structure includes four different kick payments corresponding to the hospital peer group assigned to the admission in New Hampshire's FFS payment structure:

- Peer Group 01 and 07
- Peer Group 02
- Peer Group 06
- Peer Group 09

The hospital inpatient psychiatric admission kick payment amounts are developed from CY 2018 and CY 2019 DRG distributions for DRGs 880 to 887 in conjunction with the most recent DRG table. Within each peer group, we calculated a composite base DRG price using this distribution and per admit price. The hospital inpatient psychiatric admission kick payment amounts include a 1.5% allowance for risk margin and a 2.0% allowance for premium tax. Related administrative costs are funded through the administrative allowance included in the monthly rate cell capitation payments.

The hospital inpatient psychiatric admission kick payments do not apply to dually eligible members or any admissions at New Hampshire Hospital or Hampstead Hospital. These admissions are funded through the regular monthly capitation rates.

Clarification Regarding Application of DRF Minimum Fee Schedule Arrangement

Effective July 1, 2021, DHHS implemented a minimum fee schedule arrangement that mandates MCOs pay all DRF admissions at a rate no less than the applicable New Hampshire Medicaid fee-for-service DRG fee schedule. All DRF admissions are categorized under Peer Group 06.

STEP 2: APPLY ADJUSTMENTS TO THE BASE EXPERIENCE DATA TO PROJECT SFY 2023 MEDICAL COSTS

In this step, we apply adjustment factors to reflect differences between the base period data and the projected SFY 2023 MCM program medical costs. We explain each adjustment factor in detail below.

Appendices B1 through B2 show adjusted and trended values for each rate cell along with the detailed adjustment factors by service category.

IBNR Adjustment

We set completion factors (CFs) by eligibility category and major service category for claims incurred in the base experience period to 1.000, since the majority of services have a significant amount of run-out for both CY 2018 and CY 2019 claims experience.

We applied a 1.05 underreporting adjustment to the encounter base experience data for CMHC services only. We developed the underreporting adjustment in order to correct for data reporting issues between the CMHCs and MCOs as we understand there is not an actual reduction in services provided by the CMHCs.

Reimbursement Adjustment

We reviewed the MCO provider reimbursement levels in the base experience period as a percentage of Medicaid fees to better understand MCO provider contracts and payment levels. We determined that payment levels in the base experience period were relatively consistent with previous assumptions. We maintained historical reimbursement levels compared to Medicaid fees for all services.

Hospital Inpatient Services

We used the DRG rate table effective October 2021 in conjunction with admit distributions by rate cell to develop reimbursement adjustment factors. We also made a trend adjustment to account for known changes in reimbursement levels in SFY 2023. For hospital inpatient services, we estimated an October 2022 DRG weight update of 0.45%, which we intend to update in the January amendment to capture the actual October 2022 information. We applied the 0.45% DRG weight increase as a 0.34% adjustment to trend, since it impacts only the last nine months of SFY 2023. Table 8 shows a summary of the impact of the hospital inpatient pricing adjustment for each of the base data periods:

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

Table 8 New Hampshire Department of Health and Human Services Medicaid Care Management Program Impact of Repricing for Hospital Inpatient Services Average Impact for All Rate Cells Combined		
Base Period Data	Total Impact	
Standard Medicaid Encounter Year 1	1.0882	
Standard Medicaid Encounter Year 2	1.0643	
Medically Frail Encounter Year 1	1.0926	
Medically Frail Encounter Year 2	1.0659	
Non-Medically Frail Encounter Year 2	1.0670	

We set the funding level for BCH hospital inpatient services consistent with the CY 2019 experience of \$10.8 million annually, then adjusted for incurred but not reported claims liability using BCH inpatient services specific completion patterns and trended to SFY 2023 for hospital operating cost increases. We adjusted the base data to include a number of recent retrospective MCO claims payments to BCH, as well as more recent BCH experience from the latest encounter data information through October 2021. We also made a 2.1% adjustment to reflect enrollment growth in the updated enrollment projections for SFY 2023 compared to the base period data.

Please refer to Exhibit C for more details.

Table 9 below shows a summary of unadjusted hospital inpatient paid claims experience at BCH from CY 2018 to CY 2020:

Table 9 New Hampshire Department of Health and Human Services Medicaid Care Management Program Hospital Inpatient Service Paid Claims at Boston Children's Hospital FFS and MCO Encounter Data	
Base Period Data	Total Paid
CY 2018	\$8,383,807
CY 2019	10,829,050
CY 2020	13,417,174

The hospital inpatient reimbursement factors are adjusted to reflect erroneous payments made during the base period by one of the MCOs to St. Joseph of Nashua Hospital and Mass General Hospital for facility services.

Hospital Outpatient Services

We applied a 1.063 reimbursement adjustment for hospital outpatient services for all populations to account for the 3.1% Medicaid provider rate increase effective January 1, 2020 and January 1, 2021, as mandated by House Bill 4, Section 348. We address increases to cost-based reimbursement in the trend development section of this report. We also address cost-based settlements, including enhanced payments to Boston Children's Hospital, in the hospital outpatient cost settlement adjustment section of this report.

For BCH outpatient services we also set the funding at level consistent with the CY 2019 experience of \$4.2 million annually, then adjusted for incurred but not reported claims liability using BCH inpatient services specific completion patterns and trended to SFY 2023 for hospital operating cost increases. Please refer to Exhibit C for more details. We adjusted the base data to include a number of recent retrospective MCO claims payments to BCH, as well as more recent BCH experience from the latest encounter data information through October 2021. We also made a 2.1% adjustment to reflect enrollment growth in the updated enrollment projections for SFY 2023 compared to the base period data.

Table 10 below shows a summary of unadjusted hospital outpatient paid claims experience at BCH from CY 2018 to CY 2020:

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

Table 10 New Hampshire Department of Health and Human Services Medicaid Care Management Program Hospital Outpatient Service Paid Claims at Boston Children's Hospital FFS and MCO Encounter Data	
Base Period Data	Total Paid
CY 2018	\$2,668,550
CY 2019	4,204,125
CY 2020	3,540,528

The hospital outpatient reimbursement factors are adjusted to reflect erroneous payments made during the base period by one of the MCOs to St. Joseph of Nashua Hospital and Mass General Hospital for facility services.

Professional and Other Services

The reimbursement adjustment factors for professional and other services are based on a comparison of the Medicaid fees effective during the base period to the most recent fee schedules as of January 1, 2022. We also incorporated known fee changes not included in the current fee schedules. For each CPT and HCPCS code, we compared the current FFS amount to the corresponding amount in the fee schedules effective on the date of service. We then summarized the data by rate cell and service category to obtain the adjustment factors. The most significant changes are related to the 3.1% Medicaid provider rate increase effective January 1, 2020 and January 1, 2021, and the DME minimum fee schedule (described in a separate section below).

Table 11 shows a summary of the impact of the reimbursement change for professional and other services. There is additional variation by rate cell based on the services used by the population in each rate cell.

Table 11 New Hampshire Department of Health and Human Services Medicaid Care Management Program Impact of Repricing for Professional and Other Services Average Impact for All Rate Cells Combined		
Base Period Data	Total Impact	
Standard Medicaid Encounter Year 1	1.0952	
Standard Medicaid Encounter Year 2	1.0843	
Medically Frail Encounter Year 1	1.0992	
Medically Frail Encounter Year 2	1.0790	
Non-Medically Frail Encounter Year 2	1.0732	

Please refer to Section IV of this report for details on other fee schedule changes not included in the reimbursement adjustment.

Minimum DME Fee Schedule

Effective January 1, 2020, DHHS implemented a minimum DME fee schedule to help create a sustainable environment for the local DME providers. The minimum fee schedule was set at 80% of the current Medicaid DME fee schedule. We used average unit cost from the CY 2020 encounter data to reprice the CY 2018 and CY 2019 base experience data at the HCPCS code level. The CY 2020 data reflects all recent changes to the DME fee schedule and is a reliable source to use to estimate the impact of the minimum fee schedule.

The DME minimum fee schedule also includes a number of codes previously priced manually as part of a DHHS effort to reduce the number of manually priced codes.

DHHS completed and submitted a pre-print as required by 438.6(c), which is pending approval from CMS.

Page 20

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

Federally Qualified Health Center Services

We adjusted claims for Federally Qualified Health Centers (FQHCs) and Rural Health Clinics (RHCs) to reflect the most recent per encounter rate at each facility as of July 1, 2022. The actual rate increase is 2.1% from the July 1, 2021 encounter rates to the SFY 2023 encounter rates for most FQHCs and RHCs.

Medical Trend from Base Period to SFY 2023

Since the start of the COVID-19 pandemic and PHE, service utilization levels in the New Hampshire Medicaid program have lagged behind CY 2019 utilization by rate cell and on an aggregate basis. The SFY 2023 capitation rates assume SFY 2023 utilization levels will be consistent with CY 2019. As such, we did not apply utilization trend from CY 2019 to SFY 2023 with the exception of ABA services. This utilization projection is supported by our review of historical quarterly PMPM amounts that indicate emerging costs through June 2021 were below CY 2019 even with the known provider rate increases and acuity differences. We also reviewed DHHS summaries of emerging encounter data through January 2022 that show similar cost reductions compared to CY 2019. We will continue to monitor emerging experience and revisit this assumption for the planned January 2023 rate amendment.

Table 12 below summarizes the annual medical trend rate assumptions by major service category:

Table 12 New Hampshire Department of Health and Human Services Medicaid Care Management Program Annual Trends from Base Period to SFY 2023		
Service Category	Utilization Trend CY 2018 to CY 2019 Only	Unit Cost Trend
Hospital Inpatient	0.00%	0.34% ¹
Hospital Outpatient	2.00%	3.10%
Professional	1.00%	0.00%
Community Mental Health Center	1.00%	0.00%
Other Services	1.00%	0.00%

¹ Unit cost trend for hospital inpatient services is applied as a one-time allowance for the expected 0.45% increase in DRG reimbursement on October 1, 2022 (not as an annual trend rate).

Although hospital inpatient, professional, and other services are repriced using the October 1, 2021 DRG rate table and the CY 2022 fee schedule, we also made a trend adjustment to account for expected changes in reimbursement levels in SFY 2023. For hospital inpatient services, we estimated the expected October 2022 DRG weight update to increase 0.45% based on our review of historical DRG weights. We applied the 0.45% DRG weight increase as a 0.34% adjustment since it impacts only the last nine months of SFY 2023. For professional and other services, we assumed no fee schedule changes would be implemented through June 2023 other than those discussed in this report.

Hospital outpatient reimbursement changes are tied to changes in each hospital's operating cost. We developed the 3.1% annual trend for hospital outpatient services by reviewing the average annual change in the Bureau of Labor Statistics (BLS) Producer Price Index (PPI) for hospital services from CY 2018 to October 2021 (Series ID PCU622---622---). The Hospital PPI is a measure of hospital revenue changes that can also be used as a proxy for operating cost changes because the national average operating margins for hospitals are relatively stable from year to year.

Note, the utilization and unit cost trends in Table 12 exclude trends related to the opioid addiction treatment population, which are shown as a separate adjustment and are described later in this report.

For Applied Behavioral Analysis (ABA) services, we reviewed historical data through June 2021 to develop separate utilization and unit cost trends. With MCOs expanding their ABA networks in recent years, we looked at changes in utilization and cost for members with autism receiving ABA services over time. We selected an overall utilization trend of 17.4% (varies by rate cell) and a 4.4% trend for unit cost. We then trended Q2 2021 experience data at these annual rates to develop our projected SFY 2023 costs PMPM. The resulting annual PMPM trend rate from CY 2019 to SFY 2023 is 21.7%.

Page 21

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

Table 13 below shows the rate cell specific ABA trends used.

Table 13 New Hampshire Department of Health and Human Services Medicaid Care Management Program ABA Annual Trends from Base Period to SFY 2023				
Service Category	Utilization Trend	Unit Cost Trend		
Low Income Children	17.6%	3.7%		
CHIP	18.7%	4.6%		
Foster Care / Adoption	18.0%	5.0%		
Severely Disabled Children	17.9%	6.9%		
Elderly and Disabled Adults	-4.4%	-0.7%		
Duals	-7.4%	-0.5%		
SED Children	25.9%	4.4%		

Prescription Drug Trend from Base Period to SFY 2023

Pharmacy trend assumptions are based on a combination of historical New Hampshire Medicaid data analysis, Milliman research on utilization and cost trends, and publicly available trend reports and forecasts.

Our prescription drug trend model uses January 2018 to October 2021 experience (March 2019 to October 2021 for GAHCP Non-Medically Frail) as the base period for our projections. Given the constantly changing prescription drug market, it is critical to project trends using the most current available data.

The final trends are calculated as the ratio of the average drug costs in the projection period compared to the average drug costs in the rate setting base period. Trends were calculated for brand, generic, and specialty drugs, separately for utilization and unit cost. Projected values are estimated using the prescription drug base period data as a starting point and applying anticipated shifts and trends. Each component of pharmacy trend is documented below.

Since hemophilia and other high-cost drugs (i.e., Carbaglu, Ravicti, and Zolgensma) are carved out of the MCM program for SFY 2023, we excluded these drugs from our prescription drug trend development. Exhibit H contains a list of NDCs for the excluded drugs.

Tables 14A and 14B show a summary of annual PMPM prescription drug trends by eligibility grouping.

Table 14A New Hampshire Department of Health and Human Services Medicaid Care Management Program Annual PMPM Prescription Drug Trends CY 2018 to CY 2019					
Eligibility Grouping	Generic	Brand	Specialty	Hepatitis C	Total
Low Income Children	-3.4%	-7.2%	39.3%	-82.2%	12.3%
Low Income Adults	14.5%	-6.7%	23.9%	-47.5%	-2.6%
Foster Care / Adoption	-11.5%	4.4%	15.5%	0.0%	1.0%
Severely Disabled Children	0.0%	29.5%	45.1%	0.0%	31.6%
Elderly and Disabled Adults	10.9%	1.0%	2.7%	-12.7%	2.5%
Behavioral Health Adult	-4.6%	-12.2%	0.8%	-54.3%	-12.8%
SED Children	-6.8%	-6.2%	-4.0%	0.0%	-5.9%
Duals	0.2%	12.8%	0.0%	0.0%	4.0%
Medically Frail	29.6%	6.8%	44.2%	-18.8%	13.7%
Non-Medically Frail	N/A	N/A	N/A	N/A	N/A

State of New Hampshire Department of Health and Human Services

SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

Table 14B New Hampshire Department of Health and Human Services Medicaid Care Management Program Annual PMPM Prescription Drug Trends CY 2019 to SFY 2023					
Eligibility Grouping	Generic	Brand	Specialty	Hepatitis C	Total
Low Income Children	1.2%	1.4%	15.8%	63.8%	9.1%
Low Income Adults	3.9%	-1.4%	12.0%	-9.9%	3.2%
Foster Care / Adoption	-3.2%	5.0%	16.6%	0.0%	6.5%
Severely Disabled Children	13.9%	9.8%	4.7%	0.0%	7.9%
Elderly and Disabled Adults	1.0%	3.2%	9.3%	-18.6%	4.4%
Behavioral Health Adult	-1.8%	4.0%	9.9%	-17.9%	3.8%
SED Children	-6.5%	-0.1%	22.0%	0.0%	4.6%
Duals	-0.3%	1.4%	0.0%	0.0%	13.0%
Medically Frail	4.3%	0.0%	11.9%	-25.1%	2.3%
Non-Medically Frail	5.4%	4.4%	9.8%	-9.9%	5.3%

Working with CY 2020 and CY 2021 data presented a number of challenges that required several data and process adjustments. We observed suppressed prescription drug utilization during CY 2020 and CY 2021 for many rate cells due to the pandemic, but we do not expect these reductions to continue when the PHE ends and enrollment returns to pre-PHE levels. We also observed a shift in prescribing patterns that increased the number of 90-day supply prescriptions, making the differentiation between utilization and unit cost trends less clear. As such, we developed PMPM trends for prescription drugs and then applied a 0% utilization trend accompanied by a unit cost trend equal to the PMPM trend. This anticipates that utilization in SFY 2023 will not materially vary from the base period (beyond what is captured separately in the acuity adjustment) and recognizes the pandemic may have had an effect on prescription drug utilization counts. Table 15 shows the resulting actual and projected PMPM by eligibility grouping and calendar year.

Table 15 New Hampshire Department of Health and Human Services Medicaid Care Management Program Annual Calendar Year Prescription Drug PMPM Actual Amounts and Projections						
Eligibility Grouping	2018	2019	2020	2021	2022 (proj)	2023 (proj)
Low Income Children	\$31.81	\$35.69	\$38.82	\$39.27	\$46.65	\$49.92
Low Income Adults	\$95.66	\$100.91	\$96.92	\$100.91	\$113.15	\$120.08
Foster Care / Adoption	\$78.11	\$78.86	\$71.94	\$79.83	\$96.01	\$100.83
Severely Disabled Children	\$257.17	\$338.50	\$361.62	\$388.81	\$425.84	\$455.42
Elderly and Disabled Adults	\$393.91	\$406.20	\$439.80	\$442.71	\$468.95	\$496.85
Behavioral Health Adults	\$381.73	\$355.30	\$365.70	\$401.91	\$412.56	\$430.57
SED Children	\$96.10	\$90.45	\$99.45	\$95.54	\$102.95	\$108.25
Duals	\$1.11	\$1.16	\$1.45	\$1.41	\$1.74	\$1.84
Medically Frail	\$209.85	\$254.56	\$282.04	\$268.46	\$294.96	\$314.66
Non-Medically Frail	\$0.00	\$91.92	\$101.66	\$106.38	\$110.97	\$117.86

Managed Care Savings Adjustment

Due to the change in the care management requirements in the MCO contract effective January 1, 2021, we used the same managed care savings adjustment factors as in the SFY 2022 capitation rates. These savings assumptions are based on information the MCOs presented in their responses to the 2018 MCM program request for proposal (RFP).

The RFP responses included projected savings for utilization management, care coordination and care management, and behavioral health related managed care activities. Effective January 1, 2021, the contract changed from requiring 15% of the MCM population to enroll in a care management program to only requiring 3% to enroll. As such, we only considered 20% of total savings projected by the MCOs for the care coordination and care management of

State of New Hampshire Department of Health and Human Services

SFY 2023 Capitation Rate Development for Medicaid Care Management Program

their RFP responses. This results in total estimated SFY 2023 savings of approximately \$3.00 PMPM, relative to the historical MCM encounter data.

Preferred Drug List Adjustment

Effective September 1, 2019, management of the PDL moved from MCO control to DHHS control. The State PDL includes a listing for approximately 84 therapeutic classes defined by DHHS. Approximately 78% of the total drug spend is in therapeutic classes managed in the PDL. We summarized the MCO encounter data across 495 therapeutic classes covering approximately 5,913 different medications. For each medication within a particular therapeutic class, we calculated the average cost per script, as well as its market share within that therapeutic class. We implemented shifting adjustments for off-formulary, non-preferred, and preferred drug modeling categories based on how we expect the state to manage the PDL in the future. Each adjustment applies to the market share within each therapeutic class, or cost per script, in projecting total gross drug spend.

As a result of the change in PDL control, the gross drug cost decreases compared to the previous MCO-controlled PDLs. State controlled PDLs in other states typically include brand name drugs almost exclusively, which results in a higher gross prescription drug cost than under a MCO controlled PDL. However, DHHS' PDL includes several generic options in drug classes with high expenditures because state law requires generics to be offered.

The PDL adjustment includes the impact of utilization shifts from the brand drug to new generic alternatives resulting from patent loses when the generic alternative is included on the PDL. Our utilization shift assumptions are based on Milliman research of how quickly historic brand utilization converts to generic in each month after a patent expires. Similarly, we used assumptions for what the cost of the new generic drug would be relative to the current brand drug price.

The biosimilar Semglee (insulin glargine-yfgn) recently launched in November of 2022. The manufacturer launched the same drug at two different price points:

- 1. Semglee-yfgn that is priced at ~80% of brand Lantus cost.
- 2. insulin glargine-yfgn that is priced at ~34% of brand Lantus cost.

The state PDL also recently updated to prefer insulin glargine-yfgn, and non-prefer Semglee-yfgn. To account for this new product, we shifted 7.5% of total utilization for Lantus to this new biosimilar drug.

The PDL adjustment also includes an allowance for MCO supplemental rebates. The MCOs have historically reported approximately 3.25% of gross cost as supplemental rebate revenue for a PDL fully controlled by the MCOs. Based on more recent MCO data, we estimate that MCO supplemental rebates will decrease in aggregate to 1.1% after the PDL changes to state control.

Table 16 shows the PDL adjustment factors in aggregate by rate cell grouping and year:

Table 16 New Hampshire Department of Health and Human Services Medicaid Care Management Program Preferred Drug List Adjustment				
Rate Cell Grouping	CY 2018	CY 2019		
Low Income Children ¹	0.984	0.995		
Low Income Adults	0.931	0.937		
Foster Care / Adoption	0.989	0.976		
Severely Disabled Children	0.979	0.986		
Elderly and Disabled Adults	0.956	0.965		
Dual Éligibles	1.000	1.000		
SPMI / ŠMI / LU	0.954	0.959		
Serious Emotionally Disturbed Child	0.989	0.993		
Medically Frail	0.939	0.943		
Non-Medically Frail	N/A	0.959		

¹ Low Income Children includes CHIP.

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

The adjustment factors include utilization shifts from the brand drug to new generic alternatives for brand drugs that lost their patent between the base experience period and the PDL effective date if the generic equivalent is on the PDL. Our utilization shift assumptions are based on Milliman research of how quickly historic brand utilization converts to generic in each month after a patent expires. Similarly, we used assumptions for what the cost of the new generic drug would be relative to the current brand drug price. The overall impact of the PDL adjustment is a 1.2% decrease to total MCM service costs.

Exhibit J provides a summary of the PDL analysis for selected therapeutic classes.

Medicare Covered Opioid Treatment Adjustment

Effective January 1, 2020 Medicare began paying for Opioid Treatment Programs (OTPs) through bundled payments for opioid use disorder (OUD) treatment services, including medication-assisted treatment (MAT), toxicology testing, and counseling. We estimated the reduction in cost to the MCM program by modeling the Medicaid coordination of benefit with Medicare consistent with guidance issued by DHHS to the MCOs.

High-Cost Patient Stop-Loss Adjustment

Effective September 1, 2015, DHHS implemented a stop-loss arrangement for high-cost patients where DHHS shares 50% of the cost above \$500,000 for members that have total expenses valued at Medicaid fee levels above that threshold. In order to avoid unintended interactions between the BCH risk pool funding reallocation process and the stop-loss provision of the MCO contract, the high-cost patients stop-loss excludes BCH claims.

The stop-loss attachment point of \$500,000 is indexed annually at a rate of 3.0% from its inception in SFY 2016 and rounded to the nearest \$1,000. The SFY 2023 attachment point will be \$615,000. The indexing of the attachment point addresses historical inflation and recent provider reimbursement increases and maintains a level of MCO risk consistent with program goals.

The resulting adjustment factors reduce the base experience data by about 0.05% to reflect the MCO liability under the stop-loss agreement. This program helps to reduce the variability of MCO financial results due to large cases. To evaluate the impact of this arrangement, we did not use claims re-priced at Medicaid fee levels, since the majority of expenses leading to those large cases were for prescribed medicine.

STEP 3: BLEND THE PROJECTED YEAR 1 AND YEAR 2 DATA

In this step, we blend the projected SFY 2023 medical cost for each rate cell based on actual membership in the base experience period. Blending the two base year projections increases the credibility of the capitation rates, particularly for rate cells with low enrollment. The blending process also reduces the year-to-year fluctuations in rate changes.

Opioid Addiction Treatment Trend Adjustment

We also made a trend adjustment to the base experience data underlying the MCM capitation rates to account for the estimated increase in both prevalence and cost of treatment for the opioid addiction treatment population.

Based on our review of SFY 2015 through October 2021 MCO encounter data, we determined the proportion of the total population with opioid addiction has started to decrease for several rate cells. This decreasing percentage results in adjustments of less than 1.00 for several rate cells. We believe the proposed adjustment factors properly address the funding of opioid addiction treatment in the SFY 2023 capitation rates based on available information.

Table 17 below shows the opioid addiction treatment cost trend adjustment factors:

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

Table 17 New Hampshire Department of Health and Human Services Medicaid Care Management Program Total Opioid Addiction Treatment Cost Trend Adjustment		
Rate Cell Cohort	Adjustment Factors	
Low Income Children - Age 0-11 Months	0.9765	
Low Income Children - Age 1-18 Years	0.9990	
Low Income Adults	0.9626	
CHIP	1.0028	
Foster Care / Adoption	0.9994	
Severely Disabled Children	0.9999	
Elderly and Disabled Adults - Age 19-64	1.0002	
Elderly and Disabled Adults - Age 65+	0.9990	
Dual Eligibles	0.9910	
Severe / Persistent Mental Illness - Medicaid Only	1.0069	
Severe / Persistent Mental Illness - Dual Eligibles	0.9985	
Severe Mental Illness - Medicaid Only	0.9937	
Severe Mental Illness - Dual Eligibles	0.9984	
Low Utilizer - Medicaid Only	0.9966	
Low Utilizer - Dual Eligibles	0.9975	
Serious Emotionally Disturbed Child	0.9999	
Medically Frail	0.9926	
Non-Medically Frail	0.9795	

We adjusted the opioid addiction trend factors to reflect the cost of BCH services that are projected outside of the main base experience period data. This adjustment was necessary since the original opioid addiction trend factors were developed as a global adjustment that applies to all service costs.

FFS Data Adjustment

Consistent with the SFY 2022 capitation rates, we continue to rely solely on encounter data due to the decreasing impact of the FFS experience on the MCM capitation rates in recent years. However, we continue to apply an adjustment factor developed from FFS data to reflect the portion of the base experience data historically attributed to FFS for the managed care eligible population that are newly enrolled and in their MCO selection period. Beginning January 1, 2019, DHHS implemented the "next day" enrollment process. Under this new passive enrollment, a newly Medicaid eligible member is enrolled in an MCO on the day following Medicaid eligibility confirmation.

The adjustment is meant to reflect the impact this data would have on the capitation rates without including all the detailed information through the rate setting process. The overall impact of this adjustment is 0.9984.

Preliminary and Retroactive Acuity Adjustment

For the acuity adjustment, we intend to assign each enrolled individual a risk score to capture how the average risk of the departed population (those who disenroll following redetermination) compares to the continuing population (those who meet eligibility requirements during redetermination). Since all continuing members will meet largely the same eligibility requirements that existed in the CY 2018 and CY 2019 base period, it is appropriate to use this base period as a starting point to estimate the SFY 2023 costs of the continuing population. The retroactive acuity adjustment factor will then adjust the rates to account for the average SFY 2023 acuity by capturing the number of member months associated with departed members and their relative acuity compared to continuing members at the rate cell level.

In the initial rate development, we use DHHS' initial membership projections that include the PHE unwind, and we assume Standard Medicaid departed members will have a relative acuity of 0.750, while GAHCP departed members will have a relative acuity of 0.900. We selected these relativities, which are intended to be preliminary placeholders,

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

because they produce an aggregate acuity adjustment impact on the rates that is generally consistent with the enrollment-based adjustment included in the SFY 2022 rates.

At this time, we only apply the preliminary acuity adjustment to Low Income Children and Adults, CHIP, Foster Care / Adoption, Elderly and Disabled Adults, and the GAHCP rate cells. We do not anticipate applying a retroactive acuity adjustment to the other rate cells based on the nature of services they receive (e.g., Medicare crossover, CMHC) and / or the limited enrollment growth during the PHE. However, we intend to review the other rate cells as additional redetermination and risk score information is known.

The attached Exhibit D demonstrates the calculation of each rate cell's preliminary acuity factor using the following:

- Estimated total SFY 2023 member months
- Estimated departed SFY 2023 member months
- Estimated continuing SFY 2023 member months
- Preliminary estimated acuity of departed cohort

The enrollment projections provided by DHHS assume a return to pre-COVID-19 enrollment and acuity levels by March 1, 2023.

In the event HHS continues to extend the PHE continues or a majority of redeterminations are not complete in time to accurately capture the relative acuity of the departed and continuing cohorts, we will consider re-evaluating the methodology to retroactively capture the acuity of the enrolled members in SFY 2023.

Additional Acuity Adjustment for the Medically Frail Population

Milliman and DHHS continue to review emerging medical loss ratio (MLR) experience for the Medically Frail population. Our review indicated a difference in acuity between the base data period and the emerging experience, which warranted an adjustment to the base period data.

We compared the MCOs' actual MLR during CY 2018 and CY 2019 to their September 2019 to December 2019 MLR. The resulting acuity adjustment increases the experience by 3.9% for the combined CY 2018 and CY 2019 base experience period. This acuity adjustment for the Medically Frail population is then combined with the preliminary acuity adjustment related to the end of the PHE discussed above.

May 20, 2022

IV. FINAL CAPITATION RATE ADJUSTMENTS

This section of the report describes the final adjustments to calculate the New Hampshire MCM program capitation rates from the projected SFY 2023 medical costs developed in Section III of this report.

CALCULATE FINAL PROJECTED MEDICAL COSTS

In this step, we use PMPM adjustments to account for specific changes to the projected service costs, such as adding (removing) benefits to (from) the base experience data, capturing program savings or other initiatives, and moving services to a directed payment. These adjustments include:

- Expanded mental health services under the Community Mental Health Agreement
- Facility specific FQHC adjustments
- Behavioral Health Crisis Treatment Center
- Assistive Technical Devices (ATECH)
- New Hampshire and Hampstead Hospitals
- Boston Children Hospital risk pool
- Community Residential Services
- Home visiting benefit
- Genetic testing allowance per HB 600
- Mobile Crisis Response Team and Emergency Services carve out
- Additional fee schedule changes and benefit updates
- Targeted pharmacy savings

Appendix D shows the details of our calculations.

Expanded Mental Health Services

DHHS is continuing its expansion of mental health service capacity consistent with the Community Mental Health Agreement (CMHA). The SFY 2023 capitation rates include approximately \$15.1 million of funding for additional Medicaid-funded services related to mobile crisis teams, crisis apartments, adult ACT teams, and supported employment.

We developed the PMPM add-on by rate cell using the rate cell structure for SFY 2023 and the CMHC expenditures by rate cell to allocate the CMHA funding. However, we changed our allocation method to include Serious Emotionally Disturbed Child in the funding distribution. Originally, the funds were meant to increase adult services. However, after multiple conversations with CMHCs and DHHS, we determined the funding was impacting the service levels for all behavioral health rate cells.

Facility Specific FQHC Adjustments

As of April 1, 2018, White Mountain Community Center is classified as a Federally Qualified Health Center Look-Alike (LAL) with a per encounter rate of \$176.09 for SFY 2023. We estimated the impact of the change in classification by repricing all encounters at this facility at the per encounter rate. We only repriced services subject to the per encounter rate payment as defined in the FQHC provider manual, Volume II dated January 1, 2018.

We estimate the impact of this change for SFY 2023 to be \$6,200.

Behavioral Health Crisis Treatment Center

On November 1, 2018, DHHS began operating a Behavioral Health Crisis Treatment Center (BHCTC) to serve any individuals in need of acute psychiatric treatment. The Behavioral Health Crisis Treatment Center provides services to adults ranging from crisis intervention, individual and group psychotherapy, and psychoeducational services.

Effective December 31, 2021, Riverbend's Behavioral Health Crisis Treatment Center (BHCTC) services contract ended with related services continuing through the State's redesigned Rapid Response/Mobile Crisis Response Program. We removed all BHCTC funding from the SFY 2023 MCM rates.

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

Assistive Technical Devices (ATECH) Adjustment

Effective October 1, 2018, providers bill all ATECH services to the State Plan, rather than to the Waiver. Previously, some ATECH specialty services were billed through the Waiver and DHHS discovered a duplication in State Plan and Waiver services.

We summarized all ATECH Waiver services for relevant providers and CPT codes during the base experience period to develop rate cell specific adjustments.

The impact of the ATECH adjustment is about \$4,900 annually across all populations.

New Hampshire and Hampstead Hospitals

Effective July 1, 2022, DHHS is implementing its SMI/SED 1115 waiver amendment to the Substance Use Disorder Treatment and Recovery Access (SUD-TRA) 1115 Demonstration. This waiver amendment will allow DHHS to claim federal matching for many Medicaid enrollees residing in an Institution for Mental Disease (IMD) for mental health treatment. As such, DHHS will begin providing Medicaid payments for individuals ages 21 to 64 receiving mental health services in an IMD under the standard fee-for-service Medicaid and Medicaid managed care programs. As such, these admissions to New Hampshire Hospital will now be included as an at-risk service for MCOs. All IMD stays longer than 60 days are excluded from the rate development since these stays do not qualify for the waiver.

We reviewed CY 2019 admissions at New Hampshire Hospital using data provided by DHHS, excluding all admissions over 60 days, and repriced each claim at the SFY 2023 per diem of \$1,506. As a result, the SFY 2023 rates include additional an estimated \$10,392,000 for these services.

We also included an allowance to reflect the per diem increase to \$1,506 (SFY 2023) from \$1,346 (CY 2019) for the populations already included in the MCM capitation rates (under 21 years old, 65+ year old). This adds \$541,000 to the SFY 2023 capitation rates. We also repriced historical claims data for Hampstead hospital using the same \$1,506 per diem, which is a SFY 2023 increase of \$2,875,000.

For SFY 2023, all admissions to both New Hampshire and Hampstead Hospital will be covered under a minimum fee schedule directed payment of \$1,506 per diem.

All expenditures for these two facilities are included in the capitation rates and are no longer part of the psychiatric admission kick payment.

Boston Children Hospital Risk Pool

Expenditures at BCH have varied significantly among the MCOs over the last few years. In order to better allocate funds across MCOs for services provided at BCH, DHHS created a budget-neutral risk pool to redistribute funds at the end of each contract period based on actual expenditure levels. Inpatient and outpatient facility services provided at BCH will qualify for risk pool calculation.

We set the funding level for BCH hospital inpatient and outpatient services consistent with the CY 2019 experience of \$15.0 million adjusted for retrospective payments and carved-out services (e.g., Zolgensma). We then adjusted the base period data for incurred but not reported claims liability using BCH inpatient services specific completion patterns and trended to SFY 2023 for hospital operating cost increases for a total annual projected inpatient and outpatient funding of \$17.1 million.

The BCH risk pool also includes an additional \$3.2 million to account for the following:

- Pediatric claims incurred at Massachusetts General Hospital (MGH) anticipated to move to BCH following one MCO's removal of MGH from their network
- Pediatric claims incurred at Tufts Hospital anticipated to move to BCH following the closure of Tufts' pediatric unit
- Contractual adjustments related to new contract provisions with one of the MCOs

Page 29

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

This report assumes that the reader is familiar with the State of New Hampshire's Medicaid program, its benefits, and rate setting principles. This report was prepared solely to provide assistance to DHHS to set SFY 2023 capitation rates for the Medicaid Care Management program. It may not be appropriate for other purposes. Milliman does not intend to benefit and assumes no duty or liability to other parties who receive this work. This material should only be reviewed in its entirety.

The projected funding also reflects enrollment growth in the updated enrollment projections for SFY 2023.

Community Residential Services Minimum Fee Schedule

Effective July 1, 2021, DHHS implemented a minimum fee schedule for community residential services and increasing payments consistent with the current per diem rate for transitional housing services of \$232.79. The recent increase in per diem rates is reflected in the reimbursement adjustment discussed above. However, we also allocated roughly \$958,000 in funding to reflect fourteen new beds available beginning July 1, 2022.

Home Visiting Benefit

Effective July 1, 2021, eligibility and service unit restrictions for pregnant women and for infants age 0-1 were removed following the passage of House Bill 2 (which amends RSA 167:68-a, *Home Visiting Programs*); existing eligibility and service restrictions for children ages 1-21 remain unchanged. The annual funding for this benefit is \$1,500,000 for SFY 2023.

Genetic Testing Allowance

House Bill 600 instructed DHHS to set up reimbursement for newborn genetic screening tests performed by hospitals. The bill provides a mechanism for hospitals to receive compensation to offset the cost for purchasing filter papers and administration in support of the newborn screening program. This new reimbursement was effective November 20, 2021.

The Amendment #8 SFY 2023 capitation rates include a \$75 allowance for each pediatric genetic testing performed by hospitals. The two newborn kick payments are updated to include a \$75 payment to cover the average cost of genetic tests previously borne by hospitals.

Mobile Crisis Response Team and Emergency Services Carve Out

Effective July 1, 2021, DHHS implemented a redesign of its mobile crisis response team (MCRT) infrastructure in order to provide enhanced regional delivery of mental health services for children, youth, and adults with a centralized portal and single source phone-based access point for behavioral health crises. This mental health access point provides an integrated point of entry for local and regional information for mental health and substance use crisis services. DHHS is factoring in the current regional infrastructure, which includes, but is not limited to, the Doorway program, Integrated Delivery Networks, and Regional Public Health Networks, to determine how to leverage and centralize access points for individuals and families.

Funding for MCRT and emergency services continues to be carved out of the MCM program until DHHS has a better understanding of actual utilization of these services. As a result, we removed roughly \$3.4M from the projected SFY 2023 experience in Appendix D, which is based on the historical costs for these services provided by the MCOs.

MCRT services are defined as those having a combination of the HW and U9 modifiers while emergency services are defined using HCPCS codes 90839, 90840, H2011 and S9484.

Additional Fee Schedule Changes and New Benefits

The Amendment #8 SFY 2023 capitation rates include a number of fee schedule updates as follows:

- <u>J-Code Repricing</u>: Effective July 1, 2022 several J-codes will be updated to reflect new pricing points. These changes account for an additional \$1,800,000 in funding for SFY 2023.
- Peristeen Pumps: Effective January 1, 2022, DHHS began covering the cost of peristeen pumps for eligible members. The Peristeen pump is a transanal irrigation system for people who suffer from fecal incontinence and constipation. DHHS estimated that four children and fifteen adults would be candidates for this device. DHHS also assumed an annual supply of 270 catheters (three boxes) per user, since these devices are used every other day. Based on this information, we estimate the benefit will cost approximately \$138,000 per year. We used the distribution of members with a neurogenic bowel dysfunction diagnosis by rate cell to allocate the estimated funding.

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

 <u>Breast Milk Storage Bags</u>: Effective October 1, 2021 breast pumping bags are covered at a cost of \$0.24 per unit. DHHS estimates the cost of this benefit to be about \$280,200 annually. We used the distribution of newborn deliveries by rate cell to allocate the total cost of this benefit.

Targeted Pharmacy Savings Adjustment

The SFY 2023 capitation rates include additional adjustments for targeted pharmacy savings initiatives. We summarized CY 2019 pharmacy data to identify potentially inefficient prescribing and dispensing patterns. We applied the pharmacy analyses in a stepwise process to ensure savings were not duplicated across analyses. We also applied a clinically appropriate offset to each analysis to account for situations where payment of the claim was justified. These situations may include clinically accepted off-label utilization, prior authorization, or third-party liability, among others. The pharmacy efficiency analyses include a clinical claims review, opioid analysis, and fraud, waste and abuse analysis. Lastly, we excluded all nursing facility members from these pharmacy efficiency analyses.

Table 18 below shows the overall potentially inefficient prescribing and dispensing for each pharmacy efficiency measure.

Table 18 New Hampshire Department of Health and Human Services Medicaid Care Management Program Pharmacy Efficiency Measures – Calendar Year 2019 Basis				
Pharmacy Efficiency Measure	Identified Inefficient Spend	Clinically Appropriate Adjustment	Adjusted Dollars	РМРМ
Age Limit	105,521	20%	84,417	0.04
Quantity Limit	615,376	20%	492,301	0.25
Medication Possession Ratio	46,800	20%	37,440	0.02
Therapeutic Duplication – Asthma / COPD, Corticosteroids	58,660	20%	46,928	0.02
Therapeutic Duplication – Asthma / COPD, Long-Acting Beta Agonists	4,492	20%	3,593	0.00
Therapeutic Duplication – Asthma / COPD, Long-Acting Muscarinic Antagonists	6,339	20%	5,071	0.00
Therapeutic Duplication – Autoimmune	13,168	20%	10,535	0.01
Therapeutic Duplication – HIV	190,054	20%	152,043	0.08
Opioid – Concurrent Utilization	44,533	10%	40,080	0.02
Opioid Use Disorder	34,445	10%	31,001	0.02
Opioid Agonists	4,812	10%	4,331	0.00
Total	1,124,202		907,740	0.45

Details by rate cell are provided in Exhibit G.

CALCULATE FINAL CAPITATION RATES BY RATE CELL

In this step, we apply adjustment factors to reflect third party liability recoveries, an allowance for MCO administration / margin, inclusion of directed payments, and an allowance for state premium tax. Appendix E shows the details of our calculations.

Third Party Liability Recoveries and Other Transactional Adjustment

MCOs are expected to pursue and collect third party liability (TPL) recoveries from other payers. DHHS believes there is substantial opportunity for the MCOs to enhance their recovery, provider overpayment, coordination of benefits, and subrogation efforts. In other established Medicaid programs, we typically see a range of 0.5% to 1.5% in savings for these programs. Based on our understanding of current efforts, we included a TPL savings adjustment of 0.75% in the SFY 2023 rates. We believe these assumptions represent reasonably achievable levels of savings in this program. DHHS will continue to monitor TPL recoveries in the MCM program.

The aggregate adjustment factor for TPL and other transactional adjustments is 0.9925 for SFY 2023.

Page 31

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

Fraud, Waste and Abuse Adjustment

The SFY 2023 MCM capitation rates do not include additional considerations for fraud, waste, and abuse adjustments beyond what is reflected in the base period data.

MCO Administration / Margin Allowance

Effective January 1, 2021, several parts of the MCO contract were restructured to remove certain administrative requirements. We adjusted the administrative allowance for the contract changes and additional DHHS expectations for administrative efficiency.

The composite MCO administration / margin allowance is \$42.48 PMPM across all MCM program populations, which represents 9.5% of MCO revenue prior to the directed payments and the premium tax allowance. The administration / margin allowance provides for 8.0% of revenue for administrative expenses (\$35.78 PMPM) and 1.5% for profit and risk margin (\$6.70 PMPM).

Table 19 below shows a summary of the administration / margin allowance by population.

New Hampshire Department Medicaid Care M Composite MCO Adminis	ble 19 c of Health and Human Se anagement Program stration / Margin Allowan contract Period		
	MCO Administration	Risk / Profit	Total
Population	Allowance	Margin	Allowance
Standard Medi	caid Population		
Base Population	\$27.09	\$5.07	\$32.17
CHIP	15.83	2.57	18.40
Behavioral Health Population	95.14	21.32	116.46
Standard Medicaid – PMPM	\$30.71	\$5.95	\$36.67
Standard Medicaid – Percentage of At-Risk Revenue ¹	7.7%	1.5%	9.2%
GAHCP F	Population		
Medically Frail	\$76.36	\$16.14	\$92.50
Non-Medically Frail	\$37.94	\$6.17	44.11
Behavioral Health Population	138.74	29.32	168.06
GAHCP – PMPM	\$46.43	\$8.26	\$54.69
GAHCP – Percentage of At-Risk Revenue ¹	8.4%	1.5%	9.9%
	otal		
Total – PMPM	\$35.78	\$6.70	\$42.48
Total – Percentage of At-Risk Revenue ¹	8.0%	1.5%	9.5%

¹Excluding Directed Payments and Premium Tax.

The administration allowance for the Medically Frail population is consistent with the Elderly and Disabled Adults Standard Medicaid rate cell on a percent of revenue basis. Similarly, the non-Medically Frail administration allowance is consistent with the Low Income Adults Standard Medicaid rate cell on a percent of revenue basis.

We developed the aggregate administrative cost target on a PMPM basis from the information gathered from participating MCOs and supplemented by Milliman-developed administrative cost benchmarks trended to SFY 2023. The SFY 2023 MCM program administrative allowance is consistent with the 31st percentile of Milliman's benchmarks plus a PBM administrative fee component of \$2.89 PMPM based on information provided by the MCOs.

Milliman's benchmarks are derived from a database of administrative cost, staffing, productivity, and quality information developed using data collected from more than 100 commercial health plans, Medicare Advantage Plans, Medicaid managed care organizations, third party administrators, and other organizations that administer health insurance benefits. The benchmark information and model are updated annually and validated using administrative costs reported by Medicaid MCOs in their annual financial statements. We use this database to develop customized benchmarks that can provide a basis for evaluating the administrative efficiency and effectiveness (e.g., staffing, cost, productivity, and

Page 32

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

quality) of an organization. The benchmark tool is structured around eight functional areas, 22 activities, and over 80 tasks inherent to typical health plan activities.

We also used Milliman's *Medicaid risk-based managed care: Analysis of Administrative Costs for 2020¹* Research Report to benchmark the non-benefit cost allowance included in the SFY 2023 MCM capitation rates. We compared the SFY 2023 MCM administration allowance to administrative costs reported by Medicaid MCOs in their 2019 financial statements, excluding amounts for taxes and fees. The SFY 2023 MCM program administrative allowance is consistent with the 35th percentile of the reported results included in the research report after adjusting for trends in administrative costs.

CMHC Directed Payment

The SFY 2023 MCM capitation rates include a prorated directed payment of \$5 million annually to the CMHCs across all programs and populations (MCM and GAHCP), pending approval by CMS. MCOs are required to pay these amounts directly to CMHCs according to criteria approved by CMS.

The directed payment is targeted to all Medicaid beneficiaries in the behavioral health population (members identified as SPMI, SMI, low utilizer, and SED children). We developed the PMPM directed payment by rate cell using the CMHC expenditures to allocate the total directed payment amount. Since these amounts are to be paid directly to the providers by the MCOs, we did not include an additional allowance for administrative expense or risk margin. Please see Appendix E for the rate cell specific PMPM amount for the CMHC directed payment.

DHHS has submitted the appropriate 438.6(c) pre-print and is awaiting approval from CMS.

Critical Access Hospital Directed Payment

The SFY 2023 capitation rates also include an estimated directed payment to New Hampshire critical access hospitals, pending approval by CMS. The funding amount for qualifying hospitals is \$33 million. MCOs are required to pay these amounts directly to hospitals according to criteria approved by CMS.

The directed payments is directly tied to actual hospital services provided: the number of inpatient discharges and outpatient visits reported by qualifying providers to New Hampshire MCOs. DHHS and CMS will be able to tie each payment to a specific service provided to a specific beneficiary through the data consistent with the managed care rule. Payment amounts are a uniform dollar increase initially determined by dividing the projected quality pool amount by the number of projected inpatient discharges and outpatient visits for SFY 2023. To determine the final uniform dollar amount for the qualified directed payments for inpatient discharges and the uniform dollar amount for qualified directed payments for outpatient visits, DHHS will apply weights based on the relative costs of those services and use actual utilization to ensure the quality and access pool is fully distributed.

The hospital directed payments will be made to qualifying providers out of a quality and access pool for each of the hospital classes. We developed the PMPM directed payment funding by rate cell (excluding kick payments) using the hospital inpatient and outpatient expenditures to allocate the total directed payment amount.

Since these amounts are to be paid directly to the providers by the MCOs, we did not include an additional allowance for administrative expense or risk margin. Please see Appendix E for the rate cell specific PMPM amount for the hospital directed payment.

DHHS has submitted the appropriate 438.6(c) pre-print and is awaiting approval from CMS.

Premium Tax Allowance

The capitation rates include an allowance for the 2.0% premium tax collected by the New Hampshire Insurance Department.

Withhold and Incentive Arrangement

DHHS has implemented a withhold and incentive arrangement for the SFY 2023 contract period through which DHHS will withhold 2% of the MCO rate. The terms of the withhold and incentive arrangement are outlined in the contract with

¹ https://us.milliman.com/en/insight/medicaid-managed-care-financial-results-for-2020

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

the MCOs. As required by CMS, the incentive will not exceed 5% of the capitation rates. The capitation rates shown in this report do not reflect the withhold provision. The withhold and incentive arrangement will be implemented in a retrospective manner, such that 100% of the capitation payments are made and funds will be recouped in the event that an MCO does not meet all criteria to earn back the calculated withhold amount.

The MCO contract includes the following language regarding incentive payments included in the withhold and incentive program:

"Insofar as the withhold incentive is capped at one hundred five percent (105%) of approved Capitation Payments, and the design of the Withhold and Incentive Program is to maintain withhold funds in the program, should there be a remaining amount in withheld funds within the program, additional incentives shall be available through performance metrics determined by the State so that all funds will be disbursed before the end of the contract term in accordance with separate guidance."

Furthermore, the SFY 2023 capitation rates as documented in this report are actuarially sound, since unearned withhold amounts would be returned to the program in the form of incentive payments to higher-performing MCOs. As a result, the withhold will not increase or decrease the aggregate capitation rate paid in the MCM program.

Any incentive payments made to higher-performing MCOs will not impact the minimum or maximum MLR provision of the contract.

ANTICIPATED FUTURE AMENDMENT

DHHS is considering one or more amendments to the SFY 2023 MCM capitation rates to account for the following items:

- Update to the preliminary acuity adjustment factors to reflect the actual end date of the PHE, updated enrollment projections, and New Hampshire's unwind redetermination plan
- Review of final cost-to-charge ratios that underlie the FFS hospital outpatient payment methodology
- Adjustment for additional beds at Hampstead Hospital
- Update to actual hospital inpatient DRG risk weights effective October 1, 2022
- Review of emerging utilization data to better inform trends
- Review of emerging NEMT and ABA claims data to better inform trends
- Other legislative changes

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

V. SERVICE CATEGORY ASSIGNMENT

This section of the report provides information about the service category assignment used to create the cost models included in the New Hampshire MCM program capitation rate development. This information can be used by participating MCOs to monitor their experience in a format and detail similar to the rate development process.

To prepare the attached cost models, we grouped claims into service categories. The service category assignment described below does not account for excluded or limited services. The next few paragraphs detail how the claim level detail is assigned to the service categories shown in Appendices A and B.

HOSPITAL INPATIENT

Hospital inpatient services are those items and services, provided under the direction of a physician, furnished to a patient who is admitted to a general acute care or psychiatric medical facility for facility and professional services on a continuous basis that is expected to last for a period greater than 24 hours. An admission occurs when the Severity of Illness / Intensity of Services criteria set forth by the review contractor and approved by DHHS is met. Among other services, hospital inpatient services encompass a full range of necessary diagnostic, therapeutic care including surgical, medical, general nursing, radiological, and rehabilitative services in emergency or non-emergency conditions. Additional hospital inpatient services would include miscellaneous hospital services, medical supplies, and equipment.

The hospital inpatient claims are assigned a service category based on Diagnostic Related Group (DRG) codes from version 37. Milliman's algorithm classifies hospital inpatient claims using the following groupings of CMS v24 DRG codes:

N	Table 20 lew Hampshire Department of Health and Human Services Medicaid Care Management Program Hospital Inpatient Service Groupings by DRG Code
Service Category	Diagnosis Related Group
Medical	'052'-'103','121'-'125','146'-'159','175'-'208','280'-'316','368'-'395','432'-'446','533'- '566','592'-'607','637'-'645','682'-'700','722'-'730','754'-'761','789'-'794','808'- '816','834'-'849','862'-'872','913'-'923','933'-'935','945'-'951','963'-'965','974'-'977'
Surgical	'001'-'042','113'-'117','129'-'139','163'-'168','215'-'265','326'-'358','405'-'425','453'- '517','573'-'585','614'-'630','652'-'675','707'-'718','734'-'750','799'-'804','820'- '830','853'-'858','876'-'876','901'-'909','927'-'929','939'-'941','955'-'959','969'- '970','981'-'989'
Maternity Delivery	[•] 786-788', [•] 796-798', [•] 805-807'
Maternity Non-Delivery	'769'-'770', '776'-'782'
Newborn	'795'
Psychiatric	'880'-'887'
Alcohol and Drug Abuse	'894'-'897'
Other	'998'-'999'

HOSPITAL OUTPATIENT

Hospital outpatient services are defined as those preventive, diagnostic, therapeutic, rehabilitative, surgical, and emergency services received by a patient through an outpatient / ambulatory care facility for the treatment of a disease or injury for a period of time generally not exceeding 24 hours. Outpatient / ambulatory care facilities include hospital outpatient departments, diagnostic / treatment centers, ambulatory surgical centers, emergency rooms, end stage renal disease (ESRD) clinics, and outpatient pediatric AIDS clinics (OPAC). Costs include facility charges only and do not include professional charges unless performed by staff of the facility and billed on a UB-92 (hospital) claims form. All facility-billed items not part of an inpatient admission are considered hospital outpatient services.

The hospital outpatient claims are assigned a service category based on revenue codes. Milliman's algorithm classifies hospital outpatient claims using the following groupings of revenue codes:

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program Page 35

May 20, 2022

	Table 21 New Hampshire Department of Health and Human Services
	Medicaid Care Management Program
	Hospital Outpatient Service Groupings by Revenue Code
Service Category	Revenue Code
Emergency Room	'0450'-'0459'
Surgery	'0360'-'0369','0481','0490'-'0499','0750'-'0759','0790'-'0799'
Radiology	'0320'-'0330','0333','0339'-'0349','0350'-'0359','0400'-'0403','0404','0409','0610'- '0619'
Pathology / Lab	'0300'-'0319','0923','0925'
Pharmacy	'0250'-'0269','0331'-'0332','0335','0630'-'0637'
Cardiovascular	'0480','0482'-'0489','0730'-'0739'
PT / OT / ST	'0420'-'0449','0470'-'0479','0530'-'0539','0930'-'0932','0951'-'0952'
Psychiatric	'0513','0900'-'0905','0907'-'0919'
Alcohol and Drug Abuse	'0906', '0944'-'0945'
Other	'0001','0220'-'0249','0270'-'0279','0280'-'0289','0290'-'0299','0370'-'0379','0380'- '0399','0410'-'0419','0460'-'0469','0500'-'0509','0510'-'0512','0514'- '0521','0523','0526','0528','0529','0550'-'0569','0600'-'0609','0621'- '0624','0650','0655'-'0659','0670'-'0729','0740'-'0749','0760'-'0769','0770'- '0789','0800'-'0809','0810'-'0819','0820'-'0859','0860'-'0861','0880'-'0889','0920'- '0922','0924','0929','0940'-'0943','0946'-'0947','0948','0949','0990'-'0999','2100'- '3109'

PROFESSIONAL

Professional services are assigned to a service category using a condensed version of Milliman's *Health Cost Guidelines*[™] (*HCGs*) grouping logic and other categories defined by DHHS. Professional services include the full range of preventive care services, primary care medical services, and physician specialty services. All services must be medically necessary and appropriate for the treatment of a specific diagnosis, as needed for the prevention, diagnostic, therapeutic care, and treatment of the specific condition. Physician services are performed at physician's offices, patients' homes, clinics, and skilled nursing facilities. Technical services performed in a physician's office are considered part of the professional services delivered in an ambulatory setting unless designated as a separate service.

COMMUNITY MENTAL HEALTH CENTER

Community Mental Health Center services are split into detailed service categories in order to provide more comprehensive medical cost information for the populations eligible for enhanced mental health services through the CMHCs. We reviewed the CMHC expenditures for those eligible for enhanced mental health services and developed the following service categories with the help of DHHS staff:

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

Table 22 New Hampshire Department of Health and Human Services Medicaid Care Management Program Community Mental Health Center Service Groupings by CPT Code		
Service Category	CPT Code	
Case Management	T1016	
Long Term Support Service	H0034, H2011, H2015, H2019, H2020, T1027	
Partial Hospital	H2001, H2018	
Psychotherapy	90875, 90801, 90804, 90806, 90808, 90816, 90818, 90821, 90832, 90833, 90834, 90836, 90837, 90839, 90840, 90846, 90847, 90853	
Evidence Based Practice	H2027	
Medication Management	90805, 90807, 90809, 90817, 90819, 90862, H2010, M0064, T1001	
Emergency Service 24/7	S9484	
APRTP	S9485	
Supported Employment	H2023	

PHARMACY

The pharmacy category includes pharmaceuticals as ordered by licensed prescribers and obtained at an outpatient pharmacy. Prescription drugs are identified by the presence of a National Drug Code (NDC) in the claims file. We used Medi-Span information to separate prescription drug expenditures into generic, single source brand, multi-source brand, specialty, Hepatitis C, and other scripts. We used a definition of specialty drugs consistent with Milliman's *HCGs*.

OTHER

The other service categories includes the following services:

- Home health services including intermittent skilled nursing, home health aide, physical, occupational and speech therapy services, and physician ordered supplies
- Emergency transportation, or acute care situation where normal transportation would potentially endanger the life of the patient
- Durable medical equipment that provides therapeutic benefits or enables a recipient to perform certain tasks that he or she would be unable to undertake otherwise due to certain medical conditions and / or illnesses

Other services are also assigned a service category using CPT codes. Other, unidentifiable services are assigned an "unknown" category of service.

May 20, 2022

VI. CMS RATE SETTING CHECKLIST ISSUES

This section of the report lists each item in the November 10, 2014 CMS checklist and discusses how DHHS addresses each issue and / or directs the reader to other parts of this report. CMS uses the rate setting checklist to review and approve a state's Medicaid capitation rates.

AA.1.0 – Overview of Rates Being Paid Under the Contract

The SFY 2023 managed care organization (MCO) capitation rates for the Medicaid Care Management program are developed using CY 2018 and CY 2019 MCO encounter data and FFS data for the MCO eligible population, along with other information. The GAHCP Non-Medically Frail rate cell uses March 2019 to December 2019 encounter data due to the transition of this population to the MCM program effective January 1, 2019. DHHS sets rates by rate cell for all MCOs.

Please refer to this report for background on the program and more details around the rate development.

AA.1.1 – Actuarial Certification

The Actuarial Certification of the SFY 2023 MCM capitation rates is shown in Exhibit K. The SFY 2023 MCM capitation rates have been developed in accordance with generally accepted actuarial principles and practices and are appropriate for the populations to be covered and the services to be furnished under the contract.

AA.1.2 – Projection of Expenditures

Exhibit B includes a projection of total expenditures based on estimated enrollment and SFY 2023 capitation rates.

AA.1.3 – Risk Contracts

The MCM program contract meets the criteria of a risk contract.

AA.1.4 – Modifications

The SFY 2023 rates documented in this report are the original capitation rates for the MCM population for the SFY 2023 MCM contract period. These SFY 2023 capitation rates are included in Amendment #8 to the MCM program contracts with the three MCOs.

Note: There is no AA.1.5 on the Rate Setting Checklist.

AA.1.6 – Limit on Payment to Other Providers

It is our understanding no payment is made to a provider other than the participating MCOs for services available under the contract.

AA.1.7 – Risk and Profit

The SFY 2023 MCM capitation rates include a targeted margin of 1.5% for risk, profit, and contribution to reserves. We believe that this margin is appropriate given the variability of expenses under the program.

AA.1.8 – Family Planning Enhanced Match

DHHS claims enhanced match for family planning services for the population covered under this program.

AA.1.9 – Indian Health Service (IHS) Facility Enhanced Match

DHHS does not claim enhanced match for Indian Health Services for the population covered under this program.

Page 38

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

AA.1.10 – Newly Eligible Enhanced Match

The GAHCP population is part of the newly eligible Medicaid population. Therefore, the rates for those rate cells are eligible for the enhanced Federal match under Section 1905(y). Capitation rates for the Medically Frail and Non-Medically Frail populations are developed separately from the Standard Medicaid populations. The GAHCP population eligible for enhanced behavioral health services receive the same capitation payment as similar members in the Standard Medicaid population.

AA.1.11 – Retroactive Adjustments

The SFY 2023 rates documented in this report are the original capitation rates for the SFY 2023 MCM contracts.

AA.2.0 – Based Only Upon Services Covered Under the State Plan

The CY 2018 and CY 2019 MCO encounter experience data includes a cost effective non-covered service that qualifies as an in-lieu of service and meets cost effectiveness requirements. Please see Section III of this report for more details.

AA.2.1 – Provided Under the Contract to Medicaid-Eligible Individuals

The SFY 2023 capitation rate development methodology relies on MCO encounter data for all MCM eligible populations.

AA.2.2 – Data Sources

The SFY 2023 capitation rates are developed using CY 2018 and CY 2019 MCO encounter and eligibility data. The GAHCP Non-Medically Frail rate cell uses March 2019 to December 2019 encounter data due to the transition of this population to the MCM program effective January 1, 2019.

Please refer to Sections II and III of this report for more details.

AA.3.0 – Adjustments to Base Year Data

All adjustments to the base year data are discussed in this report. In addition, each item in the checklist is addressed in items AA.3.1 - AA.3.17 below.

AA.3.1 – Benefit Differences

The base data used to calculate the capitation rates includes the services covered under the MCM contract.

Section IV of this report documents the development of PMPM add-ons for services that were not offered under the MCM contract in the base period but are part of the MCM contract for SFY 2023.

Please refer to Section IV of this report for more details.

AA.3.2 – Administrative Cost Allowance Calculations

The capitation rates include explicit administrative allowances by rate cell. Please see Section IV in the report for more details regarding the administrative allowance calculation.

AA.3.3 – Special Populations' Adjustments

The SFY 2023 capitation rates methodology does not include an adjustment for special populations as the base FFS and encounter data used to calculate the capitation rates is consistent with the eligible population.

AA.3.4 – Eligibility Adjustments

The base data only reflects experience for time periods where members were eligible to enroll in an MCO.

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

AA.3.5 – Third Party Liability (TPL)

The managed care organizations are responsible for the collection of any TPL recoveries. The capitation rates include an adjustment to reflect additional TPL recoveries that are not reflected in the base year FFS data portion.

AA.3.6 – Indian Health Care Provider Payments

The MCOs are responsible for the entirety of any IHC payments, which are fully reflected in the claims data.

AA.3.7 – DSH Payments

DSH payments are not included in the capitation rates.

AA.3.8 – FQHC and RHC Reimbursement

The MCOs are responsible for the entirety of the FQHC and RHC encounter payments, which are fully reflected in the claims data.

AA.3.9 – Graduate Medical Education (GME)

GME payments are not included as part of the capitation rates.

AA.3.10 - Copayments, Coinsurance, and Deductibles in Capitated Rates

The MCM population with an income over 100% of FPL must pay a \$1 / \$2 preferred / non-preferred copay for prescription drugs. The FFS and MCO encounter data reflect the copayment collection.

AA.3.11 – Medical Cost / Trend Inflation

Section III of this report documents the trend assumptions used to project the base period costs to SFY 2023.

AA.3.12 – Utilization Adjustments

Utilization trend is included in AA.3.11.

AA.3.13 – Utilization and Cost Assumptions

The utilization and cost assumptions are appropriate for the population to be covered.

AA.3.14 – Post-Eligibility Treatment of Income (PETI)

Long-term care services that are subject to patient liability are excluded from the MCM population capitation rates.

AA.3.15 – Incomplete Data Adjustment

The capitation rates include an adjustment to reflect IBNR claims and underreported CMHC claims. Please refer to Section III of this report for more information on the development of these adjustment factors.

AA.3.16 – Primary Care Rate Enhancement

The SFY 2023 capitation rates are priced at levels consistent with current MCO reimbursement levels with considerations for expected MCM fee schedule changes.

AA.3.17 – Health Homes

Not Applicable.

Page 40

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

AA.4.0 – Establish Rate Category Groupings

The SFY 2023 capitation rates use several rate cells developed from Medicaid eligibility categories to designate the eligible population. Please see Section II of this report and Exhibit I for more details.

There are also separate maternity and newborn kick payments.

AA.4.1 – Eligibility Categories

The eligibility categories included in the SFY 2023 capitation rates are defined in Section II of this report.

AA.4.2 – Age

Age is used for certain rate category groupings.

AA.4.3 – Gender

Gender is not used as a rating variable.

AA.4.4 – Locality / Region

Region is not used as a rating variable.

AA.4.5 – Risk Adjustments

The MCM population capitation rates will use an actuarially sound risk adjustment model to adjust the rates for each participating MCO. Section II of this report includes an overview of the risk adjustment methodology.

We will provide a separate report documenting the development of the MCO Adjusted Risk Factors that will be applied to the SFY 2023 capitation rates.

AA.5.0 – Data Smoothing

We did not perform any data smoothing.

AA.5.1 – Cost-Neutral Data Smoothing Adjustment

We did not perform any data smoothing.

AA.5.2 – Data Distortion Assessment

Our review of the base FFS and MCO encounter data did not detect any material distortions or outliers.

AA.5.3 – Data Smoothing Techniques

We determined that a data smoothing mechanism resulting from data distortions was not required.

AA.5.4 – Risk Adjustments

The MCM population capitation rates will use an actuarially sound risk adjustment model to adjust the rates for each participating MCO. Section II of this report includes an overview of the risk adjustment methodology.

We will provide a separate report documenting the development of the MCO Adjusted Risk Factors that will be applied to the SFY 2023 capitation rates.

AA.6.0 – Stop-Loss, Reinsurance, or Risk Sharing Arrangements

DHHS administers a stop-loss arrangement for high-cost patients where DHHS shares 50% of the cost above \$615,000 for members that have total expenses for capitated services valued at Medicaid fee levels above that threshold. The

SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

State of New Hampshire Department of Health and Human Services

stop-loss provision will exclude claims incurred at Boston Children's Hospital, due to the creation of a separate risk pool for those services.

The stop-loss adjustment factors reduce the base experience data by about 0.05% to reflect the MCO liability under the stop-loss agreement. This change is intended to be budget neutral to DHHS and the MCOs. It helps to reduce the variability of MCO financial results due to large cases.

The SFY 2023 MCM capitation rates also feature a risk pool for Boston Children's Hospital services described in Section II of this report.

AA.6.1 – Commercial Reinsurance

DHHS does not require entities to purchase commercial reinsurance.

AA.6.2 – Stop-Loss Program

DHHS administers a stop-loss arrangement for high-cost patients where DHHS shares 50% of the cost above \$615,000 for member that have total expenses for capitated services valued at Medicaid fee levels above that threshold. The stop-loss provision will exclude claims incurred at Boston Children's Hospital due to the creation of a separate risk pool for those services.

The adjustment factors reduce the base experience data by about 0.05% to reflect the MCO liability under the stop-loss agreement. This change is intended to be budget neutral to DHHS and the MCOs. It helps to reduce the variability of MCO financial results due to large cases.

AA.6.3 – Risk Corridor Program

The SFY 2023 MCM capitation rates also feature a minimum and maximum MLR provision, similar to a risk corridor, as described in Section II of this report.

AA.7.0 – Incentive Arrangements

The MCO contract includes a withhold and incentive program in which unearned withheld dollars will be used to finance an incentive pool that is available for additional incentive payments to be made to high-performing MCOs. The amount of an incentive pool awarded to a particular MCO will not exceed 5.0% of the MCO's qualifying capitation revenue.

Please refer to the Withhold and Incentive Program policy included in the MCO contract for more details.

AA.7.1 – Electronic Health Records (EHR) Incentive Payments

DHHS has not implemented incentive payments related to EHRs for the SFY 2023 contract period.

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

VII. RESPONSE TO 2022-2023 MEDICAID CARE RATE DEVELOPMENT GUIDE (APRIL 2022)

SECTION I. MEDICAID MANAGED CARE RATES

1. General Information

- A. Rate Development Standards
 - i. The SFY 2023 capitation rates do not include rate ranges.
 - ii. The rate certification included herein is for the SFY 2023 contract period.
 - iii. This rate certification submission was prepared in accordance with 42 CFR §438.4, 438.5, 438.6, and 438.7.
 - a. The actuarial certification report signed by John Meerschaert, FSA, MAAA certifies that the final capitation rates meet the standards in 42 CFR §438.3(c), 438.3(e), 438.4, 438.5, 438.6, and 438.7. The certification can be found in Exhibit K.
 - b. The final and certified capitation rates for all rate cells can be found in Exhibit A.
 - c. The items requested can be found in Sections I through IV of this report.
 - iv. Differences in capitation rates for covered populations are based on valid rate development standards and are not based on the rate of federal financial participation associated with the covered population.
 - v. Each rate cell is developed independently to be actuarially sound and does not cross-subsidize payments for another rate cell.
 - vi. The effective dates of changes to the Medicaid program are consistent with the assumptions used to develop the capitation rates.
 - vii. The capitation rates are developed in a way that the MCO can reasonably achieve a medical loss ratio of at least 85% for the rate year.
 - viii. The SFY 2023 capitation rates do not include rate ranges.
 - ix. The SFY 2023 capitation rates do not include rate ranges.
 - x. The rate certification submission demonstrates that the capitation rates were developed using generally accepted actuarial practices and principles.
 - a. All adjustment to the capitation rates reflect reasonable, appropriate, and attainable costs.
 - b. No adjustments to the rates are performed outside of the initial rate setting process beyond those outlined in Section III of the report.
 - c. The final contracted rates in each cell match the capitation rates in the certification.
 - xi. The capitation rates included in this submission are certified for all time periods in which they are effective. No rates for a previous time period are used for a future time period.
 - xii. Section I includes documentation of the COVID-19 considerations in the SFY 2023 capitation rate development.
 - xiii. This rate certification conforms to the procedure for rate certifications for rate and contract amendments.

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

This report assumes that the reader is familiar with the State of New Hampshire's Medicaid program, its benefits, and rate setting principles. This report was prepared solely to provide assistance to DHHS to set SFY 2023 capitation rates for the Medicaid Care Management program. It may not be appropriate for other purposes. Milliman does not intend to benefit and assumes no duty or liability to other parties who receive this work. This material should only be reviewed in its entirety.

- B. Appropriate Documentation
 - i. The SFY 2023 capitation rates do not include rate ranges.
 - ii. We believe the attached report properly documents all the elements included in the rate certification and provides CMS enough detail to determine that regulation standards are met.

Please see Sections I through IV of this report for the following details:

- a. Data used, including citations to studies, research papers, other states' analyses, or similar secondary data sources.
- b. Assumptions made, including any basis or justification for the assumption.
- c. Methods for analyzing data and developing assumptions and adjustments.
- iii. This capitation rate certification includes a specific rate for each rate cell developed from specific assumptions for each rate cell. All assumptions and adjustments underlying the certified capitation rates are properly documented.
- iv. The SFY 2023 capitation rates do not include rate ranges.
- v. We detail within our responses in this guide the section of our report where each item described in the 2022 to 2023 Medicaid Managed Care Rate Development Guide can be found.
- vi. Differences in capitation rates for covered populations are based on valid rate development standards and are not based on the rate of federal financial participation associated with the covered population.
- vii. DHHS claims enhanced match for family planning services for the population covered under this program. All other services and populations included in this rate certification are subject to the regular state FMAP except for the newly eligible population as described in Section III.
- viii. Please see Sections I and II of this report for the requested documentation.
- ix. Section IV outlines the anticipated rate amendment details.
- x. Section I includes documentation of the COVID-19 considerations in the SFY 2023 capitation rate development.

2. Data

- A. Rate Development Standards
 - i. Our report includes a thorough description of the data used and shows compliance with 42 CFR §438.5(c).
 - a. DHHS provided Milliman with validated encounter data and financial reports for at least the three most recent and complete years prior to the rating period.
 - b. The rate development methodology uses current MCO encounter data. In an email dated May 11, 2022, CMS approved the state's request for an exception to the base data standards in accordance with 42 CFR § 438.5(c)(3) for SFY 2023 rate development purposes.
 - c. The data used is derived from the Medicaid population served under the Medicaid Care Management program.
 - d. The rate development methodology uses recent MCO encounter data. In an email dated May 11, 2022, CMS approved the state's request for an exception to the base data standards in accordance with 42 CFR § 438.5(c)(3) for SFY 2023 rate development purposes.

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

B. Appropriate Documentation

- i. Milliman requested and received a full claims and enrollment database from DHHS and the MCOs. This information is summarized in Appendices A1-A2.
- ii. A detailed description of the data used in the rate development methodology can be found in Section III of this report. Section III also includes comments on the availability and quality of the data used for rate development.
- iii. The rate certification and attached report thoroughly describe any material adjustments, and the basis for the adjustments, that are made to the data. Please see Section III and IV of this report for more details.

3. Projected Benefit Costs

A. Rate Development Standards

- i. The final capitation rates shown in Exhibit A are based only upon services described in 42 CFR §438.3(c)(1)(ii) and 438.3(e).
- ii. Each projected benefit cost trend assumption is reasonable and developed in accordance with generally accepted actuarial principles and practices using actual experience of the Medicaid population and consideration of other factors that may affect projected benefit cost trends through the rating period.
- iii. Please refer to Section III of this report for the details related to the treatment of in lieu of services.

New Hampshire's Substance Use Disorder Treatment and Recovery Access 1115 Demonstration Waiver allows for the coverage of substance use disorder (SUD) services provided in an IMD. The SMI/SED 1115 waiver amendment to the Substance Use Disorder Treatment and Recovery Access (SUD-TRA) 1115 demonstration will allow DHHS to receive federal matching funds for these Institution for Mental Disease (IMD) services.

- B. Appropriate Documentation
 - i. The various Appendices and Exhibits included in this report document the final projected benefit costs by relevant level of detail and is consistent with how the State makes payments to the plans.
 - ii. Please refer to Section III of this report for the methodology and assumptions used to project contract period benefit costs. Section II of the report highlights key methodological changes since the previous rate development.
 - The rate certification includes a section on projected benefit cost trends in compliance with 42 CFR §438.7(b)(2). See Step 2 of Section III for details related to the development of projected benefit cost trends.
 - iv. This certification does not include additional services deemed by the state to be necessary to comply with the parity standards of the Mental Health Parity and Addiction Equity Act.
 - v. Please refer to Section III of this report for the details related to the treatment of in lieu of services.
 - vi. Section III includes a description of how retrospective eligibility periods are accounted for in rate development.
 - vii. Section I documents the impact on projected costs for all material changes to covered benefits or services since the last rate certification.
 - viii. The rate certification includes an estimated impact of each covered benefit or service change on the amount of projected benefit costs and a description of the data, assumptions, and methodologies used to develop the adjustment for each change related to covered benefits or services.

Page 45

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

4. Special Contract Provisions Related to Payment

A. Incentive Arrangements

i. Rate Development Standards

The SFY 2023 capitation rate methodology does include an incentive arrangement. Total payments under the incentive arrangement will not exceed 105 percent of the approved capitation payments under the contract that are attributable to the enrollees or services covered by the incentive arrangements. Please refer to the Withhold and Incentive Program policy included in the MCO contract for more details.

ii. Appropriate Documentation

The SFY 2023 capitation rate methodology does include an incentive arrangement. Total payments under the incentive arrangement will not exceed 105 percent of the approved capitation payments under the contract that are attributable to the enrollees or services covered by the incentive arrangements. Please refer to the Withhold and Incentive Program policy included in the MCO contract for more details.

- B. Withhold Arrangements
 - i. Rate Development Standards

The SFY 2023 capitation rate methodology does include a withhold arrangement. Please refer to the Withhold and Incentive Program policy included in the MCO contract for more details.

ii. Appropriate Documentation

The SFY 2023 capitation rate methodology does include a withhold arrangement. Please refer to the Withhold and Incentive Program policy included in the MCO contract for more details.

- C. Risk Sharing Mechanism
 - i. Rate Development Standards

Section III of the report documents the High-Cost Patient Stop-Loss Adjustment, minimum / maximum MLR provisions, and the Boston Children's Hospital risk pool.

ii. Appropriate Documentation

Section III of the report documents the High-Cost Patient Stop-Loss Adjustment, minimum / maximum MLR provisions, and the Boston Children's Hospital risk pool.

- D. State Directed Payments
 - i. Rate Development Standards

Section IV of the report documents the directed payments and minimum fee schedule items and are pending CMS approval. There are no additional directed payments in the program that are not addressed in the certification. The directed payment are in compliance with 42 C.F.R. § 438.6(c),

ii. Appropriate Documentation

Please review tables 1 to 3 below for the requested information.

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

Table 1 New Hampshire Department of Health and Human Services State Directed Payments				
Control name of the state directed payment	Type of payment	Brief description	Is the payment included as a rate adjustment or separate payment term?	
CMHC Directed Payment	Medicaid-specific delivery system reform or performance improvement initiative	Payments are made to CMHCs who achieve various quality measures.	Separate payment term	
Hospital Directed Payment	Uniform percentage increase for network providers that provide a particular service under the contract	Payment made to critical access hospitals (CAH) tied to actual hospital services provided; the number of inpatient discharges and outpatient visits.	Separate payment term	
DME Minimum Fee Schedule	Minimum fee schedule	The minimum fee schedule for DME services is set at 80% of the current Medicaid fee schedule.	Rate adjustment	
Designated Receiving Facility Minimum Fee Schedule	Minimum Fee Schedule	The minimum fee schedule for DRF services is set at the current Medicaid fee schedule.	Rate adjustment	
Community Residential Minimum Fee Schedule	Minimum fee schedule	The maintenance of effort (MOE) requirement is incorporated into the MCM program effective 7/1/2021. There are concerns that the return of this risk on the capitation rates in conjunction with the emergency department boarding crisis will disincentivize the CMHCs to support capacity expansion. Therefore, reimbursement for community residential services is moved to a directed payment.	Rate adjustment	
New Hampshire Hospital and Hampstead Hospitals	Minimum fee schedule	The minimum fee schedule for admissions at these hospitals is set at \$1,506 per day for SFY 2023.	Rate adjustment	

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

	Table 2 New Hampshire Department of Health and Human Services				
State Directed Payment – Rate Adjustments					
Control name of the state directed payment	Rate cells affected	Impact	Description of the adjustment	Confirmation the rates are consistent with the preprint	For maximum fee schedules, provide the information requested
DME Minimum Fee Schedule	All rate cells	Please See Appendices B1 and B2 for the reimbursement factor adjustment related to DME services	Implemented as a reimbursement adjustment to bring the base encounter data in line with the minimum fee schedule.	The directed payment is consistent with the pre-print.	Not a maximum fee schedule
Designated Receiving Facility Minimum Fee Schedule	All rate cells	Please see Exhibit A	Implemented as a reimbursement adjustment to the hospital inpatient psychiatric admission kick payment for Peer Group 06.	Minimum fee schedule set equal to the Medicaid fee schedule; therefore, no pre-print is required.	Not a maximum fee schedule
Community Residential Minimum Fee Schedule	All rate cells	Please see Appendix D	Adjusted historical expenditure to account for new fee schedule minimum.	Minimum fee schedule set equal to the Medicaid fee schedule; therefore, no pre-print is required.	Not a maximum fee schedule
New Hampshire Hospital and Hampstead Hospitals	All rate cells	Please see Appendix D	Adjusted historical expenditure to account for new fee schedule minimum.	Minimum fee schedule set equal to the Medicaid fee schedule; therefore, no pre-print is required.	Not a maximum fee schedule

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

This report assumes that the reader is familiar with the State of New Hampshire's Medicaid program, its benefits, and rate setting principles. This report was prepared solely to provide assistance to DHHS to set SFY 2023 capitation rates for the Medicaid Care Management program. It may not be appropriate for other purposes. Milliman does not intend to benefit and assumes no duty or liability to other parties who receive this work. This material should only be reviewed in its entirety.

Page 48

Table 3 New Hampshire Department of Health and Human Services State Directed Payment – Separate Payments					
Control name of the state directed payment CMHC Directed Payment	Aggregate amount included in the certification \$5,000,000	Statement that the actuary is certifying the separate payment term We certify the amount of the separate payment term disclosed in the certification.	The magnitude on a PMPM basis Please refer to Appendix E for the PMPM impact by rate cell.	Confirmation the rate development is consistent with the preprint The directed payment is consistent with the pre-print.	Confirmation that the state and actuary will submit required documentation at the end of the rating period (as <u>applicable</u>) The state and its actuary will submit required documentation at the end of the rating period.
Hospital Directed Payment	\$33,000,000	We certify the amount of the separate payment term disclosed in the certification.	Please refer to Appendix E for the PMPM impact by rate cell.	The directed payment is consistent with the pre-print.	The state and its actuary will submit required documentation at the end of the rating period.

E. Pass-Through Payments

i. Rate Development Standards

The SFY 2023 capitation rate methodology does not include any pass-through payments.

ii. Appropriate Documentation

The SFY 2023 capitation rate methodology does not include any pass-through payments.

5. Projected Non-Benefit Costs

- A. Rate Development Standards
 - i. The development of the non-benefit component of the SFY 2023 rates is compliant with 42 CFR §438.5(e) and includes reasonable, appropriate, and attainable expenses related to MCO administration, taxes, licensing and regulatory fees, contribution to reserves, risk margin, and cost of capital.
 - ii. The non-benefit costs included in the SFY 2023 capitation rates are developed as a target PMPM and applied using percentages of projected benefit costs that vary by rate cell grouping.

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program Page 49

May 20, 2022

- B. Appropriate Documentation
 - i. Please refer to Section IV of this report for a detailed description of the data and methodology used to develop of the projected non-benefit costs included in the capitation rates. The report includes a description of changes made since the last rate development.
 - ii. The projected non-benefit costs include appropriate consideration for administrative costs, taxes, licensing and regulatory fees, other assessments and fees, contribution to reserves, risk margin, and cost of capital.
 - iii. MCOs provided non-benefit cost data as part of the data validation and reconciliation to financial statements efforts. We reviewed this information for general reasonableness and to understand the operations of the MCOs. The targeted PMPM administrative allowance was negotiated with the MCOs and compared to national benchmarks adjusted for New Hampshire's contractual requirements.

6. Risk Adjustment and Acuity Adjustment

- A. Rate Development Standards
 - i. The SFY 2023 capitation rates will use the risk adjustment arrangement described in Section II of this report.
 - ii. The risk adjustment arrangement described in Section II has been developed in accordance with generally accepted actuarial principles and practices and is budget neutral to the state in total.
 - iii. Section III of this report documents the trend adjustment for opioid addiction population treatment costs to reflect the increased acuity of the covered population Section III also documents the acuity adjustment applied for the Medically Frail population.

Sections II and III of this report document preliminary and retroactive acuity adjustments related to the end of the PHE and related enrollment levels. Due to the uncertainty surrounding the unwind process following the end of the PHE, DHHS will implement a non-budget neutral retrospective acuity adjustment. This adjustment will capture the relative acuity of the actual individuals enrolled during SFY 2023 during the continuing PHE and through the redetermination process. The preliminary acuity factor included in this report will be updated retroactively to account for the actual PHE end date, the redetermination process, and relative acuity levels.

- B. Appropriate Documentation
 - i. The SFY 2023 capitation rates will use the risk adjustment arrangement described in Section II of this report.
 - ii. The SFY 2023 capitation rate methodology does not include any retrospective risk adjustment components.
 - iii. Proposed changes to the risk adjustment methodology will be documented in a separate correspondence. The risk adjustment process is and will remain budget neutral to the state in total.
 - iv. Please see Sections II and III of this report for the requested documentation regarding the trend adjustment for opioid addiction population treatment costs to reflect the increased acuity of the covered population, the acuity adjustment for the Medically Frail population, and the preliminary and retroactive acuity adjustment.

SECTION II. MEDICAID MANAGED CARE RATES WITH LONG-TERM SERVICES AND SUPPORTS

This certification does not include rates for managed long-term services and supports (MLTSS).

Page 50

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

SECTION III. NEW ADULT GROUP CAPITATION RATES

This section includes the documentation for the rates for the new adult group under 1902(a)(10)(A)(i)(VIII) of the Social Security Act.

1. Data

- A. A detailed description of the data can be found in Sections II and III of this report.
- B. The Medically Frail and Non-Medically Frail populations were covered starting in September 2014. The SFY 2023 rates are based on CY 2018 and CY 2019 encounter data for the Medically Frail population and March 2019 to December 2019 encounter data for the Non-Medically Frail population, which is the most recent data available for these populations.

2. Projected Benefit Costs

- A. Our report includes a thorough discussion of issues related to the projected benefit costs for the new adult group:
 - i. We relied on data for the Medically Frail and Non-Medically Frail populations available from previous rating periods. The rates are based on the data time periods noted above. The base data, methodology, and assumptions used to calculate the SFY 2023 capitation rates is generally similar to the methodology used to calculate the SFY 2022 capitation rates.
 - ii. Not applicable the new adult group was covered in previous rating periods.
 - iii. We made an adjustment for acuity to reflect of the impact of the COVID-19 pandemic and PHE and another adjustment to reflect observed acuity differences between the base period data and emerging experience for the Medically Frail population. We did not make any adjustments for pent-up demand, adverse selection, and demographic differences to either Medically Frail or Non-Medically Frail populations.
- B. Table 2 in Section I of the report quantifies the impact of program changes implemented for SFY 2023.

3. Projected Non-Benefit Costs

- A. The methodology used to develop the SFY 2023 non-benefit costs is consistent with those used to develop the SFY 2022 non-benefits costs. We increased the non-benefit cost allowance for all rate cells to reflect new MCO contract requirement.
- B. Please refer to Section IV of this report for more details on the development of the non-benefit costs for the Medically Frail and Non-Medically Frail populations and how these assumptions compare to the Standard Medicaid population.

4. Final Certified Rates

A. Please refer to Tables 1 and 2 in Section I of the report for a comparison of the SFY 2023 capitation rates to the SFY 2022 capitation rates.

5. Risk Mitigation Strategy

A. The SFY 2023 capitation rate for the new adult group will use the same risk adjustment, minimum loss ratio requirement with remittance, High-Cost Patient Stop-Loss arrangement, and Boston Children's Hospital risk pool, as the Standard Medicaid population, as described in Section II of this report. Moreover, DHHS is implementing a minimum and maximum MLR provision for the SFY 2023 contract period to address the uncertainty of future medical costs due to the COVID-19 pandemic and the public health emergency.

Page 51

May 20, 2022

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

B. DHHS is implementing a minimum and maximum MLR provision for the SFY 2023 contract period to address the uncertainty of future medical costs due to the COVID-19 pandemic and the public health emergency. Refer to Section II of the report for the detailed provisions of the MLR provisions.

This report assumes that the reader is familiar with the State of New Hampshire's Medicaid program, its benefits, and rate setting principles. This report was prepared solely to provide assistance to DHHS to set SFY 2023 capitation rates for the Medicaid Care Management program. It may not be appropriate for other purposes. Milliman does not intend to benefit and assumes no duty or liability to other parties who receive this work. This material should only be reviewed in its entirety.

Page 52

EXHIBIT I

State of New Hampshire Department of Health and Human Services Medicaid Care Management Program Rate Cell Assignment Logic

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022



15800 W. Bluemound Road Suite 100 Brookfield, WI 53005 USA Tel +1 262 784 2250 Fax +1 262 923 3680

milliman.com

Mathieu Doucet, FSA, MAAA Consulting Actuary

mathieu.doucet@milliman.com

January 16, 2020

Mr. Henry D. Lipman, FACHE Medicaid Director New Hampshire Department of Health and Human Services 129 Pleasant Street Concord, NH 03301

Re: SFY 2021 Rate Cell Recommendation

Dear Henry:

At your request, we are providing the New Hampshire Department of Health and Human Services (DHHS) with a recommendation for a rate cell structure for the Medicaid Care Management (MCM) program for contract year SFY 2021 and beyond.

RATE CELL ASSIGNMENT LOGIC

The proposed structure has two major changes from the SFY 2020 MCM rate cell structure.

- The behavioral health rate cells will contain both Standard Medicaid and Granite Advantage individuals. We moved the behavioral health rate cells to first priority in the hierarchy, above the expansion population rate cells. We expect this change will help facilitate the contracting arrangement between community mental health centers (CMHCs) and managed care organizations (MCOs) for all individuals with enhanced behavioral health needs. We also renamed the "Medicaid – Only" rate cells to "Non-Dual" for clarity.
- 2. The new rate cell for the Children's Health Insurance Program (CHIP) population will allow for easier Federal Medical Assistance Percentages (FMAP) reporting and claiming by DHS.

Table 1 below shows the rate cell definitions for the Standard Medicaid and Granite Advantage Health Care Program (GAHCP) populations, which includes rate cell assignment using the following hierarchy:

- Behavioral health rate cells
- Expansion population rate cells
- All other rate cells



	Mee	Table 1 re Department of Health and Huma dicaid Care Management Program SFY 2021 Rate Cell Definitions		
Rate Cell	Age / Gender Categories	Eligibility Category	Dual Status Code*	Other Criteria
		Hierarchy: First Priority		
Severe / Persistent Mental Illness – Dual Eligibles	All	Any except MGIA or MGIM	Y	CMHC certification code U1
Severe / Persistent Mental Illness – Non-Dual Eligibles	All	Any	Ν	CMHC certification code U1
Severe Mental Illness –Dual Eligibles	All	Any except MGIA or MGIM	Y	CMHC certification code U2
Severe Mental Illness – Non-Dual Eligibles	All	Any	Ν	CMHC certification code U2
Low Utilizer – Dual Eligibles	All	Any except MGIA or MGIM	Y	CMHC certification code U5
Low Utilizer – Non-Dual Eligibles	All	Any	Ν	CMHC certification code U5
Serious Emotionally Disturbed Child	All	Any	All	CMHC certification code U6 or U7
		Hierarchy: Second Priority		
Granite Advantage – Medically Frail	All	MGIM	All	
Granite Advantage – Non-Medically Frail	All	MGIA	All	
		Hierarchy: Third Priority		
Low Income Children and Adults	Separate rate cells for the following age groups: 0 to 11 months** 1 to 18 19 +	MAEM, MAEN, MAER, MAES, MAEU, MAFR, MAFU, MCER, MCEU, MCFR, MCFU, MCN, MCRA, MCRF, ME12, ME4, MMER, MMEU, MMFR, MMFU, MMRA, MMRF, MP C, MP P, MPQC, MPQP, MCIS, MGIC, MGIF, MGIN, MGIP, MGIW	Ν	
CHIP	All	MCIE, MGIE	All	
Foster Care / Adoption	All	MCP1, MCP2, MCPI, MCPN, MMP2, MMPN	N	
Severely Disabled Children	All	MAAB, MCAB, MCDC, MCHC, MMAB, MMDC, MMHC, MSPB, MSPP	Ν	Age <19 for category codes MAAB, MCAB, MMAB, MSPB, MSPP
Disabled Adults	Separate rate cells for the following age groups: 19 to 64 65+	MAAA, MAAB, MAAD, MBCC, MCAA, MCAB, MCAD, MEAD, MMAA, MMAB, MMAD, MSPB, MSPP	N	Age 19+ for category code MAAB, MCAB, MMAB, MSPB, MSPP (includes 18 year olds in other eligibility categories)
Dual Eligibles	All	Any	Y	
		Kick Payments		
Newborn Kick Payment		Any	N	First two months of life
Neonatal Abstinence Syndrome Newborn Kick Payment		Any	Ν	Same as newborn criteria with diagnosis code requirement
Maternity Kick Payment		Any tor "Y" and the presence of a Medicaid E	Ν	Service description below

Only. ** The Low Income Children and Adults – 0 to 11 Months rate cell also includes the first month of life (excluding the birth month) for babies born prior to

* The Low Income Children and Adults – 0 to 11 Months rate cell also includes the first month of life (excluding the birth month) for babies born prior to Medicaid enrollment (and, therefore, excluded from the newborn kick payment).



The neonatal abstinence syndrome (NAS) kick payment is provided for babies who have a diagnosis code of P96.1 (ICD10) or 779.5 (ICD9) in their birth month on an inpatient claim. The regular newborn kick payment is provided for all other newborns.

The maternity kick payment includes the following services:

- Hospital inpatient services with MS-DRG codes of 765 to 768, 774 to 775
- Hospital outpatient and professional services with a primary diagnosis code of:

ICD9: v27.0 - v27.9, 650, and 651.01 - 669.92 (with the 5th digit being 1 or 2) ICD10: O60 - O82, and Z37

This includes delivery related anesthesia services (CPT-4 codes 00850, 00857, 00946, 00955, 01960, 01961, 01967, 01968), as long as a relevant primary diagnosis code is present.

Delivery and post-partum care services (CPT-4 codes 59400, 59409, 59410, 59430, 59510, 59514, 59515, 59610, 59612, 59614, 59618, 59620, 59622)

The maternity kick payment includes all facility and professional claims associated with deliveries. Prenatal and postpartum services are not included in the maternity kick payment.

CAVEATS AND LIMITATIONS ON USE

This letter is designed to assist DHHS with reviewing rate cell assignment logic for the SFY 2021 MCM capitation rates. This information may not be appropriate, and should not be used, for other purposes.

The information contained in this letter has been prepared for DHHS. To the extent that the information contained in this letter is provided to third parties, this letter should be distributed in its entirety. Any user of this information must possess a certain level of expertise in actuarial science and healthcare modeling, so as not to misinterpret the information presented.

Milliman makes no representations or warranties regarding the contents of this letter to third parties. Likewise, third parties are instructed that they are to place no reliance upon this letter prepared for DHHS by Milliman that would result in the creation of any duty or liability under any theory of law by Milliman or its employees to third parties. Other parties receiving this letter must rely upon their own experts in drawing conclusions about the information presented.

Guidelines issued by the American Academy of Actuaries require actuaries to include their professional qualifications in all actuarial communications. I am a member of the American Academy of Actuaries, and I meet the qualification standards for performing the analyses in this letter.

The terms of Milliman's contract with DHHS effective July 1, 2017 apply to this letter and its use.

 \diamond \diamond \diamond \diamond



Mr. Henry Lipman NH Department of Health and Human Services January 16, 2020 Page 4 of 4

Henry, please call me at 262 784 2250 if you have any questions.

Sincerely,

Cathin T. and

Mathieu Doucet, FSA, MAAA Consulting Actuary

MD/laa

EXHIBIT K

State of New Hampshire Department of Health and Human Services Medicaid Care Management Program SFY 2023 Actuarial Certification

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

This report assumes that the reader is familiar with the State of New Hampshire's Medicaid program, its benefits, and rate setting principles. This report was prepared solely to provide assistance to DHHS to set SFY 2023 capitation rates for the Medicaid Care Management program. It may not be appropriate for other purposes. Milliman does not intend to benefit and assumes no duty or liability to other parties who receive this work. This material should only be reviewed in its entirety.



May 20, 2022

New Hampshire Department of Health and Human Services Capitated Contracts Ratesetting Actuarial Certification SFY 2023 Medicaid Care Management Program Capitation Rates

I, John D. Meerschaert, am associated with the firm of Milliman, Inc., am a member of the American Academy of Actuaries, and meet its Qualification Standards for Statements of Actuarial Opinion. I was retained by the New Hampshire Department of Health and Human Services (DHHS) to perform an actuarial certification of the Medicaid Care Management program capitation rates for SFY 2023 for filing with the Centers for Medicare and Medicaid Services (CMS). I reviewed the calculated capitation rates and am familiar with the relevant requirements of 42 CFR 438; the CMS "Attachment A, PAHP, PIHP and MCO Contracts Financial Review Documentation for At-risk Capitated Contracts Ratesetting;" the 2022 to 2023 Medicaid Managed Care Rate Development Guide; and Actuarial Standard of Practice (ASOP) 49.

I examined the actuarial assumptions and actuarial methods used in setting the capitation rates for SFY 2023. To the best of my information, knowledge and belief, the capitation rates offered by DHHS are in compliance with the relevant requirements of 42 § CFR 438.3(c), 438.3(e), 438.4, 438.5, 438.6, and 438.7. The attached actuarial report describes the capitation rate setting methodology.

In my opinion, the capitation rates are actuarially sound, as defined in Actuarial Standard of Practice (ASOP) 49, were developed in accordance with generally accepted actuarial principles and practices, and are appropriate for the populations to be covered and the services to be furnished under the contract.

In making my opinion, I relied upon the accuracy of the underlying claims and eligibility data records prepared by DHHS, as well as encounter data, financial data summaries, and other information prepared by the participating MCOs. A copy of the reliance letter received from DHHS is attached and constitutes part of this opinion. I did not audit the data and calculations, but did review them for reasonableness and consistency and did not find material defects. In other respects, my examination included such review of the underlying assumptions and methods used and such tests of the calculations as I considered necessary.

The capitation rates developed may not be appropriate for any specific MCO. Any MCO will need to review the rates in relation to the benefits provided. Each MCO should compare the rates with its own experience, expenses, capital and surplus, and profit requirements prior to agreeing to contract with DHHS. The MCO may require rates above, equal to, or below the actuarially sound capitation rates.

Actuarial methods, considerations, and analyses used in forming my opinion conform to the appropriate Standards of Practice as promulgated from time-to-time by the Actuarial Standards Board, whose standards form the basis of this Statement of Opinion.

It should be emphasized that capitation rates are a projection of future costs based on a set of assumptions. Actual costs will be dependent on each contracted MCO's situation and experience.

This Opinion assumes the reader is familiar with the New Hampshire Medicaid program, Medicaid eligibility rules, and actuarial rating techniques. The Opinion is intended for the State of New Hampshire and the Centers for Medicare and Medicaid Services and should not be relied on by other parties. The reader should be advised by actuaries or other professionals competent in the area of actuarial rate projections of the type in this Opinion, so as to properly interpret the projection results.

↓ohn D. Meerschaert Member, American Academy of Actuaries May 20, 2022

RELIANCE LETTER

State of New Hampshire Department of Health and Human Services SFY 2023 Capitation Rate Development for Medicaid Care Management Program

May 20, 2022

This report assumes that the reader is familiar with the State of New Hampshire's Medicaid program, its benefits, and rate setting principles. This report was prepared solely to provide assistance to DHHS to set SFY 2023 capitation rates for the Medicaid Care Management program. It may not be appropriate for other purposes. Milliman does not intend to benefit and assumes no duty or liability to other parties who receive this work. This material should only be reviewed in its entirety.



Lori A. Shibinette

Commissioner

Henry D. Lipman

Director

STATE OF NEW HAMPSHIRE

DEPARTMENT OF HEALTH AND HUMAN SERVICES

DIVISION OF MEDICAID SERVICES

129 PLEASANT STREET, CONCORD, NH 03301 1-844-ASK-DHHS (1-844-275-3447) Fax: 603-271-8431 TDD Access: 1-800-735-2964 www.dhhs.nh.gov

May 19, 2022

Mr. John D. Meerschaert, F.S.A. Milliman, Inc. 15800 Bluemound Road, Suite 100 Brookfield, WI 53005

Re: Actuarial Certification of SFY 2023 Capitation Rates for New Hampshire Medicaid Care Management Program Capitation Rates

Dear Mr. Meerschaert:

I, Henry Lipman, Medicaid Director for the New Hampshire Department of Health and Human Services, hereby affirm that the data prepared and submitted to Milliman, Inc. (Milliman) for the purpose of certifying the SFY 2023 New Hampshire Medicaid Care Management (MCM) program capitation rates were prepared under my direction, and to the best of my knowledge and belief are accurate and complete. This data includes:

- 1. Computer files supporting the SFY 2023 capitation rate calculation, including, but not limited to:
 - Technical Definition for NH MCM Data Book Services Scope V3.doc
 - Reference Files.xls
 - NH Provider Type Codes and Descriptions.xls
 - Eligibility Category Detail.xlsx
 - Medicaid CAHs.xls
 - OP-RHC-FQHC Reimbursement Process as of 2-5-13.doc
 - Provider Payment Algorithms2011.docx
 - QA LOG Care management 111412012 w corrected date at top.xlsx
 - Medicaid Extract and Claims Information.doc
 - NH+Medicaid+rebranded+detailed+FQHC+Provider+Manual+2-1-18.pdf
 - Newborn Reporting Procedures Guidance Statement 20121130.doc
 - NH Care Management Contract Exhibit A 031612.pdf
 - NH MCM Rate Cells Definition 2014-02-20.xls
 - Community Mental Health Agreement 1.22.15.pdf
 - Fiscal Impact Change of Scope & LAL SFY19.xlsx
 - FY20 21 Provider Rate Increases by Senate Bill and Milliman Exh 1.xlsx
 - NHM PDL 20220328.xlsx
 - PG02+PG09 Provider Numbers.xlsx
 - DME MP for Analysis 09_01_2020.xlsx
 - Initial Estimate Funding ALL MCOs (Pmts 0722-0623).xlsx
 - Neuro Psych (96131-96139) Utilization 2021.10.05.xlsx

Mr. John D. Meerschaert, F.S.A. May 19, 2022 Page 2 of 3

- Rate Increase Impacts-Peristeen A4459 & catheters.xlsx
- Metric_Quantity_Request with MMIS Units.xlsx
- 2. Fee schedule files:
 - 1) CY/SFY/FFY 2018 fee schedules:
 - · 2018 DRG Rate Sheet.xls
 - SFY 18 Hosp IP & OP.xls
 - 2018 NH Fee Schedule Covered Procedures 02232018.xlsx
 - 2018 NH Fee Schedule Manually Priced Procedures 02232018.xlsx
 - 2018 Medicaid Rates for Diagnostic and Physician Rates-20190108.xlsx
 - 2018 ASC Fee Schedule.xls
 - 2018 Hospice Rates worksheet-Final.xlsx
 - FQHC Based Rate SFY 2018.xls
 - SFY18 RATE CHANGE LOG.xlsx
 - 2019 FISCAL IMPACT Hospice Rates FINAL.xlsx
 - 2) CY/SFY/FFY 2019 fee schedules:
 - 2019 DRG Rate Sheet.xls
 - Fee Schedule 1-8-19.xlsx
 - Hospice Rates PDF File for Providers Eff 10_1_2018.pdf
 - Report of Covered Procedures 20190110.xlsx
 - Report of Manually Priced Procedures 20190110.xlsx
 - 3) CY/SFY/FFY 2023 fee schedules:
 - SFY23 RATE CHANGE LOG.xlsx
 - FFY22 DRG Rate Sheet .xlsx
 - Covered proedures Fee schedule as of 01012021.xlsx
 - Manual Priced Procedures 01012021.xlsx
 - 2022 Hospice Rates.xlsx
 - ASC Rates_20220125_Fee Sched_1.26.22.xlsx
 - J Code CSR Attachment 22020405.xlsx
- 3. January 2015 October 2021 Medicaid eligibility data and claims from MMIS and MCO encounter files, including:
 - Biweekly claims data (facility, professional and drug).
 - Biweekly enrollment data
 - Provider reference files.
 - Supplemental eligibility/ineligibility files
 - Additional FFS Hospice claims:
 - SFY 2018 Hospice data for Milliman.xlsx
 - 2020_3_4_Hospice_SFY2019 Data for Milliman.xlsx

Mr. John D. Meerschaert, F.S.A. May 19, 2022 Page 3 of 3

- Additional FFS CTS claims:
 - CTS FFS Encounter Data.csv
 - CTS PAP Encounter Data.csv
 - Additional BDAS claims:
 - tblWITS_SFY18_SvcInMedicaidSpan.xlsx
- Plan Selection Period info:
 - Fee for Service Spans for Andrew 1-1-2018 to current.xls
- IMD Claims:
 - NHH_Medicaid_2016-2019.xlsx
- 4. Other supporting documentation, including:
 - MCO contract
 - Financial Reconciliation data from participating health plans
 - MCO administrative cost survey results
 - ATECH services details
 - Behavioral Health Crisis Treatment Center (BHCTC) implementation and termination schedule
 - DRF identification process
 - Hospital directed payment details
 - Community Residential Service expansion details
 - Mobile Crisis Response Team and Emergency Services identification process
 - Home and Community Based Services directed payment details
 - Genetic Testing for Hospitals details
 - Fiscal impact and appropriation amounts for various legislative provisions
 - Other computer files
 - Conversations concerning supplied dat

ed data	01
Jar	20hr
Henry Lipman	

1

Medicaid Director

May 19, 2022

For more information about Milliman, please visit us at:

milliman.com



Milliman is among the world's largest providers of actuarial and related products and services. The firm has consulting practices in life insurance and financial services, property & casualty insurance, healthcare, and employee benefits. Founded in 1947, Milliman is an independent firm with offices in major cities around the globe.

milliman.com

CONTACT

John D. Meerschaert john.meerschaert@milliman.com

Greg J. Herrle gregory.j.herrle@milliman.com

Mathieu Doucet mathieu.doucet@milliman.com

Sarah Wunder sarah.wunder@milliman.com

Rachel Marsiglio rachel.marsiglio@milliman.com

Jacob Magnusson jacob.magnusson@milliman.com

© 2022 Milliman, Inc. All Rights Reserved. The materials in this document represent the opinion of the authors and are not representative of the views of Milliman, Inc. Milliman does not certify the information, nor does it guarantee the accuracy and completeness of such information. Use of such information is voluntary and should not be relied upon unless an independent review of its accuracy and completeness has been performed. Materials may not be reproduced without the express consent of Milliman.

Eligibility Category: Low Income Children - Age 0-11 Months

Member Months: 47,865

							Per Capita
Panafita	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Benefits Hospital Inpatient	Dollars	Aumits	Services	1,000	1,000	Charge	COSL
Medical	\$880,490	247	1,076	61.9	269.8	\$3,564.74	\$18.40
Surgical	261,151	32	194	8.0	48.6	8,160.96	5.46
Maternity Delivery	0	0	0	0.0	40.0	0.00	0.00
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00
Well Newborn	528	1	2	0.3	0.5	527.60	0.00
Psychiatric	0	0	0	0.0	0.0	0.00	0.00
Alcohol and Drug Abuse	Ő	ő	ů 0	0.0	0.0	0.00	0.00
Other	Ő	ő	ů 0	0.0	0.0	0.00	0.00
	\$1,142,168	280	1,272	70.2	318.9	\$4,079.17	\$23.86
Hospital Outpatient	¢1,112,100	200	.,		01010	¢ 1,07 0111	\$20.00
Emergency Room	\$863,979		3,505		878.7	\$246.50	\$18.05
Surgery	200,328		214		53.7	936.11	4.19
Radiology	180,149		823		206.3	218.89	3.76
Pathology	50,106		4,207		1,054.7	11.91	1.05
Pharmacy	65,217		11,416		2,862.1	5.71	1.36
Cardiovascular	16,215		141		35.3	115.00	0.34
PT/OT/ST	67,417		2,218		556.1	30.40	1.41
Psychiatric	0		0		0.0	0.00	0.00
Substance Abuse	0		0		0.0	0.00	0.00
Other	531,028		11,213		2,811.2	47.36	11.09
	\$1,974,440		33,737		8,458.1	\$58.52	\$41.25
Professional and Other State Plan Services	<i></i>				-,	,,,,,,	
Ambulatory Surgery Center	\$12,467		41		10.3	\$304.07	\$0.26
Office Visits	848,053		15,630		3,918.5	54.26	17.72
Preventive Medicine	922,037		47,180		11,828.3	19.54	19.26
Maternity	0		0		0.0	0.00	0.00
Certified Midwife	2,017		56		14.0	36.01	0.04
PT/OT/ST	24,467		931		233.4	26.28	0.51
Psychiatric and Substance Abuse	423		11		2.8	38.43	0.01
Radiology and Pathology	68,027		5,039		1,263,3	13.50	1.42
Home Health and Private Duty Nursing	276,125		8,409		2,108.2	32.84	5.77
Ambulance	85,229		5,021		1,258.8	16.97	1.78
Non-Emergency Transportation	74,717		1,563		391.9	47.80	1.56
Opioid Treatment Program	0		0		0.0	0.00	0.00
Federally Qualified and Rural Health Clinics	902,636		6,628		1,661.7	136.19	18.86
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	0		0		0.0	0.00	0.00
Durable Medical Equipment	1,330,922		537,249		134,691.7	2.48	27.81
Applied Behavioral Analysis	0		00		0.0	0.00	0.00
Other	1,219,659		39,044		9,788.6	31.24	25.48
	\$5,766,780		666,802		167,171.5	\$8.65	\$120.48
Prescription Drugs	<i>t</i> ,,		,				
Generic Scripts	\$160,832		11,904		2,984.4	\$13.51	\$3.36
Single-Source Brand	158,282		915		229.4	172.99	3.31
Multi-Source Brand	20,343		156		39.1	130.41	0.43
Specialty	592,077		146		36.6	4,055.32	12.37
Hepatitis C	0		0		0.0	0.00	0.00
Other	14		3		0.8	4.64	0.00
	\$931,547		13,124		3,290.3	\$70.98	\$19.46
Community Mental Health Center	¢001,011				0,20010	¢10.00	<i></i>
Case Management	\$0		0		0.0	\$0.00	\$0.00
Long Term Support Service	0		0		0.0	0.00	0.00
Partial Hospital	0		0		0.0	0.00	0.00
Psychotherapy	0		0		0.0	0.00	0.00
Evidence Based Practice	0		0		0.0	0.00	0.00
Medication Management	0		0		0.0	0.00	0.00
Emergency Service 24/7	0		0		0.0	0.00	0.00
APRTP	0		0		0.0	0.00	0.00
Supported Employment Services	0		0		0.0	0.00	0.00
Other	0		0		0.0	0.00	0.00
	\$0		0		0.0	\$0.00	\$0.00
All Services	\$9,814,935	280	714,935	70.2	179,238.7	\$13.73	\$205.06
	ψ0,017,000	200	7,000	10.2	110,200.1	μ ψιο./Ο	Ψ200.00

Eligibility Category: Low Income Children - Age 1-18 Years

Member Months: 729,792

							Per Capita	
	Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Monthly Paid	
Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost	
Hospital Inpatient								
Medical	\$1,749,195	454	1,963	7.5	32.3	\$3,852.85	\$2.40	
Surgical	1,062,053	128	640	2.1	10.5	8,297.29	1.46	
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00	
Maternity Non-Delivery	30,217	8	25	0.1	0.4	3,777.17	0.04	
Well Newborn	2,149	4	7	0.1	0.1	537.20	0.00	
Psychiatric	1,826,403	327	1,982	5.4	32.6	5,585.33	2.50	
Alcohol and Drug Abuse	19,219	4	43	0.1	0.7	4,804.73	0.03	
Other	0	0	0	0.0	0.0	0.00	0.00	
	\$4,689,236	925	4,660	15.2	76.6	\$5,069.44	\$6.43	
Hospital Outpatient			00.405			* ***	* 1 * 1 *	
Emergency Room	\$7,631,451		22,465		369.4	\$339.70	\$10.46	
Surgery	2,602,307		2,683		44.1	969.92	3.57	
Radiology	2,487,659		10,429		171.5	238.53	3.41	
Pathology	913,166		81,865		1,346.1	11.15	1.25	
Pharmacy	1,700,893		205,474		3,378.6	8.28	2.33	
Cardiovascular	66,701		812		13.4	82.14	0.09	
PT/OT/ST	858,692		26,719		439.3	32.14	1.18	
Psychiatric	202,104		841		13.8	240.31	0.28	
Substance Abuse	898		8		0.1	112.29	0.00	
Other	4,081,418		152,813		2,512.7	26.71	5.59	
	\$20,545,290		504,109		8,289.1	\$40.76	\$28.15	
Professional and Other State Plan Services								
Ambulatory Surgery Center	\$266,864		845		13.9	\$315.82	\$0.37	
Office Visits	6,942,743		124,357		2,044.8	55.83	9.51	
Preventive Medicine	2,772,104		110,533		1,817.5	25.08	3.80	
Maternity	59,952		268		4.4	223.65	0.08	
Certified Midwife	1,004		37		0.6	27.12	0.00	
PT/OT/ST	2,258,091		70,742		1,163.2	31.92	3.09	
Psychiatric and Substance Abuse	3,582,491		92,543		1,521.7	38.71	4.91	
Radiology and Pathology	869,194		66,476		1,093.1	13.08	1.19	
Home Health and Private Duty Nursing	1,889,624		149,301		2,455.0	12.66	2.59	
Ambulance	467,793		31,973		525.7	14.63	0.64	
Non-Emergency Transportation	912,139		18,826		309.6	48.45	1.25	
Opioid Treatment Program	307		30		0.5	10.22	0.00	
Federally Qualified and Rural Health Clinics	5,549,533		39,875		655.7	139.17	7.60	
Adult Medical Day Care	0,010,000		0		0.0	0.00	0.00	
Personal Care	563		24		0.0	23.47	0.00	
Durable Medical Equipment	3,261,882		1,339,259		22,021.5	2.44	4.47	
Applied Behavioral Analysis	2,836,623		198,849		3,269.7	14.27	3.89	
Other	6,183,633		439,548		7,227.5	14.07	8.47	
	\$37,854,539		2,683,486		44,124.6	\$14.11	\$51.87	
Prescription Drugs	ψ07,00 4 ,000		2,000,400		++,12+.0	ψι τ .ιι	ψ01.07	
Generic Scripts	\$6,884,254		225,905		3,714.6	\$30.47	\$9.43	
Single-Source Brand	6,245,400		30,643		503.9	203.81	8.56	
Multi-Source Brand	1,790,352		8,056		132.5	222.24	2.45	
Specialty	8,588,935		1,489		24.5	5,768.26	11.77	
Hepatitis C	0,500,955		1,409		0.0	0.00	0.00	
Other	1,810		66		1.1	27.43	0.00	
	\$23,510,751		266,159		4,376.5	\$88.33	\$32.22	
Community Mental Health Center	φ20,010,701		200,100		4,010.0	φ00.00	ψ02.22	
Case Management	\$644,294		1,835		30.2	\$351.11	\$0.88	
Long Term Support Service	331,450		5,552		91.3	59.70	0.45	
Partial Hospital	331,430		0,552		0.0	0.00	0.43	
Psychotherapy	250,487		3,024		49.7	82.83	0.00	
Evidence Based Practice	250,487		3,024 0		49.7	0.00	0.00	
Medication Management	80		03		0.0	26.64	0.00	
	80 94		3		0.0	20.04		
Emergency Service 24/7							0.00	
APRTP	10,431		19		0.3	549.00	0.01	
Supported Employment Services	0		0		0.0	0.00	0.00	
Other	436,114		3,241 13,678		53.3 224.9	134.56 \$122.31	0.60 \$2.29	
All Services	\$1,672,950 \$88,272,766	925	3,472,092	15.2	57,091.7	\$25.42	\$120.96	

Eligibility Category: Low Income Adults

Member Months: 124,392

							Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient				.,	.,		
Medical	\$1,672,375	488	2,177	47.1	210.0	\$3,427.00	\$13.44
Surgical	1,383,386	179	1,232	17.3	118.9	7,728.41	11.12
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	284,490	118	317	11.4	30.6	2,410.93	2.29
Well Newborn	20,402	32	70	3.1	6.8	637.57	0.16
Psychiatric	396,213	77	545	7.4	52.6	5,145.62	3.19
Alcohol and Drug Abuse	163,662	71	532	6.8	51.3	2,305.09	1.32
Other	0	0	0	0.0	0.0	0.00	0.00
	\$3,920,528	965	4,873	93.1	470.1	\$4,062.72	\$31.52
Hospital Outpatient							
Emergency Room	\$4,612,049		8,044		776.0	\$573.35	\$37.08
Surgery	1,348,164		1,545		149.0	872.60	10.84
Radiology	1,916,824		10,028		967.4	191.15	15.41
Pathology	882,166		69,143		6,670.2	12.76	7.09
Pharmacy	968,466		208,733		20,136.3	4.64	7.79
Cardiovascular	129,916		836		80.6	155.40	1.04
PT/OT/ST	337,832		11,599		1,118.9	29.13	2.72
Psychiatric	39,913		302		29.1	132.16	0.32
Substance Abuse	26,337		232		22.4	113.52	0.21
Other	1,878,692		69,028		6,659.1	27.22	15.10
	\$12,140,359		379,490		36,609.1	\$31.99	\$97.60
Professional and Other State Plan Services							
Ambulatory Surgery Center	\$128,718		422		40.7	\$305.02	\$1.03
Office Visits	2,376,773		41,715		4,024.2	56.98	19.11
Preventive Medicine	940,175		22,569		2,177.2	41.66	7.56
Maternity	1,268,542		4,945		477.0	256.54	10.20
Certified Midwife	29,592		164		15.8	180.92	0.24
PT/OT/ST	199,194		8,861		854.8	22.48	1.60
Psychiatric and Substance Abuse	2,402,881		26,424		2,549.1	90.94	19.32
Radiology and Pathology	1,666,819		62,334		6,013.3	26.74	13.40
Home Health and Private Duty Nursing	152,874		3,780		364.7	40.44	1.23
Ambulance	280,106		14,427		1,391.8	19.42	2.25
Non-Emergency Transportation	2,271,082		153,227		14,781.7	14.82	18.26
Opioid Treatment Program	1,544,437		151,119		14,578.3	10.22	12.42
Federally Qualified and Rural Health Clinics	2,430,268		18,257		1,761.2	133.12	19.54
Adult Medical Day Care	13,520		609		58.7	22.20	0.11
Personal Care	1,549		287		27.7	5.40	0.01
Durable Medical Equipment	605,390		41,770		4,029.5	14.49	4.87
Applied Behavioral Analysis	0		0		0.0	0.00	0.00
Other	2,847,864		216,880		20,922.3	13.13	22.89
	\$19,159,783		767,789		74,068.1	\$24.95	\$154.03
Prescription Drugs	* ********		150.054		15 000 1	A 15 A 3	* • • • • =
Generic Scripts	\$2,481,704		156,354		15,083.4	\$15.87	\$19.95
Single-Source Brand	3,672,688		14,053		1,355.7	261.35	29.53
Multi-Source Brand	2,624,090		12,848		1,239.4	204.24	21.10
Specialty	3,311,392		956		92.2	3,463.80	26.62
Hepatitis C	2,181,340		106		10.2	20,578.68	17.54
Other	385		22		2.1	17.50	0.00
Community Mental Health Center	\$14,271,600		184,339		17,783.1	\$77.42	\$114.73
-	\$46,558		130		12.5	\$358.14	\$0.37
Case Management							
Long Term Support Service Partial Hospital	95,524		1,190		114.8 0.0	80.27	0.77
•	0 162,903		0 3,567		344.1	0.00 45.67	0.00 1.31
Psychotherapy Evidence Based Practice	· · ·				344.1		0.00
	0		0			0.00	
Medication Management	1,598		60		5.8	26.64	0.01
Emergency Service 24/7	94		4		0.4	23.46	0.00
APRTP	30,744		56		5.4	549.00	0.25
Supported Employment Services			0		0.0	0.00	0.00
Other	190,672		2,578		248.7	73.96	1.53
All Services	\$528,093 \$50,020,363	965	7,585 1,344,076	93.1	731.7 129,662.1	\$69.62 \$37.22	\$4.25 \$402.12
All Services	φ00,020,303	905	1,344,070	yo.1	129,002.1	ېر <u>ت</u> و	φ402.1Z

Eligibility Category: CHIP

Member Months: 154,555

							Per Capita	
	Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid	
Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost	
Hospital Inpatient								
Medical	\$139,691	55	166	4.3	12.9	\$2,539.84	\$0.90	
Surgical	88,305	16	67	1.2	5.2	5,519.09	0.57	
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00	
Maternity Non-Delivery	2,712	1	1	0.1	0.1	2,711.98	0.02	
Well Newborn	, 0	0	0	0.0	0.0	0.00	0.00	
Psychiatric	367,648	56	514	4.3	39.9	6,565.14	2.38	
Alcohol and Drug Abuse	12,390	2	38	0.2	3.0	6,195.00	0.08	
Other	0	0	0	0.0	0.0	0.00	0.00	
otilei	\$610,747	130	786	10.1	61.0	\$4,698.05	\$3.95	
Hospital Outpatient	φ010,747	130	700	10.1	01.0	\$4,090.05	\$3.95	
Emergency Room	\$1,271,846		3,409		264.7	\$373.08	\$8.23	
			, , , , , , , , , , , , , , , , , , , ,					
Surgery	476,840		534		41.5	892.96	3.09	
Radiology	506,989		2,167		168.3	233.96	3.28	
Pathology	179,120		16,587		1,287.9	10.80	1.16	
Pharmacy	145,087		37,653		2,923.5	3.85	0.94	
Cardiovascular	22,657		247		19.2	91.73	0.15	
PT/OT/ST	218,361		7,088		550.3	30.81	1.41	
Psychiatric	39,228		155		12.0	253.08	0.25	
Substance Abuse	0		0		0.0	0.00	0.00	
Other	792,110		26,590		2,064.5	29.79	5.13	
-	\$3,652,237		94,430		7,331.8	\$38.68	\$23.63	
Professional and Other State Plan Services	<i></i>		.,		.,			
Ambulatory Surgery Center	\$57,184		184		14.3	\$310.78	\$0.37	
Office Visits	1,605,267		29,159		2,264.0	55.05	10.39	
Preventive Medicine	680,091		28,999		2,204.0	23.45	4.40	
					· · · ·			
Maternity	3,751		14		1.1	274.72	0.02	
Certified Midwife	253		7		0.5	36.21	0.00	
PT/OT/ST	681,587		20,382		1,582.5	33.44	4.41	
Psychiatric and Substance Abuse	640,755		13,275		1,030.7	48.27	4.15	
Radiology and Pathology	191,117		15,284		1,186.7	12.50	1.24	
Home Health and Private Duty Nursing	265,283		33,931		2,634.5	7.82	1.72	
Ambulance	65,407		4,683		363.6	13.97	0.42	
Non-Emergency Transportation	141,467		1,759		136.6	80.42	0.92	
Opioid Treatment Program	0		0		0.0	0.00	0.00	
Federally Qualified and Rural Health Clinics	737,150		5,742		445.8	128.38	4.77	
Adult Medical Day Care	0		0		0.0	0.00	0.00	
Personal Care	94		4		0.3	23.47	0.00	
Durable Medical Equipment	646,040		192,764		14,966.7	3.35	4.18	
Applied Behavioral Analysis	612,065		41,924		3,255.1	14.60	3.96	
Other			86,870		6,744.8	14.59	8.20	
Other	1,267,320				,			
Description Description	\$7,594,832		474,981		36,878.7	\$15.99	\$49.14	
Prescription Drugs	A		15 005			AAA 50	* • • •	
Generic Scripts	\$1,375,311		45,005		3,494.3	\$30.56	\$8.90	
Single-Source Brand	1,486,815		6,679		518.6	222.61	9.62	
Multi-Source Brand	276,017		1,710		132.8	161.41	1.79	
Specialty	2,332,314		400		31.1	5,830.79	15.09	
Hepatitis C	0		0		0.0	0.00	0.00	
Other	111		6		0.5	18.57	0.00	
	\$5,470,569		53,800		4,177.2	\$101.68	\$35.40	
Community Mental Health Center								
Case Management	\$80,223		226		17.5	\$354.97	\$0.52	
Long Term Support Service	51,500		1,148		89.1	44.86	0.33	
Partial Hospital	01,000		0		0.0	0.00	0.00	
Psychotherapy	36,999		434		33.7	85.25	0.00	
			434					
Evidence Based Practice	0				0.0	0.00	0.00	
Medication Management	0		0		0.0	0.00	0.00	
Emergency Service 24/7	0		0		0.0	0.00	0.00	
	0		0		0.0	0.00	0.00	
APRTP		1	1					
Supported Employment Services	0		0		0.0	0.00	0.00	
	0 73,323		578		44.9	126.86	0.47	
Supported Employment Services	0	130		10.1				

Eligibility Category: Foster Care / Adoption

Member Months: 20,976

							Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient	Donars	Admits	Services	1,000	1,000	Charge	0031
Medical	\$76,177	24	71	13.7	40.6	\$3,174.04	\$3.63
Surgical	15,552	3	5	1.7	2.9	5,184.07	0.74
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	Ő	0	0	0.0	0.0	0.00	0.00
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	306,226	40	464	22.9	265.4	7,655.65	14.60
Alcohol and Drug Abuse	0	0	0	0.0	0.0	0.00	0.00
Other	Ő	0	0	0.0	0.0	0.00	0.00
	\$397,955	67	540	38.3	308.9	\$5,939.63	\$18.97
Hospital Outpatient	+					++,++++++++++++++++++++++++++++++++++++	
Emergency Room	\$263,976		603		345.0	\$437.77	\$12.58
Surgery	98,371		106		60.6	928.03	4.69
Radiology	87,090		413		236.3	210.87	4.15
Pathology	41,746		3,563		2,038.3	11.72	1.99
Pharmacy	19,225		7,847		4,489.1	2.45	0.92
Cardiovascular	5,182		48		27.5	107.97	0.25
PT/OT/ST	72,316		2,280		1,304.3	31.72	3.45
Psychiatric	527		6		3.4	87.82	0.03
Substance Abuse	0		0		0.0	0.00	0.00
Other	199,894		6,265		3,584.1	31.91	9.53
	\$788,327		21,131		12,088.6	\$37.31	\$37.58
Professional and Other State Plan Services	, .				,		
Ambulatory Surgery Center	\$8,721		31		17.7	\$281.31	\$0.42
Office Visits	261,450		4,567		2,612.7	57.25	12.46
Preventive Medicine	86,916		3,687		2,109.3	23.57	4.14
Maternity	2,615		4		2.3	655.39	0.12
Certified Midwife	0		0		0.0	0.00	0.00
PT/OT/ST	111,678		3,807		2,177.9	29.33	5.32
Psychiatric and Substance Abuse	287,521		4,392		2,512.6	65.46	13.71
Radiology and Pathology	49,707		2,619		1,498,3	18.98	2.37
Home Health and Private Duty Nursing	761,166		18,985		10,860,9	40.09	36.29
Ambulance	32,294		2,160		1,235.7	14.95	1.54
Non-Emergency Transportation	46,203		1,257		719.1	36.76	2.20
Opioid Treatment Program	0		0		0.0	0.00	0.00
Federally Qualified and Rural Health Clinics	165,920		1,254		717.4	132.31	7.91
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	0		0		0.0	0.00	0.00
Durable Medical Equipment	384,776		185,445		106,089.2	2.07	18.34
Applied Behavioral Analysis	128,775		8,585		4,911.3	15.00	6.14
Other	260,051		17,577		10,055.4	14.79	12.40
	\$2,587,793		254,370		145,519.8	\$10.17	\$123.37
Prescription Drugs							
Generic Scripts	\$686,824		20,660		11,819.2	\$33.24	\$32.74
Single-Source Brand	384,852		1,694		969.1	227.19	18.35
Multi-Source Brand	149,925		375		214.5	399.80	7.15
Specialty	419,290		132		75.5	3,176.44	19.99
Hepatitis C	0		0		0.0	0.00	0.00
Other	22		3		1.7	7.33	0.00
	\$1,640,913		22,864		13,080.0	\$71.77	\$78.23
Community Mental Health Center							
Case Management	\$50,856		146		83.5	\$348.33	\$2.42
Long Term Support Service	20,264		284		162.5	71.35	0.97
Partial Hospital	0		0		0.0	0.00	0.00
Psychotherapy	21,360		306		175.1	69.80	1.02
Evidence Based Practice	0		0		0.0	0.00	0.00
Medication Management	0		0		0.0	0.00	0.00
Emergency Service 24/7	Ő		0		0.0	0.00	0.00
APRTP	Ő		0		0.0	0.00	0.00
Supported Employment Services	0		0		0.0	0.00	0.00
Other	48,284		393		224.8	122.86	2.30
	\$140,763		1,129		645.9	\$124.68	\$6.71
All Services	\$5,555,752	67	300,034	38.3	171,643.2	\$18.52	\$264.86
	++,000,10E	51	500,004			\$10.0E	\$201.00

Eligibility Category: Severely Disabled Children

Member Months: 11,180

							Bor Copito	
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Per Capita Monthly Paid Cost	
Hospital Inpatient	Dollars	Aumits	Services	1,000	1,000	Charge	COSI	
Medical	\$333,372	39	303	41.9	325.2	\$8,547.99	\$29.82	
Surgical	151,327	10	34	10.7	36.5	15,132.70	13.54	
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00	
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00	
Well Newborn	0	0	0	0.0	0.0	0.00	0.00	
Psychiatric	13,230	2	13	2.1	14.0	6,614.80	1.18	
Alcohol and Drug Abuse	0	2	0	0.0	0.0	0.00	0.00	
Other	0	0	0	0.0	0.0	0.00	0.00	
	\$497.928	51	350	54.7	375.7	\$9,763.30	\$44.54	
Hospital Outpatient	ψ491,920	51			515.1	\$9,703.30	φ44.04	
Emergency Room	\$86,129		180		193.2	\$478.49	\$7.70	
Surgery	86,501		82		88.0	1,054.89	7.74	
Radiology	129,661		286		307.0	453.36	11.60	
Pathology	29,623		2,570		2,758.4	11.53	2.65	
Pharmacy	26,645		6,806		7,305.0	3.91	2.38	
Cardiovascular	4,338		46		49.4	94.30	0.39	
PT/OT/ST	125,073		3,787		4,064.7	33.03	11.19	
Psychiatric	237		4		4.3	59.31	0.02	
Substance Abuse	0		0		0.0	0.00	0.00	
Other	112,739		2,574		2,762.7	43.80	10.08	
	\$600,944		16,335		17,532.7	\$36.79	\$53.75	
Professional and Other State Plan Services	\$000,044		10,000		11,002.1	\$00.70	\$00.10	
Ambulatory Surgery Center	\$4,995		22		23.6	\$227.03	\$0.45	
Office Visits	184,552		2,396		2,571.7	77.02	16.51	
Preventive Medicine	19,909		746		800.7	26.69	1.78	
Maternity	0		0		0.0	0.00	0.00	
Certified Midwife	0		0		0.0	0.00	0.00	
PT/OT/ST	315,426		13,620		14.618.6	23.16	28.21	
Psychiatric and Substance Abuse	98,286		4,711		5,056.4	20.86	8.79	
Radiology and Pathology	34,214		2,173		2,332.3	15.74	3.06	
Home Health and Private Duty Nursing	3,973,413		312,544		335,459.6	12.71	355.40	
Ambulance	26,687		3,162		3,393.8	8.44	2.39	
Non-Emergency Transportation	79,367		4,661		5,002.7	17.03	7.10	
Opioid Treatment Program	0		4,001		0.0	0.00	0.00	
Federally Qualified and Rural Health Clinics	14,479		128		137.4	113.12	1.30	
Adult Medical Day Care	0		0		0.0	0.00	0.00	
Personal Care	100		8		8.6	12.50	0.00	
Durable Medical Equipment	1,572,676		942,747		1,011,868.9	1.67	140.67	
Applied Behavioral Analysis	632,222		45,776		49,132.3	13.81	56.55	
Other	670,226		20,108		21,582.3	33.33	59.95	
	\$7,626,551		1,352,802		1,451,989.0	\$5.64	\$682.14	
Prescription Drugs	ψ1,020,001		1,002,002		1,401,000.0	φ0.04	ψ002.14	
Generic Scripts	\$533,203		15,690		16,840.4	\$33.98	\$47.69	
Single-Source Brand	397,242		1,693		1,817.1	234.64	35.53	
Multi-Source Brand	745,502		1,075		1,153.8	693.49	66.68	
Specialty	1,207,571		296		317.7	4,079.63	108.01	
Hepatitis C	0		230		0.0	0.00	0.00	
Other	268		3		3.2	89.21	0.00	
	\$2,883,787		18,757		20,132.3	\$153.74	\$257.94	
Community Mental Health Center	ψ2,003,707		10,707		20,132.3	\$155.74	ψ207.94	
Case Management	\$12,893		36		38.6	\$358.14	\$1.15	
Long Term Support Service	22,474		924		991.7	24.32	2.01	
Partial Hospital	0		324 0		0.0	0.00	0.00	
Psychotherapy	2,280		27		29.0	84.45	0.00	
Evidence Based Practice	2,200		0		29.0	0.00	0.20	
Medication Management	0		0		0.0	0.00	0.00	
Emergency Service 24/7	0		0		0.0	0.00	0.00	
APRTP	0		0		0.0			
	1					0.00	0.00	
Supported Employment Services	0		0		0.0	0.00	0.00	
Other	6,521		63		67.6	103.50	0.58	
All Services	\$44,168 \$11,653,378	51	1,050 1,389,294	54.7	1,127.0 1,491,156.6	\$42.06 \$8.39	\$3.95 \$1,042.32	
All Sel VICES	φ11,000,078	51	1,309,294	J4./	1,491,130.0	۵0.39 <u>م</u>	φ1,042.3Z	

Eligibility Category: Elderly and Disabled Adults - Age 19-64

Member Months: 76,828

						Por Capita	
	Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid
Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost
Hospital Inpatient							
Medical	\$5,080,452	1,072	7,793	167.4	1,217.2	\$4,739.23	\$66.13
Surgical	3,048,274	363	3,020	56.7	471.7	8,397.45	39.68
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	11,842	5	10	0.8	1.6	2,368.46	0.15
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	460,093	98	768	15.3	120.0	4,694.82	5.99
Alcohol and Drug Abuse	177,284	61	536	9.5	83.7	2,906.29	2.31
Other	0	0	0	0.0	0.0	0.00	0.00
	\$8,777,944	1,599	12,127	249.8	1,894.2	\$5,489.65	\$114.25
Hospital Outpatient							
Emergency Room	\$4,021,987		5,623		878.3	\$715.27	\$52.35
Surgery	1,518,876		2,043		319.1	743.45	19.77
Radiology	2,092,134		7,434		1,161.1	281.43	27.23
Pathology	606,124		51,178		7,993.6	11.84	7.89
Pharmacy	3,135,167		297,716		46,501.1	10.53	40.81
Cardiovascular	149,840		1,157		180.7	129.51	1.95
PT/OT/ST	509,828		17,848		2,787.7	28.57	6.64
Psychiatric	9,509		91		14.2	104.49	0.12
Substance Abuse	7,636		68		10.6	112.29	0.10
Other	2,410,480		68,804		10,746.7	35.03	31.37
	\$14,461,580		451,962		70,593.2	\$32.00	\$188.23
Professional and Other State Plan Services			,		· · ·		
Ambulatory Surgery Center	\$184,620		628		98.1	\$293.98	\$2.40
Office Visits	2,169,682		35,718		5,578.9	60.74	28.24
Preventive Medicine	212,085		8,207		1,281.9	25.84	2.76
Maternity	25,556		99		15.5	257.68	0.33
Certified Midwife	178		7		1.1	25.38	0.00
PT/OT/ST	301,743		13,560		2,118.0	22.25	3.93
Psychiatric and Substance Abuse	1,124,413		15,866		2,478.2	70.87	14.64
Radiology and Pathology	970,578		39,043		6,098.2	24.86	12.63
Home Health and Private Duty Nursing	5,067,181		298,047		46,552.8	17.00	65.95
Ambulance	554,479		32,680		5,104.4	16.97	7.22
Non-Emergency Transportation	2,166,935		120,212		18,776.3	18.03	28.20
Opioid Treatment Program	734,590		71,878		11,226.8	10.00	9.56
Federally Qualified and Rural Health Clinics	1,366,789		9,519		1,486.8	143.59	17.79
Adult Medical Day Care	211,997		11,153		1,742.0	19.01	2.76
Personal Care	1,333,730		246,676		38,529.0	5.41	17.36
Durable Medical Equipment	2,865,219		1,247,272		194,814.9	2.30	37.29
Applied Behavioral Analysis	55,344		4,828		754.1	11.46	0.72
Other	4,695,954		318,415		49,734.1	14.75	61.12
	\$24,041,073		2,473,808		386,391.1	\$9.72	\$312.92
Prescription Drugs	ψ24,041,070		2,470,000		000,001.1	ψ5.72	ψ012.02
Generic Scripts	\$4,422,341		261,892		40,905.7	\$16.89	\$57.56
Single-Source Brand	11,184,768		32,923		5.142.3	339.73	145.58
Multi-Source Brand	4,984,389		12,918		2,017.7	385.85	64.88
Specialty	11,446,184		1,773		276.9	6,455.83	148.98
Hepatitis C	1.421.400		69		10.8	20,600.01	18.50
Other	786		73		10.0	10.77	0.01
	\$33,459,868		309,648		48,364.8	\$108.06	\$435.52
Community Mental Health Center	<i>\\</i> 00,100,000		000,010		40,004.0	\$100.00	\$100.0 <u>2</u>
Case Management	\$56,944		159		24.8	\$358.14	\$0.74
Long Term Support Service	102,508		1,766		275.8	58.05	1.33
Partial Hospital	0		0		0.0	0.00	0.00
Psychotherapy	135,594		2,365		369.4	57.33	1.76
Evidence Based Practice	398				2.3	26.54	0.01
			15 176		2.3 27.5	26.54	0.01
Medication Management	5,835						
Emergency Service 24/7	446		19		3.0	23.46	0.01
APRTP	21,411		39		6.1	549.00	0.28
Supported Employment Services	2,840		107		16.7	26.54	0.04
Other	191,024		2,627		410.3	72.72	2.49
All Sandiago	\$517,001	1 500	7,273	040.0	1,136.0	\$71.08	\$6.73 \$1,057.65
All Services	\$81,257,467	1,599	3,254,818	249.8	508,379.3	\$24.97	\$1,US7.65

Eligibility Category: Elderly and Disabled Adults - Age 65+

Member Months: 12,505

							Por Capita	
P Ch.	Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid	
Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost	
Hospital Inpatient	\$4.445.055	050	4 700	044.0	4 000 4	¢4 405 CO	¢00.40	
Medical	\$1,115,255	252	1,733	241.8	1,663.1	\$4,425.62	\$89.19	
Surgical	490,617	65	576	62.4	552.8	7,547.96	39.23	
Maternity Delivery Maternity Non-Delivery	0	0	0	0.0	0.0 0.0	0.00 0.00	0.00	
	0	-		0.0			0.00	
Well Newborn	0	0 7	0	0.0	0.0	0.00	0.00	
Psychiatric	46,010	0	361	6.7	346.4	6,572.80	3.68	
Alcohol and Drug Abuse	0	0	0	0.0 0.0	0.0 0.0	0.00 0.00	0.00 0.00	
Other	\$1,651,882	324	2,670	310.9	2,562.2	\$5,098.40	\$132.10	
Hospital Outpatient	\$1,031,002	324	2,070	510.9	2,502.2	\$5,096.40	\$132.1U	
Emergency Room	\$537,069		594		570.0	\$904.16	\$42.95	
Surgery	166,864		256		245.7	651.81	13.34	
Radiology	404,417		1,501		1,440.4	269.43	32.34	
Pathology	103,006		10,227		9,814.2	10.07	8.24	
Pharmacy	297,998		32,796		31,472.3	9.09	23.83	
Cardiovascular	24,833		262		251.4	94.78	1.99	
PT/OT/ST	65,907		2,609		2,503.7	25.26	5.27	
Psychiatric	05,507		2,009		2,505.7	0.00	0.00	
Substance Abuse	0		0		0.0	0.00	0.00	
Other	502,812		7,594		7,287.5	66.21	40.21	
	\$2,102,905		55.839		53,585.3	\$37.66	\$168.17	
Professional and Other State Plan Services	φ2,102,505		55,055		55,505.5	ψ37.00	φ100.1 <i>1</i>	
Ambulatory Surgery Center	\$62,368		273		262.0	\$228.45	\$4.99	
Office Visits	333,980		5,418		5.199.3	61.64	26.71	
Preventive Medicine	32,790		1,182		1,134.3	27.74	2.62	
Maternity	0		0		0.0	0.00	0.00	
Certified Midwife	0		0		0.0	0.00	0.00	
PT/OT/ST	24,742		1,018		976.9	24.30	1.98	
Psychiatric and Substance Abuse	13,983		312		299.4	44.82	1.12	
Radiology and Pathology	153,374		6,631		6.363.4	23.13	12.27	
Home Health and Private Duty Nursing	742,268		8,553		8,207.8	86.78	59.36	
Ambulance	96,321		3,720		3,569.9	25.89	7.70	
Non-Emergency Transportation	173,149		9,981		9,578.2	17.35	13.85	
Opioid Treatment Program	12,939		1,266		1,214.9	10.22	1.03	
Federally Qualified and Rural Health Clinics	276,233		2,054		1,971.1	134.49	22.09	
Adult Medical Day Care	392,865		20,146		19,332.9	19.50	31.42	
Personal Care	0		0		0.0	0.00	0.00	
Durable Medical Equipment	265,862		134,547		129,116.6	1.98	21.26	
Applied Behavioral Analysis	0		0		0.0	0.00	0.00	
Other	818,483		32,312		31,007.9	25.33	65.45	
	\$3,399,356		227,413		218,234.5	\$14.95	\$271.85	
Prescription Drugs	\$0,000,000		221,110		210,20110	\$11100	\$211.00	
Generic Scripts	\$563,967		51,890		49,795.7	\$10.87	\$45.10	
Single-Source Brand	1,656,591		6,414		6,155.1	258.28	132.48	
Multi-Source Brand	402,368		1,002		961.6	401.57	32.18	
Specialty	588,867		170		163.1	3,463.92	47.09	
Hepatitis C	61,921		3		2.9	20,640.31	4.95	
Other	68		17		16.3	4.01	0.01	
	\$3,273,782		59,496		57,094.7	\$55.03	\$261.80	
Community Mental Health Center			,		. ,			
Case Management	\$358		2		1.9	\$179.07	\$0.03	
Long Term Support Service	3,021		36		34.5	83.91	0.24	
Partial Hospital	0		0		0.0	0.00	0.00	
Psychotherapy	2,142		27		25.9	79.32	0.17	
Evidence Based Practice	0		0		0.0	0.00	0.00	
Medication Management	0		0		0.0	0.00	0.00	
Emergency Service 24/7	0		0		0.0	0.00	0.00	
APRTP	0		0		0.0	0.00	0.00	
Supported Employment Services	0		0		0.0	0.00	0.00	
Other	10,711		475		455.8	22.55	0.86	
	\$16,232		540		518.2	\$30.06	\$1.30	
All Services	\$10,444,157	324	345,958	310.9	331,994.9	\$30.19	\$835.22	
	1,01,777,01	024	540,000	010.0	001,007.0	ψου.19	9000.2Z	

Eligibility Category: Dual Eligibles

Member Months: 166,049

Benefits Total Paid Total Paid Total Paid Vulnesses Average Paid Norms/Paid Hogial Inpatient 53,373,261 2,271 11,817 164.1 83,95,8 51,485.38 520,31 Surgein 986,224 520 4,830 37,8 33,46 1,000,9 5,00 0,00								Per Capita	
Heapital matient S. 373,261 2.271 11,817 164.1 6385.5 51,485.86 520.31 Surgial 99.224 522 4,630 37.8 539.65 509.05 Maternity Delivery werey 0 0 0 0.00 <t< th=""><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>Monthly Paid</th></t<>								Monthly Paid	
Medical 33.372.201 2.271 11.617 1764.1 38.36 51.86.36 S20.3 Maternity Non-Delivery 9 0		Dollars	Admits	Services	1,000	1,000	Charge	Cost	
Surgical 995/234 5/23 4.6300 37.6 5/33.6 1,020.201 5.59 Maternity Non-Delivery 11,477 7 19 0.5 1.4 1,035.67 0.000 Maternity Non-Delivery 11,477 7 19 0.5 1.4 1,035.67 0.000 Maternity Non-Delivery 497.4216 2.490 1.23 3.1 5,222.802 0.800 Acrohal and Drug Abuse 49.474.216 2.490 1.8,873 216.1 1.285.9 51.606.62 529.90 Chery 1.44.442 4.803 3.271.5 1.900.7 7.83 81.602.2 6.540 7.78 1.900.2 6.540 7.88 0.33 2.225 6.54 7.88 0.345 1.355 9.233.1 9.022.2 6.54 7.88 0.34 2.215.318 6.564.44 4.0097.5 3.99 1.33.4 Cadioxascular 1.55.95 2.81.9 1.34.3 0.34 2.245.1 3.03.45 1.99.41.33 0.34 2.24.10 9.34 2.245.1	· · ·								
Materinty Non-Delivery 1 0									
Materning Non-Delivery 11 / 479 7 19 0.5 1.4 1.838.7 0.07 Vell Newborn 497,597 156 2,181 1.13 1583 3,186.73 3.00 Chine 497,597 156 2,181 1.13 1583 3,185.76 3.00 Chine 49,974.216 2,269 10.872 2.16 1,283.9 3,165.56 49.93.22 6.25 Emergency Room 33,157.450 7.743 549.65 540.778 51.90.2 6.25 Radiology 1,328.409 1.15.03 634.1 22.90.5 5.60 1.14 1.333 90.22 6.25 Pharmacy 198,630 3.1967 2.200.5 5.60 1.14 1.333 0.22 6.25 Photogy 198,630 3.1967 2.200.5 5.60 1.14 1.333 0.22 6.26 5.00 1.14 1.333 0.22 6.26 7.99 1.14 1.33 0.24 6.00 1.33 0.23 <t< td=""><td></td><td></td><td></td><td>,</td><td></td><td></td><td></td><td></td></t<>				,					
Weil Newborn 0 <t< td=""><td></td><td></td><td></td><td>-</td><td></td><td></td><td></td><td></td></t<>				-					
Psychiatric 497,507 150 2,191 11.3 158.3 3,199.73 3.00 Actool and Dug Abure 0 0 0 0 0 0 0 0.00									
Akoho and Drug Abuse 0		-	-						
Other 0 0 0 0 0 0 0 0.00 Hesplat Oupationt 33, 157, 450 2,890 18,873 216,1 1,963,8 \$1,66,562 \$29,98 Burgency, Component 53, 157, 450 7,743 556,6 \$407,76 \$1,902 \$51,853,000 \$25,133 \$0,220 \$61,87 \$25,133 \$0,220 \$61,87 \$25,133 \$0,220 \$61,87 \$25,133 \$0,220 \$61,87 \$39,90 \$13,34 Cardiovascular 159,580 54,844 40,007,5 \$3,99 \$13,34 \$2,28 \$14,13 \$34,228 \$14,13,33 \$2,28 \$14,13 \$34,228 \$14,13 \$34,228 \$14,13 \$34,228 \$14,28 \$14,33 \$2,28 \$16,86 \$14,21 \$13,34 \$2,28 \$16,86 \$14,21 \$16,85 \$16,86 \$14,21 \$16,85 \$16,86 \$14,22 \$16,85 \$16,86 \$14,22 \$16,85 \$16,86 \$14,22 \$16,85 \$16,85 \$14,26 \$16,86 \$14,26 \$,			, ·		
S4.974.216 2.990 18.873 216.1 1.365.9 \$1.63.82 \$2.99.6 Emergency Room \$3.157.450 7.743 555.6 \$407.78 \$19.03 \$7.52 Radiology 1.037.809 11.630 \$33.13 50.22 \$2.55 Pathology 188.830 31.957 \$2.095 \$5.50 1.14 Preamagular 2.15.116 56.444 40.097.5 3.99 13.33 \$2.65 Prioting 74.665 26.105 20.314 41.33 \$2.65 \$2.014 \$2.03 \$2.014 \$2.03 \$2.014 \$2.03 \$2.014 \$2.03 \$2.014 \$2.03 \$2.015 \$2.66.20 \$2.014 \$2.05 \$2.00 \$2.014 \$2.05 \$2.00 \$2.014 \$2.05 \$2.00 \$2.014 \$2.05 \$2.02 \$2.14.20 \$2.67 \$2.05 \$2.02 \$2.14.20 \$2.67 \$2.05 \$2.02 \$2.14.20 \$2.05 \$2.05 \$2.014 \$2.05 \$2.014 \$2.05 \$2.014 \$2.015 <									
Hospital Outpatient Emergency Norm \$3,157,450 7,743 S556 \$407,78 \$19,02 Surgery 1,284,442 4,603 347,1 259,33 7,52 Radiology 1,037,800 11,803 631,3 80,22 6.25 Pathology 186,800 31,957 2,206,5 5.90 1,144 Cardiovascular 193,800 2,351 165,9 2,309,5 5.99 1,334 Cardiovascular 193,800 2,351 165,9 2,313 12,28 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,29 0,33 112,39 112,39 112,39 112,39 112,39 112,39 112,39 112,39 112,39 112,39 <t< td=""><td>Other</td><td></td><td>-</td><td></td><td></td><td></td><td></td><td></td></t<>	Other		-						
Emergency Room \$3,167,460 7,743 5666 \$407.78 \$19.02 Radiology 1.037,809 11,633 831.3 90.22 6.25 Pathology 1.88,830 31,957 2.030.5 5.39 1.14 Pharmacy 2.215,136 554,644 40.097.5 3.99 3.34 Cardiovascular 159,580 2.23.51 169.9 67.88 0.966 PYOITSIT 374,666 28,109 2.03.4 13.33 12.28 0.03 Substance Abuse 5,165 44 10.2066 19.28 16.88 Other 2.770.017 14.3.040 10.208.6 19.28 16.88 Professional and Other State Pian Services 811.21.306 76.06 52.80 18.17 0.79 Ambulardo Synopy Center 161.6906 66.447 4.30.9 94.128 0.03 Office Visits 161.6906 77.06 52.80 18.17 0.79 Preventive Medicine 130.962 7.706 62.80 18.417		\$4,974,216	2,990	18,873	216.1	1,363.9	\$1,663.62	\$29.96	
Surgery 1.248.442 4.803 347.1 229.93 7.52 Pathology 188.830 31.157 2.309.5 5.50 1.14 Pharmacy 2.215.318 554.844 40.097.5 3.99.22 6.25 Pathology 159.860 2.351 100.9 67.88 0.86 Cardiovascular 59.965 1.336 87.9 41.31 0.34 Substance Abuse 27.165 143.06 10.3.8 112.28 0.86 Other 511.210.056 786.351 56.820.0 314.26 567.53 Ambulatory Surgery Center \$106.435 870 6.2.9 \$121.19 50.63 Office Visits 1.06.806 6.8.47 4.830.9 24.19 9.7.2 Preventive Madicine 10.9.62 7.206 520.8 18.17 0.79 Maternity 2.4.57 7.23 19.8 88.86 0.05 Protosional and Other State Plan Services 44.45 13 0.9 3.2.6 0.00		A0 457 450		7 7 40		550.0	0407 70	\$10.00	
Radiology 1.037.809 11.503 881.3 90.22 6.65 Pathology 188.630 31.957 2.309.5 5.50 1.14 Pharmacy 2.215.313 554.844 40.007.5 3.99 13.34 Cardiovascular 159.580 2.351 10.99 67.88 0.96 Pri/DT/ST 374.665 2.8.109 2.031.4 13.33 2.26 Substance Abuse 5.165 46 3.3 112.29 0.03 Other 2.2770.017 143.640 10.380.6 19.28 16.88 Profescional and Other State Pian Services Anabiatory Starp Center 510.64.35 707 6.82.80 514.48 507.53 Profescional and Other State Pian Services 7.066 520.8 18.17 0.79 9.88 0.15 Anabiatory Starp Venter 1.616.909 7.026 520.8 18.17 0.79 Malermity 2.4567 7.206 520.8 18.17 0.79 Malermity 2.4577 7.13 <									
Pathology 188.630 31.957 2.309.5 5.50 1.14 Pharmacy 2.215.318 554.844 40.097.5 3.89 13.34 Cardiovascular 159.580 2.351 196.9 67.88 0.96 Psychiatric 55.980 1.355 97.9 41.31 0.34 Substance Abuse 5,165 46 3.3 112.29 0.83 Other 2.770.017 143.640 10.380.6 19.28 16.86 Professional and Other State Plan Services 37.065 766.531 56.928.0 51.426 56.753 Ambulatory Surgery Center 1.616.906 66.847 4.830.8 22.19 9.74 Ambulatory Surgery Center 1.649.902 7.206 50.208 10.17 7.73 19.8 88.88 0.15 Centlied Midwile 106.777 12.33 9.33 4.84 0.40 Proventive Medicine 106.902 7.206 53.28 10.75 3.28 10.92 Autified AdWavile 1.66									
Pharmacy Cardiovascular 2,215,318 554,844 40,007.5 3.99 13.34 Cardiovascular 159,580 2,351 1069 67.88 0.96 PT/OT/ST 374,665 2,8,109 2,031.4 13.33 2.26 Substance Abuse 5,165 4.66 3.3 112.29 0.03 Other 2,770,017 143,464 10.380.6 19.28 16.88 Professional and Other State Plan Services 766,351 56.828.0 \$14.26 \$97.9 Ambulatory Surgery Center \$10,64,35 8.70 6.2.9 \$12.19 \$0.83 Office Visis 13.69,62 7.206 520.8 18.17 0.79 Preventive Medicine 130,902 7.206 520.8 18.17 0.79 Priot/TST 106.77 12.084 13.80 8.58 0.60 Priot/TST 106.77 12.084 13.84 0.84 0.80 Protorial Anabiance 149.90 2.456 1.80.4 3.864 0.80	6,								
Cardiovascular 159,580 2,351 169.9 67.88 0.96 PiyOT/ST 374,665 28,109 2.031.4 13.33 2.66 Psychiatric 55,980 1.355 97.9 41.31 0.34 Other 2,770,017 143,640 10,380.6 19.28 16.68 Professional and Other State Plan Services 786,331 56.828.0 \$14.213,096 66.847 4.830.9 24.19 \$7.43 Protessional and Other State Plan Services 706 52.06 18.17 0.79 9.44.19 \$7.44 \$7.49 9.74 Preventive Medicine 130,962 7.20.6 52.06 18.17 0.79 Materinity 24,570 27.3 19.8 88.8 0.05 Pri/T/ST 106,779 12.064 87.3.3 8.84 0.64 Psychiatric and Substance Abuse 943,302 24.9668 1.80.4 38.02 5.72 Radiology and Pathology 511,531 49.416 3.577.1 10.35 3.08		· · ·				,			
PTIOTIST 374.665 28.109 2.031.4 13.33 2.26 Psychiatric 55.965 4.6 3.3 112.29 0.03 Other 2.770,071 143.640 10.380.6 19.28 15.65 Ambulatory Surgery Center \$112,10,956 766.351 56.82.0 \$14.26 \$57.53 Professional and Other State Plan Services -									
Psychiatric 55,980 1,355 97.9 41.31 0.34 Other 2,770,017 143,640 10.380.6 1122,0 0.13 Other 2,770,017 143,640 10.380.6 1122,0 0.13 Professional and Other State Plan Services 786,351 556,828.0 \$14,21,0 \$56,828.0 \$14,21,0 \$57,0 Ambulatory Surgery Center \$106,169.0 66,847 4,830.9 \$24,19 \$9,43.2 Offee Visits 1616,906 66,847 4,830.9 \$24,19 \$9,43.2 Preventive Medicine 130,962 7,206 520.8 18,17 0.9 Matemity 24,570 273 19.8 89.88 0.05 Profinition and Substance Abuse 943,302 24,49.68 1.804.4 38.02 5.72 Radiody and Pathology 511,531 49,416 3.57.1 10.35 3.08 Non-Emergency Transportation 4316,868 106,625 14,186.1 249.99 26.00 Opioid Treatment Program 598,760 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>									
Substance Abuse 5,165 466 3.3 112.29 0.03 Other 2,770.017 143.460 10.38.00 19.28 16.68 Professional and Other State Plan Services \$11.213.056 786.351 566.828.0 \$14.26 \$67.53 Ambulatory Surgery Center 1516.435 870 62.9 \$12.11.9 \$0.83 Ofter Visits 1616.906 66.847 4.80.9 24.19 \$9.74 Preventive Medicine 130.962 7.206 520.8 18.17 0.77 Matemity 24.570 27.3 19.8 80.88 0.17 Prometive Medicine 445 13 0.9 34.26 0.00 PriOTST 106.779 12.04 873.3 8.84 0.64 Prome Health and Private Duly Nursing 2.124.101 168.697 12.263.7 12.52 12.95 Ambulance 1.274.624 109.521 7.914.9 1164 7.65 On-Emergency Transportation 4.315.68 165.05 14.05.51				,					
Other 2.770.017 143.640 10.380.6 19.28 16.68 811.213.056 766.351 56.828.0 \$11.42 \$57.83 Professional and Other State Plan Services \$105.435 870 6.29 \$121.19 \$0.63 Ambulatory Surgery Center \$105.455 870 6.29 \$121.19 \$0.63 Office Visits 1.616.906 66.847 4.830.9 24.19 \$0.63 Preventive Medicine 1.30.692 7.706 \$20.8 18.17 0.79 Maternity 24.570 273 19.8 80.88 0.15 Centified Midwle 24.570 273 19.8 80.84 0.64 Psychiatric and Substance Abuse 949.302 24.4968 1.804.4 3.671.2 10.03 3.08 Home Health and Private Duty Nursing 2.124.101 166.697 1.226.3 11.252 12.78 Non-Emergency Transportation 4.316.888 196.285 14.185.1 2.199 26.00 Opiold Testatment Program 598.760	•								
\$11,213,056 766,351 56,828.0 \$14,28 \$87,53 Ambulatory Surgery Center \$105,435 870 62.9 \$121,19 \$6,637 Office Visits 130,962 7,206 520,8 18,17 0.79 Preventive Medicine 130,962 7,206 520,8 18,17 0.79 Preventive Medicine 146,5006 66,847 4,830.9 24,19 9,74 Preventive Medicine 146,5006 72,06 520,8 18,17 0.79 Preventive Medicine 4455 13 0.9 34,26 0.00 PritOTIST 106,779 12,084 873.3 8,84 0.04 Preventive Duty Nursing 2,124,101 169,697 12,263.7 12,52 12.79 Non-Emergency Transportation 4,316,868 199,285 14,185 21.99 26.00 Opidi Trasmant Program 5,002,878 2,8597 4,224.0 10.22 3.61 Aduit Medicial Equipment 2,4593.5 15,538 1,172.9 1.582									
Professional and Other State Plan Services Ambulatory Surgery Center \$105,435 870 62.9 \$121.19 \$50,633 Office Visits 1.616,906 66,847 4,830.9 24.19 9.74 Preventive Medicine 130,962 7,206 520.8 18.17 0.79 Certified Midwife 445 13 0.9 34.26 0.00 PriOT/ST 106,779 12,084 167,33 0.9 34.26 0.00 PriOT/ST 106,779 12,084 167,33 3.84 0.64 Psychiatric and Substance Abuse 949,302 24,968 1,004,4 3.8.02 5.7.2 Addiology and Pathology 511,531 49,416 3.57.1 10.35 3.08 Home Heatth and Private Duty Nursing 2,124,101 196,657 4,234.0 10.22 3.17 Opioid Treatment Program 596,760 58,587 4,234.0 10.22 3.61 Apulted Behavioral Analysis 1,680 112 8.1 15.00 0.110 16.01 Durable M	Other					.,			
Ambulatory Surgery Center \$105.435 870 62.2 \$121.19 \$0.63 Office Visits 1.616.906 66.847 4,830.9 24.19 974 Preventive Medicine 130.962 7.206 520.8 18.17 0.79 Maternity 24,570 273 19.8 89.88 0.15 Certified Midwife 445 13 0.9 34.26 0.00 PTIOT/ST 106.779 12.084 873.3 8.84 0.64 Peychiatric and Substance Abuse 949.302 24,968 1,804.4 38.02 5.72 Radiology and Pathology 511.531 49.416 3.571.2 10.35 3.08 Non-Emergency Transportation 4.316.686 196.525 14.185.1 21.99 26.00 Opioid Treatment Program 598.760 56.587 4.234.0 10.22 3.013 Durable Medical Equipment 2.659.248 2.616 189.1 95.52 1.50 Adult Medical Day Care 5.00.278 925.946 6661.63		\$11,213,056		786,351		56,828.0	\$14.26	\$67.53	
Office Visits 1.616.906 66.8.47 4.830.9 2.4.19 9.74 Preventive Medicine 130.962 7.206 520.8 18.17 0.79 Maternity 24,570 273 19.8 89.88 0.15 Certified Midwife 445 13 0.9 34.26 0.00 PT/0T/ST 106,779 12.084 873.3 8.84 0.64 Psychiatric and Substance Abuse 949,302 24.968 1.80.4 38.02 5.72 Radiology and Pathology 511,531 49,416 3.571.2 10.3,571.2 10.3,571.2 10.25 12.79 Ambulance 1,274,824 109,521 7.914.9 11.64 7.68 11.29 28.00 Opiold Treatment Program 598.760 56.587 4.234.0 10.22 3.61 Aduit Medical Day Care 245.835 15.538 1,112.9 15.82 1.68 Durable Medical Equipment 2.069,244 2.423.709 175.160.9 1.10 16.01 Applied Sehavior		0405 405		070		00.0		* 0.00	
Preventive Medicine '130,962 7,206 520.8 18.17 0.79 Maternity 24,570 273 19.8 88,98.8 0.15 Certified Midwire 4445 13 0.9 34.26 0.00 PT/OT/ST 106,779 12,084 873.3 8.84 0.84 Peychiatric and Substance Abuse 949,302 24,968 1.804.4 36.02 5.72.8 Radiology and Pathology 511,531 49.416 3.571.2 10.35 3.08 Home Health and Private Duty Nursing 2.124.101 169.697 12.263.7 12.52 12.7.79 Ambulance 1.274,824 199.525 14.185.1 2.19.9 2.6.00 Opioid Treatment Program 598,760 55.587 4.234.0 10.22 3.51 Federally Qualified and Rural Health Clinics 248,878 2.6.16 189.1 9.55.2 1.50 Autil Medical Equipment 2.659.244 2.423.766 1.75.10 3.64 2.42.9 Applied Behavioral Analysis 1.680	, , ,								
Matemity 24,570 273 19.8 89.88 0.15 Certified Midwife 445 13 0.9 34.26 0.00 Pri/DT/ST 106,779 12,084 873.3 8.84 0.06 Psychiatric and Substance Abuse 949,302 24,968 1,804.4 38.02 5.72 Radiology and Pathology 511,531 49,416 3,571.2 10.35 3.08 Home Health and Private Duty Nursing 2,124,101 166,697 12,263.7 12,52 12,79 Non-Emergency Transportation 4,316,868 196,252 7,914.9 11.64 7.68 Nor-Emergency Transportation 4,316,868 196,253 14,185.1 21.99 28.00 Opioid Treatment Program 598,676 58,587 4,234.0 10.22 3.61 Personal Care 2,659,244 2,62,946 66,916.3 5.40 30.13 Darable Medical Equipment 2,659,244 2,42,769 175,160.9 1.10 16.01 Other 3,997,652 4,715,317 <td></td> <td></td> <td></td> <td>,</td> <td></td> <td></td> <td></td> <td></td>				,					
Certified Midwife 445 13 0.9 34.26 0.00 PT/OT/ST 106.779 12.084 873.3 8.84 0.64 Psychiatric and Substance Abuse 949.302 24.968 1,804.4 38.02 5.72 Radiology and Pathology 511.531 494.416 3.571.2 10.35 3.08 Mome Health and Private Duty Nursing 2,124.101 166.697 12.263.7 12.52 12.79 Ambulance 1.274.824 109.521 7,914.9 11.64 7.68 Non-Emergency Transportation 43.61.6868 199.285 14.185.1 21.99 26.00 Opioid Treatment Program 598.760 55.657 4.234.0 10.22 3.61 Adutt Medical Day Care 2.458.35 15.538 1,122.9 15.82 1.48 Durable Medical Equipment 2.659.244 2.423.769 175.160.9 1.0 16.01 Applied Behavioral Analysis 1.680 142 7.53 12.26 0.00 Generic Scripts \$163.795 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>									
PT/OTST 106,779 12,084 873.3 8.84 0.64 Psychiatric and Substance Abuse 949,302 24,968 1,804.4 38.02 5.72 Radiology and Pathology 511,531 49,416 3,571.2 10.35 3.08 Home Health and Private Duty Nursing 2,124,101 108,997 12,263.7 12,52 12,79 Non-Emergency Transportation 4,316,868 196,285 14,185.1 21.99 26,00 Opicid Treatment Program 598,760 58,887 4,234.0 10.22 3.61 Federally Qualified and Rural Health Clinics 249,878 2,616 189,1 95,52 1.60 Ault Medical Day Care 24,6583 115,538 1,112.9 15,82 1.48 Parsonal Care 5.002,878 925,946 669,916.3 5.40 30.13 Durable Medical Equipment 2,659,244 2,423,769 175,160.9 1.10 16.01 Applied Behavioral Analysis 1,680 5163,795 45,752 3,306.4 \$3.550 \$3.09 <t< td=""><td>J</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	J								
Psychiatric and Substance Abuse 949,302 24,068 1,804.4 38.02 5.72 Radiology and Pathology 511,531 49,416 3,571.2 10.35 3.08 Home Health and Private Duty Nursing 2,124,101 168,897 12,263.7 12,52 12,79 Ambulance 1,274,824 109,521 7,914.9 11.64 7,68 Non-Emergency Transportation 4,316,868 196,285 141,185.1 2.199 28.00 Opioid Treatment Program 598,760 58,587 4.234.0 10.22 3.61 Federally Qualified and Rural Health Clinics 249,878 26.16 189.1 95.52 1.50 Adult Medical Day Care 2,659,244 2,423,768 175,160.9 1.10 16.01 Applied Behavioral Analysis 1,680 112 8.1 15.00 0.01 Otree 3,997,652 651,566 47,076.6 6.14 42.08 Prescriptio Drugs				-					
Radiology and Pathology 511.531 49.416 3.571.2 10.35 3.08 Home Health and Private Duty Nursing 2,124,101 169.697 12,263.7 12.52 12.79 Ambulance 1,274,824 109,521 7,914.9 11.64 7,68 Non-Emergency Transportation 4,316,868 196,285 14,185.1 21.99 26.00 Opioid Treatment Program 598,760 58,587 4,234.0 10.22 3.61 Federally Qualified and Rural Health Clinics 249,878 26,616 189.1 95.52 1.50 Adult Medical Day Care 2,602,878 925,946 66,916.3 5.40 30.13 Durable Medical Equipment 2,659,244 2,423,769 1175,160.9 1.10 16.01 Applied Behavioral Analysis 1,680 112 8.1 15.00 0.01 Other 3,997,652 651,569 47,087.6 5.14 24.08 Single-Source Brand 12,773 1,042 75.3 12.26 0.08 Single-Source Brand		106,779						0.64	
Home Health and Private Duty Nursing 2.124.101 169.697 12.263.7 12.22 12.79 Ambulance 1,274,824 109.521 7,914.9 11.64 7.68 Non-Emergency Transportation 4,316,868 196,285 14,185.1 21.99 Opioid Treatment Program 598,760 58,857 4.234.0 10.22 3.61 Federally Qualified and Rural Health Clinics 249,878 2.616 189.1 95.52 1.50 Adult Medical Day Care 245,835 15,538 1,12.9 15.82 1.48 Personal Care 5,002,878 925,546 66,916.3 5.40 30.13 Durable Medical Equipment 2,259,46 66,916.3 5.40 30.13 Durable Medical Equipment 2,283,917,650 4,715,317 340,766.5 \$5.07 \$144.04 Prescription Drugs 6 47,075.3 12.26 0.08 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 <td< td=""><td>5</td><td></td><td></td><td>,</td><td></td><td></td><td></td><td></td></td<>	5			,					
Ambulance 1.274.824 199.521 7.914.9 11.64 7.68 Non-Emergency Transportation 4.316.868 196.285 14.185.1 21.99 26.00 Opicid Treatment Program 598.760 58.567 4.234.0 10.22 3.61 Federally Qualified and Rural Health Clinics 249.873 2.616 188.1 99.552 1.50 Adult Medical Day Care 245.835 15.538 1.122.9 15.82 1.48 Personal Care 5.002.878 925.946 66.916.3 5.40 30.13 Durable Medical Equipment 2.659.244 2.423.769 175.160.9 1.10 16.01 Other 3.997.652 651.569 47.087.6 6.14 24.08 Generic Scripts \$163.795 45.752 3.306.4 \$3.58 \$0.99 Single-Source Brand 12.773 1.042 75.3 12.26 0.08 Multi-Source Brand 6.490 483 34.9 13.44 0.04 Specialty 0 0 0	Radiology and Pathology	511,531		49,416		3,571.2	10.35	3.08	
Non-Emergency Transportation 4,316,868 196,285 14,185,1 21,99 26,00 Opioid Treatment Program 598,760 58,587 4,234,0 10,22 3,61 Federally Quilified and Rural Health Clinics 249,878 2,616 189,1 95,52 1,50 Adult Medical Day Care 245,835 15,538 1,122,9 15,82 1,48 Personal Care 5,002,878 2,925,946 66,916,3 5,400 30,13 Durable Medical Equipment 2,659,244 2,423,769 175,160,9 1,10 16,01 Applied Behavioral Analysis 1,680 112 8,1 15,00 0,01 Other \$23,917,650 4,715,317 340,766,5 \$5,07 \$144,04 Prescription Drugs Generic Scripts \$163,795 45,752 3,306,4 \$3,58 \$0.99 Single-Source Brand 12,773 1,042 75,3 12,26 0,00 Walti-Source Brand 12,773 1,042 75,3 3,49,9 13,44 0,04	Home Health and Private Duty Nursing	2,124,101		169,697			12.52	12.79	
Opioid Treatment Program 598,760 58,587 4,234.0 10.22 3.61 Federally Qualified and Rural Health Clinics 249,878 2,616 189.1 95.52 1.50 Adult Medical Day Care 5,002,878 925,946 669,916.3 5.40 30.13 Durable Medical Equipment 2,659,244 2,423,769 175,160.9 1.10 16.01 Optice Behavioral Analysis 1,680 112 8.1 15.00 0.01 Other 3,997,652 651,569 47,087.6 6.14 24.03 Prescription Drugs \$163,795 45,752 3,306.4 \$3.58 \$0.99 Single-Source Brand 12,277 1,042 75.3 12.26 0.08 Multi-Source Brand 6,490 483 34.9 13.44 0.04 Specialty 0 0 0.0 0.00 0.00 0.00 Other 480 63 4.6 7.61 0.00 0.00 Other 480 63 4.6 <td< td=""><td>Ambulance</td><td></td><td></td><td></td><td></td><td>7,914.9</td><td>11.64</td><td></td></td<>	Ambulance					7,914.9	11.64		
Federally Qualified and Rural Health Clinics 249,878 2,616 189.1 95.52 1.50 Adult Medical Day Care 245,835 15,538 1,122.9 15.82 1.48 Personal Care 5,002,878 925,946 66,916.3 5.400 30.112 Durable Medical Equipment 2,659,244 2,423,769 175,160.9 1.10 16.01 Applied Behavioral Analysis 1,680 112 8.1 15.00 0.01 Other 3,997,652 651,569 47,087.6 6.14 24.08 Prescription Drugs	Non-Emergency Transportation	4,316,868		196,285		14,185.1	21.99	26.00	
Adult Medical Day Care 245,835 15,538 1,122.9 15.82 1.48 Personal Care 5,002,878 925,944 66,916.3 5.40 30.13 Durable Medical Equipment 2,659,244 2,423,769 175,160.9 1.10 16.01 Applied Behavioral Analysis 1,680 112 8.1 15.00 0.01 Other 3,997,652 651,569 47,087.6 6.14 24.08 Generic Scripts \$163,795 4,715,317 340,766.5 \$5.07 \$144.04 Specialty \$163,795 45,752 3,306.4 \$3.58 \$0.99 Single-Source Brand 12,773 1,042 75.3 12.26 0.08 Multi-Source Brand 6,490 483 34.9 13.44 0.04 Specialty 0 0 0 0.00 0.00 0.00 Other 480 63 4.6 7.61 0.00 0.00 Community Mental Health Center C 15 \$3.62 0.01	Opioid Treatment Program	598,760		58,587				3.61	
Personal Care 5,002,878 925,946 66,916.3 5.40 30.13 Durable Medical Equipment 2,659,244 2,423,769 175,160.9 1.10 16.01 Applied Behavioral Analysis 1,680 112 8.1 15.00 0.01 Other 3,997,652 651,569 47,087.6 6.14 24.08 Prescription Drugs 4,715,317 340,766.5 \$5.07 \$144.04 Generic Scripts \$163,795 45,752 3,306.4 \$3.58 \$0.99 Single-Source Brand 12,773 1,042 75.3 12.26 0.08 Multi-Source Brand 6,490 483 34.9 13.44 0.04 Specialty 0 0 0 0.00 0.00 0.00 Other 480 63 4.6 7.61 0.000 0.00 0.00 0.00 0.00 0.05 0.55 0.05 0.55 0.05 1.5 \$367.60 \$0.35 0.35 0.35 0.35 0.00 0.0	Federally Qualified and Rural Health Clinics	249,878		2,616		189.1	95.52	1.50	
Durable Medical Equipment 2,659,244 2,423,769 175,160.9 1.10 16.01 Applied Behavioral Analysis 1,680 112 8.1 15.00 0.01 Other \$23,917,650 4,715,317 340,766.5 \$5.07 \$144.04 Prescription Drugs Generic Scripts \$163,795 45,752 3,306.4 \$3.38 \$0.99 Single-Source Brand 12,773 1,042 75.3 12.26 0.08 Multi-Source Brand 6,490 483 34.9 13.44 0.04 Specially 0 0 0.0 0.00 0.00 0.00 Other 480 63 4.6 7.61 0.00 0.00 Other 480 63 4.6 7.61 0.00 0.00 Community Mental Health Center 20.06 21 11.5 \$367.60 \$0.35 Long Term Support Service 133,702 27.97 202.1 47.80 0.81 Partial Hospital 2.006 21 <	Adult Medical Day Care	245,835		15,538		1,122.9	15.82	1.48	
Applied Behavioral Analysis 1,680 112 8.1 15.00 0.01 Other 3,997,652 651,569 47,087.6 6.14 24.08 Prescription Drugs 4,715,317 340,766.5 \$5.07 \$144.04 Prescription Drugs 4,715,317 340,766.5 \$5.07 \$144.04 Generic Scripts \$163,795 45,752 3,306.4 \$3.58 \$0.99 Single-Source Brand 12,773 1,042 75.3 12.26 0.08 Multi-Source Brand 6,490 483 34.9 13.44 0.04 Specialty 0 0 0.0 0.00 0.00 0.00 Hepatitis C 0 0 0.0 0.00 <td>Personal Care</td> <td>5,002,878</td> <td></td> <td>925,946</td> <td></td> <td>66,916.3</td> <td>5.40</td> <td>30.13</td>	Personal Care	5,002,878		925,946		66,916.3	5.40	30.13	
Other 3,997,652 651,569 47,087.6 6.14 24.08 Prescription Drugs \$23,917,650 4,715,317 340,766.5 \$5.07 \$144.04 Generic Scripts \$163,795 45,752 3,306.4 \$3.58 \$0.99 Single-Source Brand 12,773 1,042 75.3 12.26 0.08 Multi-Source Brand 6,490 483 34.9 13.44 0.04 Specialty 0 0 0.00 0.00 0.00 0.00 Hepatitis C 0 0 0.00 0.00 0.00 0.00 Other 480 63 4.6 7.61 0.00 0.00 Other \$183,538 47,340 3,421.2 \$3.88 \$11.1 Community Mental Health Center \$133,702 2,797 202.1 47.80 0.81 Long Term Support Service 133,702 2,797 202.1 47.80 0.81 Partial Hospital 2,006 21 1.5 95.52 0	Durable Medical Equipment	2,659,244		2,423,769		175,160.9	1.10	16.01	
\$23,917,650 4,715,317 340,766.5 \$5.07 \$144.04 Prescription Drugs \$163,795 45,752 3,306.4 \$3.58 \$0.99 Single-Source Brand 12,773 1,042 75.3 12.26 0.08 Multi-Source Brand 6,490 483 34.9 13.44 0.04 Specialty 0 0 0.0 0.00 0.00 0.00 Hepatitis C 0 0 0.0 0.00 0.00 0.00 Other 480 63 46.6 7.61 0.00 0.00 Community Mental Health Center	Applied Behavioral Analysis	1,680		112		8.1	15.00	0.01	
Prescription Drugs Generic Scripts \$163,795 45,752 3,306.4 \$3.58 \$0.99 Single-Source Brand 12,773 1,042 75.3 12.26 0.08 Multi-Source Brand 6,490 483 34.9 13.44 0.04 Specialty 0 0 0 0.00 0.00 0.00 Hepatitis C 0 0 0 0.00 0.00 0.00 Other 480 63 4.6 7.61 0.00 Other \$183,538 47,340 3,421.2 \$3.88 \$111 Community Mental Health Center -	Other	3,997,652		651,569		47,087.6	6.14	24.08	
Generic Scripts \$163,795 45,752 3,306.4 \$3.58 \$0.99 Single-Source Brand 12,773 1,042 75.3 12.26 0.08 Multi-Source Brand 6,490 483 34.9 13.44 0.04 Specialty 0 0 0 0.00 0.00 0.00 Hepatitis C 0 0 0 0.00 0.00 0.00 Other 480 63 4.6 7.61 0.00 Community Mental Heatth Center		\$23,917,650		4,715,317		340,766.5	\$5.07	\$144.04	
Single-Source Brand 12,773 1,042 75.3 12.26 0.08 Multi-Source Brand 6,490 483 34.9 13.44 0.04 Specialty 0 0 0.0 0.00 0.00 0.00 Hepatitis C 0 0 0.0 0.05 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>									
Multi-Source Brand 6,490 483 34.9 13.44 0.04 Specialty 0 0 0 0.0 0.01 0.00 0.01 0.01 0.01	Generic Scripts			,		3,306.4			
Specialty 0 0 0 0.0 0.00 0.00 0.00 Hepatitis C 0 0 0 0.0 0.00 Stastast Stas									
Hepatitis C 0 0 0 0.0 0.00 0.	Multi-Source Brand	6,490						0.04	
Other 480 63 4.6 7.61 0.00 \$183,538 47,340 3,421.2 \$3.88 \$1.11 Community Mental Health Center	Specialty	0		0		0.0	0.00	0.00	
\$183,538 47,340 3,421.2 \$3.88 \$1.11 Community Mental Health Center \$58,449 159 11.5 \$367.60 \$0.35 Case Management \$58,449 2,797 202.1 47.80 0.81 Partial Hospital 2,006 21 1.5 95.52 0.01 Psychotherapy 95,425 3,296 238.2 28.95 0.57 Evidence Based Practice 335 24 1.7 13.95 0.00 Medication Management 7,613 279 20.2 27.29 0.055 Emergency Service 24/7 117 5 0.4 23.46 0.00 APRTP 23,933 43 3.1 556.57 0.14 Supported Employment Services 4,501 163 11.8 27.61 0.03 Other 217,110 7,553 545.8 28.74 1.31 \$543,190 14,340 1,036.3 \$37.88 \$3.27	Hepatitis C	0		0		0.0	0.00	0.00	
Community Mental Health Center 50.0	Other								
Case Management \$58,449 159 11.5 \$367.60 \$0.35 Long Term Support Service 133,702 2,797 202.1 47.80 0.81 Partial Hospital 2,006 21 1.5 995.52 0.01 Psychotherapy 95,425 3,296 238.2 28.95 0.57 Evidence Based Practice 335 224 1.7 13.95 0.00 Medication Management 7,613 279 20.2 27.29 0.05 Emergency Service 24/7 117 5 0.4 23.46 0.00 APRTP 23,933 43 3.1 556.57 0.14 Supported Employment Services 4,501 163 11.8 27.61 0.03 Other 217,110 7,553 545.8 28.74 1.31 Sta3,190 14,340 1,036.3 \$37.88 \$3.27		\$183,538		47,340		3,421.2	\$3.88	\$1.11	
Long Term Support Service 133,702 2,797 202.1 47.80 0.81 Partial Hospital 2,006 21 1.5 95.52 0.01 Psychotherapy 95,425 3,296 238.2 28.95 0.57 Evidence Based Practice 335 24 1.7 13.95 0.00 Medication Management 7,613 279 202.2 27.29 0.05 Emergency Service 24/7 117 5 0.4 23.46 0.00 APRTP 23,933 43 3.1 556.57 0.14 Supported Employment Services 4,501 163 11.8 27.61 0.03 Other 217,10 7,553 545.8 28.74 1.31 Supported Employment Services \$\$43,190 14,340 1,036.3 \$\$37.88 \$\$32.77	•								
Partial Hospital 2,006 21 1.5 95.52 0.01 Psychotherapy 95,425 3,296 238.2 28.95 0.57 Evidence Based Practice 335 24 1.7 13.95 0.00 Medication Management 7,613 279 20.2 27.29 0.05 Emergency Service 24/7 117 5 0.4 23.46 0.00 APRTP 23,933 43 3.1 556.57 0.14 Supported Employment Services 4,501 163 11.8 27.61 0.00 Other 217,110 7,553 545.8 28.74 1.31 Supported Employment Services \$543,190 14,340 1,036.3 \$337.88 \$32.77		1 I							
Psychotherapy 95,425 3,296 238.2 28.95 0.57 Evidence Based Practice 335 24 1.7 13.95 0.00 Medication Management 7,613 279 20.2 27.29 0.05 Emergency Service 24/7 117 5 0.4 23.46 0.00 APRTP 23,933 43 3.1 556.57 0.14 Supported Employment Services 4,501 163 11.8 27.61 0.03 Other 217,110 7,553 545.8 28.74 1.31									
Evidence Based Practice 335 24 1.7 13.95 0.00 Medication Management 7,613 279 20.2 27.29 0.05 Emergency Service 24/7 117 5 0.4 23.46 0.00 APRTP 23,933 43 3.1 556.57 0.14 Supported Employment Services 4,501 163 118 27.61 0.03 Other 217,110 7,553 545.8 28.74 1.31									
Medication Management 7,613 279 20.2 27.29 0.05 Emergency Service 24/7 117 5 0.4 23.46 0.00 APRTP 23,933 43 3.1 556.57 0.14 Supported Employment Services 4,501 163 118 27.61 0.03 Other 217,110 7,553 545.8 28.74 1.31									
Emergency Service 24/7 117 5 0.4 23.46 0.00 APRTP 23,933 43 3.1 556.57 0.14 Supported Employment Services 4,501 163 11.8 27.61 0.00 Other 217,110 7,553 545.8 28.74 1.31 \$543,190 14,340 1,036.3 \$37.88 \$3.27									
APRTP 23,933 43 3.1 556.57 0.14 Supported Employment Services 4,501 163 11.8 27.61 0.03 Other 217,110 7,553 545.8 28.74 1.31 \$543,190 14,340 1,036.3 \$37.88 \$3.27									
Supported Employment Services 4,501 163 11.8 27.61 0.03 Other 217,110 7,553 545.8 28.74 1.31 \$543,190 14,340 1,036.3 \$37.88 \$32.71									
Other 217,110 7,553 545.8 28.74 1.31 \$543,190 14,340 1,036.3 \$37.88 \$3.27						3.1	556.57		
\$543,190 14,340 1,036.3 \$37.88 \$3.27									
	Other								
All Services \$40,831,649 2,990 5,582,221 216.1 403,416.0 \$7.31 \$245.90									
	All Services	\$40,831,649	2,990	5,582,221	216.1	403,416.0	\$7.31	\$245.90	

Eligibility Category: Newborn Kick Payment

Member Months: 2,296

							Por Canita	
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Per Capita Monthly Paid Cost	
Hospital Inpatient	Donars	Aunits	00111005	1,000	1,000	Charge	0051	
Medical	\$4,721,086	996	6,823	433.8	2,971.7	\$4,740.05	\$2,056.22	
Surgical	49,579	4	36	1.7	15.7	12,394.86	21.59	
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00	
Maternity Non-Delivery	0	ů 0	0	0.0	0.0	0.00	0.00	
Well Newborn	803,787	1,311	2,805	571.0	1,221.7	613.11	350.08	
Psychiatric	000,707	0	2,009	0.0	0.0	0.00	0.00	
Alcohol and Drug Abuse	0	0	0	0.0	0.0	0.00	0.00	
Other	0	ů 0	0	0.0	0.0	0.00	0.00	
	\$5,574,452	2,311	9,664	1,006.5	4,209.1	\$2,412.14	\$2,427.90	
Hospital Outpatient	\$0,01 1,10L	2,011	0,001	1,000.0	.,200.1	<i>\</i> _ , _	<i>\\\\\\\\\\\\\</i>	
Emergency Room	\$85,118		291		126.7	\$292.50	\$37.07	
Surgery	6,070		14		6.1	433.56	2.64	
Radiology	14,689		97		42.2	151.43	6.40	
Pathology	8,022		1,022		445.1	7.85	3.49	
Pharmacy	1,014		472		205.6	2.15	0.44	
Cardiovascular	3,360		27		11.8	124.43	1.46	
PT/OT/ST	1,393		25		10.9	55.73	0.61	
Psychiatric	0		0		0.0	0.00	0.00	
Substance Abuse	0		0		0.0	0.00	0.00	
Other	109,575		1,278		556.6	85.74	47.72	
	\$229,241		3,226		1,405.1	\$71.06	\$99.84	
Professional and Other State Plan Services	\$220,211		0,220		1,100.1	<i></i>	<i>\</i>	
Ambulatory Surgery Center	\$0		0		0.0	\$0.00	\$0.00	
Office Visits	92,782		1,815		790.5	51.12	40.41	
Preventive Medicine	249,024		5,738		2,499.1	43.40	108.46	
Maternity	0		0,100		0.0	0.00	0.00	
Certified Midwife	549		13		5.7	42.26	0.24	
PT/OT/ST	351		10		5.2	29.24	0.15	
Psychiatric and Substance Abuse	0		0		0.0	0.00	0.00	
Radiology and Pathology	19,049		1,372		597.6	13.88	8.30	
Home Health and Private Duty Nursing	63,509		791		344.5	80.29	27.66	
Ambulance	47,851		2,132		928.6	22.44	20.84	
Non-Emergency Transportation	4,428		195		84.9	22.71	1.93	
Opioid Treatment Program	0		0		0.0	0.00	0.00	
Federally Qualified and Rural Health Clinics	171,970		1,040		453.0	165.36	74.90	
Adult Medical Day Care	0		0		0.0	0.00	0.00	
Personal Care	0		0		0.0	0.00	0.00	
Durable Medical Equipment	20,979		10,534		4,588.0	1.99	9.14	
Applied Behavioral Analysis	0		0		0.0	0.00	0.00	
Other	813,114		12,114		5,276.1	67.12	354.14	
	\$1,483,605		35,756		15,573.2	\$41.49	\$646.17	
Prescription Drugs	\$1,100,000				10,010.2	<i>\</i>	<i>Q</i> U U U	
Generic Scripts	\$4,160		499		217.3	\$8.34	\$1.81	
Single-Source Brand	2,005		23		10.0	87.15	0.87	
Multi-Source Brand	151		10		4.4	15.10	0.07	
Specialty	102				0.4	101.69	0.04	
Hepatitis C	0		0		0.0	0.00	0.00	
Other	7		2		0.9	3.40	0.00	
	\$6,424		535		233.0	\$12.01	\$2.80	
Community Mental Health Center	¢0,121				200.0	¢12.01	\$2.00	
Case Management	\$0		0		0.0	\$0.00	\$0.00	
Long Term Support Service	0		0		0.0	0.00	0.00	
Partial Hospital	0		0		0.0	0.00	0.00	
Psychotherapy	0		0		0.0	0.00	0.00	
Evidence Based Practice	0		0		0.0	0.00	0.00	
Medication Management	0		0		0.0	0.00	0.00	
Emergency Service 24/7	0		0		0.0	0.00	0.00	
APRTP	0		0		0.0	0.00	0.00	
Supported Employment Services	0		0		0.0	0.00	0.00	
Other	0		0		0.0	0.00	0.00	
	\$0		0		0.0	\$0.00	\$0.00	
All Services	\$7,293,722	2,311	49,181	1,006.5	21,420.3	\$0.00	\$3,176.71	
	ψι,293,122	2,311	49,101	1,000.5	21,420.3	φ140.3U	ψυ, ΓΓΟ. ΓΙ	

Eligibility Category: Neonatal Abstinence Syndrome Kick Payment

Member Months: 144

							Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient	Donard	71411110		.,	.,		
Medical	\$848,923	168	1,492	1,166.7	10,361.1	\$5,053.11	\$5,895.30
Surgical	15,395	2	20	13.9	138.9	7,697.51	106.91
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	0	0	0	0.0	0.0	0.00	0.00
Alcohol and Drug Abuse	0	0	0	0.0	0.0	0.00	0.00
Other	0	0	0	0.0	0.0	0.00	0.00
	\$864,318	170	1,512	1,180.6	10,500.0	\$5,084.22	\$6,002.21
Hospital Outpatient	* 0 7 00				07.0	0000.40	AOE 07
Emergency Room	\$3,726		14		97.2	\$266.12	\$25.87
Surgery	397		1		6.9	397.44	2.76
Radiology	1,056		8		55.6	132.00	7.33
Pathology	548		54		375.0	10.14	3.80
Pharmacy	8		4		27.8	2.06	0.06
Cardiovascular	326		3		20.8	108.83	2.27
PT/OT/ST	0		0		0.0	0.00	0.00
Psychiatric	0		0		0.0	0.00	0.00
Substance Abuse	0		0		0.0	0.00	0.00
Other	3,874 \$9,935		52 136		361.1 944.4	74.50 \$73.05	<u>26.90</u> \$69.00
Professional and Other State Plan Services	φ 9 ,935		130		944.4	\$73.05	\$09.00
Ambulatory Surgery Center	\$0		0		0.0	\$0.00	\$0.00
Office Visits	6,675		131		909.7	50.95	46.35
Preventive Medicine	21,342		511		3,548.6	41.76	148.21
Maternity	0		0		0.0	0.00	0.00
Certified Midwife	0		0		0.0	0.00	0.00
PT/OT/ST	325		13		90.3	25.02	2.26
Psychiatric and Substance Abuse	00		0		0.0	0.00	0.00
Radiology and Pathology	1,814		131		909.7	13.85	12.60
Home Health and Private Duty Nursing	12,161		137		951.4	88.76	84.45
Ambulance	5,847		161		1,118.1	36.32	40.60
Non-Emergency Transportation	1,197		46		319.4	26.03	8.32
Opioid Treatment Program	0		0		0.0	0.00	0.00
Federally Qualified and Rural Health Clinics	12,968		83		576.4	156.24	90.06
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	0		0		0.0	0.00	0.00
Durable Medical Equipment	7,819		3,523		24,465.3	2.22	54.30
Applied Behavioral Analysis	0		0,020		0.0	0.00	0.00
Other	103,711		2,184		15,166.7	47.49	720.21
	\$173,859		6,920		48,055.6	\$25.12	\$1,207.36
Prescription Drugs							
Generic Scripts	\$528		67		465.3	\$7.88	\$3.66
Single-Source Brand	207		3		20.8	69.07	1.44
Multi-Source Brand	17		2		13.9	8.67	0.12
Specialty	0		0		0.0	0.00	0.00
Hepatitis C	0		0		0.0	0.00	0.00
Other	0		0		0.0	0.00	0.00
	\$752		72		500.0	\$10.45	\$5.22
Community Mental Health Center							
Case Management	\$0		0		0.0	\$0.00	\$0.00
Long Term Support Service	0		0		0.0	0.00	0.00
Partial Hospital	0		0		0.0	0.00	0.00
Psychotherapy	0		0		0.0	0.00	0.00
Evidence Based Practice	0		0		0.0	0.00	0.00
Medication Management	0		0		0.0	0.00	0.00
Emergency Service 24/7	0		0		0.0	0.00	0.00
APRTP	0		0		0.0	0.00	0.00
Supported Employment Services	0		0		0.0	0.00	0.00
Other	0		0		0.0	0.00	0.00
	\$0	/	0	4 405 5	0.0	\$0.00	\$0.00
All Services	\$1,048,865	170	8,640	1,180.6	60,000.0	\$121.40	\$7,283.78

Eligibility Category: Maternity Kick Payment

Member Months: 2,897

Boartis Total Paid Total Paid Annue Sources Autrice Put Vision Put Vision Put Output Mediated Mediation Put Output								Den Oralita
Benefits Dollars Admits Services 1.000 Coit Coit Marical 568,455 20 7 11.0 20.0 52.140.14 52.341 Marical 568,5558 2.349 0.430 2.221 5.33.189 133.133 Materity Delivey 559,558 2.349 0.400 0.00		Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid
Medical 686,465 22 78 11.0 29.9 52,41.01 422,32.0 Maternity Non-Delayery 5,556,058 2,340 6,436 810.63 22,21.8 2,341.86 1,30.13 Maternity Non-Delayery 5,556,058 2,340 6,436 810.63 2,221.8 2,231.86 1,30.13 Psychiatric 0 0 0 0 0 0.00 <t< th=""><th>Benefits</th><th>Dollars</th><th>Admits</th><th>Services</th><th>1,000</th><th>1,000</th><th>Charge</th><th>Cost</th></t<>	Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost
Sungial 4.490 1 2 0.3 0.7 4.490.73 1.7.2 Maternity Delivery 555,058 3.249 6.63 10.7 2.221 6 2.231 8 19.31 33 Maternity Non-Delivery 557,166 3.1 69 10.7 2.238 1.86 30 19.71 33 Wal Newborn 0 0 0 0.0 0.00	Hospital Inpatient							
Multimetry NoDelivery 5,565,058 2,341 6,435 810.8 2,2215 2,2318 1,391,33 Weil Needborn 0 0 0 0 0 0.00	Medical		32					
Materniy Non-Delivery 57.198 31 69 10.7 2.38 11.845.03 11.7.4 Well Newborn 0 0 0 0.0 0.0 0.0 0.00 Relicibly 2.372.67 \$19.76.43 H2.8 Starzer \$19.76.43 H2.8 Starzer \$19.76.43 H2.8 Starzer \$19.76.43 H2.8 Starzer \$19.76.43 H2.8 H3.4	Surgical				0.3	0.7	4,990.37	1.72
Weil Newborn Alcchol and Drug Abuse 0				,				
Psychiatric 0 <th< td=""><td>Maternity Non-Delivery</td><td>57,196</td><td></td><td></td><td>10.7</td><td></td><td></td><td></td></th<>	Maternity Non-Delivery	57,196			10.7			
Alechol and Drug Abuse 0 0 0 0 0 0 0.00 <th< td=""><td>Well Newborn</td><td>0</td><td>0</td><td>0</td><td>0.0</td><td>0.0</td><td>0.00</td><td>0.00</td></th<>	Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Other 0 0 0 0.0 0.00 0.00 Hospial Outpatient 55:728778 2.413 6.585 832.4 2.273 53.372.473 51.976.474 Berngency Room 5.367.15 3.7 12.8 9992.50 51.372.474 51.976.474 Surgery 2.304 4 1.4 57.51 0.80 Radiobay 6.583 6.61 2.211 107.914 2.27 Pharmacy 6.383 1.44 36.55 5.16 2.24 Cardiovancular 0 0 0.00 0.00 0.00 0.00 Profestional and Other State Plan Services 0 0 0 0.00<	Psychiatric	0		0				0.00
S5.725.72 2.413 6.585 832.9 2.273.0 52.372.87 51.976.43 Emergency Room \$36,715 37 12.8 \$998.230 \$12.6 Radiology 6.683 61 2.11 10.755 8.08 Pharmacy 5.313 1,147 355.8 5.16 2.29 Pharmacy 5.313 1,147 355.8 5.16 2.04 Pharmacy 5.313 1,147 355.8 5.16 2.04 Pharmacy 0 0 0.00 0.00 0.00 0.00 Substance Abuse 0 0 0 0.00	Alcohol and Drug Abuse		0	-				
Hospital Outpatient Emergency Room \$36,715 37 12.8 \$992.30 \$12.67 Surgery 2,304 4 1.4 \$75.91 0.8 Radiology 6,683 61 22.11 10.79 2.27 Pathology 4,991 443 152.9 11.04 16.9 Pathology 4,991 443 152.9 1.04 16.9 Cardiovascular 0 0 0.0 0.00 0.00 0.00 Pathology 39.155 85.4 2.94.8 46.85 13.52 Professional and Other State Plan Services 0 0 0.0 0.00 <td>Other</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Other							
Emergency Room \$36,715 37 12.8 \$992.30 \$12.67 Radiology 6,863 61 21.1 107.91 2.27 Phathology 6,863 61 21.1 107.91 2.27 Phathology 4,891 443 15.29 11.04 16.9 Cardiovascular 0 0 0.00 0.00 0.00 0.00 Profisition 0 0 0.00 0.00 0.00 0.00 Substance Abuse 0 0 0.00 0.00 0.00 0.00 Other 39.55 864 2.848 46.85 13.52 Professional and Other Stab Plan Services 0 0 0 0.00 9.000 Ambulatory Strenger Center 0 0 0 0.00 9.000 Professional and Other Stab Plan Services 0 0 0 0.00 9.000 Protexition 4.269 93 32.1 4.54.50 1.57 1.58		\$5,725,729	2,413	6,585	832.9	2,273.0	\$2,372.87	\$1,976.43
Surgery 2,304 4 1.4 575.91 0.80 Pathology 6,583 61 21.1 107.91 2.27 Pathology 4,891 443 152.9 11.04 16.9 Cardiovascular 0 0 0.0 0.0 0.00 0.00 Cardiovascular 0 0 0.0 0.00 0.00 0.00 Psychiatric 0 0 0 0.0 0.00 0.00 0.00 Other 39.155 85.4 29.6 878.8 537.63 532.99 Ambulatory Surgery Center 50 0 0 0.0 50.00 \$30.00	· · ·							
Radiology 6.583 61 21.1 107.91 22.7 Phatholgy 4.891 4.43 152.9 11.04 169 Pharmacy 5.913 1,147 395.5 5.16 2.04 Cardiovascular 0 0 0.0 0.00 0.00 0.00 Profestional and Other State Plan Services 0 0 0.0 0.0 0.00 0.00 0.00 Other 39.155 2.844 47.88 537.63 532.90 0.0 0.0 0.00 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>								
Pathology 4.891 443 152.9 11.04 16.9 Pharmacy 5.913 1.147 395.9 5.16 2.04 Cardiovascular 0 0 0.0 0.00 0.00 0.00 Psychiatric 0 0 0.0 0.00 0.00 0.00 Substance Abuse 0 0 0.00 0.00 0.00 0.00 Other 39.155 8.84 2.94.8 4.85.85 32.99 9 Professional and Other State Plan Services 3 0 0.0 0.0 0.0								
Pharmacy Cardiovascular 5,913 1,147 395,9 5,16 2.04 Cardiovascular 0 0 0,0 0,00 0,00 0,00 Pychiatric 0 0 0 0,0 0,00 </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>								
Cardiovscular 0 0 0 0.0 0.00 0.00 PriOT/ST 0 0 0 0.0 0.00 0.00 0.00 Substance Abuse 0 0 0 0.00 0.00 0.00 0.00 Other 39,155 .854 .2646 .878.6 15.52 Professional and Other State Plan Services	Pathology							
PT(17)TST 0 0 0 0.0 0.00 0.00 Substance Abuse 0 0 0 0.0 0.00 0.00 Substance Abuse 39,155 854 244.8 45.85 13.52 Professional and Other State Plan Services \$90,561 2.546 878.8 \$37.53 \$32.99 Professional and Other State Plan Services \$0 0 0.0 \$0.00 \$0.00 Ambulatory Surgery Center \$0 0 0.0 \$0.00 \$0.00 Office Visits 4.269 93 32.21 45.90 1.47 Preventive Medicine 3.136 352 121.5 8.91 1.08 Maternity 1.503.452 3.311 1.14.31 45.40 50.00 0.00								
Psychiatric 0 0 0.0 0.00 0.00 0.00 Other 39.155 854 284.8 45.85 13.52 Professional and Other State Plan Services N 0 0.00 \$50.00				-				
Substance Abuse 0 0 0.0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 532.298 Professional and Other State Plan Services 80 0 0.00 \$97.68 \$37.53 \$32.298 Ambulatory Surgery Center \$0 0 0.00 \$0.00 \$0.00 \$0.00 \$0.00 Office Visits 4.269 93 32.1 4.55.80 1.47 Preventive Medicine 3.136 352 121.5 8.91 1.60 Matemity 1.503.452 3.311 1.143.1 454.60 51.97 Or Tior JGT 0 0 0 0.00 0.00 0.00 0.00 Radiogy and Pathology 6.975 31.9 11.01 21.83 4.45 4.53.24 4.54 Arn-Emergency Transportation 0 0 0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00	PT/OT/ST							
Other 39,155 854 224.8 43.85 13.52 Professional and Other State Plan Services Ambulatory Surgery Center \$0 0 0.0 \$0.00	Psychiatric	0		0		0.0	0.00	0.00
\$95.561 2.566 878.8 \$37.53 \$32.99 Ambulatory Surgery Center \$0 0 0.0 \$0.00	Substance Abuse	0		0		0.0	0.00	0.00
Professional and Other State Plan Services Ambulatory Surgery Center S0 0 0.0 \$\$0.00 \$\$0.00 Office Visits 4.269 93 32.1 45.90 1.45 Preventive Medicine 3.136 352 121.5 8.91 1.08 Maternity 1.503.452 3.311 1.143.1 44.401 518.97 Certified Midwife 17.412 58 20.2 297.95 6.01 PTIOT/ST 0 0 0.00 0.00 0.00 0.00 Psychiatric and Substance Abuse 0 0 0 0.00 0.00 0.00 0.00 Psychiatric and Substance Abuse 1.471 38 635 284.2 13.70 3.95 Non-Emergency Transportation 0 0 0.00 0.00 0.00 0.00 Opriod Treatment Program 0 0 0.00 0.00 0.00 0.00 Auti Medical Equipment 0 0 0.00 0.00 0.00 0.00 0.00 <t< td=""><td>Other</td><td></td><td></td><td></td><td></td><td></td><td></td><td>13.52</td></t<>	Other							13.52
Ambulatory Surgery Center S0 0 0.0 \$0.00		\$95,561		2,546		878.8	\$37.53	\$32.99
Office Visits 4.269 93 32.1 45.90 1.47 Preventive Medicine 3.136 352 121.5 8.91 1.08 Maternity 1,503.452 3.311 1.14.31 45.401 518.97 Certified Midwife 17.412 58 20.2 297.95 6.01 Pri/OTST 0 0 0.00 0.00 0.00 0.00 Psychiatric and Substance Abuse 0 0 0.0	Professional and Other State Plan Services							
Preventive Medicine 3.136 352 12.15 8.91 1.08 Matemity 1,503,452 3.311 1,143.1 454.01 518.97 Certified Midwire 17.412 58 22.97.95 6.01 PY/DT/ST 0 0 0.00 0.00 0.00 Radiology and Pathology 6.975 319 110.1 21.87 2.41 Home Health and Private Duty Nursing 1.271 38 13.1 33.44 0.44 Ambulance 11.438 885 288.2 13.70 3.95 Non-Emergency Transportation 0 0 0.0 0.00 0.00 Optid Treatment Program 0 0 0.0 0.00 0.00 0.00 Personal Care 0 0 0 0.0 0.00 0.00 0.00 Other 38.631 2.452 8.464 15.75 13.33 Prescription Drugs 51 50 0 0.0 0.00 0.00	Ambulatory Surgery Center	\$0		0		0.0	\$0.00	\$0.00
Matemity 1,503,452 3,311 1,143,1 454,01 518,97 Certified Midwife 17,412 58 20.2 297,95 6.01 PT/OT/ST 0 0 0.0 0.00 0.00 0.00 Psychiatric and Substance Abuse 0 0 0 0.0 0.00 0.00 Psychiatric and Substance Abuse 1.271 3.88 13.1 33.44 0.44 Home Health and Private Duty Nursing 1.271 3.88 288.2 13.77 3.95 Non-Emergency Transportation 0 0 0.0 0.00 0.00 0.00 Opioid Treatment Program 0 0 0 0.00	Office Visits	4,269		93		32.1	45.90	1.47
Certified Midwife 17,412 58 20.2 297.95 6.01 PT/OT/ST 0 0 0.0 0.00 0.00 0.00 Radiology and Pathology 6.975 319 11.01 21.87 2.41 Home Health and Private Duty Nursing 1.271 38 28.8.2 13.70 3.95 Non-Emergency Transportation 0 0 0.0 0.00 0.00 0.00 Pederally Qualified and Rural Health Clinics 9.233 58 20.1 18.827 3.19 Adut Medical Day Care 0 0 0 0.00 0.00 0.00 Personal Care 0 0 0 0.00 0.00 0.00 Durable Medical Equipment 0 0 0.00 0.00 0.00 0.00 Other 38.631 2.452 8464 15.75 13.33 58 Generic Scripts \$0 0 0 0.00 0.00 0.00 Mulis-Source Brand 0 <t< td=""><td>Preventive Medicine</td><td>3,136</td><td></td><td>352</td><td></td><td>121.5</td><td>8.91</td><td>1.08</td></t<>	Preventive Medicine	3,136		352		121.5	8.91	1.08
PT/OT/ST 0 0 0 0.00 0.000 0.000 Radiology and Pathology 6,975 319 110.1 21.87 2.41 Home Health and Private Duty Nursing 1.271 38 13.1 33.44 0.44 Ambulance 11.438 835 288.2 13.70 3.95 Non-Emergency Transportation 0 0 0.00 0.00 0.00 Opicid Treatment Program 0 0 0.00 0.00 0.00 0.00 Presonal Care 0 0 0 0.00 0.00 0.00 0.00 Ourable Medical Equipment 0 0 0.00 0.00 0.00 0.00 Other 38.631 2.462 846.4 15.75 13.33 Prescription Drugs 51.596.817 7.517 2.594.8 52.228 550.85 Prescription Drugs 51 0 0 0.00 0.00 0.00 Multi-Source Brand 0 0 0.00	Maternity	1,503,452		3,311		1,143.1	454.01	518.97
PT/OT/ST 0 0 0 0.0 0.00 0.00 0.00 Psychiatric and Bubstance Abuse 0 0 0.00 0.00 0.00 0.00 Radiology and Pathology 6.975 319 110.1 21.87 2.41 Home Health and Private Duty Nursing 1.271 38 13.1 33.44 0.44 Ambulance 11.438 835 286.2 13.70 3.95 Non-Emergency Transportation 0 0 0.00 0.00 0.00 0.00 Opicid Treatment Program 0 0 0.00 0.00 0.00 0.00 Federally Qualified and Rural Health Clinics 9.233 58 20.1 158.27 3.19 Autit Medical Equipment 0 0 0.00 0.00 0.00 0.00 Durable Medical Equipment 0 0 0.00 0.00 0.00 0.00 Other 38.631 2.452 846.4 15.75 13.3 3.45 Pre	Certified Midwife	17,412		58			297.95	6.01
Psychiatric and Substance Abuse 0 0 0 0.00 0.00 0.00 Radiology and Pathology 6,975 319 110.1 21.87 2.41 Home Health and Private Duty Nursing 11.271 38 13.1 33.44 0.44 Ambulance 11.438 8835 228.2 13.70 3.95 Non-Emergency Transportation 0 0 0.00 0.00 0.00 0.00 Opioid Treatment Program 0 0 0.00	PT/OT/ST							
Radiology and Pathology 6,975 319 110.1 21.87 2.41 Home Health and Private Duty Nursing 1,271 38 13.1 33.44 0.444 Ambulance 11,438 835 288.2 13.70 3.95 Non-Emergency Transportation 0 0 0.00 0.00 0.00 Opicid Treatment Program 0 0 0.00 0.00 0.00 0.00 Cold Care 0 0 0.00		0		0				
Home Health and Private Duty Nursing 1.271 38 13.1 33.44 0.44 Ambulance 11.438 835 288.2 13.70 3.95 Non-Emergency Transportation 0 0 0.00	5			-				
Ambulance 11,438 835 288.2 13.70 3.95 Non-Emergency Transportation 0 0 0.0 0.00 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>								
Non-Emergency Transportation 0 0 0 0.0 0.00 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>								
Opioid Treatment Program 0 0 0 0.0 0.00 0.00 0.00 Federally Qualified and Rural Health Clinics 9,233 58 20.1 1158.27 3.19 Adult Medical Day Care 0 0 0.0 0.00 0.00 0.00 Personal Care 0 0 0.0 0.00 0.00 0.00 Durable Medical Equipment 0 0 0.0 0.00 0.00 0.00 Opiod Text 0 0 0.0 0.00 0.00 0.00 Optice Behavioral Analysis 0 0 0.0 0.00 0.00 0.00 Other 38.631 2.452 846.4 15.75 13.33 Single-Source Brand 0 0 0.0 0.00 0.00 0.00 Single-Source Brand 0 0 0.0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>								
Federally Qualified and Rural Health Clinics 9,233 58 20.1 158.27 3.19 Adult Medical Day Care 0 0 0.0 0.00				-				
Adult Medical Day Care 0 0 0.0 0.00 0.00 Personal Care 0 0 0.0 0.00 0.00 0.00 Durable Medical Equipment 0 0 0.00 0.00 0.00 0.00 Applied Behavioral Analysis 0 0 0 0.00 0.00 0.00 Other 38,631 2,452 846.4 15.75 13.33 Prescription Drugs 7,517 2,594.8 \$212.29 \$550.80 Generic Scripts \$0 0 0.00 0.00 \$0.00 Single-Source Brand 0 0 0.00 0.00 \$0.00 Specialty 0 0 0.00 0.00 0.00 \$0.00 Other 0 0 0 0.00 0.00 \$0.00 \$0.00 Community Mental Health Center 0 0 0 0.00 \$0.00 \$0.00 Cong Term Support Service 0 0 0 0.00 0.00<				-				
Personal Care 0 0 0 0.0 0.00								
Durable Medical Equipment Applied Behavioral Analysis 0 0 0 0.0 0.00				-				
Applied Behavioral Analysis 0 0 0 0.0 0.00 0.00 Other 38,631 2,452 846.4 15.75 13.33 Prescription Drugs 2,598.8 \$212.29 \$550.855 Prescription Drugs 0 0.0 \$0.00 \$500.00 Single-Source Brand 0 0 0.00 \$0.00 \$0.00 Multi-Source Brand 0 0 0.00 0.00 \$0.00 \$0.00 Specialty 0 0 0 0.00 0.00 \$0.00				-				
Other 38,631 2,452 846.4 15.75 13.33 Prescription Drugs \$1,595,817 7,517 2,594.8 \$212.29 \$\$50.85 Generic Scripts \$0 0 0.0 \$0.00 \$0.00 \$0.00 Single-Source Brand 0 0 0.0 0.00 0.00 \$0.00<								
\$1,595,817 7,517 2,594.8 \$212.29 \$550.85 Prescription Drugs Generic Scripts \$0 0 0.0 \$0.00				-				
Prescription Drugs Generic Scripts \$0 0 0.0 \$0.00 <td>Other</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Other							
Generic Scripts \$0 0 0.0 \$0.00 \$0.00 Single-Source Brand 0 0 0 0.0 0.00 0.00 Multi-Source Brand 0 0 0 0.00 0.00 0.00 Specialty 0 0 0 0.00 0.00 0.00 Hepatitis C 0 0 0 0.00 0.00 0.00 Other 0 0 0 0.00 0.00 0.00 Commuity Mental Health Center C 0 0.00 \$0.00 \$0.00 Case Management \$0 0 0 0.00 \$0.00 \$0.00 Long Term Support Service 0 0 0.00 0.00 0.00 \$0.00 Partial Hospital 0 0 0 0.00 0.00 0.00 \$0.00 \$0.00 Evidence Based Practice 0 0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00	Broscription Drugs	\$1,595,617		7,517		2,394.0	\$212.29	\$550.65
Single-Source Brand 0 0 0.0 0.00 0.00 Multi-Source Brand 0 0 0 0.00 0.00 0.00 Specialty 0 0 0.0 0.00 0.00 0.00 Hepatitis C 0 0 0.0 0.00 <		\$0		0		0.0	\$0.00	\$0.00
Muti-Source Brand 0 0 0 0.0 0.00	•							
Specialty 0 0 0 0.0 0.00 0.00 Hepatitis C 0 0 0 0.0 0.00 \$0.00								
Hepatitis C 0 0 0.0 0.00 <th< td=""><td></td><td></td><td></td><td>-</td><td></td><td></td><td></td><td></td></th<>				-				
Other 0 0 0.0 0.00 0.00 \$0 0 0 0.0 \$0.00 \$0.00 Community Mental Health Center 0 0 0.0 \$0.00 \$0.00 Case Management \$0 0 0.0 \$0.00 \$0.00 \$0.00 Long Term Support Service 0 0 0.0 \$0.00 \$0.00 \$0.00 Partial Hospital 0 0 0 0.00 \$0.00 \$0.00 Partial Hospital 0 0 0.0 0.00 \$0.00 \$0.00 Psychotherapy 0 0 0 0.00 \$0.00 \$0.00 Evidence Based Practice 0 0 0.00 \$0.00 \$0.00 \$0.00 Medication Management 0 0 0.00 \$0.00 \$0.00 \$0.00 Emergency Service 24/7 0 0 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 <				-				
\$0 0 0.0 \$0.00 \$0.00 Community Mental Health Center 0 0 \$0.00 \$0.00 \$0.00 Case Management \$0 0 0.0 \$0.00 \$0.00 \$0.00 Long Term Support Service 0 0 0.0 \$0.00 \$0.00 \$0.00 Partial Hospital 0 0 0.0 0.00 0.00 0.00 \$0.00 Partial Hospital 0 0 0.0 0.0	•							
Community Mental Health Center \$0 0 0.0 \$0.00<	Other							
Case Management \$0 0 \$0.00 \$0.00 Long Term Support Service 0 0 0.0 0.00 0.00 Partial Hospital 0 0 0.0 0.00 0.00 0.00 Psychotherapy 0 0 0.0 0.00 0.00 0.00 Evidence Based Practice 0 0 0.00 0.00 0.00 0.00 Medication Management 0 0 0 0.00 <td>Community Montal Health Contar</td> <td>\$0</td> <td></td> <td>0</td> <td></td> <td>0.0</td> <td>\$0.00</td> <td>\$0.00</td>	Community Montal Health Contar	\$0		0		0.0	\$0.00	\$0.00
Long Term Support Service 0 0 0 0.00 0.00 Partial Hospital 0 0 0 0.00 0.00 0.00 Psychotherapy 0 0 0 0.00 0.00 0.00 Evidence Based Practice 0 0 0 0.00 0.00 0.00 Medication Management 0 0 0 0.00 0.00 0.00 Emergency Service 24/7 0 0 0 0.00 0.00 0.00 Supported Employment Services 0 0 0 0.00 0.00 0.00 Other 0 0 0.00 0.00 0.00 0.00	-	¢0		0		0.0	¢0.00	¢0.00
Partial Hospital 0 0 0 0.0 0.00 0.00 Psychotherapy 0 0 0 0.0 0.00 0.00 Evidence Based Practice 0 0 0 0.00 0.00 0.00 Medication Management 0 0 0 0.00 0.00 0.00 Emergency Service 24/7 0 0 0 0.00 0.00 0.00 APRTP 0 0 0 0.00 0.00 0.00 0.00 Supported Employment Services 0 0 0 0.00 0.00 0.00 Other 0 0 0.00 0.00 0.00 0.00 0.00								
Psychotherapy 0 0 0 0.0 0.00 0.00 Evidence Based Practice 0 0 0 0.0 0.00 0.00 Medication Management 0 0 0 0.00 0.00 0.00 Emergency Service 24/7 0 0 0 0.00 0.00 0.00 APRTP 0 0 0.00 0.00 0.00 0.00 Supported Employment Services 0 0 0.00 0.00 0.00 Other 0 0 0.00 0.00 0.00 0.00								
Evidence Based Practice 0 0 0.0 0.00 0.00 Medication Management 0 0 0 0.0 0.00 0.00 Emergency Service 24/7 0 0 0 0.00 0.00 0.00 APRTP 0 0 0.0 0.00 0.00 0.00 Supported Employment Services 0 0 0.00 0.00 0.00 Other 0 0 0.00 \$0.00 \$0.00 \$0.00								
Medication Management 0 0 0.0 0.00 0.00 Emergency Service 24/7 0 0 0 0.0 0.00 0.00 APRTP 0 0 0 0.0 0.00								
Emergency Service 24/7 0 0 0 0.0 0.00 0.00 APRTP 0 0 0 0.0 0.0								
APRTP 0 0 0.0 0.00 0.00 Supported Employment Services 0 0 0 0.00								
Supported Employment Services 0 0 0.0 0.0								
Other 0 0 0.0 0.00<								
\$0 0 0.0 \$0.00 \$0.00								
	Other			-				
All Services \$7,417,108 2,413 16,648 832.9 5,746.7 \$445.52 \$2,560.27								
	All Services	\$7,417,108	2,413	16,648	832.9	5,746.7	\$445.52	\$2,560.27

Eligibility Category: Severe/Persistent Mental Illness - Non-Dual

Member Months: 15,659

							Der Carita
	Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid
Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost
Hospital Inpatient	2011010	, (0,111)		.,	.,		
Medical	\$696,809	171	854	131.0	654.4	\$4,074.91	\$44.50
Surgical	624,061	62	562	47.5	430.7	10,065.49	39.85
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	6,889	4	11	3.1	8.4	1,722.29	0.44
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	822,722	223	1,291	170.9	989.3	3,689.33	52.54
Alcohol and Drug Abuse	30,129	12	99	9.2	75.9	2,510.73	1.92
Other	0	0	0	0.0	0.0	0.00	0.00
	\$2,180,609	472	2,817	361.7	2,158.7	\$4,619.94	\$139.25
Hospital Outpatient					,		
Emergency Room	\$1,308,298		2,081		1,594.7	\$628.69	\$83.55
Surgery	285,183		357		273.6	798.83	18.21
Radiology	351,654		1,431		1,096.6	245.74	22.46
Pathology	134,489		11,397		8,733.7	11.80	8.59
Pharmacy	180,084		46,713		35,797.1	3.86	11.50
Cardiovascular	35,630		198		151.7	179.95	2.28
PT/OT/ST	86,618		3,348		2,565.6	25.87	5.53
Psychiatric	18,098		113		86.6	160.16	1.16
Substance Abuse	5,887		52		39.8	113.21	0.38
Other	300,369		14,073		10.784.4	21.34	19.18
	\$2,706,310		79,763		61,123.9	\$33.93	\$172.82
Professional and Other State Plan Services	<i> </i>				.,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
Ambulatory Surgery Center	\$35,398		124		95.0	\$285.47	\$2.26
Office Visits	502,683		8,343		6.393.4	60.25	32.10
Preventive Medicine	52,036		1,927		1,476.7	27.00	3.32
Maternity	16,499		95		72.7	173.81	1.05
Certified Midwife	217		7		5.4	30.95	0.01
PT/OT/ST	49,887		2,138		1.638.4	23.33	3.19
Psychiatric and Substance Abuse	378,200		5,764		4,417.1	65.61	24.15
Radiology and Pathology	222,188		10,154		7.781.2	21.88	14.19
Home Health and Private Duty Nursing	191,037		6,684		5,122.1	28.58	12.20
Ambulance	163,090		7,900		6,053.9	20.64	10.41
Non-Emergency Transportation	629,910		39,725		30,442.0	15.86	40.23
Opioid Treatment Program	136,733		13,379		10,252.6	10.22	8.73
Federally Qualified and Rural Health Clinics	330,465		2,399		1,838.4	137.75	21.10
Adult Medical Day Care	0		2,000		0.0	0.00	0.00
Personal Care	39,486		7,318		5,607.9	5.40	2.52
Durable Medical Equipment	296,952		83,753		64,181.5	3.55	18.96
Applied Behavioral Analysis	0		00,700		0.0	0.00	0.00
Other	811,204		48,209		36,943.5	16.83	51.80
	\$3,855,985		237,919		182,321.8	\$16.21	\$246.24
Prescription Drugs	ψ0,000,000		201,010		102,021.0	ψ10.21	ψ2+0.2+
Generic Scripts	\$1,585,159		85,973		65,882.7	\$18.44	\$101.23
Single-Source Brand	3,319,268		7,911		6,062.3	419.58	211.97
Multi-Source Brand	941,955		2,781		2,131.1	338.71	60.15
Specialty	2,706,585		1,034		792.4	2,617.59	172.84
Hepatitis C	472,045		21		16.1	22,478.33	30.14
Other	1,115		43		33.0	22,478.33	0.07
	\$9,026,127		97,763		74,917.6	\$92.33	\$576.41
Community Mental Health Center	ψ0,020,121		51,100		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	ψυ2.00	ψυτυ.+Τ
Case Management	\$4,361,787		12,179		9,333.0	\$358.14	\$278.54
Long Term Support Service	4,199,955		155,932		119,493.7	26.93	268.21
Partial Hospital	193,483		2,008		1,538.8	96.36	12.36
Psychotherapy	1,009,563		16,666		12,771.5	60.58	64.47
Evidence Based Practice	52,650		3,069		2,351.8	17.16	3.36
Medication Management	64,395		2,160		1,655.2	29.81	4.11
Emergency Service 24/7	64,395		2,100		2,102.8	29.81	4.11
APRTP	77,409		2,744		2,102.8	549.00	4.11
						26.54	
Supported Employment Services Other	318,055 494,212		11,984 6,919		9,183.6 5,302.2	71.43	20.31 31.56
	\$10,835,884		213,802		5,302.2	\$50.68	\$691.98
All Services	\$28,604,916	472	632,064	361.7	484,362.6	\$50.68	\$1,826.71
All Sel VICes	φ20,004,910	412	032,004	301.7	404,302.0	J₹43.20	φ1,020.7 T

Eligibility Category: Severe/Persistent Mental Illness - Dual

Member Months: 22,217

							Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient	2011010			.,	.,		
Medical	\$456,422	309	1,994	166.9	1,077.0	\$1,477.09	\$20.54
Surgical	90,311	62	341	33.5	184.2	1,456.62	4.07
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	405,420	146	1,506	78.9	813.4	2,776.85	18.25
Alcohol and Drug Abuse	27,861	15	137	8.1	74.0	1,857.43	1.25
Other	0	0	0	0.0	0.0	0.00	0.00
Hospital Outpatient	\$980,014	532	3,978	287.4	2,148.7	\$1,842.13	\$44.11
Emergency Room	\$683,010		2,060		1,112.7	\$331.56	\$30.74
Surgery	213,975		714		385.7	299.68	9.63
Radiology	149,077		2,030		1,096.5	73.44	6.71
Pathology	36,316		5,725		3,092.3	6.34	1.63
Pharmacy	168,933		60,938		32,915.0	2.77	7.60
Cardiovascular	18,489		350		189.0	52.83	0.83
PT/OT/ST	60,317		5,085		2,746.6	11.86	2.71
Psychiatric	22,611		377		203.6	59.98	1.02
Substance Abuse	3,304		29		15.7	113.93	0.15
Other	395,208		26,883		14,520.6	14.70	17.79
Professional and Other State Plan Services	\$1,751,239		104,191		56,277.6	\$16.81	\$78.83
Ambulatory Surgery Center	\$26,367		190		102.6	\$138.77	\$1.19
Office Visits	326,354		13,326		7,197.9	24.49	\$1.19 14.69
Preventive Medicine	21,639		1,060		572.5	20.41	0.97
Maternity	2,947		34		18.2	87.67	0.13
Certified Midwife	2,347		0		0.0	0.00	0.00
PT/OT/ST	21,363		2,856		1,542.6	7.48	0.96
Psychiatric and Substance Abuse	775,708		13,464		7,272.4	57.61	34.92
Radiology and Pathology	86,147		8,364		4,517.7	10.30	3.88
Home Health and Private Duty Nursing	52,919		1,261		681.1	41.97	2.38
Ambulance	163,318		9,880		5,336.6	16.53	7.35
Non-Emergency Transportation	963,968		49,310		26,634.2	19.55	43.39
Opioid Treatment Program	132,329		12,948		6,993.7	10.22	5.96
Federally Qualified and Rural Health Clinics	24,445		389		210.1	62.84	1.10
Adult Medical Day Care	42,657		1,390		750.8	30.69	1.92
Personal Care	203,081		37,505		20,257.9	5.41	9.14
Durable Medical Equipment	195,861		160,550		86,719.3	1.22	8.82
Applied Behavioral Analysis	0		0		0.0	0.00	0.00
Other	615,399 \$3,654,501		97,617 410,144		52,726.7 221,534.5	6.30 \$8.91	27.70 \$164.49
Prescription Drugs	\$3,034,301		410,144		221,004.0	φ0.91	\$104.49
Generic Scripts	\$25,557		7,094		3,831.7	\$3.60	\$1.15
Single-Source Brand	507		110		59.4	4.61	0.02
Multi-Source Brand	465		27		14.6	17.21	0.02
Specialty	0		0		0.0	0.00	0.00
Hepatitis C	0		0		0.0	0.00	0.00
Other	61		12		6.5	5.05	0.00
	\$26,590		7,243		3,912.2	\$3.67	\$1.20
Community Mental Health Center			17.070			*****	****
Case Management	\$6,662,227		17,873		9,653.9	\$372.75	\$299.88
Long Term Support Service	9,295,560		323,590		174,783.5	28.73	418.41
Partial Hospital	541,369 760,384		5,480 22,108		2,960.0 11,941.4	98.79 34.39	24.37 34.23
Psychotherapy Evidence Based Practice	124,226		6,392		3,452.6		34.23 5.59
Medication Management	124,226		6,392 4,606		3,452.6 2,487.9	19.43 30.65	5.59 6.35
Emergency Service 24/7	96,838		4,000		2,487.9	23.71	4.36
APRTP	182,904		4,085		2,200.5	571.58	8.23
Supported Employment Services	839,375		30,337		16,386.2	27.67	37.78
Other	361,722		10,288		5,556.9	35.16	16.28
	\$19,005,780		425,079		229,601.7	\$44.71	\$855.48
All Services	\$25,418,124	532	950,635	287.4	513,474.6	\$26.74	\$1,144.11

Eligibility Category: Severe Mental Illness - Non-Dual

Member Months: 7,588

							Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient	Donars	Adinto	00111003	1,000	1,000	onarge	0031
Medical	\$406,091	117	631	185.0	997.9	\$3,470.86	\$53.52
Surgical	196,344	28	221	44.3	349.5	7,012.28	25.88
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	13,977	5	16	7.9	25.3	2,795.38	1.84
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	248,946	60	384	94.9	607.3	4,149.10	32.81
Alcohol and Drug Abuse	43,343	16	152	25.3	240.4	2,708.92	5.71
Other	-10,040	0	0	0.0	0.0	0.00	0.00
other	\$908,700	226	1,404	357.4	2,220.3	\$4,020.80	\$119.75
Hospital Outpatient	\$000,100		1,404	001.4	2,220.0	\$1,020.00	\$110.70
Emergency Room	\$700,092		1,128		1,783.9	\$620.65	\$92.26
Surgery	143,145		172		272.0	832.24	18.86
Radiology	159,859		774		1,224.0	206.54	21.07
Pathology	76,921		5,948		9,406.4	12.93	10.14
Pharmacy	91,439		19,981		31,598.6	4.58	12.05
Cardiovascular	25,429		122		192.9	208.44	3.35
PT/OT/ST	52,174		1,672		2,644.2	31.20	6.88
Psychiatric	13,394		80		126.5	167.43	1.77
Substance Abuse	7,066		67		106.0	105.47	0.93
Other	177,438		6,615		10,461.2	26.82	23.38
	\$1,446,959		36,559		57,815.6	\$39.58	\$190.69
Professional and Other State Plan Services	ψ1,410,000		00,000		01,010.0	\$00.00	\$100.00
Ambulatory Surgery Center	\$9,911		40		63.3	\$247.77	\$1.31
Office Visits	247.759		4,105		6,491.8	60.36	32.65
Preventive Medicine	34,587		840		1,328.4	41.18	4.56
Maternity	27,213		116		182.7	235.58	3.59
Certified Midwife	43		1		1.6	42.72	0.01
PT/OT/ST	12,592		560		885.6	22.49	1.66
Psychiatric and Substance Abuse	339,261		3,647		5,767.5	93.02	44.71
Radiology and Pathology	150,204		5,792		9,159.7	25.93	19.79
Home Health and Private Duty Nursing	95,441		8,377		13,247.7	11.39	12.58
Ambulance	85,915		3,696		5,845.0	23.25	11.32
Non-Emergency Transportation	470,591		23,308		36,860.0	20.19	62.02
Opioid Treatment Program	122,538		11,990		18,961.4	10.22	16.15
Federally Qualified and Rural Health Clinics	180,451		1,225		1,937.3	147.31	23.78
Adult Medical Day Care	11,397		492		778.1	23.16	1.50
Personal Care	863		160		253.0	5.40	0.11
Durable Medical Equipment	97,526		13,207		20,886.0	7.38	12.85
Applied Behavioral Analysis	01,020		0		0.0	0.00	0.00
Other	450,293		27,076		42,818.9	16.63	59.34
	\$2,336,584		104,632		165,467.8	\$22.33	\$307.93
Prescription Drugs	\$2,000,001		101,002		100,10110	\$22.00	<i>\\</i>
Generic Scripts	\$463,230		27,627		43,690.3	\$16.77	\$61.05
Single-Source Brand	838,568		2,179		3,445,9	384.84	110.51
Multi-Source Brand	278,301		1,198		1,894.6	232.31	36.68
Specialty	746,678		196		310.0	3,809.58	98.40
Hepatitis C	560,567		23		36.4	24,372.48	73.87
Other	1		1		1.6	1.30	0.00
	\$2,887,345		31,224		49,378.7	\$92.47	\$380.51
Community Mental Health Center	+=,==:,=:=		.,				
Case Management	\$1,456,555		4,067		6,431.7	\$358.14	\$191.95
Long Term Support Service	477,695		18,320		28,971.9	26.08	62.95
Partial Hospital	1,038		11		17.4	94.33	0.14
Psychotherapy	529,462		8,387		13,263.5	63.13	69.78
Evidence Based Practice	23,367		1,038		1,641.5	22.51	3.08
Medication Management	13,112		446		705.3	29.40	1.73
Emergency Service 24/7	5,255		224		354.2	23.46	0.69
APRTP	75,213		137		216.7	549.00	9.91
Supported Employment Services	92,253		3,477		5,498.6	26.53	12.16
Other	208,530		2,383		3,768.6	87.51	27.48
	\$2,882,480		38,490		60,869.4	\$74.89	\$379.87
All Services	\$10,462,069	226	212,309	357.4	335,751.7	\$49.28	\$1,378.76
	\$.0, TOL,000	220	212,000		550,101.1	ψ ⁺ 0.20	÷.,010.10

Eligibility Category: Severe Mental Illness - Dual

Member Months: 3,766

							D
	Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid
Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost
Hospital Inpatient							
Medical	\$64,977	49	180	156.1	573.6	\$1,326.05	\$17.25
Surgical	27,827	17	127	54.2	404.7	1,636.90	7.39
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	1,340	1	2	3.2	6.4	1,340.00	0.36
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	23,153	11	74	35.1	235.8	2,104.84	6.15
Alcohol and Drug Abuse	22,787	7	78	22.3	248.6	3,255.24	6.05
Other	0	0	0	0.0	0.0	0.00	0.00
	\$140,084	85	461	270.9	1,469.0	\$1,648.04	\$37.20
Hospital Outpatient							
Emergency Room	\$125,418		452		1,440.3	\$277.47	\$33.30
Surgery	37,008		153		487.5	241.88	9.83
Radiology	36,461		391		1,245.9	93.25	9.68
Pathology	9,784		1,517		4,834.0	6.45	2.60
Pharmacy	34,522		11,677		37,209.6	2.96	9.17
Cardiovascular	2,815		47		149.8	59.89	0.75
PT/OT/ST	10,318		859		2,737.3	12.01	2.74
Psychiatric	12,935		175		557.6	73.91	3.43
Substance Abuse	0		0		0.0	0.00	0.00
Other	60,288		3,770		12,013.4	15.99	16.01
	\$329,549		19,041		60,675.5	\$17.31	\$87.51
Professional and Other State Plan Services							
Ambulatory Surgery Center	\$4,423		28		89.2	\$157.96	\$1.17
Office Visits	63,344		2,513		8,007.8	25.21	16.82
Preventive Medicine	13,817		300		956.0	46.06	3.67
Maternity	3,505		19		60.4	184.84	0.93
Certified Midwife	0		0		0.0	0.00	0.00
PT/OT/ST	3,756		390		1,242.8	9.63	1.00
Psychiatric and Substance Abuse	80,771		1,155		3,680.5	69.93	21.45
Radiology and Pathology	24,386		1,859		5,923.8	13.12	6.48
Home Health and Private Duty Nursing	6,992		983		3,132.4	7.11	1.86
Ambulance	30,577		1,658		5,283.3	18.44	8.12
Non-Emergency Transportation	236,468		13,789		43,939.6	17.15	62.79
Opioid Treatment Program	60,963		5,965		19,007.9	10.22	16.19
Federally Qualified and Rural Health Clinics	9,986		82		261.3	121.79	2.65
Adult Medical Day Care	1,948		178		567.2	10.94	0.52
Personal Care	105,944		19,652		62,622.5	5.39	28.13
Durable Medical Equipment	36,500		20,955		66,774.6	1.74	9.69
Applied Behavioral Analysis	00,000		20,000		0.0	0.00	0.00
Other	106,707		21,698		69,142.2	4.92	28.34
	\$790,088		91,224		290,691.5	\$8.66	\$209.81
Prescription Drugs	\$700,000		01,224		200,001.0	\$0.00	\$200.01
Generic Scripts	\$3,403		640		2,039.4	\$5.32	\$0.90
Single-Source Brand	1,042		7		22.3	148.81	0.28
Multi-Source Brand	4		2		6.4	1.96	0.00
Specialty	0		0		0.0	0.00	0.00
Hepatitis C	0		0		0.0	0.00	0.00
Other	23		9		28.7	2.52	0.00
	\$4,471		658		2,096.8	\$6.80	\$1.19
Community Mental Health Center	φ-1,-17-1		000		2,000.0		
Case Management	\$777,477		2,086		6,647.2	\$372.71	\$206.46
Long Term Support Service	316,550		11,864		37,805.5	26.68	84.06
Partial Hospital	2,340		24		76.5	97.48	0.62
Psychotherapy	163,988		4,240		13,511.1	38.68	43.55
Evidence Based Practice	20,643		1,072		3,416.0	19.26	43.55 5.48
Medication Management	4,401		1,072		3,416.0 509.9	27.51	5.48 1.17
Emergency Service 24/7	2,465		100		334.6	27.51	0.65
APRTP					334.6 226.2		
	40,980		71 2 752			577.18	10.88
Supported Employment Services	76,244		2,753		8,772.6	27.69	20.25
Other	60,807		1,309		4,171.2	46.45	16.15
All Services	\$1,465,894	05	23,684	070.0	75,470.7	\$61.89	\$389.26
All Services	\$2,730,086	85	135,068	270.9	430,403.4	\$20.21	\$724.97

Eligibility Category: Low Utilizer - Non-Dual

Member Months: 1,227

Dentria Total Paid Admits Total Paid Services Total Paid 1.000 Unitation Par (Des Par Services) Morenage Paid (Des Par (Des Paid) Morenage Paid (Des Paid) Morenage Paid) MorenagePaid) Morenage Paid) Morenage P								Per Capita	
Hespital Inpatient S22.739 6 17 5.87 16.33 3.789.88 5 Surgical 11,88 2 4 19.6 33.1 5.349.01 6 Maternity Delvary 0 0 0 0.0	Renefits							Monthly Paid	
Medical \$22,739 6 17 58.7 196.3 \$32,88.6 \$32,88.6 \$32,88.6 \$32,85.6 \$34,80.7 Materinity Non-Delivery 0 0 0 0.0<		Donard	Admito	001110000	1,000	1,000	onargo	0001	
Surgiant 11.88 2 4 16.6 39.1 5.548.01 Materity Diverse 0 0 0 0.0 0.0 0.0 0.0 Materity Diverse 0 0 0.0 0.0 0.0 0.0 0.0 Psychiatic 7.382 2 3.4 15.6 332.6 3.591.15 Other 0.0 0.0 0.0 0.0 0.0 0.00 Other 0.0 0.0 0.0 0.0 0.00 0.00 Surgery 27.234 22.234 27.33 874.80 5 Surgery 27.234 2.22 2.41.1 1.1 8 Pathology 10.890 9.999 8.811.1 1.1 8 Pathology 10.890 9.999 8.811.1 1.1 8 Pathology 10.890 9.999 8.811.2 1.229 1.1 1.1 1.056 2.33.7 Pathology 13.265 4.72 6.132.		\$22,739	6	17	58.7	166.3	\$3 789 86	\$18.53	
Materiny Non-Cellway 0								9.70	
Matering Non-Delivery 0							· · ·	0.00	
Well Newborn 0 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>0.00</td></t<>								0.00	
Psychiatric 7.882 2 34 19.6 332.6 3.891.15 Actorlal and Dutpatient 0 0 0 0.0 0.0 0.00 Hespital Dutpatient 5 97.8 558.0 \$42.019 5 Emergency Room 565.96 97.8 558.0 \$42.019 5 Surgery 27.264 28 17.3.5 67.4.00 5 Emergency Room 13.546 4.162 40.709.3 1.96.9 7 Pharmacy 13.546 4.162 40.709.3 3.28 Cardiovascular 2.803 2.22.1 Protectional and Other State Plan Services 3.87 11 10.76 2.33.7 Substance Abuse 1.57.2 14 13.68 3.47 12.29 51 Professional and Other State Plan Services 3.8771 2.07 6.272 61.34.7.7 \$2.98.01 1 Preventive Mickline 3.57.71 2.05 56.42 2.98.01 1 Preventive Mickline 0 0			-	-				0.00	
Alcohol and Drug Abuse 0 0 0 0.0 0.0 0.00 Other 92.019 10 55 97.8 538.0 \$4.201.95 \$ Emergency Room \$66.969 102 997.7 \$\$656.56 \$ \$ Surgery 27.294 28 27.39 974.80 \$ <								6.02	
Other 0 0 0.0 0.00 Hospital Outpatient 5 97.8 538.0 54.2019 5 Emergency Room \$66.969 102 997.7 \$656.56 5 Surgery 27.224 28 27.33 974.80 \$ Pathology 10.800 909 8.891.1 11.98 \$ Pathology 10.800 242.1 22.0 24.1 96.49 Pf(T/TST 8.505 27 2.24.1 96.49 \$ Other 2.857 61.7 61.83 12.97 \$ Other 2.857 20 19.56 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>· · ·</td> <td>0.00</td>							· · ·	0.00	
Statute Statute <t< td=""><td></td><td></td><td>-</td><td></td><td></td><td></td><td></td><td>0.00</td></t<>			-					0.00	
Hospital Outpatient Emergency Room S66.660 102 997.7 5565.56 S Surgery 27.294 28 27.39 974.80 S Radiology 10.890 909 8.891.1 11.98 Pharmacy 13.648 4.162 40.709.3 3.28 Cardiovascular 2.605 2.77 2.64.1 96.49 Phythitic 2.837 11 107.6 2.93.77 Subtaince Abuse 1.57.7 107 31.83 3.54.7 Other 2.22.7 6.67.72 61.347.7 52.98 51 Ambulatory Surgery Center 51.87.70 6.0 0 0 0.0 0.00 Office Visits 3.346.2 11.84 Materily 11.7 7 68.5 16.71 Preventive Medicine 4.535 383 3.746.2 11.84 Materily 10 1.075.9 22.80 170 Preventive Medicine 4.535 383 3.746.2 11.84 Materily 1.17			-					\$34.25	
Emergency Room S66, 960 102 997.7 5856.56 S Radiology 30,000 128 1,252.0 240,70 Pathology 10,860 909 8,891.1 1198 Cardiovascular 2,005 27 2,64.1 96.49 Protrostore 2,005 27 2,64.1 96.49 Protrostore 2,037 11 107.6 2,99.7 Substance Abuse 1,572 14 136.9 112.29 Other 2,237 627 6,134.7 528.601 Other 3,167.06 6,272 6,134.7 528.601 Other 3,167.06 6,572 60.1 40.1 Protestional and Other State Plan Services 3,135 5,524.2 60.1 11.4 Protestional and Other State Plan Services 3,257 5,564.2 60.1 11.9 Protestional and Dither State Plan Services 3,376.5 5,624.2 60.1 11.9 Protestional and Substance Abuse 3,376.5 5,624.2	Hospital Outpatient	• • • • • • • •					Ţ.,		
Surgery 27,294 28 27,39 97,480 Radiology 10,880 909 8,891,1 11,98 Pathology 10,880 909 8,891,1 11,98 Cardiovascular 2,605 2.7 2,841,1 66,49 Pit/DT/ST 8,603 2,624 2,852,2 3,224 Psychiatric 2,237 6,27 6,132,8 3,544 Other 2,237 6,27 6,132,8 3,547 Professional and Other State Plan Services 7 6,1347.7 529,83 51 Ambulatory Surgery Center 3,4571 6,755 5,624,2 60,12 Preventive Medicine 4,535 383 3,746,2 10,14 Materinity 117 7 6,65 16,71 Certified Midwife 0 0 0 0,00 0,00 Professitian and Subtanne Abuse 5,226 110 1,075,9 2,280 Psychiatric and Subtanne Abuse 6,273 2,61,45,1 11,44 14,465,6 7,7		\$66,969		102		997.7	\$656.56	\$54.59	
Radiology 30.800 128 1.252.0 240.70 Plathology 10.890 909 8.891.1 11.88 Pharmacy 13.548 4.162 40.709.3 3.26 Cardiovascular 2.655 2.7 2.841.9 96.49 PT/OTST 8.503 264 2.582.2 32.21 Other 2.637 11 107.6 23.97 Substance Abuse 1.572 44 136.9 112.29 Other 2.237 627 6.132 35.47 Other Visits 34.571 575 5.624.2 60.12 Prescional and Other State Plan Services 34.571 575 5.624.2 60.12 Prescional and Other State Plan Services 34.671 575 5.624.2 60.12 Prescional and Ther State Plan Services 34.871 575 5.624.2 60.12 Prescional and Ther State Plan Services 34.871 575 5.624.2 60.12 Prescional and Ther State Plan Services 34.567 7.68	5 ,							22.25	
Pathology 10.890 909 8.891.1 11.88 Cardiovascular 2.605 27 2.64.1 96.49 Prychistic 2.637 11 107.6 239.77 Substance Abuse 1.5.72 14 136.9 112.24 Other 2.2.37 6.272 6.132.8 35.47 Professional and Other State Plan Services 5187.066 6.272 6.132.8 35.47 Professional and Other State Plan Services 5187.066 6.272 0.195.6 5226.01 Office Visits 34.571 575 5.624.2 60.12 Preventive Medicine 4.535 333 3.746.2 11.44 Materity Winth 117 7 6.85 16.71 Certified Midwife 0 0 0 0.0 0 ProyChatric and Substance Abuse 50.288 812 7.942.3 61.33 Radiology and Pathology 14.991 6565 6.436.0 22.76 Abum Heattan and Private Duty Nursing 1.438 1								25.11	
Pharmacy Cardiovascular 13,648 4,162 40,709.3 3.26 Cardiovascular 2,605 27 264.1 96,40 Pri/OTST 8,603 264 2,582.2 32,21 Psychiatric 2,637 11 107.6 239,71 Substance Abuse 1,572 14 136.9 112.29 Other 22,237 627 6132.8 35.47 Professional and Other State Plan Services 6272 61.347.7 529.83 51 Ambulatory Surgery Center \$5,720 20 195.6 \$226.01 7 Office Visits 34,671 75 5.624.2 60.12 7 Preventive Medicine 4,335 383 3.746.2 11.84 Maternity 117 7 68.5 16.71 Certified Midwife 0 0 0.00 0.00 Pri/OTST 2,606 110 10.75.9 22.80 Psychiatric and Substance Abuse 50,288 812 7,942.3<								8.88	
Cardiovascular 2605 27 284.1 96.49 PryDrDTST 8.503 264 2.5822 32.21 Psychiatric 2.637 11 107.6 239.77 Substance Abuse 1572 14 138.9 112.29 Other 2.22.37 627 61.347.7 529.83 51 Professional and Other State Plan Services 627 61.347.7 529.83 51 Ambulatory Surgery Center 5.67.0 20 19.66 52.62.0 61.14 Office Visits 34.571 7 68.5 16.71 62.86.0 1 Preventive Medicine 4.535 383 3.74.62 11.04 1 107.55 22.80 1 1.07.51 22.80 1 1.07.51 22.80 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.03 1.02 1.02								11.04	
PT/OT/ST 8.503 264 2.882.2 3.2.21 Psychiatric 2.637 11 107.6 2.3.77 Substance Abuse 1.572 44 136.9 112.29 Other 2.2.237 6.27 6.13.47.7 52.9.83 51 Professional and Other State Plan Services 0 1 575 5.62.42 60.12 Ambulatory Surgery Center 35.720 20 195.6 5.286.01 1 Office Visits 34.571 5.75 5.62.42 60.12 1 1.44 Maternity 117 7 66.5 16.71 1 1 1.64 1				,		,		2.12	
Psychiatric 2.637 11 107.6 239.77 Other 22.237 627 613 35.47 Professional and Other State Plan Services 0 627 613.87.7 52.88 51 Professional and Other State Plan Services 0 0.134.77 52.86 51 Ambulatory Surgery Center 34.571 575 5.62.4 60.12 Office Visits 34.571 7 68.5 0.12 Preventive Medicine 4.535 38.33 3.74.62 11.14 Maternity 117 7 68.5 16.71 Certified Midwife 0 0 0 0.00 0.00 PTO/TST 2.508 110 1.075.9 2.240 Psychiatric and Substance Abuse 50.288 812 7.044.66 7.06 Non-Energency Transportation 37.085 2.673 2.6145.1 13.37 Opid Traimsportation 37.085 2.673 2.6145.1 13.92 Opid Traimsportation 37.085 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>6.93</td></td<>								6.93	
Substance Abuse 1.572 14 136.9 112.29 Other 22.237 627 61.328.8 35.47 Professional and Other State Plan Services 5187.066 6.272 61.327.7 52.98.3 51 Ambulatory Surgery Center 34.571 575 56.24.2 60.12 7 Office Visits 34.571 575 56.24.2 60.12 7 Orffice Visits 34.571 7 66.5 16.71 7 Certified Mikwife 0 0 0.0 0.0 0.0 0 Proynamic and Substance Abuse 50.288 812 7.942.3 61.93 8 Radiology and Pathology 14.991 656 64.456.0 22.78 Home Heatth and Private Duty Nursing 10.432 1.478 1.4456.6 7.06 Ambulance 6,713 241 2.357.3 27.86 10.0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00						,		2.15	
Other 22.237 627 6.132.8 33.47 Professional and Other State Plan Services Ambulatory Surgery Center 65.720 61.347.7 \$29.83 \$1 Ambulatory Surgery Center 55.720 20 195.6 \$228.601 \$1 Office Visits 34.571 575 5.624.2 60.12 \$1 Preventive Medicine 4.535 383 3.746.2 11.84 \$1 Certified Midwife 0 0 0.0 0.00 \$2.80 \$1 Proventive Medicine 4.535 383 3.746.2 11.84 \$1 Maternity 117 7 66.5 16.71 \$2.80 \$1 Proventive Medicine 0 0 0.0 0.00 \$2.78 \$1.74 \$1.85 \$3.83 Radiology and Pathology 14.491 2.357.3 \$2.86 \$1.33.07 \$1.22 \$1.414 \$2.357.3 \$2.86 \$1.34.83 \$1.33.07 \$1.22.57 \$1.57 \$1.57 \$1.57 \$1.57 \$1.57 \$1.57<	•							1.28	
S187.066 6.272 61.347.7 \$28.83 \$1 Ambulatory Surgery Center \$5.720 20 195.6 \$260.01 Office Visits 34.571 575 5624.2 60.12 Preventive Medicine 4.535 383 3.746.2 11.84 Matemity 117 7 66.5 16.71 Cettified Midwife 0 0 0.00 0.00 PT/OTSST 2.508 110 1.075.9 22.80 Psychiatric and Substance Abuse 50.288 612 7.942.3 61.93 Radiology and Pathology 14.991 66.65 64.466.0 22.78 Home Heatin and Private Duty Nursing 10.432 1.476 14.465.6 7.06 Ambulance 6.713 2.613 2.6145.1 13.380.7 10.22 Personal Care 7.678 1.4144 13.8315.5 5.43 Durable Medical Equipment 11.533 1.336 13.067.7 8.63 Applied Behavioral Analysis 0 0								18.13	
Professional and Other State Plan Services Ambulatory Surgery Center S5,720 20 195.6 \$286.01 Office Visits 34,571 575 5,624.2 60.12 Preventive Medicine 4,535 383 3,746.2 11.84 Matemity 117 7 68.5 16.71 Certified Midwife 0 0 0.0 0.00 Proventive Medicine 4,535 383 3,746.2 11.84 Matemity 117 7 68.5 16.71 Certified Midwife 0 0 0.0 0.00 ProfOT/ST 2,508 8110 1.075.9 22.80 Pachiotyand Pathology 14,4991 658 6,436.0 22.73 Home Health and Private Duty Nursing 10,432 1.478 14.456.6 7.06 Anbulance 6,713 2.241 2.357.3 27.86 Non-Emergency Transportation 37.085 2.673 26.145.1 13.87 Opicid Treatment Program 13,380.1 1.368 13.306								\$152.48	
Ambulatory Surgery Center \$\$720 20 195.6 \$288.01 Office Visits 34.571 575 5.624.2 60.12 Preventive Medicine 4.535 383 3.746.2 11.84 Maternity 117 7 68.5 16.71 Certified Midrife 0 0 0.00 0.00 Preventive Medicine 4.535 383 3.746.2 11.84 Maternity 117 7 68.5 16.71 Certified Midrife 0 0 0.00 0.00 Psychiatric and Substance Abuse 50.288 812 7.942.3 61.93 Radiology and Pathology 14.991 668 6.436.0 22.78 Home Health And Private Duty Nursing 10.432 1.1478 14.4456.6 70.66 Non-Emergency Transportation 37.7085 2.673 2.661.45.1 13.87 Opioid Treatment Program 13.881 1.336 1.33.067.7 8.63 Autil Medical Day Care 0 0 0.00	Professional and Other State Plan Services	, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		-,		,			
Office Visits 34,571 575 5,624.2 60.12 Preventive Medicine 4,535 383 3,746.2 11.84 Maternity 117 7 68.5 16.71 Certified Midwife 0 0 0.0 0.00 Pr/OT/ST 2,508 812 7,942.3 61.93 Radiology and Pathology 14,991 658 6,436.0 22.78 Home Health and Private Duty Nursing 10,432 1,478 14,456.6 7.06 Ambulance 6,713 241 2,257.3 27.86 Non-Emergency Transportation 37.085 2,673 26,145.1 13.87 Opioid Treatment Program 13,981 1,368 13,300.7 10.22 Federally Qualified and Rural Health Clinics 24,552 191 1,868.2 128.54 Adut Medical Equipment 11,533 1,336 13,367.7 82.04 32.73 Prescription Drugs 6 0 0 0.0 0.00 0.00 0.00 0.00 <		\$5.720		20		195.6	\$286.01	\$4.66	
Preventive Medicine 4,535 333 3,746.2 11.84 Maternity 117 7 66.5 16.71 Certified Midwife 0 0 0,00 0,00 PT/OT/ST 2,508 110 1,075.9 22.80 Psychiatric and Substance Abuse 50,288 812 7,942.3 61.33 Radiology and Pathology 14,991 658 6,436.0 22.78 Home Health and Private Duty Nursing 10,432 1,478 11,456.6 7.06 Ambulance 6,713 24.11 2,357.3 27.86 7.06 Opiod Treatment Program 13,381 1,368 13,380.7 10.22 Federally Outsified and Rural Health Clinics 24,552 191 1,868.2 128.54 Aduit Medical Day Care 0 0 0.00 0.00 0.00 Durable Medical Equipment 11,533 1,336 13.367.7 820.43 23.61 9.27.3 52.73 Generic Scripts \$13,86,26 6.075 59.40.8								28.18	
Matemity 117 7 68.5 16.71 Certified Midwife 0 0 0.0 0.00 0.00 PryOT/STST 2,508 110 1,075.9 22.80 Psychiatric and Substance Abuse 50,288 812 7,942.3 61.93 Radiology and Pathology 14,491 568 6,336.0 22.78 Home Health and Private Duty Nursing 10,432 1,478 14,456.6 7.06 Ambulance 6,713 241 2,357.3 27.86 Non-Emergency Transportation 37.085 2,673 26,145.1 13.87 Opiold Treatment Program 13.981 1,368 13.3807 10.22 Federally Qualified and Rural Health Clinics 24,552 191 1,868.2 128.54 Auti Medical Day Care 0 0 0.0 0.00 0.00 Personal Care 7,678 1,414 13.337.7 \$20.43 \$22 Stingle-Source Brand 245.609 666 6,416.5 374.40 2 <tr< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>3.70</td></tr<>								3.70	
Certified Midwife 0 0 0.0 0.00 PT/OT/ST 2,508 110 1,075.9 22.80 Psychiatric and Substance Abuse 50,288 812 7,942.3 61.93 Radiology and Pathology 14,991 658 6,436.0 22.78 Home Health and Private Duty Nursing 10,432 1,478 14,456.6 7.06 Ambulance 6,713 241 2,357.3 27.66 Non-Emergency Transportation 37,065 2,673 26,145.1 13.87 Opioid Treatment Program 13,981 1,368 13,380.7 10.22 Federally Qualified and Rural Health Clinics 24,552 191 1,868.2 128.54 Adult Medical Equipment 11,533 1,336 13,067.7 8.63 Applied Behavioral Analysis 0 0 0 0.00 0.00 Other 52,76,536 13,634 133,357.7 52,04.3 \$22.52 \$13 Beneric Scripts \$136,826 6,075 \$9,420.8 \$22.52.52		· · ·				,		0.10	
PT/OT/ST 2.508 110 1.075.9 22.80 Psychiatric and Substance Abuse 50.288 812 7.942.3 61.93 Radiology and Pathology 14,991 658 6.436.0 22.78 Home Health and Private Duty Nursing 10,432 1.478 14.456.6 7.06 Non-Emergency Transportation 37.085 2.673 26.145.1 13.87 Opiod Treatment Program 13.981 1.368 13.380.7 10.22 Federally Qualified and Rural Health Clinics 24.552 191 1.868.2 128.54 Adult Medical Day Care 0 0 0 0.0 0.00 Personal Care 7.678 1.414 13.381.5 5.43 Durable Medical Equipment 11.533 1.336 13.067.7 8.63 Applied Behavioral Analysis 0 0 0 0 0 Generic Scripts \$13.626 6.075 59.420.8 \$22.52 \$1 Generic Scripts \$13.682 6.075 59.420.8 \$22.52		1						0.00	
Psychiatric and Substance Abuse 50.288 812 7.942.3 61.93 Radiology and Pathology 14.991 658 6,436.0 22.78 Ambulance 6,713 241 2,357.3 27.66 Ambulance 6,713 241 2,357.3 27.66 Opiold Treatment Program 13,981 1,368 13,380.7 10.22 Federally Qualified and Rural Health Clinics 24,552 191 1,868.2 128.64 Adult Medical Day Care 0 0 0 0.00 0.00 Durable Medical Equipment 11,533 1,336 13,067.7 8.63 Applied Behavioral Analysis 0 0 0 0.00 0.00 Other 53.832 2.366 23,161.9 22.73 520.43 52 Single-Source Brand 225,056 11,634 133,357.7 520.43 52 Specialty 140,942 69 674.9 2,042.64 1 Hepattis C 0 0 0 0 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>2.04</td></t<>								2.04	
Radiology and Pathology 14,991 668 6,436.0 22.78 Home Health and Private Duty Nursing 10,432 1,478 14,456.6 7.06 Ambulance 6,773 241 2,357.3 27.86 Non-Emergency Transportation 37,085 2,673 26,145.1 13.387 Opioid Treatment Program 13,981 1,368 13,380.7 10.22 Federally Qualified and Rural Health Clinics 24,552 191 1,868.2 128.54 Adult Medical Day Care 0 0 0 0.0 0.00 Personal Care 7,678 1,414 133.837.7 820.43 427.35 Durable Medical Equipment 11,533 1,336 133.67.7 8.63 6.00 Other 53,832 2,368 23,161.9 22.73 5 Generic Scripts \$136,826 6.075 59,420.8 \$22.52 \$1 Generic Scripts \$136,826 6.075 59,420.8 \$22.52 \$1 Generic Scripts \$136,826 6.0						,		40.99	
Home Health and Private Duty Nursing 10,432 1,478 14,456.6 7.06 Ambulance 6,713 241 2,357.3 27.86 Non-Emergency Transportation 37,085 2,673 226,145.1 13.87 Opioid Treatment Program 13,981 1,368 13,380.7 10.22 Federally Qualified and Rural Health Clinics 24,552 191 1,868.2 128.54 Adult Medical Day Care 0 0 0.00 0.00 Durable Medical Equipment 11,533 1,336 133,067.7 8.63 Applied Behavioral Analysis 0 0 0.00 0.00 Otrer 53,832 2,368 23,161.9 22.73 Fescription Drugs 5 6,675 59,420.8 \$22.52 \$1 Generic Scripts \$136,826 6,075 59,420.8 \$22.52 \$1 Specially 14,245.6 0 0 0 0 0 Otrer 0 0 0 0.0 0.00 0.00								12.22	
Ambulance 6,713 241 2,357.3 27.86 Non-Emergency Transportation 37,085 2,673 26,145.1 13.87 Opioid Treatment Program 13,981 13,380.7 10.22 Federally Qualified and Rural Health Clinics 24,552 191 1,888.2 128,54 Adult Medical Day Care 0 0 0.00 0.00 Personal Care 7,678 1,414 13,831.5 5.43 Durable Medical Equipment 11,533 1,336 13,067.7 8.63 Applied Behavioral Analysis 0 0 0.00 0.00 Other 53,832 2,368 23,161.9 22.73 Prescription Drugs \$13,64.26 6,075 59,420.8 \$22.52 \$1 Single-Source Brand 25,306 115 1,124.8 22.005 \$2 Specially 140,942 69 67,437.0 \$79.35 \$4 Other 0 0 0.0 0.00 0.00 Other 0								8.50	
Non-Emergency Transportation 37,085 2,673 26,145.1 13.87 Opioid Treatment Program 13,981 1,368 13,380.7 10.22 Federally Qualified and Rural Health Clinics 24,552 191 1,868.2 122,54 Adult Medical Day Care 0 0 0.00 0.00 0.00 Personal Care 7,678 1,414 13,831.5 5.43 0 Durable Medical Equipment 11,533 13,366.7 8.63 0 0.0 0.00 0.00 Other 53,832 2,368 23,161.9 22,73 5 4 6 6 6 6 6 6 6 6 6 6 7 6 5 5 5	, ,			,		,		5.47	
Opioid Treatment Program 13,981 1,368 13,380.7 10.22 Federally Qualified and Rural Health Clinics 24,552 191 1,868.2 128,54 Adult Medical Day Care 0 0 0.0 0.00 0 Personal Care 7,678 1,414 13,831.5 5,43 0 Durable Medical Equipment 11,533 1,336 13,067.7 8,63 Applied Behavioral Analysis 0 0 0.0 0.00 0 Other 53,832 2,368 23,161.9 22.73 \$22.73 Frescription Drugs								30.23	
Federally Qualified and Rural Health Clinics 24,552 191 1,868.2 128.54 Adult Medical Day Care 0 0 0.0 0.00 0.00 Personal Care 7,678 1,414 13,831.5 5,433 Durable Medical Equipment 11,533 1,336 13,067.7 8,63 Applied Behavioral Analysis 0 0 0.00 0.00 Other 53,832 2,368 23,161.9 22,73 Prescription Drugs 5 13,634 133,357.7 \$20.43 \$22 Generic Scripts \$136,826 6,075 59,420.8 \$22.52 \$1 Specially 140,942 69 674.9 2,042.64 1 Hepatitis C 0 0 0 0.0 0.00 0 Other 0 0 0 0.0 0.0 0.0 0 Other 0 0 0 0.0 0.0 0.0 0 Other 0 0 0								11.40	
Adult Medical Day Care 0 0 0.0 0.00 Personal Care 7,678 1,414 13,831.5 5,43 Durable Medical Equipment 11,533 1,336 13,067.7 8,63 Applied Behavioral Analysis 0 0 0.0 0.00 Other 53,832 2,368 23,161.9 22.73 Prescription Drugs 133,357.7 \$20.43 \$22 Generic Scripts \$136,826 66,075 59,420.8 \$22.52 \$1 Single-Source Brand 245,609 6656 6,416.5 37.440 2 Specialty 140,942 69 674.9 2,042.64 1 Hepatitis C 0 0 0 0.00 0.00 Other 0 0 0 0.00 0.00 0.00 Case Management \$255,712 714 6,983.8 \$358.14 \$22 Partial Hospital 0 0 0 0.00 0.00 Psychotherapy 3						,		20.01	
Personal Care 7,678 1,414 13,831.5 5.43 Durable Medical Equipment 11,533 1,336 13,067.7 8.63 Applied Behavioral Analysis 0 0 0.0 0.00 Other 53,832 2,368 23,161.9 22.73 Prescription Drugs \$278,536 13,634 133,357.7 \$20.43 \$22 Generic Scripts \$136,826 6,075 59,420.8 \$22.52 \$1 Single-Source Brand 245,609 656 6,416.5 374.40 22 Hepatitis C 0 0 0 0.00 0						,		0.00	
Durable Medical Equipment 11,533 1,336 13,067.7 8.63 Applied Behavioral Analysis 0 0 0.0 0.00 0.00 Other 53,832 2,368 23,161.9 22.73 \$2 Prescription Drugs \$278,536 13,634 133,357.7 \$20.43 \$2 Generic Scripts \$136,826 6,075 59,420.8 \$22.52 \$1 Single-Source Brand 245,609 656 6,416.5 374.40 2 Specialty 140,942 69 674.9 2,042.64 1 Hepatitis C 0 0 0 0.00 0.00 0 Other 0 0 0 0.00 0.00 0.00 0.00 0.00 0 0 0.00 <td>•</td> <td>7.678</td> <td></td> <td>1.414</td> <td></td> <td></td> <td></td> <td>6.26</td>	•	7.678		1.414				6.26	
Applied Behavioral Analysis 0 0 0.0 0.00 Other 53,832 2,368 23,161.9 22.73 Prescription Drugs 133,357.7 \$20.43 \$2 Generic Scripts \$136,826 6,075 59,420.8 \$22.52 \$1 Single-Source Brand 245,609 6656 6,416.5 374.40 2 Specialty 140,942 669 674.9 2,042.64 1 Hepatitis C 0 0 0 0.00 0.00 0 Other 0 0 0 0.00 0.00 0 0 Case Management \$255,712 714 6,983.8 \$358.14 \$2 \$2 Long Term Support Service 27,955 1,251 12,263.3 2.2.52 \$4 Medication Management \$325,712 714 6,983.8 \$358.14 \$2 Long Term Support Service 27,955 1,251 12,263.3 2.2.54 \$4 Vidence Based Practice 3								9.40	
Other 53,832 2,368 23,161.9 22.73 Prescription Drugs \$278,536 13,634 133,357.7 \$20.43 \$22 Generic Scripts \$136,826 6,075 59,420.8 \$22.52 \$13 Single-Source Brand 245,609 6566 6,416.5 374.40 22 Specialty 140,942 69 674.9 2,042.64 1 Hepatitis C 0 0 0.0 0.00 0						,		0.00	
\$278,536 13,634 133,357.7 \$20.43 \$22 Generic Scripts \$136,826 6,075 59,420.8 \$22.52 \$1 Generic Scripts \$136,826 6,075 59,420.8 \$22.52 \$1 Single-Source Brand 245,609 656 6,416.5 374.40 2 Multi-Source Brand 25,306 115 1,124.8 220.05 2 Specialty 140,942 69 674.9 2,042.64 1 Hepatitis C 0 0 0.0 0.00 0 Other 0 0 0.0 0.00 0 0 Case Management \$255,712 714 6,983.8 \$358.14 \$2 Long Term Support Service 27,955 1,251 12,283.3 22.35 23 Partial Hospital 0 0 0 0.0 0.00 0 0 0 0 0 0 0 0 0 0 0 0 0 0		53.832		2.368				43.88	
Prescription Drugs \$136,826 6,075 59,420.8 \$22.52 \$13 Single-Source Brand 245,609 656 6,416.5 374.40 22 Specially 140,942 669 674.9 2,042.64 1 Hepatitis C 0 0 0 0.0 0.00 0 Other 0 0 0 0.0 0.00 0								\$227.03	
Generic Scripts \$136,826 6,075 59,420.8 \$22.52 \$1 Single-Source Brand 245,609 656 6,416.5 374.40 22 Multi-Source Brand 25,306 115 1,124.8 22,02.64 1 Specialty 140,942 69 67.637.0 \$2,042.64 1 Hepatitis C 0 0 0 0.00 0.00 0<	Prescription Drugs					/			
Multi-Source Brand 25,306 115 1,124.8 220.05 Specialty 140,942 69 674.9 2,042.64 1 Hepatitis C 0 0 0.0 0.00 0 0 Other 0 0 0.0 0.00 0 0 0 Community Mental Health Center \$548,683 6,915 67,637.0 \$79.35 \$44 Case Management \$255,712 714 6,983.8 \$358.14 \$25 \$		\$136,826		6,075		59,420.8	\$22.52	\$111.53	
Specialty 140,942 69 674.9 2,042.64 1 Hepatitis C 0 0 0 0.0 0.00 0	Single-Source Brand	245,609		656		6,416.5	374.40	200.20	
Hepatitis C 0 0 0.0 0.00 0.00 Other 0 0 0 0.0 0.00	Multi-Source Brand	25,306		115		1,124.8	220.05	20.63	
Hepatitis C 0 0 0.0 0.00 0.00 Other 0 0 0 0.0 0.00	Specialty	140,942		69		674.9	2,042.64	114.88	
Other 0 0 0.0 0.00 \$548,683 6,915 67,637.0 \$79.35 \$4 Case Management \$255,712 714 6,983.8 \$358.14 \$22 Long Term Support Service 27,955 1,251 12,236.3 22.35 22.35 Partial Hospital 0 0 0 0.00 0.00 0.00 Psychotherapy 38,284 660 6,455.6 58.01 58.01 58.01 58.01 59.00 660 6,455.6 58.01 660 6,457.6 22.654 49.01 49.01 49.01 66.01 6,635 58.01 59.00<	Hepatitis C	0		0		0.0	0.00	0.00	
\$548,683 6,915 67,637.0 \$79.35 \$4 Community Mental Health Center \$255,712 714 6,983.8 \$358.14 \$2 Long Term Support Service 27,955 1,251 12,236.3 22.35 \$2 Partial Hospital 0 0 0.0 0.00 0.00 \$265,712 \$2		0		0		0.0	0.00	0.00	
Case Management \$255,712 714 6,983.8 \$358.14 \$2 Long Term Support Service 27,955 1,251 12,236.3 22.35 Partial Hospital 0 0 0.0 0.00 Psychotherapy 38,284 660 64,556 58.01 Evidence Based Practice 345 13 127.2 26.54 Medication Management 7,390 269 2,631.1 27.47 Emergency Service 24/7 141 6 58.7 23.46 APRTP 0 0 0.00 0.00 0.00 Supported Employment Services 6,635 250 2,445.3 26.54 Other 28,624 410 4,010.3 69.82		\$548,683		6,915		67,637.0	\$79.35	\$447.23	
Long Term Support Service 27,955 1,251 12,236.3 22.35 Partial Hospital 0 0 0 0.0 0.00 Psychotherapy 38,284 660 6,455.6 58.01 Evidence Based Practice 345 13 127.2 26.54 Medication Management 7,390 269 2,631.1 27.47 Emergency Service 24/7 141 6 58.7 23.46 APRTP 0 0 0.00 0.00 Supported Employment Services 6,635 250 2,445.3 26.54 Other 28,624 410 4,010.3 69.82	Community Mental Health Center								
Partial Hospital 0 0 0.0 0.00 Psychotherapy 38,284 660 6,455.6 58.01 Evidence Based Practice 345 13 127.2 26.54 Medication Management 7,390 269 2,631.1 27.47 Emergency Service 24/7 141 6 58.7 23.46 APRTP 0 0 0.00 0.00 Supported Employment Services 6,635 250 2,445.3 26.54 Other 28,624 410 4,010.3 69.82	Case Management	\$255,712		714		6,983.8	\$358.14	\$208.43	
Partial Hospital 0 0 0.0 0.00 Psychotherapy 38,284 660 6,455.6 58.01 Evidence Based Practice 345 13 127.2 26.54 Medication Management 7,390 269 2,631.1 27.47 Emergency Service 24/7 141 6 58.7 23.46 APRTP 0 0 0.00 0.00 Supported Employment Services 6,635 250 2,445.3 26.54 Other 28,624 410 4,010.3 69.82		27,955		1,251		12,236.3	22.35	22.79	
Psychotherapy 38,284 660 6,455.6 58.01 Evidence Based Practice 345 13 127.2 26.54 Medication Management 7,390 269 2,631.1 27.47 Emergency Service 24/7 141 6 58.7 23.46 APRTP 0 0 0.00 0.00 Supported Employment Services 6,635 250 2,445.3 26.54 Other 28,624 410 4,010.3 69.82								0.00	
Evidence Based Practice 345 13 127.2 26.54 Medication Management 7,390 269 2,631.1 27.47 Emergency Service 24/7 141 6 58.7 23.46 APRTP 0 0 0.00 0.00 Supported Employment Services 6,635 250 2,445.3 26.54 Other 28,624 410 4,010.3 69.82		38,284		660				31.21	
Medication Management 7,390 269 2,631.1 27.47 Emergency Service 24/7 141 6 58.7 23.46 APRTP 0 0 0.00 0.00 Supported Employment Services 6,635 250 2,445.3 26,54 Other 28,624 410 4,010.3 69.82								0.28	
Emergency Service 24/7 141 6 58.7 23.46 APRTP 0 0 0.0 0.00 Supported Employment Services 6,635 250 2,445.3 26.54 Other 28,624 410 4,010.3 69.82								6.02	
APRTP 0 0 0.0 0.00 Supported Employment Services 6,635 250 2,445.3 26.54 Other 28,624 410 4,010.3 69.82								0.11	
Supported Employment Services 6,635 250 2,445.3 26.54 Other 28,624 410 4,010.3 69.82		1						0.00	
Other 28,624 410 4,010.3 69.82								5.41	
								23.33	
		\$365,086		3,573		34,948.2	\$102.18	\$297.58	
	All Services		10		97.8			\$1,158.57	

Eligibility Category: Low Utilizer - Dual

Member Months: 2,080

							Per Capita	
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost	
Hospital Inpatient	2011010	,		.,	.,			
Medical	\$31,712	25	127	144.2	732.8	\$1,268.50	\$15.25	
Surgical	24,792	12	165	69.2	952.0	2,065.98	11.92	
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00	
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00	
Well Newborn	0	0	0	0.0	0.0	0.00	0.00	
Psychiatric	4,020	3	33	17.3	190.4	1,340.00	1.93	
Alcohol and Drug Abuse	0	0	0	0.0	0.0	0.00	0.00	
Other	0	0	0	0.0	0.0	0.00	0.00	
	\$60,524	40	325	230.8	1,875.2	\$1,513.10	\$29.10	
Hospital Outpatient								
Emergency Room	\$41,771		149		859.7	\$280.34	\$20.08	
Surgery	23,185		90		519.3	257.61	11.15	
Radiology	16,412		179		1,032.8	91.69	7.89	
Pathology	3,348		612		3,531.1	5.47	1.61	
Pharmacy	10,887		33,023		190,532.7	0.33	5.23	
Cardiovascular	1,749		33		190.4	52.99	0.84	
PT/OT/ST	4,341		437		2,521.4	9.93	2.09	
Psychiatric	172		6		34.6	28.67	0.08	
Substance Abuse	0		0		0.0	0.00	0.00	
Other	29,780		2,237		12,906.8	13.31	14.32	
	\$131,647		36,766		212,128.6	\$3.58	\$63.30	
Professional and Other State Plan Services								
Ambulatory Surgery Center	\$4,798		25		144.2	\$191.90	\$2.31	
Office Visits	31,227		1,281		7,391.0	24.38	15.01	
Preventive Medicine	3,673		103		594.3	35.66	1.77	
Maternity	0		0		0.0	0.00	0.00	
Certified Midwife	0		0		0.0	0.00	0.00	
PT/OT/ST	3,189		426		2,457.9	7.48	1.53	
Psychiatric and Substance Abuse	22,882		369		2,129.0	62.01	11.00	
Radiology and Pathology	8,900		848		4,892.7	10.50	4.28	
Home Health and Private Duty Nursing	86		4		23.1	21.45	0.04	
Ambulance	12,773		519		2,994.5	24.61	6.14	
Non-Emergency Transportation	54,433		2,384		13,755.0	22.83	26.17	
Opioid Treatment Program	5,498		538		3,104.1	10.22	2.64	
Federally Qualified and Rural Health Clinics	2,334		27		155.8	86.46	1.12	
Adult Medical Day Care	5,992		124		715.4	48.32	2.88	
Personal Care	0		0		0.0	0.00	0.00	
Durable Medical Equipment	20,949		17,874		103,127.5	1.17	10.07	
Applied Behavioral Analysis	0		0		0.0	0.00	0.00	
Other	67,440		14,130		81,525.8	4.77	32.43	
Prescription Drugs	\$244,173		38,652		223,010.3	\$6.32	\$117.40	
Generic Scripts	\$1,584		535		3,086.8	\$2.96	\$0.76	
Single-Source Brand	479		11		63.5	43.57	0.23	
Multi-Source Brand	34		1		5.8	34.08	0.23	
Specialty	0		0		0.0	0.00	0.02	
Hepatitis C	0		0		0.0	0.00	0.00	
Other	0		0		0.0	0.00	0.00	
Other	\$2,097		547		3,156.0	\$3.83	\$1.01	
Community Mental Health Center	ψ2,007		547		0,100.0	φ0.00	φ1.01	
Case Management	\$395,254		1,062		6,127.4	\$372.18	\$190.04	
Long Term Support Service	78,917		3,232		18,647.7	24.42	37.94	
Partial Hospital	5,316		56		323.1	94.94	2.56	
Psychotherapy	33,425		940		5,423.5	35.56	16.07	
Evidence Based Practice	1,141		43		248.1	26.54	0.55	
Medication Management	9,943		356		2,054.0	27.93	4.78	
Emergency Service 24/7	422		18		103.9	23.46	0.20	
APRTP	5,220		9		51.9	580.02	2.51	
Supported Employment Services	49,799		1,816		10,477.8	27.42	23.94	
Other	26,777		764		4,408.0	35.05	12.87	
	\$606,216		8,296		47.865.4	\$73.07	\$291.47	
All Services	\$1,044,657	40	84,586	230.8	488,035.5	\$12.35	\$502.28	
	φ.,0++,007	-70	0-1,000	200.0		ψ1 <u>2.00</u>	\$00L.LU	

Eligibility Category: Serious Emotionally Disturbed Child

Member Months: 72,959

							Dor Conito
Panafila	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid	Per Capita Monthly Paid Cost
Benefits Hospital Inpatient	Dollars	Aumits	Services	1,000	1,000	Charge	COSI
Medical	\$190,827	46	167	7.6	27.5	\$4,148.41	\$2.62
Surgical	160,966	40	107	2.5	19.9	10,731.07	2.21
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	4,469	2	6	0.3	1.0	2,234.27	0.06
Well Newborn	4,405	0	0	0.0	0.0	0.00	0.00
Psychiatric	3,519,946	409	3,989	67.3	656.1	8,606.23	48.25
Alcohol and Drug Abuse	17,762	2	24	0.3	3.9	8,881.20	0.24
Other	0	0	0	0.0	0.0	0.00	0.00
	\$3,893,970	474	4,307	78.0	708.4	\$8,215.13	\$53.37
Hospital Outpatient	\$0,000,010		1,001			\$0,210110	
Emergency Room	\$1,484,069		3,549		583.7	\$418.17	\$20.34
Surgery	309,369		328		53.9	943.20	4.24
Radiology	294,504		1,483		243.9	198.59	4.04
Pathology	160,890		13,401		2,204.2	12.01	2.21
Pharmacy	63,190		25,269		4,156.2	2.50	0.87
Cardiovascular	9,694		145		23.8	66.86	0.13
PT/OT/ST	148,376		4,842		796.4	30.64	2.03
Psychiatric	123,064		372		61.2	330.82	1.69
Substance Abuse	0		0		0.0	0.00	0.00
Other	539,182		16,843		2,770.3	32.01	7.39
	\$3,132,339		66,232		10,893.6	\$47.29	\$42.93
Professional and Other State Plan Services	\$0,102,000		00,202		10,000.0	¢20	÷ 12.00
Ambulatory Surgery Center	\$34,314		112		18.4	\$306.38	\$0.47
Office Visits	1,052,311		17,896		2,943.5	58.80	14.42
Preventive Medicine	319,877		11,262		1,852.3	28.40	4.38
Maternity	4,397		36		5.9	123.35	0.06
Certified Midwife	182		7		1.2	26.01	0.00
PT/OT/ST	346,321		12,862		2,115.5	26.93	4.75
Psychiatric and Substance Abuse	241,456		4,073		669.9	59.28	3.31
Radiology and Pathology	163,501		10,840		1,782.9	15.08	2.24
Home Health and Private Duty Nursing	154,648		24,749		4,070.6	6.25	2.12
Ambulance	188,095		15,095		2,482.8	12.46	2.58
Non-Emergency Transportation	236,278		10,520		1,730.3	22.46	3.24
Opioid Treatment Program	0		0		0.0	0.00	0.00
Federally Qualified and Rural Health Clinics	734,042		4,923		809.7	149.10	10.06
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	0		0		0.0	0.00	0.00
Durable Medical Equipment	325,797		166,771		27,429.9	1.95	4.47
Applied Behavioral Analysis	146,114		10,009		1,646.2	14.60	2.00
Other	1,109,242		66,986		11,017.6	16.56	15.20
	\$5,056,576		356,141		58,576.8	\$14.20	\$69.31
Prescription Drugs	\$0,000,010				00,010.0	\$111 <u>2</u> 0	
Generic Scripts	\$2,998,736		92,025		15,136.0	\$32.59	\$41.10
Single-Source Brand	2,046,832		7,948		1,307.3	257.53	28.05
Multi-Source Brand	329,285		1,541		253.5	213.68	4.51
Specialty	1,610,991		307		50.5	5,247.53	22.08
Hepatitis C	0		0		0.0	0.00	0.00
Other	384		10		1.6	38.44	0.01
	\$6,986,228		101,831		16,748.8	\$68.61	\$95.76
Community Mental Health Center	\$0,000, <u>220</u>				10,11010	\$00.01	
Case Management	\$17,293,669		48,296		7.943.6	\$358.08	\$237.03
Long Term Support Service	8,083,935		353,717		58,178.2	22.85	110.80
Partial Hospital	0,000,000		000,717		0.0	0.00	0.00
Psychotherapy	8,408,169		94,768		15,587.1	88.72	115.25
Evidence Based Practice	53		2		0.3	26.54	0.00
Medication Management	6,980		262		43.1	26.64	0.00
Emergency Service 24/7	18,486		788		129.6	23.46	0.10
APRTP	9,333		17		2.8	549.00	0.23
Supported Employment Services	9,333		75		12.3	26.54	0.13
Other	1,192,853		75 13,059		2,147.9	20.54 91.34	
	\$35,015,469		510,984		84,044.9	\$68.53	16.35 \$479.94
All Services	\$54,084,581	474	1,039,495	78.0	170,972.6	\$52.03	\$741.30
	ψ 04,004,00 1	4/4	1,009,490	10.0	110,912.0	φυ2.03	φ141.3U

Eligibility Category: Granite Advantage - Medically Frail

Member Months: 71,345

	Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid
Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost
Hospital Inpatient	Donard	Addinto	00111000	1,000	1,000	onargo	0001
Medical	\$3,248,455	781	4,759	131.4	800.5	\$4,159.35	\$45.53
Surgical	2,722,248	294	2,672	49.5	449.4	9,259.35	38.16
Maternity Delivery	0	0	2,012	0.0	0.0	0.00	0.00
Maternity Non-Delivery	5,906	3	8	0.5	1.3	1,968.56	0.08
Well Newborn	515	1	1	0.2	0.2	515.30	0.01
Psychiatric	934,077	178	1,374	29.9	231.1	5,247.62	13.09
Alcohol and Drug Abuse	877,819	310	3,214	52.1	540.6	2,831.68	12.30
Other	0	0	0	0.0	0.0	0.00	0.00
	\$7,789,021	1,567	12,028	263.6	2,023.1	\$4,970.66	\$109.17
Hospital Outpatient	* · , · • • , • = ·	.,	,			÷ ,,• . • . • •	
Emergency Room	\$3,533,745		6,082		1,023.0	\$581.02	\$49.53
Surgery	1,053,196		1,521		255.8	692.44	14.76
Radiology	1,706,320		5,978		1,005.5	285.43	23.92
Pathology	524,959		39,484		6,641.1	13.30	7.36
Pharmacy	2,054,776		302,111		50,814.3	6.80	28.80
Cardiovascular	146,282		828		139.3	176.67	2.05
PT/OT/ST	367,447		13,193		2,219.0	27.85	5.15
Psychiatric	112,675		603		101.4	186.86	1.58
Substance Abuse	74,768		665		111.9	112.43	1.05
Other	1,548,988		50,078		8,423.0	30.93	21.71
	\$11,123,157		420,543		70,734.3	\$26.45	\$155.91
Professional and Other State Plan Services	* · · · , · _ • , · • • ·		,		,	+	
Ambulatory Surgery Center	\$128,562		443		74.5	\$290.21	\$1.80
Office Visits	1,857,811		31,242		5.254.8	59.47	26.04
Preventive Medicine	185,244		6,293		1,058.5	29.44	2.60
Maternity	33,404		146		24.6	228.02	0.47
Certified Midwife	626		11		1.9	56.91	0.01
PT/OT/ST	212,257		9,260		1,557.5	22.92	2.98
Psychiatric and Substance Abuse	4,321,094		40,588		6,826.8	106.46	60.57
Radiology and Pathology	1.416.024		45,587		7.667.6	31.06	19.85
Home Health and Private Duty Nursing	810,213		31,710		5,333.5	25.55	11.36
Ambulance	421,765		20,192		3,396.2	20.89	5.91
Non-Emergency Transportation	3,025,336		175,281		29,481.8	17.26	42.40
Opioid Treatment Program	1,628,292		159,347		26,801.8	10.22	22.82
Federally Qualified and Rural Health Clinics	1,450,268		9,753		1,640.4	148.70	20.33
Adult Medical Day Care	24,020		1,194		200.8	20.12	0.34
Personal Care	31,582		5,853		984.5	5.40	0.44
Durable Medical Equipment	1,195,856		150,264		25,274.0	7.96	16.76
Applied Behavioral Analysis	0		0		0.0	0.00	0.00
Other	3,279,404		220,116		37,023.0	14.90	45.97
	\$20,021,758		907,280		152,602.4	\$22.07	\$280.63
Prescription Drugs	<i>t</i>		,			¥==:•:	+
Generic Scripts	\$2,225,008		141,192		23,748.1	\$15.76	\$31.19
Single-Source Brand	4,874,044		15,884		2,671.6	306.85	68.32
Multi-Source Brand	2,918,298		13,591		2,286.0	214.72	40.90
Specialty	5,568,705		1,623		273.0	3,431.12	78.05
Hepatitis C	3,626,789		177		29.8	20,490.33	50.83
Other	405		31		5.2	13.05	0.01
	\$19,213,248		172,498		29,013.7	\$111.38	\$269.30
Community Mental Health Center			,				
Case Management	\$104,935		293		49.3	\$358.14	\$1.47
Long Term Support Service	160,247		2,678		450.4	59.84	2.25
Partial Hospital	0		_,0		0.0	0.00	0.00
Psychotherapy	168,960		4,996		840.3	33.82	2.37
Evidence Based Practice	456		27		4.5	16.90	0.01
Medication Management	4,798		157		26.4	30.56	0.07
Emergency Service 24/7	727		31		5.2	23.46	0.01
APRTP	61,199		115		19.3	532.17	0.86
Supported Employment Services	3,052		115		19.3	26.54	0.04
Other	303,409		4,418		743.1	68.68	4.25
	\$807,784		12,830	1	2,158.0	\$62.96	\$11.32
All Services	\$58,954,968	1,567	1,525,179	263.6	256,531.5	\$38.65	

Eligibility Category: Low Income Children - Age 0-11 Months

Member Months: 47,995

							D. 0
	Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid
Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost
Hospital Inpatient				,	,		
Medical	\$867,084	333	1,157	83.3	289.3	\$2,603.86	\$18.07
Surgical	212,743	27	111	6.8	27.8	7,879.37	4.43
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00
Well Newborn	580	1	1	0.3	0.3	580.42	0.01
Psychiatric	0	0	0	0.0	0.0	0.00	0.00
Alcohol and Drug Abuse	0	0	0	0.0	0.0	0.00	0.00
Other	0	0	0	0.0	0.0	0.00	0.00
	\$1,080,407	361	1,269	90.3	317.3	\$2,992.82	\$22.51
Hospital Outpatient			0.050				A / 7 7 7
Emergency Room	\$851,806		3,252		813.1	\$261.93	\$17.75
Surgery	137,968		158		39.5	873.21	2.87
Radiology	183,765		835		208.8	220.08	3.83
Pathology	56,139		4,282		1,070.6	13.11	1.17
Pharmacy	64,513		6,513		1,628.4	9.91	1.34
Cardiovascular	14,442		181		45.3	79.79	0.30
PT/OT/ST	54,497		1,705		426.3	31.96	1.14
Psychiatric	0		0		0.0	0.00	0.00
Substance Abuse	0		0		0.0	0.00	0.00
Other	551,985		9,689		2,422.5	56.97	11.50
	\$1,915,114		26,615		6,654.4	\$71.96	\$39.90
Professional and Other State Plan Services						****	* *****
Ambulatory Surgery Center	\$9,707		30		7.5	\$323.58	\$0.20
Office Visits	810,171		15,754		3,938.9	51.43	16.88
Preventive Medicine	952,995		47,833		11,959.4	19.92	19.86
Maternity	0		0		0.0	0.00	0.00
Certified Midwife	3,025		88		22.0	34.38	0.06
PT/OT/ST	23,427		905		226.3	25.89	0.49
Psychiatric and Substance Abuse	352		9		2.3	39.15	0.01
Radiology and Pathology	101,332		6,577		1,644.4	15.41	2.11
Home Health and Private Duty Nursing	234,348		4,271		1,067.9	54.87	4.88
	95,854		5,226		1,306.6	18.34	2.00
Non-Emergency Transportation	68,106		1,406		351.5	48.44	1.42
Opioid Treatment Program	0		0		0.0	0.00	0.00
Federally Qualified and Rural Health Clinics	985,022		6,986		1,746.7	141.00	20.52
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	0		0		0.0	0.00	0.00
Durable Medical Equipment	1,478,410		616,585		154,160.9	2.40	30.80
Applied Behavioral Analysis	0		0		0.0	0.00	0.00
Other	1,612,720		43,194		10,799.5	37.34	33.60
Prescription Drugs	\$6,375,470		748,864		187,233.8	\$8.51	\$132.83
Generic Scripts	\$148,424		11 011		2,803.0	\$13.24	\$3.09
Single-Source Brand	95,400		11,211 607		2,803.0	157.17	1.99
Multi-Source Brand	22,190		106		26.5	209.34	0.46
Specialty	316,197		108		20.5	2,927.75	6.59
Hepatitis C	0		0		0.0	0.00	0.00
Other	141		12		3.0	11.79	0.00
Other	\$582,352		12,044		3,011.3	\$48.35	\$12.13
Community Mental Health Center	\$362,352		12,044		3,011.3	φ40.33	φ12.13
Case Management	\$0		0		0.0	\$0.00	\$0.00
Long Term Support Service	0		0		0.0	0.00	0.00
	0		0		0.0	0.00	0.00
Partial Hospital Psychotherapy	0		0		0.0	0.00	0.00
Evidence Based Practice	0		0		0.0	0.00	0.00
Medication Management	0		0		0.0	0.00	0.00
Emergency Service 24/7	0		0		0.0	0.00	0.00
APRTP	0		0		0.0	0.00	0.00
	0		0		0.0		
Supported Employment Services Other	0		0		0.0	0.00 0.00	0.00 0.00
	\$0		0		0.0	\$0.00	\$0.00
All Services	\$9,953,343	361	788,792	90.3	197,216.7	\$0.00	\$0.00
	ψ9,900,040	301	100,192	90.3	131,210.1	μ φι <u>2.0</u> 2	¢∠01.30

Eligibility Category: Low Income Children - Age 1-18 Years

Member Months: 727,538

							Den Conito
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Per Capita Monthly Paid Cost
Hospital Inpatient	Donars	Admita	Octvices	1,000	1,000	onarge	0031
Medical	\$1,382,287	413	1,388	6.8	22.9	\$3,346.94	\$1.90
Surgical	1.091.815	113	462	1.9	7.6	9,662.08	1.50
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	17,780	6	16	0.1	0.3	2,963.37	0.02
Well Newborn	1,081	2	5	0.0	0.0	540.66	0.00
Psychiatric	1,698,565	287	1,884	4.7	31.1	5,918.35	2.33
Alcohol and Drug Abuse	58,113	16	159	0.3	2.6	3,632.06	0.08
Other	0	0	0	0.0	0.0	0.00	0.00
	\$4,249,642	837	3,914	13.8	64.6	\$5,077.23	\$5.84
Hospital Outpatient	ψ1,210,012	001	0,014	10.0	04.0	φ0,011.20	φ 0.04
Emergency Room	\$8,074,420		22,163		365.6	\$364.32	\$11.10
Surgery	2,236,292		2,281		37.6	980.40	3.07
Radiology	2,403,438		10,486		173.0	229.20	3.30
Pathology	974,236		80,805		1,332.8	12.06	1.34
Pharmacy	1,404,650		179,479		2,960.3	7.83	1.93
Cardiovascular	89,077		823		13.6	108.23	0.12
PT/OT/ST	898,060		27,606		455.3	32.53	1.23
Psychiatric	110,958		402		6.6	276.02	0.15
Substance Abuse	0		402		0.0	0.00	0.00
Other	4,167,013		123,096		2,030.3	33.85	5.73
Other	\$20,358,144		447,141		7,375.1	\$45.53	\$27.98
Professional and Other State Plan Services	φ20,000,144		447,141		7,575.1	φ+0.00	φ21.50
Ambulatory Surgery Center	\$464,530		1,231		20.3	\$377.36	\$0.64
Office Visits	6,593,515		124,881		2.059.8	52.80	9.06
Preventive Medicine	2,874,922		114,457		1,887.9	25.12	3.95
	45,406		198		3.3		
Maternity Certified Midwife						228.80	0.06
PT/OT/ST	1,618		17		0.3	97.11	0.00
	2,488,675		77,110		1,271.9	32.27	3.42
Psychiatric and Substance Abuse	3,878,406		107,177		1,767.8	36.19	5.33
Radiology and Pathology	963,139		70,467		1,162.3 2.487.2	13.67	1.32
Home Health and Private Duty Nursing	2,350,926		150,792		, -	15.59	3.23
Ambulance	485,342		36,387		600.2	13.34	0.67
Non-Emergency Transportation	928,076		20,012		330.1	46.38	1.28
Opioid Treatment Program	675		66		1.1	10.22	0.00
Federally Qualified and Rural Health Clinics	6,140,579		43,023		709.6	142.73	8.44
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	525		7		0.1	75.00	0.00
Durable Medical Equipment	3,666,883		1,428,863		23,567.6	2.57	5.04
Applied Behavioral Analysis	4,274,869		277,048		4,569.6	15.43	5.88
Other	6,437,903		467,653		7,713.5	13.77	8.85
	\$41,595,990		2,919,389		48,152.3	\$14.25	\$57.17
Prescription Drugs	A7 074 050		000 404		0.045.4	*****	* 0.70
Generic Scripts	\$7,074,052		233,121		3,845.1	\$30.34	\$9.72
Single-Source Brand	6,094,399		23,838		393.2	255.66	8.38
Multi-Source Brand	957,374		3,906		64.4	245.10	1.32
Specialty	12,042,936		1,769		29.2	6,807.77	16.55
Hepatitis C	0		0		0.0	0.00	0.00
Other	1,334		105		1.7	12.71	0.00
Community Mantal Haalth Contan	\$26,170,097		262,739		4,333.6	\$99.60	\$35.97
Community Mental Health Center	****		4 000		00.0	* ~~ * ~	* 0.00
Case Management	\$699,004		1,820		30.0	\$384.07	\$0.96
Long Term Support Service	365,803		5,212		86.0	70.18	0.50
Partial Hospital	0		0		0.0	0.00	0.00
Psychotherapy	252,387		2,898		47.8	87.09	0.35
Evidence Based Practice	0		0		0.0	0.00	0.00
Medication Management	108		4		0.1	27.06	0.00
Emergency Service 24/7	119		5		0.1	23.90	0.00
APRTP	6,551		11		0.2	595.53	0.01
Supported Employment Services	0		0		0.0	0.00	0.00
Other	462,677		3,355		55.3	137.91	0.64
All Services	\$1,786,649 \$94,160,521	837	13,305 3,646,488	13.8	219.5 60,145.1	\$134.28 \$25.82	\$2.46 \$129.42

Eligibility Category: Low Income Adults

Member Months: 133,084

							Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient	Donars	Aulinto	Services	1,000	1,000	Charge	0031
Medical	\$2,388,694	688	2,764	62.0	249.2	\$3,471.94	\$17.95
Surgical	1,584,216	200	1,461	18.0	131.7	7,921.08	11.90
Maternity Delivery	1,004,210	200	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	295,486	126	312	11.4	28.1	2,345.12	2.22
Well Newborn	29,867	35	61	3.2	5.5	853.35	0.22
Psychiatric	384,674	111	464	10.0	41.8	3,465.53	2.89
Alcohol and Drug Abuse	266,773	88	639	7.9	57.6	3,031.52	2.00
Other	200,775	0	000	0.0	0.0	0.00	0.00
	\$4,949,711	1,248	5,701	112.5	514.1	\$3,966.11	\$37.19
Hospital Outpatient	φ4,949,711	1,240	5,701	112.5	514.1	\$5,500.11	ψ57.19
Emergency Room	\$5,388,709		8,560		771.8	\$629.52	\$40.49
Surgery	1,500,116		1,631		147.1	919.75	11.27
Radiology	2,070,750		11,364		1,024.7	182.22	15.56
Pathology	1,049,198		79,703		7,186.7	13.16	7.88
Pharmacy	1,153,532		240,317		21,669.0	4.80	8.67
Cardiovascular	133,910		821		74.0	163.11	1.01
PT/OT/ST	364,303		12,268		1,106.2	29.70	2.74
Psychiatric	18,448		169		1,100.2	109.16	0.14
Substance Abuse	19,106		168		15.1	113.73	0.14
Other	2,289,129		75,907		6,844.4	30.16	17.20
Otilei	\$13,987,201		430,908		38,854.3	\$32.46	\$105.10
Professional and Other State Plan Services	φ13,307,201		430,900		30,034.3	ψ32.40	\$105.10
Ambulatory Surgery Center	\$119,939		467		42.1	\$256.83	\$0.90
Office Visits	2,390,297		407		42.1	53.13	30.90 17.96
Preventive Medicine			, ,		4,056.6	52.33	7.79
	1,037,239		19,822		· · · ·	258.37	
Maternity Certified Midwife	1,367,924		5,294 221		477.4 19.9	209.84	10.28 0.35
PT/OT/ST	46,416 200.931		8,951		807.1		
	,		30,972		2,792.7	22.45 102.88	1.51 23.94
Psychiatric and Substance Abuse	3,186,477		67,905		6,122.9	28.22	23.94 14.40
Radiology and Pathology	1,916,133		, ,		298.6		
Home Health and Private Duty Nursing Ambulance	136,366		3,312			41.17 16.28	1.02 2.01
	267,581		16,436		1,482.0		
Non-Emergency Transportation	2,296,233		151,769		13,684.8	15.13	17.25
Opioid Treatment Program	1,580,381		154,633		13,943.0	10.22	11.88
Federally Qualified and Rural Health Clinics	2,728,352		19,852		1,790.0	137.43	20.50
Adult Medical Day Care	1,707		100		9.0 0.0	17.07	0.01
Personal Care	0		0			0.00	0.00
Durable Medical Equipment	885,364		48,131		4,339.9	18.39	6.65
Applied Behavioral Analysis	0		0		0.0	0.00	0.00
Other	3,475,558		287,293		25,904.8	12.10	26.12
Prescription Drugs	\$21,636,896		860,150		77,558.4	\$25.15	\$162.58
Generic Scripts	\$3,133,709		170,725		15,394.0	\$18.36	\$23.55
Single-Source Brand			12,905		1,163.6	324.98	
Single-Source Brand Multi-Source Brand	4,193,868				,	324.98 226.46	31.51
	1,802,872		7,961 1,303		717.8 117.5		13.55 32.31
Specialty	4,299,808		, i			3,299.93	
Hepatitis C	1,199,308		89 46		8.0 4.1	13,475.37	9.01
Other	745 \$14,630,309		193,029		17,405.1	16.19 \$75.79	0.01 \$109.93
Community Mental Health Center	\$14,030,309		193,029		17,405.1	\$15.19	\$109.93
Case Management	\$77,127		199		17.9	\$387.57	\$0.58
Long Term Support Service	102,029		1,388		125.2	73.51	0.77
Partial Hospital	102,029		1,300		0.0	0.00	0.00
Psychotherapy	166,931		3,129		282.1	53.35	1.25
Psychotherapy Evidence Based Practice			3,129		282.1	53.35 26.54	0.00
	80		3 103		0.3 9.3	26.54 26.71	
Medication Management	2,751						0.02
Emergency Service 24/7	332		14		1.3	23.72	0.00
APRTP	39,901		67		6.0	595.53	0.30
Supported Employment Services	259		9		0.8	28.79	0.00
Other	232,555		3,033		273.5	76.67	1.75
Other All Services	232,555 \$621,965 \$55,826,081	1,248	3,033 7,945 1,497,733	112.5	273.5 716.4 135,048.3	\$78.28 \$37.27	\$4.67 \$419.48

Eligibility Category: CHIP

Member Months: 165,409

							Dor Conito
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1.000	Average Paid Charge	Per Capita Monthly Paid Cost
Hospital Inpatient	Donars	Adinto	00111003	1,000	1,000	onarge	0031
Medical	\$348,974	85	270	6.2	19.6	\$4,105.57	\$2.11
Surgical	316,888	26	229	1.9	16.6	12,188.01	1.92
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	3,000	1	2	0.1	0.1	2.999.82	0.02
Well Newborn	534	1	2	0.1	0.1	533.97	0.00
Psychiatric	325,908	122	366	8.9	26.6	2,671.37	1.97
Alcohol and Drug Abuse	30,795	4	103	0.3	7.5	7,698.75	0.19
Other	0	0	0	0.0	0.0	0.00	0.00
	\$1,026,098	239	972	17.3	70.5	\$4,293.30	\$6.20
Hospital Outpatient	1 1 1 1 1 1 1 1						
Emergency Room	\$1,491,489		3,715		269.5	\$401.48	\$9.02
Surgery	532,009		526		38.2	1,011.42	3.22
Radiology	544,214		2,352		170.6	231.38	3.29
Pathology	205,035		17,483		1,268.3	11.73	1.24
Pharmacy	211,529		38,927		2,824.1	5.43	1.28
Cardiovascular	16,583		186		13.5	89.16	0.10
PT/OT/ST	224,365		7,023		509.5	31.95	1.36
Psychiatric	32,124		80		5.8	401.55	0.19
Substance Abuse	0		0		0.0	0.00	0.00
Other	912,987		25,550		1,853.6	35.73	5.52
	\$4,170,334		95,842		6,953.1	\$43.51	\$25.21
Professional and Other State Plan Services	, , , , , , , , , , , , , , , , , , ,				-,		* -*·-·
Ambulatory Surgery Center	\$95,433		254		18.4	\$375.72	\$0.58
Office Visits	1,572,017		30,092		2,183.1	52.24	9.50
Preventive Medicine	753,353		30,575		2,218.1	24.64	4.55
Maternity	2,578		9		0.7	276.85	0.02
Certified Midwife	239		8		0.6	29.92	0.00
PT/OT/ST	751,654		22,445		1,628.3	33.49	4.54
Psychiatric and Substance Abuse	822,085		22,172		1,608.5	37.08	4.97
Radiology and Pathology	233,780		17,624		1,278.6	13.26	1.41
Home Health and Private Duty Nursing	253,389		23,941		1,736.9	10.58	1.53
Ambulance	83,124		5,784		419.6	14.37	0.50
Non-Emergency Transportation	163,348		2,129		154.5	76.73	0.99
Opioid Treatment Program	0		_,0		0.0	0.00	0.00
Federally Qualified and Rural Health Clinics	964,255		7,425		538.7	129.87	5.83
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	0		0		0.0	0.00	0.00
Durable Medical Equipment	861,653		239,127		17,348.1	3.60	5.21
Applied Behavioral Analysis	967,867		66,548		4,827.9	14.54	5.85
Other	1,513,840		93,755		6,801.7	16.15	9.15
	\$9,038,616		561,888		40,763.6	\$16.09	\$54.64
Prescription Drugs	, . , ,		,				
Generic Scripts	\$1,468,074		48,599		3,525.7	\$30.21	\$8.88
Single-Source Brand	1,438,380		5,729		415.6	251.07	8.70
Multi-Source Brand	155,951		893		64.8	174.64	0.94
Specialty	3,771,514		480		34.8	7,857.32	22.80
Hepatitis C	30,780		1		0.1	30,779,72	0.19
Other	417		17		1.2	24.54	0.00
	\$6,865,116		55,719		4,042.3	\$123.21	\$41.50
Community Mental Health Center	+=,===,=				.,	* · - * · - ·	
Case Management	\$83,914		216		15.7	\$388.49	\$0.51
Long Term Support Service	56,245		884		64.1	63.63	0.34
Partial Hospital	00,240		0		0.0	0.00	0.00
Psychotherapy	47,550		563		40.8	84.46	0.29
Evidence Based Practice	0		000		0.0	0.00	0.00
Medication Management	0		0		0.0	0.00	0.00
Emergency Service 24/7	0		0		0.0	0.00	0.00
APRTP	6,551		11		0.8	595.53	0.00
Supported Employment Services	0,551		0		0.0	0.00	0.04
Other	59,653		455		33.0	131.11	0.00
	\$253,913		2,129		154.5	\$119.26	\$1.54
All Services	\$21,354,077	239	716,550	17.3	51,983.9	\$29.80	\$129.10
	ψ21,004,011	209	710,000	1.3	51,305.9	ψ <u>2</u> 9.00	φ123.10

Eligibility Category: Foster Care / Adoption

Member Months: 23,159

							Den Cenite
	Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid
Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost
Hospital Inpatient							
Medical	\$94,537	30	88	15.5	45.6	\$3,151.24	\$4.08
Surgical	55,465	10	32	5.2	16.6	5,546.53	2.39
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	468,468	58	621	30.1	321.8	8,077.03	20.23
Alcohol and Drug Abuse	2,465	1	2	0.5	1.0	2,464.71	0.11
Other	0	0	0	0.0	0.0	0.00	0.00
	\$620,935	99	743	51.3	385.0	\$6,272.07	\$26.81
Hospital Outpatient							
Emergency Room	\$277,510		618		320.2	\$449.05	\$11.98
Surgery	110,871		124		64.3	894.12	4.79
Radiology	135,627		502		260.1	270.17	5.86
Pathology	45,266		3,563		1,846.2	12.70	1.95
Pharmacy	21,875		7,026		3,640.5	3.11	0.94
Cardiovascular	5,199		77		39.9	67.52	0.22
PT/OT/ST	79,705		2,308		1,195.9	34.53	3.44
Psychiatric	22,366		73		37.8	306.39	0.97
Substance Abuse	0		0		0.0	0.00	0.00
Other	212,964		6,096		3,158.6	34.94	9.20
	\$911,383		20,387		10,563.5	\$44.70	\$39.35
Professional and Other State Plan Services	<i><i><i>¢€1</i>,<i>0,0,0,0</i></i></i>		20,007				<i></i>
Ambulatory Surgery Center	\$19,185		58		30.1	\$330.77	\$0.83
Office Visits	264,021		4,795		2,484.5	55.06	11.40
Preventive Medicine	99,578		4,055		2,101.1	24.56	4.30
Maternity	16		4,000		0.5	15.98	0.00
Certified Midwife	34		1		0.5	33.60	0.00
PT/OT/ST	129,267		4,276		2,215.6	30.23	5.58
	· · ·						
Psychiatric and Substance Abuse	386,633		8,174		4,235.4	47.30	16.69
Radiology and Pathology	57,851		2,914		1,509.9	19.85	2.50
Home Health and Private Duty Nursing	662,232		15,987		8,283.7	41.42	28.59
Ambulance	32,510		2,916		1,510.9	11.15	1.40
Non-Emergency Transportation	29,675		852		441.5	34.83	1.28
Opioid Treatment Program	0		0		0.0	0.00	0.00
Federally Qualified and Rural Health Clinics	211,300		1,563		809.9	135.19	9.12
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	0		0		0.0	0.00	0.00
Durable Medical Equipment	461,634		195,293		101,191.3	2.36	19.93
Applied Behavioral Analysis	145,212		9,197		4,765.4	15.79	6.27
Other	411,211		20,708		10,729.9	19.86	17.76
	\$2,910,357		270,790		140,310.1	\$10.75	\$125.67
Prescription Drugs							
Generic Scripts	\$686,613		21,650		11,218.0	\$31.71	\$29.65
Single-Source Brand	503,102		1,608		833.2	312.87	21.72
Multi-Source Brand	91,855		240		124.4	382.73	3.97
Specialty	544,962		112		58.0	4,865.73	23.53
Hepatitis C	0		0		0.0	0.00	0.00
Other	58		16		8.3	3.62	0.00
	\$1,826,589		23,626		12,241.8	\$77.31	\$78.87
Community Mental Health Center							
Case Management	\$60,993		157		81.3	\$388.49	\$2.63
Long Term Support Service	36,643		583		302.1	62.85	1.58
Partial Hospital	0		0		0.0	0.00	0.00
Psychotherapy	19,933		232		120.2	85.92	0.86
Evidence Based Practice	0		0		0.0	0.00	0.00
Medication Management	ő		0		0.0	0.00	0.00
Emergency Service 24/7	ő		0		0.0	0.00	0.00
APRTP	0		0		0.0	0.00	0.00
Supported Employment Services	0		0		0.0	0.00	0.00
Other	37,588		281		145.6	133.76	1.62
					140.0	133./0	1.02
All Services	\$155,157 \$6,424,421	99	1,253 316,799	51.3	649.2 164,149.7	\$123.83 \$20.28	\$6.70 \$277.40

Eligibility Category: Severely Disabled Children

Member Months: 10,785

							D
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Per Capita Monthly Paid Cost
Hospital Inpatient	Donars	Adinto	00111003	1,000	1,000	onarge	0031
Medical	\$161,361	34	164	37.8	182.5	\$4,745.91	\$14.96
Surgical	52,597	7	35	7.8	38.9	7,513.91	4.88
Maternity Delivery	02,007	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00
Well Newborn	0	ů 0	0	0.0	0.0	0.00	0.00
Psychiatric	42,032	6	39	6.7	43.4	7,005.36	3.90
Alcohol and Drug Abuse	12,002	0	0	0.0	0.0	0.00	0.00
Other	0	ů 0	0	0.0	0.0	0.00	0.00
	\$255,990	47	238	52.3	264.8	\$5,446.61	\$23.74
Hospital Outpatient	\$200,000		200	02.0	201.0	φ0,110.01	φ20.14
Emergency Room	\$91,379		181		201.4	\$504.86	\$8.47
Surgery	41,860		53		59.0	789.81	3.88
Radiology	119,467		224		249.2	533.34	11.08
Pathology	30,341		2,242		2,494.5	13.53	2.81
Pharmacy	45,876		6,145		6,837.1	7.47	4.25
Cardiovascular	4,102		50		55.6	82.05	0.38
PT/OT/ST	118,424		3,617		4,024.4	32.74	10.98
Psychiatric	8,652		33		36.7	262.18	0.80
Substance Abuse	0,002		0		0.0	0.00	0.00
Other	89,856		1,932		2.149.6	46.51	8.33
	\$549,958		14,477		16,107.6	\$37.99	\$50.99
Professional and Other State Plan Services	\$040,000		11,111		10,107.0		\$00.00
Ambulatory Surgery Center	\$7,335		24		26.7	\$305.61	\$0.68
Office Visits	169,630		2,140		2.381.0	79.27	15.73
Preventive Medicine	20,473		811		902.3	25.24	1.90
Maternity	20,479		0		0.0	0.00	0.00
Certified Midwife	0		0		0.0	0.00	0.00
PT/OT/ST	279,507		12,958		14,417.5	21.57	25.92
Psychiatric and Substance Abuse	73,134		4,580		5,095.9	15.97	6.78
Radiology and Pathology	27,355		1,420		1,579.9	19.26	2.54
Home Health and Private Duty Nursing	3,928,373		260,248		289,560.9	15.09	364.24
Ambulance	71,206		11,309		12,582.8	6.30	6.60
Non-Emergency Transportation	173,233		4,398		4,893.4	39.39	16.06
Opioid Treatment Program	0		4,000		4,000.4	0.00	0.00
Federally Qualified and Rural Health Clinics	18,877		166		184.7	113.71	1.75
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	6,421		1,190		1,324.0	5.40	0.60
Durable Medical Equipment	1,472,017		1,190,718		1,324,834.0	1.24	136.48
Applied Behavioral Analysis	441,295		34,607		38,504.9	12.75	40.92
Other	473,015		16,019		17,823.3	29.53	43.86
	\$7,161,870		1,540,588		1,714,111.4	\$4.65	\$664.05
Prescription Drugs	ψ1,101,010		1,040,000		1,714,111.4	φ+.00	φ004.00
Generic Scripts	\$535,357		16,009		17,812.2	\$33.44	\$49.64
Single-Source Brand	749,920		1,732		1,927.1	432.98	69.53
Multi-Source Brand	599,314		794		883.4	754.80	55.57
Specialty	1,766,230		316		351.6	5,589.33	163.76
Hepatitis C	0		0.0		0.0	0.00	0.00
Other	22		13		14.5	1.68	0.00
	\$3,650,843		18,864		20,988.7	\$193.53	\$338.50
Community Mental Health Center	ψ0,000,0 1 0		10,004		20,000.1	φ100.00	4000.00
Case Management	\$24,864		65		72.3	\$382.52	\$2.31
Long Term Support Service	29,292		986		1,097.1	29.71	2.72
Partial Hospital	25,252		0		0.0	0.00	0.00
Psychotherapy	4,014		45		50.1	89.20	0.00
Evidence Based Practice	4,014		45 0		0.0	0.00	0.00
Medication Management	0		0		0.0	0.00	0.00
Emergency Service 24/7	94		0 4		4.5	23.46	0.00
APRTP	94		4		4.5	0.00	0.00
	0		0		0.0	0.00	0.00
Supported Employment Services Other	6,671		0 59		0.0 65.6	113.07	
	\$64,935		1,159		1,289.5	\$56.03	0.62
All Services	\$04,935	47	1,575,326	52.3	1,752,762.1	\$7.42	\$0.02
	ψ11,000,090	47	1,010,020	JZ.3	1,132,102.1	μ ψι.4Ζ	ψ1,003.30

Eligibility Category: Elderly and Disabled Adults - Age 19-64

Member Months: 73,573

							Dan Canita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1.000	Average Paid Charge	Per Capita Monthly Paid Cost
Hospital Inpatient	Donars	Aumits	Services	1,000	1,000	Charge	0031
Medical	\$4,244,205	956	5,721	155.9	933.1	\$4,439.55	\$57.69
Surgical	2,787,563	280	3,084	45.7	503.0	9,955.58	37.89
Maternity Delivery	2,707,505	200	0,004	0.0	0.0	0.00	0.00
Maternity Non-Delivery	6,471	3	9	0.5	1.5	2,157.13	0.09
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	540,427	97	707	15.8	115.3	5,571.41	7.35
Alcohol and Drug Abuse	187,863	72	410	11.7	66.9	2,609.20	2.55
Other	0	0	0	0.0	0.0	0.00	0.00
	\$7,766,529	1,408	9,931	229.7	1,619.8	\$5,516.00	\$105.56
Hospital Outpatient	Ţ: ,: c c,c=c	.,	-,		.,		
Emergency Room	\$4,165,285		5,563		907.3	\$748.75	\$56.61
Surgery	1,726,839		2,063		336.5	837.05	23.47
Radiology	2,389,277		8,291		1,352.3	288.18	32.48
Pathology	608,971		50,321		8,207.5	12.10	8.28
Pharmacy	3,001,760		352,234		57,450.7	8.52	40.80
Cardiovascular	173,503		1,073		175.0	161.70	2.36
PT/OT/ST	443,516		14,938		2,436.4	29.69	6.03
Psychiatric	34,858		150		24.5	232.39	0.47
Substance Abuse	10,025		89		14.5	112.64	0.14
Other	2,365,388		71,687		11,692.4	33.00	32.15
	\$14,919,423		506,409		82,597.2	\$29.46	\$202.78
Professional and Other State Plan Services							
Ambulatory Surgery Center	\$167,604		635		103.6	\$263.94	\$2.28
Office Visits	1,944,446		35,233		5,746.6	55.19	26.43
Preventive Medicine	253,324		8,540		1,392.9	29.66	3.44
Maternity	17,205		80		13.0	216.25	0.23
Certified Midwife	0		0		0.0	0.00	0.00
PT/OT/ST	275,618		12,170		1,985.0	22.65	3.75
Psychiatric and Substance Abuse	1,222,810		16,096		2,625.3	75.97	16.62
Radiology and Pathology	1,077,329		40,862		6,664.7	26.37	14.64
Home Health and Private Duty Nursing	4,166,815		226,509		36,944.5	18.40	56.64
Ambulance	489,959		27,820		4,537.5	17.61	6.66
Non-Emergency Transportation	2,028,900		120,401		19,637.9	16.85	27.58
Opioid Treatment Program	708,557		69,330		11,308.0	10.22	9.63
Federally Qualified and Rural Health Clinics	1,295,121		8,904		1,452.3	145.45	17.60
Adult Medical Day Care	158,590		8,422		1,373.7	18.83	2.16
Personal Care	1,164,091		215,336		35,122.1	5.41	15.82
Durable Medical Equipment	3,010,074		1,289,519		210,325.4	2.33	40.91
Applied Behavioral Analysis	6,950		1,271		207.3	5.47	0.09
Other	4,809,450		365,564		59,624.9	13.16	65.37
Barren da di se Barren	\$22,796,845		2,446,692		399,064.7	\$9.32	\$309.85
Prescription Drugs	¢E 0E4 700		250 000		40.004.7	¢10 50	¢60.70
Generic Scripts	\$5,054,709		259,232		42,281.7	\$19.50	\$68.70
Single-Source Brand Multi-Source Brand	11,789,638		29,941		4,883.5	393.76 441.16	160.24 47.83
	3,518,673		7,976		1,300.9		
Specialty	11,343,439		1,867 91		304.5	6,075.76 13.710.85	154.18
Hepatitis C Other	1,247,687 232		91 65		14.8 10.6	-,	16.96 0.00
Other	\$32,954,379		299,172		48,796.1	3.57 \$110.15	\$447.92
Community Mental Health Center	\$32,954,579		299,172		40,790.1	\$110.15	φ447.9Z
Case Management	\$62,765		163		26.6	\$385.06	\$0.85
	98,831					63.43	1.34
Long Term Support Service Partial Hospital	90,031		1,558 0		254.1 0.0	0.00	0.00
Psychotherapy	148,682		2,955		482.0	50.32	2.02
Evidence Based Practice	623		2,955		482.0	13.25	0.01
Medication Management	6,132		189		30.8	32.44	0.01
Emergency Service 24/7	331		14		2.3	23.62	0.08
APRTP	53,002		89		14.5	595.53	0.00
Supported Employment Services	1,411		52		8.5	27.13	0.72
Other	214,692		2,821		6.5 460.1	76.10	2.92
	\$586,468		7,888		1,286.6	\$74.35	\$7.97
All Services	\$79,023,644	1,408	3,270,092	229.7	533,364.4	\$24.17	\$1,074.09
	ψ10,020,044	1,400	0,210,092	223.1	000,004.4	ψ <u></u>	ψ1,01-1.09

Eligibility Category: Elderly and Disabled Adults - Age 65+

Member Months: 12,753

	Trial Data	THERE	Trial Drive			A	Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient	Donars	Adinto	00111003	1,000	1,000	onarge	0031
Medical	\$1,196,490	283	1,682	266.3	1,582.6	\$4,227.88	\$93.82
Surgical	877,034	66	461	62.1	433.8	13,288.40	68.77
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	54,907	2	52	1.9	48.9	27,453.31	4.31
Alcohol and Drug Abuse	0	0	0	0.0	0.0	0.00	0.00
Other	0	0 351	0	0.0	0.0	0.00	0.00
Hospital Outpatient	\$2,128,431	301	2,195	330.3	2,065.3	\$6,063.91	\$166.89
Emergency Room	\$623,306		637		599.4	\$978.50	\$48.87
Surgery	220,009		297		279.5	740.77	17.25
Radiology	385,493		1,535		1,444.3	251.14	30.23
Pathology	116,138		11,337		10,667.2	10.24	9.11
Pharmacy	297,510		45,825		43,117.7	6.49	23.33
Cardiovascular	34,356		316		297.3	108.72	2.69
PT/OT/ST	71,425		2,738		2,576.2	26.09	5.60
Psychiatric	0		0		0.0	0.00	0.00
Substance Abuse	0		0		0.0	0.00	0.00
Other	492,073		8,154		7,672.3	60.35	38.58
Professional and Other State Plan Services	\$2,240,309		70,839		66,653.9	\$31.63	\$175.66
Ambulatory Surgery Center	\$63,868		363		341.6	\$175.94	\$5.01
Office Visits	316,640		5,789		5,447.0	54.70	24.83
Preventive Medicine	34,499		1,231		1,158.3	28.03	2.71
Maternity	0		0		0.0	0.00	0.00
Certified Midwife	0		0		0.0	0.00	0.00
PT/OT/ST	34,838		1,489		1,401.0	23.40	2.73
Psychiatric and Substance Abuse	12,651		244		229.6	51.85	0.99
Radiology and Pathology	146,866		6,834		6,430.3	21.49	11.52
Home Health and Private Duty Nursing	650,955		9,771		9,193.7	66.62	51.04
Ambulance	111,277		5,043		4,745.1	22.07	8.73
Non-Emergency Transportation	159,969		8,317		7,825.6	19.23	12.54
Opioid Treatment Program	11,344		1,110		1,044.4	10.22	0.89
Federally Qualified and Rural Health Clinics	291,428		2,106		1,981.6	138.38 19.98	22.85
Adult Medical Day Care Personal Care	389,696 2,188		19,501 103		18,348.9 96.9	21.24	30.56 0.17
Durable Medical Equipment	294,968		176,096		165,692.5	1.68	23.13
Applied Behavioral Analysis	204,000		0		0.0	0.00	0.00
Other	943,993		54,688		51,457.1	17.26	74.02
	\$3,465,181		292,685		275,393.6	\$11.84	\$271.71
Prescription Drugs							
Generic Scripts	\$715,114		52,648		49,537.6	\$13.58	\$56.07
Single-Source Brand	1,734,238		6,212		5,845.0	279.18	135.98
Multi-Source Brand	253,230		576		542.0	439.64	19.86
Specialty	656,762		143		134.6	4,592.74	51.50
Hepatitis C	0		0		0.0	0.00	0.00
Other	57 \$3,359,401		22 59,601		20.7 56,079.9	2.58 \$56.36	0.00 \$263.41
Community Mental Health Center	\$3,339,401		59,001		50,079.9	\$30.30	\$203.4T
Case Management	\$1,942		5		4.7	\$388.49	\$0.15
Long Term Support Service	420		5		4.7	83.91	0.03
Partial Hospital	0		0		0.0	0.00	0.00
Psychotherapy	2,962		41		38.6	72.24	0.23
Evidence Based Practice	0		0		0.0	0.00	0.00
Medication Management	0		0		0.0	0.00	0.00
Emergency Service 24/7	0		0		0.0	0.00	0.00
APRTP	0		0		0.0	0.00	0.00
Supported Employment Services	0		0		0.0	0.00	0.00
Other	5,067		58		54.6	87.37	0.40
All Convision	\$10,391	054	109	000.0	102.6	\$95.33	\$0.81
All Services	\$11,203,714	351	425,429	330.3	400,295.2	\$26.34	\$878.48

Eligibility Category: Dual Eligibles

Member Months: 172,429

Benefits Doal Paid Total Paid Services Jutics Part Utics Part Average Paid Monthly Paid Heopfal Inpationt 53,304,019 2,314 11,980 161.0 883.7 51,427.54 819.16 Surgial 1,102,278 555 5,468 44.2 880.4 1,736.57 64.10 Surgial 1,102,278 555 5,468 44.2 880.4 1,736.57 64.10 Well Newborn 0 0 0 0 0.0 0.00 <								Per Capita
Heepital Inpatient S. 304.019 2.314 11.980 16.1 8337 (1) 9337 (1)		Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	
Medical 53.304.019 2.214 11.080 161.0 63.37 51.422.44 S15.1 Materity Non-Delwary 10.978 655 5.468 4.2 380.4 11.736.97 6.40 Materity Non-Delwary 10.989 6 0 0.13 1.363.80 0.00 0.00 Psychiatric 291.343 122 15.25 6.5 1109.1 2.206.00 1.63 Acobal and Drug Abuse 71.317 36 2216.8 1.335.7 31.531.48 2.01.0 0.00	Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost
Surgical 1.102.978 6.535 5.466 4.42 3.004 1.73.677 6.40 Materiny Doir-Delivery 10.969 8 16 0.6 1.3 1.363.60 0.00 Materiny Doir-Delivery 10.969 8 16 0.6 1.3 1.363.60 0.00 Applied Toring Abuse 21.317 12 2.5 1.6.8 1.81.7 0.00 </td <td>Hospital Inpatient</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Hospital Inpatient							
Materinity Non-Delivery 0								
Maternik Non-Delivery 10.09 8 16 0.6 1.3 1.338.60 0.06 Psychiatric 281,343 122 15.25 8.5 108.1 2.306.09 16.83 Achol and Dug Abuse 71.31 36 281 2.5 8.5 108.1 1.981.02 0.41 Cher 42,770.596 3,115 19.20 216.8 1.339.7 \$1,531.48 \$227.07 Heaptial Outpatient \$2,770.766 3,415.4 34.1 2.806.06 8.83 Radiology 1,135.676 12,411 363.7 91.51 6.59 Synthology 1,836.46 3,31.00 2.318.9 5.67 1.10 Planmacular 2.010.891 71.477 4.475.5 2.24 11.33 10.8 Synthology 1.80.48 3.330 2.318.9 5.67 1.10 10.6 Prothology 1.317.00.42 99.95 9.06.8 2.20.7 0.43 30.47 10.23 4.83 1.1.8 10.60.0 1		, . ,		,				
Well Newborn 0 <t< td=""><td></td><td></td><td>-</td><td>-</td><td></td><td></td><td></td><td></td></t<>			-	-				
Psychiatric 281,343 122 1.525 8.5 106.1 2.206.00 1.63 Other 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0.00 <td></td> <td>· · ·</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>		· · ·						
Alcohol and Ding Abuse 71,37 38 261 2.5 18.2 1981.02 0.41 Other 0 0 0 0.0 0.0 0.0 0.0 0.0 Entregrey Room 53.01.66 3.115 19.250 216.8 314.1 280.26 6.83 Bardology 1.135.676 124.11 863.7 916.16 6.89 Pathology 1.85.41 2.472 172.0 66.89 0.98 PTOTIST 4.28.049 33.109 2.682.2 11.20 4.8 Statistice Abuse 2.99.02 192.44 92.68 952.67 956.79 Other 511.205.442 954.955 0.86.493 152.26 857.89 Preventive Medicine 128.399	Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Other 0 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>, ·</td> <td></td>							, ·	
S4.70.666 3.115 19.250 216.8 1.339.7 \$1.531.48 \$27.87 Emergency Room \$3.616.634 6.217 571.9 \$440.14 \$20.97 Surgery 1.135.676 124.11 863.7 91.51 6.58 Radiology 1.98.048 33.30 2.318 5.57 1.10 Phamagener 2.01.924 71.6565 4.74.41 863.7 91.51 6.59 Promany Inter 2.01.924 71.6565 4.74.75 6.28 11.80 4.75.8 2.85 6.64.91 51.26 12.3 2.46 13.33 2.85 6.64.91 51.22 57.87 9.768.8 2.081 0.02 57.87 9.768.8 2.081 1.82.26 57.87 9.768.8 2.081 1.22.8 587.89 9.901 62.7 \$107.39 \$2.06 7.64.8 2.056 7.64.01 8.22.67 7.159 \$2.66 7.16.8 7.173 \$2.66 7.16.8 7.16.8 7.16.8 7.173 \$2.06 7.44 0.05								
Hespital Outpatient Emergency Room \$3,616,634 8,217 571.9 \$44.014 \$20.97 Surgery 1,177,986 4,514 314.1 260.96 6.83 Pathology 1,135,876 12,411 863.7 91.51 6.59 Pathology 199,048 33,330 2,319.6 6.57 11.0 Cardiovascular 165,341 2,472 172.0 66.89 0.86 Cardiovascular 165,341 2,472 172.0 66.89 0.86 Substance Abuse 3,881 35 2,4 113.73 0.02 Other 2,033.062 139.477 97.06.8 2.041 16.84 Childro Visis 1,350.015 62.248 4.332.1 2.199 7.83 Preventive Medicine 128,939 5.742 399.6 2.246 0.75 Maternity 23.571 144 0.00 16.50 0.14 Certified Midufe 112 5 0.3 2.250 0.00 Preventive Medic	Other							0.00
Emergency Room \$3.816.834 8.217 571.9 \$440.14 \$20.07 Radiology 1.135.876 12.411 863.7 91.51 6.58 Radiology 1.98.048 33.330 2.319.6 5.57 1.10 Pharmacy 2.010.992 774.966 49.756.6 2.81 1.66 Cardiovascular 165.341 2.477 172.0 66.89 0.98 PTI/CITST 42.80.49 38.199 2.462.2 11.23 2.48 Substance Abuse 3.981 35 2.70 0.43 Substance Abuse 3.981 35 2.4 11.373 0.02 Professional and Other State Plan Services 3.817.05.44 94.965 66.490.1 512.48 66.73 Preventive Medicine 1.28.399 0.01 62.248 4.332.1 1.98 7.97.99 \$2.66 Preventive Medicine 1.28.39 0.014 10.22.6 0.27.67 1.75.92 4.61 6.7.44 0.63 Preventive Medicine 1.28.3		\$4,770,566	3,115	19,250	216.8	1,339.7	\$1,531.48	\$27.67
Surger 1.177.986 4.514 314.1 260.986 6.83 Radiology 1.135.575 12.411 883.7 91.51 6.59 Pathology 189.046 33.330 2.319.6 5.67 1.10 Cardiovascular 165.541 2.472 172.0 66.89 0.96 Protestional 428.044 3.810 2.239.6 2.81 2.389 Subtaince Abuse 2.309 139.57 7.24 172.0 66.89 0.96 Other 2.309 19.77 9.768 2.88 10.85 0.66.458.1 2.86 2.43 11.37.0 0.84 Other 2.309.062 0.64.49.55 66.458.1 2.86 7.80 9.85 1.87.26 2.86 2.86 2.86 7.80 9.85 1.85.26 2.86 1.80 1.85.27 1.69.24 4.64 0.75 1.83.27 4.83 1.85.26 2.85.78 1.789.2 4.64 6.78 1.789.2 4.64 6.78 1.789.2 4.64<								
Radiology 1.35.676 12.411 88.37 91.51 6.59 Phalnolgy 189.048 33.30 2.319.6 5.67 110 Pharmacy 2.010.992 774.956 49.756.6 2.81 11.65 Cardiovascular 165.341 2.472 17.20 66.689 0.98 PYIOTIST 428.049 38.109 2.662.2 11.23 2.48 Substance Abuse 3.981 35 2.4 113.73 0.02 Other 2.903.062 19.477 9.708.8 20.81 16.84 Professional and Othe State Plan Services 58.6759 01 62.7 \$107.93 \$742 Professional sing Ucenter 138.0015 62.246 4.332.1 \$107.39 \$745 Presentive Microine 128.593 5.742 39.6 2.46 1.35.00 0.00 PT/OT/NT 109.152 14.4665 1.020.6 7.44 0.63 0.44 1.10 0.64 7.66 0.00 0.00 0.00 0				,				
Pathology 189,049 33,330 2,319.6 5.67 1.10 Cardiovascular 165,541 2,472 172.0 66.89 0.96 PriOTIST 428,049 38,109 2,265.22 11.23 2,48 Psychiatric 74,674 1,134 99.8 52.07 0.43 Substance Abuse 3,991 35 2,24 113.73 0.002 Other 2,003,062 139,477 9,706.8 20.81 16.84 Professional and Other State Plan Services Ambulatory Surgery Center 1,500,018 62.248 4,332.1 21.69 7.83 Ambulatory Surgery Center 1,300,018 62.248 4,332.1 21.69 7.83 Office Visits 1,300,018 62.248 4,332.1 21.69 7.84 Maternity 22.571 14.44 10.0 165.80 0.11 Carlider Midvile 11.05.22 14.555 1.029.6 7.44 0.83 Proteintical Pathology 1.365.22 14.555 1.029.4		1,177,986		4,514			260.96	6.83
Pharmacy Cardiovascular 2.010.992 714.956 49.76.6 2.81 11.66 Cardiovascular 165.341 2.472 172.0 66.89 0.96 Pri/OTNST 428.049 38.109 2.852.2 11.23 2.48 Substance Abuse 3.981 35 2.4 113.73 0.02 Other 2.903.062 139.477 -9.706.8 2.081 16.84 Professional and Other State Plan Services 4 964.955 66.459.1 \$12.28 \$87.789 Professional and Other State Plan Services 4 99.455 66.459.1 \$12.28 \$87.789 Ambulatory Surgery Center 139.677 14.44 10.0 163.60 0.14 Certified Midwife 112 5 0.3 22.50 0.00 PT/OT/ST 149.665 1.020.6 7.44 0.63 1.11 2.4 Home Health and Private Duty Nursing 1.735.141 145.539 10.114.7 11.94 10.06 Arbuiance 93.295 84.035	Radiology	1,135,676				863.7		6.59
Cardiovascular 165,341 2,472 172,0 66,89 0.96 PT/OT/ST 428,049 38,109 2,265,2 11,23 2,48 Psychiatric 7,4674 1,434 99,8 52,07 0,43 Substance Abuse 3,981 35,5 2,4 11,37 0,02 Other 2,003,062 139,477 9,706,8 20,81 168,44 Professional and Other State Plan Services 596,759 901 62,7 510,7,39 80,56 Ambulatory Surgery Center 128,293 5,742 399,6 2,246 0,75 Maternity 23,571 144 100 183,56 0,14 Certified Midwle 112 45 0,32 2,00 0,00 Pri/DT/ST 108,152 14,665 1,020,6 7,44 0,63 Psychiatric and Substance Abuse 1,163,25 2,272 1,759,2 46,14 0,74 Redinkey and Pathology 1,462,25 2,426 3,543,31 11,10 6,41	Pathology	189,048		33,330		2,319.6	5.67	1.10
PI/OT/SIT 428,049 38,109 2.65.22 11.23 2.48 Substance Abuse 3.981 35 2.4 113.73 0.02 Other 2.003,062 139,477 9.706.8 2.081 15.8 Professional and Other State Plan Services 954,955 66.459.1 \$11.22 \$87.789 Ambulatory Surgery Center 398,759 901 62.7 \$107.739 \$0.56 Office Visits 1.350,018 62.248 4.332.1 21.60 7.83 Preventive Medicine 122,939 5,742 399.6 22.46 0.75 Matemity 23.571 14.44 10.0 103.60 0.14 Certified Midwife 112 5 0.3 22.50 0.00 Pi/OT/ST 109,152 14.665 1.020.6 7.44 0.63 Aduiology and Pathology 1735.141 145,339 10.114.7 11.94 10.06 Ambulance 215.863 2.099 146.1 102.24 1.25 Op	Pharmacy	2,010,992		714,956		49,756.6	2.81	11.66
Psychiatric 74 674 1.434 99.8 52.07 0.43 Substance Abuse 3.981 35 2.4 113.37 0.02 Cher 2.903.062 139.477 9.706.8 20.81 16.84 Professional and Other State Plan Services 139.477 9.54.955 66.4691 \$12.26 567.89 Protessional and Other State Plan Services 139.50.018 62.248 4.332.1 21.69 7.83 Mulatory Surgery Center 128.593 5.742 399.6 22.46 0.73 Matemity 23.571 144 10.00 163.26 0.04 Proventive Medicine 112 5 0.3 22.50 0.00 PT/OTST 109.152 14.665 1.020.6 7.44 0.63 Psychiatric and Substance Abuse 11.66.25 25.278 1.759.2 46.14 6.76 Radiology and Pathology 473.236 441.845 2.912.2 11.31 2.74 Abore Heath And Provate Duty Nursing 1.759.14 14.533	Cardiovascular	165,341		2,472		172.0	66.89	0.96
Substance Abuse 3.981 3.5 2.4 113.73 0.02 Other 2.903.060 138.477 9.706.8 20.01 168.44 Professional and Other State Plan Services 954.955 666.469.1 \$12.26 \$67.89 Ambulatory Surgery Center \$96.759 901 62.7 \$107.39 \$05.56 Office Visits 1.350.018 62.248 4.332.1 21.49 7.83 Preventive Medicine 128.939 5.742 399.6 22.46 0.75 Naternity 2.35.71 1.44 10.0 163.60 0.14 Centified Midwife 112 5 0.3 22.50 0.00 PtiOT/ST 109.152 14.655 2.978.2 11.31 2.74 Home Health and Private Duty Nursing 1.735.141 145.339 10.114.7 11.94 10.06 Non-Emergency Transportation 5.356.961 2.244.184 15.601.9 2.300 31.07 Optiot Treatment Program 712.027 69.663 4.948.1 10.2	PT/OT/ST	428,049		38,109		2,652.2	11.23	2.48
Other 2.003.062 139.477 9.706.8 20.81 16.84 Professional and Other State Plan Services S67.9 901 62.7 \$107.39 S67.89 Ambulatory Surgery Center \$86.759 901 62.7 \$107.39 S05.6 Office Visits 1.350.018 62.248 4.332.1 21.69 7.83 Preventive Medicine 1.28.571 1.44 10.0 163.60 0.14 Certified Midwife 1.12 5 0.3 22.250 0.00 Pro/TIST 109.152 14.665 1.020.6 7.44 0.63 Psychiatric and Substance Abuse 1.166.25 25.278 1.759.2 46.14 6.76 Radiodya and Pathology 473.236 44.1845 2.912.2 11.31 2.74 Abore Health and Prvate Duty Nursing 1.753.41 14.533 10.114 11.06 46.1 102.24 4.125 Adult Medical Day Care 301.360 13.948 970.7 21.61 10.25 Petoral Quarine 3.	Psychiatric	74,674		1,434		99.8	52.07	0.43
Other 2.003.062 139.477 9.706.8 20.81 16.84 Professional and Other State Plan Services S67.9 901 62.7 \$107.39 S67.89 Ambulatory Surgery Center \$86.759 901 62.7 \$107.39 S05.6 Office Visits 1.350.018 62.248 4.332.1 21.69 7.83 Preventive Medicine 1.28.571 1.44 10.0 163.60 0.14 Certified Midwife 1.12 5 0.3 22.250 0.00 Pro/TIST 109.152 14.665 1.020.6 7.44 0.63 Psychiatric and Substance Abuse 1.166.25 25.278 1.759.2 46.14 6.76 Radiodya and Pathology 473.236 44.1845 2.912.2 11.31 2.74 Abore Health and Prvate Duty Nursing 1.753.41 14.533 10.114 11.06 46.1 102.24 4.125 Adult Medical Day Care 301.360 13.948 970.7 21.61 10.25 Petoral Quarine 3.	Substance Abuse	3.981		35		2.4	113.73	0.02
S11.705.442 954.955 66.459.1 \$12.26 \$67.89 Ambulatory Surgery Center \$96,759 901 62.7 \$107.39 \$50.5 Office Visits 1.380.018 62.248 4.332.1 21.69 7.83 Preventive Medicine 128.939 5.742 399.6 22.46 0.75 Matemity 23.571 144 100 163.60 0.14 Certified Midwife 112 5 0.3 22.80 0.00 PTOORSTON 109.152 14.665 1.020.6 7.44 0.63 Paychiatric and Subtance Abuse 1.166.225 25.278 1.759.2 46.14 6.76 Ambulance 1.202.7 26.633 10.114.7 11.34 10.64 Non-Emergenery Transportation 5.36.861 2.244.14 15.601.9 23.90 31.07 Opioid Treatment Program 71.59.63 26.693 4.494.1 0.22 4.13 Prescription Drugs 27.39.026 2.774.560 62.33.9 5.40 5.86.80								
Professional and Other State Plan Services Ambulatory Surgery Center Sofe,759 901 62.74 \$107,39 \$50,56 Office Visits 1,350,018 62,248 4,332,1 21,69 7,83 Preventive Medicine 128,393 5,742 396 22,46 0,75 Maternity 23,571 144 10.0 163,86 0,14 Certified Midwife 112 5 0,3 22,60 0,00 PT/OT/ST 109,152 14,665 1,020,6 7,752 46,14 6,76 Radiology and Pathology 473,236 41,845 2,912,2 11,31 2,74 Home Health and Private Duty Nursing 1,735,141 145,339 10,114,7 11,94 10,06 Non-Emergency Transportation 5,366,061 224,184 15,601,9 23,90 31,07 Personal Care 4,841,842 895,680 62,333,9 5,41 28,08 Aptiled Behavioral Analysis 53,385 3,486 242,6 15,31 0,31 Durable Medicial Equipment								
Office Visits 1350.018 62.248 4.332.1 21.69 7.83 Preventive Medicine 128.939 5.742 399.6 22.46 0.75 Maternity 23.571 1.44 10.0 163.60 0.14 Certified Midwlie 112 5 0.3 22.56 0.00 Procentive Medicine 109.152 14.655 1.020.6 7.44 0.63 Psychiatric and Substance Abuse 1.166.225 25.278 1.759.2 46.14 6.76 Anabulance 932.935 0.44.035 5.648.3 11.10 5.41 Non-Emergency Transportation 5.356.961 22.41.84 15.601.9 23.90 31.07 Opicid Treatment Program 712.027 69.663 4.848.1 10.22.4 11.25 Auti Medicial Equipment 2.739.026 2.774.540 193.901.3 0.99 13.61 1.75 Durable Medicial Equipment 2.739.026 2.774.540 193.901.3 0.99 15.86 Applied Behavioral Analysis 5.385	Professional and Other State Plan Services			,		· · ·		
Office Visits 1350.018 62.248 4.332.1 21.69 7.83 Preventive Medicine 128.939 5.742 399.6 22.46 0.75 Maternity 23.571 1.44 10.0 163.60 0.14 Certified Midwlie 112 5 0.3 22.56 0.00 Procentive Medicine 109.152 14.655 1.020.6 7.44 0.63 Psychiatric and Substance Abuse 1.166.225 25.278 1.759.2 46.14 6.76 Anabulance 932.935 0.44.035 5.648.3 11.10 5.41 Non-Emergency Transportation 5.356.961 22.41.84 15.601.9 23.90 31.07 Opicid Treatment Program 712.027 69.663 4.848.1 10.22.4 11.25 Auti Medicial Equipment 2.739.026 2.774.540 193.901.3 0.99 13.61 1.75 Durable Medicial Equipment 2.739.026 2.774.540 193.901.3 0.99 15.86 Applied Behavioral Analysis 5.385	Ambulatory Surgery Center	\$96,759		901		62.7	\$107.39	\$0.56
Preventive Medicine 128,939 5,742 399.6 22.46 0.75 Maternity 23,5671 144 10.0 163.66 0.14 Certified Midwife 112 5 0.3 22.50 0.00 PT/OT/ST 109,152 14.665 1.020.6 7.44 0.63 Psychiatric and Substance Abuse 1.166,225 25.278 1.759.2 11.31 2.74 Amore Healt and Private Duty Nursing 1.73,3141 145,339 10.114.7 11.94 10.06 Ambulance 932,935 84,035 5,848.3 11.10 5.41 Non-Emergency Transportation 5,556,961 22.41,84 15.601.9 23.90 31.07 Opioid Treatment Program 712.027 69,663 4,848.1 10.22 4.13 Federally Qualified and Rural Health Clinics 215,863 2.099 146.1 120.24 125 Adut Medical Day Care 3,3466 242.6 15.31 0.31 0.99 15.88 Applied Behavioral Analysis 5.33.85 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>								
Matemity 23,571 144 10.0 163.60 0.14 Certified Midvife 112 5 0.3 22.50 0.00 PT/OT/ST 109.152 14.665 1.020.6 7.44 0.63 Psychiatric and Substance Abuse 1.166.225 25.278 1.759.2 46.14 6.76 Radiology and Pathology 473.236 44.1845 2.912.2 11.31 2.74 Home Health and Private Duty Nursing 1.735.141 145.339 10.114.7 11.94 10.06 Ambulance 932.935 84.035 5.548.43 11.10 5.41 Non-Emergency Transportation 5.356.961 224.144 15.601.9 23.90 31.07 Prederally Qualified and Rural Health Clinics 215.863 2.099 146.61 102.84 125.86 Durable Medical Equipment 2.739.026 2.774.540 193.091.3 0.99 15.88 Drabe Medical Equipment 2.739.020 2.774.540 193.091.3 0.99 15.88 Other 3.666.651				,				
Certified Midwife 112 5 0.3 22.50 0.00 PT/OT/ST 109.152 14.665 1.020.6 7.44 0.63 Psychiatric and Substance Abuse 1.166.225 25.278 1.759.2 46.14 6.76 Radiology and Pathology 473.236 41.845 2.912.2 11.31 2.74 Ambulance 932.935 84.035 5.848.3 11.10 5.41 Ambulance 172.027 66.663 4.848.1 10.22 4.13 Personal Care 301.360 13.948 62.70.7 21.61 1.75 Personal Care 4.841.842 896.660 62.333.9 5.44 28.20 Durable Medical Equipment 2.739.026 2.774.540 193.091.3 0.99 15.88 Applied Behavioral Analysis 53.385 3.446 24.26 15.31 0.31 Other 3.666.651 603.127 41.974.0 6.08 21.26 Seciptis \$158.794 49.207 3.424.5 \$3.23 \$0.9		· · ·						
PT/OT/ST 109,152 14,665 1,020.6 7.44 0.63 Psychiatric and Substance Abuse 1,166.225 25,273 1,759.2 46.14 6.76 Radiology and Pathology 473,236 14,845 2,912.2 11.31 2.74 Home Health and Private Duty Nursing 1,735,141 145,333 10,114.7 11.94 100.06 Non-Emergency Transportation 5,366,961 224,184 15,601.9 23.90 31.07 Opiol Treatment Program 712,027 68,663 4,848,1 102.2 4.13 Federally Qualified and Rural Health Clinics 215,863 2,099 146.1 102.84 1,25 Audit Medical Day Care 301,360 13,348 970.7 21.61 1.75 Personal Care 4,841,842 898,680 62,333.9 5.41 28.08 Durable Medical Equipment 2,739,026 2,774,540 139,061.3 0.99 15.86 Other 3,366,651 60.27 4,1974.0 6.06 21.26 13.86 Other	,							
Psychiatric and Substance Abuse 1.166.225 25.278 1.759.2 46.14 6.76 Radiology and Pathology 473.236 41.845 2.912.2 11.31 2.74 Home Health and Private Duly Nursing 1.735,141 145,339 10.114.7 11.94 10.06 Ambulance 932,935 84.035 5.848.3 11.10 5.41 Opioid Treatment Program 712,027 66.663 4.848.1 10.22 4.13 Pederal/Outilifed and Rural Health Clinics 215,863 2.099 146.1 102.24 4.125 Adult Medical Day Care 301,360 13.948 970.7 21.61 1.75 Personal Care 4,841,842 895,680 62.333.9 5.41 28.08 Durable Medical Equipment 2.739,026 2.774,540 193,091.3 0.09 15.88 Applied Behavioral Analysis 53.385 3.486 242.6 15.31 0.31 Other 3.0424.5 \$3.23 \$0.92 1.748 121.7 12.92 0.13		1 1		-				
Fadiology and Pathology 473.236 41.845 2.912.2 11.31 2.74 Home Health and Private Duty Nursing 1,735.141 145.339 10,114.7 11.94 10.06 Ambulance 932.935 84.035 5.848.3 11.10 5.41 10.06 Non-Emergency Transportation 5.356.961 224.184 15.601.9 23.90 31.07 Opicid Treatment Program 712.027 68.663 4.848.1 10.22 4.13 Federally Qualified and Rural Health Clinics 215.863 2.099 146.1 102.84 1.25 Adult Medical Day Care 301.360 23.79 24.16 15.31 0.31 Durable Medical Equipment 2.739.026 2.774.540 193.091.3 0.99 15.88 Applied Behavioral Analysis 53.385 3.486 242.6 15.31 0.31 Other 3.666.651 603.127 41.974.0 6.08 21.26 Single-Source Brand 22.992 1,748 121.7 12.92 0.31 Single-Source		· · ·				,		
Home Health and Private Duty Nursing 1,735,141 145,339 10,114.7 11.94 10.06 Ambulance 932,935 84,035 5,848.3 11.10 5.41 Opioid Treatment Program 712,027 69,663 4,848.1 10.22 4.13 Adult Medical Day Care 301,360 13,948 970.7 21.61 1.75 Pedrally Qualified and Rural Health Clinics 27,39,026 2,774,540 193,091.3 0.99 15.88 Applied Behavioral Analysis 53,385 3,486 242.6 15.31 0.31 Other 3,666,651 603,127 41,974.0 60.8 212.6 Prescription Drugs	5							
Ambulance 932.935 84.035 5.848.3 11.10 5.41 Non-Emergency Transportation 5.366.961 224.184 15.601.9 23.90 31.07 Opioid Treatment Program 712.027 69.663 4.848.1 10.22 4.13 Federally Qualified and Rural Health Clinics 215.863 2.099 146.1 102.84 1.25 Adult Medical Day Care 301.360 13.948 970.7 21.61 1.75 Personal Care 4.841.842 895.680 62.333.9 5.41 28.08 Durable Medical Equipment 2.739.026 2.774.540 193.091.3 0.99 15.88 Other 3.666.651 603.127 41.974.0 6.08 21.26 Generic Scripts \$158.794 49.207 3.424.5 \$3.23 \$0.92 Specialty 0 0 0.00 0.00 0.00 0.00 Specialty 0 0 0.00 0.00 0.00 0.00 0.00 0.00 Other \$195.						,		
Non-Emergency Transportation 5.366.961 224.184 15.601.9 23.90 31.07 Opioid Treatment Program 712.027 69.663 4.848.1 10.22 4.13 Federally Qualified and Rural Health Clinics 215.863 2.099 1416.1 102.84 1.25 Adutt Medical Day Care 301.360 13.948 970.7 21.61 1.75 Personal Care 4.841.842 895.680 62.333.9 5.41 28.08 Durable Medical Equipment 2.739.026 2.774.540 193.091.3 0.99 15.88 Applied Behavioral Analysis 53.385 3.486 242.6 15.31 0.31 Other 3.666.651 603.127 41.974.0 60.08 21.26 Prescription Drugs Generic Scripts \$158.794 49.207 3.424.5 \$3.23 \$0.92 Single-Source Brand 22.592 1,748 121.7 12.92 0.13 Multi-Source Mand 1.242 864 111.27 0.08 Specialty 0 0 </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>								
Opioid Treatment Program 712.027 69.663 4.848.1 10.22 4.13 Federally Qualified and Rural Health Clinics 215.863 2.099 146.1 102.24 1.25 Adult Medical Day Care 301.360 13.948 970.7 21.61 1.75 Personal Care 4.841.842 895.680 62.333.9 5.41 28.08 Durable Medical Equipment 2.739.026 2.774.540 193.091.3 0.99 15.88 Applied Behavioral Analysis 53.385 3.486 242.6 15.31 0.31 Other 3.666.651 603.127 41.974.0 6.08 21.26 Fescription Drugs								
Federally Qualified and Rural Health Clinics 215,863 2,099 146.1 102,84 1.25 Adult Medical Day Care 301,360 13,948 970.7 21.61 1.75 Personal Care 4,841,842 895,680 62,333 5.41 28.05 Durable Medical Equipment 2,739,026 2,774,540 193,091,3 0.99 15.88 Applied Behavioral Analysis 53,385 3,466 242,6 15.31 0.31 Other 3,666,651 603,127 41,974.0 6.08 21.26 Prescription Drugs 2 4,966,929 345,668.4 \$4.81 \$138,63 Generic Scripts \$158,794 49,207 3,424.5 \$3.23 \$0.92 Single-Source Brand 22,592 1,748 121.7 12.92 0.13 Multi-Source Brand 13,994 1,242 86.4 11.27 0.00 Specialty 0 0 0.0 0.00 0.00 0.00 Gremer Scripts \$199,859 52,276 3,63				,		,		
Adult Medical Day Care 301,360 13,948 970.7 21.61 1.75 Personal Care 4,841,842 895,680 62,333.9 5.41 28.08 Applied Behavioral Analysis 53,385 3,486 242.6 15.31 0.31 Other 3,666,651 603,127 41,974.0 6.08 21.26 Prescription Drugs \$22,903,202 4,966,929 345,668.4 \$38.83 \$3,92 Generic Scripts \$158,794 49,207 3,424.5 \$3.23 \$0.92 Single-Source Brand 22,592 1,748 121.7 12.92 0.13 Multi-Source Brand 13,994 1,242 86.4 11.27 0.08 Specialty 0 0 0 0.00 0.00 0.00 Other 480 79 5.5 6.08 0.00 0.00 Cher 480 79 5.5 6.08 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00								
Personal Care 4,841,842 895,680 62,333.9 5.41 28.08 Durable Medical Equipment 2,739,026 2,774,540 1133,091.3 0.99 15.88 Applied Behavioral Analysis 53,385 3,486 242.6 15.31 0.31 Other 3,666,651 603,127 41,974.0 6.08 21.26 Prescription Drugs \$158,794 49,207 3.424.5 \$3.23 \$0.92 Generic Scripts \$158,794 49,207 3.424.5 \$3.23 \$0.92 Single-Source Brand 13,994 1,242 864 11.27 0.08 Specialty 0 0 0 0.00 0.00 0.00 Other 480 79 5.5 6.08 0.000 0.00 Other \$195,859 52,276 3.638.1 \$3.75 \$1.14 Case Management \$90,459 247 17.2 \$366,23 \$0.52 Long Term Support Service 170,658 3.748 260.8 45.53								
Durable Medical Equipment 2,739,026 2,774,540 193,091,3 0.99 15.88 Applied Behavioral Analysis 53,385 3,486 242.6 15.31 0.31 Other \$23,903,202 4,966,929 345,668.4 \$4.81 \$138.63 Prescription Drugs \$158,794 49,207 3,424.5 \$3.23 \$0.92 Generic Scripts \$158,794 49,207 3,424.5 \$3.23 \$0.92 Single-Source Brand 22,592 1,748 121.7 12.92 0.13 Multi-Source Brand 13,994 1,242 86.4 11.27 0.00 Specialty 0 0 0 0.00 0.00 0.00 Other 480 79 5.5 6.08 0.00 Community Mental Health Center \$90,459 247 17.2 \$366.23 \$0.52 Cong Term Support Service 170,658 3.748 260.8 45.53 0.99 Parital Hospital 2,125 39 2.7 54.50								
Applied Behavioral Analysis 53,385 3,486 242.6 15.31 0.31 Other 3,666,661 603,127 41,974.0 6.08 21.26 Prescription Drugs 4,966,929 345,668.4 \$4.81 \$138.63 Generic Scripts \$158,794 49,207 3,424.5 \$3.23 \$0.92 Single-Source Brand 22,592 1,748 121.7 12.92 0.13 Specialty 0 0 0.00 0.00 0.00 0.00 Hepatitis C 0 0 0.0 0.00 0.00 0.00 Cher 480 79 5.5 6.08 0.00 0.00 Other 480 79 5.5 6.08 0.00 0.00 Community Mental Health Center Case Management \$90,459 247 17.2 \$366.23 \$0.52 Long Term Support Service 170,658 3.748 260.8 45.53 0.99 Partial Hospital 2,125 39 2.7 54				,				
Other 3,666,651 603,127 41,974.0 6.08 21.26 Vescription Drugs 345,668.4 \$4.81 \$138,63 Prescription Drugs 49,007 3424,5 \$3.23 \$0.92 Single-Source Brand 22,592 1,748 121.7 12.92 0.13 Multi-Source Brand 13,994 1,242 864.4 11.27 0.08 Specially 0 0 0 0.00 0.00 0.00 Hepatitis C 0 0 0 0.00 0.00 0.00 0.00 Other 480 79 5.5 6.08 0.00 0.00 Other 490,459 247 17.2 \$366.23 \$0.52 0.01 Community Mental Health Center 90,459 247 17.2 \$366.23 \$0.52 0.01 Case Management \$90,459 247 17.2 \$366.23 \$0.52 0.01 Long Term Support Service 170,658 3.748 260.8 45.53								
\$23,903,202 4,966,929 345,668.4 \$4.81 \$138.63 Prescription Drugs \$158,794 49,207 3,424.5 \$3.23 \$0.92 Single-Source Brand 22,592 1,748 121.7 12.92 0.13 Multi-Source Brand 13,994 1,242 86.4 11.27 0.08 Specialty 0 0 0.00 0.00 0.00 0.00 Multi-Source Brand 13,994 1,242 86.4 11.27 0.08 Specialty 0 0 0.00 0.00 0.00 0.00 Other 480 79 5.5 6.08 0.00 Community Mental Health Center Case Management \$195,859 247 17.2 \$366.23 \$0.52 Long Term Support Service 170,658 3,748 260.8 45.53 0.99 Partial Hospital 2,125 39 2.7 54.50 0.01 Psychotherapy 103,373 3,044 211.8 33.66 0.60				,				
Prescription Drugs Generic Scripts \$158,794 49,207 3,424.5 \$3.23 \$0.92 Single-Source Brand 22,592 1,748 121.7 12.92 0.13 Multi-Source Brand 13,994 1,242 86.4 11.27 0.08 Specialty 0 0 0 0.00 0.00 0.00 Hepattis C 0 0 0.0 0.00 0.00 0.00 Other 480 79 5.5 6.08 0.00 Community Mental Health Center 5 3,748 260.8 45.53 0.99 Long Term Support Service 170,658 3,748 260.8 45.53 0.99 Partial Hospital 2,125 39 2.7 54.50 0.01 Psychotherapy 103,373 3,044 211.8 33.96 0.60 Evidence Based Practice 2,200 123 8.6 17.89 0.01 Medication Management 7,222 249 17.3 29.00 0.04	Other			,				
Generic Scripts \$158,794 49,207 3,424.5 \$3.23 \$0.92 Single-Source Brand 22,592 1,748 121.7 12.92 0.13 Multi-Source Brand 13,994 1,242 86.4 112.7 12.92 0.03 Specialty 0 0 0.01 0.55 0.00 1.14		\$23,903,202		4,966,929		345,668.4	\$4.81	\$138.63
Single-Source Brand 22,592 1,748 121.7 12.92 0.13 Multi-Source Brand 13,994 1,242 86.4 11.27 0.08 Specialty 0 0 0.0 0.00 0.00 0.00 Hepatitis C 0 0 0.0 0.01 Startine for themane theatheatheathean theathean theathean theatheathean theathean		A 1 50 50 1		10.007			* ****	* ****
Multi-Source Brand 13,994 1,242 86.4 11.27 0.08 Specialty 0 0 0 0.0 0.01 10.373 3.748 260.8 45.53 0.99 9 103,373 3.044 211.8 33.96 0.60 0.0	•			,		,		
Specialty 0 0 0 0 0.0 0.00 0.00 Hepatitis C 0 0 0 0.0 0.01 10.373 3.748 260.8 45.53 0.99 9 103.373 3.044 211.8 33.96 0.61 11 23.90 0.01 Medication Management 7.222 249 17.3 29.00								
Hepatitis C 0 0 0 0.0 0.00 0.00 Other 480 79 5.5 6.08 0.00 0.00 Community Mental Health Center 5 6.08 0.00 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01								
Other 480 79 5.5 6.08 0.00 \$195,859 52,276 3,638.1 \$3.75 \$1.14 Case Management \$90,459 247 17.2 \$366.23 \$0.52 Long Term Support Service 170,658 3,748 260.8 45.53 0.99 Partial Hospital 2,125 39 2.7 54.50 0.01 Psychotherapy 103,373 3,044 211.8 33.96 0.60 Evidence Based Practice 2,200 123 8.6 17.89 0.01 Medication Management 7,222 249 17.3 29.00 0.04 Emergency Service 24/7 376 16 1.1 23.50 0.00 APRTP 45,167 76 5.3 594.31 0.26 Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5	Specialty	0		0		0.0		
\$195,859 52,276 3,638.1 \$3.75 \$1.14 Community Mental Health Center \$90,459 247 17.2 \$366.23 \$0.52 Long Term Support Service 170,658 3,748 260.8 45.53 0.99 Partial Hospital 2,125 39 2.7 54.50 0.01 Psychotherapy 103,373 3,044 211.8 33.96 0.60 Evidence Based Practice 2,200 123 8.6 17.89 0.01 Medication Management 7,222 249 17.3 29.00 0.04 Emergency Service 24/7 376 16 1.1 23.50 0.00 APRTP 45,167 76 5.3 594.31 0.26 Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71	Hepatitis C			-				
Community Mental Health Center \$90,459 247 17.2 \$366.23 \$0.52 Long Term Support Service 170,658 3,748 260.8 45.53 0.99 Partial Hospital 2,125 39 2.7 54.50 0.01 Psychotherapy 103,373 3,044 211.8 33.96 0.60 Evidence Based Practice 2,200 123 8.6 17.89 0.01 Medication Management 7,222 249 17.3 29.00 0.04 Emergency Service 24/7 376 16 1.1 23.50 0.00 APRTP 45,167 76 5.3 594.31 0.26 Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71	Other							
Case Management \$90,459 247 17.2 \$366.23 \$0.52 Long Term Support Service 170,658 3,748 260.8 45.53 0.99 Partial Hospital 2,125 39 2.7 54.50 0.01 Psychotherapy 103,373 3,044 211.8 33.96 0.60 Evidence Based Practice 2,200 123 8.6 17.89 0.01 Medication Management 7,222 249 17.3 29.00 0.04 Emergency Service 24/7 376 16 1.1 23.50 0.00 APRTP 45,167 76 5.3 594.31 0.26 Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$30,279 13,256 922.5 \$48.23 \$3.71		\$195,859		52,276		3,638.1	\$3.75	\$1.14
Long Term Support Service 170,658 3,748 260.8 45.53 0.99 Partial Hospital 2,125 39 2.7 54.50 0.01 Psychotherapy 103,373 3,044 211.8 33.96 0.60 Evidence Based Practice 2,200 123 8.6 17.89 0.01 Medication Management 7,222 249 17.3 29.00 0.04 Emergency Service 24/7 376 16 1.1 23.50 0.00 APRTP 45,167 76 5.3 594.31 0.26 Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71	-							
Partial Hospital 2,125 39 2.7 54.50 0.01 Psychotherapy 103,373 3,044 211.8 33.96 0.60 Evidence Based Practice 2,200 123 8.6 17.89 0.01 Medication Management 7,222 249 17.3 29.00 0.04 Emergency Service 24/7 376 16 1.1 23.50 0.00 APRTP 45,167 76 5.3 594.31 0.26 Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71	Case Management	\$90,459		247		17.2	\$366.23	\$0.52
Psychotherapy 103,373 3,044 211.8 33.96 0.60 Evidence Based Practice 2,200 123 8.6 17.89 0.01 Medication Management 7,222 249 17.3 29.00 0.04 Emergency Service 24/7 376 16 1.1 23.50 0.00 APRTP 45,167 76 5.3 594.31 0.26 Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71								
Evidence Based Practice 2,200 123 8.6 17.89 0.01 Medication Management 7,222 249 17.3 29.00 0.04 Emergency Service 24/7 376 16 1.1 23.50 0.00 APRTP 45,167 76 5.3 594.31 0.26 Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71	•							
Medication Management 7,222 249 17.3 29.00 0.04 Emergency Service 24/7 376 16 1.1 23.50 0.00 APRTP 45,167 76 5.3 594.31 0.26 Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71								
Emergency Service 24/7 376 16 1.1 23.50 0.00 APRTP 45,167 76 5.3 594.31 0.26 Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71						8.6	17.89	
APRTP 45,167 76 5.3 594.31 0.26 Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71	Medication Management	7,222		249		17.3	29.00	0.04
Supported Employment Services 4,394 162 11.3 27.13 0.03 Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71							23.50	
Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71	APRTP	45,167		76		5.3	594.31	0.26
Other 213,304 5,552 386.4 38.42 1.24 \$639,279 13,256 922.5 \$48.23 \$3.71	Supported Employment Services	4,394		162		11.3	27.13	0.03
\$639,279 13,256 922.5 \$48.23 \$3.71				5,552				
	All Services	\$41,214,348	3,115		216.8			

Eligibility Category: Newborn Kick Payment

Member Months: 2,767

							Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient	Donard	Admito	001110000	1,000	1,000	onargo	0000
Medical	\$5,275,089	1,094	7,771	395.4	2,808.5	\$4,821.84	\$1,906.43
Surgical	297,741	12	225	4.3	81.3	24,811.77	107.60
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00
Well Newborn	991,741	1,448	3,092	523.3	1,117.5	684.90	358.42
Psychiatric	4,009	1	4	0.4	1.4	4,009.32	1.45
Alcohol and Drug Abuse	0	0	0	0.0	0.0	0.00	0.00
Other	0	0	0	0.0	0.0	0.00	0.00
	\$6,568,581	2,555	11,092	923.4	4,008.7	\$2,570.87	\$2,373.90
Hospital Outpatient							
Emergency Room	\$83,277		270		97.6	\$308.43	\$30.10
Surgery	2,325		11		4.0	211.35	0.84
Radiology	13,935		100		36.1	139.35	5.04
Pathology	9,631		1,278		461.9	7.54	3.48
Pharmacy	426		662		239.2	0.64	0.15
Cardiovascular	1,571		19		6.9	82.70	0.57
PT/OT/ST	1,927		29		10.5	66.44	0.70
Psychiatric	0		0		0.0	0.00	0.00
Substance Abuse	0		0		0.0	0.00	0.00
Other	108,556		1,055		381.3	102.90	39.23
	\$221,647		3,424		1,237.4	\$64.73	\$80.10
Professional and Other State Plan Services							
Ambulatory Surgery Center	\$0		0		0.0	\$0.00	\$0.00
Office Visits	101,063		2,113		763.6	47.83	36.52
Preventive Medicine	291,984		6,346		2,293.5	46.01	105.52
Maternity	0		0		0.0	0.00	0.00
Certified Midwife	437		11		4.0	39.70	0.16
PT/OT/ST	1,005		40		14.5	25.12	0.36
Psychiatric and Substance Abuse	0		0		0.0	0.00	0.00
Radiology and Pathology	32,536		1,558		563.1	20.88	11.76
Home Health and Private Duty Nursing	67,334		834		301.4	80.74	24.33
Ambulance	62,079		2,507		906.0	24.76	22.44
Non-Emergency Transportation	2,349		124		44.8	18.94	0.85
Opioid Treatment Program	0		0		0.0	0.00	0.00
Federally Qualified and Rural Health Clinics	239,730		1,412		510.3	169.78	86.64
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	0		0		0.0	0.00	0.00
Durable Medical Equipment	25,234		12,408		4,484.3	2.03	9.12
Applied Behavioral Analysis	0		,0		0.0	0.00	0.00
Other	1,372,085		11,719		4,235.3	117.08	495.87
	\$2,195,835		39,072		14,120.7	\$56.20	\$793.58
Prescription Drugs					, i		
Generic Scripts	\$5,797		547		197.7	\$10.60	\$2.10
Single-Source Brand	1,379		19		6.9	72.60	0.50
Multi-Source Brand	87		10		3.6	8.73	0.03
Specialty	12		2		0.7	5.95	0.00
Hepatitis C	0		0		0.0	0.00	0.00
Other	5		2		0.7	2.59	0.00
	\$7,281		580		209.6	\$12.55	\$2.63
Community Mental Health Center							
Case Management	\$0		0		0.0	\$0.00	\$0.00
Long Term Support Service	0		0		0.0	0.00	0.00
Partial Hospital	0		0		0.0	0.00	0.00
Psychotherapy	0		0		0.0	0.00	0.00
Evidence Based Practice	0		0		0.0	0.00	0.00
Medication Management	0		0		0.0	0.00	0.00
Emergency Service 24/7	0		0		0.0	0.00	0.00
APRTP	0		0		0.0	0.00	0.00
Supported Employment Services	0		0		0.0	0.00	0.00
Other	0		0		0.0	0.00	0.00
-	\$0		0		0.0	\$0.00	\$0.00
All Services	\$8,993,345	2,555	54,168	923.4	19,576.4	\$166.03	\$3,250.21
	+=,000,010	2,000	5.,.50	020.1	. 0,01 0.1	2.00.00	+=,200.21

Eligibility Category: Neonatal Abstinence Syndrome Kick Payment

Member Months: 160

							Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1.000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient	Donard	Adinto	00111000	1,000	1,000	onargo	0000
Medical	\$871,140	165	1,568	1,031.3	9,800.0	\$5,279.64	\$5,444.62
Surgical	0	0	0	0.0	0.0	0.00	0.00
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00
Well Newborn	15,333	2	5	12.5	31.3	7,666.44	95.83
Psychiatric	0	0	0	0.0	0.0	0.00	0.00
Alcohol and Drug Abuse	0	0	0	0.0	0.0	0.00	0.00
Other	0	0	0	0.0	0.0	0.00	0.00
	\$886,473	167	1,573	1,043.8	9,831.3	\$5,308.22	\$5,540.45
Hospital Outpatient	¢0.470		45		02.0	¢004.05	¢04.74
Emergency Room	\$3,478		15		93.8	\$231.85	\$21.74
Surgery	1,441		1		6.3	1,440.61	9.00
Radiology	697 799		5		31.3	139.44	4.36
Pathology			44		275.0	18.17	5.00
Pharmacy	227 66		37 5		231.3 31.3	6.14	1.42
Cardiovascular PT/OT/ST	223		2		12.5	13.16 111.47	0.41 1.39
Psychiatric	223		2		0.0	0.00	0.00
Substance Abuse	0		0		0.0	0.00	0.00
Other	4,804		63		393.8	76.26	30.03
Other	\$11,735		172		1,075.0	\$68.23	\$73.34
Professional and Other State Plan Services	ψ11,700		112		1,070.0	φ00.20	φ/ 0.04
Ambulatory Surgery Center	\$0		0		0.0	\$0.00	\$0.00
Office Visits	7,563		151		943.8	50.08	47.27
Preventive Medicine	21,823		446		2,787.5	48.93	136.40
Maternity	0		0		0.0	0.00	0.00
Certified Midwife	0		0		0.0	0.00	0.00
PT/OT/ST	350		15		93.8	23.34	2.19
Psychiatric and Substance Abuse	0		0		0.0	0.00	0.00
Radiology and Pathology	1,113		86		537.5	12.94	6.95
Home Health and Private Duty Nursing	12,665		145		906.3	87.35	79.16
Ambulance	5,016		148		925.0	33.89	31.35
Non-Emergency Transportation	335		23		143.8	14.55	2.09
Opioid Treatment Program	0		0		0.0	0.00	0.00
Federally Qualified and Rural Health Clinics	16,773		98		612.5	171.15	104.83
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	0		0		0.0	0.00	0.00
Durable Medical Equipment	4,982		2,220		13,875.0	2.24	31.14
Applied Behavioral Analysis	0		0		0.0	0.00	0.00
Other	98,818		1,495		9,343.8	66.10	617.61
	\$169,438		4,827		30,168.8	\$35.10	\$1,058.99
Prescription Drugs					100.0		* = • <i>i</i>
Generic Scripts	\$855		64		400.0	\$13.36	\$5.34
Single-Source Brand	7		1		6.3	7.19	0.04
Multi-Source Brand	0		0		0.0	0.00	0.00
Specialty	0		0		0.0	0.00	0.00
Hepatitis C	0		0		0.0	0.00	0.00
Other	\$862		65		0.0 406.3	0.00 \$13.26	0.00 \$5.39
Community Mental Health Center	φ002		05		400.3	\$13.20	¢0.09
Case Management	\$0		0		0.0	\$0.00	\$0.00
Long Term Support Service			0		0.0	0.00	0.00
Partial Hospital	0		0		0.0	0.00	0.00
Psychotherapy	0		0		0.0	0.00	0.00
Evidence Based Practice	0		0		0.0	0.00	0.00
Medication Management	0		0		0.0	0.00	0.00
Emergency Service 24/7	0		0		0.0	0.00	0.00
APRTP	0		0		0.0	0.00	0.00
Supported Employment Services	0		0		0.0	0.00	0.00
Other	ů o		0		0.0	0.00	0.00
	\$0		0		0.0	\$0.00	\$0.00
All Services	\$1,068,508	167	6,637	1,043.8	41,481.3	\$160.99	\$6,678.17

Eligibility Category: Maternity Kick Payment

Member Months: 3,260

					Por Conito		
P	Total Paid	Total Paid	Total Paid Services	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid
Benefits Hospital Inpatient	Dollars	Admits	Services	1,000	1,000	Charge	Cost
Medical	\$242,479	95	253	29.1	77.6	\$2,552.42	\$74.38
	\$242,479 0	95	253	29.1	0.0	\$2,552.42 0.00	\$74.38 0.00
Surgical Maternity Delivery	5,906,101	2,422	6,396	742.9	1,962.0	2,438.52	1,811.69
Maternity Non-Delivery	70,975	2,422	58	6.4	1,902.0	3,379.75	21.77
Well Newborn	10,975	0	0	0.4	0.0	0.00	0.00
Psychiatric	0	0	0	0.0	0.0	0.00	0.00
Alcohol and Drug Abuse	0	0	0	0.0	0.0	0.00	0.00
Other	0	0	0	0.0	0.0	0.00	0.00
Otilei	\$6,219,555	2,538	6,707	778.5	2,057.4	\$2,450.57	\$1,907.84
Hospital Outpatient	ψ0,219,000	2,000	0,707	110.5	2,007.4	ψ2,430.37	φ1,307.04
Emergency Room	\$48,839		41		12.6	\$1,191.19	\$14.98
Surgery	16,043		14		4.3	1,145.92	4.92
Radiology	7,314		54		16.6	135.44	2.24
Pathology	5,902		544		166.9	10.85	1.81
Pharmacy	9,474		1,571		481.9	6.03	2.91
Cardiovascular	137		2		401.9	68.54	0.04
PT/OT/ST	538		20		6.1	26.92	0.04
Psychiatric	0		20		0.0	0.00	0.00
Substance Abuse	0		0		0.0	0.00	0.00
Other	64,567		1,246		382.2	51.82	19.81
Other	\$152,814		3,492		1,071.2	\$43.76	\$46.88
Professional and Other State Plan Services	ψ152,014		5,452		1,071.2	ψ43.70	φ40.00
Ambulatory Surgery Center	\$0		0		0.0	\$0.00	\$0.00
Office Visits	3,626		64		19.6	\$6.66	1.11
Preventive Medicine	1,645		171		52.5	9.62	0.50
Maternity	1,678,784		3,571		1,095.3	470.18	514.96
Certified Midwife	27,775		65		1,095.5	470.18	8.52
PT/OT/ST	21,115		4		1.2	428.48	0.01
Psychiatric and Substance Abuse	0		0		0.0	0.00	0.00
Radiology and Pathology	7,321		326		100.0	22.46	2.25
Home Health and Private Duty Nursing	701		10		3.1	70.10	0.22
Ambulance	9,738		1,062		325.8	9.17	2.99
Non-Emergency Transportation	0,700		0		0.0	0.00	0.00
Opioid Treatment Program	0		0		0.0	0.00	0.00
Federally Qualified and Rural Health Clinics	8,427		51		15.6	165.24	2.59
Adult Medical Day Care	0,427		0		0.0	0.00	0.00
Personal Care	0		0		0.0	0.00	0.00
Durable Medical Equipment	222		2		0.6	110.77	0.00
Applied Behavioral Analysis	0		0		0.0	0.00	0.00
Other	43,360		2,433		746.3	17.82	13.30
Other	\$1,781,622		7,758		2,379.9	\$229.64	\$546.51
Prescription Drugs	ψ1,701,022		1,130		2,515.5	φ229.04	\$J40.51
Generic Scripts	\$0		0		0.0	\$0.00	\$0.00
Single-Source Brand	φ0 0		0		0.0	0.00	0.00
Multi-Source Brand	0		0		0.0	0.00	0.00
Specialty	0		0		0.0	0.00	0.00
Hepatitis C	0		0		0.0	0.00	0.00
Other	0		0		0.0	0.00	0.00
Other	\$0		0		0.0	\$0.00	\$0.00
Community Mental Health Center	ψυ		0		0.0	ψ0.00	ψ0.00
Case Management	\$0		0		0.0	\$0.00	\$0.00
Long Term Support Service	φ0 0		0		0.0	0.00	0.00
Partial Hospital	0		0		0.0	0.00	0.00
Psychotherapy	0		0		0.0	0.00	0.00
Evidence Based Practice	0		0		0.0	0.00	0.00
Medication Management	0		0		0.0	0.00	0.00
	0		0		0.0	0.00	
Emergency Service 24/7 APRTP	0		0		0.0	0.00	0.00 0.00
Supported Employment Services	0		0		0.0	0.00	0.00
Other	0 \$0		0		0.0	0.00	00.0
All Services	\$0 \$8,153,990	2,538	17,957	778.5	0.0 5,508.4	\$0.00 \$454.07	\$0.00
All SELVICES	φo, 153,990	2,008	17,957	110.5	5,506.4	\$404.U7	\$2,501.22

Eligibility Category: Severe/Persistent Mental Illness - Non-Dual

Member Months: 16,446

							Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1.000	Utilization Per 1.000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient	2011410	71011110		.,	.,	onargo	
Medical	\$831,664	202	901	147.4	657.4	\$4,117.15	\$50.57
Surgical	400,660	57	343	41.6	250.3	7,029.13	24.36
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	14,423	6	19	4.4	13.9	2,403.85	0.88
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	781,999	154	1,261	112.4	920.1	5,077.91	47.55
Alcohol and Drug Abuse	31,204	11	71	8.0	51.8	2,836.72	1.90
Other	0	0	0	0.0	0.0	0.00	0.00
	\$2,059,950	430	2,595	313.8	1,893.5	\$4,790.58	\$125.26
Hospital Outpatient							
Emergency Room	\$1,400,277		2,185		1,594.4	\$640.86	\$85.15
Surgery	294,457		422		307.9	697.76	17.90
Radiology	366,206		1,611		1,175.5	227.32	22.27
Pathology	156,232		12,445		9,080.9	12.55	9.50
Pharmacy	243,001		50,310		36,710.2	4.83	14.78
Cardiovascular	42,654		198		144.5	215.43	2.59
PT/OT/ST	104,357		3,552		2,591.8	29.38	6.35
Psychiatric	30,578		210		153.2	145.61	1.86
Substance Abuse	3,911		37		27.0	105.70	0.24
Other	382,229		14,611		10,661.4	26.16	23.24
	\$3,023,902		85,581		62,446.8	\$35.33	\$183.87
Professional and Other State Plan Services							
Ambulatory Surgery Center	\$34,505		166		121.1	\$207.86	\$2.10
Office Visits	489,430		8,927		6,513.9	54.83	29.76
Preventive Medicine	88,697		2,557		1,865.8	34.69	5.39
Maternity	20,898		112		81.9	186.24	1.27
Certified Midwife	73		3		2.2	24.21	0.00
PT/OT/ST	35,872		1,518		1,107.7	23.63	2.18
Psychiatric and Substance Abuse	360,598		4,391		3,204.0	82.12	21.93
Radiology and Pathology	251,210		10,752		7,845.5	23.36	15.28
Home Health and Private Duty Nursing	230,954		5,120		3,736.0	45.11	14.04
Ambulance	175,540		8,306		6,060.7	21.13	10.67
Non-Emergency Transportation	738,807		43,613		31,823.6	16.94	44.92
Opioid Treatment Program	160,489		15,703		11,458.2	10.22	9.76
Federally Qualified and Rural Health Clinics	329,159		2,316		1,689.9	142.12	20.02
Adult Medical Day Care	4,896		199		145.2	24.60	0.30
Personal Care	4,344		687		501.3	6.32	0.26
Durable Medical Equipment	296,293		92,972		67,839.9	3.19	18.02
Applied Behavioral Analysis	0		0		0.0	0.00	0.00
Other	938,058		74,202		54,143.8	12.64	57.04
	\$4,159,821		271,544		198,140.6	\$15.32	\$252.95
Prescription Drugs							
Generic Scripts	\$1,805,926		90,111		65,752.3	\$20.04	\$109.81
Single-Source Brand	3,616,361		7,479		5,457.3	483.54	219.90
Multi-Source Brand	724,812		1,795		1,309.8	403.79	44.07
Specialty	3,436,227		1,157		844.2	2,969.95	208.95
Hepatitis C	416,980		29		21.2	14,378.62	25.36
Other	104		24		17.5	4.35	0.01
	\$10,000,410		100,595		73,402.2	\$99.41	\$608.09
Community Mental Health Center							
Case Management	\$4,903,156		12,625		9,212.2	\$388.37	\$298.14
Long Term Support Service	4,551,366		162,984		118,926.3	27.93	276.75
Partial Hospital	129,288		1,232		899.0	104.94	7.86
Psychotherapy	1,128,680		20,009		14,600.2	56.41	68.63
Evidence Based Practice	114,525		6,098		4,449.6	18.78	6.96
Medication Management	67,959		2,291		1,671.7	29.66	4.13
Emergency Service 24/7	43,639		1,860		1,357.2	23.46	2.65
	79,801		134		97.8	595.53	4.85
APRTP	10,0011						
	420,219		14,596		10,650.4	28.79	25.55
APRTP			14,596 7,560		10,650.4 5,516.4	28.79 72.71	25.55 33.42
APRTP Supported Employment Services	420,219						

Eligibility Category: Severe/Persistent Mental Illness - Dual

Member Months: 23,082

Total Paid Total Paid Total Paid Total Paid Number Paid Validation Paid Infigure Aurorage Paid Monthly Paid Hospital Inguistri 547,5,642 228 1,708 151.6 88.01 15.28.05 520.01 Materinty Poliviny 10.227 28 10 46.00 42.00 12.80.05 0.00 Materinty Poliviny 2.28 3 10 16 13.80.05 0.00										
Hospital ingation S475,644 222 1,708 151.8 6880.0 51.229.92 20.01 Surgical 101.227 88 810 45.3 42.11 1.32.13 8.39 Well Newborn 2,72 0 0 0.0 0.0 0.0 0.0 0.00										
Medical 5476,644 202 1,708 115.18 688.0 51,262,922 52,001 Maternity Non-Delivery 0 </td <td></td> <td>Dollars</td> <td>Admits</td> <td>Services</td> <td>1,000</td> <td>1,000</td> <td>Charge</td> <td>Cost</td>		Dollars	Admits	Services	1,000	1,000	Charge	Cost		
Surgical 161.227 88 810 45.8 42.11 1.32.13 6.69 Maternity Delivery 2.728 2 3 1.0 1.6 1.36.04 0.00 Maternity Delivery 2.728 2 3 1.0 1.6 1.36.04 0.00 Pap.hation 646.37 1.707 1.98 8.84 1.0.3 1.21.21 0.000 Oher 0 0 0 0 0.00 0.000 0.000 Deptil Object 1 1.21.21.37 569 4.549 2.365.0 52.13.09 52.23.75 Surgley 21.05.08 7.73 3.98.7 2.88.1 7.9.3 Radioby 182.303 2.18 3.32.25 2.28.8 7.9.3 Radioby 2.15.7 3.78 3.78.7 1.21.9 3.2.2.2 Cardowaca'ar 25.174 5.98 3.02.57 1.2.8 3.78.5 1.0.0 1.0.2.2 1.0.6 1.0.2.2 2.0.6 1.0.2.2 2.0.6		A 175 A 11		4 700				* ***		
Main O										
Maternity Non-Delivery 2,728 2 3 1.0 1.8 1.384.04 0.10.0 Psychiatic 544,330 1.70 1,932 88.4 1.004.4 3.21.70 2.23.70 Other Other 5.97.8 7 9.8 8.8 4.94.9 2.23.70 2.3.70 1.14 Other Status 5.90.945 2.3.87 1.92.8 2.3.85.9 2.2.98.50		1 1					· · ·			
Weil Newborn 0 <t< td=""><td></td><td></td><td></td><td>-</td><td></td><td></td><td></td><td></td></t<>				-						
Psychiatric 546.330 170 1.9.22 88.4 1.004.4 3.213.70 2.23.67 Acholand Chug Abue 28.378 17 66 8.8 4.96.9 1.00 0.00		1 1					· · ·			
Alcohol and Dug Abuse 28.378 177 66 8.8 49.9 1.551.66 1.14 Other 0 0 0.00		-	-							
Other 0 0 0 0 0 0 0 0.00 0.00 Hogital Outpatient 5780 4549 2958 2.365 52.3059 552.25 Bergency Nom 5790.946 2.317 1.204.6 \$341.37 \$34.27 Surgery 216.558 7.63 3.86.7 2.33.17 9.38 Dethology 1.32.67 8.33.27.5 2.38.17 9.38 Ordinovacular 25.175 3.78 1.96.6 6.66.0 1.90 ProVINST 7.2.14 5.918 3.076.7 12.19 3.12				,						
S1.212.07 569 4.449 298.8 2.386.0 \$2.130.59 \$352.52 Emergency Room \$700.045 2.317 1.204.6 \$\$341.37 \$\$34.77 Radiology 122.000 2.160.85 763 3.967 2.83.17 9.36 Radiology 182.290 2.169 1.142.7 82.09 7.00 Phathogy 132.633 63.090 3.322.5 2.28 8.66 Phathogy 130.662 63.090 3.322.5 2.98 8.66 Phatholitin 24.428 417 2.18 3.17 9.36 Stoblance Abuse 3.564 22 16.6 112.30 0.16 Other 427.806 2.67.14 13.88.5 16.01 18.33 Professional and Other State Pian Services 31.686 117 66.8 \$161.42 \$0.02 Amulatory Surgery Center 21.650 1.055 54.56.09 8.10.0 \$85.16 Protensite Medicine 21.650 1.055 54.51.0.22 8.33.3										
Hospital Outpatient -	Other									
Emergency Reom \$700,945 2,317 1,204.6 \$33,17 \$34,77 Radiology 12,508 763 39.67 23.37 \$9.36 Radiology 12,239 2,198 1,142.7 82.08 7.90 Pathology 32,233 6,3090 33,322.5 2.98 8.26 Cardiovascular 25,175 378 190.65 6.60 1.09 PT/OTIST 7,2124 5.918 3.076.7 12.19 3.12 Substance Abuse 3.644 3.2 1.66 112.30 0.16 Substance Abuse 3.1065.71 100.03 56.866.9 \$18.36 1.06 Professional and Other Stab Plan Services 31.965.71 100.03 56.866.9 \$18.36 1.07 0.0 0.00		\$1,212,307	569	4,549	295.8	2,365.0	\$2,130.59	\$52.52		
Surgery 216,068 763 996.7 283,17 93.8 Pathology 32,833 6,390 3,322.1 5,11 1,14 Pharmacy 190,652 6,63,068 3,322.5 5,28 8,286 Cardiovascular 25,175 3,76 196.5 66,660 1.09 Psychiatric 24,428 417 216.8 68,858 1.06 Stotstance Abuse 3,564 22 16.6 112.30 0.16 Other 47,2006 26,714 13,888.5 16.01 18.83 Professional and Other State Pian Services 3,184.5 101 13,384.5 16.02 2.03,22 Protessional and Other State Pian Services 3,18,866 1.17 6.0.8 \$16,142 80.30 2.0.22 1.0.85 5.0.642,7 22.38 12.85 12.85 12.85 12.85 12.85 12.85 12.85 12.85 12.85 12.85 12.85 12.85 12.85 12.85 12.85 12.85 12.85 12.85				o o / =			** ****	* • • • - -		
Radiology 182.390 2,198 1,1427 82.88 7.90 Pathology 32.633 6.390 33.222.1 5.11 1.14 Pharmacy 190,562 63.908 33.225.5 2.88 8.26 Cardiovascular 25,175 37.8 1.96.5 66.60 1.09 PT/OT/ST 72,124 5.918 3.076.7 12.19 3.122 Profestorial and Other State Plan Services 3.594 32 16.6 112.30 0.16 Other 427.806 2.87.74 13.885.8 10.05 58.856 31.80.714 109.035 65.866.9 31.80.2 20.82 Profestorial Substance Abuse 51.88.851 13.354 6.942.7 51.88.2 20.82 0.44 Maternity 4.874 41 21.1 120.12 0.21 0.21 Certified Mavife 0 0 0 0.00 0.00 0.00 Proventive Module 0 0 0 0.00 0.00 0.00		1 1 1				· · · · · ·				
Pathologin 33.633 6.390 3.3221 5.11 1.14 Pharmacy 190.562 63.900 3.3225.5 2.88 8.28 Cardiovascular 25.175 3.78 196.5 66.60 1.09 Psychiatric 24.428 4.17 216.8 58.58 1.06 Other 24.700 26.714 13.885.5 16.01 18.55 Professional and Other State Plan Services 31.985.714 109.035 556.685.9 \$18.03 585.16 Ambulatory Surgery Center 28.8,86 1.17 60.8 \$16.1.42 0.9.2 Office Visits 29.8,811 13.354 6.9.42.7 22.38 12.05 Proventive Medicine 21.650 1.055 54.83 20.52 0.94 Maternity 4.874 4 1.01 1.02 0.1 Portified Midwife 1.02 2.2 1.03 56.96 86.0 Portified Midwife 1.02 2.2 3.30 6.00 0.0 0.0 <	0,									
Pharmacy Cardiovascular Cardiovascular PT/OT/ST Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular PS/blafic Cardiovascular										
Cardiovascular 25,175 378 196,5 66,60 1.09 Psychiatric 24,428 417 216,8 56,58 1.06 Substance Abuse 3,594 32 16,6 112,30 0.16 Other 427,806 26,714 13,888,5 16,01 18,53 Professional and Other State Plan Services 511,865,714 190,035 56,868,9 \$10,01 8,55,16 Proventive Medicine 21,66,0 11,75 6,04,42,7 22,38 12,95 Preventive Medicine 21,65,0 1,05,5 54,84,5 20,52 0.94 Maternity 4,874 41 21,1 120,12 0.21 Certified Midvife 0 0 0 0.0 0.00 0.00 PryChistric and Substance Abuse 960,366 16,023 8,330,3 59,94 41,81 Radiology and Pathology 76,748 7,113 3,698,0 11,07 3.41 Home Health and Private Duty Nursing 828,87 10,226 5,316,5		1 1								
PT(VT)ST 72,124 5,918 3,076.7 12,19 3,12 Substance Abuse 3,594 32 16.6 112.30 0.16 Substance Abuse 3,594 32 16.6 112.30 0.16 Other 427,806 26,714 13,888.5 16.01 18.53 Professional and Other State Plan Services	Pharmacy									
Psychiatric 24,428 417 121.8. 56.58 1.06 Other 427,806 26,714 13,888.5 16.01 18.53 Professional and Other State Plan Services 51,965,714 190,035 56,688.9 \$10.01 8.53 Professional and Other State Plan Services 51,865,714 190,035 56,688.9 \$10.01 8.55 16.01 20.02 20.02 20.02 20.02 10.05 54.85 20.52 0.94 Maternity 4,874 41 21.1 120.12 0.21 0.21 0.00	Cardiovascular									
Substance Abuse 3,594 32 16.6 112.30 0.16 Other 427.806 26.774 13.8885 16.01 18.853 Professional and Other State Plan Services 1109.035 566.866.9 \$18.866 117 60.8 \$161.42 \$0.855.16 Ambulatory Surgery Center 29.86.81 1.3.534 6.942.7 22.38 12.95 Preventive Medicine 21.650 1.055 548.5 20.52 0.94 Matemity 4.874 441 21.1 120.12 0.21 0.10 0.00 </td <td>PT/OT/ST</td> <td></td> <td></td> <td>,</td> <td></td> <td>· · · · · ·</td> <td></td> <td></td>	PT/OT/ST			,		· · · · · ·				
Other 427,806 267,74 198,85 16.01 18,885 Professional and Other State Plan Services Ambulatory Surgery Center \$1,985,774 109,035 56,869 \$118.03 \$55,16 Professional and Other State Plan Services \$18,886 117 60,8 \$161,42 \$50,82 Office Visits 288,811 13,354 6,942,7 22,38 \$20,82 Preventive Medicine 21,650 1,055 548,5 20,22 0.94 Maternity 4,674 41 21,1 120,12 0.21 Certified Midwife 0 0 0.00 0.00 0.00 Profoxing Plantic and Substance Abuse 990,386 16,023 8,330,3 59,94 41,61 Home Health and Private Duty Nursing 88,288 1,089 878,1 52,27 3,83 Ano-Emergency Transportation 984,959 44,711 26,340,3 19,59 41,37 Opicia Treatment Program 115,055 11,27 5,862,5 10.02 4,38 Apolied Dep Care	Psychiatric	24,428		417		216.8	58.58	1.06		
S1 965.714 199.035 56.686.9 \$18.03 \$85.16 Professional of ther State Plan Services \$18,886 117 60.8 \$161.42 \$0.82 Ambulatory Surgery Center 296,811 13.354 6.942.7 22.38 12.95 Preventive Medicine 21,650 1.055 548.5 20.52 0.94 Matemity 4,874 41 21.1 120.12 0.21 Cartified Midwle 0 0 0.0 0.00 0.00 Provintive Madicine Abuse 960.386 16.023 8.330.3 59.94 4.16 Pachistria and Substance Abuse 960.386 16.023 8.781.1 52.27 3.84 Radiology and Pathology 78,748 7.13 3.6980.11.0 3.41 Home Health and Private Duty Nursing 88.288 1.1693 878.1 52.2 3.83 Opicid Trastmare Program 115.055 11.257 5.852.5 10.2 4.98 Fedorally Cualified and Rural Health Clinics 23.791 366 190.3 <t< td=""><td>Substance Abuse</td><td>3,594</td><td></td><td>32</td><td></td><td>16.6</td><td>112.30</td><td>0.16</td></t<>	Substance Abuse	3,594		32		16.6	112.30	0.16		
Professional and Other State Plan Services Ambulatory Surgery Center Stat. 886 117 60.8 \$161.42 \$0.82 Office Visits 298.811 13.354 6.942.7 22.38 12.95 Office Visits 21.850 1.055 548.5 20.52 0.94 Maternity 4,874 4 0 0.00 0.00 Cartified Midwle 21.850 14.015 2.241 1,165.1 6.825 0.61 Psychiatric and Substance Abuse 960.386 16.023 8.330.3 59.94 4.161 Professity 78,748 7.113 3.060.0 11.07 3.41 Home Health and Private Duty Nursing 88.288 1.689 878.1 52.27 3.83 Anbulance 13.8487 10.226 5.316.5 13.56 6.02 Non-Emergency Transportation 954.959 48.741 25.340.3 19.959 41.37 Opcied Treatment Program 115.055 11.267 5.852.5 10.22 4.88 Aptiled Behavioral Analysis 0 </td <td>Other</td> <td>427,806</td> <td></td> <td>26,714</td> <td></td> <td>13,888.5</td> <td>16.01</td> <td>18.53</td>	Other	427,806		26,714		13,888.5	16.01	18.53		
Ambulatory Surgery Center \$18,886 117 60.8 \$161.42 \$0.82 Office Visits 298,811 13,354 6,942.7 22.38 12.95 Preventive Medicine 21,650 1,055 548.5 20.52 0.94 Maternity 4,874 41 21.1 120.12 0.21 Preventive Medicine 0 0 0.00 0.00 PT/OT/ST 14,015 2.241 1,165.1 6.25 0.61 Psychiatric and Substance Abuse 960,366 16.023 8,330.3 59.94 41.81 Radiogy and Pathology 78,748 7,113 3,668.0 11.07 3.41 Non-Emergency Transportation 954,959 48,741 25,540.3 19.59 41.37 Opiod Treatment Program 115,055 11,257 5,852.5 10.22 4.88 Opiod Treatment Program 12,857 3,640 18,945.0 5.40 8.522 Darable Medical Equipment 218,100 18,874 98.127.3 1.16 9.		\$1,965,714		109,035		56,686.9	\$18.03	\$85.16		
Office Visits 298,811 13,354 6,942.7 22.38 12.95 Preventive Medicine 21,650 1,055 548.5 20,52 0,94 Matemity 4,874 41 21.1 120.12 0,21 Certified Midwife 0 0 0,0 0,00 0,00 Psychiatric and Substance Abuse 960,386 16,023 6,330.3 59.94 41.61 Radiology and Pathology 78,748 7,113 3,698.0 11.07 3.41 Home Heatth and Private Duty Nursing 88,288 1,689 878.1 52.27 3.83 Ambulance 133,887 10.226 5.316.5 13.58 6.02 Opicid Treatment Program 115,055 11.257 5.852.5 10.22 4.88 Federally Qualified and Rural Health Clinics 23,791 366 100.3 65.00 10.3 Adult Medical Equipment 216,000 188,744 98,127.3 1.16 9.45 Applied Behavigs 0 0 0.00 0.0	Professional and Other State Plan Services									
Preventive Medicine 21,650 1,055 548,5 20,52 0.94 Matemity 4,374 0 0 0,00 0,00 0,00 0,00 PT/OT/ST 14,015 2,241 1,165,1 6,25 0,81 Psychiatric and Substance Abuse 960,386 16,023 8,300,3 59,94 41,61 Radiology and Pathology 78,748 7,113 3,686,0 11,07 3,41 home Health and Private Duty Nursing 88,288 10,226 5,316,5 13,38 66,02 Non-Emergency Transportation 964,959 48,741 25,340,3 19,99 41,37 Opiod Treatment Program 115,055 11,257 5,852,5 10,22 4,98 Personal Care 196,705 36,440 18,945,0 5,40,3 15,01 0,75 Parsonal Care 196,705 36,440 18,874 98,122 3,13,0 23,41,76,7 58,21 \$160,22 Opted Parkviral Analysis 0 0 0,0 0,0 0,0	Ambulatory Surgery Center	\$18,886		117		60.8	\$161.42	\$0.82		
Matemity 4.874 41 21.1 120.12 0.21 Certified Midwife 0 0 0.00 0.00 0.00 PryChistric and Substance Abuse 960.386 16.023 8.330.3 55.94 41.61 Psychiatric and Substance Abuse 960.386 16.023 8.330.3 55.94 41.61 Home Health and Private Duty Nursing 88.288 1.689 878.1 52.27 3.83 Ambulance 138.887 10.226 5.316.6 13.58 6.02 Non-Emergency Transportation 954.959 44.741 25.340.3 19.59 44.37 Opicid Treatment Program 115.055 11.257 5.852.5 10.22 4.98 Federally Qualified and Rural Health Clinics 23.791 3.66 190.3 65.00 10.37 Auti Medical Equipment 218,100 188,744 98,127.3 1.16 94.5 Durable Medical Equipment 218,100 11.876 58.163.9 4.90 23.73 MultiSouce Brand 2.020	Office Visits	298,811		13,354		6,942.7	22.38	12.95		
Matemity 4.874 41 21.1 120.12 0.21 Certified Midwife 0 0 0.00 0.00 0.00 PryChistric and Substance Abuse 960.386 16.023 8.330.3 55.94 41.61 Psychiatric and Substance Abuse 960.386 16.023 8.330.3 55.94 41.61 Home Health and Private Duty Nursing 88.288 1.689 878.1 52.27 3.83 Ambulance 138.887 10.226 5.316.6 13.58 6.02 Non-Emergency Transportation 954.959 44.741 25.340.3 19.59 44.37 Opicid Treatment Program 115.055 11.257 5.852.5 10.22 4.98 Federally Qualified and Rural Health Clinics 23.791 3.66 190.3 65.00 10.37 Auti Medical Equipment 218,100 188,744 98,127.3 1.16 94.5 Durable Medical Equipment 218,100 11.876 58.163.9 4.90 23.73 MultiSouce Brand 2.020	Preventive Medicine			1,055		548.5	20.52	0.94		
Certified Midwife 0 0 0.0 0.00 0.00 0.00 0.00 PT/OT/ST 14.015 2.241 1.165.1 6.25 0.61 Psychiatric and Substance Abuse 960.386 16.023 8.330.3 55.94 41.61 Radiology and Pathology 78.748 7.113 3.698.0 11.07 3.41 Nom-Emergency Transportation 954.959 447.41 25.340.3 19.59 41.37 Opiol Treatment Program 115.055 11.257 5.852.5 10.22 4.98 Adult Medical Day Care 17.221 1.147 596.3 15.01 0.75 Personal Care 196.705 38.440 18.945.0 5.40 8.52 Durable Medical Equipment 218.100 118.8744 98.127.3 1.16 9.45 Applied Behavioral Analysis 0 0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00	Maternity	4,874				21.1	120.12	0.21		
PT/OT/ST 14,015 2,241 1,165.1 6.25 0.61 Psychiatric and Substance Abuse 960,386 16,023 8,330.3 59.94 41.61 Radiology and Pathology 78,748 7,113 3.698.0 11.07 3.41 Home Health and Private Duty Nursing 88,288 1,689 878.1 52.27 3.83 Non-Emergency Transportation 954,959 48,741 25,340.3 19,59 41.37 Opiold Treatment Program 115,055 11,257 5,852.5 10.22 498 Federally Qualified and Rural Health Clinics 23,791 366 190.3 65.00 1.03 Adult Medical Day Care 17,221 1,1147 596.3 15.01 0.75 Personal Care 196,705 36,440 18,945.0 5.40 852 Durable Medical Equipment 218,100 188,744 99,127.3 1.16 9.45 Single-Source Brand 2,020 221 114.9 9.44 0.00 Single-Source Brand 2,020 <t< td=""><td>5</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	5									
Psychiatric and Substance Abuse 960,386 16,023 8,330.3 95.94 41.61 Radiology and Pathology 78,748 7,113 3,698.0 11.07 3.41 Home Health and Private Duty Nursing 88,288 1,688 875.1 52.27 3.83 Ambulance 138,887 10,226 5,316.5 13,58 6.02 Non-Emergency Transportation 954,959 448,741 25,340.3 19.59 41.37 Opioid Treatment Program 115,055 11,257 5.882.5 10.22 4.98 Personal Care 17,221 1,147 596.3 15.01 0.75 Personal Care 196,705 36,440 18,945.0 5.40 8.52 Durable Medical Equipment 21,8100 118,874 99,127.3 1.16 9.45 Opher 547,689 111.876 55,19.4 4.90 22.37 Prescription Drugs \$28,090.065 450,430 2234,176.7 58.21 \$160.22 Single-Source Brand 2,020 221	PT/OT/ST	14.015		2.241			6.25			
Radiology and Pathology 78,748 7,113 3,688,00 11,07 3.41 Home Health and Private Duty Nursing 88,288 1,689 878,1 52,27 3.83 Ambulance 138,887 10,226 5,316,5 13,58 6.02 Non-Emergency Transportation 954,959 48,741 25,340,3 19,59 41,37 Opiold Treatment Program 111,257 5,852,5 10,22 4,98 Federally Qualified and Rural Health Clinics 23,791 366 190,3 65.00 1.03 Adult Medical Day Care 17,221 1,147 5566,3 15.01 0.75 Personal Care 196,705 36,440 18,945,0 5.40 8.52 Durable Medical Equipment 218,100 188,744 98,127.3 1.16 9.45 Applied Behavioral Analysis 0 0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00										
Home Health and Private Duty Nursing 88.288 1.689 678.1 52.27 3.83 Ambulance 138,887 10,226 5.316.5 13.58 6.02 Non-Emergency Transportation 954,959 48,741 25.340.3 19.59 41.37 Opioid Treatment Program 115,055 11.257 5.852.5 10.22 4.98 Federally Qualified and Rural Health Clinics 23.791 366 190.3 65.00 10.3 Adult Medical Day Care 17.221 1,147 596.3 15.01 0.75 Personal Care 196,705 36,440 18,945.0 5.40 8.52 Durable Medical Equipment 218,100 188,744 98,127.3 1.16 9.45 Applied Behavioral Analysis 0 0 0.00 0.00 0.00 0.00 0.00 0.00 2.373 1.16 9.45 2.45 1.62.27 \$.88.1 \$.16.22 \$.69.1 5.5 2.8 1.46 9.52.73 1.16 0.00 0.00 0.00 0.00 </td <td>5</td> <td></td> <td></td> <td>,</td> <td></td> <td></td> <td></td> <td></td>	5			,						
Ambulance 138,887 10,226 5,316.5 13,58 6.02 Non-Emergency Transportation 954,959 48,741 25,340.3 19,59 41.37 Non-Emergency Transportation 954,959 48,741 25,340.3 19,59 41.37 Opiold Treatment Program 115,055 11,257 5,852.5 10.22 4.98 Federally Qualified and Rural Health Clinics 23,791 366 190.3 65.00 10.3 Auth Medical Day Care 17,221 1,147 596.3 15.01 0.75 Personal Care 196,705 36,440 18,845.0 5.40 8.52 Durable Medical Equipment 218,100 188,744 98,127.3 1.16 9.45 Applied Behavioral Analysis 0 0 0.00 <td< td=""><td></td><td></td><td></td><td></td><td></td><td>- ,</td><td></td><td>-</td></td<>						- ,		-		
Non-Emergency Transportation 954,959 48,741 25,340.3 19,59 41,37 Opioid Treatment Program 115,055 11,257 5,852.5 10,22 4,98 Federally Qualified and Rural Health Clinics 23,791 366 190.3 65.00 1.03 Adult Medical Day Care 17,221 1,147 596.3 15.01 0.75 Durable Medical Equipment 218,100 188,744 98,127.3 1.16 9.45 Applied Behavioral Analysis 0 0 0 0.00										
Opioid Treatment Program 115.055 11,257 5.852.5 10.22 4.98 Federally Qualified and Rural Health Clinics 23,791 366 190.3 65.00 1.03 Adult Medical Day Care 17,221 1,147 596.3 15.01 0.75 Personal Care 196,705 36,440 18,945.0 5.40 8.52 Durable Medical Equipment 218,100 188,744 98,127.3 1.16 9.45 Applied Behavioral Analysis 0 0 0.00 0.00 0.00 0.00 Other 547,689 111,876 58,163.9 4.90 23.73 Prescription Drugs										
Federally Qualified and Rural Health Clinics 23,791 366 190.3 65.00 1.03 Adult Medical Day Care 17,221 1,147 596.3 15.01 0.75 Personal Care 196,705 36.440 18,945.0 5.40 8.52 Durable Medical Equipment 218,100 188,744 98,127.3 1.16 9.45 Applied Behavioral Analysis 0 0 0 0.000 0.000 0.000 Other 547,689 111,876 58,163.9 4.90 23.73 Prescription Drugs				,		· · · · · ·		-		
Adult Medical Day Care 17,221 1,147 596.3 15.01 0.75 Personal Care 196,705 36,440 18,945.0 5.40 8.52 Durable Medical Equipment 218,100 188,744 98,127.3 1.16 9.45 Applied Behavioral Analysis 0 0 0 0.00 0.00 0.00 Other 547,689 111,876 58,163.9 4.90 23.73 Generic Scripts \$2,699,065 450,430 234,176.7 \$8.21 \$160.22 Generic Scripts \$28,133 7,642 3,973.0 \$3.68 \$1.22 Single-Source Brand 1,183 78 40.6 15.16 0.05 Specialty 0 0 0 0.00 0.00 0.00 Other 55 28 14.6 1.97 0.00 0.00 Community Mental Health Center C 2 14.6 1.97 0.00 Community Mental Health Center S13.391 7,969 4,143.1										
Personal Care 196,705 36,440 18,945.0 5.40 8.52 Durable Medical Equipment 218,100 188,744 98,127.3 1.16 9.45 Applied Behavioral Analysis 0 0 0.00<										
Durable Medical Equipment 218,100 188,744 98,127.3 1.16 9.45 Applied Behavioral Analysis 0 0 0.0 0.00 <td< td=""><td>•</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>	•									
Applied Behavioral Analysis 0 0 0 0.0 0.00 0.00 0.00 Other 547,689 111,876 58,163.9 4.90 23.73 Prescription Drugs \$3,698,065 450,430 234,176.7 \$8.21 \$160.22 Prescription Drugs \$28,133 7,642 3,973.0 \$3.68 \$1.22 Single-Source Brand 2,020 221 114.9 9.14 0.09 Multi-Source Brand 1,183 78 40.6 15.16 0.00 Specialty 0 0 0 0.00 0.00 0.00 Hepatitis C 0 0 0.00 0.00 0.00 0.00 Cher 55 28 14.6 1.97 0.00 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>										
Other 547,689 111,876 58,163.9 4.90 23.73 Prescription Drugs \$3,698,065 450,430 234,176.7 \$8.21 \$160.22 Generic Scripts \$28,133 7,642 3,973.0 \$3.68 \$1.22 Single-Source Brand 2,020 221 114.9 9.14 0.09 Multi-Source Brand 1,183 78 40.6 15.16 0.05 Specialty 0 0 0 0.00 0.00 0.00 Repatitis C 0 0 0 0.00 0.00 0.00 Other 55 28 14.6 1.97 0.00 Other 55 28 14.6 1.97 0.00 Other 55 28 14.6 1.97 0.00 Community Mental Health Center 57,121,612 18,344 9,537.0 \$388.23 \$308.54 Long Term Support Service 9,796,998 330,655 171,906.4 29.63 424.45 Partial										
\$3,698,065 450,430 234,176.7 \$8.21 \$160.22 Prescription Drugs \$28,133 7,642 3,973.0 \$3.68 \$1.22 Single-Source Brand 2,020 221 114.9 9.14 0.09 Multi-Source Brand 1,183 78 40.6 15.16 0.05 Specialty 0 0 0.0 0.00 0.00 0.00 Multi-Source Brand 55 28 14.6 1.97 0.00 0		1		-						
Prescription Drugs \$28,133 7,642 3,973.0 \$3.68 \$1.22 Single-Source Brand 2,020 221 114.9 9.14 0.09 Multi-Source Brand 1,183 78 40.6 15.16 0.05 Specialty 0 0 0 0.00 0.00 0.00 Hepatitis C 0 0 0 0.00 0.00 0.00 Other 55 28 14.6 1.97 0.00 Cher \$31,391 7,969 4,143.1 \$3.94 \$1.36 Community Mental Health Center Community Mental Health Center 7,969 4,143.1 \$3.94 \$1.36 Case Management \$7,121,612 18,344 9,537.0 \$388.23 \$308.55 Partial Hospital 619,319 6,064 3,152.7 102.13 26.83 Psychotherapy 676,755 20,801 10,814.4 32.53 29.32 Evidence Based Practice 168,798 8,515 4,426.9 19.42	Otilei									
Generic Scripts \$28,133 7,642 3,973.0 \$3.68 \$1.22 Single-Source Brand 2,020 221 114.9 9.14 0.09 Multi-Source Brand 1,183 78 40.6 15.16 0.05 Specialty 0 0 0.0 0.00 0.00 0.00 Hepatitis C 0 0 0 0.00 0.00 0.00 0.00 Cotter 55 28 14.6 1.97 0.00 Community Mental Health Center \$31,391 7,969 4,143.1 \$3.94 \$1.36 Case Management \$7,121,612 18,344 9,537.0 \$388.23 \$308.54 Long Term Support Service 9,796,998 330,655 171,906.4 29.63 424.45 Partial Hospital 619,319 6,064 3,152.7 102.13 26.83 Psychotherapy 676,755 20,801 10,814.4 32.53 29.32 Evidence Based Practice 168,798 8,515 4,426.9	Proscription Drugs	\$3,090,005		450,450		234,170.7	φ0.21	\$100.22		
Single-Source Brand 2,020 221 114.9 9.14 0.09 Multi-Source Brand 1,183 78 40.6 15.16 0.05 Specialty 0 0 0.0 0.00 0.00 0.00 Hepatitis C 0 0 0.0 0.00 0.00 0.00 Other 55 28 14.6 1.97 0.00 Community Mental Health Center		\$28 133		7 642		3 973 0	\$3.68	\$1.22		
Multi-Source Brand 1,183 78 40.6 15.16 0.05 Specialty 0 0 0 0.0 0.00						· · · · ·				
Specialty 0 0 0 0.0 0.00										
Hepatitis C 0 0 0 0.0 0.00 0.										
Other 55 28 14.6 1.97 0.00 Comunity Mental Health Center \$31,391 7,969 4,143.1 \$3.94 \$1.36 Case Management \$7,121,612 18,344 9,537.0 \$388.23 \$308.54 Long Term Support Service 9,796,998 330,655 171,906.4 29.63 424.45 Partial Hospital 619,319 6,064 3,152.7 102.13 26.83 Psychotherapy 676,755 20,801 10,814.4 32.53 29.32 Evidence Based Practice 168,798 8,515 4,426.9 19.82 7.31 Medication Management 143,560 4,704 2.445.6 30.52 6.22 Emergency Service 24/7 66,788 2,831 1,471.8 23.59 2.89 APRTP 172,711 290 150.8 595.56 7.48 Supported Employment Services 826,998 28,618 14,878.4 28.90 35.83 Other 313,603 10,646 5,534.8 29.46 <td></td> <td></td> <td></td> <td>-</td> <td></td> <td></td> <td></td> <td></td>				-						
\$31,391 7,969 4,143.1 \$3.94 \$1.36 Community Mental Health Center \$7,121,612 18,344 9,537.0 \$388.23 \$308.54 Long Term Support Service 9,796,998 330,655 171,906.4 29.63 424.45 Partial Hospital 619,319 6,064 3,152.7 102.13 26.83 Psychotherapy 676,755 20,801 10,814.4 32.53 29.32 Evidence Based Practice 168,798 8,515 4,426.9 19.82 7.31 Medication Management 143,560 4,704 2,445.6 30.52 6.22 Emergency Service 24/7 66,788 2,831 1,471.8 23.59 2.89 APRTP 172,711 290 150.8 595.56 7.48 Supported Employment Services 826,998 28,618 14,878.4 29.46 13.59 Other 313,603 10,646 5,534.8 29.46 13.59 Long Termine Services \$19,907,143 431,468 224,318.7 \$46	•			-						
Community Mental Health Center \$7,121,612 18,344 9,537.0 \$388.23 \$308.54 Case Management 9,796,998 330,655 171,906.4 29.63 424.45 Partial Hospital 619,319 6,064 3,152.7 102.13 26.83 Psychotherapy 676,755 20,801 10,814.4 32.53 29.32 Evidence Based Practice 168,798 8,515 4,426.9 19.82 7.31 Medication Management 143,560 4,704 2,445.6 30.52 6.22 Emergency Service 24/7 66,788 2,831 1,471.8 23.59 2.89 APRTP 172,711 290 150.8 595.56 7.48 Supported Employment Services 826,998 28,618 14,878.4 28.90 35.83 Other 313,603 10,646 5,534.8 29.46 13.59 \$19,907,143 431,468 224,318.7 \$46.14 \$862.47	Other									
Case Management \$7,121,612 18,344 9,537.0 \$388.23 \$308.54 Long Term Support Service 9,796,998 330,655 171,906.4 29.63 424.45 Partial Hospital 619,319 6,064 3,152.7 102.13 26.83 Psychotherapy 676,755 20,801 10,814.4 32.53 29.32 Evidence Based Practice 188,798 8,515 4,426.9 19.82 7.31 Medication Management 143,560 4,704 2,445.6 30.52 6.22 Emergency Service 24/7 66,788 2,831 1,471.8 23.59 2.89 APRTP 172,711 290 150.8 595.56 7.48 Supported Employment Services 826,998 28,618 14,878.4 29.94 13.59 Other 313,603 10,646 5,534.8 29.46 13.59 \$19,907,143 431,468 224,318.7 \$46.14 \$862.47	Community Montal Health Contor	\$31,391		7,909		4,143.1	<u> </u>	\$1.30		
Long Term Support Service 9,796,998 330,655 171,906.4 29.63 424.45 Partial Hospital 619,319 6,064 3,152.7 102.13 26.83 Psychotherapy 676,755 20,801 10,814.4 32.53 29.32 Evidence Based Practice 168,798 8,515 4,426.9 19.82 7.31 Medication Management 143,560 4,704 2,445.6 30.52 6.22 Emergency Service 24/7 66,788 2,831 1,471.8 23.59 2.89 APRTP 172,711 290 150.8 595.56 7.48 Supported Employment Services 826,998 28,618 14,878.4 28.90 35.83 Other 313,603 10,646 5,534.8 29.46 13.59 \$19,907,143 431,468 224,318.7 \$46.14 \$862.47	-	¢7 404 040		40.044		0 507 0	¢000.00	¢200 F4		
Partial Hospital 619,319 6,064 3,152.7 102.13 26.83 Psychotherapy 676,755 20.801 10,814.4 32.53 29.32 Evidence Based Practice 168,798 8,515 4,426.9 19.82 7.31 Medication Management 143,560 4,704 2,445.6 30.52 6.22 Emergency Service 24/7 66,788 2,831 1,471.8 23.59 2.89 APRTP 172,711 290 150.8 595.56 7.48 Supported Employment Services 826,998 28,618 14,878.4 28.90 35.83 Other 313,603 10,646 5,534.8 29.46 13.59 \$19,907,143 431,468 224,318.7 \$46.14 \$862.47										
Psychotherapy 676,755 20,801 10,814.4 32.53 29.32 Evidence Based Practice 168,798 8,515 4,426.9 19.82 7.31 Medication Management 143,560 4,704 2,445.6 30.52 6.22 Emergency Service 24/7 66,788 2,831 1,471.8 23.59 2.89 APRTP 172,711 290 150.8 595.56 7.48 Supported Employment Services 826,998 28,618 14,878.4 28.90 35.83 Other 313,603 10,646 5,534.8 29.46 13.59 \$19,907,143 431,468 224,318.7 \$46.14 \$862.47										
Evidence Based Practice 168,798 8,515 4,426.9 19.82 7.31 Medication Management 143,560 4,704 2,445.6 30.52 6.22 Emergency Service 24/7 66,788 2,831 1,471.8 23.59 2.89 APRTP 172,711 290 150.8 595.56 7.48 Supported Employment Services 826,998 28,618 14,878.4 28.90 35.83 Other 313,603 10,646 5,534.8 29.46 13.59 \$19,907,143 431,468 224,318.7 \$46.14 \$862.47										
Medication Management 143,560 4,704 2,445.6 30.52 6.22 Emergency Service 24/7 66,788 2,831 1,471.8 23.59 2.89 APRTP 172,711 290 150.8 595.56 7.48 Supported Employment Services 826,998 28,618 14,878.4 28.90 35.83 Other 313,603 10,646 55,534.8 29.46 13.59 \$19,907,143 431,468 224,318.7 \$46.14 \$862.47						· · · · ·				
Emergency Service 24/7 66,788 2,831 1,471.8 23.59 2.89 APRTP 172,711 290 150.8 595.56 7.48 Supported Employment Services 826,998 28,618 14,878.4 28.90 35.83 Other 313,603 10,646 55,534.8 29.46 13.59 \$19,907,143 431,468 224,318.7 \$46.14 \$862.47										
APRTP 172,711 290 150.8 595.56 7.48 Supported Employment Services 826,998 28,618 14,878.4 28.90 35.83 Other 313,603 10,646 5,534.8 29.46 13.59 \$19,907,143 431,468 224,318.7 \$46.14 \$862.47										
Supported Employment Services 826,998 28,618 14,878.4 28.90 35.83 Other 313,603 10,646 5,534.8 29.46 13.59 \$19,907,143 431,468 224,318.7 \$46.14 \$862.47										
Other 313,603 10,646 5,534.8 29.46 13.59 \$19,907,143 431,468 224,318.7 \$46.14 \$862.47										
\$19,907,143 431,468 224,318.7 \$46.14 \$862.47	Supported Employment Services									
	Other									
All Services \$26,814,619 569 1,003,451 295.8 521,690.4 \$26.72 \$1,161.74										
	All Services	\$26,814,619	569	1,003,451	295.8	521,690.4	\$26.72	\$1,161.74		

Eligibility Category: Severe Mental Illness - Non-Dual

Member Months: 7,448

							Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient				.,	.,		
Medical	\$570,897	121	1,006	194.9	1,620.8	\$4,718.16	\$76.65
Surgical	253,436	25	448	40.3	721.8	10,137.42	34.03
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	8,988	4	19	6.4	30.6	2,246.92	1.21
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	402,004	70	523	112.8	842.6	5,742.91	53.97
Alcohol and Drug Abuse	94,463	23	172	37.1	277.1	4,107.08	12.68
Other	0	0	0	0.0	0.0	0.00	0.00
	\$1,329,787	243	2,168	391.5	3,492.8	\$5,472.37	\$178.53
Hospital Outpatient			4 9 9 9				* / * / *
Emergency Room	\$777,336		1,206		1,943.0	\$644.56	\$104.36
Surgery	93,534		127		204.6	736.48	12.56
Radiology	150,532		689		1,110.0	218.48	20.21
Pathology	82,748		5,641		9,088.2	14.67	11.11
Pharmacy	37,885		13,989		22,537.5	2.71	5.09
Cardiovascular	10,107		65		104.7	155.50	1.36
PT/OT/ST	47,931		2,826		4,552.9	16.96	6.44
Psychiatric	17,025		112		180.4	152.01	2.29
Substance Abuse	2,695		24		38.7	112.29	0.36
Other	155,739 \$1,375,531		4,977 29,656		8,018.4 47,778.5	31.29 \$46.38	20.91 \$184.68
Professional and Other State Plan Services	\$1,375,551		29,000		47,770.5	\$40.30	\$104.00
Ambulatory Surgery Center	\$16,536		62		99.9	\$266.70	\$2.22
Office Visits	232,158		4,161		6,703.7	55.79	φ <u>2.22</u> 31.17
Preventive Medicine	44,605		1,011		1,628.8	44.12	5.99
Maternity	26,542		108		173.8	246.02	3.56
Certified Midwife	47		2		3.2	23.50	0.01
PT/OT/ST	17,331		763		1,229.3	22.71	2.33
Psychiatric and Substance Abuse	525,452		4,388		7,069.5	119.75	70.55
Radiology and Pathology	175,560		6,111		9,845.4	28.73	23.57
Home Health and Private Duty Nursing	62,411		1,033		1,664.3	60.42	8.38
Ambulance	94,777		4,690		7,556.0	20.21	12.72
Non-Emergency Transportation	338,803		22,568		36,359.1	15.01	45.49
Opioid Treatment Program	136,098		13,316		21,453.3	10.22	18.27
Federally Qualified and Rural Health Clinics	193,661		1,299		2,092.8	149.09	26.00
Adult Medical Day Care	0		0		0.0	0.00	0.00
Personal Care	39,438		7,309		11,775.5	5.40	5.29
Durable Medical Equipment	96,451		15,214		24,511.1	6.34	12.95
Applied Behavioral Analysis	0		0		0.0	0.00	0.00
Other	488,659		29,687		47,828.4	16.46	65.61
	\$2,488,528		111,722		179,994.1	\$22.27	\$334.10
Prescription Drugs							
Generic Scripts	\$503,323		27,485		44,280.8	\$18.31	\$67.57
Single-Source Brand	740,793		1,819		2,930.6	407.25	99.46
Multi-Source Brand	213,338		852		1,372.6	250.40	28.64
Specialty	688,035		176		283.6	3,909.29	92.37
Hepatitis C	174,034		15		24.2	11,602.27	23.37
Other	46		8		12.9	5.71	0.01
Community Mantal Haalth Conton	\$2,319,569		30,355		48,904.7	\$76.41	\$311.42
Community Mental Health Center	A4 400 050		0 704		0.000.4		\$407.04
Case Management	\$1,469,658		3,784		6,096.4	\$388.39	\$197.31
Long Term Support Service	411,654		15,157		24,419.3 0.0	27.16	55.27
Partial Hospital Psychotherapy	463,382		0 6,911		0.0 11,134.2	0.00 67.05	0.00 62.21
Evidence Based Practice	463,382 10,957		456		734.7	24.03	1.47
Medication Management	5,581		456 199		320.6	24.03 28.05	0.75
Emergency Service 24/7	5,581		236		320.6	28.05	0.75
APRTP	58,362		230 98		157.9	595.53	7.84
Supported Employment Services	57,954		2,025		3,262.5	28.62	7.78
Other	208,684		2,025		3,202.5	89.56	28.02
	\$2,691,769		31,196		50,259.6	\$86.29	\$361.39
All Services	\$10,205,184	243	205,097	391.5	330,429.6	\$49.76	\$1,370.12
	+,200,104	2.10	200,001		230,120.0	÷	÷.,0.0.12

Eligibility Category: Severe Mental Illness - Dual

Member Months: 3,417

	Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid
Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost
Hospital Inpatient							
Medical	\$57,165	40	171	140.5	600.6	\$1,429.13	\$16.73
Surgical	290,280	17	312	59.7	1,095.8	17,075.29	84.96
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	5,456	4	8	14.0	28.1	1,364.00	1.60
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	49,756	22	307	77.3	1,078.3	2,261.62	14.56
Alcohol and Drug Abuse	3,698	2	8	7.0	28.1	1,849.10	1.08
Other	0	0	0	0.0	0.0	0.00	0.00
	\$406,355	85	806	298.5	2,830.9	\$4,780.64	\$118.94
Hospital Outpatient							
Emergency Room	\$121,721		369		1,296.0	\$329.87	\$35.63
Surgery	24,475		126		442.5	194.25	7.16
Radiology	26,518		339		1,190.7	78.22	7.76
Pathology	7,443		1,197		4,204.2	6.22	2.18
Pharmacy	9,748		7,917		27,806.5	1.23	2.85
Cardiovascular	1,762		39		137.0	45.18	0.52
PT/OT/ST	10,505		802		2,816.8	13.10	3.07
	1,201		28		2,810.8	42.89	0.35
Psychiatric							
Substance Abuse	674		6		21.1	112.29	0.20
Other	52,401		2,717		9,542.8	19.29	15.34
Professional and Other State Dian Services	\$256,448		13,540		47,555.9	\$18.94	\$75.06
Professional and Other State Plan Services	04.400		10		45.7		* 0.40
Ambulatory Surgery Center	\$1,469		13		45.7	\$113.03	\$0.43
Office Visits	46,986		2,044		7,179.0	22.99	13.75
Preventive Medicine	3,651		132		463.6	27.66	1.07
Maternity	1,227		16		57.3	75.21	0.36
Certified Midwife	0		0		0.0	0.00	0.00
PT/OT/ST	2,263		208		730.5	10.88	0.66
Psychiatric and Substance Abuse	197,320		2,813		9,880.0	70.15	57.75
Radiology and Pathology	15,394		1,122		3,940.7	13.72	4.51
Home Health and Private Duty Nursing	9,533		948		3,329.6	10.06	2.79
Ambulance	20,961		1,224		4,299.0	17.13	6.14
Non-Emergency Transportation	173,744		11,802		41,451.6	14.72	50.85
Opioid Treatment Program	53,354		5,220		18,334.0	10.22	15.62
Federally Qualified and Rural Health Clinics	9,615		78		274.0	123.27	2.81
Adult Medical Day Care	3,189		225		790.3	14.17	0.93
Personal Care	15,108		2,800		9,834.3	5.40	4.42
Durable Medical Equipment	38,767		28,925		101,591.9	1.34	11.35
Applied Behavioral Analysis	0		0		0.0	0.00	0.00
Other	64,855		11,220		39,407.5	5.78	18.98
	\$657,437		68,790		241,608.9	\$9.56	\$192.42
Prescription Drugs	<i><i><i>q</i>001,101</i></i>				211,000.0	\$0.00	<i><i><i></i></i></i>
Generic Scripts	\$2,810		684		2,402.4	\$4.11	\$0.82
Single-Source Brand	529		30		105.4	17.64	0.15
Multi-Source Brand	545		12		42.1	45.44	0.16
Specialty	0		0		0.0	0.00	0.00
Hepatitis C	0		0		0.0	0.00	0.00
Other	10		1		3.5	10.41	0.00
	\$3,895		727		2,553.4	\$5.36	\$1.14
Community Mental Health Center	ψ0,000		121		2,000.4	φ0.00	φ1.14
Case Management	\$671,525		1,729		6,072.7	\$388.39	\$196.55
Long Term Support Service	335,146		12,010		42,182.1	27.91	98.09
Partial Hospital	15,612		12,010		533.9	102.71	4.57
Psychotherapy	107,638		2,656		9,328.5	40.53	4.57
Evidence Based Practice	9,135		674		2,367.3	13.55	2.67
Medication Management	4,760		176		618.2	27.05	1.39
Emergency Service 24/7	5,553		223		783.2	24.90	1.63
APRTP	20,844		35		122.9	595.53	6.10
Supported Employment Services	79,508		2,772		9,736.0	28.68	23.27
Other	36,796		881		3,094.3	41.77	10.77
	\$1,286,515		21,308		74,839.1	\$60.38	\$376.55
All Services	\$2,610,650	85	105,171	298.5	369,388.1	\$24.82	\$764.11

Eligibility Category: Low Utilizer - Non-Dual

Member Months: 1,412

Total Paid Total Paid Total Paid Total Paid Jump Average Paid Worthy Paid Hospial Inpatient 22,785 7 100 55.5 90.0 52,854 7 Marently Non-Delway 35,650 0 147 42.5 390.0 32,854 92.25 Marently Non-Delway 0 0 0 0 0 0 0.00								Per Capita
Hospital match S22,788 7 106 5.95 500.99 7.32,54 25,256.41 Surgical 35,603 5 47 4.2.5 300.44 7,152.44 25,256 Weil Newborn 0 0 0 0.00 <td< th=""><th>Benefits</th><th></th><th></th><th></th><th></th><th></th><th></th><th>Monthly Paid</th></td<>	Benefits							Monthly Paid
Medical 52,758 7 106 59.5 60.09 53,256,41 51,52,52 Metarrity Mon-Dailwey 0		Donard	7 1011110		.,	.,		
Sungical Maternity Delivery 35,663 5 47 42.5 399.4 7,12.54 25.84 Maternity Delivery 0 0 0 0.0 0.0 0.0 0.00 <	Medical	\$22,788	7	106	59.5	900.9	\$3.255.41	\$16.14
Malerinty Non-Dillery 0	Surgical							
Maternity Non-Delivery 0	Maternity Delivery						· · ·	0.00
Weil Newborn 0 <t< td=""><td>Maternity Non-Delivery</td><td>0</td><td>0</td><td>0</td><td>0.0</td><td>0.0</td><td>0.00</td><td>0.00</td></t<>	Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00
Adorb and Drug Abuse 0	Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Other 0 0 0 0 0 0 0.00 Hospial Outpatient 578.357 17 11445 1512.8 \$56.992.1 \$55.494 Enregrency Room S68.815 B8 747.9 \$773.35 \$49.44 Surgery 23.866 23 118.13 112.12 20.879 Pharmacy 22.966 23 118.13 212.70 20.819 Cardiovascular 2.626 22.610 101.00 116.81 PTOTNST 6.043 274 2.328.6 22.25.6 4.28 Deprict Astrone Abure 0 0 0.00 0.00 0.00 0.00 Other 2.4169 6.534 5.313 44.86.23 381.6 7.12 Professional and Other State Plan Services 4.777 5.733 44.86.23 383.00 5.318.46 Anbulardy Signery Center 3.2575 5.50 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00	Psychiatric	19,906	5	25	42.5	212.5	3,981.19	14.10
Stright of Upper Line Stright of Line Profile Stright of Line Stright of Line <td< td=""><td>Alcohol and Drug Abuse</td><td>0</td><td>0</td><td>0</td><td></td><td></td><td>0.00</td><td>0.00</td></td<>	Alcohol and Drug Abuse	0	0	0			0.00	0.00
Hospital Outpatient reservence reservence <threservence< th=""> reservence <th< td=""><td>Other</td><td>0</td><td>0</td><td>0</td><td>0.0</td><td>0.0</td><td>0.00</td><td>0.00</td></th<></threservence<>	Other	0	0	0	0.0	0.0	0.00	0.00
Emergency Room 599,815 888 747.9 5703.35 549.4 Radiology 29,866 139 1,181.3 212.70 20,94 Pharbacy 26,938 3,017 25,540.5 8,33 19,98 Cardiovascular 2,628 226 221.0 101.00 1.86 PTOTYST 6,043 274 2,5346.5 8,33 19,98 Substance Abuse 0 0 0,00		\$78,357	17	178	144.5	1,512.8	\$4,609.21	\$55.49
Surgéry 23,886 23 195.5 1,038.52 16.92 Pathology 12,407 931 7,12.3 13.33 8.79 Pathology 12,407 931 7,12.3 13.33 8.79 Cardiovascular 2,628 2,26 2,26 2,21.0 101.00 1.86 Psychatric 88 1 8.5 87.82 0.06 0.00 0.00 0.00 Other 24,169 634 5,382.2 3.81.2 7.78 0.06 0.00 <td>Hospital Outpatient</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Hospital Outpatient							
Radiology 29,666 139 1,181.3 212.70 20,94 Phanbogy 12,407 931 7,912.3 13,33 8.79 Phamacy 26,938 3,017 25,640.5 8.83 19,08 Cardiovascular 2,626 226 221.0 101,00 1.86 PTOTIST 6,043 274 2,328.6 22.05 4.28 Substance Abuse 0 0 0 0.0 0.00 0.00 Other 24,169 6,34 53,842 38.12 17.12 Profesional and Other State Plan Services 37,791 31 26,251 2,851 Other Visity 97,675 510 2,923.0 2,336 5,81 Other Visity 2,662 5 2,923.0 2,336 5,81 Maternity 2,562 5 4,24 5,31 3,170.0 2,00 0,00 0,00 Provinitive Mediane Abuse 33,256 7,68 6,766.4 2,24.3 1,235 3,170.	Emergency Room	\$69,815		88		747.9	\$793.35	\$49.44
Pathologing 12,407 931 7,912.3 13.33 8.79 Cardiovascular 26,938 3,017 25,940.5 8.83 19,08 Cardiovascular 26,626 26 221.0 101.00 1.86 Psychiatric 8.8 1 8.5 87.82 0.00 Other 2,0 0 0 0.00 0.00 0.00 Other 2,169 6.34 5,388.2 38.12 17.12 Professional and Other State Plan Services 3 4,2623.7 55.31 24,263.5 55.51 25.55.1	Surgery	23,886		23		195.5	1,038.52	16.92
Phamagy 26,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 10,000 0	Radiology	29,566		139		1,181.3	212.70	20.94
Cardiovscular 2.626 22 101.00 1.66 PryOrt/ST 6.043 2.74 2.326.6 22.05 4.28 Psychiatric 88 1 8.5 87.82 0.06 Subtance Abuse 0 0 0.00 0.00 0.00 0.00 Other 24,169 6.34 5.388.2 38.12 17.12 Professional and Other State Plan Services 519.557 5.13 43.023.7 538.09 \$13.84.8 Protentise Medicine 8.2075 590 5.014.2 55.38 22.314 Preventive Medicine 8.201 351 2.983.0 23.36 5.81 Maternity 2.562 5 4.24 513.46 1.81 Certified Movife 0 0 0 0.00 0.00 0.00 ProVIT/ST 4.483 215 1.827.2 22.85 3.17 Psychiatric and Substance Abuse 33.266 7576 6.442.0 4.24 5.31 Ambulance	Pathology	12,407		931		7,912.3	13.33	8.79
PTOTIST 6.043 274 2.82.65 22.05 4.88 Substance Abuse 0 0 0.0 0.00 0.00 Other 24.169 634 5.882.38.09 \$138.48 Professional and Other State Plan Services 31 283.5 \$25.113 43.82.37 \$38.09 \$138.48 Professional and Other State Plan Services 31 283.5 \$25.51 \$5.51 \$2.95.57 \$5.90 \$5.112.4 \$5.83.8 \$2.85.51 Office Visits 32.675 5.90 \$5.112.4 \$5.83.8 \$2.81.6 \$8.41 Preventive Medicine 8.201 351 2.985.0 \$2.33.6 \$5.81 Stational Advisor 0 0 0 0.00 0.00 0.00 PrOTIST 4.483 2.15 \$1.827.2 2.08.5 \$3.17 \$2.35 \$3.170.0 \$2.06 \$3.170.0 \$2.06 \$3.170.0 \$2.06 \$3.15 \$4.42.0 \$4.38.7 \$2.35 Raidolog and Pathology 7.781 3.170.0	Pharmacy	26,938		3,017		25,640.5	8.93	19.08
Psychiatric B8 1 B5 B7.82 0.06 Other 24,169 634 5.388.2 38.12 17.12 Professional and Other State Plan Services 5195.557 5.133 43.623.7 533.60 \$138.48 Professional and Other State Plan Services 577 5310 525.13	Cardiovascular	2,626		26		221.0	101.00	1.86
Substance Abuse 0 0 0 0.00 0.00 0.00 Other 24.169 634 5.383 43.623.7 \$38.09 \$138.48 Professional and Other State Plan Services 5.7781 31 263.5 \$2251.02 \$5.513.4 Ambulatory Surgery Center \$7.781 31 263.5 \$2251.02 \$5.513.4 Proventive Medicine 8.201 351 2.983.0 23.36 \$61 Proventive Medicine 8.201 351 2.983.0 0.00 <	PT/OT/ST	6,043		274		2,328.6	22.05	4.28
Other 24,169 634 5,388.2 38,12 17,12 S195,537 5,133 43,6237 \$38,60 \$138,48 Professional and Other State Plan Services Ambulatory Surgery Center 57,781 31 283,5 \$2,510 \$5,51 Office Visits 32,675 550 5,014,2 55,38 23,46 181,81 Perventive Medicine 8,201 35,51 2,983,0 23,38 5,81 Maternity 2,562 5 42,4 51,346 1,81 Carified Mdwife 0 0 0,00 0,00 0,00 ProVinis 4,483 215 1,827,2 20,85,31,77 20,85,31,77 Radiology and Pathology 17,834 795 6,76,64,22,43,12,63 10,72,83 10,72,83 10,73,33 3,170,0 20,10 5,31 Applied Pathology Transportation 59,956 4,874 41,422,5 12,30 42,46 Opicid Testment Program 23,170 20,10 15,43 43,65 42,61 10,73<	Psychiatric	88		1		8.5	87.82	0.06
S105.57 5.133 43.623.7 \$38.00 \$138.48 Ambulatory Surgery Center \$7.781 31 263.5 \$251.02 \$5.51 Office Visits 32.675 550 5.014.2 \$5.53 23.36 \$5.14 Preventive Medicine 8.201 351 2.983.0 23.36 \$5.14 Preventive Medicine 8.201 351 2.983.0 23.36 \$5.14 Certified Midwife 0 0 0.0 0.00 0	Substance Abuse	0		0		0.0	0.00	0.00
Professional and Other State Plan Services Ambulatory Surgery Center \$7.781 31 263.5 \$2251.02 \$5.51 Office Visits 32.675 500 5.014.2 55.38 23.14 Preventive Medicine 8.201 351 2.983.0 22.38 5.81 Maternity 2.662 5 42.4 513.46 1.81 Cartified Midwife 0 0 0.00 0.00 0.00 PTO/TST 4.483 215 1.827.2 20.65 3.17 Psychiatric and Substance Abuse 33.256 758 6.442.0 4.387 23.55 Acidogy and Pathology 17.634 795 6.756.4 22.43 12.63 Non-Emergency Transportation 59.966 4.874 41.422.5 12.30 42.46 Opicid Treatment Program 23.188 2.269 19.283.5 10.22 16.42 Apulted Day Care 10.728 515 4.376.5 20.83 5.60 Apulted Behavioral Analysis 0 0 0.00 <td< td=""><td>Other</td><td>24,169</td><td></td><td>634</td><td></td><td>5,388.2</td><td>38.12</td><td>17.12</td></td<>	Other	24,169		634		5,388.2	38.12	17.12
Ambulatory Surgery Center \$7,781 31 263.5 \$221.02 \$5.51 Office Visits 32,675 500 5,014.2 65.38 23.36 5.81 Maternity 2,662 5 4.2.4 513.46 1.81 Certified Midwife 0 0 0.00 0.00 0.00 ProVinitria di Substance Abuse 33,256 778 6,442.0 43.37 23.55 Radiology and Pathology 17,834 795 6,756.4 22.43 12.65 Anne Health and Private Duty Nursing 59,055 4,874 41,422.5 12.23 42.46 Opioid Treatment Program 22,189 2,269 19,283.5 10.02 16.42 Opioid Treatment Program 23,189 2,269 19,283.5 10.22 14.42 Opioid Treatment Program 23,189 2,269 19,283.5 10.22 16.42 Opiod Treatment Program 23,189 2,864.3 5.40 10.73 Durable Medical Equipment 19,226 3,188 26,923.8		\$195,537		5,133		43,623.7	\$38.09	\$138.48
Office Visis 52675 590 5.014.2 65.38 22.14 Preventive Midicine 8.201 351 2.983.0 23.36 6.81 Matemity 2.562 5 42.4 513.46 1.81 Certified Midwife 0 0 0 0.00 0.00 0.00 Psychiatric and Substance Abuse 33.266 758 6.442.0 43.87 22.55 Radiology and Pathology 17.834 795 6.756.4 22.43 12.65 Home Health and Private Duty Nursing 59.956 4.874 41.422.5 12.30 42.46 Opiold Treatment Program 23.189 2.269 19.283.5 10.22 16.42 Federally Qualified and Rural Health Clinics 26.756 186 1.580.8 143.85 19.95 Adult Medical Day Care 15.151 2.808 23.864.3 5.40 10.73 Durable Medical Equipment 19.226 29.700 252.410.5 514.41 52.97.6 Adult Medicical Caupiment 19.394 <	Professional and Other State Plan Services							
Preventive Medicine 8,201 351 2,983.0 23.36 5.81 Maternity 2,562 5 42.4 513.46 18.11 Cartified Midwire 0 0 0.00 0.00 0.00 PT/OT/ST 4,483 215 1.827.2 20.85 3.17 Sprchiatric and Substance Abuse 33.256 758 6.766.4 22.43 12.63 Nome Health and Private Duty Nursing 59.055 8.449 71.805.3 6.99 4.81 Ambulance 7,497 373 3.1700 20.10 5.31 Non-Emergency Transportation 59.956 4.874 41.422.5 12.30 12.22 16.42 Opioid Treatment Program 23.189 2.269 19.283.5 10.22 16.42 Personal Care 15.151 2.806 23.864.3 5.40 10.73 Durable Medical Equipment 19.226 3.168 26.923.8 6.07 13.62 Applied Behavioral Analysis 0 0 0 0.00	Ambulatory Surgery Center	\$7,781				263.5	\$251.02	\$5.51
Matemily 2,562 5 42.4 513.46 1.81 Certified Midwife 0 0 0.00 0.00 0.00 Prochiatic and Substance Abuse 33.256 756 6.442.0 43.87 22.55 Radiology and Pathology 17.834 795 6.756.4 22.43 12.63 Home Heath and Private Duty Nursing 59.935 8.449 71.805.3 6.99 41.81 Non-Emergency Transportation 59.956 4.874 41.422.5 12.30 42.46 Opioid Treatment Program 23.189 2.269 19.283.5 10.62 16.42 Federally Qualified and Rural Health Clinics 26.756 186 1.580.8 143.85 19.82 Autil Medical Equipment 19.226 3.168 26.92.3 6.07 10.73 Darable Medical Equipment 91.394 4.313 36.654.8 21.19 64.73 Single-Source Brand 32.5419 74.4 63.23.0 \$143.31 \$297.26 Cherric 20.902 2	Office Visits	32,675					55.38	23.14
Certified Midwife 0 0 0.0 0.00	Preventive Medicine	8,201		351		2,983.0	23.36	5.81
PT/OT/ST 4.483 215 1.827.2 20.85 3.17 Psychiatric and Substance Abuse 33.256 758 6.442.0 43.87 23.55 Radiology and Pathology 17.834 795 6.756.4 22.43 12.63 Home Health and Private Duty Nursing 59.035 8.449 71,805.3 6.99 41.81 Non-Emergency Transportation 59.956 4.874 41,422.5 12.30 42.46 Opiod Treatment Program 23.189 2.260 19.283.5 10.22 16.42 Federally Qualified and Rural Health Clinics 26.756 186 1.580.8 113.85 18.95 Adult Medical Day Care 10.728 515 4.376.8 20.83 7.60 Durable Medical Equipment 19.225 29.700 22.640.3 5.40 10.73 Durable Medical Equipment 91.384 4.313 36.654.8 21.19 64.73 Single-Source Brand 32.541.9 7.757 64.43.86.0 \$18.92 \$101.50 Single-Source Brand 23.594 60 509.9 399.89 16.99	Maternity	2,562		5		42.4	513.46	1.81
Psychiatric and Substance Abuse 33 256 758 6 442.0 43.87 22.55 Radiology and Pathology 17,834 795 6,756.4 22.43 12.63 home Healtn and Private Duty Nursing 59,035 8.449 71,805.3 6.99 41.81 Ambulance 7,497 373 3,170.0 20.10 5.31 Doid Treatment Program 23,189 2.269 19,283.5 10.22 16.42 Cederally Qualified and Rural Health Clinics 26,756 186 1,580.8 143.85 18.95 Adult Medical Day Care 10,728 515 4.376.8 20.864.3 5.40 Personal Care 0 0 0 0.00 0.00 0.00 Other 91,394 4.313 36.654.8 21.19 64.73 Perscription Drugs S 1413.7576 64.386.0 \$18.92 \$10.73 Single-Source Brand 325,419 744 63.23.0 437.39 23.047 Multi-Source Brand 23.994 60	Certified Midwife	0		0		0.0	0.00	0.00
Rediblogy and Pathology 17.834 795 6,756.4 22.43 12.63 Home Health and Private Duty Nursing 59,035 8,449 71.805.3 6.99 41.81 Ambulance 7,497 373 3,170.0 20.10 5.31 Non-Emergency Transportation 59,956 4.874 41,422.5 12.30 42.46 Opiold Treatment Program 22,869 19,283.5 10.22 16.42 Federally Qualified and Rural Health Clinics 26,756 186 1,580.8 143.85 18.95 Adult Medical Equipment 19,226 3,168 26,923.8 6.07 13.62 Applied Behavioral Analysis 0 0 0 0.00 0.00 0.00 Other 91,394 4,313 36,654.8 21.19 64,73 Single-Source Brand 325,419 74.44 63,230 437.39 230.47 Multi-Source Brand 325,419 74.4 63,230 437.39 230.47 Multi-Source Brand 23,90,49 60 <td< td=""><td>PT/OT/ST</td><td>4,483</td><td></td><td>215</td><td></td><td>1,827.2</td><td>20.85</td><td>3.17</td></td<>	PT/OT/ST	4,483		215		1,827.2	20.85	3.17
Home Health and Private Duty Nursing 50,035 8,449 71,805.3 6.99 41.81 Ambulance 7,497 373 3,170.0 20.10 5.31 Non-Emergency Transportation 56,956 4,874 41,422.5 12.30 42.46 Opioid Treatment Program 23,189 22,269 19,283.5 10.22 16.42 Federally Qualified and Rural Health Clinics 28,756 186 1,580.8 143.85 18.95 Adult Medical Day Care 10,728 515 4,376.8 20.83 7.60 Personal Care 15,151 2,808 23,864.3 5.40 10.73 Durable Medical Equipment 19,225 3,168 26,923.8 6.07 13.62 Applied Behavioral Analysis 0 0 0 0.00 0.00 0.00 Other 91.394 4.313 36,654.8 21.19 56.72 5 514.13 \$297.26 Prescription Drugs Generic Scripts \$14,33.11 7.576 64,386.0 \$18.92 \$1	Psychiatric and Substance Abuse	33,256		758		6,442.0	43.87	23.55
Ambulance 7,497 373 3,170.0 20.10 5.31 Non-Emergory Transportation 59,956 4,874 41,422.5 12.30 42.46 Opiol Treatment Program 23,189 2,269 19,283.5 10.22 16.42 Federally Qualified and Rural Health Clinics 26,756 186 1,580.8 143.85 18.95 Adult Medical Day Care 10,728 515 4,376.8 20.83 7.60 Personal Care 15,151 2,808 23,864.3 5.40 10.73 Durable Medical Equipment 19,226 3,168 26,923.8 6.07 13.62 Applied Behavioral Analysis 0 0 0.0 0.00 0.00 Other 91,394 4,313 36.654.8 21.19 64.73 Generic Scripts \$143,311 7,576 64.386.0 \$18.92 \$101.50 Single-Source Brand 325.419 74.44 6.323.0 437.39 230.47 Multi-Source Brand 357.99,006 8,463 71.924.3	Radiology and Pathology	17,834		795		6,756.4	22.43	12.63
Non-Emergency Transportation 59,956 4,874 41,422.5 12.30 42,46 Opioid Treatment Program 23,189 2,269 19,283.5 10,22 16,42 Federally Qualified and Rural Health Clinics 26,756 186 1,580.8 143.85 18.95 Adult Medical Day Care 10,728 515 4,376.8 20.83 5.40 10.73 Durable Medical Equipment 19,226 3,168 26,923.8 6.07 13.62 Applied Behavioral Analysis 0 0 0.0 0.00 0.00 Other 91,394 4,313 36,654.8 21.19 64.73 Prescription Drugs	Home Health and Private Duty Nursing					71,805.3		41.81
Opioid Treatment Program 23,189 2,269 19,283,5 10,22 16,42 Federally Qualified and Rural Health Clinics 26,756 186 1,580.8 143.85 18,95 Adult Medical Day Care 10,728 515 4,376.8 20.83 7.60 Personal Care 15,151 2,808 23,864.3 5.40 10.73 Durable Medical Equipment 19,226 3,168 26,923.8 6.07 13.62 Applied Behavioral Analysis 0 0 0.00 0.00 0.00 Other 91,394 4,313 36,654.8 21.19 64.73 Prescription Drugs	Ambulance							5.31
Federally Qualified and Rural Health Clinics 26,756 166 1,580.8 143.85 18,95 Adult Medical Day Care 10,728 515 4,376.8 20.83 7,60 Personal Care 15,151 2,808 22,864.3 5.40 10,73.62 Durable Medical Equipment 19,226 3,168 22,864.3 5.40 0.00 Applied Behavioral Analysis 0 0 0.00 0.00 0.00 0.00 Other 91,394 4,313 36,654.8 21.19 64,73 Prescription Drugs - - - - - Generic Scripts \$143,811 7,576 64,386.0 \$18.92 \$101.50 Single-Source Brand 325,419 744 6,323.0 437.39 230.47 Multi-Source Brand 20,994 60 509.9 399.89 16.99 Specialty 195,381 81 688.4 2,412.11 138.37 Hepattis C 0 0 0 0.00 0.00	Non-Emergency Transportation	59,956		4,874		41,422.5	12.30	42.46
Adult Medical Day Care 10,728 515 4,376.8 20.83 7.60 Personal Care 15,151 2,808 23,864.3 5.40 10,73.62 Durable Medical Equipment 19,226 3,168 26,923.8 6.07 13.62 Applied Behavioral Analysis 0 0 0 0.00 0.00 0.00 Other 91,394 4,313 36,654.8 21.19 64.73 Generic Scripts \$143,311 7,576 64,386.0 \$18.92 \$101.50 Single-Source Brand 325,419 744 6,323.0 437.33 230.47 Mult-Source Brand 23,994 60 509.9 399.89 16.99 Specialty 195,381 81 688.4 2,412.11 138.37 Hepatitis C 20.902 2 17.0 10,450.75 14.80 Other 0 0 0 0.00 0.00 0.00 Community Mental Health Center - - - - -	Opioid Treatment Program							
Personal Care 15,151 2,808 23,864.3 5.40 10.73 Durable Medical Equipment 19,226 3,168 26,923.8 6.07 13,62 Durable Medical Equipment 91,394 4,313 36,654.8 21.19 64,73 Perscription Drugs \$419,725 29,700 252,410.5 \$14.13 \$207.26 Prescription Drugs \$143,311 7,576 64,386.0 \$18.92 \$101.50 Single-Source Brand 325,419 744 6,323.0 437.39 230.47 Specialty 195,381 81 688.4 2,412.11 138.37 Hepatitis C 0 0 0.00 0.00 0.00 Other 0 0 0.00 0.00 0.00 0.00 Case Management \$307,296 791 6,722.5 \$388.49 \$217.63 Case Management \$307,296 791 6,722.5 \$388.49 \$217.63 Long Term Support Service 39,158 1,559 13,504.4 24.64	Federally Qualified and Rural Health Clinics							
Durable Medical Equipment Applied Behavioral Analysis 19,226 0 3,168 0 26,923.8 6.07 13.62 Applied Behavioral Analysis 0 0 0.0 0.00	Adult Medical Day Care	10,728		515		4,376.8		7.60
Applied Behavioral Analysis 0 0 0 0.0 0.00 0.00 Other 91,394 4,313 36,654.8 21.19 64.73 Prescription Drugs \$419,725 29,700 252,410.5 \$14.13 \$297.26 Prescription Drugs \$143,311 7,576 64,386.0 \$18.92 \$101.50 Single-Source Brand 325,419 744 6,323.0 437.39 230.47 Multi-Source Brand 23,994 60 509.9 399.89 16.99 Specialty 195,381 81 688.4 2,412.11 138.37 Hepatitis C 20,902 2 17.0 10,450.75 14.80 Other 0 0 0 0.00 0.00 0.00 0.00 Case Management \$307.296 791 6,722.5 \$388.49 \$217.63 Long Term Support Service 39,158 1,589 13,504.4 24.64 27.73 Partial Hospital 0 0 0 0.00	Personal Care							10.73
Other 91,394 4,313 36,654.8 21.19 64.73 Prescription Drugs \$419,725 29,700 252,410.5 \$14.13 \$297.26 Generic Scripts \$143,311 7,576 64,386.0 \$18.92 \$101.50 Single-Source Brand 325,419 744 6,323.0 437.39 230.47 Multi-Source Brand 23,994 60 509.9 399.89 16.99 Specially 195,381 81 688.4 2,412.11 138.37 Hepatitis C 20,902 2 17.0 10,450.75 14.80 Other 0 0 0.0 0.00 0.00 0.00 Other 0 0 0.0 0.0 0.0 0.0	Durable Medical Equipment	19,226		3,168				
\$419,725 29,700 252,410.5 \$14.13 \$297.26 Prescription Drugs Generic Scripts \$143,311 7,576 64,386.0 \$18.92 \$101.50 Single-Source Brand 325,419 744 6,323.0 437.39 230.47 Multi-Source Brand 23,994 60 509.9 399.89 16.99 Specialty 195,381 81 688.4 2,412.11 138.37 Hepatitis C 20,902 2 17.0 10,450.75 14.80 Other 0 0 0.00 0.00 0.00 Case Management \$307,296 791 6,722.5 \$388.49 \$217.63 Long Term Support Service 39,158 1,589 13,504.4 24.64 27.73 Partial Hospital 0 0 0 0.00 0.00 0.00 Psychotherapy 42,413 557 4,733.8 76.15 30.04 Evidence Based Practice 106 4 34.0 26.54 0.08 Re	Applied Behavioral Analysis							
Prescription Drugs s 143,311 7,576 64,386.0 \$18.92 \$101.50 Generic Scripts \$143,311 7,576 64,386.0 \$18.92 \$101.50 Single-Source Brand 325,419 744 6,323.0 437.39 230.47 Multi-Source Brand 23,994 60 509.9 399.89 16.99 Specialty 195,381 81 688.4 2,412.11 138.37 Hepatitis C 20,902 2 17.0 10,450.75 14.80 Other 0 0 0.0 0.00 0.00 Community Mental Health Center 71.924.3 \$83.78 \$502.13 Long Term Support Service 39,158 1,589 13,504.4 24.64 27.73 Partial Hospital 0 0 0.00	Other							
Generic Scripts \$143,311 7,576 64,386.0 \$18.92 \$101.50 Single-Source Brand 325,419 744 6,323.0 437.39 230.47 Multi-Source Brand 23,994 60 509.9 399.89 16.99 Specialty 195,381 81 688.4 2,412.11 138.37 Multi-Source Brand 20,902 2 17.0 10,450.75 14.80 Other 0 0 0.0 0.00 0.00 0.00 Community Mental Health Center \$307,296 791 6,722.5 \$388.49 \$217.63 Long Term Support Service 39,158 1,589 13,504.4 24.64 27.73 Partial Hospital 0 0 0 0.00 0.00 0.00 Psychotherapy 42,413 557 4,733.8 76.15 30.04 Evidence Based Practice 106 4 34.0 26.54 0.08 Medication Management 5,223 195 1,657.2 26.78		\$419,725		29,700		252,410.5	\$14.13	\$297.26
Single-Source Brand 325,419 744 6,323.0 437.39 230.47 Multi-Source Brand 23,994 60 509.9 399.89 16.99 Specialty 195,381 81 668.4 2,412.11 138.37 Hepatitis C 20,902 2 17.0 10,450.75 14.80 Other 0 0 0 0.00 0.00 0.00 Community Mental Health Center \$307,296 791 6,722.5 \$388.49 \$217.63 Long Term Support Service 39,158 1,588 13,504.4 24.64 27.73 Partial Hospital 0 0 0 0.00 0.00 Psychotherapy 42,413 557 4,733.8 76.15 30.04 Evidence Based Practice 106 4 340.0 26.78 3.70 Evidence Based Practice 106 4 340.0 26.78 3.70 Evidence Based Practice 106 4 340.0 26.78 3.70								
Multi-Source Brand 23,994 60 509.9 399.89 16.99 Specialty 195,381 81 688.4 2,412.11 138.37 Hepatitis C 20,902 2 17.0 10,450.75 14.80 Other 0 0 0 0.0 0.00 0.00 Community Mental Health Center Case Management \$307,296 791 6,722.5 \$388.49 \$217.63 Long Term Support Service 39,158 1,589 13,504.4 24.64 27.73 Partial Hospital 0 0 0 0.00 0.00 0.00 Psychotherapy 42,413 557 4,733.8 76.15 30.04 Evidence Based Practice 106 4 34.0 26.54 0.08 Medication Management 5,223 195 1,657.2 26.78 3.70 Emergency Service 24/7 4,551 194 1,648.7 23.46 3.222 APRTP 5,955 10 85.0 595.53								
Specialty 195,381 81 688.4 2,412.11 138.37 Hepatitis C 20,902 2 17.0 10,450.75 14.80 Other 0 0 0 0.0 0.00 0.00 \$709,006 8,463 71,924.3 \$83.78 \$502.13 Community Mental Health Center ************************************								
Hepatitis C 20,902 2 17.0 10,450.75 14.80 Other 0 0 0 0.0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.000 0.00 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>								
Other 0 0 0.0 0.00 0.00 \$709,006 8,463 71,924.3 \$83.78 \$502.13 Case Management \$307,296 791 6,722.5 \$388.78 \$217.63 Long Term Support Service 39,158 1,589 13,504.4 24.64 27.73 Partial Hospital 0 0 0 0.00 0.00 0.00 Psychotherapy 42,413 557 4,733.8 76.15 30.04 Evidence Based Practice 106 4 34.0 26.78 3.70 Medication Management 5,223 195 1,657.2 26.78 3.70 Emergency Service 24/7 4,551 194 1,648.7 23.46 3.22 APRTP 5,955 10 85.0 595.53 4.22 Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 37.31 28.14 Other 39,732								
\$709,006 8,463 71,924.3 \$83.78 \$502.13 Community Mental Health Center \$307,296 791 6,722.5 \$388.49 \$217.63 Long Term Support Service 39,158 1,589 13,504.4 24.64 27.73 Partial Hospital 0 0 0 0.0 0.00 0.00 Psychotherapy 42,413 557 4,733.8 76.15 30.04 Evidence Based Practice 106 4 34.0 26.54 0.08 Medication Management 5,223 195 1,657.2 26.78 3.70 Emergency Service 24/7 4,551 194 1,648.7 23.46 3.22 Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 73.31 28.14	Hepatitis C							
Community Mental Health Center \$307,296 791 6,722.5 \$388.49 \$217.63 Case Management \$307,296 39,158 1,589 13,504.4 24.64 27.73 Partial Hospital 0 0 0 0.0 0.00 0.00 0.00 Psychotherapy 42,413 557 4,733.8 76.15 30.04 Evidence Based Practice 106 4 34.0 26.54 0.08 Medication Management 5,223 195 1,657.2 26.78 3.70 Emergency Service 24/7 4,551 194 1,648.7 23.46 3.22 APRTP 5,955 10 85.0 595.53 4.22 Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 73.31 28.14 \$451,736 4,156 35,320.5 \$108.69 \$319.93	Other							
Case Management \$307,296 791 6,722.5 \$388.49 \$217.63 Long Term Support Service 39,158 1,589 13,504.4 24.64 27.73 Partial Hospital 0 0 0 0.0 0.00 0.00 Psychotherapy 42,413 557 4,733.8 76.15 30.04 Evidence Based Practice 106 4 34.0 26.54 0.08 Medication Management 5,223 195 1,657.2 26.78 3.70 Emergency Service 24/7 4,551 194 1,648.7 23.46 3.222 APRTP 5,955 100 85.0 595.53 4.22 Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 73.31 28.14 Medication \$451,736 4,156 35,320.5 \$108.69 \$319.93		\$709,006		8,463		71,924.3	\$83.78	\$502.13
Long Term Support Service 39,158 1,589 13,504.4 24.64 27.73 Partial Hospital 0 0 0 0.00 0.00 0.00 0.00 Psychotherapy 42,413 557 4,733.8 76.15 30.04 Evidence Based Practice 106 4 34.0 26.54 0.08 Medication Management 5,223 195 1,657.2 26.78 3.70 Emergency Service 24/7 4,551 194 1,648.7 23.46 3.22 APRTP 5,955 10 85.0 595.53 4.22 Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 7.3.31 28.14 * \$451,736 4,156 35,320.5 \$10.869 \$319.93	-							
Partial Hospital 0 0 0 0.0 0.00 0.00 0.00 Psychotherapy 42,413 557 4,733.8 76.15 30.04 Evidence Based Practice 106 4 34.0 26.678 3.70 Medication Management 5,223 195 1,657.2 26.78 3.70 Emergency Service 24/7 4,551 194 1,648.7 23.46 3.22 APRTP 5,955 10 85.0 595.53 4.22 Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 73.31 28.14 **** \$451,736 4,156 35,320.5 \$108.69 \$319.93								
Psychotherapy 42,413 557 4,733.8 76.15 30.04 Evidence Based Practice 106 4 34.0 26.54 0.08 Medication Management 5,223 195 1,657.2 26.78 3.70 Emergency Service 24/7 4,551 194 1,648.7 23.46 3.22 APRTP 5,955 10 85.0 595.53 4.22 Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 73.31 28.14 The second sec	Long Term Support Service							
Evidence Based Practice 106 4 34.0 26.54 0.08 Medication Management 5,223 195 1,657.2 26.78 3.70 Emergency Service 24/7 4,551 194 1,648.7 23.46 3.22 APRTP 5,955 10 85.0 595.53 4.22 Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 73.31 28.14	Partial Hospital							
Medication Management 5,223 195 1,657.2 26.78 3.70 Emergency Service 24/7 4,551 194 1,648.7 23.46 3.22 APRTP 5,955 10 85.0 595.53 4.22 Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 73.31 28.14								
Emergency Service 24/7 4,551 194 1,648.7 23.46 3.22 APRTP 5,955 10 85.0 595.53 4.22 Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 73.31 28.14 \$\$451,736 4,156 35,320.5 \$108.69 \$319.93								
APRTP 5,955 10 85.0 595.53 4.22 Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 73.31 28.14 \$451,736 4,156 35,320.5 \$108.69 \$319.93								
Supported Employment Services 7,301 274 2,328.6 26.65 5.17 Other 39,732 542 4,606.3 73.31 28.14 \$451,736 4,156 35,320.5 \$108.69 \$319.93								
Other 39,732 542 4,606.3 73.31 28.14 \$451,736 4,156 35,320.5 \$108.69 \$319.93								
\$451,736 4,156 35,320.5 \$108.69 \$319.93	Supported Employment Services							
	Other					,		
All Services \$1,854,360 17 47,630 144.5 404,791.8 \$38.93 \$1,313.30								
	All Services	\$1,854,360	17	47,630	144.5	404,791.8	\$38.93	\$1,313.30

Eligibility Category: Low Utilizer - Dual

Member Months: 2,242

							Per Canita	
Dec. #4	Total Paid	Total Paid	Total Paid	Admits Per	Utilization Per	Average Paid	Per Capita Monthly Paid	
Benefits	Dollars	Admits	Services	1,000	1,000	Charge	Cost	
Hospital Inpatient Medical	¢40.000	20	237	107.0	1 060 0	\$2 140.00	\$19.10	
	\$42,820	20 5	60	26.8	1,268.3 321.1	\$2,140.98 1,602.70	3.57	
Surgical Maternity Delivery	8,014 0	5	0	20.0	0.0	0.00	0.00	
Maternity Non-Delivery	0	0	0	0.0	0.0	0.00	0.00	
Well Newborn	0	0	0	0.0	0.0	0.00	0.00	
	2,999	3	40	16.1	214.1	999.77	1.34	
Psychiatric Alcohol and Drug Abuse	1,364		40	5.4	16.1	1,364.00	0.61	
Other	0	0	0	0.0	0.0	0.00	0.01	
Other	\$55,197	29	340	155.2	1,819.5	\$1,903.33	\$24.62	
Hospital Outpatient	ψ00,197	25	540	100.2	1,019.5	\$1,903.33	ψ24.02	
Emergency Room	\$37,542		112		599.4	\$335.20	\$16.74	
Surgery	17,371		76		406.7	228.56	7.75	
Radiology	14,658		194		1,038.2	75.56	6.54	
Pathology	2,869		646		3,457.1	4.44	1.28	
Pharmacy	11,445		4,426		23,685.8	2.59	5.10	
Cardiovascular	3,207		50		267.6	64.14	1.43	
PT/OT/ST	4,592		586		3,136.0	7.84	2.05	
Psychiatric	1,699		33		176.6	51.49	0.76	
Substance Abuse	0		0		0.0	0.00	0.00	
Other	34,757		2,292		12,265.7	15.16	15.50	
	\$128,141		8,415		45,033.0	\$15.23	\$57.15	
Professional and Other State Plan Services	\$120,111		0,110		10,000.0	\$10.20	<i>\\</i>	
Ambulatory Surgery Center	\$977		7		37.5	\$139.64	\$0.44	
Office Visits	26,318		1,242		6,646.6	21.19	11.74	
Preventive Medicine	3,873		374		2.001.5	10.36	1.73	
Maternity	97		1		5.4	97.28	0.04	
Certified Midwife	0		0		0.0	0.00	0.00	
PT/OT/ST	1,844		310		1.659.0	5.95	0.82	
Psychiatric and Substance Abuse	18,304		478		2,558.0	38.29	8.16	
Radiology and Pathology	5,688		534		2,857.7	10.65	2.54	
Home Health and Private Duty Nursing	16,450		2,628		14.063.8	6.26	7.34	
Ambulance	5,570		400		2,140.6	13.92	2.48	
Non-Emergency Transportation	66,059		3,293		17,622.5	20.06	29.46	
Opioid Treatment Program	10,578		1,035		5,538.8	10.22	4.72	
Federally Qualified and Rural Health Clinics	1,939		15		80.3	129.28	0.86	
Adult Medical Day Care	1,039		95		508.4	10.94	0.46	
Personal Care	3,194		592		3,168.1	5.40	1.42	
Durable Medical Equipment	35,499		25,273		135,248.7	1.40	15.83	
Applied Behavioral Analysis	0		0		0.0	0.00	0.00	
Other	54,932		11,717		62,703.7	4.69	24.50	
	\$252,361		47,994		256,840.4	\$5.26	\$112.54	
Prescription Drugs								
Generic Scripts	\$2,498		824		4,409.6	\$3.03	\$1.11	
Single-Source Brand	1,084		16		85.6	67.75	0.48	
Multi-Source Brand	5		1		5.4	5.31	0.00	
Specialty	0		0		0.0	0.00	0.00	
Hepatitis C	0		0		0.0	0.00	0.00	
Other	1		1		5.4	1.15	0.00	
	\$3,589		842		4,506.0	\$4.26	\$1.60	
Community Mental Health Center								
Case Management	\$420,346		1,082		5,790.3	\$388.49	\$187.46	
Long Term Support Service	49,924		1,992		10,660.2	25.06	22.26	
Partial Hospital	2,276		21		112.4	108.37	1.01	
Psychotherapy	31,632		781		4,179.5	40.50	14.11	
Evidence Based Practice	239		9		48.2	26.54	0.11	
Medication Management	8,546		303		1,621.5	28.21	3.81	
Emergency Service 24/7	68		9		48.2	7.50	0.03	
APRTP	6,551		11		58.9	595.53	2.92	
Supported Employment Services	39,162		1,360		7,278.1	28.80	17.46	
Other	25,134		805		4,308.0	31.22	11.21	
	\$583,877		6,373		34,105.2	\$91.62	\$260.39	
All Services	\$1,023,164	29	63,964	155.2	342,304.1	\$16.00	\$456.29	

Eligibility Category: Serious Emotionally Disturbed Child

Member Months: 75,719

Benefits Total Paid Total Paid Advants Put Capital Services Multiply Paid Cost Heghti Inpationt 5295.53 66 200 100 45.5 49.47.72 53.30 Burgical 122.529 21 71 3.3 1.13 9.160.00 2.5 Burgical 122.529 21 71 3.3 1.13 9.160.00 2.00 0.00 <t< th=""><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>Dan O</th></t<>								Dan O
Benefits Dollars Admits Services 1.000 0.000 Origin Cost Medical 5295,035 66 229 10.8 4.43.4 53.55 53.55 Medical 5295,035 66 229 1 3.0 1.0.8 4.44.75 53.55 Materinity Non-Cellevery 3.229 1 1 0.0 0.0 0.00		Total Paid	Total Paid	Total Paid	Admite Por	Iltilization Por	Average Paid	Per Capita Monthly Paid
Hesplating 2295,638 68 299 10.8 45.8 54.347.62 53.90 Surgical 192,529 21 71 3.3 11.3 9,168.00 2.8 Wall Needom 3.42 0 0 0.0	Benefits							
Medical \$226.528 68 299 10.8 44.8 54.47.82 53.30 Maternity Non-Delivery 3.42 0 0 0.0 0.00		Donars	Adinto	00111003	1,000	1,000	onarge	0031
Surgical 192,529 21 71 3.3 11.3 9,686.06 2.24 Maternity Delvery 3,429 1 1 0.2 3,4262 0.05 Wat Neurolom 3,472,0 0 0 0.0 0.0 0.0 0.0 0.00 Psychatin 3,472,0 440 4.27 66.7 67.8 8,0002,03 446.80 Other 0 0 0 0.0 0.0 0.0 0.00	· · ·	\$295.638	68	289	10.8	45.8	\$4.347.62	\$3.90
Materinty Non-Delivery 0								
Weil Newborn 0 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td>· · ·</td><td></td></t<>							· · ·	
Psychiatric 3.847.203 4400 4.279 0.87 678.1 8.062.03 44.855 Other 0 0 0 0 0 0.00 0.00 Standball 0.00 0.00 0.00 0.00 0.00 0.00 Happtal Dutpation 51.812.603 3.544 651 7.552 651 7.562 Inengenry Room 22.997 22.997 1.575 4.30 2.440 0.005 2.430 Reidery 23.997 24.997 2.456 1.575 2.430 1.401 2.282 1.24 2.44 Prathology 18.4520 14.410 2.286 1.223 1.281 2.430 Protescalar 21.091 1.550 2.2606 3.33 1.264 0.71 Stotance Abuse 0 0 0.00 0.00 0.00 0.00 0.00 0.00 Other 55.07 2.2600.33 2.2600.33 2.27 3.244 2.465 3.27.072 4.2	Maternity Non-Delivery	3,429	1	1	0.2	0.2	3,428.62	0.05
Psychiatric 3,47/29 440 4.279 69.7 678.1 8.062.03 46.85 Achol and Dup Abuse 49.008 7 11 11 18.8 7.00 0.00 Hapital Outpatient 51 753 47.757 651 7538 57.61 2.47 5538 Inergency Noron \$1.812.030 3.594 550.6 544.867 521.30 Reidsley 226.634 1.575 651 722.48 120.67 4.30 Reidsley 226.64 1.575 240.6 224.84 2.44 0.71 Reidsley 53.37 20.267 3.31.19 2.24 2.44 0.71 Cardiovascular 21.091 150 2.33.91 2.35 2.35 2.33.90 2.35 Professional and Other State Plan Services 551.070 0.0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 <th< td=""><td>Well Newborn</td><td>0</td><td>0</td><td>0</td><td>0.0</td><td>0.0</td><td>0.00</td><td>0.00</td></th<>	Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Alcohol and Dug Abuse 440.08 7 117 1.1 18.5 7.001.20 0.65 Other 9 0.0 0.0 0.0 0.0 0.00 0.00 Hespital Outpatient 537 4.767 851 773.3 87.612.47 853.39 Emergency Room S1.612.608 3.894 4.68 1.020.56 3.81 Reinburgy 234.347 2.281 4.615 7.601.26 3.81 Planmacy 234.347 2.2067 2.211.3 2.244 0.71 Cardiovasolar 21.091 160 2.38 140.61 0.28 Profusion 177.814 5.245 8.81.2 3.809 2.29 Substance Abuse 0 0 0.0 0.00 0.00 0.00 Other 55.457 9.608.3 140.3 2.27 9.33.3 7.28 Destation & Substance Abuse 55.454 11.73.3 12.33 12.33 12.33 12.33 12.33 12.33 12.33 <td></td> <td>3,547,293</td> <td>440</td> <td>4,279</td> <td></td> <td></td> <td></td> <td></td>		3,547,293	440	4,279				
Other 0 0 0 0.0 0.00 0.00 Hepital Outpatient 537 47.77 55.1 75.3 57.812.47 55.3 Bernegency, Room \$1.812.608 3.984 568.44 568.4 523.3 Surgeny 234.447 289 4.88 1.020.56 3.30 Bernegency, Room 53.447 20.87 2.222.7 2.33.14 2.44 Pharmacy 53.437 20.267 2.33.8 140.61 0.28 Cardiovascular 72.091 156 2.33.9 2.255 2.55 Pychistic 55.1079 177.84 5.245 2.800.4 3.33.7 2.84 Profestical Atane 0 0 0.00 <td< td=""><td></td><td></td><td></td><td>,</td><td></td><td></td><td></td><td></td></td<>				,				
94.087.987 537 4.757 85.1 753.9 \$7,612.47 \$53.99 Emergency Room \$1,612.008 3.694 5606 \$448.69 \$21.30 Radiology 325.534 1,575 246.67 4.30 Pathology 184.520 14.401 2.262.3 12.81 2.44 Pharmacylater 51.47 20.0 3.21.1.8 2.66 3.31.2 13.30 2.25 PriortiST 177.814 5.645 53.1.2 13.30 2.26 12.44 13.30 2.26 12.93 12.84 2.44 0.4 0.0 0.0 0.00			0				· · ·	
Hospital Outpatient Emergency Room \$1,612,608 3.594 5686 544.869 52.130 Surgery 224,947 289 45.8 1,020,58 3.30 Pathology 325,534 1,575 246.6 20.675 4.30 Pathology 184,560 14.401 2.282.3 12.81 2.44 Cardiovascular 21,091 15.60 23.8 14.061 0.28 Cardiovascular 21,091 15.63 2.282.3 14.061 0.26 Systeme Ause 0 0 0 0.00 0.00 0.00 Other 551.079 16.538 2.282.6 3.33 7.28 Arnotatory Surgery Center 53.379.091 62.292 9.900.3 554.04 54.45 Arnotatory Surgery Center 33.42.2 51.079 13.58 1.783 2.277.8 53.84 1.370 Preventive Medicine 324.546 11.893 1.384.8 27.79 4.63 DriOtTy T 380.022 14.376 2.274.5 <td></td> <td>\$4,087,897</td> <td>537</td> <td>4,757</td> <td></td> <td></td> <td></td> <td></td>		\$4,087,897	537	4,757				
Surgery 224,947 228 45.8 1.020.58 3.90 Pathology 325,634 1,757 249.6 2026.75 4.30 Pathology 184,520 14.401 2.282.3 12.81 2.44 Cardiowascular 21.091 150 2.3.8 140.61 0.2.8 Diotascular 177.814 5.245 831.2 3.3.80 2.0.8 Substance Abuse 551.077 16.33 2.620.6 0.33.3 7.28 Other 551.077 16.33 2.620.6 0.33.3 7.28 Professional and Other State Plan Services 554.04 144.3 2.2.7 S384.22 9.07.3 Professional and Other State Plan Services 554.04 14.93 2.2.7 S384.22 9.07.3 Professional and Other State Plan Services 54.043 144.3 2.2.7 S384.22 9.07.3 Professional and Other State Plan Services 54.043 1.93.9 2.2.97.3 2.3.91 0.0.6 Cartileed Midwile 18.2 2.0 <t< td=""><td>Hospital Outpatient</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	Hospital Outpatient							
Radiology 226,634 1,775 249.6 208.75 4.30 Pharmacy 53,437 20,267 3,211.9 2.64 0.71 Cardiovascular 21,091 150 2.38 144.61 0.282.3 Pri/OT/ST 177,814 5,245 831.2 33.90 2.35 Substance Abuse 0 0 0.0 0.00 0.00 0.00 Other 551.079 16,556 2.2620 3.33 1.28 4.44 Professional and Other State Plan Services 54,943 443 2.27 3.84.22 3.84.22 3.73 Cristics Michaion Singery Center 1.98,989 143.3 2.978.3 S.34.22 3.93 1.42 4.00 0.6 4.547 0.00 1.00 0.00 1.00 0.00 1.00 0.00 1.00 0.0 1.03 1.284.84 2.978.3 S.34.22 3.33 1.281 1.668 2.578 3.33 1.281 0.6 4.471 0.0 1.00 1.00 1.00	Emergency Room	\$1,612,608		3,594		569.6	\$448.69	\$21.30
Pathology 184,620 14.401 2.282.3 12.81 2.44 Pharmacy 53.437 20.827 3.211.9 2.64 0.71 Cardiovascular 21.091 150 2.3.8 140.61 0.2.85 Psychiatric 177.814 5.245 6.812.2 33.90 2.0.9 Substance Abuse 0 0 0.00 0.00 0.00 0.00 Other 551.079 16.536 2.620.6 3.3.3 7.8 Professional and Other State Plan Services 551.941 454.463 7.33 2.978.3 551.81 13.70 Professional and Other State Plan Services 4 2.27 5.384.22 50.73 Office Visits 1.308.989 18.733 2.978.3 551.81 13.70 Preventive Medicine 324.546 11.803 1.84.8 2.22.33 2.17 5.16 Preventive Medicine 30.802 14.85 2.27.05 3.23.8 10.05 Preventive Medicine 324.5443 31653	Surgery	294,947		289		45.8	1,020.58	3.90
Pharmacy Cardiovascular 53,437 21,091 20,267 150 3,211.9 2.64 0.71 Cardiovascular 21,091 150 23.8 144.661 0.28 PT/C17ST 177,814 5,245 831.2 33.80 2.35 Substance Abuse 0 0 0 0.0 0.00 0.00 Other 551.079 16,356 2,6206 33.33 7.28 Arbulatory Surgery Center 54,943 143 2.27 \$38.42 \$3.73 Preventive Medicine 324,546 11,1933 1,844.8 2.29 4.29 Maternity 4,852 2.0 3.2 2.38.91 0.06 Cartified Midwife 182 4 0.6 44.47 0.00 Cartified Midwife 182 4 0.6 44.47 0.00 Psychiatric and Substance Abuse 346,007 7.135 1,130.8 44.7 4.00 Pacidociay and Pathology 191.717 12.22 1.938.1 15.68 2.53	Radiology	325,634		1,575		249.6	206.75	4.30
Cardiovscular 21,091 150 23.8 140,61 0.28 Piythatric 157,961 470 74.5 33.80,09 2.09 Substance Abuse 0 0 0,00 0.00 0.00 0.00 Other 551,079 16,556 2,202,6 33.33 7.28 Professional and Other State Plan Services 33.370,001 6,257 9,900,3 554,04 54,44 Ambulatory Supery Center 554,943 143 2,277,3 551,8 13.70 Preventive Medicine 324,546 11,893 1,884,8 27,29 4,29 Maternity 4,852 20 3,2 238,91 0.06 Certified Midvife 182 4 0,6 4,47 0.00 Pri/DT/ST 390,022 14,376 2,276,3 27,13 5,15 Psychiatric and Substance Abuse 348,007 7,135 1,130,8 48,77 46,0 Ambulance 181,971 14,018 2,221,6 12,98 2,40	Pathology	184,520		14,401		2,282.3	12.81	2.44
PTIOTIST 177.814 5.245 833.12 33.90 2.35 Substance Abuse 0 0 0.0 0.00 0.00 0.00 0.00 Substance Abuse 0 0 0.0 0.00 <td>Pharmacy</td> <td>53,437</td> <td></td> <td>20,267</td> <td></td> <td>3,211.9</td> <td>2.64</td> <td>0.71</td>	Pharmacy	53,437		20,267		3,211.9	2.64	0.71
Psychiatric 157,961 470 74,5 338,09 2.09 Other 551,079 16,536 2,202,6 33,33 7.28 Professional and Other State Plan Services 62,527 9,909,3 554,04 54,44 Ambulatory Supery Center 554,94,9 14,37 2,277,3 551,27 4,292 Professional and Other State Plan Services 13,370,091 6,252 9,209,3 551,61 13,70 Preventive Medicine 324,546 11,893 1,284,8 22,9 4,29 Maternity 4,852 20 3,2 238,91 0.06 Certified Midwife 182 4 0,6 4,47 0.00 ProfVirJST 390,022 14,376 2,277,3 27,13 5,15 Psychiatric and Substance Abuse 349,007 7,135 1,130,8 48,77 4,60 Ambulace 143,971 14,229 1,338,1 15,68 2,53 More Healt and Private Duity Nursing 24,34 3,166,5 5,116,2 2,64	Cardiovascular	21,091		150		23.8	140.61	0.28
Substance Abuse 0 0 0.0 0.00 0.00 Other 551/079 16,556 2,620,6 33.33 7.28 Professional and Other State Plan Services 83,379,091 66,2527 9,909,3 \$54,043 \$44,63 Ambulatory Surgery Center \$54,943 11,38 2,278 \$55,161 113,70 Preventive Medicine 324,546 11,893 1,848,8 27,28 42,99 Oritor Visits 182 4 0,6 45,47 0,00 Preventive Medicine 182 4 0,6 45,47 0,00 Profestional Pathology 191,717 12,229 1,838,1 15,66 2,273 Profestional Protocy 191,717 12,229 1,538,1 15,66 2,234 Non-Emregency Transportation 215,301 9,664 7,615 2,400 Non-Emregency Transportation 215,301 9,644 1,515,7 2,246 Opicid Traitment Program 0 0 0 0,00 0,00	PT/OT/ST	177,814		5,245		831.2	33.90	2.35
Other 551.079 16.536 2.620.6 33.33 728 Professional and Other State Plan Services S3.379.091 62.527 9.909.3 \$54.04 \$44.83 Arbulatory Surgery Center S54.943 14.3 2.27 \$38.422 \$0.73 Office Visits 1.036.989 18.703 2.978.3 55.18 \$0.73 Preventive Medicine 324.546 11.893 1.864.8 27.29 14.29 Maternity 4.852 20 3.2 2.83.91 0.06 Certified Midwle 1.852 4 0.6 45.47 0.00 Pi(107)ST 3.80.022 14.376 2.278.3 2.71.3 5.15 Psychiatric and Substance Abuse 346.007 7.135 1.130.8 48.77 4.60 Non-Emergency Transportation 215.301 9.664 1.515.7 2.251 2.84 Opcid Treatment Program 0 0 0.00 0.00 0.00 0.00 Outradised acting Qualified and Rural Heath Clinics 705.568 <t< td=""><td>Psychiatric</td><td>157,961</td><td></td><td>470</td><td></td><td>74.5</td><td>336.09</td><td>2.09</td></t<>	Psychiatric	157,961		470		74.5	336.09	2.09
Fordesional and Other State Plan Services 554,943 143 22.7 5384,22 55.7 Ambulatory Surgery Center 554,943 143 2.27 \$384,22 50.73 Office Visits 1,036,889 18,793 2.973,3 55.18 13.70 Preventive Medicine 324,546 11,893 1,884,8 27.29 4.29 Ambulatory Surgery Center 4852 20 3.2 238,91 0.06 Certified Midwife 182 4 0.6 45.47 0.00 Proyenitaria and Substance Abuse 348,007 7.135 1,130.8 48.77 4.60 Proyenitaria and Substance Abuse 348,007 7.135 1,130.8 47.71 3.10 Ambulance 181,971 14.018 2.221.6 12.28 2.40 Non-Emergency Transportation 215,301 9.564 0.0 0.0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00	Substance Abuse	0		0		0.0	0.00	0.00
Professional and Other State Plan Services Ambulatory Surgery Center 554,943 143 227 Sask 22 Sorvard Office Visits 1,036,869 18,793 2,378.3 56.18 13.70 Office Visits 324,546 11,893 1,884.8 27.29 4.29 Matemity 4,852 2 0.6 45.47 0.00 Orfice Visits 390,022 14,376 2,278.3 27.13 5.15 Pacholisy and Pathology 191,177 12,229 1,384.1 15.68 2.53 Home Health and Private Duty Nursing 234,643 31,653 5.016.4 7.41 3.10 Non-Emergency Transportation 215,301 9,564 1,515.7 2.28 2.40 Opicid Treatment Program 0 0 0 0.000 0.000 0.00 Outrable decidal Equipment 419,646 188,943 29,943.8 2.22 5.13.22 9.32 Aptiled Behavioral Analysis 213,334 14,611 2,315.6 14.60 2.89 Other <td>Other</td> <td>551,079</td> <td></td> <td>16,536</td> <td></td> <td>2,620.6</td> <td>33.33</td> <td>7.28</td>	Other	551,079		16,536		2,620.6	33.33	7.28
Ambulatory Surgery Center \$54,943 143 22.7 \$384,22 50.73 Office Visits 1.086,8989 18,793 2.278.3 55.18 13.70 Preventive Medicine 324,546 11,893 1,884.8 27.29 4.29 Maternity 4,852 20 3.2 238.91 0.06 Certified Midwife 182 4 0.6 45.47 0.00 Pri/OT/ST 390,022 11,376 1,310.8 48.77 4.60 Radiology and Pathology 191.717 12.229 1,338.1 15.68 2.53 Ambulance 181.971 14,018 2,221.6 12.98 2.40 Opicid Treatment Program 0 0 0.0 0.00 0.00 0.00 Opresonal Care 0 0 0.0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 <td< td=""><td></td><td>\$3,379,091</td><td></td><td>62,527</td><td></td><td>9,909.3</td><td>\$54.04</td><td>\$44.63</td></td<>		\$3,379,091		62,527		9,909.3	\$54.04	\$44.63
Office Visits 1036.989 18,793 2.978.3 55.18 13.70 Preventive Medicine 324,546 11,803 1,884.8 27.29 4.29 Maternity 4,852 20 3.2 238.91 0.06 Carified Midwife 162 4 0.6 45.47 0.00 PriOT/ST 390.022 14.376 2.278.3 55.18 1.30.8 PriOT/ST 390.022 14.376 2.278.3 55.18 4.60 Radiology and Pathology 191,717 12.229 1.938.1 15.68 2.53 Home Health and Private Duty Nursing 234,643 31.653 5.016.4 7.41 3.10 Ambulance 181,971 14.018 2.221.6 12.84 2.94 2.94 2.94 2.94 2.94 2.94 2.94 2.94 2.94 2.94 2.94 2.94 2.94 2.94 2.94 3.10 2.94 3.10 2.94 3.10 2.94 3.10 2.94 3.10 2.94	Professional and Other State Plan Services							
Preventive Medicine 1224,546 11,893 1,884.8 27.29 4.29 Maternity 4.852 20 3.2 238.91 0.06 Cartified Midwife 182 4 0.6 46.47 0.00 Pri/DT/ST 390,022 14,376 2.278.3 27.13 5.15 Psychiatire and Substance Abuse 344,007 7.135 1.130.8 48.77 4.60 Nome Heath and Private Duty Nursing 234.643 31.653 5.016.4 7.41 3.10 Ambulance 181,971 14,018 2.221.6 12.98 2.40 Non-Emergency Transportation 215.501 9.564 1.515.7 22.51 2.84 Opiod Treatment Program 0 0 0.00 0.00 0.00 Personal Care 0 0 0.00 0.00 0.00 0.00 Durable Medical Equipment 419.646 188,943 29.943.8 2.22 5.54 Other 1.258,533 60.595 12.772.8 16.62 <	Ambulatory Surgery Center	\$54,943		143		22.7	\$384.22	\$0.73
Matemity 4.852 20 3.2 238.91 0.06 Certified Midrife 182 4 0.6 45.47 0.00 PY/OTST 390.022 14.376 2.278.3 27.13 5.15 Psychiatric and Substance Abuse 346.007 7.135 1.130.8 48.77 460 Psychiatric and Substance Abuse 346.007 7.135 1.130.8 48.77 460 Ambulance 191,717 12.229 1.338.1 15.68 2.53 Home Health and Private Duty Nursing 234.643 31.653 5.016.4 7.41 3.10 Non-Emergency Transportation 215.301 9.564 1.515.7 22.51 2.84 Adult Medical Day Care 0 0 0.0 0.00 0.00 0.00 Durable Medical Equipment 419.646 188.943 2.94.38 2.22 5.54 Applied Behavioral Analysis 213.34 14.611 2.315.6 14.66 2.82 Other 1.258.53 30.655 12.777.2	Office Visits	1,036,989		18,793		2,978.3	55.18	13.70
Certified Midwife 182 4 0.6 45.47 0.00 PT/OT/ST 300.022 14.376 2.278.3 27.13 5.15 Psychiatric and Substance Abuse 346.007 7.135 1.130.8 48.77 4.60 Radiology and Pathology 191,717 7.229 1.938.1 15.68 2.53 Ambulance 181,971 14.018 2.221.6 12.98 2.40 Ambulance 181,971 14.018 2.221.6 12.98 2.40 Ono-Emergency Transportation 215.301 9.564 7.155 2.84 4000 0.0 0.00	Preventive Medicine	324,546		11,893		1,884.8	27.29	4.29
PT/OT/ST 390,022 14,376 2.278.3 27.13 5.15 Psychiatric and Substance Abuse 348,007 7,135 1,130.8 48,77 4.60 Radiology and Pathology 191,717 12,229 1,938.1 15.66 2.53 Home Health and Private Duty Nursing 234,643 31,653 5.016.4 7.41 3.10 Non-Emergency Transportation 215,301 9,564 1,515.7 22.51 2.84 Opiol Treatment Program 0 0 0.00 0.000 0.000 Colduit Medical Day Care 0 0 0.00 0.00 0.00 0.00 Qualified and Rural Health Clinics 705,568 44055 729.8 153.22 9.32 Autit Medical Equipment 419,646 188,943 29.943.8 2.222.5 5.54 Applied Behavioral Analysis 213.334 14,611 2.315.6 14.60 2.82 Other 1,285.633 0.955 12.772.8 15.58.9 53.072.0 Prescription Drugs 5	Maternity	4,852		20		3.2	238.91	0.06
Psychiatric and Substance Abuse 348,007 7,15 1,130.8 48,77 4,60 Radiology and Pathology 191,717 12,229 1,938.1 15,68 2,53 home Health and Private Duty Nursing 234,643 31,653 5,016.4 7,41 3,10 Ambulance 181,971 14,018 2,221.6 12,98 2,40 Non-Emergency Transportation 215,301 9,564 1,515.7 22.21 2,40 Opiod Treatment Program 0 0 0,00 <td>Certified Midwife</td> <td>182</td> <td></td> <td>4</td> <td></td> <td>0.6</td> <td>45.47</td> <td>0.00</td>	Certified Midwife	182		4		0.6	45.47	0.00
Radiology and Pathology 191,717 12,229 1,938.1 15.68 2.53 Home Health and Private Duty Nursing 234,643 31,653 5,016.4 7,41 3.10 Ambulance 181,971 14,018 2,221.6 12.98 2.40 Non-Emergency Transportation 215,301 9,964 1,515.7 2.2.51 2.84 Opicid Treatment Program 0 0 0.01 5.558/58/52 <td>PT/OT/ST</td> <td>390,022</td> <td></td> <td>14,376</td> <td></td> <td>2,278.3</td> <td>27.13</td> <td>5.15</td>	PT/OT/ST	390,022		14,376		2,278.3	27.13	5.15
Home Health and Private Duty Nursing 234,643 31,653 5,016.4 7,41 3.10 Ambulance 181,971 14,018 2,221.6 12.98 2.40 Non-Emergency Transportation 215,301 9,664 1,515.7 22.51 2.84 Opiol Treatment Program 0 0 0.0 0.00 <t< td=""><td>Psychiatric and Substance Abuse</td><td>348,007</td><td></td><td>7,135</td><td></td><td>1,130.8</td><td>48.77</td><td>4.60</td></t<>	Psychiatric and Substance Abuse	348,007		7,135		1,130.8	48.77	4.60
Ambulance 181,971 14,018 2,221.6 12.98 2.40 Non-Emergency Transportation 215,301 9,564 1,515.7 22.51 2.84 Non-Emergency Transportation 0 0 0.00 <td>Radiology and Pathology</td> <td>191,717</td> <td></td> <td>12,229</td> <td></td> <td>1,938.1</td> <td>15.68</td> <td>2.53</td>	Radiology and Pathology	191,717		12,229		1,938.1	15.68	2.53
Non-Emergency Transportation 215,301 9,564 1,515.7 22.51 2.84 Opioid Treatment Program 0 0 0 0.0	Home Health and Private Duty Nursing	234,643		31,653		5,016.4	7.41	3.10
Opioid Treatment Program 0 0 0 0.0 0.00 0.00 0.00 Federally Qualified and Rural Health Clinics 705,568 4,605 729,8 153,22 9,32 Adult Medical Day Care 0 0 0.0 0.00 0.00 0.00 Durable Medical Equipment 419,646 188,943 229,943.8 2.22 5,54 Applied Behavioral Analysis 213,334 14,611 2,315,6 14,60 2.82 Other 1,258,533 80,595 12,772,8 15,62 16,62 Prescription Drugs	Ambulance	181,971		14,018		2,221.6	12.98	2.40
Federally Qualified and Rural Health Clinics 705,568 4,605 729.8 153.22 9.32 Aduit Medical Day Care 0 0 0.0 0.00 Start.55 Star	Non-Emergency Transportation	215,301		9,564		1,515.7	22.51	2.84
Adult Médical Day Care 0	Opioid Treatment Program	0		0		0.0	0.00	0.00
Personal Care 0 0 0 0.0 0.00 0.00 Durable Medical Equipment 419,646 188,943 29,943.8 2.22 5.54 Applied Behavioral Analysis 213,334 14,611 2.315.6 14.60 2.82 Other 1,258,533 80,595 12,772.8 15.62 16.62 Prescription Drugs 408,582 64,752.5 \$13.66 \$73.70 Generic Scripts \$2,994,585 98,371 15,589.9 \$30.44 \$39.55 Single-Source Brand 2,077,325 6,994 1,108.4 297.02 27.43 Multi-Source Brand 1,534,775 2.84 45.0 5,404.14 20.27 Specialty 1,534,775 2.84 45.0 5,404.14 20.27 Other 734 36 5.7 20.99 0.01 Other 734 36 5.7 20.39 0.01 Community Mental Health Center 6 6 5.450.5 2.483 110.59	Federally Qualified and Rural Health Clinics	705,568		4,605		729.8	153.22	9.32
Durable Medical Equipment 419,646 188,943 29,943.8 2.22 5.54 Applied Behavioral Analysis 213,334 14,611 2,315.6 14.60 2.82 Other 1,258,533 80,595 12,772.8 15.62 16.62 Prescription Drugs 664,752.5 \$13.66 \$73.70 Generic Scripts \$2,994,585 98,371 15,589.9 \$30.44 \$39.55 Single-Source Brand 2,077,325 6,994 1,108.4 297.02 27.43 Multi-Source Brand 185,080 827 131.1 224.68 2.45 Specialty 1,534,775 284 44.00 0.00 0.00 Other 734 36 5.7 20.39 0.01 Community Mental Health Center \$19,731,407 50,796 8,050.2 \$38.44 \$260.59 Long Term Support Service 8,373,692 337,268 53,450.5 24.83 110.59 Partial Hospital 0 0 0 0.00 0.00 0.00	Adult Medical Day Care	0		0		0.0	0.00	0.00
Applied Behavioral Analysis 213,334 14,611 2,315.6 14.60 2.82 Other 1,258,533 80,595 12,772.8 15.62 16.62 Prescription Drugs 64,752.5 \$13.66 \$7.70 Generic Scripts \$2,994,585 98,371 15,589.9 \$30.44 \$39.55 Single-Source Brand 2,077,325 6,6994 1,108.4 297.02 27.43 Multi-Source Brand 185,808 827 131.1 224.68 2.45 Specialty 1,534,775 284 45.0 5,404.14 20.27 Hepatitis C 0 0 0.0 0.00 <	Personal Care	0		0		0.0	0.00	0.00
Other 1,258,533 80,595 12,772.8 15.62 16.62 Prescription Drugs \$5,580,252 408,582 64,752.5 \$13.66 \$73.70 Generic Scripts \$2,994,585 98,371 15,589.9 \$30.44 \$39.55 Single-Source Brand 2,077,325 6,994 1,108.4 297.02 27.43 Muti-Source Brand 15,54,775 284 45.0 5,404.14 20.27 Hepatitis C 0 0 0.00 0.00 0.00 0.00 Other 734 36 5.7 20.39 0.01 Community Mental Health Center \$6,793,227 106,512 16,880.1 \$63.78 \$89.72 Case Management \$19,731,407 50,796 8,050.2 \$388.44 \$260.59 Long Term Support Service 8,373,692 337,268 53,450.5 24.83 110.59 Partial Hospital 0 0 0 0.0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 <	Durable Medical Equipment	419,646		188,943		29,943.8	2.22	5.54
\$5,580,252 408,582 64,752.5 \$13.66 \$73.70 Prescription Drugs Generic Scripts \$2,994,585 98,371 15,589.9 \$30.44 \$39.55 Single-Source Brand 2,077,325 6,994 1,108.4 297.02 27.43 Multi-Source Brand 155,480.8 827 131.1 224.66 2.45 Specialty 1,534,775 284 45.0 5,404.14 20.27 Hepatitis C 0 0 0.0 0.00 0.00 0.00 0.00 Other 734 36 5.7 20.39 0.01 Case Management \$19,731,407 50,796 8,050.2 \$388.44 \$260.59 Long Term Support Service 8,373,692 337,268 53,450.5 24.83 110.59 Partial Hospital 0 0 0 0.0 0.00 0.00 Psychotherapy 9,213,704 100,389 15,909.7 91.78 121.68 Evidence Based Practice 453 21 3.3 2	Applied Behavioral Analysis	213,334		14,611		2,315.6	14.60	2.82
Prescription Drugs \$2,994,585 98,371 15,589.9 \$30.44 \$39.55 Single-Source Brand 2,077,325 6,994 1,108.4 297.02 27.43 Multi-Source Brand 185,808 827 131.1 224.68 2.45 Specialty 1,534,775 284 45.0 5,404.14 20.27 Hepatitis C 0 0 0 0.00 0.00 0.00 Other 734 36 5.7 20.39 0.01 0.00 Community Mental Health Center - - - 26,793,227 106,512 16,880.1 \$63.78 829.72 Community Mental Health Center - <td>Other</td> <td>1,258,533</td> <td></td> <td>80,595</td> <td></td> <td>12,772.8</td> <td>15.62</td> <td>16.62</td>	Other	1,258,533		80,595		12,772.8	15.62	16.62
Generic Scripts \$2,994,585 98,371 15,589.9 \$30.44 \$39.55 Single-Source Brand 2,077,325 6,994 1,108.4 297.02 27.43 Multi-Source Brand 185,808 827 131.1 224.68 2.45 Specialty 1,534,775 284 45.0 5,404.14 20.27 Hepatitis C 0 0 0 0.00 0.00 0.00 Other 734 36 5.7 20.39 0.01 Case Management \$19,731,407 50,796 8,050.2 \$388.44 \$260.59 Long Term Support Service 8,373,692 337,268 53,450.5 24.83 110.59 Partial Hospital 0 0 0 0.00 0.00 0.00 Pschotherapy 9,213,704 100,389 15,909.7 91.78 121.68 0.13 Evidence Based Practice 453 21 3.3 21.58 0.01 Medication Management 9,537 358 56.7 2		\$5,580,252		408,582		64,752.5	\$13.66	\$73.70
Single-Source Brand 2,077,325 6,994 1,108.4 297.02 27.43 Multi-Source Brand 185,808 827 131.1 224.68 2.45 Specialty 1,534,775 284 45.0 5,040.14 20.27 Hepatits C 0 0 0 0.00	Prescription Drugs							
Multi-Source Brand 185,808 827 131.1 224.68 2.45 Specialty 1,534,775 284 45.0 5,404.14 20.27 Hepatitis C 0 0 0.0 0.00 0.00 Other 734 36 5.7 20.39 0.01 Community Mental Health Center \$6,793,227 106,512 16,880.1 \$63.78 \$89.72 Case Management \$19,731,407 50,796 8,050.2 \$388.44 \$260.59 Long Term Support Service 8,373,692 337,268 53,450.5 24.83 110.59 Partial Hospital 0 0 0 0.00 0.00 0.00 Psychotherapy 9,213,704 100,389 15,909.7 91.78 121.68 0.01 Medication Management 9,537 358 56.7 26.64 0.13 Evidence Based Practice 453 21 3.3 21.58 0.01 Medication Management 9,537 358 56.7 26.64	Generic Scripts	\$2,994,585		98,371		15,589.9	\$30.44	\$39.55
Specialty 1,534,775 284 45.0 5,404.14 20.27 Hepatitis C 0 0 0 0.0 0.01 0.01	Single-Source Brand	2,077,325		6,994		1,108.4	297.02	27.43
Hepatitis C 0 0 0 0.0 0.00 0.00 Other 734 36 5.7 20.39 0.01 Community Mental Health Center \$6,793,227 106,512 16,880.1 \$63.78 \$88.72 Case Management \$19,731,407 50,796 8,050.2 \$388.44 \$260.59 Long Term Support Service 8,373,692 337,268 53,450.5 24.83 110.59 Partial Hospital 0 0 0 0.00 0.00 0.00 Psychotherapy 9,213,704 100,389 15,909.7 91.78 121.68 0.01 Evidence Based Practice 453 21 3.3 21.58 0.01 Medication Management 9,537 358 56.7 26.64 0.13 Emergency Service 24/7 15,366 655 103.8 23.46 0.20 APRTP 8,551 297 47.1 28.79 0.11 Supported Employment Services 8,551 297 47.1	Multi-Source Brand	185,808		827		131.1	224.68	2.45
Other 734 36 5.7 20.39 0.01 \$6,793,227 106,512 16,880.1 \$63.78 \$89.72 Comunity Mental Health Center 106,512 16,880.1 \$63.78 \$89.72 Case Management \$19,731,407 50,796 8,050.2 \$338.44 \$260.59 Long Term Support Service 8,373,692 337,268 53,450.5 24.83 110.59 Partial Hospital 0 0 0 0.00 0.00 0.00 Psychotherapy 9,213,704 100,389 15,909.7 91.78 121.68 Evidence Based Practice 453 21 3.3 21.58 0.01 Medication Management 9,537 358 56.7 26.64 0.13 Emergency Service 24/7 15,366 655 103.8 23.46 0.20 APRTP 8,551 297 47.1 28.79 0.11 Supported Employment Services 8,551 297 47.1 28.79 0.11	Specialty	1,534,775		284		45.0	5,404.14	20.27
\$6,793,227 106,512 16,880.1 \$63.78 \$89.72 Community Mental Health Center \$19,731,407 50,796 8,050.2 \$388.44 \$260.59 Long Term Support Service 8,373,692 337,268 53,450.5 24.83 110.59 Partial Hospital 0 0 0.0 0.00 0.00 0.00 Psychotherapy 9,213,704 100,389 15,909.7 91.78 121.68 Evidence Based Practice 453 21 3.3 21.58 0.01 Medication Management 9,537 358 56.7 26.64 0.13 Emergency Service 24/7 15,366 655 103.8 23.46 0.20 APRTP 16,675 28 4.4 595.53 0.22 Supported Employment Services 8,551 2.97 47.1 28.79 0.111 Other 1,245,309 13,608 2,156.6 91.51 16.45 \$38,614,694 503,420 79,782.4 \$76.70 \$509.97	Hepatitis C	0		0		0.0	0.00	0.00
Community Mental Health Center 50.00 70.00 <	Other						20.39	0.01
Case Management \$19,731,407 50,796 8,050.2 \$388.44 \$260.59 Long Term Support Service 8,373,692 337,268 53,450.5 24.83 110.59 Partial Hospital 0 0 0 0.0 0.00 0.00 Psychotherapy 9,213,704 100,389 15,909.7 91.78 121.68 Evidence Based Practice 453 21 3.3 21.58 0.01 Medication Management 9,537 358 56.7 26.64 0.13 Emergency Service 24/7 15,366 655 103.8 23.46 0.20 APRTP 16,675 28 4.4 595.53 0.22 Supported Employment Services 8,551 297 47.1 28.79 0.11 Other 1,245,309 13,608 2,156.6 91.51 16.45 Cher \$38,614,694 503,420 79,782.4 \$76.70 \$509.97		\$6,793,227		106,512		16,880.1	\$63.78	\$89.72
Long Term Support Service 8,373,692 337,268 53,450.5 24.83 110.59 Partial Hospital 0 0 0 0.0 0.00 0.00 0.00 Psychotherapy 9,213,704 100,389 15,909.7 91.78 121.68 Evidence Based Practice 453 21 3.3 21.58 0.01 Medication Management 9,537 358 56.7 26.64 0.13 Emergency Service 24/7 15,366 655 103.8 23.46 0.20 APRTP 16,675 28 4.4 595.53 0.22 Supported Employment Services 8,551 297 47.1 28.79 0.11 Other 1,245,309 13,608 2,156.6 91.51 16.45 Suport \$38,614,694 503,420 79,782.4 \$76.70 \$509.97	Community Mental Health Center							
Partial Hospital 0 0 0 0.0 0.00 0.00 Psychotherapy 9,213,704 100,389 15,909.7 91.78 121.68 Evidence Based Practice 453 21 3.3 21.64 0.13 Medication Management 9,537 358 56.7 26.64 0.13 Emergency Service 24/7 15,366 655 103.8 23.46 0.20 APRTP 16,675 28 4.4 595.53 0.22 Supported Employment Services 8,551 297 47.1 28.79 0.11 Other 1,245,309 13,608 2,156.6 91.51 16.45	Case Management	\$19,731,407		50,796		8,050.2		\$260.59
Psychotherapy 9,213,704 100,389 15,909.7 91.78 121.68 Evidence Based Practice 453 21 3.3 21.58 0.01 Medication Management 9,537 358 56.7 22.64 0.13 Emergency Service 24/7 15,366 655 103.8 23.46 0.20 APRTP 16,675 28 4.4 595.53 0.22 Supported Employment Services 8,551 297 47.1 28.79 0.11 Other 1,245,309 13,608 2,156.6 91.51 16.45		8,373,692		337,268				
Evidence Based Practice 453 21 3.3 21.58 0.01 Medication Management 9,537 358 56.7 26.64 0.13 Emergency Service 24/7 15,366 655 103.8 23.46 0.20 APRTP 16,675 28 4.4 595.53 0.22 Supported Employment Services 8,551 297 47.1 28.79 0.11 Other 1,245,309 13,608 2,156.6 91.51 16.45	•							
Medication Management 9,537 358 56.7 26.64 0.13 Emergency Service 24/7 15,366 655 103.8 23.46 0.20 APRTP 16,675 28 4.4 595.53 0.22 Supported Employment Services 8,551 297 47.1 28.79 0.11 Other 1,245,309 13,608 2,156.6 91.51 16.45								
Emergency Service 24/7 15,366 655 103.8 23.46 0.20 APRTP 16,675 28 4.4 595.53 0.22 Supported Employment Services 8,551 297 47.1 28.79 0.11 Other 1,245,309 13,608 2,156.6 91.51 16.45 \$38,614,694 503,420 79,782.4 \$76.70 \$509.97								
APRTP 16,675 28 4.4 595.53 0.22 Supported Employment Services 8,551 297 47.1 28.79 0.11 Other 1,245,309 13,608 2,156.6 91.51 16.45 \$38,614,694 503,420 79,782.4 \$76.70 \$509.97								
Supported Employment Services 8,551 297 47.1 28.79 0.11 Other 1,245,309 13,608 2,156.6 91.51 16.45 \$38,614,694 503,420 79,782.4 \$76.70 \$509.97								
Other 1,245,309 13,608 2,156.6 91.51 16.45 \$38,614,694 503,420 79,782.4 \$76.70 \$509.97								
\$38,614,694 503,420 79,782.4 \$76.70 \$509.97								
	Other							
All Services \$58,455,162 537 1,085,798 85.1 172,078.2 \$53.84 \$772.00								
	All Services	\$58,455,162	537	1,085,798	85.1	172,078.2	\$53.84	\$772.00

Eligibility Category: Granite Advantage - Medically Frail

Member Months: 84,734

							Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient				,	,		
Medical	\$5,025,483	1,138	6,446	161.2	912.9	\$4,416.07	\$59.31
Surgical	3,895,525	368	4,126	52.1	584.3	10,585.67	45.97
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	14,904	6	33	0.8	4.7	2,483.93	0.18
Well Newborn	0	0	0	0.0	0.0	0.00	0.00
Psychiatric	1,054,213	216	1,547	30.6	219.1	4,880.62	12.44
Alcohol and Drug Abuse	1,567,815	402	4,632	56.9	656.0	3,900.04	18.50
Other	0	0	0	0.0	0.0	0.00	0.00
	\$11,557,940	2,130	16,784	301.6	2,376.9	\$5,426.26	\$136.40
Hospital Outpatient	¢4,000,005		7.040		4 0 4 0 2	#070 F7	¢50.04
Emergency Room	\$4,992,095		7,346		1,040.3	\$679.57	\$58.91
Surgery	1,434,907		1,806		255.8	794.52	16.93
Radiology	2,201,714		7,728		1,094.4	284.90	25.98
Pathology	659,261		50,015		7,083.1	13.18	7.78
Pharmacy	4,229,112		420,622		59,568.1	10.05	49.91
Cardiovascular PT/OT/ST	258,921		1,130		160.0 2,499.7	229.13 30.13	3.06
	531,870 133,057		17,651 625		2,499.7 88.5	212.89	6.28 1.57
Psychiatric							
Substance Abuse Other	86,633 2,179,523		781 61,644		110.6 8,730.0	110.93 35.36	1.02 25.72
Other	\$16,707,093		569,348		80,630.6	\$29.34	\$197.17
Professional and Other State Plan Services	φ10,101,000		000,040		00,000.0	φ20.04	<i><i></i></i>
Ambulatory Surgery Center	\$173,111		651		92.2	\$265.92	\$2.04
Office Visits	2,424,031		40,044		5,671.0	60.53	28.61
Preventive Medicine	256,861		8,319		1,178.1	30.88	3.03
Maternity	27,370		123		17.4	222.24	0.32
Certified Midwife	413		21		3.0	19.66	0.00
PT/OT/ST	254,505		11,041		1,563.6	23.05	3.00
Psychiatric and Substance Abuse	6,019,331		50,039		7,086.5	120.29	71.04
Radiology and Pathology	2,169,100		64,921		9,194.1	33.41	25.60
Home Health and Private Duty Nursing	1,114,990		41,133		5,825.2	27.11	13.16
Ambulance	518,958		25,385		3,595.0	20.44	6.12
Non-Emergency Transportation	3,203,961		197,253		27,934.8	16.24	37.81
Opioid Treatment Program	1,832,751		179,328		25,396.3	10.22	21.63
Federally Qualified and Rural Health Clinics	1,850,785		12,137		1,718.8	152.49	21.84
Adult Medical Day Care	14,519		816		115.6	17.79	0.17
Personal Care	48,028		8,901		1,260.6	5.40	0.57
Durable Medical Equipment	1,414,337		232,642		32,946.6	6.08	16.69
Applied Behavioral Analysis	345		23		3.3	15.00	0.00
Other	4,901,197		348,027		49,287.3	14.08	57.84
	\$26,224,594		1,220,804		172,889.3	\$21.48	\$309.49
Prescription Drugs	¢0.000.000		400 754		00 700 0	¢10.10	¢40.50
Generic Scripts	\$3,608,263		188,751		26,730.8	\$19.12	\$42.58
Single-Source Brand	7,335,478		18,750		2,655.4	391.23	86.57
Multi-Source Brand	2,554,617		10,717		1,517.7	238.37	30.15
Specialty	8,718,880		1,997		282.8	4,365.99	102.90
Hepatitis C Other	3,459,060 1,511		280 43		39.7 6.1	12,353.79 35.14	40.82 0.02
Other	\$25,677,810		220,538		31,232.4	\$116.43	\$303.04
Community Mental Health Center	φ20,011,010		220,000		01,202.4	φ110.40	\$000.01
Case Management	\$92,485		241		34.1	\$383.75	\$1.09
Long Term Support Service	155,601		2,231		316.0	69.74	1.84
Partial Hospital	0		2,201		0.0	0.00	0.00
Psychotherapy	195,115		8,730		1,236.3	22.35	2.30
Evidence Based Practice	2,270		230		32.6	9.87	0.03
Medication Management	4,162		156		22.1	26.68	0.05
Emergency Service 24/7	878		37		5.2	23.72	0.01
APRTP	97,071		163		23.1	595.53	1.15
Supported Employment Services	1,229		83		11.8	14.81	0.01
Other	533,744		8,771		1,242.1	60.85	6.30
	\$1,082,554		20,642		2,923.3	\$52.44	\$12.78
All Services	\$81,249,990	2,130	2,048,116	301.6	290,052.5	\$39.67	\$958.88

Eligibility Category: Granite Advantage - Non-Medically Frail

Member Months: 383,770

	Tetal Duit	Tetal Daile	Tatal Data	A duality Day		A	Per Capita
Benefits	Total Paid Dollars	Total Paid Admits	Total Paid Services	Admits Per 1,000	Utilization Per 1,000	Average Paid Charge	Monthly Paid Cost
Hospital Inpatient	Donard	Admito	00111000	1,000	1,000	onargo	0001
Medical	\$7,876,994	1,732	9,536	54.2	298.2	\$4,547.92	\$20.53
Surgical	5,293,368	670	4,759	21.0	148.8	7,900.55	13.79
Maternity Delivery	0	0	0	0.0	0.0	0.00	0.00
Maternity Non-Delivery	77,397	27	134	0.8	4.2	2,866.54	0.20
Well Newborn	2,200	4	7	0.1	0.2	550.01	0.01
Psychiatric	1,499,344	339	2,162	10.6	67.6	4,422.84	3.91
Alcohol and Drug Abuse	1,913,217	607	5,639	19.0	176.3	3,151.92	4.99
Other	0	0	0	0.0	0.0	0.00	0.00
	\$16,662,520	3,379	22,237	105.7	695.3	\$4,931.20	\$43.42
Hospital Outpatient	¢40.040.000		40.470		500.0	¢000.00	¢04.00
Emergency Room	\$12,046,890		18,173		568.2	\$662.90	\$31.39
Surgery	4,052,507		4,614		144.3 665.6	878.31	10.56
Radiology	5,097,659		21,287			239.47	13.28
Pathology	1,721,447		131,530		4,112.8	13.09	4.49
Pharmacy	5,442,435		669,445		20,932.7	8.13 238.43	14.18
Cardiovascular PT/OT/ST	604,412 1,047,768		2,535 33,861		79.3 1,058.8	230.43	1.57 2.73
Psychiatric	177,852		1,038		32.5	171.34	0.46
Substance Abuse	140,558		1,038		32.5	116.94	0.40
Other	4,736,178		176,906		5,531.6	26.77	12.34
Other	\$35,067,706		1,060,591		33,163.4	\$33.06	\$91.38
Professional and Other State Plan Services	<i>\\</i> 00,007,100		1,000,001		00,100.4	\$00.00	φ01.00
Ambulatory Surgery Center	\$337,776		1,301		40.7	\$259.63	\$0.88
Office Visits	5,539,769		94,622		2,958.7	58.55	14.44
Preventive Medicine	1,194,959		35,844		1,120.8	33.34	3.11
Maternity	155,601		697		21.8	223.10	0.41
Certified Midwife	6,788		69		2.1	98.87	0.02
PT/OT/ST	644,057		28,245		883.2	22.80	1.68
Psychiatric and Substance Abuse	9,288,844		71,057		2,221.9	130.72	24.20
Radiology and Pathology	3,580,582		129,911		4,062.2	27.56	9.33
Home Health and Private Duty Nursing	556,076		11,055		345.7	50.30	1.45
Ambulance	839,074		43,834		1,370.6	19.14	2.19
Non-Emergency Transportation	3,675,689		232,458		7,268.7	15.81	9.58
Opioid Treatment Program	2,340,352		228,991		7,160.3	10.22	6.10
Federally Qualified and Rural Health Clinics	4,037,515		28,233		882.8	143.01	10.52
Adult Medical Day Care	12,380		526		16.4	23.54	0.03
Personal Care	0		0		0.0	0.00	0.00
Durable Medical Equipment	1,359,400		120,822		3,778.0	11.25	3.54
Applied Behavioral Analysis	0		0		0.0	0.00	0.00
Other	9,434,689		675,538		21,123.2	13.97	24.58
Description Description	\$43,003,551		1,703,203		53,257.0	\$25.25	\$112.06
Prescription Drugs Generic Scripts	\$7 162 570		383,547		11,993.0	\$18.67	\$18.66
Single-Source Brand	\$7,162,579		,		1,124.8	357.88	
Multi-Source Brand	12,873,915		35,973		378.5	244.18	33.55 7.70
Specialty	2,955,816 12,418,620		12,105 2,788		87.2	4,454.31	32.36
Hepatitis C	3,300,686		2,788		8.4	12,224.76	8.60
Other	1,149		84		2.6	12,224.70	0.00
Other	\$38,712,765		434,767		13,594.6	\$89.04	\$100.87
Community Mental Health Center	<i>\\</i> 00,712,700		-10-1,1 01		10,004.0		 100.01
Case Management	\$100,435		258		8.1	\$389.28	\$0.26
Long Term Support Service	234,810		3,111		97.3	75.48	0.61
Partial Hospital	0		0		0.0	0.00	0.00
Psychotherapy	296,138		8,629		269.8	34.32	0.77
Evidence Based Practice	557		21		0.7	26.54	0.00
Medication Management	4,438		166		5.2	26.74	0.01
Emergency Service 24/7	835		35		1.1	23.86	0.00
APRTP	103,092		174		5.4	592.48	0.27
Supported Employment Services	1,094		50		1.6	21.88	0.00
Other	500,604		6,935		216.8	72.19	1.30
	\$1,242,003		19,379		606.0	\$64.09	\$3.24
All Services	\$134,688,545	3,379	3,240,177	105.7		\$41.57	\$350.96

Appendix ST New Hampshire Department of Heakth and Human Services FY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Low Income Children - Age 0-11 Months

Benefits	Per Capita Monthly Paid Cost	IBNR Adiustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiustment	PDL Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adiustment	Projected Per Capita Monthly Paid Cost
Hospital Inpatient			and the second second second			a line internet fins a			a second fact of	and the second last r	
Medical	\$18.40	1.0000	1.0874		-	1.0000	1.0000	1.0000	·	1.0000	
Surgical	5.46	1.0000	1.1049	•	.	1.0000	1.0000	1.0000	•	1.0000	
Maternity Delivery	0.00	0000	0000 F			0000 1	0000	10000		0000 F	
Wall Newhorn	0.00	1 0000	1 1153			1 0000	1 0000	1 0000		1 0000	
Psychiatric	00.00	1.0000	1.0000	·		1.0000	1.0000	1.0000		1.0000	
Alcohol and Drug Abuse	0.00	1.0000	1.0000	1.0000	1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	00.0
Otter	\$23.86	1-000	0000.1			1-0000	0000-1	00001		1.0000	
Hospital Outpatient											
Emergency Room	\$18.05	1.0000	1.0604	<i>(</i> -	1.1468	1.0000	1.0000	1.0000		-	
Surgery	4.19	1.0000	1.0604	-	1.1468	1.0000	1.0000	1.0000			
Radiology	3.76	1.0000	1.0604	,	1.1468	1.0000	1.0000	1.0000			
Pathology	c0.1 30.1	1.0000	1.0604	- •	1.1468	1.0000	1.0000	1.0000			
ruanuacy Cardiovascular	0.34	1,0000	1 0604		1.1400	1 0000	1 0000	1 0000			
PT/OT/ST	141	1.0000	1.0604	- (-	1.1468	1.0000	1 0000	1.0000		-	
Psychiatric	00 0	1 0000	1.0000		1.1468	1.0000	1 0000	1.0000		-	
Substance Abuse	00.00	1.0000	1.0000	1.0200	1.1468	1.0000	1.0000	1.0000	1.0000	1.0000	00.0
Other	11.09	1.0000	1.0604		1.1468	1.0000	1.0000	1.0000			
	\$41.25										\$51.17
Protessional and Other State Plan Services Ambulatory Surgery Center	\$U 26	1 0000	1 3531		1 0000	1 0000	1 0000	1 0000			
Office Visits	17.72	1.0000	1.0601		1.0000	1.0000	1.0000	1.0000			
Preventive Medicine	19.26	1.0000	1.0635		1.0000	1.0000	1.0000	1.0000		~	
Maternity	00.00	1.0000	1.0000	·	1.0000	1.0000	1.0000	1.0000	•	,	
Certified Midwife	0.04	1.0000	1.0669		1.0000	1.0000	1.0000	1.0000		,	
PI/01/SI Devekiatria and Substance Abuca	10.0	1.0000	1.0630		1.0000	1.0000	1.0000	1.0000			
Feytimatic and Bathology Rediology and Pathology	142	1 0000	1 0565		1.0000	1 0000		1 0000			
Home Health and Private Duty Nursing	5.77	1.0000	1.0629	·	1.0000	1.0000	1.0000	1.0000			
Ambulance	1.78	1.0000	1.0630	·	1.0000	1.0000	1.0000	1.0000	•	·	
Non-Emergency Transportation	1.56	1.0000	1.0630		1.0000	1.0000	1.0000	1.0000		~	
Opioid Treatment Program	0.00	1.0000	1.0000		1.0000	1.0000	1.0000	1.0000	•		
Federally Qualified and Kural Health Clinics	18.86	1.0000	1.1683		1.0000	1.0000	1.0000	1.0000			
Adult Medical Day Care Porsonal Care	00.0	1.0000	1.0000		1.0000	1.0000	1.0000	1.0000		_ (
Durable Medical Equipment	27.81	1.0000	1.0462	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	
Applied Behavioral Analysis	00.00	1.0000	1.0000	·	1.0000	1.0000	1.0000	1.0000		·	
Other	25.48	1.0000	1.0791		1.0000	1.0000	1.0000	1.0000		~	27.77
	\$120.48										\$131.32
Prescription Urugs Generic Scrints	92 23	1 0000	1 0000		1 0064	1 0000	1 0046	1 0000			
Single-Source Brand	3.31	1.0000	1.0000		0.9747	1.0000	0.9098	1.0000			
Multi-Source Brand	0.43	1.0000	1.0000	·	0.9747	1.0000	1.2839	1.0000			
Specialty	12.37	1.0000	1.0000	•	2.3276	1.0000	0.9860	1.0000			
Hepatitis C	0.00	1.0000	1.0000	1.0000	1.0000	1.0000	0.9894	1.0000	1.0000	1.0000	
Otter	\$19.46	0000-1	1.0000		1.0004	1.0000	1.97.0	0000-1			\$35.25
Community Mental Health Center											
Case Management	\$0.00	1.0500	1.0000		1.0000	1.0000	1.0000	1.0000			
Long Term Support Service	0.00	1.0500	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Partial Hospital Developherany	000	1.0500	1.0000		1.0000	1.0000	00001	0000 1			
r sychourerapy Evidence Based Practice		1 0500	1 0000		1 0000	1 0000	1 0000	1 0000			
Medication Management	00.00	1.0500	1.0000		1.0000	1.0000	1 0000	1.0000			
Emergency Service 24/7	00.00	1.0500	1.0000		1.0000	1.0000	1.0000	1.0000			
APRTP	00.00	1.0500	1.0000		1.0000	1.0000	1.0000	1.0000			
Supported Employment Services	0.00	1.0500	1.0000		1.0000	1.0000	1.0000	1.0000			
Other	00.0\$	nnen I	1.0000		0000-I	1.0000	0000-I	1.0000			
All Services	\$205.06										\$243.94

Appendix B1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Low Income Children - Age 1-18 Years

Banafite	Per Capita Monthly Paid Cost	IBNR Adinetment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Eartors	Managed Care Savings ∆diustment	PDI Adiustment	Medicare OUD	Stop Loss Adiustment	NEMT Adiustment	Projected Per Capita Monthly Paid Cost
Hospital Inpatient	1000		NIGHTON DA	1000	01000				unumentau	Hampalau	1000
Medical	\$2.40	1.0000	1.0973	-		1.0000	-	1.0000	1.0000	1.0000	\$2.65
Surgical	1.46	1.0000	1.1011	,		1.0000		1.0000	1.0000	1.0000	1.61
Maternity Delivery Maternity Non-Delivery	00.0	1.0000	1.0000	- •		1.0000		1.0000	1.0000	1.0000	0.00
	00.0	1.0000	1.1306			1.0000	,	1.0000	1.0000	1.0000	0000
Psychiatric	2.50	1.0000	1.0991	<i>t</i> -		1.0000	~	1.0000	1.0000	1.0000	2.77
Alcohol and Drug Abuse	0.03	1.0000	1.1293	1.0000	1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	0.03
Olliel	0.00 \$6.43	1.0000	1.000			1.0000		1.0000	1.000	1-0000	S7.10
Hospital Outpatient											
Emergency Room	\$10.46	1.0000	1.0604	1.0200	1.1468	1.0000	1.0000	1.0000	1.0000	1.0000	\$12.97
Surgery	3.57	1.0000	1.0604	•		1.0000		1.0000	1.0000	1.0000	4.42
Radiology	3.41	1.0000	1.0604			1.0000		1.0000	1.0000	1.0000	4.23
Phormood	67°L	1.0000	1.0004			00001		1.0000	1.0000	0000 1	0.0
Cardiovascular	0.09	1.0000	1.0604		·	1.0000	·	1.0000	1.0000	1.0000	0.11
PT/OT/ST	1.18	1.0000	1.0604	·		1.0000	·	1.0000	1.0000	1.0000	1.46
Psychiatric	0.28	1.0000	1.0604	·		1.0000	·	1.0000	1.0000	1.0000	0.34
Substance Abuse	0.00	1.0000	1.0604			1.0000		1.0000	1.0000	1.0000	0.00
Olliel	3.39 \$28.15	1.0000	1.0004			1.0000		10000	1.000	1-0000	0.94 \$34.92
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$0.37	1.0000	1.0307			1.0000	-	1.0000	1.0000	1.0000	\$0.38
Office Visits	9.51	1.0000	1.0606	1.0100	.	1.0000	-	1.0000	1.0000	-	10.19
Preventive Medicine	3.80	1.0000	1.0781			1.0000		1.0000	1.0000		4.14
Maternity Contified Midwife	0.00	1.0000	1.0030			1.0000		1.0000	1.0000	- •	60.0
	80.8	1 0000	1 0630			1 0000	- x	1 0000	1 0000		3.32
Psychiatric and Substance Abuse	4.91	1.0000	1.0576			1.0000		1.0000	1.0000		5.24
Radiology and Pathology	1.19	1.0000	1.0565		-	1.0000	·	1.0000	1.0000	-	1.27
Home Health and Private Duty Nursing	2.59	1.0000	1.0629	1.0100		1.0000		1.0000	1.0000	1.0000	2.78
Ambulance Non-Emercency Transnortation	1.25	1.0000	1.0630			1,0000	,	1.0000	1.0000		0.09
Opioid Treatment Program	00.00	1.0000	1.0636			1.0000		1.0000	1.0000		0000
Federally Qualified and Rural Health Clinics	7.60	1.0000	1.1678		-	1.0000		1.0000	1.0000	-	8.97
Adult Medical Day Care	00.00	1.0000	1.0000		-	1.0000	-	1.0000	1.0000	-	0.00
Personal Care	0.00	1.0000	1.0631			1.0000		1.0000	1.0000		0.00
Durable Medical Equipment	4.47		1.0403			1.0000		1.0000	1.0000		4.70
Applied Benavioral Analysis Other	3.89	1.0000	1.0630	2.2/36	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	9.39
	\$51.87										\$61.76
Prescription Drugs											
Generic Scripts	\$9.43	1.0000	1.0000	1.0000	1.0064	1.0000	1.0046	1.0000	1.0000	1.0000	\$9.54
Single-Source Brand	8.56	1.0000	1.0000			1.0000			1.0000	1.0000	7.59
Mulu-Source Brand Snecially	11 77	1 0000	1 0000			1 0000			1.0000	1.0000	27.01
Hepatitis C	0.00	1.0000	1.0000			1.0000			1.0000	1.0000	0.00
Other	0.00	1.0000	1.0000			1.0000			1.0000	1.0000	0.00
	\$32.22										\$47.21
Community mental Health Center Case Management	\$0.88	1.0500	1.1526		,	1.000		1.000	1.000	1.0000	\$1.08
Long Term Support Service	0.45	1.0500	1.0768			1.0000		1.0000	1.0000	1.0000	0.52
Partial Hospital	00.00	1.0500	1.0000			1.0000		1.0000	1.0000	1.0000	00.00
Psychotherapy	0.34	1.0500	1.0421			1.0000		1.0000	1.0000	1.0000	0.38
Evidence Based Practice	0.00	1.0500	1.0000			1.0000		1.0000	1.0000	1.0000	0.00
Medication Management Emernency Service 24/7	00.0	1 0500	1.0031	1.0100	1.0000	1 0000	1 0000	1.0000	1.0000	1.0000	00.0
APRTP	0.01	1.0500	1.1530			1.0000	·	1.0000	1.0000	1.0000	0.02
Supported Employment Services	00.00	1.0500	1.0000			1.0000		1.0000	1.0000	1.0000	00.00
Other	0.60	1.0500	1.0609			1.0000		1.0000	1.0000	1.0000	0.67
All S-min-2	\$2.29										\$2.67 6457.60
All Services	08'071¢										00.0010

Appendix B1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MOO Encounter Base Experience Data

Eligibility Category: Low Income Adults

Ronafite	Per Capita Monthly Paid Cost	IBNR Adjustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiustment	PDI Adiustment	Medicare OUD Adjustment	Stop Loss Adiustment	NEMT Adjustment	Projected Per Capita Monthly Paid Cost
Hospital Inpatient	1000		rujusmism.	00000	0000	mannenfau		Mannenfau		unumenter.	1000 100
Medical	\$13.44	-	1.0871	1.0000	1.0060	0.9902		1.0000		1.0000	\$14.56
Surgical	11.12	-	1.0956	1.0000	1.0060	0.9902		1.0000	•	1.0000	12.14
Matemity Delivery	0.00	- •	1.0000	1.0000	1.0060	0.9902		1.0000		1.0000	0.00
Malernity Nort-Delivery Mall Newhorn	0.16		1.0/13	1.0000	1.0060	0.9902	1.0000	1.0000		1.0000	2.44 0.18
Psychiatric	3.19	. –	1.1132	1.0000	1.0060	0.9902	·	1.0000		1.0000	3.53
Alcohol and Drug Abuse	1.32	1.0000	1.1235	1.0000	1.0060	0.9902	1.0000	1.0000	1.0000	1.0000	1.47
Otter	\$31.52		0000.1	1.0000	1-000	0.8802		1.000		1.0000	\$34.33
Hospital Outpatient											000
Emergency Room	\$37.08	1.0000	1.0604	1.0200	1.1468	0.9902	1.0000	1.0000	1.0000	1.0000	\$45.54
Surgery	10.84		-	1.0200	1.1468	0.9902		1.0000	-	1.0000	13.31
Radiology	15.41		~	1.0200	1.1468	0.9902		1.0000		1.0000	18.93
Pathology	2.09			1.0200	1.1468	0.9902		1.0000		1.0000	8.71
Pnarmacy Continuoscular	67.7		- (1.0200	1.1400	0.9902		0000 1		1.0000	900 A
PT/OT/ST	2 2 2			1 0200	1 1468	2066.0		1 0000	,	1 0000	3.34
Psychiatric	0.32			1 0200	1.1468	0.9902		1.0000		1.0000	0.39
Substance Abuse	0.21	-	-	1.0200	1.1468	0.9902		1.0000		1.0000	0.26
Other	15.10	-		1.0200	1.1468	0.9902		1.0000	1.0000	1.0000	18.55
	297.60										\$119.88
Professional and Other State Plan Services	¢1 00		•	1 0100	0000 1	00000	0000 1	0000 1	0000	1 0000	61.10
Attibutatory Surgery Center Office Visits	10.10			1.0100	1 0000	0.9902		1 0000	1.0000	1.0000	21.15 20.32
Preventive Medicine	7.56		- (-	1.0100	1.0000	0.9902		1.0000	1.0000	1.0000	8.67
Maternity	10.20			1.0100	1 0000	0.9902		1.0000	1 0000	1 0000	10.84
Certified Midwife	0.24	-	-	1.0100	1.0000	0.9902	-	1.0000	1.0000	1.0000	0.25
PT/OT/ST	1.60	-	-	1.0100	1.0000	0.9902	-	1.0000	v -	1.0000	1.70
Psychiatric and Substance Abuse	19.32		~	1.0100	1.0000	0.9902	,	1.0000		1.0000	21.83
Radiology and Pathology	13.40			1.0100	1.0000	0.9902		1.0000		1.0000	14.20
Ambulance	1.23		•	1.0100	1.0000	0.9902	- •	1.0000		1.0000	1.31
Non-Emergency Transportation	18.26		- (-	1.0100	1.0000	0.9902		1.0000	,	1.0000	19.41
Opioid Treatment Program	12.42			1.0100	1.0000	0.9902		1.0000	<i>t</i>	1.0000	13.21
Federally Qualified and Rural Health Clinics	19.54	-	-	1.0100	1.0000	0.9902		1.0000	-	1.0000	22.74
Adult Medical Day Care	0.11	-	-	1.0100	1.0000	0.9902		1.0000		1.0000	0.12
Personal Care	0.01			1.0100	1.0000	0.9902	.	1.0000		1.0000	0.01
Uurable Medical Equipment	4.8/			1.0100	1.0000	0.9902		1.0000		1.0000	4./8
Other	22.89	1.0000	1 0351	1.0100	1.0000	0.9902	1,0000	1,0000	1,0000	1.0000	23.70
	\$154.03										\$166.61
Prescription Drugs											
Generic Scripts	\$19.95	1.0000	1.0000	1.0000	1.3069	0.9902	1.3702	1.0000	1.0000	1.0000	\$35.38
Single-Source Brand Multi-Source Brand	28.53		1.0000		0.8891	0.9902				1.0000	£ 70
Sherialty	26.62	`	1 0000		1.8419	0.9902			•	1 0000	47.46
Hepatitis C	17 54		1.0000	· ·	0.3641	0.9902				1.0000	6.03
Other	0.00	`	1.0000	~	1.3069	0.9902			,	1.0000	0.00
	\$114.73										\$117.74
Community Mental Health Center Case Management	25 U\$		1 1530	1 0100	1 0000	1000	1 0000	1 0000		1 0000	\$0.46
Long Term Support Service	12.00	- (-	1.0690	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	0.87
Partial Hospital	0.00	1.0500	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Psychotherapy	1.31	-	1.0582	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	1.47
Evidence Based Practice	0.00		1.0000	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Medication Management	0.01		1.0631	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	0.01
Entergency Service 24/7 APRTP	0.00	- (1.0051	1 0100	1 0000	1 0000	1 0000	1 0000		1.0000	0.30
Supported Employment Services	0.00		1.0000	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Other	1.53		1.0630	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	1.73
	\$4.25										\$4.84 0.10.10
All Services	\$402.12										\$443.40

Appendix B1	New Hampshire Department of Health and Human Services	·Y 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023	Data Adjustments for CY 2018 MCO Encounter Base Experience Data
-------------	---	--	---

Eligibility Category: CHIP

State 100 1000 <th< th=""><th>Remefits</th><th>Per Capita Monthly Paid Cost</th><th>IRNR Adiustment</th><th>Reimbursement Adiustment</th><th>Utilization Trend Factors</th><th>Unit Cost Trend Factors</th><th>Managed Care Savings Adiustment</th><th>PDI Adiustment</th><th>Medicare OUD Adiustment</th><th>Stop Loss Adiustment</th><th>NEMT Adiustment</th><th>Projected Per Capita Monthly Paid Cost</th></th<>	Remefits	Per Capita Monthly Paid Cost	IRNR Adiustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiustment	PDI Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adiustment	Projected Per Capita Monthly Paid Cost
attention attention <t< td=""><td></td><td>1000</td><td></td><td>NINING BU</td><td>210000</td><td>2000</td><td>unamen nu</td><td>unamentaria i</td><td>mannenfau</td><td>Minimemfau</td><td>linineninu</td><td>1 200</td></t<>		1000		NINING BU	210000	2000	unamen nu	unamentaria i	mannenfau	Minimemfau	linineninu	1 200
Mathemy (Non-Dollway		\$0.90	1.0000	1.0786			1.0000		1.0000	,		\$0.98
Distribution 0.00 1.000 1.000 1.000 1.000 with Distribution 0.00 1.000 1.000 1.000 1.000 1.000 with Distribution 0.00 1.00		0.57	1.0000	1.0809	<i>•</i>	~	1.0000		1.0000		·	0.62
Inter-Delevery 0.02 1.000 1.9460 1.000 Inter-Delevery 0.03 1.000 1.000 1.000 Very 0.03 1.000 1.000 1.000 1.000 Very 0.04 1.000 1.000 1.000 <td>y Delivery</td> <td>0.00</td> <td>1.0000</td> <td>1.0000</td> <td>-</td> <td>-</td> <td>1.0000</td> <td></td> <td>1.0000</td> <td>.</td> <td>·</td> <td>0.00</td>	y Delivery	0.00	1.0000	1.0000	-	-	1.0000		1.0000	.	·	0.00
Anton and Ding Alters Cost Sala Cost From Sala Cost From From Sala Cost From From Sala Cost From From Sala Cost From From From Sala Cost From From From From From From From From	ly Non-Delivery	0.02	1.0000	1.0460	(- (1.0000		1.0000			0.02
Bind Drug Abtes 0.00 1.000 1.000 1.000 1.000 State State 1.000 1.000 1.000 1.000 1.000 State State 1.000 1.000 1.000 1.000 1.000 1.000 State State 1.000 1.	w.com	0.00	1 0000	1 1191			1,0000	1.0000	1 0000	1 0000	1 0000	0.00
Turbulation 0.00 1,000 1,000 1,000 1,000 1,000 Morkinet \$\$23 1,000 1,004 1,000 1,000 1,000 No \$\$23 1,000 1,000 1,000 1,000 1,000 No \$\$23 1,000 1,000 1,000 1,000 1,000 No \$\$13 1,000 1,000 1,000 1,000 1,000 No \$\$14 1,000 1,000 1,000 1,000 1,000 No \$\$100 1,000 1,000 1,000 1,000 1,000 No \$\$100 \$\$1000 \$\$10	and Drug Abuse	0.08	1.0000	1 0000			1 0000		1.0000	•		0.08
Hall Outpatient S3.36 1000 10004 12200 Nov Room 3.23 10000 10004 12200 Nov Room 3.23 10000 10004 12200 Nov Room 114 10000 10004 12200 Nov Room 3.33 10000 10004 12200 Nov Room 3.33 10000 10004 12200 Nov Room 3.33 10000 10001 10001 Nov Room 4.34 100000 10001 10000	2	0.00	1.0000	1.0000		1	1.0000		1.0000			0.00
Intervention Same 1000		\$3.95										\$4.38
prov Room 8.23 1.0000 1.0604 1.2200 00 Vector 1.33 1.0000 1.0604 1.2200 01 Vector 1.33 1.0000 1.0604 1.2200 01 Vector 1.33 1.0000 1.0604 1.2200 01 Vector 2.33 1.0000 1.0604 1.2200 01 Vector 2.33 1.0000 1.0604 1.2200 Vector 2.33 1.0000 1.060	I Outpatient											
$V_{\rm eff}$	ncy Room	\$8.23	1.0000	1.0604		1.1468	1.0000	1.0000	1.0000	1.0000	1.0000	\$10.21
Optimized Constrained Constrained <thconstrained< th=""> <thconstrained< th=""> <</thconstrained<></thconstrained<>		3.09	1.0000	1.0604		1.1468	0000	- •	1.0000		- •	3.83
m_{eff} <	AC AC	3.28 1 16	1.0000	1 0604	- •	1.1408	1.0000		1.0000		1.0000	4.07
mascular 0.1 0.00 0.000 <th< td=""><td>رب در</td><td>01-1 76 U</td><td>1 0000</td><td>1 0604</td><td></td><td>1 1468</td><td></td><td></td><td>1 0000</td><td></td><td></td><td>1.16</td></th<>	رب در	01-1 76 U	1 0000	1 0604		1 1468			1 0000			1.16
(517) (14) (100) (160) (120) (120) ince Abuse $(2,2)$ (100) (100) (120) (122) ince Abuse $(2,3)$ (100) (100) (100) (122) ince Abuse $(2,3)$ (100) (100) (100) (122) into Share State $(12,3)$ (100) (100) (100) (122) into Share State $(12,3)$ (100) (100) (100) (100) into Share $(12,3)$ (100) (100) (100) (100) into Share (120) (100) (100) (100) (100) into Share (120) (120) (100) (120) (100) into Share (120) (120) (120) (120) (120) into Share (120) (120) (120) (120) (120) into Share (120) (120) (120)	ascular	0.15	1.0000	1 0604		1.1468	1 0000		1.0000	- (-		0.18
Intic 0.25 1.000 1.064 1.2200 Inter Abuse 0.33 1.000 1.064 1.2200 Inter Abuse 0.33 1.000 1.064 1.2200 Inter Abuse 0.33 1.000 1.064 1.020 Inter Abuse 0.33 1.000 1.063 1.010 Inter Medicine 0.33 1.000 1.063 1.010 Inter Matchine 0.33 1.000 1.063 1.010 Inter Matchine 0.33 1.000 1.053 1.010 Inter Matchine 0.33 <th< td=""><td>ST</td><td>1.41</td><td>1.0000</td><td>1 0604</td><td>- 1</td><td>1.1468</td><td>1 0000</td><td></td><td>1.0000</td><td></td><td></td><td>1.75</td></th<>	ST	1.41	1.0000	1 0604	- 1	1.1468	1 0000		1.0000			1.75
and the set of the s	tric	0.25	1 0000	1 0604		1.1468	1 0000	-	1 0000		-	0.31
Sinal 1000 1000 10200 10200 Sinal $$2.33$ $$5.36$ $$1.300$ $$1.000$ <	ice Abuse	0.00	1.0000	1.0000	<i>(</i> -	1.1468	1.0000	-	1.0000	-	-	0.00
Storal and Other State Plan Services $$2.3.63$ $$1000$ 10633 1100 leity Surgey Center $$0.37$ 10000 10633 11000 leity Surgey Center $$0.37$ 10000 10633 11010 leity Surgey Center $$0.37$ 10000 10633 110100 rive Medicine $$0.200$ 10000 10633 110100 rive Medicine $$0.200$ 10000 10633 10100 rive Medicine $$0.200$ 10000 10633 10100 rive Medicine $$0.21000$ 10000 10633 10100 rive Medicine $$0.200$ 10000 10000 10000 rive Medicine $$0.200$ 10000 10000 10000 rive Medicine $$0.200$ $$0.200$ $$0.000$ 10000 10000 rive Medicine $$0.200$ $$0.000$ $$0.000$ $$0.000$ $$0.000$ rive Medicine $$0.000$ $$0.000$ $$0.000$ $$0$		5.13	1.0000	1.0604		1.1468	1.0000		1.0000			6.36
Seriend and Other State Pan Serves 50.37 10000 10807 11000 10807 11000 10807 11000 10807 110100 10807 110100 10807 110100 10807 110100 10807 110100 10807 110100 10807 110100 10807 110100 10807 110100 10807 110100 <td></td> <td>\$23.63</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>\$29.31</td>		\$23.63										\$29.31
Riory Surgery Center 50.31 1.0000 1.0633 1.0100 Riory Surgery Center 55.3 1.0000 1.0633 1.0100 Visits 4.41 1.0000 1.0633 1.0100 Visits 4.41 1.0000 1.0633 1.0100 Visits 4.41 1.0000 1.0633 1.0100 Rice Matchine 0.01 1.0000 1.0633 1.0100 Rice Matchine 0.02 1.0000 1.0633 1.0100 Rice Matchine 0.03 1.0000 1.0633 1.0100 Rice Matchine 0.03 1.0000 1.0633 1.0100 Rice Matchine 0.03 1.0000 1.0633 1.0100 Rice Matchine 0.00 1.0000 1.0633 1.0100 Rice Matchin	tional and Other State Plan Services	:::										: ::
Means In Years Model (Markine (Markine) Markin (Markine) Markine)	tory Surgery Center	\$0.37	1.0000	1.0633			1.0000		1.0000		- •	\$0.40 11.13
International 0.00 1.000 1.0620 1.000 1.0620 1.000 eff Midwife 0.01 1.0000 1.0620 1.0100 1.0620 1.0100 eff Midwife 0.01 1.0000 1.0620 1.0100 1.0620 1.0100 eff Midwife 0.02 1.0000 1.0620 1.0100 1.0620 1.0100 lefter and Substance Abuse 0.122 1.0000 1.0620 1.0100 1.0620 1.0100 lefter and Substance Abuse 0.02 0.000 1.0000 1.0620 1.0100 lefter and 0.02 0.000 1.0000 1.0630 1.0100 lefter and 0.000 1.0000 1.0630 1.0100 1.0100 Meter and Surat Health Clinics 0.00 1.0000 1.0000 1.0100 1.0100 Meter and Surat Health Clinics 0.00 1.0000 1.0000 1.0000 1.0000 1.0100 and Care 0.000	isits ive Medicine	4.40	1 0000	1 0001					1 0000		- •	4.78
circle 0.00 1.000 1.052 1.0100 (S) 4.41 1.0000 1.0654 1.0100 (S) 4.15 1.0000 1.0654 1.0100 (A) 1.0000 1.0654 1.0100 (A) 1.0000 1.0654 1.0100 (A) 1.0000 1.0654 1.0100 (A) 0.02 1.0000 1.0653 1.0100 (A) 0.020 1.0000 1.0653 1.0100 (A) 0.000 1.0650 1.0100 1.0100 (A) 0.000 1.0650 1.0100 1.0100 (A) 1.0000 1.0650 1.0100 1.0100 (A) 1.0000 1.0650 1.0100 1.0100 (A) 1.0000 1.0654 1.0100 1.0100 (A) 0.000 1.0000 1.0650 1.0100 (A) 1.0000 1.0650 1.0100 1.0100 (A) A 1.0000 1		0.02	1 0000	1 0630			1 0000		1 0000			0.03
IST 4.41 1,000 1,033 1,010 Init: and Substance Abuse 1,1 1,000 1,053 1,010 Init: and Substance Abuse 1,2 1,000 1,053 1,010 Inent: and Substance Abuse 1,2 1,000 1,053 1,010 Inenterator 1,2 1,000 1,053 1,010 Intergenor 0,02 1,000 1,053 1,010 Intergenor 1,7 1,000 1,053 1,010 Aby Called and Kural Health Clinics 0,0 1,000 1,063 1,010 Abordical Day Care 0,0 1,000 1,063 1,010 Abordical Day Care 0,0 1,000 1,000 1,010 Abordical Day Care 1,0 0,0 1,000 1,000 1,010	1 Midwife	0.00	1.0000	1.0629			1.0000		1.0000			0.00
latic and Substance Abuse 4.15 1.000 1.06:3 1.100 eigy and Patholegy 1.72 1.0000 1.06:3 1.0100 Health and Phreete Duty Nursing 1.72 1.0000 1.06:30 1.0100 Health and Phreete Duty Nursing 1.72 1.0000 1.06:30 1.0100 Incore 0.02 1.0000 1.06:30 1.0100 Incentration 0.03 1.0000 1.06:30 1.0100 Incentration 0.00 1.0000 1.06:30 1.0100 Medical Equipment 4.18 1.0000 1.06:31 1.0100 Medical Equipment 3.96 1.0000 1.06:31 1.0100 Behavioral Analysis 8.93.01 1.0000 1.06:31 1.0100 Source Brand Soure Brand 1.50:30 </td <td>ST</td> <td>4.41</td> <td>1.0000</td> <td>1.0630</td> <td></td> <td>t</td> <td>1.0000</td> <td>-</td> <td>1.0000</td> <td>ι,</td> <td>-</td> <td>4.73</td>	ST	4.41	1.0000	1.0630		t	1.0000	-	1.0000	ι,	-	4.73
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	tric and Substance Abuse	4.15	1.0000	1.0633		-	1.0000	£-	1.0000		£-	4.45
Treatm and Prvate Luy Nursing 1.72 1.0000 1.0620 1.0100 Incertan Ury Nursing 1.72 1.0000 1.0630 1.0100 Incertan Ury Nursing 1.77 1.0000 1.0630 1.0100 Incertan Ury Nursing 0.22 1.0000 1.0630 1.0100 In Contrint Program 0.000 1.0000 1.0630 1.0100 aly Qualified and Rural Health Chrics 0.00 1.0000 1.0630 1.0100 aly Qualified and Rural Health Chrics 0.00 1.0000 1.0630 1.0100 aly Qualified and Rural Health Chrics 0.00 1.0000 1.000 1.000 all Carlos 0.00 1.0000 1.000 1.000 1.0100 all Carlos 0.000 1.0000 1.000 1.000 1.0100 all Carlos 0.000 1.0000 1.000 1.000 1.000 all Carlos 0.000 1.0000 1.000 1.0000 1.000 <t< td=""><td>gy and Pathology</td><td>1.24</td><td>1.0000</td><td>1.0554</td><td></td><td></td><td>1.0000</td><td>. .</td><td>1.0000</td><td></td><td>.</td><td>1.32</td></t<>	gy and Pathology	1.24	1.0000	1.0554			1.0000	. .	1.0000		.	1.32
Interaction 0.342 1.0000 1.0500 1.0100 inregrout Transportation 0.32 1.0000 1.0500 1.0100 inregrout Transportation 0.00 1.0000 1.0500 1.0100 inregrout Transportation 0.00 1.0000 1.0000 1.0100 Merical Direct 0.00 1.0000 1.0631 1.0100 Merical Direct 0.00 1.0000 1.0631 1.0100 Merical Direct 0.00 1.0000 1.0631 1.0100 Merical Direct 0.00 1.0000 1.0000 1.0000 d Bahavioral Analysis 3.96 1.0000 1.0000 1.0000 d Bahavioral Analysis 3.27 <td>fealth and Private Duty Nursing</td> <td>1.72</td> <td>1.0000</td> <td>1.0628</td> <td></td> <td></td> <td>1.0000</td> <td></td> <td>1.0000</td> <td></td> <td></td> <td>1.84</td>	fealth and Private Duty Nursing	1.72	1.0000	1.0628			1.0000		1.0000			1.84
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	ilite aerrenev Transnortation	0.42	1.0000	1 0630				- •	1.0000		- •	0.4.0
ally Qualified and Furtal Health Clinics 4.71 10000 1185.5 10100 10000 </td <td>Treatment Program</td> <td>00.00</td> <td>1.0000</td> <td>1 0000</td> <td></td> <td></td> <td>1 0000</td> <td></td> <td>1.0000</td> <td>,</td> <td></td> <td>0000</td>	Treatment Program	00.00	1.0000	1 0000			1 0000		1.0000	,		0000
Modical Early Care 0.00 1.0000 1.0000 1.0100 1.0000	ly Qualified and Rural Health Clinics	4.77	1.0000	1.1635		-	1.0000	-	1.0000	(-	~	5.60
and Care 0.00 1.000 1.0631 1.0100 le Media Funjment 0.00 1.0000 1.0631 1.0100 d Behavioral Analysis 3.96 1.0000 1.0631 1.0100 d Behavioral Analysis 3.96 1.0000 1.0631 2.7316 iption Drugs 54 314 1.000 1.0001 1.0001 iption Drugs 54 314 1.0000 1.0000 1.0000 Scripts 54 314 1.0000 1.0000 1.0000 Scripts 54 31 1.0000 1.0000 1.0000 Scripts 54 31 1.0000 1.0000 1.0000 Scripts 54 31 1.0000 1.0000 1.0000 Scripts 55 2 1.0000 1.0000 1.0000 Scripts 55 2 1.0500 1.0000 1.0000 Management 55 2 1.0500 1.0000 1.0100 Inscriptal 55 2 1.0500 1.0000 1.0100 Inscriptered </td <td>edical Day Care</td> <td>00.00</td> <td>1.0000</td> <td>1.0000</td> <td></td> <td>-</td> <td>1.0000</td> <td><i>t</i>-</td> <td>1.0000</td> <td></td> <td></td> <td>0.00</td>	edical Day Care	00.00	1.0000	1.0000		-	1.0000	<i>t</i> -	1.0000			0.00
enerical cultification 4.18 1.0000 1.0001 1.0001 1.0101 1.0100 1.0000	al Care	00.0	1.0000	1.0631			1.0000		1.0000			00.0
a Determination 3.30 1.000 1.050 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.010 1.000	Medical Equipment	4.18	1.0000	1600.1		- •	0000	•	0000		•	4.20
Statistic <	Dellaviolal Allalysis	0.90 8.20	1 0000	1 1011			1.0000	1.0000	1 0000	1 0000	1 0000	9 12
interior Drugs S8.90 1,0000 1,0100		\$49.14										\$60.60
Scripts 58.90 1,0000 1,0100<	ption Drugs											
Source Brand 9.62 bit with a stand 1.000 bit with a stand 1.0100 bit with a stand	Scripts	\$8.90	1.0000	1.0000		1.0064	1.0000		1.0000	~	·	00.6\$
Anstruct 1.73 1.0000 1.0100 0.0	source Brand	9.62	1.0000	1.0000		0.9747	1.0000		1.0000	(- (8.53
fis C 0.00 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0100 <td></td> <td>15 09</td> <td></td> <td>0000</td> <td>-</td> <td>2.3276</td> <td>1 0000</td> <td></td> <td>1 0000</td> <td></td> <td></td> <td>34.63</td>		15 09		0000	-	2.3276	1 0000		1 0000			34.63
0.00 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0100 <td>sc</td> <td>00.00</td> <td></td> <td>1 0000</td> <td></td> <td>1.0000</td> <td>1.0000</td> <td>0.9894</td> <td>1.0000</td> <td>1.0000</td> <td>1.0000</td> <td>0.00</td>	sc	00.00		1 0000		1.0000	1.0000	0.9894	1.0000	1.0000	1.0000	0.00
S35.40 S35.40 1.0500 1.1525 1.0100 1 \$0.52 1.0500 1.1525 1.0100 1 10100 1 0.03 1.0500 1.0500 1.0500 1.0100 1 10100 1 0.03 1.0500 1.0500 1.0000 1.0100 1 10100 1 0.04 0.00 1.0500 1.0500 1.0000 1.0100 1 1 0.00 0.00 1.0500 1.0000 1.0100 1		0.00	`	1.0000		1.0064	1.0000		1.0000	,		0.00
\$0.52 1.0500 1.1525 1.0100 0.33 1.0500 1.0660 1.0100 0.33 1.0500 1.0660 1.0100 0.00 1.0500 1.0616 1.0100 0.01 1.0500 1.0000 1.0100 0.024 1.0500 1.0000 1.0100 0.00 0.00 1.0500 1.0000 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.01 1.0500 1.0000 1.0100 0.1050 1.0500 1.0500 1.0100 0.1050 1.0500 1.0500 1.0100 0.1050 1.0500 1.0100 1.0100 0.1050 1.0500 1.0582 1.0100 0.1050 1.0560 1.0560 1.0100	14	\$35.40										\$54.40
0.33 1.0500 1.01660 1.0100 0.00 1.0500 1.0100 1.0100 0.24 1.0500 1.0100 1.0100 0.01 1.0500 1.0100 1.0100 0.02 1.0500 1.0100 1.0100 0.01 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.01 1.0500 1.0000 1.0100 0.01 1.0500 1.0000 1.0100 0.01 1.0500 1.0000 1.0100 0.027 1.0500 1.0000 1.0100 1.1.77 1.0500 1.0700 1.0100 1.1.78 1.0500 1.0700 1.0100 1.1.78 1.0500 1.0700 1.0100 1.1.78 1.0500 1.0700 1.0100 1.1.58 1.0700 1.0100 1.0100	anagement	\$0.52	1.0500	1.1525			1.0000	1.0000	1.0000	-	·	\$0.63
0.00 1.1500 1.0000 1.0100 1.0100 0.24 1.15500 1.0600 1.0100 1.0100 0.00 1.15500 1.0600 1.0100 1.0100 0.00 1.15500 1.0600 1.0100 1.0100 0.00 1.15500 1.0600 1.0100 1.0100 0.00 1.0500 1.0000 1.0100 1.0100 0.00 1.0500 1.0000 1.0100 1.0100 0.01 1.0500 1.0000 1.0100 1.0100 0.01 1.0500 1.0000 1.0100 1.0100 0.047 1.0500 1.0600 1.0100 1.0100 51:57 1.0500 1.0700 1.0100 1.0100	erm Support Service	0.33	1.0500	1 0660		~	1.0000		1.0000	-	~	0.38
0.24 1.1500 1.0616 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.01 1.0500 1.0000 1.0100 0.01 1.0500 1.0000 1.0100 0.47 1.0500 1.0582 1.0100 1.57 1.0500 1.0582 1.0100	Hospital	0.00	1.0500	1.0000	·	.	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
0.00 1.0000 1.0000 1.0100 1.0100 0.00 1.0500 1.0600 1.0100 1.0100 0.00 1.0500 1.0000 1.0100 1.0100 0.00 1.0500 1.0600 1.0100 1.0100 0.01 1.0500 1.0000 1.0100 1.0100 0.07 1.0500 1.0500 1.0100 1.0100 0.47 1.0500 1.0582 1.0100 1.0100 \$11.57 \$1.570 1.0582 1.0100 1.0100	nerapy	0.24	00501	1.0016			0000		1.0000			12.0
0.00 1.500 1.000 1.0100 0.00 1.6500 1.0000 1.0100 0.00 1.9500 1.0000 1.0100 0.01 1.9500 1.0000 1.0100 0.02 1.0500 1.0100 1.0100 0.01 1.9500 1.0500 1.0100 1.157 1.0500 1.0582 1.0100 1.137 1.3500 1.0582 1.0100	e Based Practice ion Management	0.00	1.0500						1.0000		•	00.0
0.00 1.0500 1.0000 1.0100 <td>nov Service 24/7</td> <td>00.0</td> <td>1 0500</td> <td>1 0000</td> <td></td> <td></td> <td>100001</td> <td></td> <td>1 0000</td> <td></td> <td></td> <td>0.00</td>	nov Service 24/7	00.0	1 0500	1 0000			100001		1 0000			0.00
0.00 1.0500 1.0000 1.0100 1 0.47 1.0500 1.0582 1.0100 1 81.57 1.0500 1.0582 1.0100 1 81.57 1.0500 1.0582 1.0100 1		0.00	1.0500	1.0000			1.0000		1.0000			0.00
0.47 1.0500 1.0582 1.0100 1 \$1.57 51.38 \$113.88	ed Employment Services	00.00	1.0500	1.0000	·		1.0000	·	1.0000	-	<i>t</i>	00.00
\$1		0.47	1.0500	1.0582			1.0000		1.0000	-		0.53
	ioc	\$1.5/										\$1.81 ¢160.60
		0000110	-									000010

Eligibility Category: Foster Care / Adoption

Benefits	Per Capita Monthly Paid Cost	IBNR Adiustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiustment	PDL Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adiustment	Projected Per Capita Monthly Paid Cost
Hospital Inpatient			a tradition fait (strating fast r		a statute of the stat		analian fat s	
Medical	\$3.63	1.0000	1.1060		-	0.9938		1.0000	<i>t</i> -	1.0000	\$4.02
Surgical	0.74	1.0000	1.0809			0.9938		1.0000		1.0000	0.80
Maternity Delivery Maternity Non Dalivery	00.0	1,0000	1,0000			0.9930		0000 1		1.0000	00.0
	00.00	1.0000	1.0000			0.9938	·	1.0000	•	1.0000	000
Psychiatric	14.60	1.0000	1.0900		-	0.9938		1.0000	, -	1.0000	15.91
Alcohol and Drug Abuse	00.0	1.0000	1.0000	1.0000	1.0060	0.9938	1.0000	1.0000	1.0000	1.0000	00.0
Ottel	\$18.97	0000-1	0000			0.9900		0000-1		0000-1	\$20.72
Hospital Outpatient											
Emergency Room	\$12.58	1.0000	1.0604	1.0200	1.1468	0.9938	1.0000	1.0000	1.0000	1.0000	\$15.51
Surgery	4.69	1.0000	1.0604	•	1.1468	0.9938	1.0000	1.0000	-	1.0000	5.78
Radiology	4.15	1.0000	1.0604		1.1468	0.9938		1.0000		1.0000	5.12
Phomonogy	66.1	1.0000	1.0004		1.1408	0.9938	- •	0000 1		1.0000	CH-7
r nannacy Cardiovascular	0.95	1 0000	1.0604		1.1468	0.9938		1.0000	•	1 0000	0.30
PT/OT/ST	3.45	1.0000	1.0604		1.1468	0.9938		1.0000	``	1.0000	4.25
Psychiatric	0 03	1.0000	1 0604		1.1468	0.9938		1 0000		1.0000	0.03
Substance Abuse	0.00	1.0000	1.0000		1.1468	0.9938	-	1.0000	,	1.0000	0.00
Other	9.53	1.0000	1.0604		1.1468	0.9938	1.0000	1.0000	,	1.0000	11.75
	\$37.58										\$46.32
Protessional and Other State Plan Services Ambulatory Surgery Center	\$0.42	1 0000	1 2084		1 0000	0 9938	1 0000	1 0000	1 0000	1 0000	\$0 5U
Office Visits	12.46	1.0000	1.0598		1.0000	0.9938	1.0000	1.0000	1.0000	1.0000	13.26
Preventive Medicine	4.14	1.0000	1.0793		1.0000	0.9938		1.0000		1.0000	4.49
Maternity	0.12	1.0000	1.0630	-	1.0000	0.9938	-	1.0000		1.0000	0.13
Certified Midwife	00.00	1.0000	1.0000	-	1.0000	0.9938	1.0000	1.0000	-	1.0000	00.0
PT/OT/ST	5.32	1.0000	1.0631	,	1.0000	0.9938	1.0000	1.0000		1.0000	5.68
Psychiatric and Substance Abuse	13.71	1.0000	1.0630		1.0000	0.9938	_	1.0000		1.0000	14.62
Radiology and Pathology Homo Hoalth and Drivato Duty Nursing	2.3/	1.0000	086U.I 0290 1	•	1.0000	0.9938		1.0000		1.0000	70.7
Ambulance	1.54	1.0000	1.0630		1.0000	0.9938		1.0000		1.0000	1.64
Non-Emergency Transportation	2.20	1.0000	1.0630		1.0000	0.9938		1.0000		1.0000	2.35
Opioid Treatment Program	00.00	1.0000	1.0000	-	1.0000	0.9938	-	1.0000	-	1.0000	00.0
Federally Qualified and Rural Health Clinics	7.91	1.0000	1.1581		1.0000	0.9938	-	1.0000	-	1.0000	9.19
Adult Medical Day Care	0.00	1.0000	1.0000		1.0000	0.9938	1.0000	1.0000		1.0000	0.00
Personal Care	0.00	1.0000	0000.1		1.0000	0.9938	1.0000	1.0000		1.0000	0.00
Durable Medical Equiprirent Annlied Rehavioral Analysis	6.14	1.0000	1.0630		1.0000	0.9930	1.0000	1.0000		1.0000	12 27
Other	12.40	1.0000	1.0470	1.0100	1.0000	0.9938	1.0000	1.0000	1.0000	1.0000	13.03
	\$123.37										\$136.27
Prescription Drugs											
Generic Scripts	\$32.74	1.0000	1.0000		0.7899	0.9938		1.0000		1.0000	\$25.55
Single-source brain Multi-Source Brand	7 15	1 0000	1 0000	·	2802-1	0.9930		10000		1 0000	8.40
Specialty	19.99	1.0000	1.0000		1.9796	0.9938		1.0000	•	1.0000	38.71
Hepatitis C	0.00	1.0000	1.0000	1.0000	1.0000	0.9938	0.9894	1.0000	1.0000	1.0000	0.00
Other	0.00	1.0000	1.0000	·	0.7899	0.9938		1.0000		1.0000	0.00
Community Mental Health Conter	\$/8.23										\$94.32
Community mental reality center Case Management	\$2.42	1.0500	1.1513		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$2.96
Long Term Support Service	0.97	1.0500	1.0691	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.10
Partial Hospital	0.00	1.0500	1.0000		1.0000	1.0000	1.0000	1.0000		1.0000	00.0
Psychotherapy	1.02	1.0500	1.0566		1.0000	1.0000	1.0000	1.0000		1.0000	1.14
Evidence Based Practice Modionion Monocomout	0.00	1.0500	1.0000		1.0000	1.0000	1.0000	1.0000		1.0000	0.00
ivieurcauon management Emernency Service 24/7	00.0	1 0500	1 0000	-	1 0000	1.0000	1 0000	1.0000	,	1.0000	00.0
APRTP	0.00	1.0500	1.0000	-	1.0000	1.0000	1.0000	1.0000	,	1.0000	0.00
Supported Employment Services	00.00	1.0500	1.0000		1.0000	1.0000	1.0000	1.0000	v -	1.0000	0.00
Other	2.30	1.0500	1.0610		1.0000	1.0000	1.0000	1.0000		1.0000	2.59
All Candada	1/ Q\$										6/-18
	00-1070										01-0000

Appendix B1 New Hampshire Department of Heath and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Severely Disabled Children

Renefite	Per Capita Monthly Paid Cost	IBNR Adiustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiustment	PDI Adiustment	Medicare OUD Adjustment	Stop Loss Adiustment	NEMT Adiustment	Projected Per Capita Monthly Paid Cost
Hospital Inpatient	1000		NINING NO.	1 4441		unamenta.		undimention in	NINING NO.	unamentau.	
Medical	\$29.82	1 0000	1 0943	1 0000	1 0060	0 9890	·	1 0000		1 0000	\$32.40
Sumical	13.54	1.0000	1.1079	1.0000	1.0060	0.9899		1.0000		1.0000	14.93
Matemity Delivery	00.00	1 0000	1.0000	1.0000	1.0060	0.9899		1.0000		1.0000	00.00
Maternity Non-Delivery	00.00	1.0000	1.0000	1.0000	1.0060	0.9899		1.0000		1.0000	00.0
Well Newborn	0.00	1.0000	1.0000	1.0000	1.0060	0.9899		1.0000		1.0000	0.00
Psychiatric	1.18	1.0000	1.1351	1.0000	1.0060	0.9899	·	1.0000		1.0000	1.34
Alcohol and Drug Abuse	00.0	1.0000	1.0000	1.0000	1.0060	0.9899	1.0000	1 0000	1.0000	1.0000	0.00
Ollei	0.00 S44.54	1-000	1.000	1.000	nonn I	0.9089	1-0000	1-0000		0000-1	0.00 \$48.77
Hospital Outpatient											1014
Emergency Room	\$7.70	1.0000	1.0604	1.0200	1.1468	0.9899	1.0000	1.0000		·	\$9.46
Surgery	7.74	1.0000	1.0604	1.0200	1.1468	0.9899	1.0000	1.0000		·	9.50
Radiology	11.60	1.0000	1.0604	1.0200	1.1468	0.9899	1.0000	1.0000			14.24
Pathology	2.65	1.0000	1.0604	1.0200	1.1468	0.9899	-	1.0000			3.25
Pharmacy	2.38	1.0000	1.0604	1.0200	1.1468	0.9899		1.0000			2.93
Cardiovascular	0.39	1.0000	1.0604	1.0200	1.1468	0.9899	-	1.0000			0.48
PT/OT/ST	11.19	1.0000	1.0604	1.0200	1.1468	0.9899	,	1.0000			13.74
Psychiatric	0.02	1.0000	1.0604	1.0200	1.1468	0.9899		1.0000			0.03
Substance Abuse	0.00	1.0000	1.0000	1.0200	1.1468	0.9899	1.0000	1.0000	1.0000	1.0000	0.00
Ollei	\$53.75		1.000	1.0200	1.1400	n-2029	1-0000	1-0000			\$66.00
Professional and Other State Plan Services	01.000										00.000
Ambulatory Surgery Center	\$0.45	1.0000	0.8035	-	1.0000	0.9899	1.0000	1.0000		1.0000	\$0.36
Office Visits	16.51	1.0000	1.0603	-	1.0000	0.9899	-	1.0000		1.0000	17.50
Preventive Medicine	1.78	1.0000	1.0678	-	1.0000	0.9899	-	1.0000		1.0000	1.90
Maternity	00.00	1.0000	1.0000	-	1.0000	0.9899	,	1.0000		1.0000	00.0
Certified Midwife	0.00	1.0000	1.0000	-	1.0000	0.9899	-	1.0000		1.0000	0.00
PI/OI/ST	28.21	1.0000	1.0630		1.0000	0.9899		1.0000	•	1.0000	29.99
Psychiatric and Substance Abuse	0.79	1.0000	1.0431	- •	1.0000	0.9099	- •	0000 1		1.0000	9.17
Radiology and Fadiology Home Health and Private Duty Nursing	355.40	1 0000	1.00/4		1 0000	0.9099	- •	1 0000		1 0000	3.77.67
Ambulance	2.39	1.0000	1.0630		1.0000	0.9899		1.0000		1.0000	2.54
Non-Emergency Transportation	7.10	1.0000	1.0630		1.0000	0.9899	· ·	1.0000		1.0000	7.54
Opioid Treatment Program	00.00	1.0000	1.0000	-	1.0000	0.9899	-	1.0000	·	1.0000	0.00
Federally Qualified and Rural Health Clinics	1.30	1.0000	1.1574	-	1.0000	0.9899	-	1.0000		1.0000	1.50
Adult Medical Day Care	0.00	1.0000	1.0000		1.0000	0.9899		1.0000		1.0000	0.00
Personal Care Duroble Medical Equipment	140.67	1.0000	1.000	- •	00001	0.9699	- •	0000 1		00001	10.01
Annlied Rehavioral Analysis	10.04	1 0000	1 0630	- (1	1 0000	0.9090	- •	1 0000		1 0000	130.52
Other	59.95	1.0000	1.1023	1.0100	1.0000	0.9899	1.0000	1.0000	1.0000	1.0000	66.07
	\$682.14										\$852.12
Prescription Drugs											
Generic Scripts	\$47.69 01.50	1.0000	1.0000	1.0000	1.5799	0.9899		1.0000		1.0000	\$76.04
Siligle-Source Brand Multi-Source Brand	50.03 66.68	1 0000	1.0000	1.0000	1.7965	0.9099		1.0000		1 0000	100.01
Specialty	108.01	1.0000	1.0000	1.0000	1 7020	0.9899		1.0000	-	1.0000	181.23
Hepatitis C	0.00	1.0000	1.0000	1.0000	1.0000	0.9899	0.9894	1.0000	1.0000	1.0000	0.00
Other	0.02	1.0000	1.0000	1.0000	1.5799	0.9899		1.0000		1.0000	0.04
Commission Mondel Haadth Control	\$257.94										\$425.88
	\$1.15	1.0500	1.1530	1.0100	1.0000	1.0000	1.0000	1.0000		·	\$1.41
Long Term Support Service	2.01	1.0500	1.1276	1.0100	1.0000	1.0000	1.0000	1.0000	, -	·	2.40
Partial Hospital	00.00	1.0500	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000	,	·	0.00
Psychotherapy	0.20	1.0500	1.1000	1.0100	1.0000	1.0000	1.0000	1.0000			0.24
Evidence Based Practice Modination Monocomput	0.00	1.0500	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000			0.00
Medication Mariagement Emernency Service 24/7	00.0	1.0500	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000			00.0
APRTP	00.0	1.0500	1.0000	1.0100	1 0000	1.0000	1.0000	1.0000	,		0.00
Supported Employment Services	00.00	1.0500	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Other	0.58	1.0500	1.0603	1.0100	1.0000	1.0000	1.0000	1.0000			0.66
	\$3.95										54.71 54.007.47
All Services	\$1,042.32	-					-				\$1,397.47

5

Eligibility Category: Elderly and Disabled Adults - Age 19-64

Banafite	Per Capita Monthly Paid Cost	IRNR Aditistment	Reimbursement Adiuctment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiuctment	PDI Adiustment	Medicare OUD	Stop Loss Adiustment	NEMT Adjuctment	Projected Per Capita Monthly Paid Cost
Hospital Inpatient	6031			1 441013	1 401013	שמתפווו		Valasment	Valasment	Valasment	raid cost
Medical	\$66.13	1.0000	1.0849	<i>t</i> -	1.0060	0.9891	1.0000	1.0000	0.9946		\$70.99
Surgical	39.68	1.0000	1.0963		1.0060	0.9891	1.0000	1.0000	0.9946		43.04
Matemity Delivery	00.00	1.0000	1.0000	-	1.0060	0.9891	1.0000	1.0000	0.9946		00.0
Maternity Non-Delivery	0.15	1.0000	1.1035	1.0000	1.0060	0.9891	1.0000	1.0000	0.9946		0.17
Psychiatric	2.99	1.0000	1.1847		1.0060	0.9891	1.0000	1.0000	0.9946		7.02
Alcohol and Drug Abuse	2.31	1.0000	1.1382	1.0000	1.0060	0.9891	1.0000	1.0000	0.9946	1.0000	2.60
Other	0.00	1.0000	1.0000	-	1.0060	0.9891	1.0000	1.0000	0.9946		0.00
	\$114.25										\$123.82
Hospital Outpatient Emergency Room	852 35	1 0000	1 0604	1 0200	1 1468	0 9801	1 0000	1 0000	0 0046	1 0000	86.1 88
Surgerv	19.77	1.0000	1.0604	1.0200	1.1468	0.9891	1.0000	1.0000			24.12
Radiology	27.23	1.0000	1.0604	1.0200	1.1468	0.9891	1.0000	1.0000			33.23
Pathology	7.89	1.0000	1.0604	1.0200	1.1468	0.9891	1.0000	1.0000			9.63
Pharmacy	40.81	1.0000	1.0604	1.0200	1.1468	0.9891	1.0000	1.0000			49.79
Cardiovascular	1.95	1.0000	1.0604	1.0200	1.1468	0.9891	1.0000	1.0000			2.38
PT/OT/ST	6.64	1.0000	1.0604	1.0200	1.1468	0.9891	1.0000	1.0000			8.10
Psychiatric	0.12	1.0000	1.0604	1.0200	1.1468	0.9891	1.0000	1.0000			0.15
Substance Apuse Other	0.10 31.37	1.0000	1.0604	1.0200	1.1468	0.9891	1.0000	1.0000	0.9946	1.0000	21.0
10110	\$188.23		-			0					\$229.67
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$2.40	1.0000	1.1518	-	1.0000	0.9891	1.0000	1.0000	0.9946	1.0000	\$2.75
Office Visits	28.24	1.0000	1.0632			0.9891	1.0000	1.0000	0.9946		29.83
PTEVERITIVE IMEDICITIE Matemity	0.33	1.0000	1.1514			0.9091	1,0000		0.9940		0.10
Certified Midwife	0.00	1.0000	1.0629			0.9891	1.0000	1.0000	0.9946		0.00
PT/OT/ST	3.93	1.0000	1.0630			0.9891	1.0000	1.0000	0.9946		4.15
Psychiatric and Substance Abuse	14.64	1.0000	1.1716	1.0100	-	0.9891	1.0000	1.0000	0.9946		17.04
Radiology and Pathology	12.63	1.0000	1.0567			0.9891	1.0000	1.0000	0.9946		13.26
Home Health and Private Duty Nursing Ambulance	00.00 7 7 7	1.0000	1.0029			0.9891	1.0000	0000.1	0.9940		CO 60
Non-Emergency Transportation	28.20	1.0000	1.0630	1.0100	1.0000	0.9891	1.0000	1.0000	0.9946		29.79
Opioid Treatment Program	9.56	1.0000	1.0636	-	1.0000	0.9891	1.0000	1.0000	0.9946		10.10
Federally Qualified and Rural Health Clinics	17.79	1.0000	1.1587		1.0000	0.9891	1.0000	1.0000	0.9946	1.0000	20.48
Adult Medical Day Care	2.76	1.0000	1.0630	- ·	1.0000	0.9891	1.0000	1.0000	0.9946	1.0000	2.91
r ersonal Care Durable Medical Equipment	37.29	1.0000	0.9353	1.0100	1.0000	0.9891	1.0000	1.0000	0.9946	1.0000	34.66
Applied Behavioral Analysis	0.72	1.0000	1.0630	. –	1.0000	0.9891	1.0000	1.0000	0.9946	1.0000	0.75
Other	61.12	1.0000	1.1011	-	1.0000	0.9891	1.0000	1.0000	0.9946	1.0000	66.87
	\$312.92										\$331.64
Prescription Drugs	¢e7 fr		1 0000		1 1 1 7 1	0000	0000 1	0000 1	0.0046	0000 1	¢70.00
Generic Scripts Sinde Source Brand	145.58	1.0000	1 0000		1.14/1	0.9891	0 9069		0.9940		\$/0.90 146.23
Multi-Source Brand	64.88	1.0000	1.0000		1.1260	0.9891	0.6328	1.0000	0.9946		45.47
Specialty	148.98	1.0000	1.0000		1.4004	0.9891	0.9884	1.0000	0.9946		202.86
Hepatitis C	18.50	1.0000	1.0000	1.0000	0.4252	0.9891	0.9575	1.0000	0.9946	1.0000	7.41
Otter	\$435.52	0000-I	0000-1		1.14/1	0.3031	0.45.0	1.000	0.9940		\$472.97
Community Mental Health Center											
Case Management	\$0.74	1.0500	1.1530	1.0100	1.0000	1.0000	1.0000	1.0000	0.9946	-	\$0.90
Long Term Support Service	1.33	1.0500	1.0827		1.0000	1.0000		1.0000	0.9946		1.52
Partial Hospital Pevchotherany	1 76	1.0500	1.0000		1.0000	1.000		1.0000	0.9946		0.00
Evidence Based Practice	0.01	1.0500	1.0629		1.0000	1.0000		1.0000	0.9946		0.01
Medication Management	0.08	1.0500	1.0630	·	1.0000	1.0000		1.0000	0.9946	t-	0.09
Emergency Service 24/7	0.01	1.0500	1.0631	1.0100	1.0000	1.0000	1.0000	1.0000	0.9946	1.0000	0.01
APRTP Sumodod Employment Services	0.28	1.0500	1.1530		1.0000	1.0000		1.0000	0.9946		0.34
Supported Employment Services Other	2.49	1 0500	1.1330	·	1 0000	1.0000		1.0000	0.9946		9.04 2.79
	\$6.73										\$7.67
All Services	\$1,057.65										\$1,165.78

Appendix B1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Elderly and Disabled Adults - Age 65+

Banadite	Per Capita Monthly Paid Cost	IRNR Adjinstment	Reimbursement Adiuctment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings ∆diustment	PDI Adiustment	Medicare OUD	Stop Loss Adiustment	NEMT Adjustment	Projected Per Capita Monthly Paid Cost
Benerits Hospital Inpatient	0.051	IDIN VAJASTIN	Hugunen	Lactors	Lactors	Aujustifierit		Aujusunent	Aujusument	Aujustifierit	Fall COSt
Medical	\$89.19	1.0000	1.0712		-	0.9917		1.0000		1.0000	\$95.31
Surgical	39.23	1.0000	1.0708	1.0000	1.0060	0.9917	1.0000	1.0000	1.0000	1.0000	41.92
Matemity Delivery	0.00	1.0000	1.0000		-	0.9917		1.0000		1.0000	0.00
Maternity Non-Delivery	00.0	1.0000	1.0000			0.9917		1.0000		1.0000	00.0
eveninterio	3.68	1 0000	1 0145			0.9917	·	1 0000		1,0000	3 72
Alcohol and Drug Abuse	000	1 0000	1 0000			0.9917		1 0000	,	1 0000	000
Other	00.00	1.0000	1.0000		. –	0.9917		1.0000		1.0000	00.00
	\$132.10										\$140.95
Hospital Outpatient											
Emergency Room	\$42.95	1.0000	1.0604	1.0200	1.1468	0.9917	1.0000	1.0000	•	1.0000	\$52.83
Surgery	13.34	1.0000	1.0604	-		0.9917	-	1.0000	·	1.0000	16.41
Radiology	32.34	1.0000	1.0604	-	1.1468	0.9917	-	1.0000		1.0000	39.78
Pathology	8.24	1.0000	1.0604	-	-	0.9917	-	1.0000		1.0000	10.13
Pharmacy	23.83	1.0000	1.0604	-	~	0.9917	-	1.0000	,	1.0000	29.31
Cardiovascular	1.99	1.0000	1.0604	-	-	0.9917	-	1.0000		1.0000	2.44
PT/OT/ST	5.27	1.0000	1.0604	-	~	0.9917	-	1.0000		1.0000	6.48
Psychiatric	0.00	1.0000	1.0000	-		0.9917	(-)	1.0000		1.0000	0.00
Substance Abuse	0.00	1.0000	1.0000			0.9917		1.0000	1.0000	1.0000	0.00
Uner	¢168.17	nnnn-1	1-000 L		1.1400	11.66.0		0000-1		nnnn I.	49.40 ¢206.87
	11.001¢										10.0020
Professional and Other State Plan Services	00 14	1 0000	1 1000			L 100 0		0000 1	·	1 0000	0.C C 4
	55 + 6 1 - 90		1.1000		1 0000	198.0	- •	0000		1,0000	
Ornor Visita Proventive Medicine	2 6.7	1,0000	1 3366			1100 0		1 0000		1 0000	3.51
Matemity	20.2					0.0017	1 0000	1,0000		1 0000	000
reaconny Cartifiad Midwifa	00.0	00001				0.0017	1 0000	1 0000		1 0000	00.0
DT/DT/ST	1 98	1 0000	1 0631			0 9917	- (1 0000		1 0000	2.11
Psychiatric and Substance Ahuse	1 12	1 0000	1 0630	1 0100		0.9917		1 0000	1 0000	1 0000	119
Radiology and Pathology	12.27	1.0000	1.0567			0.9917		1 0000		1.0000	12.98
Home Health and Private Duty Nursing	59.36	1.0000	1.0629		-	0.9917	t-	1.0000		1.0000	63.20
Ambulance	7.70	1.0000	1.0630		-	0.9917	-	1.0000	-	1.0000	8.20
Non-Emergency Transportation	13.85	1.0000	1.0630	1.0100	1.0000	0.9917	-	1.0000	1.0000	1.0000	14.74
Opioid Treatment Program	1.03	1.0000	1.0636	-	. .	0.9917		1.0000	. .	1.0000	1.10
Federally Qualified and Kural Health Clinics	27.09	1.0000	1.1699		- •	0.991/		1.0000	- •	1.0000	25.89
Adult Medical Day Care Personal Care	31.42 0.00	1.0000	1.0030	1.0100	_	1991/	1.0000	1.0000	1.0000	1.0000	0.00
Puroble Medical Equipment	21.00	0000	0.0110			11000	- •	1.0000		1,0000	0.00
Durable Medical Equipriterit Amplied Rehavioral Analysis	0000	1 0000	1 0000			0.9917	- •	1 0000		1 0000	
Other	65.45	1.0000	1 0595	·	1.0000	0.9917		1.0000		1.0000	69.46
	\$271.85										\$289.15
Prescription Drugs											
Generic Scripts	\$45.10	1.0000	1.0000	1.0000	1.1471	0.9917	1.0929	1.0000	1.0000	1.0000	\$56.07
Single-Source Brand	132.48		1.0000			1991/		1.0000		1.0000	134.15
Mulu-Source Dianu Soociothy	32.10		1.0000			0.9917		1.0000		1.0000	41.22 BA BA
Operating Henatitis C	4 95	1.0000	1.0000			0.9917		1.0000		1.0000	2.00
Other	0.01	1.0000	1.0000			0.9917		1.0000		1.0000	0.01
	\$261.80										\$279.61
Community Mental Health Center											
Case Management	\$0.03	1.0500	1.1530		- ·	1.0000	1.0000	1.0000		1.0000	\$0.04
Long Lerm Support Service	0.24	1.0500	1.0629	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.20
raruarrospitar Psychotherany	0.17	1 0500	1 0630				1 0000	1 0000	•	1 0000	0.10
Fvidence Based Practice	0.00	1.0500	1.0000			1.0000	1.0000	1.0000	,	1.0000	0.00
Medication Management	0.00	1.0500	1.0000			1.0000	1.0000	1.0000		1.0000	0.00
Emergency Service 24/7	0.00	1.0500	1.0000		. –	1.0000	1.0000	1.0000		1.0000	00.0
APRTP	0.00	1.0500	1.0000		-	1.0000	1.0000	1.0000		1.0000	00.00
Supported Employment Services	0.00	1.0500	1.0000		- ·	1.0000	1.0000	1.0000		1.0000	0.0
Other	0.86 ¢1.20	1.0500	1.0630		-	1.0000	1.0000	1.0000		1.0000	0.9/
All Consisse	00.14 CC 3223										\$1.47 \$018.05
All OELVICES	77.000										00.0166

Appendix B1 New Hampshire Department of Heath and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Dual Eligibles

Benefits	Per Capita Monthly Paid Cost	IBNR Adiustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiustment	PDL Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adiustment	Projected Per Capita Monthly Paid Cost
Hospital Inpatient						a la company fan a				a ration from from t	
Medical	\$20.31	-	1.0712	1.0000	1.0060	1.0000	1.0000	1.0000		1.0000	\$21.89
Surgical	5.99	-	1.0708	1.0000	1.0060	1.0000	1.0000	1.0000		1.0000	6.46
Matemity Delivery	00.00	- ·	1.0000	1.0000	1.0060	1.0000	1.0000	1.0000		1.0000	0.00
Maternity Non-Delivery Moli Nowborn	10.0		1.0000	1.0000	1.0000	1.0000	00001			1.0000	/0.0
Psychiatric	3.00		1.0145	1.0000	1.0060	1.0000	1.0000	1.0000		1.0000	3.06
Alcohol and Drug Abuse	0.58	1.0000	1.0000	1.0000	1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	0.59
Other	0.00		1.0000	1.0000	1.0060	1.0000	1.0000	1.0000		1.0000	0.00
Unanitad Ordenations	928.30										00.26¢
Emergency Room	\$19.02		1.0604	1.0200	1.1468	1.0000	1.0000	1.0000		1.0000	\$23.59
Surgery	7.52	-	1.0604	1.0200	1.1468	1.0000	1.0000	1.0000	-	1.0000	9.33
Radiology	6.25	-	1.0604	1.0200	1.1468	1.0000	1.0000	1.0000		1.0000	7.75
Pathology	1.14	-	1.0604	1.0200	1.1468	1.0000	1.0000	1.0000		1.0000	1.41
Pharmacy	13.34	-	1.0604	1.0200	1.1468	1.0000	1.0000	1.0000		1.0000	16.55
Cardiovascular	0.96		1.0604	1.0200	1.1468	1.0000	1.0000	1.0000		1.0000	1.19
PI/01/SI	7.20	_	1.0604	1.0200	1.1468	1.0000	1.0000	1.0000		1.0000	2.80
Esycillaure Substance Abuse	0.04		1.0004	1.0200	1.1400	1,0000	1,0000	1.0000		0000	7470
Other	16.68	1.0000	1.0604	1.0200	1.1468	1.0000	1.0000	1.0000	1.0000	1.0000	20.69
	\$67.53										\$83.76
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$0.63	-	1.1088	1.0100	1.0000	1.0000	1.0000	1.0000	-	1.0000	\$0.71
Office Visits	9.74		1.0610	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	10.43
PTEVERITUVE IMEDICIFIE	0.19	- •	00001	1.0100	1.0000	1.0000	1.0000	00001		1.0000	1.00
Certified Midwife	000		1 0000	1 0100	1 0000	1 0000	1 0000			1 0000	000
PT/OT/ST	0.64		1.0631	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	0.69
Psychiatric and Substance Abuse	5.72	-	1.0630	1.0100	1.0000	1.0000	1.0000	1.0000	-	1.0000	6.14
Radiology and Pathology	3.08	-	1.0567	1.0100	1.0000	1.0000	1.0000	1.0000	-	1.0000	3.29
Home Health and Private Duty Nursing	12.79		1.0629	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	13.73
Ambulance	89.7 20.00	- •	1.0630	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	8.24
Noi-Entergency Transportation Onioid Treatment Program	3.61		1.0030	1.0100	1.0000	1 0000	1 0000	0.2354		1 0000	1012
Federally Qualified and Rural Health Clinics	1.50		1 1576	1 0100	1 0000	1 0000	1.0000	1 0000		1 0000	1.76
Adult Medical Day Care	1.48	-	1.0630	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	1.59
Personal Care	30.13	1.0000	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	30.43
Durable Medical Equipment	16.01	1.0000	2.0254	1.0100	1.0000	1.0000	1.0000	1.0000		1.0000	32.76
Applied benavioral Analysis Other	24.08	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	25.76
	\$144.04										\$165.59
Prescription Drugs											
Generic Scripts	\$0.99		1.0000	1.0000	0.9902	1.0000		1.0000		1.0000	\$0.97
Single-Source Brand	0.08		1.0000	1.0000	1.1838	1.0000		1.0000		1.0000	0.09
Multi-Source pranta Snecialty			1 0000	1 0000	1 0000	1 0000				1 0000	000
Hepatitis C	0.00	1.0000	1.0000	1.0000	1.0000	1.0000	0.9894	1.0000	1.0000	1.0000	0.00
Other	0.00		1.0000	1.0000	0.9902	1.0000		1.0000		1.0000	0.00
	\$1.11										\$1.11
Community Mental Health Center Case Management	\$0.35		1 1530		1 0000	1 0000	1 0000	1 0000		1 0000	\$0.43
Long Term Support Service	0.81		1.0629		1.0000	1.0000	1.0000	1.0000		1.0000	0.91
Partial Hospital	0.01	-	1.0000		1.0000	1.0000	1.0000	1.0000	,	1.0000	0.01
Psychotherapy	0.57		1.0630	-	1.0000	1.0000	1.0000	1.0000		1.0000	0.65
Evidence Based Practice	0.00		1.0000		1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Medication Management	0.05		1.0000		1.0000	1.0000	1.0000	1.0000		1.0000	0.05
Emergency Service 24/7	0.00		1.0000	- •	1.0000	1.0000	1.0000	00001		1.0000	0.00
Supported Employment Services	0.03	1.0500	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.03
Other	1.31		1.0630		1.0000	1.0000	1.0000	1.0000	•	1.0000	1.47
	\$3.27										\$3.71
All Services	\$245.90	_		_	_		_			_	\$286.22

Eligibility Category: Newborn Kick Payment

Banadite	Per Capita Monthly Paid Cost	IBNR Adirectment	Reimbursement Adjuctment	Utilization Trend Factors	Utilization Trend Unit Cost Trend	Managed Care Savings Adiustment	PDI Adiustment	Medicare OUD	Stop Loss Adiustment	NEMT	Projected Per Capita Monthly Paid Cost
Hospital Innatiant	C051	IDINY VAJASHIBII	Aujustifient	Lactors	Lactors	Aujustifierit		Aujusuneru	Aujustifierit	Aujustifierit	Lala COSt
Medical	\$2 056 22	1 0000	1 0768			1 0000	1 0000	1 0000	1 0000	1 0000	\$2 227 45
Surgical	21.59	1.0000	1.0944	¢-	~~~	1.0000	1.0000	1.0000	1.0000	1.0000	23.77
Maternity Delivery	00.00	1.0000	1.0000	<i>t</i> -	<u> </u>	1.0000	1.0000	1.0000	1.0000	1.0000	00.00
Maternity Non-Delivery	0.00	1.0000	1.0000			1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Well Newborn Devekiatric	90.0cc	1.0000	1.12/0	- •		1.0000	1.0000	1.0000	1.0000	1.0000	0.000
Alcohol and Drug Abuse	00.00	1.0000	1.0000	1.0000	1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	00.0
Other	0.00	1.0000	1.0000		-	1.0000	1.0000	1.0000	1.0000	1.0000	00.0
	\$2,427.90										\$2,648.12
Hospital Outpatient	10 100	0000	1090		T	0000	0000 1	1 0000	0000	0000 1	00 J F
Elliergency Room Surriery	2.64	1 0000	1.0604	1.0200		1,0000	1.0000	1.0000	1,0000	1.0000	97.540 87.58
Radiology	6.40	1.0000	1.0604			1.000	1.0000	1.0000	1.000	1.0000	7.94
Pathology	3.49	1.0000	1.0604			1.0000	1.0000	1.0000	1.0000	1.0000	4.33
Pharmacv	0.44	1.0000	1.0604	·		1.0000	1.0000	1.0000	1.0000	1.0000	0.55
Cardiovascular	1.46	1.0000	1.0604			1.0000	.	1.0000	1.0000	1.0000	1.82
PT/OT/ST	0.61	1.0000	1.0604			1.0000	-	1.0000	1.0000	1.0000	0.75
Psychiatric	00'0	1.0000	1.0000			1.0000	-	1.0000	1.0000	1.0000	00.00
Substance Abuse	00.00	1.0000	1.0000		1.1468	1.0000	-	1.0000	1.0000	1.0000	0.00
Other	47.72 ¢00.04	1.0000	1.0604			1.0000		1.0000	1.0000	1.0000	59.20
	922.04										\$123.04
Professional and Other State Plan Services Ambulatory Surviery Center	00.0\$	1 0000	1 0000			1 0000	1 0000	1 0000	1 0000	1 0000	\$0 D0
Office Visits	40.41	1.0000	1.0592			1.0000	1.0000	1.0000	1.000	1.0000	43.23
Preventive Medicine	108.46	1.0000	1.0628			1.0000	1.0000	1.0000	1.0000	1.0000	116.42
Maternity	0.00	1.0000	1.0000		-	1.0000	1.0000	1.0000	1.0000	1.0000	00.0
Certified Midwife	0.24	1.0000	1.0629		-	1.0000	1.0000	1.0000	1.0000	1.0000	0.26
PT/OT/ST	0.15	1.0000	1.0630		-	1.0000	1.0000	1.0000	1.0000	1.0000	0.16
Psychiatric and Substance Abuse	0.00	1.0000	1.0000			1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Radiology and Pathology	8.30	1.0000	1.055/		- •	1.0000	1.0000	1.0000	1.0000	1.0000	68.8 0.60
nome neatin and Private Duty Nuising Ambulance	20.84	1 0000	1.0620			1 0000	1.0000	1.0000		1.0000	25.22
Non-Emergency Transportation	1.93	1.0000	1.0630			1.0000	1.0000	1.0000	1.0000	1.0000	2.07
Opioid Treatment Program	00.00	1.0000	1.0000		-	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Federally Qualified and Rural Health Clinics	74.90	1.0000	1.1682		-	1.0000	1.0000	1.0000	1.0000	1.0000	88.37
Adult Medical Day Care	0.00	1.0000	1.0000		÷ ,	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Personal Care	0.00	1.0000	1.0000			1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Durable Medical Equipment Amaliod Bohavioral Analysis	9.14	1.0000	0.9/96			1.0000	1.0000	1.0000	1.0000	1.0000	9.04
Other	354.14	1.0000	1.0625	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	380.03
	\$646.17										\$700.50
Prescription Drugs											
Generic Scripts Single Service Brand	\$1.81	1.0000	1.0000			1.0000	0.9894	1.0000	1.0000	1.0000	\$1.80 0.64
olligie oourde brand Multi-Source Brand	0.07	1 0000	1 0000			1 0000		1 0000	1 0000	1 0000	0.06
Specialty	0.04	1.0000	1.0000			1.0000		1 0000	1.0000	1.0000	0.10
Hepatitis C	0.00	1.0000	1.0000	1.0000	1.0000	1.0000	0.9894	1.0000	1.0000	1.0000	0.00
Other	00.00	1.0000	1.0000			1.0000		1.0000	1.0000	1.0000	0.00
0	\$2.80										\$2.81
Community mental realth Center Case Management	\$0.00	1.0500	1.0000			1.0000	1.0000	1.0000	1.0000	1.0000	\$0.00
Long Term Support Service	0.00	1.0500	1.0000			1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Partial Hospital	0.00	1.0500	1.0000			1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Psychotherapy	00.00	1.0500	1.0000		-	1.0000	1.0000	1.0000	1.0000	1.0000	00.00
Evidence Based Practice	0.00	1.0500	1.0000			1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Medication Management	0.00	00001	1.0000		- •	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
EIIIBIJBIIU) SBINUS 24/7 APRTP		1 0500	1 0000	1 0100	1.0000	1 0000	1 0000	1 0000	1 0000	1.0000	00.0
Supported Employment Services	00.00	1.0500	1.0000			1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Other	0.00	1.0500	1.0000			1.0000	1.0000	1.0000	1.0000	1.0000	0.00
	\$0.00										\$0.00 20.125 00
All Services	\$3,1/0./1										\$3,475.28

Appendix B1 New Hampshire Department of Hashin and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Neonatal Abstinence Syndrome Kick Payment

Ithert Signation 1000 1148	Reimbursement Utilization Trend	Unit Cost Trend Savings Factors Adjustment	PDI Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adiustment	Capita Monthly Paid Cost
State State 1000 11700 1000 Delwork 0.00 1.000 1.000 1.000 Delwork 0.00 1.000 1.000 1.000 Delwork 0.00 1.000 1.000 1.000 of Drg Aluse 0.00 1.000 1.000 1.000 of Aluse 0.00 1.0000 1.0000 1.0000 <td></td> <td></td> <td></td> <td>NINIMEM INC.</td> <td>MININGPIPU</td> <td>Mannenfau</td> <td>1000</td>				NINIMEM INC.	MININGPIPU	Mannenfau	1000
Different Definition 1000<	1.0760		·			1.0000	\$6,381.07
Distlevery (with Checking) 0.00 1.000 1.	1.1323	-		•	•	1.0000	121.77
Menchenery 1000 11000	1.0000			-		1.0000	00.0
Cutation 1.000 1.000 1.000 1.000 1.000 India 0.00 1.000 1.000 1.000 1.000 1.000 India 0.00 1.000 1.000 1.000 1.000 1.000 1.000 V 2.012 1.000 1.000 1.000 1.000 1.000 1.000 1.000 V 2.012 1.000 1.000 1.000 1.000 1.000 1.000 1.000 V 2.01 1.000 <td>1.0000</td> <td></td> <td></td> <td></td> <td></td> <td>1.0000</td> <td>00.0</td>	1.0000					1.0000	00.0
mid Drug Athene 0.00 1.0000	0000	1 0060 1 1 0000	1 0000	1 0000	1 0000	1 0000	000
Outpatient 55.020 1.000	1.0000		•			1.0000	00.0
Outpatient 56 002.21 1000 10004 13200 V Roum 25.57 10000 10004 13200 V Soular 2.33 10000 10004 13200 V Soular 2.33 10000 10004 12200 V Soular 2.33 10000 10604 12200 V Soular 2.34 10000 10001 1000 V Soular 2.35 10000 10001 10000	1.0000					1.0000	
Outpatient SS-87 10000 1.9044 1.3200 Y 2.73 1.0000 1.9044 1.3200 Y 2.73 1.0000 1.9044 1.3200 Culat 2.73 1.0000 1.9044 1.3200 Culat 2.73 1.0000 1.9644 1.2200 Culat 0.00 1.0000 1.9644 1.2200 Culat 0.00 1.0000 1.9644 1.2200 Culat 0.00 1.0000 1.0000 1.9644 1.2200 Culat 0.00 1.0000 1.0000 1.9644 1.2200 Subscription 0.00 1.0000 1.9604 1.7200 1.7000 Subscription 0.00 1.0000 1.9604 1.7000 1.7000 Subscription 0.00 1.0000 1.9604 1.7000 1.7000 Subscription 0.00 1.0000 1.9604 1.7000 1.7000 Subscription 0.00 1.0000 1.9600 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>\$6,502.85</td>							\$6,502.85
cyl 25.37 1.000 1.064 1.2200 v 7.33 1.000 1.0644 1.2200 v 2.73 1.000 1.0644 1.2200 v 2.73 1.000 1.0644 1.2200 v 2.73 1.000 1.0644 1.2200 v 2.01 1.000 1.000 1.000 1.000 v 2.02 1.000 1.000 1.000 1.000 v 2.01 1.000 1.000 1.000 1.000 v 2.01 1.000 <							
Y 7.10 1.0000 1.0004 1.00004 1.0004	1.0604	1 1468 1 0000	1.0000	1.0000	1.0000	1.0000	\$32.09
3.53 (1) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2	1.0004					1,0000	24.0
5.6 5.00 1.0000 1.0004 1.0204 1.0200 0.00 0.000 1.0000 1.0004 1.0200 1.0200 0.00 0.000 1.0000 1.0000 1.0200 1.0200 0.00 0.000 1.0000 1.0000 1.0200 1.0200 0.00 1.0000 1.0000 1.0000 1.0200 1.0200 0.00 1.0000 1.0000 1.0000 1.0200 1.0200 0.00 1.0000 1.0000 1.0000 1.0200 1.0200 0.00 1.0000 1.0000 1.0000 1.0000 1.0200 0.00 1.0000 1.0000 1.0000 1.0000 1.0000 0.00 1.0000 1.0000 1.0000 1.0100 1.0100 0.00 1.0000 1.0000 1.0000 1.0100 1.0100 0.00 1.0000 1.0000 1.0000 1.0100 1.0100 0.00 1.0000 1.0000 1.0000 1.0	1.0004					0000 1	9.10
5-3 5-000 1,0100 1,0100	1.0004					1.0000	4.12
Set 0.00 2.6.90 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0200 1.0000 Set Set 0.00 0.00 1.0000 1.0000 1.0000 1.0200 Set Set 1.0000 1.0000 1.0000 1.0000 1.0200 Set Set 1.0000 1.0000 1.0000 1.0000 1.0200 Set Set 1.0000 1.0000 1.0000 1.0000 1.0000 Set Set 1.0000 1.0000 1.0000 1.0000 1.0000 Set Set 1.0000 1.0000 1.0000 1.0000 1.0000 Doub 1.0000 1.0000 1.0000 1.0100 1.0100 Set Set 1.0000 1.0000 1.0100 1.0100 Set Set 1.0000 1.0000 1.0100 1.0100 Set Set Set 1.0000 1.0100 1.0100 Set Set Set Set Set Set <td>1000</td> <td></td> <td></td> <td></td> <td></td> <td>1,0000</td> <td>0.0</td>	1000					1,0000	0.0
54 0.00 25.00 1.0000 1.000 1.0000 1.0001 1.0000 1.0001 1.0001 1.0001 1.01010 1.01010 1.01010 <td>00001</td> <td>·</td> <td></td> <td></td> <td></td> <td>1 0000</td> <td>0.00</td>	00001	·				1 0000	0.00
55 3000 1.0100 1.0100 1.0100 1.0100 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0000 1.0100 1.0100 1.0100 1.0100 1.0000 1.0000 1.0000 1.0000 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100		,			,	1 0000	0.00
25.00 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0100 </td <td>1.0000</td> <td></td> <td></td> <td></td> <td></td> <td>1.0000</td> <td>00.0</td>	1.0000					1.0000	00.0
55.00 56.00 1.0000 1.0001 1.0001 1.0100 1.0110 148.35 1.0000 1.0000 1.0000 1.0000 1.0100 1.0110 148.35 1.0000 1.0000 1.0000 1.0000 1.0100 1.0100 2.26 1.0000 1.0000 1.0000 1.0000 1.0100 1.0100 2.28 1.0000 1.0000 1.0000 1.0000 1.0100 1.0100 2.28 1.0000 1.0000 1.0000 1.0000 1.0100 1.0100 8.32 1.0000 1.0000 1.0000 1.0000 1.0100 1.0100 9.06 1.0000 1.0000 1.0000 1.0000 1.0100 1.0100 9.12.013 1.0000 1.0000 1.0000 1.0000 1.0100 1.0100 9.12.013 1.0000 1.0000 1.0000 1.0000 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100 1.0100	1.0604	·	·			1.0000	33.37
S0.00 1,0000 1,0000 1,0000 1,0000 1,0000 1,0100 </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>\$85.58</td>							\$85.58
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$							
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	1 0000		-	1.0000		1.0000	\$0.00
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	1.0607			1.0000	•	1.0000	49.66
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	1.0628		~	1.0000		1.0000	159.10
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	1 0000			1.0000		1.0000	00.00
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	1.0000			1.0000		1.0000	0.00
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	670001		- •	0000 1		1.0000	247
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	1 0551					1 0000	13.43
43.50 1,0000 1,0633 1,0100 8.32 1,0000 1,0000 1,0100 9.06 1,0000 1,0000 1,0100 9.06 1,0000 1,0000 1,0100 9.06 1,0000 1,0000 1,0100 9.06 1,0000 1,0000 1,0100 9.06 1,0000 1,0000 1,0100 9.06 1,0000 1,0000 1,0100 9.06 1,0000 1,0000 1,0100 9.01 1,0000 1,0000 1,0100 9.02 1,0000 1,0000 1,0000 9.1207.36 1,0000 1,0000 1,0000 9.1207.13 1,0000 1,0000 1,0000 9.1207.14 1,0000 1,0000 1,0000 9.1207.15 1,0000 1,0000 1,0000 9.1207.16 1,0000 1,0000 1,0000 9.1207.17 1,0000 1,0000 1,0000 9.1207.16 1,0000 1,0000	1.0629			1 0000		1.0000	90.66
3.32 1,0000 1,0500 1,0100 90.06 1,0000 1,670 1,610 90.06 1,0000 1,670 1,610 90.06 1,0000 1,670 1,010 54.30 1,0000 1,0000 1,010 54.30 1,0000 1,0000 1,010 54.30 1,0000 1,0000 1,010 55.25 1,0000 1,0000 1,0000 0.02 1,0000 1,0000 1,0000 1.44 1,0000 1,0000 1,0000 1.42 1,0000 1,0000 1,0000 0.12 1,0000 1,0000 1,0000 0.12 1,0000 1,0000 1,0000 0.12 1,0000 1,0000 1,0000 0.12 1,0000 1,0000 1,0000 0.12 1,0000 1,0000 1,0000 0.12 1,0000 1,0000 1,0000 0.00 1,0000 1,0000 1,0000	1.0630			1.0000		1.0000	43.59
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	1.0630		~	1.0000		1.0000	8.93
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	1.0000		-	1.0000		1.0000	00.00
0.00 1.0000 1.0000 1.0100 1.0000 1.0100 0.000 0.000 0.000 0.000 0.000 1.0100 0.000 0.000 0.000 0.000 0.000 1.0100 0.000 0.000 0.000 <t< td=""><td>1.1670</td><td></td><td></td><td>1.0000</td><td></td><td>1.0000</td><td>106.15</td></t<>	1.1670			1.0000		1.0000	106.15
Construction 1,0000 1,0000 1,0000 1,0000 1,0100 1	1.0000	•		1.0000		1.0000	00.0
Control 1,0000 1,0100	1.0000			1.0000		1.0000	0.00
72.0.10 1.0000 1.0100 0.000 0.000 0.000 0.000 0.000 1.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100 0.0100	0.9844			1.0000		1.0000	53.98 0.00
\$1,207.36 \$1,207.36 1,0000 1,0100 0,000 0,000 1,0100 0,000 0,000 1,0100 0,010	1.0624	1.0000 1.0000	1.0000	1.0000	1.0000	1.0000	772.79
S3.65 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,0000 1,01							\$1,300.71
35.66 1,0000 1,0100 </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>							
0.144 1,0000 1,0000 1,0000 1,0000 0.10 1,0000 1,0000 1,0000 1,0000 0.00 1,0000 1,0000 1,0000 1,0000 0.00 1,0000 1,0000 1,0000 1,0000 55.22 1,0000 1,0000 1,0000 1,0000 55.22 1,0500 1,0000 1,0100 1,0100 0.00 1,0500 1,0000 1,0100 1,0100 0.00 1,0500 1,0000 1,0100 1,0100 0.00 1,0500 1,0000 1,0100 1,0100 0.00 1,0500 1,0000 1,0100 1,0100 0.00 1,0500 1,0000 1,0100 1,0100 0.00 1,0500 1,0000 1,0100 1,0100 0.00 1,0500 1,0000 1,0100 1,0100 0.00 1,0500 1,0000 1,0100 1,0100 0.00 1,0500 1,0000 1,0100 1,0100 </td <td>1.0000</td> <td>1.0064 1.0000</td> <td>0.9894</td> <td>1.0000</td> <td>1.0000</td> <td>1.0000</td> <td>\$3.65</td>	1.0000	1.0064 1.0000	0.9894	1.0000	1.0000	1.0000	\$3.65
Center 0.01 1.0000 1.0100 0.000	1.0000			-	-	1.0000	1.39
Center 50.00 1.0000 1.0100 </td <td>1.0000</td> <td></td> <td></td> <td>- •</td> <td></td> <td>1.0000</td> <td>0.12</td>	1.0000			- •		1.0000	0.12
0.000 1.0000 1.0000 1.0000 Center 55.22 1.0000 1.0000 1.0000 S5.22 0.00 1.0560 1.0000 1.0100 Center 50.00 1.0560 1.0000 1.0110 S0.00 1.0560 1.0000 1.0100 1.0100 0.00 0.00 1.0560 1.0000 1.0100 0.00 0.000 1.0560 1.0000 1.0100 0.00 0.000 1.0560 1.0000 1.0100 0.00 0.000 1.0560 1.0000 1.0100 0.00 0.00 1.0560 1.0000 1.0100 0.00 0.00 1.0560 1.0000 1.0100 0.00 0.00 1.0560 1.0000 1.0100 0.00 0.00 1.0500 1.0000 1.0100 0.00 0.00 1.0500 1.0000 1.0100 0.00 0.00 1.0500 1.0000 1.0100						1.0000	00.0
\$5.22 \$5.22 \$5.22 \$5.22 \$5.00 \$1.0500 \$1.01000 \$1.0100 \$1.0100	1 0000			-		1.0000	0.00
Center \$0.00 1.0500 1.0100 </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>\$5.15</td>							\$5.15
\$5.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0100 1.0100 0.00 1.0500 1.0000 1.0100							
0.000 1.0300 1.0400 </td <td>1.0000</td> <td></td> <td></td> <td>1.0000</td> <td></td> <td>1.0000</td> <td>\$0.00</td>	1.0000			1.0000		1.0000	\$0.00
0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0000 1.0100 arvices 0.00 1.0500 1.0100 1.0100 services 0.00 1.0500 1.0100 1.0100	10000		·	1 0000		1.0000	0.00
0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0100 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0100 1.0100 0.00 1.0500 1.0100 1.0100 0.00 1.0500 1.0000 1.0100 atvices 0.00 1.0500 1.0100 50.00 1.0500 1.0000 1.0100	1 0000			1.0000		1.0000	0.00
0.00 1.0500 1.0100 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0100 1.0100 0.00 1.0500 1.0000 1.0100 0.00 1.0500 1.0100 1.0100 anvices 0.00 1.0500 1.0100 50.00 1.0500 1.0000 1.0100	1.0000			1.0000		1.0000	00.0
0.00 1.0500 1.0000 1.0100 1.0100 Services 0.00 1.0500 1.0100 1.0100	1.0000	·	·	1.0000		1.0000	00.00
0.00 1.0500 1.0100 <td>1.0000</td> <td></td> <td></td> <td>1.0000</td> <td></td> <td>1.0000</td> <td>00.00</td>	1.0000			1.0000		1.0000	00.00
nployment Services 0.00 1.0500 1.0000 1.01000 1.01000 1.0100 1.0100 1.0100 1.0100 1.0100 1.01	1.0000			1.0000		1.0000	0.00
	00001	1.0000 1.0000 1.0000	1.0000	1.0000	1.0000	1.0000	0.00
	0000-			0000		0000-1	\$0.00 \$0.00
All Services	-						\$7 894.29

Appendix B1 New Hampshire Department of Hallh and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Maternity Kick Payment

Demokise	Per Capita Monthly Paid	IBND Aditictment	Reimbursement Adirectment	Utilization Trend	Unit Cost Trend	Managed Care Savings Adiment	BDI Adirectment	Medicare OUD	Stop Loss Adiuctment	NEMT	Projected Per Capita Monthly Boid Coct
Hosnital Innatiant	0021		Hundreine	Lactors	Lactors	Aujusumeru		Aujusuneru	Aujustifierit	Aujustifierit	Laid COSt
Medical	\$23.64	1.0000	1.0898	1.0000	·	1.0000	·	1.0000	·	1.0000	\$25.92
Surgical	1.72	1.0000	1.0759	1.0000	-	1.0000		1.0000		1.0000	1.86
Matemity Delivery	1,931.33	1.0000	1.0578	1.0000	-	1.0000		1.0000		1.0000	2,055.23
Maternity Non-Delivery	19.74	1.0000	1.0611	1.0000	- ·	1.0000		1.0000		1.0000	21.08
Well Newborn Devokiatric	0.00	1.0000	1.0000	0000 1	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Alcohol and Drug Abuse	00.0	1 0000	1 0000	1 0000		1 0000				1 0000	0.00
Other	0.00	1 0000	1 0000	1.0000		1.0000		1.0000		1.0000	0.00
	\$1,976.43										\$2,104.08
Hospital Outpatient											
Emergency Room	\$12.67	1.0000	1.0604	1.0200	1.1468	1.0000	1.0000	1.0000	1.0000	1.0000	\$15.72
Surgery	0.80	1.0000	1.0604	1.0200		1.0000	·	1.0000	1.0000	1.0000	0.99
Radiology	2.27	1.0000	1.0604	1.0200		1.0000		1.0000	1.0000	1.0000	2.82
Pathology	1.69	1.0000	1.0604	1.0200		1.0000		1.0000	1.0000	1.0000	2.09
Pharmacy	2.04	1.0000	1.0604	1.0200		1.0000		1.0000	1.0000	1.0000	2.53
Cardiovascular DT/DT/PT	0.00	0000	1,0000	1.0200		1.0000		0000	1.0000	1.0000	0.00
Priviteiro	00.0		0000	00201		1.0000	·	0000	1.0000	1.0000	0.00
Esycillauro Substanco Abirco	00.0	1.0000	1.0000	1.0200		1,0000		1.0000	1,0000	1,0000	00.0
Other	13.52	1 0000	1 0604	1 0200		1 0000	·	1 0000	1 0000	1 0000	16.76
	\$32.99		-	0000					~~~~		\$40.92
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$0.00	1.0000	1.0000	-	1.0000	1.0000	-	1.0000	-	1.0000	\$0.00
Office Visits	1.47	1.0000	1.0622	-	1.0000	1.0000		1.0000	-	1.0000	1.58
Preventive Medicine	1.08	1.0000	1.0606		-	1.0000	t	1.0000	-	1.0000	1.16
Maternity	518.97	1.0000	1.0629	1.0100	-	1.0000		1.0000	~	1.0000	557.14
Certified Midwife	6.01	1.0000	1.0615		- ·	1.0000	(-)	1.0000	,	1.0000	6.44
PT/OT/ST	0.00	1.0000	1.0000		- •	1.0000		1.0000		1.0000	0.00
Psychiatric and Substance Abuse	0.00	1.0000	1.0000			1.0000		1.0000		1.0000	0.00
Radiology and Paurology Home Health and Driveto Duty Nursing	0.44	1.0000	1.0021			1.0000	_ (1.0000		1.0000	24.0
Ambulance	3.95	1 0000	1 0630			1 0000		1 0000		1 0000	4.24
Non-Emergency Transportation	00.0	1.0000	1.0000		-	1.0000	~	1.0000		1.0000	0.00
Opioid Treatment Program	0.00	1.0000	1.0000		-	1.0000		1.0000	-	1.0000	00.00
Federally Qualified and Rural Health Clinics	3.19	1.0000	1.1666		-	1.0000	-	1.0000	-	1.0000	3.76
Adult Medical Day Care	0.00	1.0000	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Personal Care	0.00	1.0000	1.0000		- •	1.0000	- •	1.0000		1.0000	00.0
Durable Medical Equipment Annlied Rehavioral Analysis	00.0	1.0000	1.0000			1 0000		1.0000		1.0000	0.00
Other	13.33	1.0000	1.0617			1.0000		1.0000		1.0000	14.30
	\$550.85										\$591.67
Prescription Drugs											
Generic Scripts	00.04	1.0000	1.0000	1.0000		1.0000		0000	1.0000	1.0000	00.04
Multi-Source Brand	000	1 0000	1 0000	1 0000	,	1 0000		1 0000	1 0000	1 0000	0.00
Specialty	00.00	1.0000	1.0000	1.0000	·	1.0000	-	1.0000	1.0000	1.0000	00.00
Hepatitis C	0.00	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	00.00
Other	00.00	1.0000	1.0000	1.0000		1.0000		1.0000	1.0000	1.0000	00.0
	\$0.00										\$0.00
Community Mental Health Center Case Management	\$0.00	1 0500	1 000	1 0100	,	1 0000	·	1 0000	1 0000	1 0000	\$0.00
Long Term Support Service	00.00	1.0500	1.0000	1.0100		1.0000		1.0000	1.0000	1.0000	00.00
Partial Hospital	0.00	1.0500	1.0000	1.0100	v -	1.0000	·	1.0000	1.0000	1.0000	00.0
Psychotherapy	00.00	1.0500	1.0000	1.0100		1.0000		1.0000	1.0000	1.0000	00.00
Evidence Based Practice	0.00	1.0500	1.0000	1.0100		1.0000		1.0000	1.0000	1.0000	0.00
Medication Management	0.00	1.0500	1.0000	1.0100		1.0000		1.0000	1.0000	1.0000	0.00
Enreigency Service 24/7 APRTP	00.0	1 0500	1,0000	1 0100			1 0000	1 0000	1 0000	1 0000	00.0
Supported Employment Services	00.00	1.0500	1.0000	1.0100		1.0000		1.0000	1.0000	1.0000	00.00
Other	00.00	1.0500	1.0000	1.0100	Ì	1.0000		1.0000	1.0000	1.0000	00.0
	\$0.00										\$0.00
All Services	\$2,560.27										\$2,736.67

Appendix B1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Eding April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Severe/Persistent Mental Illness - Non-Dual

Renefits	Per Capita Monthly Paid Cost	IRNR Adjustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiustment	PDI Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adjustment	Projected Per Capita Monthly Paid Cost
Locaital Innationt	2021		Unjustine	1 444413	1 401013			Unimente			Laid COSI
Modical	CAA FO		1 0852	1 0000	Ŧ	0 0875	Ţ	1 0000	1 0000	1 0000	\$47.07
Surrical	39.85	1 0000	1 0961	1 0000		0.9875		1 0000	1 0000	1 0000	43.40
Matemity Delivery	0.00	1.0000	1.0000	1.0000		0.9875		1.0000	1.0000	1.0000	0.00
Maternity Non-Delivery	0.44	1.0000	1.1072	1.0000		0.9875		1.0000	1.0000	1.0000	0.48
Well Newborn	0.00	1.0000	1.0000	1.0000	-	0.9875	.	1.0000	1.0000	1.0000	00.00
Psychiatric	52.54	1.0000	1.1964	1.0000	-	0.9875	-	1.0000	1.0000	1.0000	62.45
Alcohol and Drug Abuse	1.92	1.0000	1.1164	1.0000	1.0060	0.9875	1.0000	1.0000	1.0000	1.0000	2.13
Oliei	\$139.25	0000-I	1.000	1.000		C/08'0	1-000	0000.1	1.000	1-0000	\$156.43
Hosnital Outnatient	A-00-4										01-001-0
Fmergency Room	\$83.55	1.0000	1.0604	1.0200		0.9875	1.0000	1.0000	1.0000	1.0000	\$102.34
Surgerv	18.21	1 0000	1 0604	1.0200	•	0.9875	1.0000	1.0000	1 0000	1.0000	22.31
Radiology	22.46	1.0000	1.0604	1.0200		0.9875	·	1.0000	1.0000	1.0000	27.51
Pathology	8.59	1.0000	1.0604	1.0200		0.9875		1.0000	1 0000	1.0000	10.52
Pharmacv	11.50	1 0000	1 0604	1 0200		0.9875		1 0000	1 0000	1 0000	14 09
Cardiovascular	2.28	1.0000	1.0604	1.0200		0.9875		1.0000	1.000	1.0000	2.79
PT/OT/ST	5.53	1.0000	1.0604	1.0200		0.9875		1.0000	1 0000	1.0000	6.78
Psychiatric	116	1 0000	1 0604	1 0200		0.9875		1 0000	1 0000	1 0000	142
Substance Abuse	0.38	1.0000	1.0604	1.0200		0.9875		1.0000	1.0000	1.0000	0.46
Other	19.18	1.0000	1.0604	1.0200	1.1468	0.9875	1.0000	1.0000	1.0000	1.0000	23.50
	\$172.82										\$211.69
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$2.26	1.0000	1.1204	1.0100	-	0.9875	-	1.0000	1.0000	-	\$2.53
Office Visits	32.10	1.0000	1.0641	1.0100	~	0.9875	-	1.0000	1.0000	-	34.07
Preventive Medicine	3.32	1.0000	1.1389	1.0100	-	0.9875	-	1.0000	1.0000	-	3.77
Maternity	1.05	1.0000	1.0630	1.0100		0.9875		1.0000	1.0000		1.12
Certified Midwite	0.01	1.0000	1.0629	1.0100		0.98/5		1.0000	1.0000	- •	0.01
PT/01/ST Denshiptris and Substance Abuse	34.45	0000	1.0030	0010.1		C/26.0	- •	0000 1	1.0000	- •	3.38
Psychiatric and Substance Abuse	CI.47	1.0000	1.3294	1.0100		2/26/0	- •	0000	1.0000	- •	32.02
Kaulology and Paulology Homo Hoalth and Privato Duty Nursing	14.19	1.0000	1 0620 1	1.0100		2/08.0	- •	1.0000	1.0000		14.90
Ambulance	10.41	1 0000	1 0630	1 0100		0.9875		1 0000	1 0000		11 04
Non-Emergency Transportation	40.23	1.0000	1.0630	1.0100		0.9875	1.0000	1.0000	1.0000		42.65
Opioid Treatment Program	8.73	1.0000	1.0636	1.0100	-	0.9875	1.0000	1.0000	1.0000	-	9.26
Federally Qualified and Rural Health Clinics	21.10	1.0000	1.1572	1.0100	-	0.9875	1.0000	1.0000	1.0000	-	24.36
Adult Medical Day Care	0.00	1.0000	1.0000	1.0100		0.9875		1.0000	1.0000		00.0
reisoitai Gale Durabla Madical Equipment	2C-2 18 06	1.0000	0.0265	1.0100		0.90/3	- •	1.0000	1.0000		17 52
Annlied Behavioral Analysis	000	1 0000	1 000	1 0000		0.9875		1 0000	1 0000		000
Other	51.80	1.0000	1.0387	1.0100	1.0000	0.9875	1 0000	1.0000	1 0000	1.0000	53.66
	\$246.24										\$265.97
Prescription Drugs											
Generic Scripts	\$101.23	1.0000	1.0000	1.0000		0.9875	1.1429	1.0000	1.0000	1.0000	\$102.41
Single-Source Brand Multi-Source Brand	71.97	1.0000	1.0000	1.0000		C/86.0 7280 0		1.0000	1.0000	1.0000	193.34
Mulu-Source Brand Snecialty	172.84	1 0000	1 0000	1 0000		0.9075		1 0000	1 0000	1 0000	236.18
Hepatitis C	30.14	1.0000	1.0000	1.0000	0.2293	0.9875	0.9443	1.0000	1.0000	1.0000	6.44
Other	0.07	1.0000	1.0000	1.0000		0.9875		1.0000	1.0000	1.0000	0.06
	\$576.41										\$562.00
Community Mental Health Center Case Management	\$278 F.A	1 0500	1 1530	1 0100		10000	1 0000	1 0000	1 0000	1 0000	63AD 6D
Long Term Support Service	268.21	1.0500	1.2672	1.0100		1.0000	1.0000	1.0000	1.0000	1.0000	360.44
Partial Hospital	12.36	1.0500	1.1459	1.0100		1.0000	1.0000	1.0000	1.0000	1.0000	15.01
Psychotherapy	64.47	1.0500	1.0676	1.0100		1.0000	1.0000	1.0000	1.0000	1.0000	73.00
Evidence Based Practice	3.36	1.0500	1.0626	1.0100	-	1.0000	1.0000	1.0000	1.0000	1.0000	3.79
Medication Management	4.11	1.0500	1.0631	1.0100		1.0000	1.0000	1.0000	1.0000	1.0000	4.64
Emergency Service 24/7	4.11	1.0500	1.0631	1.0100		1.0000	1.0000	1.0000	1.0000	1.0000	4.63
Sunnorted Employment Services	20.31	1 0500	1 1529	1 0100	1 0000	1 0000	1 0000	1 0000	1 0000	1 0000	24.83
Other	31.56	1 0500	1.0628	1.0100	- (-	1.0000	1 0000	1.0000	1.0000	1.0000	35.57
	\$691.98										\$868.56
All Services	\$1,826.71										\$2,064.64

Appendix B1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development. PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Severe/Persistent Mental Illness - Dual

Benefits	Per Capita Monthly Paid Cost	IBNR Adjustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adjustment	PDL Adjustment	Medicare OUD Adjustment	Stop Loss Adiustment	NEMT Adjustment	Projected Per Capita Monthly Paid Cost
Hospital Inpatient			•								
Medical	\$20.54	-	1.0852	1.0000	-	1.0000	-		.	-	\$22.43
Surgical	4.07		1.0961	1.0000		1.0000	- •				4.48
Maternity Delivery Maternity Non-Delivery	000		1 0000	1 0000		1 0000					0000
Well Newborn	0.00		1.0000	1.0000		1.0000		·			00.0
Psychiatric	18.25	-	1.1964	1.0000	-	1.0000	-	·		~	21.96
Alcohol and Drug Abuse	1.25	1.0000	1.1164	1.0000	1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	1.41
Ollei	0.00 \$44.11		1-000	1.0000		10000					\$50.28
Hospital Outpatient											
Emergency Room	\$30.74	1.0000	1.0604	1.0200	-	1.0000	1.0000	1.0000	1.0000	-	\$38.13
Surgery	9.63		1.0604			1.0000	1.0000		1.0000	-	11.95
Radiology	6.71		1.0604			1.0000	1.0000	-	1.0000		8.32
Pathology	1.63		1.0604	- •	- •	1.0000	1.0000		1.0000		2.03
r nannacy Cardiovascular	0.83		1 0604			1 0000	1 0000		1 0000		1.03
PT/OT/ST	2.71		1 0604			1.0000	1.0000		1.0000		3.37
Psychiatric	1.02		1 0604	·	~	1.0000	1.0000		1.0000	-	1.26
Substance Abuse	0.15	-	1.0604		1.1468	1.0000	1.0000		1.0000	1.0000	0.18
Other	17.79		1.0604			1.0000	1.0000		1.0000		22.07
	\$78.83										\$97.77
Professional and Other State Plan Services		τ	1001	0100		0000			•		10.13
	91.14		1.1204	1.0100	- •	1,0000					40-04 15-70
Preventive Medicine	26.0		1.1389	1.0100		1.0000					1.12
Maternity	0.13		1.0630	1.0100	-	1.0000					0.14
Certified Midwife	0.00		1.0000	1.0100	-	1.0000	-		-	~	00.00
PT/OT/ST	0.96		1.0630	1.0100		1.0000				, - ,	1.03
Psychiatric and Substance Abuse	34.92		1.3294	1.0100		1.0000			- •	- •	40.88
Home Health and Private Duty Nursing	2.38		1.0629	1 0100		1.0000					2.56
Ambulance	7.35		1.0630	1.0100	~	1.0000					7.89
Non-Emergency Transportation	43.39	-	1.0630	1.0100	-	1.0000	-		-	-	46.58
Opioid Treatment Program	5.96		1.0636	1.0100		1.0000					1.48
Federally Qualified and Rural Healur Cunics Adult Medical Day Care	1.10	_	1.0630	1 0100		1,0000					2.06
Personal Care	9.14		1.0624	1.0100	~ ~	1.0000					9.81
Durable Medical Equipment	8.82	-	2.3497	1.0100	-	1.0000	-		~	¢-	20.92
Applied Behavioral Analysis	0.00	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Other	\$164.49		1.038/	1.010	-	1.000					\$192.06
Prescription Drugs	21-1-0-1-0										00-7010
Generic Scripts	\$1.15	-	1.0000	1.0000	0	1.0000		·-	1.0000	1.0000	\$1.13
Single-Source Brand	0.02		1.0000	1.0000		1.0000			1.0000	-	0.03
Multi-Source Brand	0.02		1.0000	1.0000		1.0000			1.0000		0.02
Speciality Henatitis C	00.0	1 0000	1.0000	1.0000	1 0000	1,0000	0.9894	1.0000	1,0000	1.0000	00.0
Other	00.00		1 0000	1 0000	0	1.0000			1 0000		00.00
	\$1.20										\$1.18
Community Mental Health Center	00 0004		1 100	00100	0000	0000	0000 1		0000		00 0000
Case Management Long Term Support Service	\$299.88 418.41		1.1530	1.0100		1.0000	1.0000		1.0000	1.0000	\$300.09 562.28
Partial Hospital	24.37		1.1459	1.0100	· -	1.0000	1.0000		1.0000	- -	29.61
Psychotherapy	34.23	-	1.0676	1.0100	-	1.0000	1.0000	-	1.0000	-	38.75
Evidence Based Practice	5.59		1.0626	1.0100		1.0000	1.0000		1.0000		6.30
Medication Management Emergency Service 24/7	0.30		1.0031	1.0100		1.0000	1.0000		00001		01.10 4.91
APRTP	8.23		1.1530	1.0100		1.0000	1.0000		1.0000		10.07
Supported Employment Services	37.78	1.0500	1.1529	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	46.20
Other	16.28		1.0628	1.0100		1.0000	1.0000		1.0000		18.35
All Services	04-0000 11 11 12										\$1,030.33
		-									·

Appendix B1 New Hampshire Department of Heath and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Severe Mental Illness - Non-Dual

Renetits	Monthly Paid Cost	IBNR Adjustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiustment	PDI Adiustment	Medicare OUD Adjustment	Stop Loss Adiustment	NEMT Adjustment	Projected Per Capita Monthly Paid Cost
Locaited Innetient	0021		Aujustriterit	Lactors	Lactors	Aujustriterit		Aujustrient	Hujusufielle	Aujustinerit	
nospital inpatient Medical	\$53 E7	1 0000	1 0856			0 9875	1 0000	1 0000	1 0000	1 0000	\$57.71
Surgical	25.88	1.0000	1.0895			0.9875		1.0000	1.0000	1.0000	28.00
Matemity Delivery	00.00	1.0000	1.0000	-	-	0.9875	-	1.0000	1.0000	1.0000	0.00
Maternity Non-Delivery	1.84	1.0000	1.0475	-		0.9875	,	1.0000	1.0000	1.0000	1.92
Well Newborn	0.00	1.0000	1.0000			0.9875		1.0000	1.0000	1.0000	0.00
Psychiatric	32.81	0000	1.2020			0.9870	- 1	1.0000	1.0000	0000	39.17
Alcoriol and Urug Abuse Other	00.0	1,0000	1.1223	1 0000	1.0060	0.9875	1.0000	1.0000	1,0000	1.0000	0.00
Long Long Long Long Long Long Long Long	\$119.75					2000					\$133.18
Hospital Outpatient											
Emergency Room	\$92.26	1.0000	1.0604	1.0200	1.1468	0.9875	1.0000	1.0000	1.0000	1.0000	\$113.01
Surgery	18.86	1.0000	1.0604	-		0.9875	-	1.0000	1.0000	1.0000	23.11
Radiology	21.07	1.0000	1.0604	-		0.9875	-	1.0000	1.0000	1.0000	25.80
Pathology	10.14	1.0000	1.0604	-		0.9875	-	1.0000	1.0000	1.0000	12.42
Pharmacy	12.05	1.0000	1.0604	-		0.9875	-	1.0000	1.0000	1.0000	14.76
Cardiovascular	3.35	1.0000	1.0604	-		0.9875	-	1.0000	1.0000	1.0000	4.10
PT/0T/ST	6.88	1.0000	1.0604	-		0.9875	~	1.0000	1.0000	1.0000	8.42
Psychiatric	1.77	1.0000	1.0604	-		0.9875	1.0000	1.0000	1.0000	1.0000	2.16
Substance Abuse	0.93	1.0000	1.0604	,	1.1468	0.9875	1.0000	1.0000	1.0000	1.0000	1.14
Other	23.38	1.0000	1.0604			0.9875	1.0000	1.0000	1.0000	1.0000	28.64
	\$190.09										\$233.51
Professional and Other State Plan Services	i c	0000 1	1 004	0010		0.007	0000 1	10000	0000 1		01 F
Ambulatory Surgery Center	10.14	0000	1.2011			0.98/0	00001	0000	1.0000	0000	00.14
Office Visits Drownthin Modicing	2075 0175	1.0000	CU/U-1 8041 1	1.0100		2700.0	1.0000	1.0000	1.0000	1.0000	00-4-00 7-10
FTEVEITUVE INEGIOLITE Matamity	1004	1.0000	1.1400			0.9075		1.0000	1 0000	1.0000	3.80
Matering Cartifiad Midwifa	100	1 0000	1 0620			0.0875			1 0000	00001	0.01
PT/OT/ST	1.66	1.0000	1.0630	·		0.9875		1.0000	1.0000	1.0000	1.76
Psychiatric and Substance Abuse	44.71	1.0000	1 2214	·		0.9875		1 0000	1 0000	1.0000	54.47
Radiology and Pathology	19.79	1.0000	1.0584	1.0100	-	0.9875	1.0000	1.0000	1.0000	1.0000	20.90
Home Health and Private Duty Nursing	12.58	1.0000	1.0628	-	-	0.9875	1.0000	1.0000	1.0000	1.0000	13.33
Ambulance	11.32	1.0000	1.0630	-	-	0.9875	1.0000	1.0000	1.0000	1.0000	12.00
Non-Emergency Transportation	62.02	1.0000	1.0630	-	-	0.9875	1.0000	1.0000	1.0000	1.0000	65.75
Opioid Treatment Program	16.15	1.0000	1.0636			0.9875	1.0000	1.0000	1.0000	1.0000	17.13
Federally Qualified and Rural Health Clinics Adult Modical Dou Corre	23.10	0000	1.1002	- •	- ,	3700.0	1.0000	0000	1.0000	0000	71.02
Auut interical Day Gare Personal Care	110	1 0000	1 0624	1.0100	1 0000	0.9075	1 0000	1 0000	1 0000	1 0000	eC.1
Durable Medical Equipment	12.85	1 0000	1 1852			0.9875		1 0000	1 0000	1.0000	15.19
Applied Behavioral Analysis	0.00	1.0000	1.0000	-	-	0.9875	-	1.0000	1.0000	1.0000	0.00
Other	59.34	1.0000	1.0536		1	0.9875	1	1.0000	1.0000	1.0000	62.36
	\$307.93										\$337.54
Prescription Drugs											
Generic Scripts	\$61.05	1.0000	1.0000			0.9875		1.0000	1.0000	1.0000	\$61.76
Single-Source Brand Multi-Source Brand	10.011	1.0000	0000.1			6/96.0 5280 0		1.0000	1.0000	1.0000	100.80
Specialty	98.40	1.0000	1.0000			0.9875		1 0000	1.0000	1.0000	134.46
Hepatitis C	73.87	1.0000	1.0000	1.0000	0.2293	0.9875	0.9443	1.0000	1.0000	1.0000	15.79
Other	0.00	1.0000	1.0000			0.9875		1.0000	1.0000	1.0000	0.00
	\$380.51										\$327.18
Community mental realth Center Case Management	\$191.95	1 0500	1 1530			1 0000	1 0000	1 0000	1 0000	1 0000	\$234 72
Long Term Support Service	62.95	1.0500	1.1592	,		1.0000	1.0000	1.0000	1.0000	1.0000	77.39
Partial Hospital	0.14	1.0500	1.1512	·	v -	1.0000	1.0000	1.0000	1.0000	1.0000	0.17
Psychotherapy	69.78	1.0500	1.0681	•	-	1.0000	1.0000	1.0000	1.0000	1.0000	79.04
Evidence Based Practice	3.08	1.0500	1.0627			1.0000	1.0000	1.0000	1.0000	1.0000	3.47
Medication Management	1.73	1.0500	1.0631			1.0000	1.0000	1.0000	1.0000	1.0000	1.95
Emergency Service 24/7	69.0	00501	1.0031		- •	1.0000	1.0000	0000	1.0000	1.0000	0./8
Supported Employment Services	12.16	1.0500	1.1526	1.0100	1.0000	1.0000	1.0000	1 0000	1.0000	1.0000	14.86
Other	27.48	1.0500	1.0629			1.0000	1.0000	1.0000	1.0000	1.0000	30.98
	\$379.87										\$455.47
All Services	\$1,378.76										\$1,486.94

Appendix B1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Severe Mental Illness - Dual

Desertite	Per Capita Monthly Paid		Reimbursement	Utilization Trend	Unit	Managed Care Savings	DDI Adiretteret	Medicare OUD	Stop Loss	NEMT	Projected Per Capita Monthly
Denerits Hospital Innatient	COSI	IDIN VAJASTIN	Aujusurien	Lactors	Lactors	Aujusument		Hundrineric	Aujusument	Aujustinent	raid COSI
Medical	\$17.25	1.0000	1.0856	<i>t</i> -		1.0000		1.0000		1.0000	\$18.84
Surgical	7.39	1.0000	1.0895	<i>•</i>	-	1.0000		1.0000		1.0000	8.10
Matemity Delivery	0.00	1.0000	1.0000	,	.	1.0000		1.0000		1.0000	0.00
Maternity Non-Delivery	0.35	1.0000	1.0475	- •	- •	1,0000		1.0000		1.0000	0.37
Psychiatric	6.15	1.0000	1.2020			1,0000	·	1.0000	•	1.0000	7.43
Alcohol and Drug Abuse	6.05	1.0000	1.1223	1.0000	1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	6.83
Other	0.00	1.0000	1.0000			1.0000		1.0000		1.0000	0.00
Hannited Ordenations	\$31.20										00.144
Hospital Outpatient Emericancy Room	533 3D	1 0000	1 0604				·	1 0000		1 0000	\$41 31
Surgery	9.83	1.0000	1.0604				·	1.0000		·	12.19
Radiology	9.68	1.0000	1.0604					1.0000			12.01
Pathology	2.60	1.0000	1.0604	-			·	1.0000		·	3.22
Pharmacy	9.17	1.0000	1.0604					1.0000			11.37
Cardiovascular	0.75	1.0000	1.0604	-				1.0000			0.93
PT/OT/ST	2.74	1.0000	1.0604					1.0000			3.40
Psychiatric Substance Abirco	3.43	1.0000	1,0004	- •				1.0000			4.26
Other	16.01	1 0000	1 0604	1 0200	1 1468	1 0000	1 0000	1 0000	1 0000	1 0000	19.86
	\$87.51										\$108.55
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$1.17	1.0000	1.2011		-	1.0000	-	1.0000	-	1.0000	\$1.42
Office Visits	16.82	1.0000	1.0705	1.0100		1.0000		1.0000		1.0000	18.19
Preventive Medicine Motomity	3.07	0000	1.1408			10000	- •	1.0000		1.0000	4.23
Certified Midwife	00.0	1 0000	1 0000			1 0000		1 0000		1 0000	00.0
PT/OT/ST	1.00	1.0000	1.0630			1.0000	, v	1.0000		1.0000	1.07
Psychiatric and Substance Abuse	21.45	1.0000	1.2214		-	1.0000	•	1.0000	~	1.0000	26.46
Radiology and Pathology	6.48	1.0000	1.0584		~	1.0000	~	1.0000	,	1.0000	6.92
Home Health and Private Duty Nursing	1.86	1.0000	1.0628			1.0000		1.0000		1.0000	1.99
Non-Emergency Transportation	62.79	1.0000	1.0630	1.0100	1.0000	1.0000		1.0000	1.0000	1.0000	67.41
Opioid Treatment Program	16.19	1.0000	1.0636	-	-	1.0000		0.2369	-	1.0000	4.12
Federally Qualified and Rural Health Clinics	2.65	1.0000	1.1602	-	-	1.0000		1.0000	-	1.0000	3.11
Adult Medical Day Care	0.52	1.0000	1.0630		÷ ,	1.0000	(- ·	1.0000		1.0000	0.56
Personal Care	28.13	1.0000	1.0624			1.0000		1.0000		1.0000	30.19
Durable Medical Equipriterit Amhliad Rahaviaral Analysis	00.0	1.0000	1 0000					1.0000		1.0000	0.00
Other	28.34	1.0000	1.0536			1.0000	1.0000	1.0000		1.0000	30.15
	\$209.81										\$229.72
Prescription Drugs	¢0.00		1 0000								¢0 b0
Single-Source Brand	0.28	1.0000	1.0000					1.0000	•	1.0000	0.32
Multi-Source Brand	0.00	1.0000	1.0000					1.0000		1.0000	00.0
Specialty	0.00	1.0000	1.0000	1.0000	1.0000	1.0000	0.9894	1.0000	1.0000	1.0000	0.00
Hepatitis C	0.00	1.0000	1.0000					1.0000		1.0000	0.00
Curei	\$1.19	0000-	0000-1					0000-		0000-	\$1.22
Community Mental Health Center											
Case Management	\$206.46	1.0500	1.1530			1.0000		1.0000	-	1.0000	\$252.46
Long Term Support Service	84.06	1.0500	1.1592			1.0000	• •	1.0000	(- (1.0000	103.33
Perchatharany	13.55	1.0500	1.151.1			10000		1.0000		1.0000	07.0
Evidence Based Practice	5.48	1.0500	1.0627			1 0000	·	1.0000	•	1.0000	6.18
Medication Management	1.17	1.0500	1.0631			1.0000		1.0000		1.0000	1.32
Emergency Service 24/7	0.65	1.0500	1.0631		-	1.0000		1.0000	•	1.0000	0.74
APRTP 2 · · · · · · · · · · · · · · · · · · ·	10.88	1.0500	1.1530			1.0000	•	1.0000		1.0000	13.31
Supported Employment Services Other	20.25	0060.1	1.1526	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	c7.42 18.20
	\$389.26									2000	\$470.36
All Services	\$724.97										\$851.43

Appendix B1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Low Utilizer - Non-Dual

	Per Capita Monthly Paid		Reimbursement	Utilization Trend	Unit	Managed Care Savings		Medicare OUD	Stop Loss	NEMT	Projected Per Capita Monthly
Benefits	Cost	IBNK Adjustment	Adjustment	Factors	Factors	Adjustment	PDL Adjustment	Adjustment	Adjustment	Adjustment	Paid Cost
Hospital Inpatient	C10F3		1 0602			0.0076		1 0000	ţ		¢10.60
Surrica	02.0	1 0000	1 0837			0.9875		1 0000		1 0000	419-03-03-03-03-03-03-03-03-03-03-03-03-03-
Matemity Delivery	0.00	1.0000	1.0000	1.0000		0.9875	·	1.0000		1.0000	00.00
Maternity Non-Delivery	0.00	1.0000	1.0000	1.0000		0.9875		1.0000		1.0000	00.00
Well Newborn	00.00	1.0000	1.0000	1.0000		0.9875		1.0000	、	1.0000	00.00
Psychiatric	6.02	1.0000	1.5299	1.0000		0.9875		1.0000	.	1.0000	9.15
Alcohol and Drug Abuse	00.0	1.0000	1.0000	1.0000	1.0060	0.9875	1.0000	1.0000	1.0000	1.0000	0.00
Other	\$34.25	0000-1	0000.1	0000-1		C 106'0		0000-1		0000-1	\$39.27
Hospital Outpatient											
Emergency Room	\$54.59	1.0000	1.0604	1.0200	v	0.9875	t	1.0000	1.0000	1.0000	\$66.86
Surgery	22.25	1.0000	1.0604	1.0200	1.1468	0.9875	1.0000	1.0000	1.0000	1.0000	27.25
Radiology	25.11	1.0000	1.0604	1.0200		0.9875	.	1.0000	1.0000	1.0000	30.76
Pathology	8.88	1.0000	1.0604	1.0200		0.9875	-	1.0000	1.0000	1.0000	10.87
Pharmacy	11.04	1.0000	1.0604	1.0200		0.9875	-	1.0000	1.0000	1.0000	13.53
Cardiovascular	2.12	1.0000	1.0604	1.0200	·	0.9875	-	1.0000	1.0000	1.0000	2.60
PT/OT/ST	6.93	1.0000	1.0604	1.0200	·	0.9875	-	1.0000	1.0000	1.0000	8.49
Psychiatric	2.15	1.0000	1.0604	1.0200		0.9875	-	1.0000	1.0000	1.0000	2.63
Substance Abuse	1.28	1.0000	1.0604	1.0200		0.9875	-	1.0000	1.0000	1.0000	1.57
Other	18.13	1.0000	1.0604	1.0200		0.9875	,	1.0000	1.0000	1.0000	22.20
	\$152.48										\$186.77
Professional and Other State Plan Services						1000					
Ambulatory Surgery Center	54 D0	1.0000	1.2932	1.0100	1.0000	C/86.0	- 1	1.0000		1.0000	20.02
Office Visits	28.18	1.0000	1.0729	1.0100	1.0000	0.98/5		1.0000		1.0000	30.15
Preventive Medicine	3./0	1.0000	91.0.1	1.0100	0000 1	C/98.0	- 1	00001		1.0000	0.40
Niaternity	01.0	1.0000	1.0032	1.0100	0000	0.9875	0000	0000		1.0000	01.0
Certified MidWite	0.00	1.0000	1.000	0010.1	0000	6/96/0 3200 0	•	0000		1.0000	00.00
PT/UT/ST Devekietie and Substance Abuce	2.04	1.0000	1.0000	0010.1	1.0000	C/26.0	- •	00001	_ 、	0000 1	2.17 FR 25
Feyciliaure and Gethology Dediology and Dethology	10.00	1.0000	1 0573	1.0100	1,0000	0.0875	- •	1.0000		1,0000	12.80
Home Health and Private Duty Nursing	8 50	1 0000	1 0628	1 0100	1 0000	0.9075	- •	1 0000		1 0000	0010
Ambulance	5.47	1 0000	1 0630	1 0100	1 0000	0.9875		1 0000		1 0000	5.80
Non-Emergency Transportation	30.23	1.0000	1.0630	1.0100	1.0000	0.9875		1.0000		1.0000	32.05
Opioid Treatment Program	11.40	1.0000	1.0636	1.0100	1.0000	0.9875	-	1.0000	-	1.0000	12.09
Federally Qualified and Rural Health Clinics	20.01	1.0000	1.1617	1.0100	1.0000	0.9875	-	1.0000	-	1.0000	23.19
Adult Medical Day Care	00.00	1.0000	1.0000	1.0100	1.0000	0.9875	t-	1.0000	~	1.0000	00.00
Personal Care	6.26	1.0000	1.0624	1.0100	1.0000	0.9875	~	1.0000		1.0000	6.63
Durable Medical Equipment	9.40	1.0000	0.9517	1.0100	1.0000	0.9875	-	1.0000	~	1.0000	8.92
Applied Behavioral Analysis	0.00	1.0000	1.0000	1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	0.00
Ollel	\$227.03	0000-1	1.07.00	1-0100	1-0000	0.906.0		00001		1-0000	\$256.14
Prescription Drugs	00-1770										1-0010
Generic Scripts	\$111.53	1_0000	1.0000	1.0000	0.8964	0.9875		1.0000	1.0000	1.0000	\$112.83
Single-Source Brand	200.20	1.0000	1.0000	1.0000	1.0065	0.9875		1.0000	1.0000	1.0000	182.60
Multi-Source Brand	20.63	1.0000	1.0000	1.0000	1.0065	0.9875		1.0000	1.0000	1.0000	8.08
Specialty	114.88	1.0000	1.0000	1.0000	1.4015	0.9875	0.9873	1.0000	1.0000	1.0000	156.98
	0.00	1.0000	1.0000	0000 1	0.906 0	0.9076		1.0000	1.0000	1.0000	000
Oliei	\$447.23	0000-1	1-0000	1-0000	10000	C 106-0		0000	0000-1	0000-1	\$460.49
Community Mental Health Center											
Case Management	\$208.43	1.0500	1.1530	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$254.87
Long Term Support Service	22.79	1.0500	1.1423	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	27.60
Partial Hospital	0.00	0060.1	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Psychomerapy Evidence Based Bractice	12.15	0050.1	1.0/33	0010.1	1,0000	1.0000	1.0000	1.0000	1.0000	00001	90.05 0.22
Evidence based Fractice Medication Management	6.02	1 0500	1.0631	10100	1 0000	1 0000	1 0000	1 0000	1 0000	1,0000	5 70 6 70
Emergency Service 24/7	0.11	1.0500	1.0631	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.13
APRTP	00.0	1.0500	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Supported Employment Services	5.41	1.0500	1.1262	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	6.46
Other	23.33	1.0500	1.0621	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	26.28
All Candada	2787¢										\$338.04
All del vices	10.001,10										01.000,16

Appendix B1 New Hampshire Department of Hathin and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Low Utilizer - Dual

Renefits	Per Capita Monthly Paid Cost	IRNR Adjustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiustment	PDI Adiustment	Medicare OUD Adjustment	Stop Loss Adiustment	NEMT Adjustment	Projected Per Capita Monthly Paid Cost
Hospital Inpatient			a la			a statistic to fact a			strating in the		
Medical	\$15.25	-	1.0692	1.0000	-	1.0000	1.0000	1.0000	-	1.0000	\$16.40
Surgical	11.92	-	1.0837	1.0000	-	1.0000	1.0000	1.0000	-	1.0000	12.99
Maternity Delivery	0.00	-	1.0000	1.0000	- ·	1.0000	1.0000	1.0000		1.0000	0.00
Maternity Non-Delivery	00.0		1.0000	1.0000		1.0000	1.0000	1.0000		1.0000	000
Psychiatric	1.93		1.5299	1 0000		1 0000	1 0000	1 0000		1 0000	2.97
Alcohol and Drug Abuse	00.0	1.0000	1.0000	1.0000	1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Other	0.00		1.0000	1.0000	-	1.0000	1.0000	1.0000		1.0000	0.00
	\$29.10										\$32.37
Hospital Outpatient	00 000										
Emergency Room	\$20.08		1.0604	1.0200	1.1468	1.0000	1.0000	1.0000	1.0000	1.0000	\$24.91
Badiology	282		1 0604	1.0200	1 1468	1 0000	1.0000		1 0000	1,0000	0 70
Pathology	1.63		1.0604	1.0200	1 1468	1.0000	1 0000		1 0000	1 0000	9.19 2.00
Pharmacv	5.23	- (-	1.0604	1.0200	1.1468	1.0000	1.0000	1.0000	1 0000	1.0000	6.49
Cardiovascular	0.84	-	1.0604	1.0200	1.1468	1.0000	1.0000	1.0000	1.0000	1.0000	1.04
PT/OT/ST	2.09		1.0604	1.0200	1.1468	1.0000	1.0000	1.0000	1.0000	1.0000	2.59
Psychiatric	0.08	~	1.0604	1.0200	1.1468	1.0000	1.0000	1.0000	1.0000	1.0000	0.10
Substance Abuse	0.00	1.0000	1.0000	1.0200	1.1468	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Other	14.32		1.0604	1.0200	1.1468	1.0000	1.0000	1.0000	1.0000	1.0000	17.76
	\$63.30										10.8/\$
Professional and Other State Plan Services	10 00		CCUC 1	1 0100		1 0000	1 0000			1 0000	60.04
Annuatory Surgery Center Office Mistre	15.01		1 0720	1.0100		1.0000	1.0000			1.0000	16.27
Preventive Medicine	1 77	- 、	1 0716	1 0100		1 0000	1 0000	1 0000	- 、	1 0000	191
Maternity	00.0	- 、	1 0000	1 0100		1 0000	1 0000	1 0000		1 0000	000
Certified Midwife	00.00		1.0000	1.0100		1.0000	1.0000	1.0000		1.0000	00.0
PT/OT/ST	1.53		1.0630	1.0100		1.0000	1.0000	1.0000		1.0000	1.65
Psychiatric and Substance Abuse	11.00	-	1.3782	1.0100	-	1.0000	1.0000	1.0000		1.0000	15.31
Radiology and Pathology	4.28	~	1.0573	1.0100	t-	1.0000	1.0000	1.0000	~	1.0000	4.57
Home Health and Private Duty Nursing	0.04		1.0628	1.0100	-	1.0000	1.0000	1.0000		1.0000	0.04
Ambulance	6.14		1.0630	1.0100		1.0000	1.0000	1.0000	,	1.0000	6.59
Non-Emergency Transportation	26.17	1.0000	1.0630	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	28.10
Opioid Treatment Program Fodorolly Ocidition and Durol Hoolth Olivion	40.7		1.0030	00101	- •	00001	0000	1.042.0	- •	0000 F	0.0
reversity Qualified and Rural realure Cilinos Adult Medical Day Care	21.12		1.01.1	1.0100		1 0000	1,0000			1,0000	2010
Personal Care	0.00	- (1.0000	1.0100		1.0000	1.0000	1.0000		1.0000	0.00
Durable Medical Equipment	10.07		1.9197	1.0100		1.0000	1.0000	1.0000		1.0000	19.53
Applied Behavioral Analysis	0.00		1.0000	1.0000	-	1.0000	1.0000	1.0000	-	1.0000	0.00
Other	32.43		1.0700	1.0100		1.0000	1.0000	1.0000		1.0000	35.04
1	\$117.40										\$136.96
Prescription Drugs	÷0		0000 1	0000 1	0000 0	0000 1	10000	0000	0000	0000 1	11 04
Generic Scripts Single Source Brand	0/.0¢		00001	1.0000	1 1838	1.0000	0.9094	1.0000	1.0000	1.0000	C/ 04
Multi-Source Brand	0.02		1.0000	1.0000	1.1838	1.0000	0.9894	1.0000	1.0000	1.0000	0.02
Specialty	0.00		1.0000	1.0000	1.0000	1.0000	0.9894	1.0000	1.0000	1.0000	00.0
Hepatitis C	0.00	1.0000	1.0000	1.0000	1.0000	1.0000	0.9894	1.0000	1.0000	1.0000	0.00
Other	0.00		1.0000	1.0000	0.9902	1.0000	0.9894	1.0000	1.0000	1.0000	0.00
Community Mental Health Center	10.1\$										\$1.04
Case Management	\$190.04		1.1530	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$232.38
Long Term Support Service	37.94	~	1.1423	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	45.97
Partial Hospital	2.56		1.0000	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	2.71
Psychotherapy	16.07		1.0753	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	18.33
Evidence Based Practice	0.55 07		1.0629	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.62 5 20
Demonstruction Management	4.78		1.0031	00101	1.0000	1.0000	0000 1	0000 1	1.0000		85.0 25.0
APRTP	2.51		00001	1 0100	1 0000	1 0000	1 0000		1 0000	1 0000	2.66
Supported Employment Services	23.94	1.0500	1.1262	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	28.60
Other	12.87		1.0621	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	14.50
	\$291.47										\$351.38
All Services	\$502.28		_		-		-			_	\$600.26

Appendix B1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data

Eligibility Category: Serious Emotionally Disturbed Child

Benefits	Per Capita Monthly Paid Cost	IBNR Adiustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiustment	PDL Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adiustment	Projected Per Capita Monthly Paid Cost
Hospital Inpatient	-		stington fast d	0		the second second					
Medical	\$2.62	1.0000	1.0842	1.0000	- ·	0.9938	-		·	1.0000	\$2.84
Surgical Metomity Dolivery	12.2	1.0000	1.08/8	1.0000		0.9938				1.0000	2.40
Maternity Non-Delivery	0.00	1 0000	1 0000	1 0000		0.9938				1 0000	0.07
Well Newborn	0.00	1.0000	1.0000	1.0000	-	0.9938	· ·			1.0000	00.0
Psychiatric	48.25	1.0000	1.0972	1.0000	-	0.9938	~			1.0000	52.92
Alcohol and Drug Abuse	0.24	1.0000	1.1410	1.0000	1.0060	0.9938	1.0000	1.0000	1.0000	1.0000	0.28
Otter	\$53.37	0000	0000	0000-1		0.9900				00001	\$58.50
Hospital Outpatient											
Emergency Room	\$20.34	1.0000	1.0604	1.0200	v -	0.9938	1.0000	1.0000	1.0000	1.0000	\$25.07
Surgery	4.24	1.0000	1.0604		~	0.9938	-		1.0000	1.0000	5.23
Radiology	4.04	1.0000	1.0604			0.9938			1.0000	1.0000	4.98
Phomonogy	12.2	1.0000	1.0004	- •		0.9938	- •		1.0000	0000 1	21.2
r namacy Cardiovascular	0.13	1.0000	1.0604		1.1468	0.9938			1.0000	1.0000	0.16
PT/OT/ST	2.03	1.0000	1.0604	-		0.9938	~		1.0000	1.0000	2.51
Psychiatric	1.69	1.0000	1.0604			0.9938	<i>(</i> -		1.0000	1.0000	2.08
Substance Abuse	0.00	1.0000	1.0000	1.0200	1.1468	0.9938			1.0000	1.0000	0.00
Other	7.39 ¢42.02	1.0000	1.0604			0.9938	1.0000		1.0000	1.0000	9.11 ¢£2.02
Professional and Other State Blan Services	041.00										70.700
Ambulatory Surgery Center	\$0.47	1.0000	0.9118		·	0.9938	1.0000	1.0000		1.0000	\$0.43
Office Visits	14.42	1.0000	1.0605		-	0.9938	-	1.0000	-	1.0000	15.35
Preventive Medicine	4.38	1.0000	1.0972	1.0100	-	0.9938	-	1.0000	v -	1.0000	4.83
Maternity	0.06	1.0000	1.0630		- ·	0.9938		1.0000		1.0000	0.06
Certified Midwite	0.00	1.0000	1.0629			0.9938		1.0000		1.0000	0.00
PT/OT/ST Psychiatric and Substance Abuse	3.31	1.0000	1.0031	1.0100	1.0000	0.9938		1.0000		1.0000	3.56
Radiology and Pathology	2.24	1.0000	1.0562			0.9938		1.0000		1.0000	2.38
Home Health and Private Duty Nursing	2.12	1.0000	1.0627	-	-	0.9938	-	1.0000	·	1.0000	2.26
Ambulance	2.58	1.0000	1.0630		-	0.9938	-	1.0000		1.0000	2.75
Non-Emergency Transportation	3.24	1.0000	1.0630			0.9938		1.0000		1.0000	3.46
Opioid Treatment Program Federally Orialified and Pural Health Clinics	10.00	1.0000	1 1725			0.9930		1.0000	- •	1.0000	11.84
	00.00	1.0000	1 0000			0.9938		1.0000		1 0000	00.0
Personal Care	00.00	1.0000	1.0000		-	0.9938	-	1.0000		1.0000	0.0
Durable Medical Equipment	4.47	1.0000	1.0758		-	0.9938	-	1.0000		1.0000	4.82
Applied Behavioral Analysis	2.00	1.0000	1.0630	4.6722	1.0000	0.9938	1.0000	1.0000	1.0000	1.0000	9.88
Olliel	\$69.31 \$69.31	1-0000	74/01			0.9930	0000-1	0000-1		1-0000	\$83.08
Prescription Drugs											0000
Generic Scripts	\$41.10	1.0000	1.0000	1.0000		0.9938	0.9937		1.0000	1.0000	\$29.95
Single-Source Brand	28.05	1.0000	1.0000	1.0000		0.9938			1.0000	1.0000	24.81
Multi-Source Brand	4.51	1.0000	1.0000	1.0000		0.9938			1.0000	1.0000	4.18
specially Hepatitis C	0.00	1.0000	1.0000	1.0000	1.0000	0.9938	0.9894	1 0000	1.0000	1.0000	0.00
Other	0.01	1.0000	1.0000	1 0000		0.9938			1.0000	1.0000	0.00
	\$95.76										\$100.78
Community Mental Health Center	00 2004	1 0500	1 1	0010		0000				0000	10 0000
Case Mariagerrien. Long Tarm Support Sanire	110.80	1.0300	1.1330			1.0000				1.0000	10 202¢
Partial Hospital	00.00	1.0500	1.0000	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	00.0
Psychotherapy	115.25	1.0500	1.0566	1.0100	-	1.0000	~			1.0000	129.14
Evidence Based Practice	0.00	1.0500	1.0000	1.0100	~	1.0000	~			1.0000	0.00
Medication Management	0.10	1.0500	1.0631	1.0100		1.0000				1.0000	0.11
APRTP	0.13	1.0500	1 1530	1 0100		1.0000	,			1.0000	0.16
Supported Employment Services	0.03	1.0500	1 1530	1 0100		1.0000	<i>(</i> -			1.0000	0.03
Other	16.35	1.0500	1.0621	1.0100	-	1.0000				1.0000	18.42
	\$479.94										\$572.78
All Services	\$/41.30										\$868.06

Appendix St. New Hampshire Department of Health and Human Services SFY 2023 Mediciald Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2018 MCO Encounter Base Experience Data
--

Eligibility Category: Granite Advantage - Medically Frail

Benefits	Per Capita Monthly Paid Cost	IBNR Adiustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Managed Care Savings Adiustment	PDL Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adiustment	Projected Per Capita Monthly Paid Cost
Hospital Inpatient	1000		uninenfau	21000	C10008 1	unumentau		unineninu.	unamentau	Hallionfau	1000 818 1
Medical	\$45.53	-	1.0872	1.0000	1.0060	0.9860	1.0000	1.0000	1.0000	1.0000	\$49.10
Surgical	38.16		1.0962	1.0000	1.0060	0.9860	1.0000	1.0000	1.0000	1.0000	41.49
Maternity Delivery	0.0	1.0000	1.0000	1.0000	1.0060	0.9860	1.0000	1.0000	1.0000	1.0000	0.00
	0.0	1.0000	1.120	1.0000	1.0060	0.9860	1.0000	1 0000	1.0000	1.0000	0.01
Psychiatric	13.09	1.0000	1.0661	1.0000	1.0060	0.9860	-	1.0000	1.0000	1.0000	13.84
Alcohol and Drug Abuse	12.30	1.0000	1.1292	1.0000	1.0060	0.9860	1.0000	1.0000	1.0000	1.0000	13.78
Olitei	\$109.17		0000-	0000-1	0000-1	0000-0		0000-1	0000-	0000-1	\$118.32
Hospital Outpatient											
Emergency Room	\$49.53	1.0000	1.0604	1.0200	1.1468	0.9860	1.0000	1.0000	1.0000	1.0000	\$60.58
Surgery	14.76	1.0000	1.0604	1.0200	1.1468	0.9860	1.0000	1.0000	1.0000	1.0000	18.05
Radiology	23.92	1.0000	1.0604	1.0200	1.1468	0.9860	1.0000	1.0000	1.0000	1.0000	97.67
Pharmacv	05.7	1 0000	1.0604	1.0200	1.1468	0.9660		1 0000	1 0000	1.0000	35.22
Cardiovascular	2.05	1.0000	1.0604	1.0200	1.1468	0.9860		1.0000	1.0000	1.0000	2.51
PT/OT/ST	5.15	1.0000	1.0604	1.0200	1.1468	0.9860		1.0000	1.0000	1.0000	6.30
Psychiatric	1.58	-	1.0604	1.0200	1.1468	0.9860		1.0000	1.0000	1.0000	1.93
Substance Abuse	1.05		1.0604	1.0200	1.1468	0.9860	(- ·	1.0000	1.0000	1.0000	1.28
Other	\$155.01	1.0000	1.0604	1.0200	1.1468	0.9860		1.0000	1.0000	1.0000	26.35 \$100.68
Professional and Other State Plan Services	0.0010										00.0014
Ambulatory Surgery Center	\$1.80	1.0000	1.1005	1.0100	1.0000	0.9860	1.0000		1.0000	1.0000	\$1.97
Office Visits	26.04	1.0000	1.0696	1.0100	1.0000	0.9860	1.0000		1.0000	1.0000	27.74
Preventive Medicine	2.60	1.0000	1.1326	1.0100	1.0000	0.9860	1.0000		1.0000	1.0000	2.93
Maternity	0.47	1.0000	1.0626	1.0100	1.0000	0.9860	1.0000		1.0000	1.0000	0.50
Certified Midwite PT/OT/ST	0.01	1.0000	1.0629	1 0100	1 0000	0.9860	1.0000	-	1.0000	1,0000	3.15
Psychiatric and Substance Abuse	60.57	1.0000	1.2267	1.0100	1.0000	0.9860	1.0000		1.0000	1.0000	73.99
Radiology and Pathology	19.85	1.0000	1.0636	1.0100	1.0000	0.9860	1.0000		1.0000	1.0000	21.02
Home Health and Private Duty Nursing	11.36	1.0000	1.0629	1.0100	1.0000	0.9860			1.0000	1.0000	12.02
Ambulance	5.91	1.0000	1.0630	1.0100	1.0000	0.9860			1.0000	1.0000	6.26
Non-Emergency Transportation Onioid Treatment Program	42.4U 22.82	1.0000	1.0030	1.0100	1.0000	0.9860			1.0000	1.0000	44.89 24.17
Federally Qualified and Rural Health Clinics	20.33	1.0000	1.1601	1.0100	1.0000	0.9860			1,0000	1,0000	23.48
Adult Medical Day Care	0.34	1.0000	1.0630	1.0100	1.0000	0.9860	·		1.0000	1.0000	0.36
Personal Care	0.44	1.0000	1.0624	1.0100	1.0000	0.9860			1.0000	1.0000	0.47
Durable Medical Equipment	16.76	1.0000	0.9316	1.0100	1.0000	0.9860			1.0000	1.0000	15.55
Applied benavioral Analysis Other	45.97	1.0000	1.000	1.0000	1.0000	0.9860	1.0000	1.0000	1.0000	1.0000	48.68
	\$280.63										\$307.19
Prescription Drugs											
Generic Scripts	\$31.19	1.0000	1.0000	1.0000	1.5041	0.9860			1.0000	1.0000	\$69.97
Siligle-source brand Multi-Source Brand	40.90	1,0000	1 0000	1,0000	1.0683	0.9860			1.0000	1,0000	04.70 13.64
Specialty	78.05	1.0000	1.0000	1.0000		0.9860			1.0000	1.0000	162.03
Hepatitis C	50.83	1.0000	1.0000	1.0000	0.2950	0.9860	0.9564	1.0000	1.0000	1.0000	14.14
Other	\$269.30		1.000	1.000		0.9860			1.0000	1.000	0.01 \$324.57
Community Mental Health Center											
Case Management	\$1.47	1.0500	1.1530	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$1.80
Long Term Support Service	2.25	1.0500	1.0919	1.0100		1.0000		1.0000	1.0000	1.0000	2.60
r aruar nospital Psvchotherapv	2.37	1.0500	1.0636	1.0100		1.0000		1.0000	1.0000	1.0000	2.67
Evidence Based Practice	0.01	1.0500	1.0627	1.0100	-	1.0000	¢.	1.0000	1.0000	1.0000	0.01
Medication Management	0.07	1.0500	1.0630	1.0100	-	1.0000	-	1.0000	1.0000	1.0000	0.08
Emergency Service 24/7	0.01		1.0631	1.0100	- •	1.0000		1.0000	1.0000	1.0000	0.01
APRTP Sunnorted Employment Services	0.04	1.0500	1261.1	1.0100	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	50 U
Other	4.25		1.0629	1.0100	. –	1.0000		1.0000	1.0000	1.0000	4.79
	\$11.32										\$13.06
All Services	\$826.34										\$953.81

Appendix B2 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Data Adjustments for CY 2019 MCO Encounter Base Experience Data

Eligibility Category: Low Income Children - Age 0-11 Months

Benefits Monthy Paid Benefits 0.00 Prospital Inpatient \$18.07 Medical 4.43 Maternity Non-Delivery 0.00 Alcohol and Drug Abuse 0.00 Alcohol and Orthatic 0.00 Supstance Abuse 0.00 Other 1.14 Psychiatic 0.00 Other 1.016 Psychiatic 0.00 Other 1.14 Psychiatic 0.00	Paid Bard IBNR Adjustment 8.07 1.0000 0.00 1.0000 0.01 1.0000 0.01 1.0000 0.01 1.0000 0.01 1.0000 0.01 1.0000 0.01 1.0000 0.01 1.0000 0.00 1.0000 0.00 1.0000			Unit Cost Trend Factors		PDL Adjustment	Medicare OUD Adjustment	Stop Lo Adjustm	NEMT Adjustment	Capita Monthly Paid Cost
patient control of the second		1.0603	1 40003	1 40003	1	הר השומאוווטווו	mannenfau	Inspine	nuo linen fou	1 414 0031
elivery letivery on-Delivery on utpatient Room Har Room Abuse Abuse Abuse all and Other State Plan Services Surgery Center Surgery Center and Substance Abuse and Pathology th and Private Duty Nursing		1.0603			0000	•				
elivery onn-Delivery onn d Drug Abuse utpatient Room Abuse Abuse Abuse all and Other State Plan Services s Surgery Center and Substance Abuse and Substance Abuse and Petrhology th and Private Duty Nursing			1.0000	1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	\$19.27
elivery on-Delivery al Drug Abuse utpatient Room Mause Abuse Abuse al and Other State Plan Services s Surgery Center and Substance Abuse and Substance Abuse and Pathology th and Private Duty Nursing		1.1028	1.0000	1.0060	1.0000	1.0000	1.0000		1.0000	4.92
on-Delivery orn d Drug Abuse utpatient Room Room Abuse Abuse Abuse Abuse Abuse and Other State Plan Services s Medicine idwife and Substance Abuse and Pathology th and Private Duty Nursing		1.0000	1.0000	1.0060	1.0000	1.0000			1.0000	0.0
al Drug Abuse utpatient Room ular Abuse Abuse Surgery Center Surgery Center Medicine Medicine and Substance Abuse and Pathology th and Private Duty Nursing		1.0000	1.0000	1.0060	1.0000	1.0000		1.0000	1.0000	0.0
J Drug Abuse urpatient Room Ular Abuse Abuse Abuse all and Other State Plan Services Surgery Center Surgery Center and Substance Abuse and Substance Abuse and Pathology th and Private Duty Nursing		1.0000	1.0000	1.0060	1 0000	1.0000	1.0000	·	1.0000	00.0
utpatient Room Ular Abuse Abuse Sugery Center Sugery Center Medicine Modicine and Substance Abuse and Pathology thand Private Duty Nursing		1.0000	1.0000	1.0060	1.0000	1.0000	·		1.0000	00.0
It patient I Room ular Abuse Abuse Surgery Center Surgery Center Surgery Center and Substance Abuse and Substance Abuse and Patholoy th and Private Duty Nursing		1.0000	1.0000	1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
urbatient ular Abuse Abuse Surgery Center Medicine Medicine and Substance Abuse and Substance Abuse and Substance Abuse fith and Private Duty Nursing										\$24.20
ular ular Abuse Surgery Center Surgery Center Medicine dwife and Substance Abuse and Pathology th and Private Duty Nursing		1 0660	0000	7077	0000	0000	0000		0000	30 000
ular Abuse		1.0558		1 1124				1.0000		3.38
ular Abuse 1 Abuse 2 Surgery Center 1 Medicine 5tate Plan Services 2 Medicine 4 In and Substance Abuse 4 and Substance Abuse 4 and Pathology 1 In and Private Duty Nursing 4		1 0558	1 0000	1 1124	1 0000	1 0000	1 0000		1 0000	4.50
ular Abuse		1.0558	1.0000	1 1124	1 0000	1.0000			1.0000	1.37
vascular (7.51 (7.51) (7.51) (7.51) ance Abuse ance Abuse and Abuse Visits Visits Visits Visits Medicine (1.51) (1		1.0558	1.0000	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	1.58
//ST iatric ance Abuse Ance Abuse Iatroy Surgery Center Visits Visits Midwife Midwife Midwife Midwife Anthology Iatro Ia		1.0558	1.0000	1.1124	1.0000	1.0000			1.0000	0.35
iatric ance Abuse ssional and Other State Plan Services 33 Visits Visits Midwife ed Midwife ed Midwife for for for for for for for for for for		1.0558	1.0000	1.1124	1.0000	1.0000			1.0000	1.33
ance Abuse and Abuse and Abuse and Abuse associated and Other State Plan Services associated and Other State Plan Services as Visits Usits and Substance Abuse and Midwife and Substance Abuse boy and Pathoboy and Pathoboy buse and Pathoboy buse and Pathoboy buse and Pathoboy and Pathoboy buse and Pathoboy and Pathoboy buse and Pathoboy and Pathoboy buse and Pathoboy buse and Pathobo buse and Pathobo buse and Pathoboy buse and Pathobo b		1.0000	1.0000	1.1124	1.0000	1.0000	•		1.0000	0.00
sional and Other State Plan Services 55 I alory Surgery Center Visite Plan Services 55 Visits Visite Medicine 55 Intrive Medicine 55 Midwife 55 Intrive Medicine 55 In	00 1.0000	1.0000	1.0000	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
ery Center State Plan Services Center Center Services State Plan Services State Blan Services			00001	1.1.124	nnnn I	nnnn I.	nnnn I		nnnn I.	13.51
ery Center ine substance Abuse athology 4 Private Duty Nursing	2									0.040
ine substance Abuse athology 1 Private Duty Nursing		1.3523	1.0000	1.0000	1.0000	1.0000			1.0000	\$0. 27
ine 1 Jubstance Abuse athology A Private Duty Nursing		1.0600	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	17.89
ubstance Abuse athology 9 Private Duty Nursing		1.0631	1.0000	1.0000	1.0000	1.0000	v -		1.0000	21.11
ubstance Abuse athology 1 Private Duty Nursing	00 1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	·	1.0000	0.00
and Substance Abuse and Pathology th and Private Duty Nursing		1.0677	1.0000	1.0000	1.0000	1.0000	•		1.0000	0.07
and Pathology th and Private Duty Nursing	1.0000	1.0030	0000	0000		1.0000	0000	0000	0000	79'N
th and Private Duty Nursing		1 0584	1 0000			1 0000			1 0000	2.23
		1.0629	1.0000	1.0000	1.0000	1.0000	•		1.0000	5.19
		1.0630	1.0000	1.0000	1.0000	1.0000			1.0000	2.12
ation	1.42 1.0000	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	.	1.0000	1.51
		1.0000	1.0000	1.0000	1.0000	1.0000			1.0000	0.00
Federally Qualified and Kural Health Clinics 20.52	00001	2051.1	0000	0000		00001	00001	0000		23.32
		1 0000								
Durable Medical Equipment 30.80	80 1.0000	1.0383	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	31.98
		1.0000	1.0000	1.0000	1.0000	1.0000	•		1.0000	00.00
		1.0641	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	35.75
\$132.83	83									\$141.98
Prescription Drugs				00101		0 0007		1 0000		53 20
e e e e e e e e e e e e e e e e e e e		1 0000		1 0503		0.9688				20.20
		1.0000	1.0000	1.0503	1.0000	0.9934			1.0000	0.48
		1.0000	1.0000	1.6705	1.0000	0.9882			1.0000	10.88
itis C	00 1.0000	1.0000	1.0000	1.0000	1.0000	0.9366	1.0000	1.0000	1.0000	0.00
		0000-1	0000-1	1.0420	0000-1	0.3044			0000-1	\$16.58
Community Mental Health Center	2									
63		1.0000	1.0000	1.0000	1.0000	1.0000	,	-	1.0000	\$0.00
		00001	0000	0000		00001	00001	0000	00001	000
		1.0000	1.0000	1.0000	1.0000	1.0000			1.0000	00.0
	1.0	1.0000	1.0000	1.0000	1.0000	1.0000	· ·		1.0000	00.00
	1.0	1.0000	1.0000	1.0000	1.0000	1.0000	-	1.0000	1.0000	00.00
incy Service 24/7		1.0000	1.0000	1.0000	1.0000	1.0000			1.0000	0.00
Supported Employment Services	0.00	1.0000	1.0000	00001	1.0000	1.0000		1.0000		
	1	1.0000	1.0000	1.0000	1.0000	1.0000			1.0000	00.0
	00									\$0.00
All Services \$207.38	38									\$229.63

Milliman

Appendix B2 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development- PHE Ending April 2023 Data Adjustments for CV 2019 MCO Encounter Base Experience Data

Eligibility Category: Low Income Children - Age 1-18 Years

	Per Capita Monthly Paid		Reimbursement	Utilization Trend	Unit Cost Trend	Managed Care Savings		Medicare OUD	Stop Loss	NEMT	Projected Per Capita Monthly
Benefits	Cost	IBNR Adjustment	Adjustment	Factors	Factors	Ħ	PDL Adjustment	Adjustment	Adjustment	Adjustment	Paid Cost
Hospital Inpatient	61 00			0000	1 0000	0000	0000	0000 1	·	0000	50 C3
Medical	4 - 90 1 50		1.0399		1.0060	1 0000		1.0000			32.U3 1.62
Maternity Delivery	00 0				1 0060	1.0000	1.0000			1.0000	00 0
Maternity Non-Delivery	0.02	1.0000		-	1.0060	1.0000	1.0000		1.0000	1.0000	0.02
Well Newborn	0.0		1.0879		1.0060	1.0000	1.0000	1.0000		1.0000	0.00
Psychiatric Alcohol and Drug Abruse	2.02				1.000		1.0000			1.0000	
Other	00.0	1.0000	1.0000	1.0000	1.0060	1,0000	1.0000	1,0000	1.0000	1.0000	00.0
	\$5.84										\$6.26
Hospital Outpatient											
Emergency Room	\$11.10			1.0000	1.1124	1.0000	1.0000	1.0000		1.0000	\$13.04
Surgery	3.07	1.0000			1.1124	1.0000	1.0000			1.0000	3.61
Radiology	3.30		1.0558	- •	1.1124	1.0000	1.0000			1.0000	3.88
Pharmoody	1.04	0000 1		- •	4711.1 4717 1		0000	1.0000		1.0000	/G.I
Priariiacy Cardiovascular	0.1-0 0.1-0				1.1124 1.124					1.0000	17.2
Calulovasculai DT/OT/ST	1 23				41111						1.45
Pevchiatric	31.0			- +	1 1124		1.0000				2t- C
Substance Abuse	00.0		1.0000		1 1124	1.0000	1.0000			1.0000	00.0
Other	5.73			-	1.1124	1.0000	1.0000	·	1.0000	1.0000	6.73
	\$27.98										\$32.87
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$0.64			1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	\$0.52
Office Visits	90.6			- ·	1.0000	1.0000	1.0000	1.0000		1.0000	9.61
Preventive Medicine	3.95				1.0000	00001	1.0000	1.0000		1.0000	4.20
Materilly				- •	1.0000		0000				/0.0
				- •	1.0000		0000			00001	0.00
Priverial Psychiatric and Substance Abuse	2.33										101
Radiology and Pathology	1.32				1.0000	1.0000	1.0000	1.0000		1.0000	1.40
Home Health and Private Duty Nursing	3.23		1.0629	-	1.0000	1.0000	1.0000	1.0000		1.0000	3.43
Ambulance	0.67			-	1.0000	1.0000	1.0000	1.0000		1.0000	0.71
Non-Emergency Transportation	1.28	1.0000		-	1.0000	1.0000	1.0000	1.0000		1.0000	1.36
Opioid Treatment Program	0.0		1.0636		1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Federally Qualified and Kural Health Clinics	8.44				1.0000	1.0000	1.0000	1.0000		1.0000	9.9
Adult Medical Day Care Dersonal Care							1.0000		1.0000	1.0000	
Durable Medical Fourinment	5.04				1 0000	00001	1,0000				4.98
Applied Behavioral Analysis	5.88				1 0000	1.0000	1.0000	1.0000		1.0000	11.83
Other	8.85			1	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	9.43
	\$57.17							Ī			\$66.44
Prescription Drugs											
Generic Scripts	\$9.72	1.0000	1.0000	1.0000	1.0420	1.0000	0.9927	1.0000		1.0000	\$10.06 2.50
Single-Source Brand Multi-Source Brand	0.00			,	1.0503		0.9000	1 0000	1.0000	1.0000	75.1
Specialty	16.55		1.0000	·	1.6705	1.0000	0.9882	1.0000		1.0000	27.33
Hepatitis C	00.00	1.0			1.0000	1.0000	0.9366	1.0000		1.0000	0.00
Other	0.00	1.0000	1.0000	1.0000	1.0420	1.0000	0.9844	1.0000	1.0000	1.0000	0.00
	\$35.97							Ī			\$47.28
Community Mental Health Center Case Management	40 0\$	1 0500	1 0630					1 0000			\$1.07
Long Term Support Service	0.50				1.0000	1.0000	1.0000	1.0000		1.0000	0.56
Partial Hospital	00.0		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	00'0
Psychotherapy	0.35			-	1.0000	1.0000	1.0000	1.0000		1.0000	0.37
Evidence Based Practice	00.0	1.0500		- ·	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Medication Management	0.00				1.0000	1.0000				1.0000	00.0
APRTP	0.0	1.0500			1.0000	1.0000			1.0000	1.0000	0.01
Supported Employment Services	00.0	1.0	1.0000	-	1.0000	1.0000	1.0000	1.0000		1.0000	00.0
Other	0.64	1.0500	·	1.0000	1.0000	1.0000			1.0000	1.0000	0.71
All C	\$2.46										\$2./3
All Services	9123.42										00.0010

Appendix B2 New Hampshire Department of Heatht and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development- PHE Ending April 2023 Data Adjustments for CY 2019 MCO Encounter Base Experience Data

Eligibility Category: Low Income Adults

	Per Capita					Managed Care					Projected Per
Benefits	Monthly Paid Cost	IBNR Adjustment	Reimbursement Adjustment	Utilization Trend Factors	Unit Cost Trend Factors	Savings Adjustment	PDL Adjustment	Medicare OUD Adjustment	Stop Loss Adjustment	NEMT Adjustment	Capita Monthly Paid Cost
Hospital Inpatient		10000						0000 1		6	0074
Nearcal Surviced	00.114		5/GU.I.		1.0060	0.9902	0000 1	00001	1.0000		\$18.90 12.57
Maternity Delivery	00.0	1.0000	1,0000	1 0000	1.0060	0.9902	1 0000	1.0000		1 0000	00.0
Maternity Non-Delivery	2.22	1.0000	1.1462	1.0000	1.0060	0.9902	1.0000	1.0000		1.0000	2.54
Well Newborn Devrehiatric	0.22	1.0000	1.0894	1.0000	1.0060	0.9902	1.0000	1.0000		1.0000	3 40
Alcohol and Drug Abuse	2.00	1.0000	1.0634	1.0000	1.0060	0.9902	1.0000	1.0000		1.0000	2.12
Other	00.00	1.0000	1.0000	1.0000	1.0060	0.9902	1.0000	1.0000		1.0000	00.0
	\$37.19										\$39.87
Hospital Outpatient	07 073		1 0660		NC11 1						¢ 47 10
Ellieigency Rooll Surgery	11.27		1.0558		1 1124	2066 U		1 0000	·		947.10 13.11
Radiology	15.56	1.0000	1.0558	1.0000	1.1124	0.9902	1.0000	1.0000		1.0000	18.10
Pathology	7.88	1.0000	1.0558		1.1124	0.9902	1.0000	1.0000	1.0000	1.0000	9.17
Pharmacy	8.67	1.0000	1.0558	·	1.1124	0.9902	1.0000	1.0000		1.0000	10.08
Cardiovascular	1.01	1.0000	1.0558		1.1124	0.9902	1.0000	1.0000		1.0000	1.17
PI/OI/SI Devolution	2./4	1.0000	1.0558	1.0000	1.1124	0.9902	1.0000	1.0000	1.0000	1.0000	3.18
Psychiatric Substance Abuse	0.14		8000 L		1124	0.9902	1.000	1.0000		1.0000	0.15
Other	17.20	1.0000	1.0558		1.1124	0.9902	1.0000	1.0000	1.0000	1.0000	20.01
	\$105.10										\$122.24
Professional and Other State Plan Services											
Ambulatory Surgery Center	06.0\$	1.0000	1.0973	1.0000	1.0000	0.9902	1.0000	1.0000	1.0000	1.0000	\$0.98
Uttice Visits Dreventive Medicine	7 70	1.0000	1.0608		1.0000	0.9902	0000 1	1.0000	1.0000	1.0000	18.8/ 8 00
Maternity	10.28	1.0000	1.0630	1.0000	1.0000	0.9902	1.0000	1.0000		1.0000	10.82
Certified Midwife	0.35	1.0000	1.0628		1.0000	0.9902	1.0000	1.0000	Ţ	1.0000	0.37
PT/OT/ST	1.51	1.0000	1.0630	·	1.0000	0.9902	1.0000	1.0000	-	1.0000	1.59
Psychiatric and Substance Abuse	23.94	1.0000	1.0730		1.0000	0.9902	1.0000	1.0000		1.0000	25.44
Radiology and Pathology	14.40	1.0000	1.0596	1.0000	1.0000	0.9902	1.0000	1.0000	1.0000	1.0000	15.11
Ambrilance	201		1.0029	ţ		2066.0		1 0000			0.10 2 12
Non-Emergency Transportation	17.25	1.0000	1.0630	1.0000	1.0000	0.9902	1.0000	1.0000	-	1.0000	18.16
Opioid Treatment Program	11.88	1.0000	1.0636	1.0000	1.0000	0.9902	1.0000	1.0000	-	1.0000	12.51
Federally Qualified and Rural Health Clinics	20.50	1.0000	1.1368	1.0000	1.0000	0.9902	1.0000	1.0000	1.0000	1.0000	23.08
Adult Medical Day Care Dersonal Care	000	1.0000	1.0630		1.0000	0.9902	1.000	1.0000		1.0000	000
Durable Medical Equipment	6.65	1.0000	1.0174	·	1.0000	0.9902	1 0000	1.0000		1,0000	6.70
Applied Behavioral Analysis	00.0	1.0000	1.0000	1.0000	1.0000	0.9902	1.0000	1.0000	~	1.0000	00.00
Other	26.12 #100 F0	1.0000	1.0702	1.0000	1.0000	0.9902	1.0000	1.0000	<u> </u>	1.0000	27.68
Drescription Drugs	0C-701¢										91/341
Generic Scripts	\$23.55	1.0000	1.0000	1.0000	1.1417	0.9902	1.1071	1.0000	1.0000	1.0000	\$29.47
Single-Source Brand	31.51	1.0000	1.0000		0.9525	0.9902	0.9337			1.0000	27.75
Multi-Source Brand	13.55	1.0000	1.0000	1.0000	0.9525	0.9902	0.3733			1.0000	4.77
Speciality Henafitis (32.31 9.01				1.48/2	0.9902	0.9/91				40.09 A 01
Other	0.01	1 0000	1.0000		1.1417	0.9902	0.9275	1 0000	1.0000	1.0000	0.0
	\$109.93										\$114.60
Community Mental Health Center	04 04	1 0500	1 0616		0000	0000	0000	0000		0000	10.04
Lona Term Support Service	00.0¢	1.0500	1.0598	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	o. 85
Partial Hospital	00.0	-	1.0000	·	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	00.0
Psychotherapy	1.25	-	1.0571		1.0000	1.0000		1.0000		1.0000	1.39
Evidence Based Practice	0.00		1.0629	1.0000	1.0000	1.0000		1.0000	1.0000	1.0000	0.00
Emergency Management Emergency Service 24/7	00.0		1.0514		1.0000	1.0000	·	1 0000		1.0000	0.00
APRTP	0.30	1.0500	1.0630	-	1.0000	1.0000		1.0000		1.0000	0.33
Supported Employment Services	0.00	1.0500	1.0629	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Other	\$4.67	nnen I	8700.I	0000-1		0000-I		1.0000		0000-1	\$5.20
All Services	\$419.48										\$455.33

Eligibility Category: CHIP

	Per Capita					Managed Care					Projected Per
Benefits	Monthly Paid Cost	BNR Adiustment	Reimbursement Adiustment	Utilization Trend	Unit Cost Trend Factors	Savings Adiustment	PDL Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adiustment	Capita Monthly Paid Cost
Hospital Inpatient											
Medical	\$2.11	-	1.0646	1.0000	1.0060	1.0000	1.0000	1.0000	6066.0	1.0000	\$2.24
Surgical	1.92	÷ ,	1.0526	1.0000	1.0060	1.0000	1.0000	1.0000	6066 0	1.0000	2.01
Maternity Uellvery	00.0		1.0000	0000 1	1.0000	00001	0000	1.0000	6066 D	1.0000	0.0
	0.00		1.0879	1.0000	1.0060	1.0000	1.0000			1.0000	0.00
Psychiatric	1.97	-	1.0386	1.0000	1.0060	1.0000	1.0000	- -	0	1.0000	2.04
Alcohol and Drug Abuse	0.19	1.0000	1.0000	1.0000	1.0060	1.0000	1.0000		0.9909	1.0000	0.19
Other	0.00	-	1.0000	1.0000	1.0060	0000.1	1.0000	1.0000	6066.0	1.0000	0.00
Hoenital Outnationt	07.0¢										00.00
Final Culpatient	CU 0\$	1 0000	1 0558	1 0000	1 1124			1 0000			\$10.49
Surgerv	3.22		1.0558	1.0000	1.1124	1.0000	1.0000		0	1.0000	3.74
Radiology	3.29	1 0000	1.0558	1 0000	1 1124	1.0000	1 0000	1.0000		1.0000	3.83
Pathology	1.24		1.0558	1.0000	1.1124	1.0000	1.0000		0	1.0000	1.44
Pharmacy	1.28	-	1.0558	1.0000	1.1124	1.0000	1.0000	-		1.0000	1.49
Cardiovascular	0.10	1.0000	1.0558	1.0000	1.1124	1.0000	1.0000	-		1.0000	0.12
PT/OT/ST	1.36	-	1.0558	1.0000	1.1124	1.0000	1.0000	1.0000		1.0000	1.58
Psychiatric	0.19	- ·	1.0558	1.0000	1.1124	1.0000	1.0000	- ·	0 0	1.0000	0.23
Substance Abuse	0.00	1.0000	1.0000	1.0000	1.1124	1.0000	1.0000	1.0000	0.9909	1.0000	0.00
	\$25.21		00001	0000-1	1.112	0000-1	0000-		D	0000-1	\$29.34
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$0.58	-	0.8585	1.0000	1.0000	1.0000	1.0000	1.0000	0	1.0000	\$0.49
Office Visits	9.50	-	1.0607	1.0000	1.0000	1.0000	1.0000	1.0000	o	1.0000	66'6
Preventive Medicine	4.55	-	1.0758	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	4.85
Maternity	0.02		1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	0.9909	1.0000	0.02
Certified Midwite	0.00		1.0689	1.0000	1.0000	1.0000	1.0000	1.0000	6066.0	1.0000	0.00
Privo 1/31 Psychiatric and Substance Abuse	10.1 70.1		1 0553	1,0000							e/ 4
Radiology and Pathology	1.41	- - -	1.0579	1.0000	1.0000	1.0000	1 0000	1.0000	0	1.0000	1.48
Home Health and Private Duty Nursing	1.53		1.0628	1.0000	1.0000	1.0000	1.0000	1.0000	0	1.0000	1.61
Ambulance	0.50	-	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.53
Non-Emergency Transportation	66.0	1.0000	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	0	1.0000	1.04
Opioid Treatment Program	0.00	- 1	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Federally Qualified and Rural Health Clinics	58.C		CC51.1	0000 1	0000		0000 F	1.0000	6066 0	1.0000	00.0
Addit Medical Day Care Personal Care	000			1.0000	1 0000	1 0000					
Durable Medical Equipment	5.21		0.9895	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	5.11
Applied Behavioral Analysis	5.85	-	1.0630	2.1849	1.0000	1.0000	1.0000	1.0000	o	1.0000	13.47
Other	9.15	-	1.1335	1.0000	1.0000	1.0000	1.0000	1.0000	0	1.0000	10.28
	\$54.64										\$65.41
Prescription Drugs	00 04		10000	0000 1	0010 1	0000	1000 0	0000	c	0000	0100
Generic Scripts Single Source Brand	00.00	- •	1.0000	1.0000	1.0420		0.992/0		0,000 0		01.94
Multi-Source Brand	0.94	1.0000	1.0000	1.0000	1.0503	1.0000	0.9934	1.0000		1.0000	0.97
Specialty	22.80	-	1.0000	1.0000	1.6705	1.0000	0.9882	1.0000		1.0000	37.30
Hepatitis C	0.19		1.0000	1.0000	1.0000	1.0000	0.9366	1.0000	00	1.0000	0.17
Other	0.00 \$41 50	nnnn I.	nnnn I.	0000-L	1.0420	nnnn-L	0.9844	0000.1		nnnn-L	0.00 \$56.31
Community Mental Health Center	00.140										0.00¢
Case Management	\$0.51	£-	1.0630	1.0000	1.0000	1.0000	1.0000	-	0.9909	1.0000	\$0.56
Long Term Support Service	0.34	- ·	1.0626	1.0000	1.0000	1.0000	1.0000	~	0	1.0000	0.38
Partial Hospital			1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	6066 ⁻ 0	1.0000	0.00
Esychotherapy Evidence Based Dractice	00 0	- •	1.0400	1.0000						1.0000	
Medication Management	00.0		1,0000	1.0000	1.0000	1.0000	1.0000	1 0000		1.0000	00.0
Emergency Service 24/7	00'0	-	1.0000	1.0000	1.0000	1.0000	1.0000	-	0.9909	1.0000	00'0
APRTP	0.04		1.0630	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.04
Supported Employment Services Other	0.00	1.0500	1.0000	1.0000	1.0000	1.0000	1.0000		6066 U	1.0000	0.00
	\$1.54		2400-1	2000	-	-	0000-	-	0000		\$1.69
All Services	\$129.10										\$159.25

Eligibility Category: Foster Care / Adoption

	Per Capita					Managed Care					Projected Per
Benefits	Monthly Paid Cost	IBNR Adiustment	Reimbursement Adiustment	Utilization Trend Unit Cost Trend Factors Factors	Unit Cost Trend Factors	Savings Adiustment	PDL Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adiustment	Capita Monthly Paid Cost
Hospital Inpatient	8										
Medical	\$4.08	-	1.0642	1.0000	1.0060	0.9938	·	1.0000		1.0000	\$4.34
Surgical	2.39	1.0000	1.0785	1.0000	1.0060	0.9938	1.0000	1.0000	1.0000	1.0000	2.58
Maternity Delivery	0.0	, ,	1.0000	1.0000	1.0060	0.9938				1.0000	0.00
Maternity Non-Delivery Well Newhorn	000	1.0000	1.0000	1.0000	1.0060	0.9938		1.000	1.0000	1.0000	00.0
Psychiatric	20.23		1.0548	1.0000	1.0060	0.9938				1.0000	21.33
Alcohol and Drug Abuse	0.11		1.0599	1.0000	1.0060	0.9938	1.0000	1.0000	¢-	1.0000	0.11
Other	0.00	1.0000	1.0000	1.0000	1.0060	0.9938	1.0000		1.0000	1.0000	0.00
Hoovital O. thatiant	18.02¢										\$28.3/
Find The State of the State of	\$11 98	1 0000	1 0558		1 1124	0 9938			1 0000		\$13 99
Surgerv	4.79		1.0558	1.0000	1.1124	0.9938	- \	1.0000	1.0000	1.0000	5.59
Radiology	5.86	-	1.0558	1.0000	1.1124	0.9938	-	1 0000		1.0000	6.84
Pathology	1.95	-	1.0558	1.0000	1.1124	0.9938	-	1.0000	÷	1.0000	2.28
Pharmacy	0.94	-	1.0558	1.0000	1.1124	0.9938		1.0000		1.0000	1.10
Cardiovascular	0.22	-	1.0558	1.0000	1.1124	0.9938	-	1.0000	-	1.0000	0.26
PT/OT/ST	3.44	1.0000	1.0558	1.0000	1.1124	0.9938	- ·	1.0000		1.0000	4.02
Psychiatric	19.0		1.0558	1.0000	1.1124	0.9938	- 1	1.0000		1.0000	1.13
Substance Abuse	0.00	1.0000	1.000	1.0000	1.1124	0.9938	1.000	1.0000	1.0000	1.0000	0.00
	\$39.35			-				-		-	\$45.93
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$0.83		0.9921	1.0000	1.0000	0.9938	·	1.0000		1.0000	\$0.82
Office Visits	11.40	,	1.0598	1.0000	1.0000	0.9938	1.0000	1.0000	1.0000	1.0000	12.01
Preventive Medicine	4.30		1.0/45	1.0000	1.0000	0.9938		1.0000		1.0000	4.59
Malerriity Cartifiad Midwifa	0000		1.0000			0.9930			1.0000		000
PT/OT/ST	5.58		1.0630	1.0000	1.0000	0.9938		1 0000		1.0000	5.90
Psychiatric and Substance Abuse	16.69	1.0000	1.0583	1.0000	1.0000	0.9938	-	1.0000	1.0000	1.0000	17.56
Radiology and Pathology	2.50	~	1.0590	1.0000	1.0000	0.9938	-	1.0000		1.0000	2.63
Home Health and Private Duty Nursing	28.59		1.0629	1.0000	1.0000	0.9938	1.0000	1.0000		1.0000	30.20
Ambulance Non Emoranous Transmission	1.40		1.0630	1.0000	00001	0,9938	- •	0000 1		1.0000	1.48
Opioid Treatment Program	00.00	1.0000	1.0000	1.0000	1.0000	0.9938		1.0000	1.0000	1.0000	00.0
Federally Qualified and Rural Health Clinics	9.12	-	1.1326	1.0000	1.0000	0.9938	-	1.0000		1.0000	10.27
Adult Medical Day Care	0.00	1.0000	1.0000	1.0000	1.0000	0.9938	1.0000	1.0000	1.0000	1.0000	0.00
Personal Care	00.0	. .	1.0000	1.0000	1.0000	0.9938	, - •	1.0000		1.0000	00.00
Durable Medical Equipment	19.93	1.0000	0.8612	1.0000	1.0000	0.9938	1.0000	1.0000	1.0000	1.0000	17.06
Applied Deliavioral Arialysis Other	17.76		1.0445	1.0000	1.0000	0.9938		1.0000	1.0000	1.0000	18.43
	\$125.67										\$133.18
Prescription Drugs											
Generic Scripts	\$29.65	1.0000	1.0000	1.0000	0.8925	0.9938	0.9916	1.0000	1.0000	1.0000	\$26.07
Single-Source Brand Multi-Source Brand	2/ 1/2		1.0000	1.0000	1.18/2	0.9938		1.000		1.0000	24.97
Specialty	23.53	1.0000	1.0000	1.0000	1.7134	0.9938		1.0000		1.0000	37.69
Hepatitis C	00.0	~	1.0000	1.0000	1.0000	0.9938	0.9894	1.0000	1.0000	1.0000	00.0
Other	0.00	1.0000	1.0000	1.0000	0.8925	0.9938		1.0000	1.0000	1.0000	0.00
Community Mental Health Center	\$/8.01										\$93.UQ
Case Management	\$2.63	1.0500	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$2.94
Long Term Support Service	1.58	,	1.0629	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	1.77
Partial Hospital Devototherativ	00.0	1.0500	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Freidence Based Practice	000		00001	1 0000	1 0000	1 0000	1 0000	1 0000		1 0000	00.0
Medication Management	00.00		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Emergency Service 24/7	00.0	1.0500	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	00.0
APRTP Supported Employment Services			1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	000
Other	1.62		1.0629	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	1.81
											\$7.45
All Services	\$277.40	_									\$308.01

Eligibility Category: Severely Disabled Children

Domofiles	Monthly Paid	IDND Adirictment	Reimbursement	Utilization Trend	Unit Cost Trend		DDI Adiretment	Medicare OUD	Stop Loss	NEMT	Capita Monthly
Hospital Inpatient	1000			1 401013	1 401013			Valasmon	ununenteu	שממנוובווו	1 414 0031
Medical	\$14.96		·-	1.0000	1.0060	0.9899	1.0000	1.0000	1.0000	1.0000	\$15.79
Surgical	4.88		-	1.0000	1.0060	0.9899	1.0000	1.0000	1.0000	1.0000	5.23
Maternity Delivery	00.00		~	1.0000	1.0060	0.9899	1.0000	1.0000	1.0000	1.0000	00.0
Maternity Non-Delivery	0.0	1.0000	1.0000	1.0000	1.0060	0.9899	1.0000	1.0000	1.0000	1.0000	0.00
Vell Newboll Psychiatric	00.0			1.0000	1 0060	0.9899	1.0000				4.85
Alcohol and Drug Abuse	00.00			1 0000	1.0060	0.9899	1.0000	1 0000		1 0000	00.0
Other	00.0			1.0000	1.0060	0.9899	1.0000	1.0000		1.0000	00.0
	\$23.74										\$25.86
Hospital Outpatient											:
Emergency Room	\$8.47	1.0000	1.0558	1.0000	1.1124	0.9899	1.0000	1.0000	1.0000	1.0000	\$9.85
Surgery	3.88		, ,	1.0000		0.9899	1.0000	1.0000	1.0000	1.0000	4.51
Radiology	11.08	1.0000	1.0558	1.0000		0.9899	1.0000	1.0000	1.0000	1.0000	88.21
Parmology	10.2				4711 I	0.9039				0000	3.2/ A DE
Pharmacy	62.4		- 1	0000		0.9899		00001		00001	0.4
Cardiovascular DT/DT/ST	0.00					0.9899	0000			0000	47.0 17.0
PI/OI/SI	0.00				4711 I	0.9099			0000 1	0000	// 7I
Estance Abico	000			1.0000	4711.1 4714 1	0.9099			1.0000	0000	00.0
Outber	0.00				1 1124	9999 U		1 0000			0.0
	\$50.99			-	-	2	-	-		-	\$59.29
Professional and Other State Plan Services	-										
Ambulatory Surgery Center	\$0.68	1.0000	0.7512	1.0000	1.0000	0.9899	1.0000	1.0000	1.0000	1.0000	\$0.51
Office Visits	15.73	~	1.0605	·		0.9899	1.0000	1.0000	1.0000	1.0000	16.51
Preventive Medicine	1.90				·	0.9899	1.0000	1.0000		1.0000	2.01
Maternity	00.00					0.9899	1.0000	1.0000		1.0000	00.0
Certified Midwife	00.00					0.9899	1.0000	1.0000	1.0000	1.0000	00.0
PT/OT/ST	25.92			·	·	0.9899	1.0000	1.0000	1.0000	1.0000	27.27
Psychiatric and Substance Abuse	6.78					0.9899	1.0000	1.0000	1.0000	1.0000	7.03
Radiology and Pathology	2.54					0.9899	1.0000	1.0000	1.0000	1.0000	2.65
Home Health and Private Duty Nursing	364.24	1.0000	1.0629	•	1.0000	0.9899	1.0000	1.0000	1.0000	1.0000	383.23
	00.01					0.9899	0000.1	1.000	0000.1	0000.1	00.01 00.01
Non-Emergency I ransportation	00.01					0.9899			0000 F	nnnn I	00.0
Conduct Treatment Program	0.00					0.9099			0000 1	0000	00.0
Adult Modical Dev Caro	000		1.1324			0.9099			1.0000	0000	
Personal Care	0.60		1 0624	1 0000		0.9899	1 0000	1 0000	1 0000	1 0000	0.63
Durable Medical Equipment	136.48		1 4876	1 0000		0,9899	1 0000	1 0000		00001	000
Applied Behavioral Analysis	40.92		1.0630	1.9403	1.0000	0.9899	1.0000	1.0000	1.0000	1.0000	83.54
Other	43.86		0.7963	1.0000		0.9899	1.0000	1.0000	1.0000	1.0000	34.57
	\$664.05										\$784.75
Prescription Drugs											
Generic Scripts	\$49.64		(-	1.0000	1.5793	0.9899	0.9905	1.0000	1.0000	1.0000	\$76.87
Single-Source Brand	69.53	1.0000		1.0000	1.38/4	0.9899	0.9544	1.0000	1.0000	1.0000	91.14
Nutre-Source Diaria Seecialty	16.00		- (1,00.1	0.9099	0.0910		1.0000		188 83
Hepatitis C	00.0	1.0000	1.0000	1.0000	1.0000	0.9899	0.9894	1.0000	1.0000	1.0000	0.00
Other	00.0			1.0000	1.5793	0.9899	0.9751	1.0000		1.0000	00.0
	\$338.50										\$424.84
Community Mental Health Center											:
Case Management	\$2.31	1.0500	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	2.57
Lorig Terri Support Service Partial Hosnital	00.0										
Psychotherapy	0.37		, t-	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.42
Evidence Based Practice	00.00		· ·	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	00.0
Medication Management	00.0		(-	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	00.00
Emergency Service 24/7	0.01		1.0631	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.01
APRIP Supported Employment Services	00.0		-	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	000
outported Employment dervices	0.62	1.0500	- (-	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.00
	\$6.02										\$6.72
All Services	\$1,083.30										\$1,301.46

Eligibility Category: Elderly and Disabled Adults - Age 19-64

	Per Capita					Managed Care					Projected Per
Benefits	Monthly Paid Cost	BNR Adiustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors	Savings Adiustment	PDL Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adiustment	Capita Monthly Paid Cost
Hospital Inpatient			•				•				
Medical	\$57.69		~	1.0000	1.0060	0.9891	1.0000	1.0000	0.9971	1.0000	\$60.36
Surgical	37.89		~	1.0000	1.0060	0.9891	1.0000	1.0000	0.9971	1.0000	39.92
Maternity Delivery	00.0			1.0000	1.0060	0.9891	1.0000	1.0000		1.0000	00.0
Maternity Non-Delivery	60.0	1.0000	1.1511	1.0000	1.0060	0.9891	1.0000	1.0000	0.9971	1.0000	0.10
Psychiatric	7.35			1.0000	1.0060	0.9891	1.0000	1.0000		1.0000	8.12
Alcohol and Drug Abuse	2.55		`	1.0000	1.0060	0.9891	1.0000	1.0000	0.9971	1.0000	2.69
Other	00.00		1.0000	1.0000	1.0060	0.9891	1.0000	1.0000	0.9971	1.0000	00.00
	\$105.56										\$111.19
Hospital Outpatient											
Emergency Room	\$56.61	1.0000	1.0558	1.0000	1.1124	0.9891	1.0000	1.0000	0.9971	1.0000	\$65.58
Surgery	23.47		1.0558		1.1124	0.9891	1.0000	1.0000		1.0000	27.19
Radiology	32.48	1.0000	1.0558	1.0000	1.1124	0.9891	1.0000	1.0000		1.0000	37.62
Pharmack	07.0		0000.1	- •	11124	0.9091				0000	90.5 90.7 N
Primiecy Cardiovascular	40.00		1.0330		4711.1 47174	0.9091	1.0000		0.0071	1.0000	4/ 20
	2.20 6.03		1.0330	·	4711.1	0.9091	1.0000			1.0000	6/.7 80 9
Devrehiatric	20.0		1 0558	1.0000	11124	0.9091	1.0000	1.0000	1.000 0		0.55
Substance Abuse	0.14		1 0558		1 1124	0.9891	1 0000			00001	0.16
Other	32.15		1.0558	. –	1 1124	0.9891	1 0000	1.0000	0.9971	1 0000	37.24
	\$202.78										\$234.90
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$2.28	-	1.1878	1.0000	1.0000	0.9891	1.0000	1.0000	0.9971	1.0000	\$2.67
Office Visits	26.43	1.0000	1.0610	1.0000	1.0000	0.9891	1.0000	1.0000	0.9971	1.0000	27.65
Preventive Medicine	3.44		1.1330	1.0000	1.0000	0.9891	1.0000	1.0000		1.0000	3.85
Maternity	0.23		1.0630	1.0000	1.0000	0.9891	1.0000	1.0000	0.9971	1.0000	0.25
Certified Midwife	0.0		1.0000	1.0000	1.0000	0.9891	1.0000	1.0000		1.0000	0.00
PI/OI/SI	3./5	1.0000	1.0630	1.0000	1.0000	0.9891	1.0000	1.0000	0.99/1	1.0000	3.93
Padiatory and Dathology	10.02		1.0720			0.0800			1.000.0		11.00
Home Health and Private Duty Nursing	56.64		1.00.1	1 0000	1 0000	0.9091	1 0000	1 0000	0.9971	00001	59.37
Ambulance	6.66	÷	1.0630	1 0000	1.0000	0.9891	1 0000	1 0000	0.9971	1 0000	6.98
Non-Emergency Transportation	27.58	1.0000	1.0630	1.0000	1.0000	0.9891	1.0000	1.0000	0.9971	1.0000	28.91
Opioid Treatment Program	9.63	.	1.0636	1.0000	1.0000	0.9891	1.0000	1.0000	0.9971	1.0000	10.10
Federally Qualified and Rural Health Clinics	17.60	1.0000	1.1345	1.0000	1.0000	0.9891	1.0000	1.0000	0.9971	1.0000	19.70
Adult Medical Day Care	2.16		1.0630	1.0000	1.0000	0.9891	1.0000	1.0000	0.9971	1.0000	2.26
	79.61		1.0624	0000	1.0000	0.9891	1.0000	1.0000	1/66.0	1.0000	16.58
uurable medical Equipment Annlied Behavioral Analysis	10.04		0.94 IO 1 0630		1.0000	0.9691	1.0000		0.9971	1.0000	0.10 0.10
Other	65.37		1.0913	1.0000	1.0000	0.9891	1.0000	1.0000	0.9971	1.0000	70.35
	\$309.85										\$323.53
Prescription Drugs											
Generic Scripts	\$68.70	1.0000		• •	1.0340	0.9891	1.0248	1.0000		1.0000	\$71.79 465.00
Multi-Source Brand	47.83		1 0000	1 0000	1 1153	0.9891	0.6783	1 0000	1.9971	0000	35.68
Specialty	154 18		·	,	1.3640	0.9891	0.9884	1.0000		1 0000	205.00
Hepatitis C	16.96	-	1.0000	-	0.4873	0.9891	0.9666	1.0000		1.0000	7.88
Other	00.00	1.0000	1.0000	1.0000	1.0340	0.9891	0.9544	1.0000	0.9971	1.0000	00.00
Community Montel Haatth Conter	\$447.92										\$485.69
Community mental health Center	\$0 \$5	1 0500	1 0611	1 0000	1 000	1 0000	1 0000	1 0000	0 9971	1 0000	\$0.95
Lona Term Support Service	1.34			1.0000	1.0000	1.0000	1.0000	1.0000	0.9971	1.0000	1.50
Partial Hospital	00.0		-	1.0000	1.0000	1.0000	1.0000	1.0000	0.9971	1.0000	00.0
Psychotherapy	2.02		·	-	1.0000	1.0000	1.0000	1.0000	0.9971	1.0000	2.25
Evidence Based Practice	0.01			1.0000	1.0000	1.0000	1.0000	1.0000	0.9971	1.0000	0.01
Medication Management Emergency Service 24/7	80.0		1.0620		1.0000	00001	1.0000	1.0000		1.0000	60'0
APRTP	0.72		·	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.80
Supported Employment Services	0.02		·	·	1.0000	1.0000	1.0000	1.0000	0.9971	1.0000	0.02
Other	2.92		1.0629	1.0000	1.0000	1.0000	1.0000	1.0000	0.9971	1.0000	3.25
All Services	19/14/19/19/19/19/19/19/19/19/19/19/19/19/19/									T	\$8.88
All Oct VIGGS	÷			-	-	-			-	-	÷

Eligibility Category: Elderly and Disabled Adults - Age 65+

Benefits Hospital Inpatient Medical Surgical Matemity Delivery	Menther Capita					Manage Course					Due is also d Dau
Hospital Inpatient Medical Surgical Matemity Delivery		IPND Adjuctment	Reimbursement	Utilization Trend	Unit Cost Trend	Savings	DDI Adiretment	Medicare OUD	Stop Loss	NEMT	Capita Monthly
Medical Surgical Maternity Delivery	1500		שממוווכווו	1 40101 3	1 401013			Vajasment	עממצוווכוור		1 814 0031
Surgical Maternity Delivery	\$93.82	1.0000	-	1.0000		0.9917	1.0000	1.0000	1.0000	1.0000	\$97.51
Maternity Delivery	68.77		1.0439	1.0000		0.9917	1.0000	1.0000	1.0000	1.0000	71.62
	00.0		-	1.0000		0.9917	1.0000	1.0000	1.0000	1.0000	00.0
Maternity Non-Delivery	000	1.0000	1.0000	1.0000	1.0060	0.9917	1.0000	1.0000	1.0000	1.0000	0.00
Devchiatric	0.00					0.0017	1.0000			1.0000	0.00
Alcohol and Drug Abuse	00.0			1.0000		0.9917	1.0000	1 0000	1.0000	1,0000	00.0
Other	00.00		-	1.0000		0.9917	1.0000	1.0000	1.0000	1.0000	00.00
	\$166.89										\$173.89
Hospital Outpatient											
Emergency Room	\$48.87	1.0000	1.0558	1.0000	1.1124	0.9917	1.0000	1.0000	1.0000	1.0000	\$56.93
Surgery	11.25		1.0558	1.0000	1.1124	0.991/	1.0000	1.0000	1.0000	1.0000	20.09
Radiology	30.23	1.0000	1.0558	1.0000	1.1124	/ 188.0	1.0000	1.0000	0000	1.0000	35.27
Pharmood	9. IL		1.0000		11/24	1.991/				0000	10.01
Pharmacy	23.33 7.60		9000 F		1.1.124	1.99.0				0000	11.12
Cargiovascular DT/OT/OT	2.09		8000 I.	00001	1.1124	1.99.0	0000	1,0000		0000	0.14 4
Pri/Ol/SI Developtio					1.1124	0.0017				0000	70.0
Psychiatric	000		0000 1		11124	1991/					00.0
Other	38.58	1 0000	1 0558		1 1124	0.9917				00001	44.94
	\$175.66		-	-	-		-	-	-	-	\$204.62
Professional and Other State Plan Services	-										
Ambulatory Surgery Center	\$5.01	1.0000	1.0882	1.0000	1.0000	0.9917	1.0000	1.0000	1.0000	1.0000	\$5.40
Office Visits	24.83	1.0000	1.0609	1.0000		0.9917	1.0000	1.0000	1.0000	1.0000	26.12
Preventive Medicine	2.71		1.3083	1.0000	·	0.9917	1.0000	1.0000	1.0000	1.0000	3.51
Maternity	00.00		1.0000	1.0000		0.9917	1.0000	1.0000	1.0000	1.0000	0.00
Certified Midwife	00.00		1.0000	1.0000	•	0.9917	1.0000	1.0000	1.0000	1.0000	0.00
PT/OT/ST	2.73	1.0000	1.0630	1.0000	1.0000	0.9917	1.0000	1.0000	1.0000	1.0000	2.88
Psychiatric and Substance Abuse	0.99		1.0530	00001		1.99.0	0000	1.0000	0000	0000	1.05
Hame Upoth and Private Duty Number	76.11		1.0249	0000 1		/188.0	0000 1	0000	0000 F	0000 F	GU.21
Ambulance	8 73		1 0630	1 0000	·	0.9917				00001	00.6
Non-Emergency Transportation	12.54		1.0630	1.0000	·	0.9917	1 0000	1 0000	1 0000	1 0000	13.22
Opioid Treatment Program	0.89		1.0636	1.0000	1.0000	0.9917	1.0000	1.0000	1.0000	1.0000	0.94
Federally Qualified and Rural Health Clinics	22.85		1.1369	1.0000	1.0000	0.9917	1.0000	1.0000	1.0000	1.0000	25.77
Adult Medical Day Care	30.56		1.0630	1.0000	1.0000	0.9917	1.0000	1.0000	1.0000	1.0000	32.21
Personal Care	0.17	1.0000	1.0630	1.0000	1.0000	0.9917	1.0000	1.0000	1.0000	1.0000	0.18
Durable Medical Equipment	23.13		0.9848	1.0000	1.0000	0.9917	1.0000	1.0000	1.0000	1.0000	22.59
Applied Demayoral Amarysis Other	74.02		1.0508	1 0000	1.0000	0.9917	1 0000	1.0000	1.0000	1.0000	77 13
	\$271.71										\$286.05
Prescription Drugs											
Generic Scripts	\$56.07		1.0000	1.0000	1.0340	0.9917	1.0248	1.0000	1.0000	1.0000	\$58.92
Single-Source Brand	135.98	1.0000	1.0000	1.0000	1.1153	0.9917	0.9380	1.0000	1.0000	1.0000	141.08
Multi-Source Brand	19.86		1.0000	1.0000	1.1153	0.991/	0.6/83	1.0000	1.0000	1.0000	14.90 68.05
Specially Henstitis C					0.4873	0.9917	0.9004				
Other	00.0		1.0000	1.0000	1.0340	0.9917	0.9544	1 0000	1.0000	1.0000	00.0
	\$263.41										\$283.76
Community Mental Health Center											
Case Management	\$0.15	1.0500	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$0.17
Long Term Support Service Partial Hosnital	200 0		1.0029			1.0000		1.0000	1.0000	1.0000	0.00
Psychotherapy	0.23		1.0630	1.0000	1 0000	1.0000	1.0000	1 0000	1 0000	1,0000	0.26
Evidence Based Practice	00.0		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	00.0
Medication Management	00.0		1.0000	1.0000		1.0000	1.0000	1.0000		1.0000	00.0
Emergency Service 24/7	00.0		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	00'0
APRTP	00.0		1.0000	1.0000		1.0000	1.0000	1.0000		1.0000	0.00
Supported Employment Services Other	0.40	1.0500	1.0630	1.0000		1.0000	1.0000	1.0000		1.0000	0.44
	\$0.81										\$0.91
All Services	\$878.48										\$949.22

Eligibility Category: Dual Eligibles

	Dor Canita					Manado Cam					Droioctod Dor
Banofite	Monthly Paid	IBND Adjustment	Reimbursement	Utilization Trend	Unit Cost Trend	Savings Adiustment	DDI Adirietment	Medicare OUD	Stop Loss	NEMT	Capita Monthly
Hospital Inpatient	500			00000	0000	unumphu		mannahar	Molliophy	monitori	2000 515 -
Medical	\$19.16	1.0000	1.0418	1.0000	1.0060	1.0000		1.0000		1.0000	\$20.08
Surgical	6.40	1.0000	1.0439	1.0000	1.0060	1.0000		1.0000	·	1.0000	6.72
Maternity Delivery	0.00	1.0000	1.0000	1.0000	1.0060	1.0000	·	1.0000	•	1.0000	00.0
Maternity Non-Delivery	0.06	1.0000	1.0000	1.0000	1.0060	1.0000		1.0000		1.0000	90.0
Psychiatric	163		1 1078	1,0000	1 0060						1.82
Alcohol and Drug Abuse	0.41	1.0000	1.0000	1.0000	1.0060	1.0000	1.0000	1.0000		1.0000	0.42
Other	0.0	1.0000	1.0000	1.0000	1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	00.00
	\$27.67										\$29.10
Hospital Outpatient	¢00.07		1 0660		VCFF F		Ţ			0000	671 GA
Surgericy Nouri	683 683		1 0558	1 0000	1 1124		1 0000			00001	#0.124
Radiolony	6.59		1 0558	1 0000	1 1124	1 0000		1 0000		00001	7.74
Pathology	1.10	1.0000	1.0558	1.0000	1.1124	1.0000	1.0000	1.0000		1.0000	1.29
Pharmacy	11.66	-	1.0558	1.0000	1.1124	1.0000	·	1.0000		1.0000	13.70
Cardiovascular	96.0	-	1.0558	1.0000	1.1124	1.0000	·	1.0000		1.0000	1.13
PT/OT/ST	2.48	- ·	1.0558	1.0000	1.1124	1.0000		1.0000		1.0000	2.92
Psychiatric	0.43		1.0558	1.0000	1.1124	1.0000		1.0000		1.0000	0.51
Substance Abuse Other	16.84		1.0558	1.0000	1.1124			1.0000	1.0000	1.0000	0.03 19.78
	\$67.89			-				-	-)) -	\$79.74
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$0.56	1.0000	1.0882	1.0000	1.0000	1.0000	1.0000	1.0000	¢-	1.0000	\$0.61
Office Visits	7.83	1.0000	1.0609	1.0000	1.0000	1.0000		1.0000		1.0000	8.31
Preventive Mealcine Matemity	c/.n		1.3083	00001	00001						0.98 1 1 1
rviater III.y Certified Midwife				1.0000							
PT/OT/ST	0.63	1 0000	1.0630	1 0000	1 0000	1 0000	·	1.0000		1 0000	0.67
Psychiatric and Substance Abuse	6.76	~	1.0630	1.0000	1.0000	1.0000	-	1.0000		1.0000	7.19
Radiology and Pathology	2.74	<u> </u>	1.0549	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	2.90
Ambultance	10.06		1.0629	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	10./U F 7F
Non-Emergency Transportation	31.07		1 0630	0000	1 0000	1 0000	1 0000	1 0000		0000 1	33.02
Opioid Treatment Program	4.13	. –	1.0636	1.0000	1.0000	1.0000	1.0000	0.2354		1.0000	1.03
Federally Qualified and Rural Health Clinics	1.25	1.0000	1.1340	1.0000	1.0000	1.0000	1.0000	1.0000	v -	1.0000	1.42
Adult Medical Day Care	1.75	1.0000	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	1.86
Personal Care Durable Medical Fouriement	15.88		1.0030	1.0000	1.0000		- •				29.82 30.22
Applied Behavioral Analysis	0.31			1,0000					•		0.31
Other	21.26	1.0000	1.0508	1.0000	1.0000	1.0000	-	1.0000	1.0000	1.0000	22.34
	\$138.63										\$157.29
Prescription Drugs	000	0000	0000	0000 1	0.0005	0000	1000 0	1 0000		0000	000
Generic Scripts Single Source Brand	\$0.9Z			1.0000	0.9885		0.9894	00001	00001		09.0¢
Multi-Source Brand	0.08		1,0000	1,0000	1.0498	1.0000		1 0000		1,0000	0.08
Specialty	00'0	-	1.0000	1.0000	1.0000	1.0000		1.0000		1.0000	00.0
Hepatitis C	0.00	1.0000	1.0000	1.0000	1.0000	1.0000	0.9894	1.0000	1.0000	1.0000	0.00
Oliei	\$1.14	0000-1	0000-1	0000-1	0.900	0000-1		0000-1		0000-1	\$1.12 \$1.12
Community Mental Health Center	•										
Case Management	\$0.52	1.0500	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	\$0.59
Long Term Support Service	0.99	1.0500	1.0629	1.0000	1.0000	1.0000		1.0000		1.0000	1.10
Parial Hospital Developherany	0.00	1.0500	1.0000	1.0000	1.0000	1.0000		1.0000	1.0000	1.0000	0.01
Evidence Based Practice	0.01	1.0500	1.0000	1.0000	1.0000	1.0000		1.0000		1.0000	0.01
Medication Management	0.04	1.0500	1.0000	1.0000	1.0000	1.0000		1.0000		1.0000	0.04
Emergency Service 24/7	00.00	1.0500	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.0
APR I P Supported Employment Services	0.26	1.0500	1.0000	1.0000	1.0000	1.0000		1.0000		1.0000	0.28
Other	1.24	1.0500	1.0630	1.0000	1.0000	1.0000		1.0000	1.0000	1.0000	1.38
	\$3.71										\$4.11
All Services	\$239.02				_						\$271.36

Eligibility Category: Newborn Kick Payment

	Per Capita					Managed Care					Projected Per
Benefits	Monthly Paid Cost	IBNR Adjustment	Reimbursement Adjustment	Utilization Trend Factors	Unit Cost Trend Factors		PDL Adjustment	Medicare OUD Adjustment	Stop Loss Adjustment	NEMT Adjustment	Capita Monthly Paid Cost
I Inpatient											
Medical	\$1,906.43	-	1.0509	1.0000	1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	\$2,015.42
Surgical	107.60	. .,	1.0852		1.0060	1.0000	1.0000			1.0000	117.48
Maternity Delivery Maternity Non Delivery					1,000					1.0000	
Well Newborn	358.42				1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	390.68
Psychiatric	1.45		•		1.0060	1.0000	1.0000		•	1.0000	1.65
Alcohol and Drug Abuse	00.0	.	1.0000	1.0000	1.0060	1.0000	1.0000			1.0000	000
	0.00 \$2 373 90	_	0000-1	0000-1	0000-1	0000-1	0000.1	-	0000.1	0000-1	\$2 525 23
Hospital Outpatient											
Emergency Room	\$30.10		1.0558	-	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	\$35.35
Surgery	0.84		1.0558	-	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	66'0
Radiology	5.04			-	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	5.92
Plathology	3.48	- · ·	1.0558		1.1124	1.0000	1.0000	1.0000		1.0000	4.09
Pritarinacy Cardiovascular	0.57	1.0000	-	1.0000	11124	1.0000		1,0000		1.0000	0.67
	02.0	- .			1 1124	1 0000	1 0000	1 0000		1 0000	0.87
Psychiatric	00.0	1.0000	1 0000	1.0000	1.1124	1.0000	1 0000	1.0000	1.0000	1.0000	00.0
Substance Abuse	00.0	-		ŗ	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Other	39.23	+		1.0000	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	46.08
	\$80.10										\$94.09
Professional and Other State Plan Services	0000	0000 1	0000 1	1 0000		1 0000		1 0000	1 0000	0000	00 04
Ambulatory Surgery Center	36 F2		1.000	1.0000					1.0000	1.0000	38.60
Preventive Medicine	105.52			1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	112.17
Maternity	00.0		1.0000		1.0000	1.0000	1.0000	1.0000		1.0000	00.0
Certified Midwife	0.16		1.0629		1.0000	1.0000	1.0000	1.0000	-	1.0000	0.17
PT/OT/ST	0.36	÷	1.0630	·	1.0000	1.0000	1.0000	1.0000	-	1.0000	0.39
Psychiatric and Substance Abuse	00.0	. .,			1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Radiology and Pathology Home Health and Private Duty Nursing	9/11.		1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		75,87
Ambulance	22 44	~		ţ	1.0000	1.0000	1 0000	1.0000	1.0000	1 0000	23.85
Non-Emergency Transportation	0.85	1.0000		·	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	06.0
Opioid Treatment Program	00.0	-		~	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	00.0
Federally Qualified and Rural Health Clinics	86.64	1.0000	1.1355	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	98.38 2.50
Adult Medical Day Care Dersonal Care	000		1.000	- •	1.0000	1.0000	1.0000	1.0000	1.0000	00001	0.00
Durable Medical Fourinment	9.10	1 0000	0.8983					1 0000	00001	1 0000	0.0 019
Applied Behavioral Analysis	0.00				1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Other	495.87	1.0000	1.0625	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	526.87
	\$793.58										\$847.92
Prescription Drugs	¢2,10				0010 1	1 0000	1000 0	1 0000	0000 1	0000	91 09
Generic Source Brand	0.50		1.0000		1.0503	1.0000	0.9894	1.0000		1.0000	0.52
Multi-Source Brand	0.03	-		ţ	1.0503	1.0000	0.9894			1.0000	0.03
Specialty	00.0	1.0000			1.6705	1.0000	0.9894		1.0000	1.0000	0.01
Hepatitis C	0.0	.			1.0000	1.0000	0.9894			1.0000	0.0
Outer	\$2.63	0000-1	0000-1	0000-1	1.0420	0000-1	0.3034	_	0000-	0000-1	\$2.72
Community Mental Health Center											
Case Management	\$0.00	,	1.0000	~	1.0000	1.0000	•	~	1.0000	1.0000	\$0.00
Long Term Support Service	0.0	÷.+	1.0000		1.0000	1.0000	1.0000		1.0000	1.0000	0.0
Partial Tospital Psychotherapy				1 0000		1 0000		1 0000	1 0000	1 0000	000
Evidence Based Practice	00.0	-			1.0000	1.0000		~	1.0000	1.0000	0.00
Medication Management	00.00	-			1.0000	1.0000		•		1.0000	0.00
Emergency Service 24/7	00.0	1.0500	1.0000	1.0000	1.0000	1,0000	1.0000	1.0000	1.0000	1,0000	0.0
APKTP Supported Employment Services											00.0
Other	0.00				1.0000	1.0000	`			1.0000	0.00
	\$0.00										\$0.00
All Services	\$3,250.21										\$3,469.95

Eligibility Category: Neonatal Abstinence Syndrome Kick Payment

	Per Capita					Managed Care					Projected Per
Benefits	Montrily Paid Cost	IBNR Adjustment	Adjustment	Factors	Factors	Adjustment	PDL Adjustment	Adjustment	Adjustment	Adjustment	Capita Montrily Paid Cost
Hospital Inpatient	CE 111 CO	0000	1050		0000	0000				0000	86 764 76
ivieuicai Surgicai	20.444,000		10000.1		1.0060	1.0000			1.0000	1.0000	00.0 00.0
Maternity Delivery	00.0	£.		1.0000	1.0060	1.0000				1.0000	0.00
Maternity Non-Delivery	0.00	1.0000			1.0060	1.0000		1.0000		1.0000	00.0
vvell Newborn Psvchiatric	00.0			1.0000	1.0060				1.0000		0.00
Alcohol and Drug Abuse	00.0	-	1.0000	,	1.0060	1.0000	1.0000			1.0000	00.0
Other	0.00	-		-	1.0060	1.0000				1.0000	0.00 #1 010
Hoomital Orthoptions	\$5,540.45										50,008,03
Hospital Outpatient Emergency Room	\$21.74	1.0000		1.0000	1.1124	1.0000				1.0000	\$25.53
Surgery	00.6	1.0000		. –	1.1124	1.0000	1.0000	1.0000		1.0000	10.58
Radiology	4.36	-		·	1.1124	1.0000	-		1.0000	1.0000	5.12
Pathology	5.00	1.0000		1.0000	1.1124	1.0000	1.0000			1.0000	5.87
Pharmacy	1.42	÷		·	1.1124	1.0000	·			1.0000	1.67
Cardiovascular	0.41	, ,			1.1124	1.0000				1.0000	0.48
P1/01/S1 Devehiatric	00.0	1.0000			4211.1 AC11.1					1.0000	1.04
Substance Abuse	00.0		1.0000	•	1.1124	1 0000	`			1.0000	0.00
Other	30.03			. –	1.1124	1.0000	. –	-		1.0000	35.27
	\$73.34										\$86.15
Professional and Other State Plan Services						0000		0000		0000	00 00
Ambulatory Surgery Center		0000 1	1.0000	0000 1	0000	0000	0000 F	0000	0000	1.0000	\$0.00 F0.06
OIIICE VISILS Preventive Medicine	136.40			·				·			00.0C
Maternity	00.0			·	1.0000	1.0000	•			1.0000	0.00
Certified Midwife	00.0	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	Ţ	1.0000	1.0000	0.00
PT/OT/ST	2.19			·	1.0000	1.0000	·	·		1.0000	2.33
Psychiatric and Substance Abuse	00.0	÷		·	1.0000	1.0000	•			1.0000	00.0
Radiology and Pathology	6.95	, ,	1.0558	1.0000	1.0000	1.0000	1.0000			1.0000	7.34
Ambulance Ambulance	31.35		-							1.0000	33.33
Non-Emergency Transportation	2.09				1.0000	1.0000				1.0000	2.22
Opioid Treatment Program	00.00	-		-	1.0000	1.0000	¢-			1.0000	0.00
Federally Qualified and Rural Health Clinics	104.83	£		-	1.0000	1.0000	•	•		1.0000	119.55
Adult Medical Day Care	0.0	1.0000	1.0000		1.0000	1.0000			1.0000	1.0000	0.00
Personal Care				0000 1				0000		0000	0.00
Durable Medical Equipment Applied Rehavioral Apalysis	0.00	- +						- 、		1.0000	000
Other	617.61	-	1.0627	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	656.32
	\$1,058.99										\$1,127.13
Prescription Drugs	10 10				0070 7	0000 1				0000	еп п.
Generic Scripts Sindle-Source Brand	10 0	1 0000			1.0503		0.9894				0.05
Multi-Source Brand	00.0			·	1.0503	1.0000				1.0000	0.00
Specialty	00.0	-		1.0000	1.6705	1.0000		1.0000	1.0000	1.0000	0.00
Hepatitis C	0.0	1.0000	1.0000		1.0000	1.0000	0.9894			1.0000	0.0
Outed	\$5.39	-	0000-1	-	1.0440	0000-1				0000-1	\$5.56
Community Mental Health Center											
Case Management	\$0.00				1.0000	1.0000				1.0000	\$0.00 0
Long I erm Support Service Dartial Hosnital				1.0000	1.0000	1.0000	1.0000		1.0000	1.0000	0.00
Psychotherapy	00.0	- . -	-		1.0000	1 0000	,			1.0000	00.0
Evidence Based Practice	00.00	1.0500		-	1.0000	1.0000	·	1.0000		1.0000	0.00
Medication Management	0.0	÷ ,			1.0000	1.0000	• •			1.0000	0.00
Emergency Service 24/7 APRTP	000				1 0000					00001	00.0
Supported Employment Services	00.0	1.0500	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Other	0.0	+			1.0000	1.0000			1.0000	1.0000	0.00
All Consissed	\$0.00				Ť	T					\$0.00
All Services	11.010,00										04.0.10,10

Eligibility Category: Maternity Kick Payment

	Per Capita Monthly Paid		Reimbursement	Utilization Trend	Unit Cost Trend	Managed Care Savings		Medicare OUD	Stop Loss	NEMT	Projected Per Capita Monthly
Benefits	Cost	IBNR Adjustment			Factors	Ŧ	PDL Adjustment	Adjustment	Adjustment	Adjustment	Paid Cost
Hospital Inpatient	11 1				0000			1 0000		0000	00 004
Nearcal	\$/4.38 0.00		6/01.1		1.0060		1.0000	1.0000	00001	1.0000	0000
Maternity Delivery	1.811.69			·	1 0060		·			1 0000	1.945.55
Maternity Non-Delivery	21.77	1.0000		1.0000	1.0060			1.0000		1.0000	25.28
Well Newborn	00.0	÷.	1.0000		1.0060					1.0000	0.00
Psychiatric	0.0	÷.,			1.0060					1.0000	0.0
Alcoriol and Drug Abuse Other	00 0	1 0000	1 0000		1.0060		1 0000	1.0000	0000		
	\$1,907.84									-	\$2,053.72
Hospital Outpatient											
Emergency Room	\$14.98		1.0558	-	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	\$17.60
Surgery	4.92	1.0000		1.0000	1.1124	1.0000	1.0000	1.0000		1.0000	5.78
Radiology	2.24	, .		- ·	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	2.64
Pathology	1.81	1.0000	1.0558	- ·	1.1124	1.0000	1.0000	1.0000		1.0000	2.13
Pharmacy	19.2			0000	4211.1	00001	1.0000	1.0000		1.0000	3.41
Cardiovascular DT/OT/ST	40.0		0330 F	- 1	4711.1 4717 1						cn'n
Pri/Or/Or Devokistris	1.0				4711.1 V 1171					1.0000	
r sycillatio Substance Ablise					1 1124	00001	1 0000	1 0000		1 0000	000
Other	19.81	1.0000	1.0558	1.0000	1.1124	1.0000	1,0000	1.0000	1.0000	1.0000	23.26
	\$46.88										\$55.06
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$0.00	, .			1.0000	1.0000	1.0000	1.0000		1.0000	\$0 ^{.00}
Office Visits	1.11	. ,		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.18
Preventive Mealcine Matemity	00.0 90.113			- •			0000 1	1.0000		00001	0.04 547 34
Materinity Certified Midwife	014-00 04-00	1 0000	1 0620		1,0000	1000		1 0000		1 0000	10.140 20.0
PT/OT/ST	0.01				1 0000	1 0000	1 0000	1 0000		1 0000	0.01
Psychiatric and Substance Abuse	00.0		1.0000	·	1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Radiology and Pathology	2.25	-		·	1.0000	1.0000	1.0000	1.0000		1.0000	2.39
Home Health and Private Duty Nursing	0.22	1.0000		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.23
Attibutatioe Non-Emergency Transportation	00.0		1,0000	·							0.0
Opioid Treatment Program	00.0			·	1.0000	1.0000	1.0000	1.0000		1.0000	00.0
Federally Qualified and Rural Health Clinics	2.59	-	1.1361	·	1.0000	1.0000	1.0000	1.0000		1.0000	2.94
Adult Medical Day Care	00.0	1.0000		-	1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Personal Care	00.0	, ,	1.0000		1.0000	1.0000	1.0000	1.0000		1.0000	00.0
Durable Medical Equipment	/0.0	- · ·			1.0000	1.0000	1.0000	1.0000		1.0000	/0.0
Other	13.30		1.0614	1.0000	1.0000	1.0000	1.0000	1 0000	1.0000	1.0000	14.12
	\$546.51										\$581.00
Prescription Drugs											
Generic Scripts	\$0.00	÷ ,			1.0000	1.0000				1.0000	\$0.00 0.00
Single-Source Brand Multi-Source Brand	000	1.0000			1.0000				1.0000	1.0000	00.0
Specialty	00.0	- . -			1.0000	1.0000				1.0000	00.0
Hepatitis C	00.0	Ť			1.0000	1.0000	1.0000			1.0000	0.00
Other	00.0	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	1.0000	1.0000	0.00
Community Montal Hoalth Contar	\$0.00										\$0 [.] 00
Community mental reacti center Case Management	\$0.00		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$0.00
Long Term Support Service	00.0	-			1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Partial Hospital	00.0	÷.,	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Psychotherapy	0.0	- ,			1.0000	1.0000	1.0000			1.0000	0.00
Evidence based Practice Medication Management	000		1.0000		1.0000		1.0000				00.0
Emergency Service 24/7	00.0	1.0500		·	1.0000	1.0000	1.0000		1.0000	1.0000	00'0
APRTP	00.0	÷.	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.00
Supported Employment Services Other	00.0	. .			1.0000	1.0000	1.0000			1.0000	00.0
222	\$0.00						0000-	0000-	-		\$0.00
All Services	\$2,501.22										\$2,689.78
	ш										

Eligibility Category: Severe/Persistent Mental Illness - Non-Dual

	Per Capita Monthly Paid		Reimbursement	Utilization Trend	Unit Cost Trend	Managed Care Savings		Medicare OUD	Stop Loss	NEMT	Projected Per Capita MonthIv
Benefits	Cost	IBNR Adjustment	Adjustment	Factors	Factors	Ŧ	PDL Adjustment	Adjustment	Adjustment	Adjustment	Paid Cost
Hospital Inpatient	¢60.67		Ŧ		1 0060	0 0876					¢53.06
Suraical	24.36		1.0596	1.0000	1.0060	0.9875	1.0000	1.0000	1.0000	1.0000	25.64
Maternity Delivery	00.0		·	1.0000	1.0060	0.9875	1.0000	1.0000		1.0000	00'0
Maternity Non-Delivery	0.88	1.0000	1.1482	1.0000	1.0060	0.9875	1.0000	1.0000	1.0000	1.0000	1.00
Well Newborn	0.00			1.0000	1.0060	0.9875	1.0000	1.0000		1.0000	0.00
Alcohol and Drug Abuse	1.90		·	1.0000	1.0060	0.9875	1.0000	1.0000	1.0000	1.0000	2.00
Other	00.00			1.0000	1.0060	0.9875	1.0000	1.0000		1.0000	0.00
	\$125.26										\$137.07
Hospital Outpatient											
Emergency Room	\$85.15	1.0000	1.0558	1.0000	1.1124	0.9875	1.0000	1.0000	1.0000	1.0000	\$98.76
Surgery	17.90			1.0000	1.1124	0.9875	1.0000	1.0000		1.0000	20.77
	72.27	1.0000	1.0558	1.0000	1.1124	0.98/5	1.0000	1.0000	1.0000	1.0000	58.67 14.00
Pharmood					4711.1 4714 1	0.90/0					11.02
r nannacy Cardiovascular	0.4				11124	0.9875	1 0000				* -1- 7 -1-
DT/DT/ST	6.35		·	1 0000	1 1124	0.9875			00001		7.36
Psychiatric	1.86			1.0000	1.1124	0.9875	1.0000	1.0000		1.0000	2.16
Substance Abuse	0.24			1.0000	1.1124	0.9875	1.0000	1.0000	1.0000	1.0000	0.28
Other	23.24		·	1.0000	1.1124	0.9875	1.0000	1.0000		1.0000	26.96
	\$183.87										\$213.27
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$2.10		1.1268	1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	\$2.33
Office Visits	29.76			1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	31.18
Materia Medicine	50.0 20.0				0000	0.98/5			0000 F		0.11
	12.1					0.90/0	1.0000				000
	0.00	1.0000				0.9875					0.00
Psychiatric and Substance Abuse	21.93		1 1020	1 0000	1 0000	0.9875	1 0000	1 0000	1 0000	1 0000	23.86
Radiology and Pathology	15.28		·	1 0000	1.0000	0.9875	1 0000	1 0000		1.0000	15.96
Home Health and Private Duty Nursing	14.04			1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	14.74
Ambulance	10.67		·	1.0000	1.0000	0.9875	1.0000	1.0000		1.0000	11.20
Non-Emergency Transportation	44.92	1.0000		1.0000	1.0000	0.9875	1.0000	1.0000		1.0000	47.16
Opioid Treatment Program	9.76			1.0000	1.0000	0.9875	1.0000	1.0000		1.0000	10.25
Federally Qualified and Kural Health Clinics	20.02		1.1344	0000	0000	0.98/5	0000 1	0000	0000		74.77
Adult Medical Day Care Dereanal Care	0.30					0.96/0	1.0000				1.5.0
Durable Medical Equipment	18.02	1.0000	0.9008	1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	16.03
Applied Behavioral Analysis	00.00		· -	1.0000	1.0000	0.9875	1.0000	1.0000		1.0000	00.0
Other	57.04	1.0000	-	1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	58.00
	\$252.95										\$263.47
Prescription Drugs	100 01 100 01		Ţ	0000	20000	0.0076	1 0627		0000 1	0000	ac 7014
Generic Scripts Single Source Brand	103.01 ¢				1 1463	0.90/5	12001 0020 0		1.0000		07. 727 23
Multi-Source Brand	44.07	1.0000		1.0000	1.1463	0.9875	0.4284	1.0000		1.0000	21.37
Specialty	208.95		v -	1.0000	1.3906	0.9875	0.9856	1.0000		1.0000	282.78
Hepatitis C	25.36	1.0000	1.0000	1.0000	0.5020	0.9875	0.9788	1.0000	1.0000	1.0000	12.30
Other	0.01 *****	1.0000		1.0000	0.9397	0.98/0	0.9489	1.0000	1.0000	1.0000	0.01 ¢are 07
Community Mental Health Center	en onde										16 0000
Case Management	\$298.14	1.0500	·	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	\$332.76
Long Term Support Service	276.75			1.0000	1.0000	1.0000	1.0000	1.0000	Ţ	1.0000	332.66
Partial Hospital	7.86	1.0500	1.0629	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	8.77
Psychotherapy	68.63			1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	76.56
Evidence Based Practice	6.96			1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	11.1
Integration Management	4. IJ						1.0000		1.0000		4.01 2.06
APRTP	4.85		·	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	5.42
Supported Employment Services	25.55		·	1.0000	1.0000	1.0000	1.0000	1.0000	.	1.0000	28.52
Other	33.42			1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	37.30
	\$728.97										\$837.33
All Services	\$1,099.14										\$Z, IU/ . IZ

Eligibility Category: Severe/Persistent Mental Illness - Dual

	Per Capita					Managed Care					Projected Per
Benefits	Monthly Paid Cost	IBNR Adjustment	Reimbursement Adjustment	Utilization Trend Factors	Unit Cost Trend Factors		PDL Adjustment	Medicare OUD Adjustment	Stop Loss Adjustment	NEMT Adjustment	Capita Monthly Paid Cost
Hospital Inpatient											
Medical Survival	50.01 6 00				1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	\$21.90 7.45
Carigical Maternity Delivery	00.0		·		1.0060	1.0000	1.0000	1.0000		1.0000	00.0
Maternity Non-Delivery	0.12		1.1482	1.0000	1.0060	1.0000	1.0000	1.0000		1.0000	0.14
Well Newborn	00.0		·	-	1.0060	1.0000	1.0000	1.0000		1.0000	0.0
Psychiatric	23.67				1.0060	1.0000	1.0000	1.0000	1.0000	1.0000	27.91
	00.0	1.0000	·		1.0060	1.0000	1 0000	1.0000	1.0000	1.0000	0.00
	\$52.52										\$58.61
Hospital Outpatient											
Emergency Room	\$34.27	1.0000	(- ,	- 1	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	\$40.25
Surgery	9.36		-	- •	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	10.99
Radiology Dethology	1.41		•		1.1124						9.28
Pharmacv	8.26		- (-		1.1124	1 0000	1.0000	1.0000		1.0000	02.1
Cardiovascular	1.09		, t-	. –	1.1124	1.0000	1.0000	1.0000		1.0000	1.28
PT/OT/ST	3.12		r-	-	1.1124	1.0000	1.0000	1.0000		1.0000	3.67
Psychiatric	1.06		<i>(</i>	-	1.1124	1.0000	1.0000	1.0000		1.0000	1.24
Substance Abuse	0.16		1.0558		1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	0.18
Otter	\$85.16	-			1.1124	0000-I	0000-1	0000-1	0000-I	nnnn-i	\$100.03
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$0.82		•	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$0.92
Office Visits	12.95		-	1.0000	·	1.0000	1.0000	1.0000		1.0000	13.74
Preventive Medicine	0.94		1.1476	1.0000	·	1.0000	1.0000	1.0000		1.0000	1.08
Maternity	0.21		- •	1.0000		1.0000	1.0000	1.0000		1.0000	0.22
Certified Intawre	0.61										0.00
Psychiatric and Substance Abuse	41.61		1.1020	1 0000	1.0000	1.0000	1.0000	1 0000	1.0000	1,0000	45.85
Radiology and Pathology	3.41		-	¢-	·	1.0000	1.0000	1.0000	1.0000	1.0000	3.61
Home Health and Private Duty Nursing	3.83		-	~	·	1.0000	1.0000	1.0000	1.0000	1.0000	4.07
Ambulance	6.02					1.0000	1.0000	1.0000	1.0000	1.0000	6.40 42.00
Non-Entregency Transportation Onioid Treatment Program	10.14				·			0.2309			1 22
Federally Qualified and Rural Health Clinics	1.03				. –	1.0000	1.0000	1.0000		1.0000	1.17
Adult Medical Day Care	0.75		-	-	-	1.0000	1.0000	1.0000	-	1.0000	0.79
Personal Care	8.52			- ·	1.0000	1.0000	1.0000	1.0000		1.0000	9.05
Durable Medical Equipment	9.45	1.0000	2.1/46	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	20.55
Other	23.73				1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	24.43
	\$160.22										\$177.73
Prescription Drugs											:
Generic Scripts Sinda-Source Brand	\$1.22		1.0000	1.0000	0.9885	1.0000	0.9894	1.0000	1.0000	1.0000	\$1.19 0.00
Multi-Source Brand	0.05		·	,		1.0000	0.9894	1 0000		1 0000	0.05
Specialty	00.00		1.0000	,		1.0000	0.9894	1.0000		1.0000	0.00
Hepatitis C	0.00	1.0000		1.0000	1.0000	1.0000	0.9894	1.0000	1.0000	1.0000	0.00
Other	\$1.36		÷	0000-L		00001	0.9894	0000-1		nnnn-L	0.00 \$1.34
Community Mental Health Center											
Case Management	\$308.54	1.0500		<i>(</i> - ·	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$344.36
Long Lerm Support Service	424.45 76.83				1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	510.20 20.05
Psychotherapy	29.32		·		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	32.71
Evidence Based Practice	7.31				1.0000	1.0000	1.0000	1.0000		1.0000	8.16
Medication Management	6.22			-	1.0000	1.0000	1.0000	1.0000		1.0000	6.94
Emergency Service 24/7	2.89				1.0000	1.0000	1.0000	1.0000		1.0000	3.23
APR I P Supported Employment Services	35.83		1.0629		00001	00001	0000 1		00001		66.98 39.99
Other	13.59	1.0500			1.0000	1.0000	1.0000	1.0000		1.0000	15.16
	\$862.47										\$999.04
All Services	\$1,161./4										\$1,336.75

Eligibility Category: Severe Mental Illness - Non-Dual

	:										
Donofile	Monthly Paid	IPND Adimet	Reimbursement	Utilization Trend	Unit Cost Trend	Manageo care Savings Adiuctmont	DDI Adiretment	Medicare OUD	Stop Loss	NEMT	Capita Monthly Daid Coct
Hospital Inpatient	COSt		Aujustillerit	LACIUIS	Lactors	Aujustiment		Adjustifiellt	Aujustinent	Hujusuine	
Medical	\$76.65	1.0000	1.0555	1.0000	1.0060	0.9875	1.0000	1.0000	1.0000	1.0000	\$80.37
Surgical	34.03		-	1.0000	1.0060	0.9875	1.0000	1.0000	•	1.0000	36.12
Maternity Delivery	00.0		(-)	1.0000	1.0060	0.9875	1.0000	1.0000		1.0000	0.00
Maternity Non-Delivery Wall Newborn	00.0	1.0000	1.1108	1.0000	1.0060	0.98/5 0.9875	1.0000	1.000		1.0000	1.33
Psychiatric	53.97			1.0000	1.0060	0.9875	1.0000	1.0000	1.0000	1.0000	62.30
Alcohol and Drug Abuse	12.68			1.0000	1.0060	0.9875	1.0000	1.0000		1.0000	13.48
Other	0.00 ¢170 E2	1.0000	1.000	1.0000	1.0060	C/86.0	1.0000	1.0000	1.0000	1.0000	0.00 #402 64
Hoenital Outnationt	00.0/10										10.0510
Emergency Room	\$104.36	1.0000	1.0558	1.0000	1.1124	0.9875	1.0000	1.0000	1.0000	1.0000	\$121.05
Surgery	12.56		1.0558	1.0000	1 1124	0.9875	1.0000	1.0000	1.0000	1.0000	14.57
Radiology	20.21		1.0558	1.0000	1.1124	0.9875	1.0000	1.0000	·	1.0000	23.44
Pathology	11.11	1.0000	1.0558	1.0000	1.1124	0.9875	1.0000	1.0000	,	1.0000	12.89
Pharmacy	5.09		1.0558	1.0000	1.1124	0.9875	1.0000	1.0000		1.0000	5.90
Cardiovascular	1.36		1.0558	1.0000	1.1124	0.9875	1.0000	1.0000	-	1.0000	1.57
PI/OI/SI	0.44	1.0000	1.0550	1.0000	1.1124	0.98/5	1.0000	1.0000		1.0000	7.65
rsycriaαric Substance Δhise	92.2		1.0330		1 1124	0.96/0	1.0000		1.0000	1.0000	2077 0 4 0
Other	20.91		1.0558	1.0000	1.1124	0.9875	1.0000	1.0000		1.0000	24.25
	\$184.68										\$214.20
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$2.22		1.2081	1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	\$2.65
Office Visits Dravantive Medicine	31.17	1.0000	1.0609	1.0000	1.0000	0.98/0	1.0000	1.0000	1.0000	1.0000	50.25 98 9
Maternity	3.56		1 0630	1 0000	1 0000	0.9875		1 0000		00001	3.74
Certified Midwife	0.01		1.0629	1.0000	1.0000	0.9875	-	1.0000	-	1.0000	0.01
PT/OT/ST	2.33	1.0000	1.0630	1.0000	1.0000	0.9875	-	1.0000		1.0000	2.44
Psychiatric and Substance Abuse	70.55		1.0788	1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	75.16
Radiology and Pathology	23.57		1.0597	1.0000	1.0000	0.9875	1.0000	1.0000		1.0000	24.66
Ambiilance	10.70		1.0629			0.96/0	1.0000	1.0000			0.80 13.36
Non-Emergency Transportation	45.49		1.0630	1.0000	1 0000	0.9875	1.0000	1 0000		1.0000	47.75
Opioid Treatment Program	18.27		1.0636	1.0000	1.0000	0.9875	1.0000	1.0000		1.0000	19.19
Federally Qualified and Rural Health Clinics	26.00		1.1339	1.0000	1.0000	0.9875	1.0000	1.0000		1.0000	29.11
Adult Medical Day Care	0.0		1.0000	1.0000	1.0000	0.9875	1.0000	1.0000		1.0000	0.00
Personal Care Durable Medical Equipment	12.05		1.U024 0.0516			0.9875					10.00
Applied Behavioral Analysis	00.0		1.0000	1.0000	1.0000	0.9875	1.0000	1.0000	•	1,0000	0.00
Other	65.61		1.0306	1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	66.77
2 - -	\$334.10										\$350.87
Prescription Drugs	¢67.67				0 0307	0 0876	1 0627	1 0000			¢66.00
Single-Source Brand	99.46		1.0000	1.0000	1.1463	0.9875	0.9329	1.0000		1.0000	105.03
Multi-Source Brand	28.64		1.0000	1.0000	1.1463	0.9875	0.4284	1.0000		1.0000	13.89
Specialty	92.37		1.0000	1.0000	1.3906	0.9875	0.9856	1.0000	-	1.0000	125.02
Departus C	23.37	1.0000		1.0000	0.5020	0.98/5	0.9/88	1.0000	1.0000	1.0000	11.34
	\$311.42		0000-		000.0	0.00.0	0000	0000-			\$321.30
Community Mental Health Center											
Case Management	\$197.31		1.0629	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	\$220.22
Long Term Support Service Partial Hosnital	/7.60				1.0000			1.0000	-	1 0000	01.02
Psychotherapy	62.21		1.0624	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	69.40
Evidence Based Practice	1.47		1.0624	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	1.64
Medication Management	0.75		1.0628	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.84
Emergency Service 24/7	0.74	1.0500	1.0613 1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.83 8 76
Supported Employment Services	7.78		1.0630	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	8.68
Other	28.02	1.0	1.0629	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	31.27
-	\$361.39										\$403.44
All Services	\$1,3/U.12										\$1,400.41

Eligibility Category: Severe Mental Illness - Dual

	Per Capita Monthly Doid		Doimbureomont	fili-ofion Trond	Init Cost Trond	Managed Care		Medicens OID	Stor Loce	NEMT	Projected Per
Benefits	Cost	IBNR Adjustment	Adjustment	Factors	Factors	t	PDL Adjustment	Adjustment	Adjustment	Adjustment	Paid Cost
Hospital Inpatient	¢16 72		Ţ	0000 1	1 0000	0000	0000 1	0000		0000	¢ 17 77
Interical Surgical	84.96	1.0000	1.0687	1.0000	1.0060	1.0000		1.0000	1.0000	1.0000	91.34
Maternity Delivery	00.00		·	ţ	1.0060	1.0000	·	-		1.0000	0.00
Maternity Non-Delivery	1.60	1.0000	1.1108		1.0060	1.0000	, ·		1.0000	1.0000	1.78
well newborn Psychiatric	14.56		·		1 0060	1.0000				1 0000	17.02
Alcohol and Drug Abuse	1.08		1.0696	1.0000	1.0060	1.0000	1.0000			1.0000	1.16
Other	00.00	1.0000	1.0000		1.0060	1.0000	·	1.0000	1.0000	1.0000	00.0
	\$118.94										\$129.08
Hospital Outpatient	40E 60					0000	·	•		0000	0 11 OF
Emergency Room Surgery	7 16		1.0338	1.0000	11124		1.0000		1.0000		00.14¢ 14.8
Cargory Radiology	7.76				1 1124	1 0000				00001	61.6
Pathology	2.18				1 1124	1 0000			1.0000	1.0000	2.56
Pharmacy	2.85			·	1.1124	1.0000			-	1.0000	3.35
Cardiovascular	0.52			1.0000	1.1124	1.0000	·		-	1.0000	0.61
PT/OT/ST	3.07				1.1124	1.0000	1.0000	1.0000		1.0000	3.61
Psychiatric	0.35		•	•	1.1124	1.0000				1.0000	0.41
Substance Abuse	0.20	1.0000		1.0000	1.1124	1.0000	1.0000		1.0000	1.0000	0.23
Outer	\$75.06	0000.1		0000-1	1.1124	0000-I	-			0000-1	\$88.16
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$0.43	-	1.2081	1.0000	1.0000	1.0000	¢-			1.0000	\$0.52
Office Visits	13.75	1.0000	1.0609	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	14.59
Preventive Medicine	1.07		·		1.0000	1.0000	-			1.0000	1.24
Maternity	0.36				1.0000	1.0000	- ·	-		1.0000	0.38
Certified Miawite	0.00				1.0000	0000	- •	1.0000		00001	0.00
P I/O I/S I Psychiatric and Substance Abuse	00.00 27.75		·	·	1.0000						0./0
Radiology and Pathology	4.51				1.0000	1 0000				1.0000	4.77
Home Health and Private Duty Nursing	2.79	1.0000	1.0629	1.0000	1.0000	1.0000	1.0000	-	1.0000	1.0000	2.97
Ambulance	6.14			·	1.0000	1.0000	-		-	1.0000	6.52
Non-Emergency Transportation	50.85 15 67		1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	54.05
Opioid Heatment Program Federally Origified and Bural Heath Clinics	19.01			1.0000							0.04 0.01
Adult Medical Day Care	0.93	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.93
Personal Care	4.42			1.0000	1.0000	1.0000				1.0000	4.70
Durable Medical Equipment	11.35		-	1.0000	1.0000	1.0000	-	-		1.0000	21.44
Applied Behavioral Analysis	00.00		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Uther	\$192.47	0000-L	1.0300	nnnn L	nnnn L	0000-L				00001	19.56 \$201 82
Prescription Drugs	71.70 ÷				Î						20.1020
Generic Scripts	\$0.82		•		0.9885	1.0000				1.0000	\$0.80
Single-Source Brand	0.15	1.0000			1.0498	1.0000				1.0000	0.16
Multi-Source Brand	0.16				1.0498	1.0000				1.0000	0.17
Speciality Henatitis C							0.9694			1 0000	
Other	00.00				0.9885	1.0000				1.0000	0.00
	\$1.14										\$1.13
Community Mental Health Center	¢108 EE	1 0500	1 0000	0000 1	0000	0000 1			T	0000 1	0710 36
Case Management Lond Term Support Service		1 0500	- •		1 0000		1.0000		1.0000		00.91 24 100 72
Partial Hospital	4.57		·	1.0000	1.0000	1.0000	·	·	-	1.0000	5.10
Psychotherapy	31.50		·		1.0000	1.0000	·	·		1.0000	35.14
Evidence Based Practice	2.67		1.0624	•	1.0000	1.0000	·			1.0000	2.98
Medication Management	1.39				1.0000	1.0000				1.0000	1.55
APRTP	6.10		1.0630	1.0000	1.0000	1.0000	1.0000	1,0000	1.0000	1,0000	6.81
Supported Employment Services	23.27		·		1.0000	1.0000	·			1.0000	25.97
Other	10.77	1.0500	1.0629	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	12.02
All Services	\$3/6.55			Ť	T		T			T	\$420.48 \$840.67
				-							0.0100

Eligibility Category: Low Utilizer - Non-Dual

	Per Capita					Managed Care					Projected Per
Benefits	Monthly Paid Cost	IBNR Adjustment	Reimbursement Adjustment	Utilization Trend Factors	Unit Cost Trend Factors	Savings Adjustment	PDL Adjustment	Medicare OUD Adjustment	Stop Loss Adjustment	NEMT Adjustment	Capita Monthly Paid Cost
Hospital Inpatient											
Medical	\$16.14	1.0000	1.0475	1.0000	1.0060	0.9875		1.0000	1.0000	1.0000	\$16.79
Surgical	25.26	1.0000	1.0426	1.0000	1.0060	0.9875	1.0000	1.0000	1.0000	1.0000	26.16 6.00
Maternity Delivery	00.0	1.0000	1.0000	1.0000	1.0060	0.98/5		1.0000		1.0000	00.0
Maternity Non-Jeilvery Well Newhorn	000				1.0060	C/86.0	1.0000	1.0000	1.0000	1.0000	000
Psychiatric	14.10	-	1.3767	1.0000	1.0060	0.9875				1.0000	19.28
Alcohol and Drug Abuse	0.00	÷.	1.0000	1.0000	1.0060	0.9875	1.0000	·		1.0000	0.00
Other	0.00	1.0000	1.0000	1.0000	1.0060	0.9875	1.0000	1.0000	1.0000	1.0000	00.00
	\$55.49										\$62.24
Hospital Outpatient						1200 0		0000	0000	0000	
Emergency Koom	440.44		1.0530 b	1.0000	4711.1 4777	6/96/0 22000	1.0000	1.0000	1.0000	1.0000	CS./C¢
DadioParty	10.00		1.0330		4711.1 1 1124	0.90/0	- •				20'61
Dethology	10.07 07 8		1.0330		4111.1 11124	0.96/0					10.10
Pharmacv	19.08	1 0000	1 0558	1 0000	1 1124	0.9875				00001	22 13
Cardiovascular	186	1 0000	1 0558	1 0000	1 1124	0.9875	1 0000	1 0000		1 0000	2 16
PT/OT/ST	4.28	1 0000	1 0558	1.0000	1 1124	0.9875				1 0000	4.96
Psychiatric	0.06	1.0000	1.0558	1.0000	1 1124	0.9875	1 0000		1 0000	1.0000	0.07
Substance Abuse	0.00	1.0000	1.0000	1.0000	1.1124	0.9875		-	1.0000	1.0000	0.00
Other	17.12	1.0000	1.0558	1.0000	1.1124	0.9875	1.0000	1.0000	1.0000	1.0000	19.85
	\$138.48										\$160.62
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$5.51	1.0000	1.1160	1.0000	1.0000	0.9875	1.0000	1.0000		1.0000	\$6.07
Office Visits	23.14	1.0000	1.0609	1.0000	1.0000	0.9875	1.0000	1.0000	·	1.0000	24.24
Preventive Medicine	5.81	1.0000	1.1242	1.0000	1.0000	0.9875		1.0000		1.0000	6.45
Maternity	1.81	1.0000	1.0630	1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	1.90
Certified Midwife	0.00	1.0000	1.0000	1.0000	1.0000	0.98/5		1.0000	1.0000	1.0000	0.00
PI/OI/SI	3.17	0000	1.0630	0000	0000	C/98/0		1.0000	00001	0000	3.33
Psychiatric and Substatice Abuse	20.02 12 63		1.0311			0.90/0	- •				91.02
Home Health and Private Duty Nursing	41.81	1.0000	1.0628	1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	43.88
Ambulance	5.31	1 0000	1 0630	1 0000	1 0000	0.9875		1.0000	1 0000	1 0000	5.57
Non-Emergency Transportation	42.46	1.0000	1.0630	1.0000	1.0000	0.9875	-	1.0000	1.0000	1.0000	44.57
Opioid Treatment Program	16.42	1.0000	1.0636	1.0000	1.0000	0.9875	<i>(</i>	1.0000	1.0000	1.0000	17.25
Federally Qualified and Rural Health Clinics	18.95	1.0000	1.1354	1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	21.25
Adult Medical Day Care	2.60	1.0000	1.0630	1.0000	1.0000	0.9875	-	1.0000	1.0000	1.0000	7.98
Personal Care	10.73	1.0000	1.0624	1.0000	1.0000	0.9875	1.0000	1.0000	1.0000	1.0000	11.26
Durable Medical Equipment	13.62	1.0000	0.9725	1.0000	1.0000	0.98/5	1.0000	1.0000	1.0000	1.0000	13.08
Applied Benavioral Analysis	0.00		1.0000	0000	0000	6/98/0 0 0 075	0000	1.0000	0000 1		0.00
	\$297.26	0000-				0.00.0		0000-1			\$316.64
Prescription Drugs											
Generic Scripts	\$101.50	1.0000	1.0000	1.0000	0.9397	0.9875		1.0000		1.0000	\$99.16
Single-Source Brand	230.47	1.0000	1.0000	1.0000	1.1463	0.9875		1.0000	1.0000	1.0000	243.39
Multi-Source Brand	16.99	1.0000	1.0000	1.0000	1.1463	0.98/5		1.0000		1.0000	8.24
Specially Henafitis C	14.80		1.0000		0.5020	0.96/5		1.0000			7 18
Other	00.0	1.0000	1.0000	1.0000	0.9397	0.9875	0.9489	1.0000	·	1.0000	00.0
	\$502.13										\$545.24
Community Mental Health Center											
Case Management	\$217.63	1.0500	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$242.90
Long Term Support Service Dartial Hosnital	21.13	1.050	1.0029	1.0000	1.0000		1.0000	1.0000	1.0000	1.0000	66.05 UU U
Psychotherapy	30.04	1.0500	1.0628	1.0000	1.0000	1.0000		1.0000	1.0000	1.0000	33.52
Evidence Based Practice	0.08	1.0500	1.0629	1.0000	1.0000	1.0000	-	1.0000	1.0000	1.0000	0.08
Medication Management	3.70	1.0500	1.0631	1.0000	1.0000	1.0000	~	1.0000	1.0000	1.0000	4.13
Emergency Service 24/7	3.22	1.0500	1.0631	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	3.60
APRTP	4.22	1.0500	1.0630	1.0000	1.0000	1.0000	- ·	1.0000	1.0000	1.0000	4.71
Supported Employment Services	71.C 78.14	1 0500	1.1091		1.0000		1.0000	1.0000	0000 1	00001	0.02 31 41
0.00	\$319.93		0000-								\$357 32
All Services	\$1,313.30										\$1,442.06

Eligibility Category: Low Utilizer - Dual

	Per Canita					Managed Care					Projected Per
Benefits	Monthly Paid Cost	BNR Adiustment	Reimbursement Adiustment	Utilization Trend Factors	Unit Cost Trend Factors		PDL Adiustment	Medicare OUD Adiustment	Stop Loss Adiustment	NEMT Adiustment	Capita Monthly Paid Cost
Hospital Inpatient											
Medical	\$19.10	,	1.0475	1.0000	1.0060	1.0000		1.0000		1.0000	\$20.12 0.12
Surgical Matemity Delivery	3.5/	1.0000	1.0426	1.0000	1.0060	1.0000		1.0000	1.0000	1.0000	3.75
Maternity Non-Delivery	00.0		1.0000	1.0000	1.0060	1.0000		1.0000		1.0000	0.0
Well Newborn	00.00	-	1.0000	1.0000	1.0060	1.0000		1.0000		1.0000	0.00
Psychiatric	1.34		1.3767	1.0000	1.0060	1.0000		1.0000		1.0000	1.85
Alconol and Drug Abuse Other	00.0	1.0000	1.0000	1.0000	1.0060	1.0000				1.0000	1900
	\$24.62										\$26.34
Hospital Outpatient											
Emergency Room	\$16.74	÷.,	1.0558	1.0000	1.1124	1.0000	1.0000	1.0000		1.0000	\$19.66
Surgery	7.75	- · ·	1.0558	1.0000	1.1124	1.0000	1.0000	1.0000		1.0000	9.10
Radiology	6.54 ac 1		1.0558	1.0000	1.1124	1.0000	1.0000	1.0000		1.0000	/.68
Pharmacv	5.10		1.0558	1.0000	1.1124	1.0000	1.0000	1.0000		1.0000	5.99
Cardiovascular	1.43	-	1.0558	1.0000	1.1124	1.0000	1.0000	1.0000		1.0000	1.68
PT/OT/ST	2.05	1.0000	1.0558	1.0000	1.1124	1.0000	1.0000	1.0000		1.0000	2.41
Psychiatric	0.76	-	1.0558	1.0000	1.1124	1.0000	1.0000	1.0000		1.0000	0.89
Substance Abuse	0.00	1.0000	1.0000	1.0000	1.1124	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
	\$57.15		0000-	0000-		0000-		0000-1		0000-1	\$67.12
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$0.44		1.1160	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	\$0.49
Office Visits	11.74	1.0000	1.0609	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	12.45
Preventive mealcine Maternity	700	- +	1.1242					1 0000	1.0000		0.05
Certified Midwife	00.0		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	00.0
PT/OT/ST	0.82		1.0630	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.87
Psychiatric and Substance Abuse	8.16		1.0911	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	8.91
Radiology and Pathology	2.54	1.0000	1.0555	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	2.68
Ambulance	2.48		1.0630	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	2.64
Non-Emergency Transportation	29.46	-	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	31.31
Opioid Treatment Program	4.72	,	1.0636	1.0000	1.0000	1.0000	1.0000	0.2461		1.0000	1.23
Federally Qualified and Rural Health Clinics	0.86	. .	1.1354	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.98
Adult Medical Day Care Personal Care	1.40		1.0630	1.0000	1.0000	1.0000		1.0000		1.0000	0.43
Durable Medical Equipment	15.83	~	1.7998	1 0000	1.0000	1.0000	1 0000	1.0000		1.0000	28.49
Applied Behavioral Analysis	00.0	~	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	00.0
Other	24.50 #117 E1	1.0000	1.1149	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	27.31
Description Duran	99.211\$										\$129.16
Prescription Drugs Generic Scripts	\$111	~	1 0000	1 0000	0 9885	1 0000	0 9894	1 0000		1 0000	\$1.09
Single-Source Brand	0.48		1.0000	1.0000	1.0498	1.0000	0.9894	1.0000		1.0000	0.50
Multi-Source Brand	00.0	÷ .	1.0000	1.0000	1.0498	1.0000	0.9894	1.0000		1.0000	00.0
Specialty	0.00	, ,	1.0000	1.0000	1.0000	1.0000	0.9894	1.0000		1.0000	0.0
Departus C	00 0		1.0000	1.0000	0.9885	1.0000	0.9894	1.0000	1.0000	1.0000	000
	\$1.60		0000-	0000-	00000	0000-	0000				\$1.59
Community Mental Health Center		•									00 0004
Case Management Long Term Support Service	\$187.46	1.0500	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$209.22 24.85
Partial Hospital	1.01		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	1.07
Psychotherapy	14.11	-	1.0628	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	15.74
Evidence Based Practice	0.11	÷ ·	1.0629	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.12
Medication Management	3.81	÷ ,	1.0631	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	4.25
Emergency Service 24/7 APRTP	26.0 2.62		1.0031	1 0000	1 0000	00001		1 0000		00001	3.26
Supported Employment Services	17.46	1.0500	1.1091	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	20.34
Other	11.21	-	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	12.51
All Continue	\$260.39										\$291.39 ¢646.64
	010010										0.0100

Eligibility Category: Serious Emotionally Disturbed Child

						and Canada					Ductorford Day
Ranofite	Monthly Paid	IBNP Adjustment	Reimbursement	Utilization Trend	Unit Cost Trend		DDI Adiretment	Medicare OUD	Stop Loss	NEMT Adjustment	Capita Monthly Daid Cost
Hospital Inpatient	2021		עממצווומווי	1 40003	1 400013		ו הר שמומווונוווי	valasment	ununenteu	Valasiiisiii	1 414 0031
Medical	\$3.90		1.0624	1.0000	1.0060	0.9938	1.0000	1.0000	1.0000	1.0000	\$4.15
Surgical	2.54		1.0664	1.0000	1.0060	0.9938	1.0000	1.0000	1.0000	1.0000	2.71
Maternity Delivery	00.00			1.0000	1.0060	0.9938	1.0000		·	1.0000	00'0
Maternity Non-Delivery	0.05	1.0000		1.0000	1.0060	0.9938	1.0000	1.0000		1.0000	0.04
	0.00			0000	nonn I	0.9938	0000 1		0000 1		0.00
Psychiatric Alcohol and Drug Abuse	10.65		1.0432	1 0000	1 0060	0.9938	1 0000			1 0000	0.72
Other	00 0			1 0000	1 0060	0.9938	1 0000			1 0000	00 0
	\$53.99										\$56.57
Hospital Outpatient											
Emergency Room	\$21.30	1.0000	1.0558	1.0000	1.1124	0.9938	1.0000	1.0000	1.0000	1.0000	\$24.86
Surgery	3.90			1.0000	1.1124	0.9938	1.0000	1.0000	1.0000	1.0000	4.55
Radiology	4.30	1.0000	1.0558	1.0000	1.1124	0.9938	1.0000	1.0000	1.0000	1.0000	5.02
Pathology	2.44			1.0000	1.1124	0.9938	1.0000	1.0000	1.0000	1.0000	2.84
Pharmacy	0.71			1.0000	1.1124	0.9938	1.0000	1.0000		1.0000	0.82
Cardiovascular	0.28		•	1.0000	1.1124	0.9938	1.0000	1.0000		1.0000	0.33
PI/01/SI	2.35			1.0000	1.1124	0.9938	1.0000	1.0000		1.0000	2.74
Psychiatric	60.7	0000	8660 I	00001	1.1124	0.9938	00001	0000.1	0000 1	0000 1	2.43
Substance Abuse	0.00		1.0000	1.0000	1.1124	0.9938	1.0000	1.0000		1.0000	0.00
Outed	274.63		0000.1	0000-1	1.1124	0,000,0	0000-1	0000-1	0000-1	0000-I	\$52.00
Professional and Other State Plan Services											20100
Ambulatory Surgery Center	\$0.73	1.0000	0.7671	1.0000	1.0000	0.9938	1.0000	1.0000	1.0000	1.0000	\$0.55
Office Visits	13.70		1.0605	1.0000	1.0000	0.9938	1_0000	1.0000	1.0000	1.0000	14.43
Preventive Medicine	4.29			1.0000	1.0000	0.9938	1.0000	1.0000		1.0000	4.66
Maternity	0.06			1.0000	1.0000	0.9938	1.0000	1.0000		1.0000	0.07
Certified Midwife	00.00			1.0000	1.0000	0.9938	1.0000	1.0000		1.0000	00.0
PT/OT/ST	5.15		1.0630	1.0000	1.0000	0.9938	1.0000	1.0000		1.0000	5.44
Psychiatric and Substance Abuse	4.60		1.0566	1.0000	1.0000	0.9938	1.0000	1.0000		1.0000	4.83
Radiology and Pathology	2.53		1.0577	1.0000	1.0000	0.9938	1.0000	1.0000		1.0000	2.66
Home Health and Private Duty Nursing	3.10	1.0000	1.0627	1.0000	1.0000	0.9938	1.0000	1.0000	1.0000	1.0000	3.27
	04.7		1.0030	0000 1	0000	0.000				00001	+0.7
Doining Treatment Program		1 0000		1,0000	00001	0.9590					
Federally Qualified and Rural Health Clinics	9.32		1 1341	1 0000	1 0000	0.9938	1.0000	1.0000	1.0000	1.0000	10.50
Adult Medical Day Care	00.00		1.0000	1.0000	1.0000	0.9938	1.0000	1.0000	1.0000	1.0000	00.0
Personal Care	00.0		1.0000	1.0000	1.0000	0.9938	1.0000	1.0000	~	1.0000	00'0
Durable Medical Equipment	5.54		1.0117	1.0000	1.0000	0.9938	1.0000	1.0000	-	1.0000	5.57
Applied Behavioral Analysis	2.82	, .	1.0630	3.3169	1.0000	0.9938	1.0000	1.0000	.	1.0000	9.87
Other	16.62 ¢73.70	1.0000	1.0624	1.0000	1.0000	0.9938	1.0000	1.0000	1.0000	1.0000	11.55 30 A02
Dracorintion Druce	01.010										404 40
Generic Scrints	\$39 55	1 0000	1 0000		0 7915	0 9938	0 9907	1 0000	1 0000		\$30.82
Single-Source Brand	27.43			1.0000	0.9966	0.9938	0.9729	1.0000		1.0000	26.43
Multi-Source Brand	2.45			1.0000	0.9966	0.9938	0.9367	1.0000		1.0000	2.28
Specialty	20.27			1.0000	2.0057	0.9938	0.9886	1.0000		1.0000	39.94
Hepatitis C	00.0	-		1.0000	1.0000	0.9938	0.9894	1.0000		1.0000	0.00
Other	0.01	1.0000	1.0000	1.0000	0.7915	0.9938	0.9829	1.0000	1.0000	1.0000	0.01
Contraction (11 - 11 - 11 - 11 - 11 - 11 - 11 - 11	\$89.72										\$99.48
Configuration of the second seco	\$260.59	1.0500	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$290.84
Long Term Support Service	110.59			1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	123.42
Partial Hospital	00.0		1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	-	1.0000	00'0
Psychotherapy	121.68			1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	129.79
Evidence Based Practice	0.01		1.0627	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.01
Medication Management	0.10		- •			1.0000	1.0000				0.14
APRTP	0.22	1.0500	1.0630	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.25
Supported Employment Services	0.11		-	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.13
Other	16.45 #F00.07	1.0500	1.0628	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	18.35
All Carvinae	19.90c¢				Ī						\$565 24
All Set VICes	>>>		-	-	-	-			-		11.0000

Eligibility Category: Granite Advantage - Medically Frail

	:					<u>.</u>					
D644	Monthly Paid	PND Addition	Reimbursement	Utilization Trend	Unit	Manageo care Savings	DDI Adiment	Medicare OUD	Stop Loss	NEMT	Capita Monthly
Hosnital Invatiant	COSE	IDNK Adjustment	Adjustment	ractors	Factors	Adjustment	PDL Agjustment	Aajustment	Adjustment	Adjustment	Paid Cost
Medical	\$59.31	1 0000	1 0551	1 0000	1 0060	0.9860	1 0000	1 0000	1 0000	1 0000	\$62.07
Suraical	45.97			1 0000	1.0060	0.9860		1.0000	1.0000	1.0000	48.52
Maternity Delivery	00.00		1.0000	1.0000	1.0060	0.9860	-	1.0000	1.0000	1.0000	00.0
Maternity Non-Delivery	0.18	1.0000	·	1.0000	1.0060	0.9860	-	1.0000		1.0000	0.19
Well Newborn	0.0		1.0000	1.0000	1.0060	0.9860	- ·	1.0000		1.0000	0.0
Psychiatric	12.44			1.0000	1.0060	0.9860	- 1	1.0000	1.0000	1.0000	13.//
Alconol and Urug Abuse	00.0			1.0000	1.0060	0.9860	1.0000	1,0000		1.0000	19.68
	\$136.40		0000-1	0000-1	0000-1	0000-0	0000-1	0000-1	0000-1	0000-1	\$144.22
Hospital Outpatient											
Emergency Room	\$58.91		1.0558	1.0000	1.1124	0.9860	-	1.0000	1.0000	1.0000	\$68.23
Surgery	16.93		1.0558	1 0000	1 1124	0.9860	1.0000	1.0000	1.0000	1 0000	19.61
Radiology	25.98		1.0558	-	1 1124	0.9860	~	1.0000		1 0000	30.09
Pathology	7 78	1 0000	1.0558	1 0000	1 1124	0.9860	1.0000	1.0000	1.0000	1.0000	9.01
Pharmacy	49.91		1.0558	-	1.1124	0.9860	-	1.0000		1.0000	57.80
Cardiovascular	3.06		1.0558	1.0000	1.1124	0.9860	-	1.0000		1.0000	3.54
PT/OT/ST	6.28		1.0558	-	1.1124	0.9860	-	1.0000		1.0000	7.27
Psychiatric	1.57		1.0558	-	1.1124	0.9860	-	1.0000		1.0000	1.82
Substance Abuse	1.02		1.0558	1.0000	1.1124	0.9860	1.0000	1.0000	1.0000	1.0000	1.18
Other	25.72		1.0558	-	1.1124	0.9860	1.0000	1.0000		1.0000	29.79
	\$197.17										\$228.35
Professional and Other State Plan Services											
Ambulatory Surgery Center	\$2.04		1.0519	1.0000	1.0000	0.9860	1.0000	1.0000	1.0000	1.0000	\$2.12
Office Visits	28.61		1.0612	1.0000	1.0000	0.9860	1.0000	1.0000	-	1.0000	29.93
Preventive Medicine	3.03	1.0000	1.1436	1.0000	1.0000	0.9860	-	1.0000	1.0000	1.0000	3.42
Maternity	0.32		1.0627	1.0000	1.0000	0.9860	-	1.0000	1.0000	1.0000	0.34
Certified Midwife	00.00		1.0630	1.0000	1.0000	0.9860	-	1.0000	1.0000	1.0000	0.01
PT/OT/ST	3.00	1.0000	1.0645	1.0000	1.0000	0.9860	-	1.0000	1.0000	1.0000	3.15
Psychiatric and Substance Abuse	71.04		1.0720	1.0000	1.0000	0.9860	-	1.0000	1.0000	1.0000	75.09
Radiology and Pathology	25.60		1.0683	1.0000	1.0000	0.9860	-	1.0000		1.0000	26.97
Home Health and Private Duty Nursing	13.16	1.0000	1.0629	1.0000	1.0000	0.9860	1.0000	1.0000	1.0000	1.0000	13.79
Ambulance	71.9		1.0630	1, 2000	0000.1	0.9860		0000.1		0000.1	0.42
Non-Emergency Iransportation	19.15		1.0000		nnnn'l	0.9890			nnnn I		39.03 27.60
Coloral Ireament Program	20.17 F0.12					0.9000	- •			0000 1	00.77
Adult Modical Dou Correction Design Chillics	10-17		1.1340			0.9000				0000	24.44 0.10
Personal Care	0.57		1 0624	1 0000	1 0000	0.9860		1 0000		1 0000	0.59
Durable Medical Equipment	16.69		1 2965			0.9860	1 0000	1 0000		00001	2134
Applied Behavioral Analysis	00.0		1.0630	1.0000	1.0000	0.9860	1.0000	1.0000		1.0000	0.00
Other	57.84		1.0375	1.0000	1.0000	0.9860	1.0000	1.0000	1.0000	1.0000	59.17
	\$309.49										\$329.27
Prescription Drugs											
Generic Scripts	\$42.58		-	·		0.9860		1.0000		1.0000	\$56.30
Single-Source Brand	10.08		0000		8000	0.986.0			0000	0000	66.08
Nutr-Jource Dialia			- (0.980.0					00 271
Hepatitis C	40.82	1.0000	- (-	1.0000	0.3632	0.9860	0.9760	1.0000	1.0000	1.0000	14.27
Other	0.02			1 0000		0.9860		1.0000		1.0000	0.02
	\$303.04										\$308.90
Community Mental Health Center											
Case Management	\$1.09		1.0625	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$1.22
Long Lerm Support Service	1.84			1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	50.2 00.0
Failiar Fospital Devrhotherany	0.00		1.0000								0.00
Evidence Based Practice	0.03							1 0000	00001	1 0000	0.03
Medication Management	0.05			1.0000	1.0000	1.0000		1.0000	1.0000	1 0000	0.05
Emergency Service 24/7	0.01		. –	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.01
APRTP	1.15		1.0630	1.0000	1.0000	1.0000	-	1.0000		1.0000	1.28
Supported Employment Services	0.01	1.0500		1.0000	1.0000	1.0000	- ·	1.0000	1.0000	1.0000	0.02
Other	¢1278		-	nnnn I.	nnnn-L	nnnn-I.		nnnn I.	nnnn I.	nnnn-L	¢11.05
All Services	\$958.88										\$1 024 99
											00-1-40'I @

Eligibility Category: Granite Advantage - Non-Medically Frail

	Per Capita					Managed Care					Projected Per
Benefits	Monthly Paid Cost	IBNR Adjustment	Reimbursement Adjustment	Utilization Trend Factors	Unit Cost Trend Factors	Savings Adjustment	PDL Adjustment	Medicare OUD Adjustment	Stop Loss Adjustment	NEMT Adjustment	Capita Monthly Paid Cost
Hospital Inpatient											
Medical	\$20.53		1.0561	1.0000	1.0060	0.9923	1.0000	1.0000		1.0000	\$21.64
Surgical Metomity Dolivost	13./9		1.0592	1.0000	1.0060	0.9923	1.0000	1.0000	1.0000	1.0000	14.58
Maternity Don-Delivery	0.20	1.0000	1.0458	1.0000	1.0060	0.9923	1.0000	1.0000	1.0000	1.0000	0.21
Well Newborn	0.01		1.0833	1.0000	1.0060	0.9923	1.0000	1.0000		1.0000	0.01
Psychiatric	3.91		1.1524	1.0000	1.0060	0.9923	1.0000	1.0000		1.0000	4.49
Alcohol and Drug Abuse Other	4.99 0.00		1.06/3	1.0000	1.0060	0.9923	1.0000	1.0000		1.0000	5.31
	\$43.42	-		-		2422	-	0000	-	-	\$46.25
Hospital Outpatient											
Emergency Room	\$31.39	1.0000	1.0558	1.0000	1.1096	0.9923	1.0000	1.0000	1.0000	1.0000	\$36.50
Surgery	10.56		1.0558	1.0000	1.1096	0.9923	1.0000	1.0000		1.0000	12.28
Radiology	13.28		1.0558	1.0000	1.1096	0.9923	1.0000	1.0000		1.0000	15.44 5.22
Pharmacv	14.43		1.0330	1.0000	1 1090	0.9923		1.0000	1.0000		07.22 16.49
Cardiovascular	1.57		1.0558	1.0000	1.1096	0.9923	1.0000	1.0000		1.0000	1.83
PT/OT/ST	2.73		1.0558	1.0000	1.1096	0.9923	1.0000	1.0000		1.0000	3.17
Psychiatric	0.46		1.0558	1.0000	1.1096	0.9923	1.0000	1.0000		1.0000	0.54
Substance Abuse	0.37		1.0558	1.0000	1.1096	0.9923	1.0000	1.0000		1.0000	0.43
Other	\$01.34 \$01.38	nnnn I.	ACCU.	nnnn I.	GRUT. I	0.9923	nnnn'l	1.0000		Innnn I.	\$106.24
Professional and Other State Plan Services	00-00										12.0010
Ambulatory Surgery Center	\$0.88		1.0495	1.0000	1.0000	0.9923	1.0000	1.0000	1.0000	1.0000	\$0.92
Office Visits	14.44		1.0608	1.0000	1.0000		1.0000	1.0000		1.0000	15.20
Preventive Medicine	3.11		1.1406	1.0000	1.0000		1.0000	1.0000		1.0000	3.52
Maternity	0.41		1.0626	1.0000	1.0000		1.0000	1.0000		1.0000	0.43
Certitited Midwite	0.02	1.0000	1.0606	1.0000	1.0000	0.9923	1.0000	1.0000	1.0000	1.0000	0.02
Psychiatric and Substance Abuse	24.20		1.0741	·	1.0000		1.0000	1.0000		1.0000	25.80
Radiology and Pathology	9.33		1.0866	·	1.0000	0.9923	1.0000	1.0000	1.0000	1.0000	10.06
Home Health and Private Duty Nursing	1.45		1.0629		1.0000		1.0000	1.0000	-	1.0000	1.53
Ambulance	2.19		1.0630	- 1	1.0000		1.0000	1.0000	1.0000	1.0000	2.31
Non-Emergency Transportation Onioid Treatment Program	9.00		1.0030		00001	0.9923		1.0000			01.10 6.44
Federally Qualified and Rural Health Clinics	10.52		1.1335	1.0000	1.0000	0.9923	1.0000	1.0000		1.0000	11.83
Adult Medical Day Care	0.03		1.0630	1.0000	1.0000	0.9923	1.0000	1.0000	1.0000	1.0000	0.03
Personal Care	00.0	. .	1.0000	1.0000	1.0000	0.9923	1.0000	1.0000		1.0000	0.00
Durable Medical Equipment	3.54		1.0063	1.0000	1.0000	0.9923	1.0000	1.0000		1.0000	3.54
Other	24.58	1.0000	1.0595	1.0000	1.0000	0.9923	1.0000	1.0000	1.0000	1.0000	25.85
	\$112.06										\$119.34
Prescription Drugs	00 010			0000		0000		0000		1	
Generic Scripts	\$18.66		1.0000	1.0000	1.19/1	0.0023	1.0611			1.0000	\$23.52 36 E4
Single-source Brand Multi-Source Brand	55.55 7 70	1.0000	1.0000	1.0000	1.1593	0.9923	0.4400	1.0000	1.0000	1.0000	3.90
Specialty	32.36		1.0000	1.0000	1.3776	0.9923	0.9801			1.0000	43.36
Hepatitis C	8.60	1.0000	1.0000	1.0000	0.7009	0.9923	0.9753	1.0000	1.0000	1.0000	5.83
Other	0.00 \$100.87		0000-1	0000.1	1.13/1	0.3323	0.3432			nnnn-1	\$113.16
Community Mental Health Center	200										
Case Management	\$0.26	1.0500	1.0608	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	\$0.29
Long Lerm Support Service	0.61		1.0633	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.68
Partial Tospital Psychotherapy	00.0	1 0500	1.0625	1 0000	1 0000			1 0000			0.00
Evidence Based Practice	00.0		1.0629	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	00.0
Medication Management	0.01		1.0593	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	0.01
Emergency Service 24/7	00.0		1.0454	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	0.00
Supported Employment Services	00.00	1.0500	1.0629	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	00.0
Other	1.30		1.0629	1.0000	1.0000	1.0000	1.0000	1.0000		1.0000	1.46
	\$3.24										\$3.61
	08.0024	_	_	_	_	_	_		_	_	\$200 DA

Eligibility Category	CY 2018 Encounter Member Months	CY 2018 Encounter Projected Per Capita Monthly Paid Cost	CY 2019 Encounter Member Months	CY 2019 Encounter Projected Per Capita Monthly Paid Cost	Blended Projected Per Capita Monthly Paid Cost	Opioid Adjustment	FFS Data Adjustment	Acuity Adjustment	Final Blended Projected Per Capita Monthly Paid Cost
Low Income Children - Age 0-11 Months	47,865	\$243.94	47,995	\$229.63	\$236.78	0.9765	0.9921	1.0000	\$229.38
Low Income Children - Age 1-18 Years	729,792	153.66	727,538	155.58	154.62	0666.0	0.9973	0.9959	153.41
Low Income Adults	124,392	443.40	133,084	455.33	449.56	0.9626	0.9949	0.9663	416.00
CHIP	154,555	150.50	165,409	159.25	155.02	1.0028	0.9985	0.9959	154.59
Foster Care / Adoption	20,976	305.43	23,159	308.01	306.78	0.9994	0.9980	0.9959	304.73
Severely Disabled Children	11,180	1,397.47	10,785	1,301.46	1,350.33	0.9999	0.9981	1.0000	1,347.59
Elderly and Disabled Adults - Age 19-64	76,828	1,165.78	73,573	1,164.18	1,165.00	1.0002	1.0000	0.9573	1,115.40
Elderly and Disabled Adults - Age 65+	12,505	918.05	12,753	949.22	933.79	0666.0	0.9971	0.9573	890.38
Dual Eligibles	166,049	286.22	172,429	271.36	278.65	0.9910	0.9947	1.0000	274.66
Newborn Kick Payment	2,296	3,475.28	2,767	3,469.95	3,472.37	1.0000	1.0101	1.0000	3,507.45
Neonatal Abstinence Syndrome Kick Payment	144	7,894.29	160	7,075.46	7,463.33	1.0000	1.0015	1.0000	7,474.75
Maternity Kick Payment	2,897	2,736.67	3,260	2,689.78	2,711.85	1.0000	0.9961	1.0000	2,701.32
	007 00						0000	0000	
Severe/Persistent Mental Illness - Non-Dual	23,493	\$2,064.64	28,956	\$2,107.12	\$2,088.09	1.0069	0000.1	1.0000	\$2,102.37
Severe/Persistent Mental Illness - Dual	22,217	1,431.66	23,082	1,336.75	1,383.30	0.9985	0.9998	1.0000	1,380.99
Severe Mental Illness - Non-Dual	17,089	1,486.94	29,203	1,483.41	1,484.72	0.9937	0.9994	1.0000	1,474.48
Severe Mental Illness - Dual	3,766	851.43	3,417	840.67	846.31	0.9984	1.0014	1.0000	846.14
Low Utilizer - Non-Dual	1,796	1,300.70	2,553	1,442.06	1,383.68	0.9966	1.0000	1.0000	1,378.94
Low Utilizer - Dual	2,080	600.26	2,242	515.61	556.34	0.9975	1.0000	1.0000	554.96
Serious Emotionally Disturbed Child	73,352	868.06	76,603	856.24	862.02	0.9999	0.9999	1.0000	861.79
Standard Subtotal1	1,487,934	\$380.44	1,532,782	\$394.62	\$387.63	0.9952	0.9982	0.9885	\$380.65
Granite Advantage - Medically Frail	71,345	\$953.81	84,734	\$1,024.99	\$992.45	0.9926	0.9982	0.9947	\$978.08
Granite Advantage - Non-Medically Frail	0	00.00	383,770	388.59	\$388.59	0.9795	1.0000	0.9663	367.79
GAHCP Subtotal2	71,345	\$953.81	468,504	\$503.69	\$563.18	0.9862	0.9991	0.9808	\$544.24
Total1	1,559,279	\$406.67	2,001,286	\$420.15	\$414.25	0.9933	0.9984	0.9869	\$405.45
¹ Member Months totals exclude kick payments. ² GAHCP Subtotal excludes behavioral health rate cells.									

New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Blended Projected Medical Cost Based on CY 2018 and CY 2019 Experience Appendix C

	inal Projected Benefit Cost	\$200.20 163.55 163.55 163.55 162.35 162.35 162.35 162.35 162.35 11116.47 111116.47 11116.47 11116.47 11116.47 11116.47 111116.47 111116.47 111116.47 111116.47 111116.47 111116.47 111116.47 111116.47 111116.47 111116.47 111116.47 111116.47 111116.47 111116.47 111116.47 1111116.47 1111116.47 111111111111111111111111111111111111	\$971.33 365.82
	Pharmacy Fin Savings B		(\$0.96) (0.39)
	SUD Residential Code H0010	600 600 600 600 600 600 600 600	\$1.02
	Remote Patient SUC Monitoring C.	000 000 000 000 000 000 000 000 000 00	\$0.03 0.00
	NEMT Changes	0.02 0.02 0.02 0.02 0.02 0.02 0.02 0.02	\$2.88
	Birthing Centers NE	80.51 0.00 0.00 0.00 0.00 0.00 0.00 0.00	\$0.03 0.00
		2002 0.159 0.159 0.159 0.159 0.199 0.198 0.200000000	\$2.74 1.09
	MCRT / ES Services Carve Additional Pricing Out Changes	80.00 80.000 80.000 80.000 80.000 80.00000000	(52.97)
	Genetic Testing	600 600 600 600 600 600 600 600	000
	Home Visiting G	4 5 5 5 5 5 5 5 5 5 5 5 5 5	\$0.00 0.06
	Community Residential Services	8000 8000	\$0.06 0.01
ng April 2023	aoston Children's Hospital	866 40 9.846 9.846 7.846 7.846 7.846 7.846 7.846 0.037 8.070 1,000 0.000 0.001 0.000 0.001 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.000000	\$0.36 0.27
lopment - PHE Ending April 2023	Transition of Mass General Pediatric and Tufts Children Iospital Claims Bc to BCH	(5.147) (5.147	(50.01)
Capitation Rate Deve sensit Add-Ons	State Owned Hospital Professional Services	80.00 0.17 0.17 0.18 0.18 0.18 0.18 0.18 0.18 0.18 0.18	\$0.20 0.04
Care Management C B	State Owned Hospital Stafing Capacity	80.00 (0.33) (0.	(\$0.56)
SFY 2023 Medicaid	Hampstead Repricing	80.00 81	\$1.00
	NHH Repricing	2000 2000 2000 2000 2000 2000 2000 200	\$0.00 0.01
			\$2.22 0.75
	P Payon Removed for Kick NHH Indusion for Payment Ages 21 to 64		(\$12.78) (4.03)
	ATECH Adjustment	800 800 800 800 800 800 800 800 800 800	20.00 0.00
	Behavioral Health Crisis Treatment Center (BHCTC)		(50.04)
	PMPM Add-On for White Mountain Behavioral Health Expanded Mmail FOUC Look-dig Crisis Treatment Health Services Adjustment Contex (FH-CTC)	0000 0000 0000 0000 0000 0000 0000 0000 0000	50.01 0.00
	PMPM Add-On for White Mountai Expanded Mental FQHC Look-41 Heakh Services Automent		00 0 0 00 0
	Projected Per Capita Monthly Paid Cost	8228-33 153.44 154.40 154.80 154.80 154.80 154.80 137.80 140.80 1	5978.08 367.79
	Efiglibility Cattagory	Lookhourer Challen - Ans D. 1 Month, Lookhourer Challen - Ans D. 1 Month, Carlo Carl	Granite Advantage - Medically Frail Granite Advantage - Non-Medically Frail

D-11 Months \$300.20 0.3925 1 1-18 Years \$300.20 0.3925 1 1-18 Years \$16.3.55 0.9925 1 1-18 Years \$16.3.55 0.9925 1 1-40e 15-4 \$11.18.47 0.9925 1 -Age 55+ \$1.18.47 0.9925 1 -Age 55+ \$1.118.47 0.9925 1 -Age 55+ \$21.18.1 0.9925 1 -Asse - Non-Dual \$2.167.15 0.9925 3 -Asse - Dual 1.492.00 0.9925 1	\$297.95 \$297.95 162.32 9.1% 413.22 9.1% 1584.68 1.584.68 1.584.68 1.584.68 1.584.68 1.584.68 1.584.68 1.584.68 1.584.68 1.584.68 1.584.68 1.584.68 1.584.68 1.584.68 1.584.68 1.584.58 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.500 1.	% \$30.27 % \$16.49 % 41.98 % 16.36 % 16.36 % 12.28 % 12.09 % 84.76 % 67.53	1.5% 1.5% 1.5% 1.5% 1.5% 1.5% 1.5%	\$5.00 55.00 5.2.72 5.33 5.33 5.33 5.33 0.00 5.33 0.00 5.33 0.00 14.50 0.00 14.50 0.00	\$11.82 \$14 14.43 3.30 7.70 13.28	2.0% 2.0% 2.0%				
\$300.20 0.9925 4 \$5.55 0.9925 16.3.55 0.9925 \$16.355 0.9925 16.2.31 0.9925 \$15.56 0.9925 1 19925 \$1,50.66 0.9925 1 1.9925 \$1,118,47 0.9925 1 1.184.7 \$1,118,47 0.9925 1 1.186.8 \$1,118,47 0.9925 1 1.9925 \$1,118,47 0.9925 1 1.186.37 \$1,118,47 0.9925 1 1.186.37 \$1,118,47 0.9925 1 1.186.37 0.9925 1 \$1,167,30 0.9925 10 0.9925 10 3.167.20 1.9925 3 3 3 1.422.00 0.9925 1 3					с	2.0% 2.0% 2.0%				
163.35 0.9925 416.35 0.9925 416.35 0.9925 162.31 0.9925 15.366 0.9925 1,118,47 0.9925 1,118,47 0.9925 1,118,47 0.9925 1,118,47 0.9925 281,131 0.9925 281,131 0.9925 281,131 0.9925 10,48,87 0.9925 10,44,87 0.9925 3,167,30 0.9925 3,167,30 0.9925 11,44,15 0.9925 3,167,30 0.9925 1,45,00 0.9925 3,167,30 0.9925 1,432,00 0.9925 1,432,00 0.9925 1,432,00 0.9925 1,432,00 0.9925 1,442,00 0.9925 1,442,00 0.9925		·				2.0% 2.0%	\$7.04	\$352.08	100%	\$352.08
416.35 0.9925 162.31 0.9925 150.18 0.9925 1118.47 0.9925 1118.47 0.9925 1118.47 0.9925 1118.47 0.9925 1118.47 0.9925 1118.47 0.9925 1118.47 0.9925 1118.47 0.9925 1118.47 0.9925 1118.47 0.9925 1118.47 0.9925 281.81 0.9925 3.167.30 0.9925 3.167.30 0.9925 15.415 0.9925 3.167.30 0.9925 15.42.55 0.9925 15.42.00 0.9925 1.432.00 0.9925 1.442.00 0.9925		·				2.0%	3.81	190.48	100%	190.48
162.31 0.9925 11.320.18 0.9925 11.1.320.18 0.9925 11.1.347 0.9925 11.1.3466 0.9925 11.1.3468 0.9925 11.1.3468.87 0.9925 13.3467.30 0.9925 13.3467.30 0.9925 13.3467.30 0.9925 13.3467.30 0.9925 13.3472.00 0.9925 11.442.00 0.9925						2.0%	9.73	486.29	100%	486.29
320.18 0.9925 1 1,596.66 0.9925 1 1,118.47 0.9925 1 1,118.47 0.9925 1 891.07 0.9925 1 891.07 0.9925 1 891.07 0.9925 1 891.07 0.9925 10 91.07 0.9925 10 3,167,30 0.9925 10 3,167,30 0.9925 3 55,164,15 0.9925 3 55,164,15 0.9925 3 1653.75 0.9925 2 1,492.00 0.9925 3		·					3.74	187.20	100%	187.20
1,566,56 0.9925 1 1,18,47 0.9925 1 391,07 0.9925 1 391,07 0.9925 1 391,07 0.9925 4 391,07 0.9925 10 31,67,30 0.9925 10 3,167,30 0.9925 3 5,167,30 0.9925 3 5,167,30 0.9925 3 1,432,00 0.9925 3 1,442,00 0.9925 1 1,442,00 0.9925 1						2.0%	7.41	370.51	100%	370.51
1,118,47 0.9925 1 891,07 0.9925 4 881,81 0.9925 4 281,81 0.9925 4 281,81 0.9925 4 10,488 0.9925 4 10,488 0.9925 10 2,167,30 0.9925 10 2,167,30 0.9925 10 2,167,30 0.9925 3 15,53,75 0.9925 2 1,492,00 0.9925 1 1,492,00 0.9925 1						2.0%	35.61	1,780.54	100%	1,780.54
ant 22,07 0,925 4 4,991,02 0,9925 4 4,991,02 0,9925 4 3,167,30 0,9925 10 3,167,30 0,9925 10 52,194,15 0,9925 3 1,482,00 0,9925 11 1,482,00 0,9925 11						2.0%	25.83	1,291.59	100%	1,291.59
281.81 0.9925 4 281.02 0.9925 4 10,468.87 0.9925 10 3,167.30 0.9925 3 \$167.30 0.9925 3 \$165.375 0.9925 1 \$1,653.75 0.9925 1 1,492.00 0.9925 1						2.0%	20.61	1,030.39	100%	1,030.39
ant 4,991,02 0,9925 4 4,991,02 0,9925 4 3,167,30 0,9925 10 52,194,15 0,9925 1 1,653,75 0,9925 1 1,422,00 0,9925 1 1						2.0%	6.44	321.91	100%	321.91
ant 10,468,87 0.9925 10 3,167,30 0.9925 10 \$2,194,15 0.9925 \$2 1,553,75 0.9925 \$2 1,422,00 0.9925 1		•				2.0%	106.41	5,320.36	100%	5,320.36
3,167,30 0.9925 3 \$2,194,15 0.9925 \$2 1,553.76 0.9925 1 1,492.00 0.9925 1		% 382.08	-			2.0%	223.19	11,159.68	100%	11,159.68
\$2,194,15 0.9925 \$2 1,553.75 0.9925 1 1,492.00 0.9925 1						2.0%	67.53	3,376.29	100%	3,376.29
1,653.75 0.9925 1 1,492.00 0.9925 1						2.0%	\$49.54	\$2,476.79	100%	\$2,476.79
1,492.00 0.9925 1	1,641.35 5.6%	% 98.75	1.5%	26.50 53.23	21.49	2.0%	37.58	1,878.89	100%	1,878.89
						2.0%	33.50	1,674.92	100%	1,674.92
						2.0%	22.11	1,105.42	100%	1,105.42
Low Utilizer - Non-Dual 1,386.99 0.9925 1,376.59						2.0%	31.00	1,549.86	100%	1,549.86
						2.0%	13.48	673.86	100%	673.86
Serious Emotionally Disturbed Child 987.53 0.9925 980.13						2.0%	22.84	1,142.02	100%	1,142.02
Granite Advantage - Medically Frail \$971.33 0.9925 \$964.04						2.0%	\$22.27	\$1,113.50	100%	\$1,113.50
0.9925	363.08 9.1%	% 36.88	1.5%	6.09	9.12	2.0%	8.47	423.65	100%	423.65

Appendix E New Hampshire Department of Heat Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Final Capitation Rate Development

Exhibit A1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development Comparison of Composite Rates with PHE Ending April 2023 to Amendment #8 SFY 2023 Capitation Rates Excluding Directed Payments	Exhibit A1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development nposite Rates with PHE Ending April 2023 to Amendment #8 SFY 202 Excluding Directed Payments	and Human Services itation Rate Developme to Amendment #8 SFY 2 ments	nt 023 Capitation Rates	
Eligibility Category	SFY 2023 Projected Member Months	Amendment #8 - SFY 2023 Rates	Amendment #9 - SFY 2023 Rates	Proposed Rate Change
		00 \$	600700	6 0
Low Income Children - Age U-11 Months Low Income Children - Age 1-18 Years	42,402 801,829	\$330.99 183.65	\$540.01 185.24	0.9%
Low Income Adults CHIP	219,693 263.326	477.91 181.09	471.57 183.83	-1.3%
Foster Care / Adoption	33,296	364.50	362.65	
Severely Disabled Children Elderly and Disabled Adults - Age 19-64	10,161 72,021	1,794.09	1,766.99 1,237.79	-1.5%
Elderly and Disabled Adults - Age 65+ Dual Eligibles	16,861 191.012	1,009.28 306.95	986.14 307.18	-2.3%
Newborn Kick Payment	2,736	4,721.98	5,320.36	
Maternity Kick Payment	2,423	2,875.73	3,376.29	17.4%
Severe/Persistent Mental Illness - Non-Dual	34,556	\$2,493.51	\$2,428.23	-2.6%
Severe/Persistent Mental Illness - Dual	24,758	1,858,45	1,802.65	-3.0%
Severe Mental Illness - Non-Dual Severe Mental Illness - Dual	3,988	1,670.97	1,051.17	-1.2%
Low Utilizer - Non-Dual	5,188 3.463	1,547.51 663 34	1,534.96 653 43	-0.8%
Serious Emotionally Disturbed Child	79,044	1,102.75	1,092.88	
Granite Advantage - Medically Frail	125,223	\$1,097.65	\$1,074.95	-2.1%
Granite Advantage - Non-Medically Frail Kick Davmant	949,176	419.51	414.34	-1.2%
PG 01 & 07		\$3,681.70	\$3,681.70	%0 [.] 0
PG 02		2,836.28 8 000 60	2,836.28 8 000 60	0.0%
PG 06		17,606.39	17,606.39	%0.0
Base Population Rate Cells	1,387,276	\$349.04	\$348.81	-0.1%
CHIP Population Rate Cell ¹	271,945	211.15	213.50	1.1%
Standard Behavioral Health Population Rate Cells ²	126,434	1,459.07	1,431.84	-1.9%
GAHCP Rate Cells	1,074,399	505.08	498.04	-1.4%
GAHCP Behavioral Health Population Rate Cells	43,059	2,054.51	2,015.13	-1.9%
Total ³	2,903,112	\$467.51	\$463.25	-0.9%
¹ CHIP population rate cell composite includes the CHIP and SED rate cells. ² Standard Behavioral Health Population Rate Cells composite excludes CHIP. ³ Member Months totals exclude kicks.	ate cells. udes CHIP.			

Exhibit A2 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development Comparison of Composite Rates with PHE Ending April 2023 to Amendment #8 SFY 2023 Capitation Rates Including Directed Payments	Exhibit A2 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development nposite Rates with PHE Ending April 2023 to Amendment #8 SFY 202 Including Directed Payments	and Human Services itation Rate Developme to Amendment #8 SFY 2 ments	nt 023 Capitation Rates	
Eligibility Category	SFY 2023 Projected Member Months	Amendment #8 - SFY 2023 Rates	Amendment #9 - SFY 2023 Rates	Proposed Rate Change
Low Income Children - Age 0-11 Months	42,402	\$350.35	\$352.08	
Low Income Children - Age 1-18 Years Low Income Adults	801,829 219 693	189.00 494.04	190.48 486.29	0.8% -1.6%
CHIP	263,326	185.02	187.20	
Foster Care / Adoption	33,296	374.07	370.51	•
Severely Disabled Children Elderly and Disabled Adults - Age 19-64	72,021	1,600.17	1,700.34	-1.5%
Elderly and Disabled Adults - Age 65+	16,861	1,059.06	1,030.39	
Uual Eligibles Newhorn Kick Davment	191,012 2 736	322.91	321.91	-0.3%
Neonatal Abstinence Syndrome Kick Payment	21	9,707.32	11,159.68	-
Maternity Kick Payment	2,423	2,875.73	3,376.29	17.4%
Severe/Persistent Mental Illness - Non-Dual	34,556	\$2,553.10	\$2,476.79	-3.0%
Severe/Persistent Mental Illness - Dual	24,758	1,939,34	1,878.89	
Severe Mental Illness - Non-Dual Severe Mental Illness - Dual	27,115	1,698.74 1 1 5 8 11	1,674.92	-1.4%
Low Utilizer - Non-Dual	5,188	1,567.60	1,549.86	
Low Utilizer - Dual	3,463	686.42	673.86	
Serious Emotionally Disturbed Child	/ 9,044	1,154.84	1,142.02	-1.1%
Granite Advantage - Medically Frail	125,223	\$1,150.99	\$1,113.50	
Granite Advantage - Non-Medically Frail	949,176	432.57	423.65	-2.1%
Kick Payment PG 01 & 07		\$3.681.70	\$3.681.70	0.0%
PG 02		2,836.28	2,836.28	
PG 09 PG 06		8,000.60 17,606.39	8,000.60 17,606.39	0.0%
Base Population Rate Cells	1,387,276	\$361.13	\$360.19	-0.3%
CHIP Population Rate Cell ¹	271,945	216.61	218.32	
Standard Behavioral Health Population Rate Cells ²	126,434	1,514.99	1,483.27	
GAHCP Rate Cells	1,074,399	522.83	510.76	
GAHCP Behavioral Health Population Rate Cells	43,059	2,097.00	2,050.34	
Total ³	2,903,112	\$483.43	\$476.60	-1.4%
¹ CHIP population rate cell composite includes the CHIP and SED rate cells. ² Standard Behavioral Health Population Rate Cells composite excludes CHIP. ³ Member Months totals exclude kicks.	ate cells. udes CHIP.			

.02

9/7/2023

Exhibit B

New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development - PHE Ending April 2023 Estimated Fiscal Impact - Assuming 50% FMAP for Standard, 65% FMAP for CHIP, and 90% FMAP for GAHCP

Eligibility Category	Estimated Member Months	Average Total Capitation Rate	Federal Capitation Rate Liability	Federal Capitation Total Cost Liability	State Capitation Rate Liability	State Capitation Total Cost Liability
Low Income Children - Age 0-11 Months	42,402	\$352.08	\$176.04	\$7,464,474	\$176.04	\$7,464,474
Low Income Children - Age 1-18 Years	801,829	190.48	95.24	76,366,226	95.24	76,366,226
Low Income Adults	219,693	486.29	243.15	53,417,366	243.15	53,417,366
CHIP	263,326	187.20	121.68	32,041,518	65.52	17,253,125
Foster Care / Adoption	33,296	370.51	185.26	6,168,271	185.26	6,168,271
Severely Disabled Children	10,161	1,780.54	890.27	9,046,324	890.27	9,046,324
Elderly and Disabled Adults - Age 19-64	72,021	1,291.59	645.80	46,510,800	645.80	46,510,800
Elderly and Disabled Adults - Age 65+	16,861	1,030.39	515.20	8,686,831	515.20	8,686,831
Dual Eligibles	191,012	321.91	160.96	30,744,283	160.96	30,744,283
Newborn Kick Payment	2,736	5,320.36	2,660.18	7,278,252	2,660.18	7,278,252
Neonatal Abstinence Syndrome Kick Payment	21	11,159.68	5,579.84	117,177	5,579.84	117,177
Maternity Kick Payment	2,423	3,376.29	1,981.77	4,801,338	1,394.52	3,378,569
Severe/Persistent Mental Illness - Non-Dual	34,556	\$2,476.79	\$1,796.50	\$62,079,030	\$680.29	\$23,507,741
Severe/Persistent Mental Illness - Dual	24,758	1,878.89	939.45	23,258,704	939.45	23,258,704
Severe Mental Illness - Non-Dual	27,115	1,674.92	1,312.04	35,576,007	362.88	9,839,356
Severe Mental Illness - Dual	3,988	1,105.42	552.71	2,204,172	552.71	2,204,172
Low Utilizer - Non-Dual	5,188	1,549.86	1,127.36	5,848,859	422.50	2,191,974
Low Utilizer - Dual	3,463	673.86	336.93	1,166,647	336.93	1,166,647
Serious Emotionally Disturbed Child	79,044	1,142.02	579.31	45,790,993	562.71	44,479,156
Granite Advantage - Medically Frail	125,223	\$1,113.50	\$1,002.15	\$125,491,885	\$111.35	\$13,943,543
Granite Advantage - Non-Medically Frail	949,176	423.65	381.29	361,906,516	42.37	40,211,835
Base Donutation Rate Calls	1 387 776	\$360 10	\$180 NG	\$740 830 576	¢180.00	\$740 830 576
CHIP Population Rate Cells	271 945	218 32	141 01	38 500 021	76.41	20,000,000
	105 101	70 00 7		00,000,021		00 767 050
Deriavioral Feature Opulation Nate Cells GAHCP Rate Cells	1 074 399	510.76	459.68	493,882,010	51 04 51 08	54,875,779
GAHCP Behavioral Health Rate Cells	43.059	2.050.34	1,845.30	79,456,859	205.03	8,828,540
Total	2,903,112	\$476.60	\$329.14	\$955,537,170	\$147.46	\$428,091,425

*CHIP population rate cell composite includes the CHIP and SED rate cells. **Standard Behavioral Health Population Rate Cells composite excludes CHIP. ***Member Months totals exclude kicks.

B	Exhibit C1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development Blended Projected BCH Medical Cost Based on Combined CY 2018 and CY 2019 Experience - PHE Ending April 2023	New Hampshi SFY 2023 Medicaid Medical Cost Base	Exhibit C1 re Department of Hea d Care Management (a on Combined CY 2	Exhibit C1 New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development Medical Cost Based on Combined CY 2018 and CY 2019 Experience	Services Sevelopment Experience - PHE	Ending April 2025		
Eligibility Category	CY 2018 Encounter Member Months	CY 2018 Encounter Projected Per Capita Monthly Paid Cost	CY 2019 Encounter Member Months	CY 2019 Encounter Projected Per Capita Monthly Paid Cost	Acuity Adjustment	Blended Projected Per Capita Monthly Paid Cost	Additional PMPM	Final Blended Projected Per Capita Monthly Paid Cost
	1100	cc o océ	11 005			ц СС Э́	C C C C C C C C C C C C C C C C C C C	C C C U H E
Low Income Children - Age 0-11 Months Low Income Children - Age 1-18 Years	729.792	\$20.33 5.36	727.538	\$46.78 8.04	0.9959	79.9 6.67	\$10.34 3.17	0.8000
Low Income Adults	124,392	0.08	133,084	0.58	0.9663	0.33	0.15	0.48
CHIP	154,555	4.71	165,409	5.93	0.9959	5.32	2.53	7.85
Foster Care / Adoption	20,976	5.39	23,159	21.53	0.9959	13.80	6.56	20.37
Severely Disabled Children	11,180	198.88	10,785	153.88	1.0000	176.79	84.05	260.83
Elderly and Disabled Adults - Age 19-64	76,828	2.58	73,573	3.57	0.9573	2.93	1.39	4.33
Elderly and Disabled Adults - Age 65+	12,505	00.0	12,753	00.0	0.9573	0.00	0.00	00.00
Dual Eligibles	166,049	0.07	172,429	0.20	1.0000	0.14	0.07	0.20
Newborn Kick Payment	2,296	623.40	2,767	561.89	1.0000	589.78	280.39	870.17
Neonatal Abstinence Syndrome Kick Payment	144	2,165.64	160	111.72	1.0000	1,084.63	515.65	1,600.28
Maternity Kick Payment	2,897	0.00	3,260	0.00	1.0000	00.00	00.00	00.0
Severe/Persistent Mental Illness - Non-Dual	23,493	\$0.06	28,956	\$0.13	1.0000	\$0.10	\$0.05	\$0.15
Severe/Persistent Mental Illness - Dual	22,217	00.0	23,082	0.59	1.0000	0.30	0.14	0.44
Severe Mental Illness - Non-Dual	17,089	0.20	29,203	0.33	1.0000	0.28	0.13	0.41
Severe Mental Illness - Dual	3,766	00.0	3,417	00.0	1.0000	0.00	00.00	00.00
Low Utilizer - Non-Dual	1,796	00.00	2,553	00.0	1.0000	00.0	00.00	00.00
Low Utilizer - Dual	2,080	0.00	2,242	00.0	1.0000	00.0	0.00	0.00
Serious Emotionally Disturbed Child	/3,352	9.51	/6,603	22.41	1.0000	16.10	(.65	23./5
Standard Subtotal ¹	1,487,934	\$7.39	1,532,782	\$9.80	0.9885	\$8.59	\$4.08	\$12.67
Granite Advantage - Medically Frail	71,345	0.38	84,734	0.12	0.9947	0.24	\$0.11	0.36
Granite Advantage - Non-Medically Frail	0	0.00	383,770	0.19	0.9663	0.19	0.09	0.27
GAHCP Subtotal ²	71,345	\$0.38	468,504	\$0.18	0.9808	\$0.20	\$0.10	\$0.30
Total ¹	1,559,279	\$7.07	2,001,286	\$7.55	0.9869	\$7.32	\$3.48	\$10.79
¹ Member Months totals exclude kicks.								
 GAHCP Subtotal excludes behavioral health rate cells. 								

Exhibit C2

New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rate Development Total Projected BCH Funding - PHE Ending April 2023

Eligibility Category	Projected BCH Funding Using Base Data	Total Additional BCH Funding	Total BCH Funding
Low Income Children - Age 0-11 Months	\$1,635,350	\$777,472	\$2,412,822
Low Income Children - Age 1-18 Years	5,345,215	2,541,203	7,886,417
Low Income Adults	71,507	33,996	105,502
CHIP	1,401,154	666,131	2,067,285
Foster Care / Adoption	459,610	218,506	678,116
Severely Disabled Children	1,796,382	854,030	2,650,412
Elderly and Disabled Adults - Age 19-64	211,297	100,454	311,751
Elderly and Disabled Adults - Age 65+	0	0	0
Dual Eligibles	26,437	12,569	39,006
Newborn Kick Payment	1,613,644	767,153	2,380,797
Neonatal Abstinence Syndrome Kick Payment	22,777	10,829	33,606
Maternity Kick Payment	0	0	0
Severe/Persistent Mental Illness - Non-Dual	3,430	1,631	5,061
Severe/Persistent Mental Illness - Dual	7,425	3,530	10,955
Severe Mental Illness - Non-Dual	7,586	3,607	11,193
Severe Mental Illness - Dual	0	0	0
Low Utilizer - Non-Dual	0	0	0
Low Utilizer - Dual	0	0	0
Serious Emotionally Disturbed Child	1,272,283	604,864	1,877,147
Standard Subtotal ¹	\$13,874,097	\$6,595,973	\$20,470,070
Granite Advantage - Medically Frail	30,237	14,375	44,613
Granite Advantage - Non-Medically Frail	175,759	83,559	259,317
GAHCP Subtotal ²	\$205,996	\$97,934	\$303,930
Total ¹	\$14,080,093	\$6,693,907	\$20,774,000

¹ Member Months totals exclude kicks.

² GAHCP Subtotal excludes behavioral health rate cells.

Exhibit D New Hampshire Department of Health and Human Services SFY 2023 Medicaid Care Management Capitation Rates Acuity Adjustment Factors January 2021 to June 2021 Actual To Expected

			Acuity
	Included Acuity	Actual to	Adjustment with
Eligibility Group	Adjustment	Expected	Full Difference
Children	-4.4%	1.5%	-2.8%
Adults	-1.1%	-7.2%	-8.2%
Disabled Adults	0.5%	-13.0%	-12.4%
BBH Adults	-1.7%	-4.9%	-6.5%
Disabled Children	-3.6%	12.4%	8.8%

SFY 2022 Actual To Expected

			Acuity
	Included Acuity	Actual to	Adjustment with
Eligibility Group	Adjustment	Expected	Full Difference
Children	-2.1%	3.3%	1.2%
Adults	-3.3%	-2.0%	-5.3%
Disabled Adults	-1.8%	-2.9%	-4.6%
BBH Adults	-2.1%	9.6%	7.5%
Disabled Children	%6 ^{.0-}	-7.7%	-8.6%

Acuity Adjustment

		• •	Average Acuity Adjustment
Eligibility Group	June 2021	SFY 2022	Dampened by 50%
Children	-2.8%	1.2%	-0.4%
Adults	-8.2%	-5.3%	-3.4%
Disabled Adults	-12.4%	-4.6%	-4.3%
BBH Adults	-6.5%	7.5%	%0 [.] 0
Disabled Children	8.8%	-8.6%	0.0%

	2	New Hampshir Aedicaid Care M	Exhibit E New Hampshire Department of Health and Human Services Medicaid Care Management Program - PHE Ending in April 2023 CMHC Funding	iit E ⁶ Health and Hu ram - PHE End unding	man Services ing in April 2023					
		CN	CMHC Service Costs	ts						
	SFY 2023					Behavioral Health Crisis				
	Projected Member				Expanded Mental Health	Treatment Center	MCRT / ES Services	CMHC Directed	Commuity Residential	
Rate Cell	Months	CY 2018	CY 2019	Blended	Services	(BHCTC)	Carve Out	Payment	Services	Total
Low Income Children - Age 0-11 Months	42,402	\$0.00	\$0 [.] 00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
Low Income Children - Age 1-18 Years	801,829	2.67	2.73	2.70	00.0	00.0	-0.58	0.00	00.0	2.12
Low Income Adults	219,693	4.84	5.20	5.03	00.0	-0.03	-1.02	0.00	0.01	3.98
CHIP	263,326	1.81	1.69	1.75	0.00	00.0	-0.45	00.0	00.0	1.30
Foster Care / Adoption	33,296	7.79	7.45	7.61	00.0	00.0	-1.55	00.00	0.00	6.06
Severely Disabled Children	10,161	4.71	6.72	5.70	00.0	00.0	-0.59	00.00	0.00	5.10
Elderly and Disabled Adults - Age 19-64	72,021	7.67	8.88	8.26	00.00	-0.06	-1.61	00.0	0.20	6.80
Elderly and Disabled Adults - Age 65+	16,861	1.47	0.91	1.18	00.0	00.0	-0.15	00.0	0.00	1.03
Dual Eligibles	191,012	3.71	4.11	3.91	00.0	-0.01	-0.82	00.00	0.04	3.12
Newborn Kick Payment	2,736	00.00	00.00	00.00	00.0	00.0	0.00	00.0	0.00	0.00
Neonatal Abstinence Syndrome Kick Payment	21	0.00	00.00	00.0	00.0	0.00	0.00	00.00	0.00	0.00
Maternity Kick Payment	2,423	0.00	0.00	0.00	0.00	0.00	0.00	0.00	00.0	0.00
Severe/Persistent Mental Illness - Non-Dual	15.089	\$868.56	\$837.33	\$852 <u>.</u> 56	\$69 <u>.</u> 38	-\$0.06	-\$10.14	\$22.97	\$6.66	\$941.37
Severe/Persistent Mental Illness - Dual	24,758	1,090,33	999.04	1,043,81	160.79	0 04	-13.95	53.23	27.43	1.271.29
Severe Mental Illness - Non-Dual	7,908	455.47	403.44	429.70	19.85	00.0	-4.33	6.57	0.13	451 93
Severe Mental Illness - Dual	3,988	470.36	420.48	446.63	64.51	00.0	-13.88	21.36	1.32	519.94
Low Utilizer - Non-Dual	2,239	358.04	357.32	357.65	17.41	00.0	-2.56	5.76	0.70	378.97
Low Utilizer - Dual	3,463	351.38	291.39	320.26	33.72	00.0	-3.37	11.16	0.01	361.79
Serious Emotionally Disturbed Child	77,608	572.78	563.15	567.87	97.69	00.0	-6.90	32.34	0.01	691.02
Standard Subtotal	1,785,655	\$54.50	\$52.24	\$53.35	\$7.38	\$0.01	\$1.28	\$2.44	\$0.45	\$62.35
Granite Advantage - Severe/Persistent Mental Illness	19,466	\$868.56	\$837.33	\$852.56	\$69.38	-\$0.06	-\$10.14	\$22.97	\$6.66	\$941.37
Granite Advantage - Severe Mental Illness	19,207	455.47	403.44	429.70	19.85	00.0	-4.33	6.57	0.13	451.93
Granite Advantage - Low Utilizer	2,949	358.04	357.32	357.65	17.41	00.0	-2.56	5.76	0.70	378.97
Granite Advantage - Serious Emotionally Disturbed Child	1,436	572.78	563.15	567.87	97.69	00.0	-6.90	32.34	0.01	691.02
Granite Advantage - Medically Frail	125,223	13.06	14.26	13.71	00.0	-0.04	-2.97	00.0	0.06	10.76
Granite Advantage - Non-Medically Frail ¹	949,176	00.00	3.61	3.61	00.0	-0.07	-1.01	00.00	0.01	2.53
GAHCP Subtotal	1,117,457	\$26.10	\$27.85	\$28.51	\$1.72	\$0.07	\$1.46	\$0.57	\$0.13	\$29.41
4	011 000 0	1	10.01 \$	01 01 0	00 14	00 04	1014	01 F#	0000	- 10 01 -
l otal	2,903,112	\$43.57	\$42.85	\$43.79	\$5.20	\$0.03	-51.35	\$1.72	\$0.33	\$49.67
CY 2018 CMHC Service Costs for the Granite Advantage - Non-Medically Frail population is set equal to CY 2019 as CY 2018 data was not available.	Medically Frail popu.	lation is set equa	l to CY 2019 as C	:Y 2018 data wa	s not available.					

222

Exhibit F New Hampshire Department of Health and Human Services Medicaid Care Management Program SFY 2023 Enrollment Projections - PHE Ending April 2023

Eligibility Category	Jul-22	Aug-22	Sep-22	Oct-22	Nov-22	Dec-22	Jan-23	Feb-23	Mar-23	Apr-23	May-23	Jun-23	Total
Low Income Children - Age 0-11 Months	3,768	3,711	3,657	3,599	3,578	3,539	3,501	3,462	3,425	3,387	3,387	3,387	42,402
Low Income Children - Age 1-18 Years	67,706	67,691	67,572	67,145	67,137	66,921	66,705	66,490	66,276	66,062	66,062	66,062	801,829
Low Income Adults	17,996	18,030	18,186	18,464	18,260	18,298	18,335	18,372	18,410	18,448	18,448	18,448	219,693
CHIP	20,922	21,164	21,656	21,783	21,876	21,987	22,098	22,210	22,323	22,436	22,436	22,436	263,326
Foster Care / Adoption	2,584	2,643	2,683	2,694	2,742	2,771	2,802	2,832	2,863	2,894	2,894	2,894	33,296
Severely Disabled Children	842	860	849	838	848	848	847	847	846	846	846	846	10,161
Elderly and Disabled Adults - Age 19-64	5,910	5,995	6,038	5,918	6,031	6,027	6,024	6,020	6,017	6,013	6,013	6,013	72,021
Elderly and Disabled Adults - Age 65+	1,336	1,340	1,393	1,343	1,408	1,415	1,423	1,430	1,438	1,445	1,445	1,445	16,861
Dual Eligibles	15,500	15,694	15,996	15,611	16,008	16,014	16,020	16,026	16,032	16,038	16,038	16,038	191,012
Newborn Kick Payment	237	198	260	217	228	228	228	228	228	228	228	228	2,736
Neonatal Abstinence Syndrome Kick Payment	•		7	I	2	2	2	2	7	7	2	7	21
Maternity Kick Payment	212	193	231	171	202	202	202	202	202	202	202	202	2,423
													ı
Severe/Persistent Mental Illness - Non-Dual	2,722	2,899	2,910	3,047	2,894	2,894	2,894	2,894	2,894	2,894	2,894	2,717	34,556
Severe/Persistent Mental Illness - Dual	2,050	1,930	2,215	2,058	2,063	2,063	2,063	2,063	2,063	2,063	2,063	2,063	24,758
Severe Mental Illness - Non-Dual	2,073	2,384	2,218	2,445	2,280	2,280	2,280	2,280	2,280	2,280	2,280	2,035	27,115
Severe Mental Illness - Dual	315	311	378	325	332	332	332	332	332	332	332	332	3,988
Low Utilizer - Non-Dual	399	430	435	467	433	433	433	433	433	433	433	429	5,188
Low Utilizer - Dual	278	274	316	286	289	289	289	289	289	289	289	289	3,463
Serious Emotionally Disturbed Child	6,385	6,241	7,171	6,544	6,585	6,585	6,585	6,585	6,585	6,585	6,585	6,605	79,044
Standard Subtotal	150,786	151,596	153,673	152,566	152,764	152,696	152,630	152,567	152,505	152,446	152,446	152,039	1,828,714
Granite Advantage - Medically Frail	10.011	10.106	026.6	10.246	10.379	10.515	10.652	10.791	10.932	10.932	10.932	9.756	125,223
Granite Advantage - Non-Medically Frail	75,918	76,809	76,862	78,153	78,895	79,644	80,401	81,165	81,935	81,935	81,935	75,523	949,176
GAHCP Subtotal	85,928	86,915	86,832	88,399	89,275	90,159	91,053	91,956	92,867	92,867	92,867	85,279	1,074,399
Total	236,714	238,511	240,505	240,964	242,038	242,855	243,683	244,522	245,373	245,314	245,314	237,319	2,903,112

Exhibit G New Hampshire Department of the Ath and Human Services Medicaid Care Management Program CY 2019 Pharmacy Efficiency Savings

					The	Therapeutic Duplication	U						
					Asthma / COPD					Opioid			
			Medication		Long-Acting	Long-Acting Muscarinic			Concurrent	Opioid Use			
Eligibility Category	Age Limit	Quantity Limit	Posession Ratio	Corticosteroids	Beta Agonists	Antagonists	Autoimmune	ΗIV	Utilization	Disorder	Agonists	Adjusted Total	PMPM
Low Income Children - Age 0-11 Months	•		11			•						Ø	00.0
Low Income Children - Age 1-18 Years	50,209	51,658	5,501	5,679	į	330						90,702	0.12
Low Income Adults	5,773	62,219	9,152	3,935	49	•			2,090	10,290	1,029	76,971	0.58
CHIP	4,889	12,582	161	1,937	•	•	•	•	•	•	•	15,655	60 [.] 0
Foster Care / Adoption	2,217	13,385	143	•	•	•		•	•	-	•	12,596	0.54
Severely Disabled Children	2,729	25,173	299	488		•						23,351	2.17
Elderly and Disabled Adults - Age 19-64	11,298	127,214	13,814	15,427	1,807	1,842		135,595	4,926	7,553	832	257,578	3.50
Elderly and Disabled Adults - Age 65+	•	3,622	2,677	2,742	67	499	4,297	I	449			11,552	0.91
Dual Eligibles	•	2,151	15					•	•		•	1,733	0.01
Newborn Kick Payment	•		•	•		•							
Neonatal Abstinence Syndrome Kick Payment		•		•	•		•				•		
Matemity Kick Payment	•	•		1					i	•	•		
Severe/Persistent Mental Illness - Non-Dual	4,366	72,021	5,231	932	103	506	1	212	30,966	1,829	185	96,379	3.33
Severe/Persistent Mental Illness - Dual	•	3,798	2			•						3,040	0.13
Severe Mental Illness - Non-Dual	2,279	31,227	1,021	3,356		•	8,871	•	1,249	9,459	1,372	48,275	1.65
Severe Mental Illness - Dual	•	37	-	•		•						30	0.01
Low Utilizer - Non-Dual	•	14,794	63		•		•		37		17	11,934	4.67
Low Utilizer - Dual	1		•	•			•		•	•	•		
Serious Emotionally Disturbed Child	7,596	50,639	1,045	365	•							47,716	0.62
Standard Subtotal	91,356	470,520	39,635	34,859	2,056	3,178	13,168	135,807	39,718	29,131	3,434	697,519	0.46
Granita Advantada - Madically Frail	1615		4 275	0 277	975			14 571	1 880	3 257	564	77 978	0 02
Gramice Advantage - Non-Medically Frail	12.551	84.518	2.890	14.524	1.461	3.161		39.676	2.927	2.057	814	132.244	0.34
GAHCP Subtotal	14,166	144,855	7,165	23,801	2,436	3,161		54,247	4,815	5,314	1,378	210,221	0.45
Total	105.521	615.376	46.800	58.660	4.492	6.339	13.168	190.054	44.533	34.445	4.812	907.740	0.45
	-						-	-	-		-	-	

		Hyperammonemia	
Zolgensma	Hemophilia	(Carbaglu & Ravicti)	Zynteglo and Skysona
71894012002	00026037220	52276031205	735543111XX
71894012103	00026037230	52276031260	735542111XX
71894012203	00026037250	75987005006	
71894012303	00026037920	76325010004	
71894012404	00026037930	76325010025	
71894012504	00026037950	71863011460	
71894012604	00026378220		
71894012705	00026378225		
71894012805	00026378330		
71894012905	00026378335		
71894013006	00026378550		
71894013106	00026378555		
71894013206	00026378660		
71894013307	00026378665		
71894013407	00026378770		
71894013507	00026378775		
71894013608	00026379220		
71894013708	00026379330		
71894013808	00026379550		
71894013909	00026379660		
71894014009	00026379770		
71894014109	00026382125		
	00026382225		
	00026382425		
	00026382650		
	00026382850		
	00026394225		
	00026394425		
	00026394625		
	00026394825		
	00026482101		
	00026482201		
	00026482401		
	00026482601		
	00026482801		
	00053623302		
	00053761505		
	00053761510		
	00053761520		
	00053762005		
	00053762010		
	00053762020		
	00053763302		
	00053763402		
	00053765601		
	00053765602		
	00053765604		
	00053765605		
	00053766801		
	00053766802		
	00053766804		
	00053813001		
	00053813002		
	00053813004		
	00053813005		
	00053813102		
	00053813202		
	00053813302		
	00053813402		
	00053813502		
	00169701001		
	00169701301		
	00169702001		

		Hyperammonemia	
Zolgensma	Hemophilia	(Carbaglu & Ravicti)	Zynteglo and Skysona
	00169704001		
	00169705001		
	00169706001		
	00169706101		
	00169706201		
	00169720101		
	00169720201		
	00169720501 00169720801		
	00169720801		
	00169781501		
	00169782001		
	00169782501		
	00169783001		
	00169785001		
	00169790101		
	00169790201		
	00169790501		
	00169810001		
	00169815001		
	00169820001		
	00169830001 00169850001		
	00169850001		
	00944038101		
	00944130210		
	00944130310		
	00944130410		
	00944283110		
	00944283210		
	00944283310		
	00944283401		
	00944283410		
	00944283501		
	00944283510		
	00944284110 00944284210		
	00944284310		
	00944284410		
	00944284510		
	00944292102		
	00944292202		
	00944292302		
	00944292402		
	00944293001		
	00944293101		
	00944293201		
	00944293301 00944293501		
	00944293501		
	00944293502		
	00944293504		
	00944293801		
	00944293802		
	00944293803		
	00944294001		
	00944294002		
	00944294003		
	00944294004		
	00944294010		
	00944294110		
	00944294210 00944294310		
	00944294310		

		Hyperammonemia	
Zolgensma	Hemophilia	(Carbaglu & Ravicti)	Zynteglo and Skysona
	00944294410		
	00944294510		
	00944294610		
	00944294810		
	00944296010		
	00944296110		
	00944296210		
	00944296310		
	00944296410		
	00944296510		
	00944302602		
	00944302802		
	00944303002		
	00944303202		
	00944303402		
	00944304510		
	00944304610		
	00944304710		
	00944305102		
	00944305202		
	00944305302		
	00944305402		
	00944394002		
	00944394202		
	00944394402		
	00944394602		
	00944425202 00944425402		
	00944425602		
	00944425802		
	00944462201		
	00944462301		
	00944462401		
	00944462501		
	00944462601		
	00944462602		
	00944462701		
	00944462702		
	00944462801		
	00944462802		
	00944500101		
	00944500105		
	00944500110		
	00944755102		
	00944755302		
	13533066520		
	13533066530		
	13533066550		
	50242092001 50242092101		
	50242092201 50242092301		
	52769046001		
	532709040001		
	53270027105		
	53270027106		
	53270027205		
	53270027206		
	58394000101		
	58394000105		
	58394000106		
	58394000201		
	58394000205		

		Hyperammonemia	
Zolgensma	Hemophilia	(Carbaglu & Ravicti)	Zynteglo and Skysona
	58394000206		
	58394000301		
	58394000305		
	58394000306		
	58394000502		
	58394000504		
	58394000602		
	58394000604		
	58394000702		
	58394000704		
	58394000802		
	58394000803 58394001102		
	58394001102		
	58394001201		
	58394001201		
	58394001202		
	58394001302		
	58394001401		
	58394001402		
	58394001501		
	58394001502		
	58394001603		
	58394002203		
	58394002303		
	58394002403		
	58394002503		
	58394063303		
	58394063403		
	58394063503		
	58394063603		
	58394063703		
	63833038602		
	63833038702		
	63833051802		
	63833061502		
	63833061602		
	63833061702		
	63833089151		
	63833089190		
	63833891501		
	64193022203		
	64193022204 64193022205		
	64193022302 64193022402		
	64193022402 64193022502		
	64193022502		
	64193042302		
	64193042302		
	64193042502		
	64193042602		
	64193044502		
	64208775201		
	64208775301		
	64406048308		
	64406048408		
	64406048508		
	64406048608		
	64406048708		
	64406048808		

		Hyperammonemia	
Zolgensma	Hemophilia	(Carbaglu & Ravicti)	Zynteglo and Skysona
	64406048908		
	64406080101 64406080201		
	64406080201		
	64406080401		
	64406080501		
	64406080601		
	64406080701		
	64406080801		
	64406080901		
	64406081001		
	64406091101		
	64406092201		
	64406093301		
	64406094401		
	64406096601		
	64406097701		
	67467018101 67467018102		
	67467018102		
	67467018201		
	68516320002		
	68516320003		
	68516320004		
	68516320005		
	68516320101		
	68516320202		
	68516320302		
	68516320401		
	68516320502		
	68516320602		
	68516320701		
	68516320802		
	68516320902 68516360002		
	68516360002		
	68516360005		
	68516360006		
	68516360102		
	68516360202		
	68516360302		
	68516360402		
	68516360502		
	68516360602		
	68516360702		
	68516360802		
	68516360902		
	68516460001		
	68516460002 68516460101		
	68516460201		
	68516460302		
	68516460402		
	68516460501		
	68516460601		
	68516460702		
	68516460802		
	68516460902		
	68516461002		
	68516461101		
	68516461201		
	68516461302		
	68516461402		

		Hyperammonemia	
Zolgensma	Hemophilia	(Carbaglu & Ravicti)	Zynteglo and Skysona
	68516461502		
	68982013901		
	68982014001		
	68982014101		
	68982014201		
	68982014301		
	68982014401		
	68982014501		
	68982014601		
	68982014701		
	68982014801		
	68982014901		
	68982015001		
	68982015101		
	68982015201		
	68982018201 68982018202		
	68982034701		
	68982034801		
	69911047402		
	69911047502		
	69911047602		
	69911047702		
	69911047802		
	69911048002		
	69911048102		
	69911086402		
	69911086502		
	69911086602		
	69911086702		
	69911086902		
	70504028205		
	70504028305		
	70504028405		
	70504028506		
	70504028606		
	70504028705		
	70504028805		
	70504028905		
	71104048308		
	71104048408		
	71104048508 71104048608		
	71104048708		
	71104048808		
	71104048908		
	71104049008		
	71104049108		
	71104049208		
	71104080101		
	71104080201		
	71104080301		
	71104080401		
	71104080501		
	71104080601		
	71104080701		
	71104080801		
	71104080901		
	71104081001		
	71104091101		
	71104092201		
	71104093301		
	71104094401		

		Hyperammonemia	
Zolgensma	Hemophilia	(Carbaglu & Ravicti)	Zynteglo and Skysona
	71104096601		
	71104097701		
	76125025020		
	76125025620		
	76125050030		
	76125066730		
	76125066750		
	76125066830		
	76125067250		
	76125067351		
	76125067650		

New Hampshire Depart Medicaid Ca Impact of PDL Cha	Hampshire Department of Health and Human Services Medicaid Care Management Program Impact of PDL Change - Top 25 Classes - CY 2018	luman Services ram : - CY 2018		
Top 25 Classes	Percentage of Pharmacy Expenditure in Base Period	Total Projected Expenditure Before PDL	Total Projected Expenditure After PDL	Impact of PDL Change
Autoimmune Agents	8.50%	\$13,623,234	\$13,586,841	-0.27%
Hepatitis C Agents - Treatments	5.99%	\$9,602,412	\$9,248,310	-3.69%
Opioid Partial Agonists - buprenorphine/naloxone film	4.22%	\$6,762,446	\$4,665,448	-31.01%
Amphetamine Stimulants	4.13%	\$6,623,697	\$6,620,700	-0.05%
Atypical Antipsychotics	3.61%	\$5,785,418	\$5,639,217	-2.53%
Methylphenidate Stimulants	3.42%	\$5,474,070	\$5,458,374	-0.29%
Growth Hormones	2.90%	\$4,653,422	\$4,686,584	0.71%
Long Acting Beta Agonist and Corticosteroid Combinations	2.89%	\$4,625,611	\$4,627,518	0.04%
Long Acting Injectable Antipsychotics	2.68%	\$4,286,837	\$4,286,837	0.00%
Cystic Fibrosis Agents - CFTR Modulator	2.63%	\$4,218,635	\$4,218,635	%00.0
Steroid Inhalants	2.26%	\$3,615,941	\$3,630,901	0.41%
Insulin - Rapid Acting	2.06%	\$3,303,971	\$3,304,565	0.02%
Antiretrovirals - HIV Combinations	1.85%	\$2,972,536	\$2,972,536	0.00%
Insulin - Long Acting - insulin glargine	1.51%	\$2,417,385	\$2,661,071	10.08%
Multiple Sclerosis Agents	1.43%	\$2,296,185	\$2,277,213	-0.83%
Opioid Antagonists - Naltrexone	1.43%	\$2,291,643	\$2,291,643	0.00%
Anticonvulsants - Misc.	1.33%	\$2,131,621	\$2,029,342	-4.80%
Fibromyalgia Agents - Lyrica	1.27%	\$2,031,878	\$496,535	-75.56%
Cystic Fibrosis Agents - Mucolytic Agents	1.23%	\$1,966,533	\$1,966,533	%00.0
GABA Modulators	1.12%	\$1,800,047	\$1,800,047	0.00%
Diagnostic Tests	1.10%	\$1,760,954	\$1,760,954	0.00%
Anticonvulsants - Misc Banzel/Vimpat	1.06%	\$1,700,929	\$1,700,929	%00.0
Anticonvulsants - Benzodiazepines - Onfi	1.06%	\$1,700,336	\$1,595,958	-6.14%
Pancreatic Enzymes	1.01%	\$1,624,186	\$1,614,366	- 0.60%
Antineoplastic Enzyme Inhibitor - Breast/Ovarian	1.01%	\$1,623,954	\$1,623,954	0.00%
Total	61.71%	\$98,893,879	\$94,765,008	-4_18%

Exhibit J1a

	Exhibit J1b			
New Hampshire Department of Health and Human Services	nent of Health and H	luman Services		
Medicaid Care Management Program Impact of PDL Change - Top 25 Classes - CY 2019	Medicaid Care Management Program t of PDL Change - Top 25 Classes - C	јгат s - СҮ 2019		
	Percentage of			
	Pharmacy Expenditure in	Expenditure	Total Projected Expenditure After	Impact of PDL
Autoimmune Agents		\$26.081.966	\$25 931 741	-0.58%
Hepatitis C Agents - Treatments	4.93%	\$10,738,709	\$10,576,736	-1.51%
Opioid Partial Agonists - buprenorphine/naloxone film	4.19%	\$9,127,238	\$5,195,807	-43.07%
Cystic Fibrosis Agents - CFTR Modulator	3.71%	\$8,085,963	\$8,085,963	%00.0
Amphetamine Stimulants	3.49%	\$7,614,536	\$7,591,075	-0.31%
Atypical Antipsychotics	3.23%	\$7,049,628	\$6,848,086	-2.86%
Antiretrovirals - HIV Combinations	2.72%	\$5,923,259	\$5,923,259	%00.0
Long Acting Beta Agonist and Corticosteroid Combinations	2.50%	\$5,441,239	\$5,436,281	%60 [.] 0-
Methylphenidate Stimulants	2.44%	\$5,310,032	\$5,290,537	-0.37%
Long Acting Injectable Antipsychotics	2.38%	\$5,177,804	\$5,177,804	00.00%
Growth Hormones	2.09%	\$4,556,444	\$4,573,697	0.38%
Insulin - Rapid Acting	1.83%	\$3,993,866	\$3,991,509	-0.06%
Steroid Inhalants	1.79%	\$3,898,631	\$3,923,173	0.63%
Multiple Sclerosis Agents	1.75%	\$3,823,366	\$3,685,112	-3.62%
Opioid Antagonists - Naltrexone	1.72%	\$3,740,126	\$3,740,126	%00.0
Anticonvulsants - Misc.	1.56%	\$3,398,571	\$3,285,143	-3.34%
Incretin Mimetic Agents (GLP-1 Receptor Agonists)	1.53%	\$3,339,911	\$3,822,778	14.46%
Insulin - Long Acting - insulin glargine	1.52%	\$3,308,725	\$3,342,366	1.02%
Anticoagulants	1.21%	\$2,632,904	\$2,632,957	%00.0
Opioid Partial Agonists	1.06%	\$2,309,943	\$2,050,784	-11.22%
Cystic Fibrosis Agents - Mucolytic Agents	1.02%	\$2,218,462	\$2,218,462	0.00%
Antineoplastic Enzyme Inhibitor - Breast/Ovarian	0.97%	\$2,107,980	\$2,107,980	0.00%
Pancreatic Enzymes	0.95%	\$2,077,907	\$2,059,888	-0.87%
Anticonvulsants - Misc Banzel/Vimpat	0.93%	\$2,021,030	\$2,021,030	0.00%
Diagnostic Tests	0.92%	\$2,000,833	\$2,000,833	0.00%
Total	62 <u>.</u> 39%	\$135,979,073	\$131,513,131	-3_28%

Exhibit J2a New Hampshire Department of Health and Human Services Medicaid Care Management Program Impact of PDL Change - Drug Classes with Largest Impact - CY 2018

	Percentage of Pharmacy Expenditure in	Total Projected Expenditure	Total Projected Expenditure After Impact of PDL	Impact of PDL
Drug Classes with Largest Impact	Base Period	Before PDL	PDL	Change
Fibromyalgia Agents - Lyrica	1.27%	\$2,031,878	\$496,535	-75.56%
Antianginal Agents	0.02%	\$31,598	\$8,157	-74.19%
Antifungals - Nail Topicals	0.00%	\$982	\$303	-69.13%
Ophthalmics - Carbonic Anhydrase Inhibitors	0.01%	\$20,636	\$6,973	-66.21%
Intestinal Cholesterol Absorption Inhibitors	0.01%	\$17,392	\$9,906	-43.05%
ACE Inhibitors	0.06%	\$93,371	\$53,716	-42.47%
Irritable Bowel Syndrome (IBS) Agents - Diarrhea	0.03%	\$55,293	\$77,980	41.03%
Beta Blockers Non-Selective	0.14%	\$228,652	\$135,759	-40.63%
Extended Release Opioid Agonists	0.60%	\$959,787	\$570,147	-40.60%
Ophthalmics - Misc.	0.01%	\$10,556	\$6,297	-40.35%
Benzodiazepines	0.09%	\$150,369	\$96,408	-35.89%
Androgens	0.12%	\$188,036	\$124,556	-33.76%
Antiparkinson Agents	0.04%	\$60,805	\$40,466	-33.45%
Potassium Removing Resins	0.01%	\$15,472	\$10,390	-32.85%
Opioid Partial Agonists - buprenorphine/naloxone film	4.22%	\$6,762,446	\$4,665,448	-31.01%
Proton Pump Inhibitors	0.43%	\$681,554	\$481,849	-29.30%
Prostaglandins - Ophthalmic	0.03%	\$54,835	\$39,071	-28.75%
Angiotensin II Receptor Antagonists	0.03%	\$41,669	\$29,903	-28.24%
Short Acting Beta Agonists - Ventolin	0.96%	\$1,541,699	\$1,125,288	-27.01%
Multiple Sclerosis Agents - dimethyl fumarate	0.52%	\$832,910	\$613,050	-26.40%
Glucocorticosteroids - Uceris	0.03%	\$45,388	\$33,621	-25.93%
Scabicides & Pediculicides	0.22%	\$347,626	\$265,391	-23.66%
Granulocyte Colony-Stimulating Factors	0.26%	\$418,906	\$324,307	-22.58%
Peripheral Opioid Receptor Antagonists	0.07%	\$120,105	\$95,840	-20.20%
Antihyperlipidemics - Misc.	0.01%	\$14,310	\$11,615	-18.83%
Total	9.19%	\$14,726,275	\$9,322,976	-36.69%

Exhibit J2b New Hampshire Department of Health and Human Services Medicaid Care Management Program

Impact of PDL Change - Drug Classes with Largest Impact - CY 2019

	Percentage of Pharmacy Expenditure in	Total Projected Expenditure	Total Projected Expenditure After	Impact of PDL
Drug Classes with Largest Impact	Base Period	Before PDL	PDL	Change
Antifungals - Nail Topicals	0.00%	\$6,912	\$1,799	-73.97%
Potassium Removing Resins	0.01%	\$16,304	\$5,956	-63.47%
Extended Release Opioid Agonists	0.52%	\$1,141,553	\$421,029	-63.12%
Ophthalmics - Carbonic Anhydrase Inhibitors	0.01%	\$19,661	\$7,338	-62.68%
Fibromyalgia Agents - Lyrica	0.82%	\$1,792,824	\$777,448	-56.64%
Antihyperlipidemics - Misc.	0.02%	\$37,639	\$21,208	-43.66%
Opioid Partial Agonists - buprenorphine/naloxone film	4.19%	\$9,127,238	\$5,195,807	-43.07%
Ophthalmics - Misc.	0.01%	\$15,490	\$8,955	-42.19%
Multiple Sclerosis Agents - dimethyl fumarate	0.61%	\$1,325,154	\$851,056	-35.78%
Prostaglandins - Ophthalmic	0.04%	\$87,420	\$58,000	-33.65%
Proton Pump Inhibitors	0.46%	\$1,000,062	\$740,557	-25.95%
Beta Blockers Non-Selective	0.12%	\$256,764	\$192,206	-25.14%
Peripheral Opioid Receptor Antagonists	0.05%	\$116,861	\$92,283	-21.03%
Benzodiazepines	%60.0	\$205,076	\$163,977	-20.04%
Intestinal Cholesterol Absorption Inhibitors	0.01%	\$21,117	\$17,247	-18.33%
ACE Inhibitors	0.09%	\$197,461	\$162,625	-17.64%
Androgens	0.09%	\$204,144	\$169,494	-16.97%
Antiparkinson Agents	0.03%	\$75,614	\$63,515	-16.00%
Antianginal Agents	0.01%	\$18,438	\$15,494	-15.96%
Insulin - Regular/NPH	1.53%	\$3,339,911	\$3,822,778	14.46%
Irritable Bowel Syndrome (IBS) Agents - Diarrhea	0.77%	\$1,669,735	\$1,911,092	14.45%
Insulin - Long Acting - insulin glargine	0.02%	\$40,275	\$45,927	14.03%
Alpha-Beta Blockers / Beta Blockers Cardio-Selective	0.14%	\$297,604	\$263,669	-11.40%
Migraine Serotonin Agonists	0.12%	\$270,957	\$240,324	-11.31%
Opioid Partial Agonists	1.06%	\$2,309,943	\$2,050,784	-11.22%
Total	10.83%	\$23,594,156	\$17,300,567	-26.67%